# **Distribution Agreement**

In presenting this thesis or dissertation as a partial fulfillment of the requirements for an advanced degree from Emory University, I hereby grant to Emory University and its agents the non-exclusive license to archive, make accessible, and display my thesis or dissertation in whole or in part in all forms of media, now or hereafter known, including display on the world wide web. I understand that I may select some access restrictions as part of the online submission of this thesis or dissertation. I retain all ownership rights to the copyright of the thesis or dissertation. I also retain the right to use in future works (such as articles or books) all or part of this thesis or dissertation.

Signature:

Sherod Thaxton

Date

The Social Geometry of Death: Social Structure and Capital Punishment in Georgia, 1993 – 2000

By

Sherod Thaxton Doctor of Philosophy

Sociology

Robert Agnew Advisor

Stephen B. Bright Committee Member Yuk Fai Cheong Committee Member

Cathryn Johnson Committee Member Jeffery Mullis Committee Member

Accepted:

Lisa A. Tedesco, Ph.D. Dean of the Graduate School

Date

The Social Geometry of Death: Social Structure and Capital Punishment in Georgia, 1993 – 2000

By

Sherod Thaxton B.A., University of California, Davis, 1997 M.A., Emory University, 2000 J.D., University of Chicago, 2008

Advisor: Robert Agnew, Ph.D.

An abstract of A dissertation submitted to the Faculty of the Graduate School of Emory University in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Sociology 2009

### Abstract

# The Social Geometry of Death: Social Structure and Capital Punishment in Georgia, 1993 – 2000

## By

### Sherod Thaxton

In 1976 the U.S. Supreme Court lifted its four year moratorium on executions after deciding the modified death penalty statutes introduced by Florida, Georgia, and Texas contained constitutionally acceptable safeguards to significantly reduce the rampant arbitrariness, capriciousness, and bias that previously plagued the capital punishment process in America. Nonetheless, empirical studies at both the state and federal levels conducted since the death penalty was reinstated have consistently discovered that the administration of capital punishment remains both highly inconsistent and discriminatory. Not only do researchers continue to find that the death penalty is not exclusively reserved for the most heinous murder cases, but also that race/ethnicity, gender, and geography significantly influence the likelihood that a defendant is charged with a capital offense and receives the death penalty. These studies have resulted in considerable debate among legal scholars as to why three decades of procedural reforms to the capital punishment process have failed to satisfactorily reduce the very problems that initially led the Court to invalidate existing death penalty statutes. In a departure from most death penalty analysts, this project argues that previous attempts to reform the capital punishment system have been largely unsuccessful because they do not adequately identify and explain the complex ways in which legally legitimate and illegitimate factors impact legal decision-making. Drawing heavily from the socio-legal and social control literatures, this study advances a theory of legal behavior operating solely at the sociological level to explain capital charging-and-sentencing patterns and tests hypotheses derived from the theory using recent death penalty data from Georgia.

The Social Geometry of Death: Social Structure and Capital Punishment in Georgia, 1993 – 2000

By

Sherod Thaxton B.A., University of California, Davis, 1997 M.A., Emory University, 2000 J.D., University of Chicago, 2008

Advisor: Robert Agnew, Ph.D.

A dissertation submitted to the Faculty of the Graduate School of Emory University in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Sociology 2009

### Acknowledgments

Writing this dissertation was quite a journey, and I want to acknowledge the good people along the way who helped me get here.

I have to begin with Bob Agnew, my mentor and dissertation advisor, who played an immeasurable role in my graduate education. I realized he would be a great mentor after our very first conversation during my prospective student visit. Bob was instrumental in my choice to pursue my doctoral studies prior to attending law school and he has assisted me with every academic endeavor since. Educators like Bob make the academy a noble endeavor, change lives and inspire others to follow suit.

I also tremendously benefited from the other patient and selfless members of my committee. Steve Bright's involvement in this project was a "big deal." Listening to him lecture or observing him in court always left me reenergized and recommitted to staying on this path. The fact that I learned so much from Fai Cheong in such a short time is a testament to his pedagogical prowess. His assistance not only helped facilitate the specific statistical analyses employed in my dissertation, but my overall understanding of, and appreciation for quantitative social science as well. Cathy Johnson's support and encouragement proved invaluable as the time and the dissertation got longer and longer. Finally, I was truly fortunately to have Jeff Mullis join the faculty around the time I discovered Donald Black's work. Our numerous discussions were invaluable to the development and refinement of my theoretical arguments.

Maggie Stephens, Cathy David, and Daphne Outlaw looked out for me the entire time I was at Emory and even after I left for law school. I know everyone misses Cathy because she taught us all the ropes.

While a law student at the University of Chicago, Bernard Harcourt, Tracey Meares, and Tom Miles read portions of my dissertation and offered important feedback. A special thanks to Professor Harcourt, who as a death penalty defense lawyer and a social scientist, really helped me think about both the scholarly and practical aspects of my work.

Michael Mears and Pamela Blume Leonard gave me my introduction to the Georgia capital defense community by hiring me at the Office of the Multi-County Public Defender in 1998. That job permitted my direct involvement with issues of caprice and bias in the death penalty, ultimately leading me to write my dissertation on Georgia's capital punishment process.

First people, then organizations. I would especially like to thank the A. Worley Brown Endowment for Southern Studies and the Southern Studies Dissertation Workshop (headed by the late Elizabeth Fox Genovese), the Open Society Institute, the American Bar Foundation (especially Bob Nelson and Susan Shapiro), and the John M. Olin Program in Law and Economics at the University of Chicago Law School for providing both financial support and exposure to exciting intellectual and activist communities.

Finally, I thank my parents, siblings, extended family, and friends who supported me throughout the entire process—particularly the Atlanta and Chicago Wilborns who opened their hearts and their homes to me. I am forever grateful for the opportunity to have a home away from home with my family.

During the final year of writing my dissertation, my father, Maxie Sherod Thaxton, and my stepfather, Frank Eugene Johnson, passed away. Not a day goes by that I don't recall and benefit from the wisdom of these two extraordinary men. I am just beginning to truly understand and appreciate how important they were in making me the person I am.

# **Table of Contents**

List of Tables		
List of Figure	S	
Chapter 1: Int	roduction	1
Chapter 2: Ba	ckground on Capital Punishment in the United States	
2.1 Ear	ly Colonial Period Through the Civil War	14
2.2 Pos	t-Civil War Period Through the 1960s	
2.3 The	1970s to the Present	32
Chapter 3: So	cial Science and the Debate Over Capital Punishment	
3.1 Ma	or Issues Surrounding the Death Penalty	
3.1.1	Deterrence	71
3.1.2	Incapacitation	77
3.1.3	Cost/Expense	
3.1.4	Wrongful Convictions/Error Rates	
3.1.5	Retribution	
Chapter 4: Ex	tra-Legal Factors and the Death Penalty	
4.1 Bac	kground on Race and the Law in the United States	
4.1.1	Early Colonial Period through the Civil War	
4.1.2	Post Civil War Period through the 1960s	
4.1.3	The 1970s to the Present	101
4.2 Res	earch on Extra-Legal Factors and Capital Punishment	108
4.2.1	Pre-Furman Empirical Studies	108
4.2.2	Post-Furman Empirical Studies	114
4.2.3	Race and the Federal and Military Death Penalty	126
Chapter 5: Ge	orgia and the Death Penalty	136
5.1 Exe	cutions in the South, 1608 – 2008	136
5.1.1	Executions in Georgia, 1735 – 2008	137
5.1.2	Race and Executions in Georgia, 1774 – 2008	139
5.2 Geo	orgia and the Modern Death Penalty	142
5.2.1	Landmark Georgia Death Penalty Cases	142
5.2.2	Race/Ethnicity, Gender, Region, and Georgia's Death Penalty	155
5.2.3	Miscarriages of Justice: Innocence and Prejudicial Error	158
5.2.4	Life Without the Possibility of Parole	161
5.3 Sur	nmary	165
Chapter 6: Pr	esentation and Evaluation of Theoretical Perspectives	
6.1 Doi	ninant Theoretical Perspectives	
6.1.1	Scientific Criteria of Evaluation	
6.1.2	Description of Dominant Perspectives	
6.1.3	Evaluation of Dominant Perspectives	
6.2 The	Blackian Paradigm	208
6.2.1	Assumptions of the Blackian Paradigm	208
6.2.2	Description of Black's Theory of Law	
6.2.3	Evaluation of the Blackian Paradigm	
6.2.4	The Conceptualization of Race in the Blackian Paradigm	
6.2.5	Evaluation of Black's Theory of Law	

6.3 Res	earch Hypotheses	
6.3.1	Hypothesis One	
6.3.2	Hypothesis Two	290
6.3.3	Hypothesis Three	
6.3.4	Hypothesis Four	292
6.3.5	Hypothesis Five	293
6.3.6	Hypothesis Six	
6.3.7	Hypothesis Seven	
6.3.8	Hypothesis Eight	
6.3.9	Summary	
6.4 A C	omment on Verification, Falsification, and Corroboration	301
Chapter 7: Da	ta and Analytical Approach	
7.1 Data	Collection	
7.1.1	Years Considered	
7.1.2	Decision Stages Examined.	
7.1.3	Data Sources	
7.2 Ana	lytical Approach	
7.2.1	Culpability Measures	
7.2.2	Multilevel Regression Models	
7.2.3	Missing Data and Selection Bias Issues	
7.3 Sum	imary	
Chapter 8: Rea	sults	
8.1 Prel	iminary Matters	
8.1.1	Multiple Imputation	
8.1.2	Statistical Inference and Parameter Interpretation	
8.1.3	Death Eligibility	
8.2 Mod	lel Specifications	
8.3 Ass	essment of Hypotheses	
8.3.1	Hypothesis One	
8.3.2	Hypothesis Two	
8.3.3	Hypothesis Three	
8.3.4	Hypothesis Four	
8.3.5	Hypothesis Five	
8.3.6	Hypothesis Six	
8.3.7	Hypothesis Seven	
8.3.8	Hypothesis Eight	
8.4 Spec	cification Checks	
8.5 Sum	imary	
Chapter 9: Sur	nmary and Discussion	
9.1 Con	clusion	408
References		412
Cases		504
Appendix A: I	Progression of Georgia Death Penalty Case	531
Appendix B: Variable Definitions		532
Appendix C: I	Random Intercept Logistic Regression Models (Null & Partial)	535
Appendix D: Supplemental Analyses (Partial & Plea Bargaining Models)		536

Appendix E: Supplemental Analyses (Pooled Estimates)	
--	--

# List of Tables

Table 1: Death Notices in Georgia by County (1993–2000)	509
Table 2: Death Notices in Georgia by Judicial Circuit (1993-2000)	510
Table 3: Summary Statistics for Case-Level Variables	511
Table 4: Summary Statistics for Judicial Circuit-Level Variables	512
Table 5: Summary Statistics for Murder Conviction Subsample	513
Table 6: Summary Statistics for Death Eligible Subsample	514
Table 7: Summary Statistics for Death Noticed Subsample	515
Table 8: Random Intercept Logistic Regression Models (H1-H3)	516
Table 9: Random Intercept Logistic Regression Models (H <sub>4</sub> )	517
Table 10: Random Intercept Logistic Regression Models (H <sub>5</sub> )	519
Table 11: Random Intercept Logistic Regression Models (H <sub>6</sub> -H <sub>7</sub> )	521
Table 12: Cross-Level Interactions (H <sub>8</sub> )	523
Table 13: Multivariate Logit Selection Models	525
Table 14: Summary of Results for Hypotheses	527

# List of Figures

Figure 1: Capital Cases Resulting in LWOP	528
Figure 2: Capital Cases Resulting in Straight Life Sentences	528
Figure 3: Nonlinear Effect of Victim's Age	529
Figure 4: Nonlinear Effect of Defendant's Age	529
Figure 5: Nonlinear Effect of Defendant/Victim Age Difference	530
Figure 6: Nonlinear Effect of Circuit-Level Horizontal Status	530

### **Chapter 1: Introduction**

Capital punishment is at the center of much public debate in the United States and recent evidence suggests that the nation is evenly split over the appropriateness of the "ultimate penalty." When given the option of sentencing convicted murderers to life in prison without the possibility of parole, a recent Gallup Poll discovered that only 55 percent of respondents supported capital punishment (Jones 2005). A similar survey conducted by CBS News one month prior to the aforementioned Gallup Poll revealed that 39 percent of Americans supported the death penalty, 39 percent supported life without the possibility of parole, and six percent supported life imprisonment with the possibility of parole (CBS News Polls 2005).<sup>1</sup> Current support for the death penalty is the lowest in over 30 years, having peaked at nearly 75 percent in the mid-1990s (see Baumer, Messner, and Rosenfeld 2003; Ellsworth and Gross 1994).

The decline in support for capital punishment has been primarily attributed to the growing worldwide abolitionist movement *and* the American public's increasing concern over the risk of executing an innocent person (Bright 2001; Dieter 2003; Unnever and Cullen 2005). International opposition to capital punishment first took center stage in 1966 when the United Nations (UN) General Assembly adopted the International Covenant on Civil and Political Rights and advocated restricting the use of the death penalty.<sup>2</sup> Five years later, with strong encouragement from Amnesty International, the UN adopted another resolution calling for the progressive restriction of capital offenses

<sup>&</sup>lt;sup>1</sup> Several academic studies published in peer-reviewed journals have also discovered that support for the death penalty declines considerably when life without parole is offered as an alternative sentencing option (see, e.g., Bowers 1993; McGarrell and Sandys 1996; Moon *et al.* 2000; Sandys and McGarrell 1995). <sup>2</sup> G.A. Res. 2200A (XXI), U.N. GAOR, 21st Sess., Supp. No. 16, U.N. Doc. A/6316 (December 16, 1966).

and the eventual abolition of the death penalty.<sup>3</sup> When Amnesty International convened a global conference on capital punishment in Stockholm, Sweden in 1977, sixteen nations had abolished capital punishment for all crimes. The goal of progressively restricting the use of the death penalty and the desirability of abolishing capital punishment was reiterated by the UN in 1977 and 1989.<sup>4</sup> Each year since 1997, the UN Commission on Human Rights has passed resolutions calling on countries that have not abolished the death penalty to establish a moratorium on executions.<sup>5</sup> In 1999, the UN began urging countries not to extradite individuals to nations without the assurance that these individuals would not be subject to the death penalty.<sup>6</sup> The UN also requested death penalty nations to make information regarding the imposition of the death penalty and the scheduling of executions available to the public. The United States remains the only Western industrialized nation to retain the death penalty for ordinary crimes (i.e., for crimes other than treason and violations of military law) and the vast majority of executions occur in just a few nations: China, Iran, Saudi Arabia, the United States, and Vietnam (Hood 2003).

With respect to the risk of executing the innocent, 118 inmates were released from death row between 1973 and 2004 because of evidence of their innocence, compared to

 <sup>&</sup>lt;sup>3</sup> G.A. Res. 2857 (XXVI), U.N. GAOR, 26th Sess., Supp. No. 29, U.N. Doc. A/8429 (December 20, 1971).
 <sup>4</sup> G.A. Res. 32/61, U.N. GAOR, 32d Sess., Supp. No. 45, U.N. Doc. A/32/45 (December 8, 1977); G.A. Res. 44/128, U.N. GAOR, 44th Sess., Supp. No. 49, Annex., U.N. Doc. A/44/49 (December 15, 1989).
 <sup>5</sup> C.H.R. Res. 1997/12, 53d Sess., U.N.Doc. E/CN.4/RES/1997/12 (April 3, 1997); C.H.R. Res. 1998/8, 54th Sess., U.N. Doc. E/CN.4/RES/1998/8 (April 3, 1998); C.H.R. Res. 1999/61, 55th Sess., U.N. Doc. E/CN.4/RES/1998/8 (April 3, 1998); C.H.R. Res. 1999/61, 55th Sess., U.N. Doc. E/CN.4/RES/1998/8 (April 2, 2000/65, 56th Sess., U.N. Doc. E/CN.4/RES/2000/65 (April 26, 2000); C.H.R. Res. 2001/68, 57th Sess., U.N. Doc. E/CN.4/RES/2001/68 (April 25, 2001); C.H.R. Res. 2002/77, 58th Sess., U.N. Doc. E/CN.4/RES/2002/77 (April 25, 2002); C.H.R. Res. 2003/67, 59th Sess., U.N. Doc. E/CN.4/RES/2003/67 (April 24, 2003); C.H.R. Res. 2004/67, 60th Sess., UN Doc. E/CN.4/RES/2004/67 (April 21, 2004); C.H.R. Res. 2005/59 61st Sess., UN Doc. E/CN.4/2005/59 (April 20, 2005).

<sup>&</sup>lt;sup>6</sup> C.H.R. Res. 1999/61. In 2001, the Supreme Court of Canada refused to extradite two defendants to the United States without assurances that the death penalty would not be imposed, noting significant problems with the American capital punishment system (see *United States of America v. Burns*, [2001] S.C.R. 283, 360).

944 executed inmates during the same time period—a ratio of one exoneration for every eight executions (Death Penalty Information Center 2008). Particularly troubling to death penalty opponents is the fact that the number of annual exonerations has dramatically increased over the past several years, suggesting the risk of executing the innocent may be as great as ever. From 1973 to 1988—the first half of the modern death penalty era—there were an average of 2.2 exonerations per year. During the second half of this period, 1989 to 2004, the average number of annual exonerations increased to slightly over five. Between 2000 and 2004, inclusive, the average number exonerations climbed to 7.2 per year (Death Penalty Information Center 2008).<sup>7</sup> Several analysts have noted that these figures underestimate the prevalence of wrongful death sentences because they only capture the individuals who are able to show evidence of their innocence while on death row. Excluded from these statistics are individuals who did not receive a full acquittal, but were nevertheless released from death row because of probable or possible innocence (Death Penalty Information Center 2008). Also excluded from these figures are individuals who were sentenced to death, but should have been convicted of second-degree murder or manslaughter and, therefore, ineligible for the death penalty in the first place. A recent study of the all death sentences handed down between 1973 and 1995 revealed that 68 percent of these sentences were eventually overturned because of serious error and 82 percent of those cases resulted in a sentence other than death upon retrial (see Chapter Three).<sup>8</sup>

<sup>&</sup>lt;sup>7</sup> The rate of exonerations for non-capital crimes has risen sharply during this time period as well (see Gross *et al.* 2005).

<sup>&</sup>lt;sup>8</sup> Seven percent of these cases resulted in the defendant ultimately being found innocent of the capital crime for which she or he was originally sentenced (Liebman *et al.* 2000c).

Illinois Governor George Ryan's growing concern over the possible execution of the wrongfully convicted compelled him to place a moratorium on executions in January 2000.<sup>9</sup> Governor Ryan noted that 12 individuals had been executed in Illinois since 1977, while 13 had been exonerated during the same time period. Two years later, in May 2002, outgoing Maryland Governor Paris Glendening followed Governor Ryan's lead and imposed a moratorium on executions in Maryland until the completion of a study he commissioned to examine potential arbitrariness and discrimination in the administration of Maryland's death penalty.<sup>10</sup> The study—conducted by researchers at the University of Maryland—uncovered substantial racial and geographic disparities in the state's capital punishment system. After taking into account numerous legally legitimate and illegitimate case characteristics, the study revealed that offenders accused of murdering

<sup>&</sup>lt;sup>9</sup> After placing the moratorium on the death penalty, Governor Ryan—a proponent of the death penalty appointed a special commission to study how the death penalty system in Illinois could be reformed, and not to debate whether or not the death penalty should be abolished. Ryan's executive order stated: "The Commission, upon concluding its examination and analysis of the capital punishment process, shall submit to the Governor a written report detailing its findings and providing comprehensive advice and recommendations to the Governor that will further ensure the administration of capital punishment in the State of Illinois will be fair and accurate" (see Exec. Order No. 424 Ill. Reg. 7439, March 4, 2000). In April 2002, the Governor's Commission—comprised of members across the political spectrum (see Sanger 2003, p. 103 n.12)—issued its report of eighty-five specific recommendations for corrections to the Illinois death penalty system, supported by over 200 pages of analyses and appended materials. The commission was unanimous in its belief that no system, given human nature and frailties, could ever be devised or constructed that would work perfectly and guarantee that no innocent person is ever again sentenced to death.

<sup>&</sup>lt;sup>10</sup> In January 2006, outgoing New Jersey Governor Richard J. Codey signed legislation requiring a temporary suspension of all executions in his State and creating a study commission to evaluate New Jersey's capital punishment system. The bill passed in the New Jersey General Assembly (H.R. 2347 & S. 709, 211th Leg., Reg. Sess. [N.J. 2004]), and called for the establishment of a thirteen member bipartisan commission to examine critical issues such as racial and regional disparities, costs, risk of erroneous executions, and whether alternatives to the death penalty are feasible. Governor Codey's signing of the bill marked the first time in the nation's history that a State imposed a moratorium on executions via legislation (Post 2006). One week later, the California Assembly Appropriations Committee sidelined a similar bill (H.R. 1121, 2005 Reg. Sess., [Cal. 2005]) that would have imposed a 24-month moratorium on executions so that the Legislature could review the recommendations of the blue-ribbon panel study evaluating the effectiveness, accuracy, and fairness of the State's capital punishment system (Marshall 2006). In December 2007, New Jersey lawmakers voted to abolish the death penalty (H.R. 3716 & S. 171, 212th Leg., Reg. Sess. [N.J. 2006]), and Governor Jon Corzine subsequently signed the bill into law and commuted the death sentences of eight death row inmates to life without the possibility of parole (Peters 2007).

white victims were significantly more likely to be noticed for the death penalty and receive a death sentence at trial. The study also discovered that black offenders accused of murdering white victims had the greatest likelihood of prosecutors seeking the death penalty against them and juries sentencing them to death (Paternoster *et al.* 2004). At the time the study was released, all of Maryland's 13 death row inmates were convicted of murdering white victims, although more than three-quarters of all homicide victims in the state were non-white (Paternoster and Brame 2003). Despite these findings, newly elected Maryland Governor Robert Ehrlich lifted the moratorium upon taking office as he had promised to do during his campaign.<sup>11</sup> Race may have also played an important role in the administration of capital punishment in Illinois. Of the 18 death row inmates in Illinois who have been exonerated since 1977, fourteen (78 percent) were racial/ethnic minorities (12 blacks and 2 Hispanics).<sup>12</sup> Partly as a result of these dramatic racial disparities, Governor Ryan commuted the death sentences of all 167 inmates sentenced to death in Illinois to life without the possibility of parole two days before he left office.<sup>13</sup> At a speech delivered at Northwestern University, Governor Ryan remarked, "Our capital system is haunted by the demon of error: error in determining guilt and error in determining who among the guilty deserves to die. What effect was race having? What

<sup>&</sup>lt;sup>11</sup> Governor Ehrlich's successor, Martin O'Malley, called for a repeal of the death penalty during this third State of the State address in January 2009. Governor O'Malley, a vocal opponent of capital punishment, personally sponsored a bill to abolish the death penalty in Maryland. He justified his opposition to the death penalty on the grounds that it does not deter potential murderers, wastes resources that could better spent fighting crime, and leaves the state open to the possibility of executing the innocent (Wagner 2009).
<sup>12</sup> Nine of the twelve (75 percent) individuals executed in Illinois since the state's death penalty was reinstated in 1977 were convicted of murdering white victims, although the vast majority of homicide victims in Illinois during this same period were black (Death Penalty Information Center 2008; Fox 2005).
<sup>13</sup> There were actually 156 inmates on death row at the time of Governor Ryan's blanket commutation. Eleven additional inmates had recently been sentenced to death, but were not physically on death row at the time of Governor Ryan's commutation (Possley and Mills 2003).

effect was poverty having? Because of all these reasons, today I am commuting the sentences of all death row inmates" (Possley and Mills 2003).<sup>14</sup>

While the arbitrary, capricious, discriminatory, and error prone administration of the death penalty in Maryland and Illinois may have come as a shock to many Americans, death penalty scholars and abolitionists have highlighted significant problems with the capital charging-and-sentencing process for over six decades. In fact, the capital defense bar began collecting and using data on the racially discriminatory application of the death penalty for rape cases in the early 1950s to challenge the constitutionality of capital punishment (see Chapter Two). Since that time, a large research literature has accumulated documenting the influence of race, gender,<sup>15</sup> region, and other legally illegitimate factors on the death penalty system across the nation. Despite the substantial evidence demonstrating the serious flaws in the administration of death penalty at every stage of the capital charging-and-sentencing process and at all levels of government, thirty-five states, the U.S. Government, and the U.S. Armed Forces still retain the death penalty. Even more troubling for death penalty abolitionists is the fact that more than three decades of procedural modifications in the modern era of the death penalty have been ineffective at eliminating the rampant arbitrariness, caprice, and discrimination that plagued previous eras (Liebman 2007; Ogletree 2002).

Why, despite the efforts of the state legislatures and courts to develop procedural rules to make the imposition of the death penalty fair and consistent, is the capital punishment system still fraught with arbitrariness, caprice, discrimination, and mistake?

<sup>&</sup>lt;sup>14</sup> Blacks comprise 42 percent of inmates currently on death row in the United States, but account for half (50.4 percent) of all exonerations since 1973 (Death Penalty Information Center 2008).

<sup>&</sup>lt;sup>15</sup> Throughout the text, the term "gender" refers to "biological sex"—an ascribed characteristic—and not sexual identity, and achieved characteristic. This distinction is important because the focus of this project is on external, observable aspects of social structure, not largely unobservable internal states.

Possible answers to this question have been heavily debated. At one extreme are the pessimistic appraisals of U.S. Supreme Court Justices John Harlan and Antonin Scalia in the landmark *McGautha* and *McCleskey* cases, respectively. Both Justices concluded that unconscious biases—including racial bias—in prosecutorial, judicial, and jury discretion in capital cases are real, inescapable, and ineradicable (see Chapter Two).<sup>16</sup> Chief Justice William Rehnquist even doubted the ability of state supreme courts to conduct adequate comparative proportionality review of death sentences handed down by the trial courts to detect and eliminate (or significantly reduce) arbitrariness and bias (cf. Kaufman-Osborn 2004).<sup>17</sup> The inevitability of racial discrimination in the death penalty process and the impossibility of its prevention, however, have been seriously challenged by both death penalty abolitionists and retentionists, members of federal and state legislatures, and legal scholars (see Chapters 4 and 5). In fact, several studies suggest that the levels of arbitrariness and discrimination in the administration of the death penalty were moderately reduced, but not eliminated, because of the capital-sentencing reforms mandated by the U.S. Supreme Court in the early- and mid-1970s (see Baldus, Woodworth, and Pulaski 1990, Chapter 5). At the opposite extreme is the optimistic

<sup>&</sup>lt;sup>16</sup> See *McGautha v. California*, 402 U.S. 183, 204 (1971) and *McCleskey v. Kemp*, 481 U.S. 279 (1987). Justice Harlan believed that there was no practical way to guide a jury's ultimate decision whether to impose the death sentence. Justice Scalia, although acknowledging the clear patterns of bias in the capital punishment system, believed that the "unconscious" sympathies and antipathies of jurors were an irreversible evil of the capital punishment process (see Davis 1998, p. 50). While recognizing the problems of arbitrariness and bias the death penalty system, both Justices supported States' rights to continue administering capital punishment in the same fashion that allowed such patterns of bias to develop in the first place.

<sup>&</sup>lt;sup>17</sup> "The plurality seems to believe [...] that provision for appellate review will afford a check upon the instances of juror arbitrariness in a discretionary system. But it is not at all apparent that appellate review of death sentences, through a process of comparing the facts of one case in which a death sentence was imposed with the facts of another in which such a sentence was not imposed, will afford any meaningful protection against whatever arbitrariness results from jury discretion. All that such review of death sentences can provide is a comparison of fact situations which must, in their nature, be highly particularized, if not unique, and the only relief which it can afford is to single out the occasional death sentence which, in the view of the reviewing court, does not conform to the standards established by the legislature" (*Woodson v. North Carolina*, 428 U.S. 280, 316 [1976]) (Rehnquist, J., dissenting).

view that the arbitrariness, caprice, and bias in the administration of capital punishment is detectable and repairable, although new procedural protections must be developed to significantly reduce or eliminate bias and error because existing rules have been inadequate (Baldus, Pulaski, and Woodworth 1994b; Berger 1991; Liebman 2000).<sup>18</sup> However, numerous death penalty analysts and socio-legal scholars are doubtful of the ability of proposed procedural safeguards to satisfactorily purge arbitrariness and bias from the capital punishment process (see, e.g., Kan and Phillips 2003; Liebman 2007; Ogletree 2002).<sup>19</sup>

Much of the debate (and confusion) over how to best identify and remedy problems with the administration of the death penalty stems from the inability of death penalty analysts to adequately explain the complex ways in which legally legitimate and illegitimate factors influence legal decision-making. Modifications to death penalty statutes are likely to be ineffective at eliminating these problems if they do not address the underlying sources of these disparate patterns (Black 1989).<sup>20</sup> Unfortunately research on the capital punishment system has been dominated by evaluation studies (i.e., studies of legal effectiveness) and, as a result, most death penalty scholars have failed to specifically develop or test theoretical propositions aimed at ordering, predicting, and

<sup>&</sup>lt;sup>18</sup> Based on interviews with 1,201 former capital jurors from 354 trials in fourteen states (Alabama, California, Florida, Georgia, Indiana, Kentucky, Louisiana, Missouri, North Carolina, Pennsylvania, South Carolina, Tennessee, Texas, and Virginia), Bowers and Foglia (2003) conclude that the constitutionally mandated requirements established to guide jury discretion and to eliminate arbitrary sentencing are not working.

<sup>&</sup>lt;sup>19</sup> There are some death penalty analysts who argue that the current capital-sentencing systems of the state and federal governments are, in fact, producing even-handed justice by limiting the death penalty to only the most heinous cases (see, e.g., Carrington 1978; Cassell 2003). The vast majority of empirical studies suggest, however, that nearly all jurisdictions with death penalty statutes administer capital punishment in an arbitrary or discriminatory fashion (see Chapter Four).

<sup>&</sup>lt;sup>20</sup> Problems with adequately identifying the sources of disparate charging-and-sentencing patterns are not unique to capital murder cases. Nearly twenty years of sentencing reform efforts for non-capital crimes at the federal and state levels have been largely unsuccessful at eliminating the influence of legally illegitimate case characteristics on legal decision-making (see Chapter Four).

explaining variation in legal behavior in the capital charging-and-sentencing process (see, generally, Black 1972; Mears 1998a).

The current project presents an examination of the capital charging-andsentencing process in Georgia from 1993 through 2000. Unlike previous studies of capital punishment, this study derives and tests specific research hypotheses from a general theory of law operating solely at the sociological level—that is, without reference to psychology (Black 1976). While this type of theorizing was at the foundation of early sociological thought (e.g., Durkheim [1895] 1962; Simmel 1909), it has been largely (and prematurely) abandoned by many contemporary socio-legal theorists. Georgia's capital punishment system was selected for these analyses for three important reasons. First, Georgia is one of the leading death-sentencing jurisdictions in the nation, ranking fifth in total number of executions since the practice of capital punishment officially began in America in 1608 (Espy and Smykla 2004). Second, Georgia has been the most influential state in shaping death penalty policy in the United States in the modern era of the death penalty. Well over a dozen cases originating in Georgia have set legal precedent with respect to the administration of capital punishment (Cook 1996). Third, Georgia's capital punishment system has received the most scholarly attention, allowing for a comparison of charging-and-sentencing patterns in the jurisdiction across various time periods (see also Chapter Five).

Chapter Two offers a historical overview of the practice of capital punishment in America. The history of the death penalty is divided into three eras: (1) the early colonial period through the Civil War; (2) the period following the Civil War through the 1960s; and (3) the 1970s to the present period. The chapter discusses the major developments that occurred in each period with respect to the administration of the death penalty and the enduring impact of these changes on the death penalty process. Chapter Three presents a detailed discussion of the major issues surrounding the use of the death penalty and how social science research has influenced the debate over capital punishment in America. Chapter Four provides an extensive review of the literature on the use of the death penalty and the influence of legally legitimate and illegitimate case characteristics on the capital charging-and-sentencing process at both the state and federal levels, with a particular emphasis on the role of race/ethnicity and region. To provide context, the chapter begins with a brief overview of the history of the discriminatory treatment of racial/ethnic minorities—particularly blacks—in the American legal system.

Chapter Five discusses the history of the administration of capital punishment in Georgia. The chapter is divided into several subsections: (1) an overview of the racially disproportionate application of the death penalty in Georgia and the American South, (2) a discussion of the landmark U.S. Supreme Court cases originating in Georgia and their impact on the current capital charging-and-sentencing system nationwide, (3) a description of the impact of race/ethnicity, gender, and region on administration of capital punishment in Georgia in the modern era, (4) an examination of the prevalence of miscarriages of justice (i.e., wrongful convictions and prejudicial error) in Georgia's modern death penalty system, and (5) a description of Georgia's capital charging-and-sentencing process and a discussion of the state's life without the possibility of parole legislation and its possible impact on the capital charging-and-sentencing process.

Chapter Six discusses the minimal role that theory has played in research on the impact of legally legitimate and illegitimate factors on the criminal sentencing process

and suggests that this relative inattention to theory has undermined sentencing reform efforts for both capital and non-capital crimes. The chapter provides an overview of the three dominant theoretical perspectives employed in sentencing research and evaluates these theories according to widely accepted scientific standards. After arguing that all three perspectives fail to successfully satisfy these established scientific criteria, a fourth perspective is presented that is believed to more completely realize the goals of scientific theory. The chapter also discusses the problems that the dominant theoretical perspectives have with conceptualizing race/ethnicity, gender, and region, and how these shortcomings undermine the ability of these theories to explain the continuing significance of race/ethnicity, gender, and region in the criminal sentencing process despite the efforts of state and federal governments to purge the effects of these factors from the criminal justice process. It is argued that the fourth theoretical perspective more completely captures the social significance of race/ethnicity, gender, and region on administration of law, in general, and on the capital charging-and-sentencing process more specifically. The chapter concludes by presenting several research hypotheses that are subsequently subject to empirical examination.

Chapter Seven describes how the data on Georgia's capital charging-andsentencing process were collected for this study and the analytical techniques used. Particular attention is given to why the analytical approach employed in this study is well suited for the research questions posed in the previous chapter. Methodological problems that have plagued previous empirical research on the capital charging-and-sentencing process in Georgia and other jurisdictions across the United States are highlighted and a discussion of how this study addresses those problems is presented. Chapter Eight reports the results from the empirical analyses and Chapter Nine summarizes the study's major findings, offers directions for future research, and discusses the tenability of current capital-sentencing reform efforts to eliminate racial/ethnic, gender, and regional bias in light of the study's results.

#### **Chapter 2: Background on Capital Punishment in the United States**

Capital punishment has been a stable feature of social policy in this nation for nearly four centuries.<sup>21</sup> The first recorded execution—in what ultimately would become the United States—was that of Captain George Kendall in the Jamestown Colony of Virginia in 1608 (see Espy and Smykla 2004).<sup>22</sup> Since that time, more than fifteen thousand individuals have been legally executed in the United States (Espy and Smykla 2004). While the death penalty has been fixture in American society, it has also been wrought with controversy and debate (Banner 2002). A brief overview of the legal history of the death penalty in the United States is provided below.

Death penalty scholars have divided the history of capital punishment in the United States into several different eras (see, e.g., Bedau 1997a; Haines 1996). Although these eras vary considerably with respect to duration and importance, and the lines of demarcation are somewhat blurred, they are all marked by significant changes in the administration of capital punishment in this country. For the purposes of this discussion, the history of the death penalty is divided into three periods: (1) the early colonial period through the Civil War; (2) the period following the Civil War through the 1960s; and (3) the 1970s to the present.

<sup>&</sup>lt;sup>21</sup> Although the U.S. Supreme Court imposed an official moratorium on capital punishment on June 29, 1972 (see Section 2.3), the actual practice of sentencing individuals to death was interrupted for less than six months (Haines 1996, pp. 45–46).

<sup>&</sup>lt;sup>22</sup> Kendall, who was of the leader of the seven member governing council of the Jamestown Colony, was accused of mutiny, tried in front a jury of his peers, and shot to death. Twenty-fours later, in 1632, Jane Champion became the first woman executed in the colonies for an unknown offense.

### 2.1 EARLY COLONIAL PERIOD THROUGH THE CIVIL WAR

The early capital punishment statutes of the original thirteen colonies varied considerably. Several colonies, modeling their statutes directly from English law, authorized capital punishment for a wide variety of crimes against the state, persons, and property (Banner 2002). Other colonies developed more restrictive capital laws, authorizing executions for a very limited number of crimes. Regardless of the scope of their capital statutes, all colonies authorized public execution by hanging as the mandatory punishment for certain crimes.

Prior to the American Revolution, public opposition to capital punishment was confined to a very small segment of society, namely religious groups such as the Quakers (Bedau 1997a). Following the Revolution, however, a growing prison reform movement provided fertile ground for death penalty abolitionists (Haines 1996). During this period, many vocal critics of capital punishment began to emerge, advocating either a complete repeal of the death penalty or severe restrictions on its use. Largely drawing inspiration from Cesare Beccaria's ([1764] 1819) seminal work, *On Crimes and Punishments*, abolitionists began gaining an audience, arguing that the death penalty did not provide a greater deterrent than life imprisonment coupled with hard labor.<sup>23</sup> As a result of the pressure from the abolitionists, several important reforms were made to the capital punishment statutes following the adoption of the U.S. Constitution through the Civil War. The earliest of these reforms was the creation of "degrees of murder." Departing from English legal tradition, many of the former colonies began dividing murder into two

<sup>&</sup>lt;sup>23</sup> The centerpiece of Beccaria's argument was that the certainty, not the severity, of punishment was most important in deterring crime. He believed that juries were more likely to feel sympathy for those facing a death sentence, feeling that the punishment was too harsh, and therefore may decide to acquit the defendant rather than impose the ultimate penalty.

categories: first- and second-degree, and reserved the death penalty for individuals convicted of first-degree homicide (Banner 2002). In 1793, Pennsylvania was the first jurisdiction to make the distinction, which resulted from a compromise between abolitionists (mostly Quakers) and retentionists, and most other states quickly followed suit (Keedy 1949). As a result, all jurisdictions that currently have capital punishment statutes reserve the death penalty for first-degree murder (or, equivalently, "capital murder").

The second major development during this period was the elimination of public executions (Haines 1996). As death penalty historians have noted, public executions were well-attended, highly formalized and highly symbolic events (Banner 2002; Bessler 1997; Frazier 2006). Individuals awaiting execution were often dressed in formal attire (i.e., dresses for females and suits for males) and openly transported from the jail to the execution site through the middle of town in order to attract spectators. After arriving at the execution site, religious leaders would offer sermons, followed by an official reading of the death warrant or some other statement by the sheriff or another high ranking legal authority (e.g., the judge who presided over the case). Individuals condemned to death would often be allowed to make a final speech before a prayer was offered and the execution was performed. Depending on the particularities of the crime (e.g., its severity/heinousness and the social status of the victim) and the background of the executed (e.g., her or his race/ethnicity and social status), the corpse was either prepared for burial, displayed publicly, mutilated, or turned over to medical authorities for dissection (Bessler 1997).

Beginning in early nineteenth century, many legal authorities and middle- and upper-class Americans joined abolitionists in opposing public executions. This was truly an unlikely alliance because most legal authorities and middle- and upper-class individuals generally approved of the death penalty; however they strongly believed that public executions were "uncivilized," "revolting," and "sadistic" (Bessler 1997, p. 67; cf. Johnson 2006b). Indeed, many death penalty proponents in state legislatures believed that executions posed a growing threat to the legitimacy of the death penalty, primarily through provoking sympathy for the condemned, leading to unruliness that threatened the public order, and brutalizing the entire community with the visibility of botched executions (e.g., decapitations and slow strangulations) (Banner 2002; Bessler 1997). Although pressure against public executions dates as far back as 1787,<sup>24</sup> it was not until nearly a half-century later in 1830 when Connecticut enacted the first laws to end public executions (Bessler 1997).<sup>25</sup> Soon after Connecticut's adoption of private hangings, similar laws were enacted in Massachusetts, New Hampshire, New Jersey, New York, Pennsylvania, and Rhode Island (Bessler 1997; Haines 1996). Fifteen additional states restricted executions to prisons (or prison yards) by 1849 (Haines 1996). The adoption of private executions, however, took much longer in other parts of the nation. In fact, an

<sup>&</sup>lt;sup>24</sup> Perhaps the most vocal critic of public executions at this time was Dr. Benjamin Rush, a signer of the Declaration of Independence. In a 1787 speech delivered in the home of Benjamin Franklin, Rush declared that "crimes should be punished in private, or not punished at all" (Bessler 1997, p. 40)

<sup>&</sup>lt;sup>25</sup> In 1828, New York passed a law permitting, but not mandating, sheriffs to conduct private executions. The law, however, was completely ineffective and executions remained public in New York until they were officially banned in 1835 (Bessler 1997, p. 41). Similar "discretionary" laws were passed in Alabama (1840), Georgia (1859), and Virginia (1856) and these three states did not mandate private executions until the end of the nineteenth century (Bessler 1997, p. 228 n.7).

entire century would pass before public executions were completely eliminated in the United States (Bessler 1997).<sup>26</sup>

The third major reform during this period was the authorization of trial juries to make binding sentencing recommendations in capital cases. Borrowing heavily from English law, most jurisdictions in the United States required a mandatory sentence of death in capital cases when a defendant was convicted of murder. Juries would often recommend that the defendant, although guilty of the capital offense, receive a sentence of life imprisonment rather than a death sentence, but judges would ignore such advice (Bedau 1997a). Beginning in the mid-nineteenth century, however, states began allowing juries to require the trial judge to grant leniency to the convicted offender. While the origins and growth of this practice are debated by legal scholars, many believe that the pressure to introduce discretion into the capital sentencing process resulted from both a general populist trend at the time and to alleviate concerns that juries would acquit capital defendants who they believed to be guilty of the crime, but whose offense did not warrant a death sentence (Banner 2002; Bedau 1997a). Some of the earliest jurisdictions to allow jury discretion in capital cases were Alabama (1841), Tennessee (1841), and Louisiana (1846). Other states began adopting discretionary jury sentencing statutes throughout the remainder of the nineteenth century, although mandatory sentences for certain crimes

<sup>&</sup>lt;sup>26</sup> By most accounts, the last public execution in the United States was that of Rainey Bethea in Owensboro, Kentucky on August 14, 1936. Bethea, a twenty-two year-old black male, was executed for the murder of an affluent seventy year-old white woman. It is estimated that nearly twenty thousand spectators hailing from five states witnessed the execution without restriction. Only 37 days had elapsed between Bethea's arrest and execution. The state of Kentucky received considerable negative publicity for the hanging, and as a result, the Kentucky General assembly mandated private executions two years later in 1938 (Bessler 1997, p. 67).

States, however, differentially define what constitutes a "public" execution, so what would be considered a private execution in one state could be considered public in another. Several executions following Bethea's, in Missouri and Kentucky, were conducted within an enclosure, but numerous witnesses were still able to view these events (Bessler 1997). More recently, the June 11, 2001 execution of Oklahoma City bomber, Timothy McVeigh, was witnessed by over 300 people (many by closed circuit television).

remained lawful well into the mid-twentieth century. As a result of a number of U.S. Supreme Court rulings in the 1970s, guided jury discretion in capital cases became a near universal feature of capital statutes in the nation (see Section 2.3).<sup>27</sup>

The fourth major development during this period was the significant narrowing of the scope of capital offenses.<sup>28</sup> While the scope of capital statutes varied considerably across the nation, many jurisdictions in the early nineteenth century began significantly limiting the number of crimes that were punishable by death. In contrast to English criminal law during this period that encompassed wide range of capital offenses (e.g., burglary, armed robbery, arson, blasphemy, witchcraft, bestiality, sodomy, and idolatry), many jurisdictions began limiting capital punishment to offenders convicted of murder, kidnapping, rape, treason, or espionage (Banner 2002). Although some jurisdictions did authorize the death penalty for crimes other than the aforementioned offenses, they rarely, if ever, executed individuals for those crimes (Bedau 1997a). There were, however, some important exceptions. Jurisdictions in the West and South tended to enact much broader capital statutes than other jurisdictions. For example, horse thievery, claim jumping, and cattle rustling were often punished by death in jurisdictions in the West (Bedau 1997a). In the South, stealing hogs, receiving certain stolen goods (e.g., horses), minor theft, and embezzling tobacco were all capital offenses (Banner 2002).

Southern jurisdictions also enacted "slave codes" that authorized a death sentence against slaves for a wide variety of offenses, many of which were nonviolent (see

<sup>&</sup>lt;sup>27</sup> Some states (e.g., Florida) allowed the trial judge to overrule the sentencing recommendation of the jury if she or he desired. Although most judges overruled jury sentences in favor of more lenient punishments, this was not always the case. In fact, it was not until *Ring v. Arizona* (536 U.S. 584 [2002]) that the U.S. Supreme Court finally ruled that juries were solely authorized to decide critical issues in death penalty cases (see Section 2.3).

<sup>&</sup>lt;sup>28</sup> At the time of the American Revolution, every colony, save Rhode Island, had ten or more capital crimes listed in their statutes (Acker and Lanier 2003).

Chapter Four). Moreover, in several of these jurisdictions, the killing of a slave by a white person was not prosecutable as criminal murder until the American Revolution (Spindel 1989) and slave owners faced basically no legal risk when they killed their slaves (Hindus 1980; Schwartz 1988). As a result of the plantation economy in the South, whites were also subject to the death penalty for a variety of nonviolent offenses. For example, whites convicted of aiding slaves to escape, stealing slaves, and inciting insurrection among slaves were subject to the death penalty (Banner 2002). And while rape was a capital offense in many jurisdictions across the United States until the late 1970s, the death penalty was disproportionately applied to blacks convicted of raping white women (see LaFree 1989; Wolfgang and Riedel 1973, 1975). In recent years, legislators in many states have frequently introduced bills that would authorize capital punishment for a host of non-homicide offenses (Bedau 1997a; Sack 1997). In fact in 1994, the U.S. Congress enacted the Federal Death Penalty Act, authorizing the death penalty for several dozen crimes, including four non-homicide offenses.<sup>29</sup>

The fifth important reform during this period was the actual repeal of death penalty statutes in a limited number of jurisdictions. Opponents of capital punishment, with a few qualifications, succeeded in abolishing the death penalty in Michigan (1847), Rhode Island (1852), and Wisconsin (1853). Repeals of the death penalty in Michigan and Rhode Island were partial, however, because Michigan retained the death penalty for treason and Rhode Island retained the death penalty for the killing of a prison guard by

<sup>&</sup>lt;sup>29</sup> The non-homicide crimes which may result in a federal death sentence are: espionage (18 U.S.C. § 794), treason (18 U.S.C. § 2381), trafficking large quantities of drugs (18 U.S.C. § 3591(b)), and "attempting, authorizing or advising the killing of any officer, juror, or witness in cases involving a continual criminal enterprise, regardless of whether such a killing actually occurs" (18 U.S.C. § 3591(b)(2)).

someone serving a life-term prison sentence (Bedau 1997a).<sup>30</sup> The momentum to repeal the death penalty in other jurisdictions would come to an abrupt halt, however, as the nation became embroiled in civil war (Banner 2002). During the American Civil War, a conflict that would last more than five years and claim over 600,000 lives, there were no significant changes in the administration of the death penalty and virtually no mention of debate over capital punishment in legislative journals (Davis 1957).

Well over 3,000 executions were carried out during the first era of capital punishment in America (Espy and Smykla 2004). This period, encompassing more than 250 years of executions, and nearly 100 years of reform, also witnessed some profound changes in the administration of the death penalty. In particular, five major developments occurred during this era: (1) the advent of degrees of murder; (2) the widespread, though incomplete, elimination of public executions; (3) the authorization of capital juries to make *binding* sentencing recommendations; (4) the narrowing of capital offenses, particularly with respect to non-violent and religious offenses; and (5) the complete or partial repeal of death penalty statutes in a limited number of jurisdictions. Death penalty opponents and reformers would have wait until the Post-Civil War period before any significant reform would occur again.

### 2.2 POST-CIVIL WAR PERIOD THROUGH THE 1960s

Following the Civil War, several states quickly abolished the death penalty, only to reinstate it shortly thereafter. For example, the death penalty was abolished for six years in Iowa (1872 - 1877), seven years in Maine (1876 - 1882), and four years in

<sup>&</sup>lt;sup>30</sup> Michigan would completely reinstate the death penalty in 1963 (Bowers 1984).

Colorado (1897 – 1900) (Bowers 1984).<sup>31</sup> In the early twentieth century, more jurisdictions abolished the death penalty. Similar to Iowa, Maine, and Colorado, many of these jurisdictions would reinstate capital punishment just a few years later. Washington abolished its death penalty in 1913 and reinstated six years later in 1919. Repeals of the death penalty were also short-lived in Oregon (1914 – 1920), Tennessee (1915 – 1919), Arizona (1916 – 1918), and Missouri (1917 – 1919). Both Kansas (1907) and South Dakota (1915) abolished their death penalties, reinstating them less than three decades later in 1935 and 1939, respectively.<sup>32</sup> Some legal historians have argued that the reinstatement of capital punishment in these states resulted from fear and anxiety over rising crime during Prohibition (1916 – 1932) and the Great Depression (1929 – 1940) (see Bedau 1997a). There were some jurisdictions during this period, however, that abolished the death penalty for a significant number of years. Minnesota, for example,

<sup>&</sup>lt;sup>31</sup> Maine would re-abolish the death penalty in 1887, as would Iowa in 1965 (Bedau 1997a).

<sup>&</sup>lt;sup>32</sup> The U.S. Supreme Court lifted the nationwide moratorium on the death penalty in 1976, but Kansas did not reinstate its death penalty until 1994. In December 2004, by a vote of four to three, the Kansas Supreme Court ruled that state's death penalty statute was unconstitutional because of the manner in which it required juries to weigh aggravating and mitigating factors (*State v. Marsh*, 102 P.3d 445 [Kan. 2004]). The Kansas law required jurors to impose a death sentence when aggravating and mitigating factors were completely balanced. The court held that the requirement that a "tie" automatically benefit the prosecution constituted cruel and unusual punishment. The decision overturned the death sentences of all six individuals on Kansas' death row. The court did not rule, however, that the death penalty was unconstitutional, per se, and simply held that the current statute needed to be modified such that a tie would benefit the defendant (Liptak 2004). Two years later, the U.S. Supreme Court reversed the Kansas Supreme Court's decision, ruling that the state's death penalty statute was constitutional (*Kansas v. Marsh*, 548 U.S. 163 [2006]).

Six months prior to the Kansas Supreme Court's decision, the New York Court of Appeals (the highest state court in New York) ruled that the state's death penalty statute was unconstitutional because of its "deadlock" provision (see *People v. Lavalle*, 817 N.E.2d 341 [N.Y. Ct. App. 2004]). This provision required judges to inform capital jurors that the judge was required to impose a sentence that may permit the defendant to be released on parole in 20 to 25 years if the juror was deadlocked as to whether to impose the death sentence. The court held, by a margin of four to three, that the provision violated the state constitution's due process clause (Glaberson 2004). At the time of the court's decision, four individuals were on death row in the state. Like the Kansas court, the New York court did not hold that the death penalty statute was inherently unconstitutional, but merely the deadlock provision must be modified to the satisfaction of the court. New York reinstated its death penalty in 1995, some 19 years after the U.S. Supreme Court lifted its nationwide band on the death penalty.

abolished the death penalty in 1911 and has yet to reinstate it (Bessler 2003).<sup>33</sup> North Dakota, in 1915, limited the death penalty to individuals convicted of first-degree murder while *already* serving a life sentence for first-degree murder and completely abolished in the death penalty in 1975 (Bowers 1984). Up through 1960s, many other jurisdictions would abolish the death penalty. Alaska (1957), Hawaii (1957), West Virginia (1965), and Iowa (1965)<sup>34</sup> permanently abolished their death penalties, whereas Vermont (1964) and New Mexico (1969) enacted partial repeals.<sup>35</sup> Several jurisdictions would also experience brief repeals of the death penalty in the 1950s and 1960s. Delaware's repeal, for example, lasted only three years (1958 – 1961) and Oregon's *second* repeal in 1964 would last two decades.

In addition to the cycle of abolition and restatement of the death penalty during the period following the American Civil War through the 1960s, several other interesting developments occurred. Perhaps the first major issue to take the national stage was the method of execution employed by jurisdictions with the death penalty (Paternoster, Brame, and Bacon 2007). Prior to the late 1880s, the dominant method of execution was hanging, although many jurisdictions also used firing squads to execute individuals

<sup>&</sup>lt;sup>33</sup> The last official execution in Minnesota occurred in Ramsey County in 1906. Nearly a century later, in December 2003, Minnesota Governor Tim Pawlenty began a public campaign to reinstate the death penalty in his jurisdiction. Pawlenty announced that he supported a statewide referendum for a constitutional amendment to reinstate capital punishment. Although such a referendum would take the decision to reinstate the death penalty out of the hands of the legislature and place it in the hands of voters, the legislature is still required to authorize the inclusion of such a referendum on the ballot. Bills to reinstate the death penalty by simply legislative action were also introduced in both houses of the legislature (H.R. 1602, 83<sup>rd</sup> Reg. Sess. [Minn. 2003]; S. 1860, 83<sup>rd</sup> Reg. Sess. [Minn. 2003]), however both bills died in committee.

<sup>&</sup>lt;sup>34</sup> This was the second time that Iowa abolished the death penalty (see above), and it has yet to be reinstated.

<sup>&</sup>lt;sup>35</sup> Both Vermont and New Mexico's partial repeals in 1964 and 1969, respectively, authorized the death penalty for the killing of an on-duty law officer and a killing committed by someone who already convicted of murder. Vermont completely abolished the death penalty in 1987 (Bedau 1997a) and New Mexico repealed its death penalty in 2009 (Associated Press 2009).

convicted of capital crimes (Bessler 1997).<sup>36</sup> Hangings, however, were often viewed as inhumane and barbaric, and they would frequently result in decapitation, slow strangulation, and disfigurement of the executed (see Driggs 1993; Moran 2002). For several decades leading up to the 1880s, legal authorities in New York debated the efficacy and humaneness of hanging individuals condemned to death. In 1886, a commission was appointed by New York Governor David Bennett Hill to explore alternative methods of execution. The Gerry Commission, as it was to be called, queried numerous experts, medical and otherwise, as to the most appropriate method of execution. Over thirty-four different alternatives were presented to the commission, with the most popular methods being lethal injection and electrocution. While lethal injection appeared to be the most painless alternative, many medical professionals strongly objected to using medical technology to kill convicted criminals.<sup>37</sup> One of the commission members, a New York dentist by the name of Alfred Southwick, began to strongly advocate for establishing electrocution as the preferred alternative to the rope and enlisted the support of electrical pioneer Thomas Edison (Denno 1994). The push towards execution by electrocution indirectly developed out of an earlier movement to provide a painless way to kill cattle before they were slaughtered and Southwick's personal belief that death by electrocution was quick and painless (Moran 2002).<sup>38</sup> Southwick, along with the support of Edison, convinced the Gerry Commission to

<sup>&</sup>lt;sup>36</sup> In the case of *Wilkerson v. Utah* (99 U.S. 130 [1878]), the U.S. Supreme Court rejected claims that death by firing squad violated the Eighth Amendment protection against cruel and unusual punishment.

<sup>&</sup>lt;sup>37</sup> The humaneness of lethal injection was underscored by the fact that some proponents of the death penalty, including individuals in the medical community, objected to the use of lethal injection as a means of execution because they believed it to be "too painless" (Moran 2002, p. 79)

<sup>&</sup>lt;sup>38</sup> It is reported that Southwick became a vocal supporter of death by electrocution after witnessing an accidental electrocution of a man who made contact with a live wire in Buffalo, New York. After witnessing this event, Southwick began performing similar experiments on animals which further solidified his belief that death by execution was quick and relatively painless (Denno 1994; Moran 2002).

endorse electrocution as the preferable method of execution and the Commission presented its report to the New York legislature in 1888. Later that year, the New York legislature passed the Electrical Execution Act mandating death by electrocution for anyone sentenced to death in the state and became the first jurisdiction to abolish execution by hanging (Moran 2002). The first scheduled execution by electrocution in New York was that of William Kemmler, convicted of murdering his girlfriend during an intoxicated rage and sentenced to death. Kemmler challenged New York's use of the electric chair on the grounds that it violated his Eighth Amendment right to protection from cruel and unusual punishment (In re Kemmler, 136 U.S. 436 [1890]). Similar to the *Wilkerson* decision twelve years earlier (see *supra*, note 36), in which U.S. Supreme Court denied the defendant's claim that Utah's firing squad was cruel and unusual punishment, the court held that the electric chair did not violate Kemmler's Eighth Amendment right. The execution of Kemmler, however, did not exhibit the quick and painless character that proponents of electrocution guaranteed. On August 6, 1890, Kemmler was initially given a surge of electricity lasting 17 seconds and pronounced dead by an attending physician. Shortly after the doctor's pronouncement, however, Kemmler's body began to twitch and other physicians on site discovered that Kemmler still had a pulse and immediately recommended that Kemmler be given another current of electricity. After the second current was applied, it is reported that the scent of burnt
flesh filled the death chamber (Moran 2002).<sup>39</sup> Nonetheless, execution by electrocution was eventually adopted by more that three dozen states (Banner 2002).<sup>40</sup>

By the 1920s, however, the electric chair increasingly came under attack. In response to criticism over the use of electrocution to execute those convicted of capital crimes, states began adopting lethal gas as the dominant method of execution. Proponents of lethal gas argued that it was a much more humane method of execution than the firing squad, hanging, and the electric chair. Nevada, which was one of the few states to authorize the use of the firing squad, was the first state to authorize the use of lethal gas in 1923, although a decade would elapse before other states would follow suit (see Denno 1994).<sup>41</sup> But while adopting "advances" in methods of execution that appeared to be more humane, many states retained and employed older methods of executions. For example, while lethal injection had been authorized for some time in all death penalty states, save Nebraska, firing squads were used in Utah in the mid-1990s, hangings were carried out in Delaware and Washington in the mid-1990s, and the electric chair remains authorized in ten states.<sup>42</sup>

The second major development in this period was the passing of control of executions from local authorities to state authorities. In the early twentieth century, local authorities began relinquishing control of executions to state authorities and the trend was

<sup>&</sup>lt;sup>39</sup> Nearly sixty years after *Kemmler*, in *Louisiana ex rel. Francis v. Resweber* (329 U.S. 459 [1947]), the Supreme Court ruled that a second electrocution of a 15 year old boy, six days after the initial botched attempt to electrocute him failed, was neither cruel and unusual punishment nor constituted double jeopardy.

<sup>&</sup>lt;sup>40</sup> By 1912, seven states had switched from hanging to electrocution: Ohio (1896), Massachusetts (1898), New Jersey (1908), Virginia (1908), North Carolina (1909), Kentucky (1910), and South Carolina (1912). By 1929, 68 percent of all executions were by electrocution.

<sup>&</sup>lt;sup>41</sup> In 1933, both Colorado and Arizona officially adopted the gas chamber as the preferred method of execution in their jurisdictions. Over the next 20 years, eight other states—all in the West and South—switched to lethal gas, with New Mexico being the last in 1955 (Banner 2002).

<sup>&</sup>lt;sup>42</sup> The eight states that currently authorize the use of the electric chair are: Alabama, Arkansas, Florida, Kentucky, Oklahoma, South Carolina, Tennessee, and Virginia (Death Penalty Information Center 2008).

basically completed by the 1960s (Banner 2002; Bessler 1997). Several factors were responsible for this shift, among them being: (1) an attempt by state authorities to increase the legitimacy of capital punishment by centralizing executions, (2) a concern over the prevalence of botched executions resulting from the inexperience of local executioners, (3) the expense associated with changing the method of execution in most states from hanging to electrocution, and (4) the elimination of executions from the public eye (recall, by most accounts, the last official public execution occurred in 1936). But while state authorities assumed responsibility for executing individuals condemned to death, the actual administration of the death penalty-the entire capital process leading up to execution—remained under local control (Banner 2002). Local prosecutors continued to exercise discretion in selecting cases for the death penalty and local judges/juries continued to exercise discretion in sentencing defendants charged with capital crimes to death. The administration of capital punishment, then, remained highly variable within and between death penalty states (see Chapter Four), with the executioner being the only uniform feature of the entire process.

The third major change during this period was the expansion of the role of federal appellate courts. Beginning in the early nineteenth century, it was fairly common for capital defendants to receive at least perfunctory review of their death sentence by state appellate courts, although very few states enacted laws mandating appellate review in every death penalty case (Bedau 1997a). In the 1920s, state prisoners on death row began receiving relief in federal courts based on violations of constitutional protections by state criminal courts. Rather than challenging the constitutionality of the death penalty in general, however, state death row inmates attacked the death penalty on a case-

by-case basis, claiming that their constitutional rights had been violated by the state criminal proceedings in capital cases (Bedau 1997a). These appeals, however, were rather infrequent until the expansion of federal habeas corpus during the early 1960s (Haines 1996). Nonetheless, the significant drop in annual executions beginning in the 1950s has been partly attributed to the "increasing sensitivity of the federal courts to postconviction appellate litigation on behalf of capital defendants" (Bedau 1997a, p. 13). Whereas executions were relatively frequent and stable in the 1930s and 1940s (148 per year), they dropped to half the annual average in the 1950s (72 per year) and one-quarter the annual average in 1960s (38 per year). In fact, there were only twenty-one executions in 1963, fifteen in 1964, seven in 1965, one in 1966, two in 1967, and no executions from 1968 to 1972, when the U.S. Supreme Court placed a moratorium on capital punishment (more below) (Espy and Smykla 2004).<sup>43</sup>

A fourth major development during this period was a shift in strategy by antideath penalty attorneys. Prior to the 1960s, abolitionists mainly attacked capital punishment by lobbying for the repeal of death penalty statutes, pursuing clemency for individual convicts, and attempting to educate the general public regarding the inhumanity of the death penalty (Banner 2002; Haines 1996). Beginning in the 1950s, however, attorneys associated with the National Association for the Advancement of Colored People Legal Defense and Educational Fund (hereinafter, LDF),<sup>44</sup> a civil rights law firm founded in 1939 focusing primarily on protecting and preserving the rights of blacks in the United States, began attacking the constitutionality of capital punishment based upon racial disparities in the administration of the death penalty in southern states

<sup>&</sup>lt;sup>43</sup> There would not be another "legal" execution in the United States until 1977 (more below).

<sup>&</sup>lt;sup>44</sup> The LDF made use of a wide network of cooperating attorneys across the nation to challenge discriminatory practices against blacks (Haines 1996).

for the crime of rape (see also Chapter Four). This effort would be fueled several years later by the Supreme Court's decision to deny granting certiorari in a case that challenged the constitutionality of the death penalty for rape in which no life is taken (*Rudolph v.* Alabama, 375 U.S. 889 [1963]). Although the court voted six to three against granting certiorari, Justice Arthur Goldberg-with Justices William Brennan and William Douglas concurring—filed a dissenting opinion questioning whether: (1) the death penalty for rape was a violation of "evolving standards of decency" in view of the worldwide trend against the death penalty for rape; 45 (2) the death sentence is a disproportionate penalty for any crime in which no life is taken; and (3) lesser punishments can serve the same legal purpose as execution, therefore making the death penalty for rape unnecessarily cruel (Kirchmeier 2002). Two years later in 1965, the LDF broadened the scope of their attack and began a "multi-issue assault on the constitutionality of capital punishment in general, that is, for white as well as black offenders and in murder as well as rape cases" (Haines 1996, p. 14). In the same year, the American Civil Liberties Union (hereinafter, ACLU), an organization established in 1920 that directed its efforts at preserving the liberties guaranteed by the Bill of Rights, allied itself with the LDF after adopting an abolitionist policy (Haines 1996). This would mark the first serious attempt by death penalty opponents to challenge the very constitutionality of the death penalty in the courts (Bedau 1997a).<sup>46</sup>

<sup>&</sup>lt;sup>45</sup> In *Trop v. Dulles* (365 U.S. 86, 100–101 [1958]), the Supreme Court first articulated the "evolving standards of decency" test when interpreting the Eighth Amendment's prohibition against cruel and unusual punishment, noting that such an interpretation "must draw its meaning from the evolving standards of decency that mark the progress of a maturing society." Although the Court did not articulate any specific criteria, the Court did look to current practices in the international community (Glass 2006, p. 1344).
<sup>46</sup> As noted above, death penalty opponents previously challenged the constitutionality of the *method* of execution employed by death penalty states in *Wilkerson* (firing squad) and *In re Kemmler* (electrocution); however the very constitutionality of capital punishment, itself, was never seriously challenged (Bedau 1997a).

The LDF had previously, albeit unsuccessfully, challenged the racially discriminatory application of the death penalty for rape cases in Virginia in 1950 (Hampton v. Commonwealth, 58 S.E.2d 288 [Va. 1950]) and had prepared to raise the issue again until the defendant's sentence was reduced to a life sentence (Hamilton v. Alabama, 386 U.S. 52 [1961]) (Haines 1996). The LDF, recognizing the importance of gathering information on sentencing patterns in rape cases in the South, funded a study during the summer of 1965 to collect such data (see Wolfgang and Riedel 1973). This data was subsequently presented to the court in Maxwell v. Bishop (398 F.2d 138 [8th Cir. 1968) to challenge discrimination in the imposition of the death penalty for rape in the South. William Maxwell, a black male, was convicted in 1962 and sentenced to death for the rape of a white woman in Arkansas. Maxwell's attorneys argued that Arkansas juries disproportionately sentenced black males convicted of raping white women to death. After refusing to hear the case, the Eighth Circuit Court of Appeals was ordered to do so by the U.S. Supreme Court. On appeal, Maxwell's attorneys raised two additional issues, namely that (1) the trial was unconstitutional because the court combined the guilt/innocence and sentencing decisions into one phase (called a single-verdict jury) and (2) the death penalty process lacked sentencing standards. The court, while acknowledging that the data clearly demonstrated a historical pattern of biased sentencing, rejected the claim on the grounds that the data did not prove discrimination by the jury that convicted and sentenced Maxwell. (As will be discussed below, the U.S. Supreme Court would reject future evidence of racial discrimination in the administration of the death penalty on similar grounds.) The court avoided the single-verdict procedure and sentencing standards questions altogether.<sup>47</sup>

<sup>&</sup>lt;sup>47</sup> The U.S. Supreme Court would later grant Maxwell a new trial, but it avoided the central constitutional

In another case in the late 1960s, Boykin v. Alabama (395 U.S. 238 [1969]), involving a man who was sentenced to death after pleading guilty to several armed robberies in which no one was killed, the LDF directly challenged capital punishment for armed robbery as "cruel and unusual punishment" because it was so infrequently and arbitrarily applied in such cases. Similar to *Maxwell*, Boykin's attorneys also challenged his sentence on grounds that Alabama juries lacked sentencing standards on which to base their decision. The U.S. Supreme court granted Boykin a new trial based on a procedural issue (Boykin had been allowed to plead guilty without understanding he would still be eligible for the death penalty), but avoided the LDF's two key arguments: (1) execution for armed robbery in which no one is killed constituted cruel and unusual punishment and, therefore, was a violation of the Eighth Amendment and (2) the absence of sentencing standards. In light of the court's focus on the narrowest of issues in Maxwell and Boykin and its complete silence on the central Eighth Amendment issue, opponents of capital punishment grew increasingly doubtful that the "cruel and unusual punishment" argument would ever be an effective attack on the death penalty (Haines 1996).

The LDF did experience some partial victories with respect to their procedural assault on the death penalty in the late 1960s. In *United States v. Jackson* (390 U.S. 570 [1968]), the defendant was sentenced to death for kidnapping under the "Lindbergh Law," which allowed defendants accused of kidnapping eligible for the death penalty only if their case went before a jury (Haines 1996, p. 32). The Lindbergh Law was overturned in a U.S. District Court because it unduly coerced defendants to waive their

arguments raised in the case, preferring to grant a new trial based upon a jury nullification issue (Haines 1996).

Fifth (self-incrimination) and Sixth (trial by jury) Amendment rights. On appeal, the U.S. Supreme Court upheld the constitutionality of the death penalty for kidnapping, but ruled that portions of the law violated the accused constitutional rights. In *Witherspoon v. Illinois* (391 U.S. 510 [1968]), the LDF and ACLU challenged the constitutionality of "death qualification" of jurors in capital cases. They presented the court with evidence suggesting that "death qualified" juries were more likely to convict than non-death qualified juries (see also Dieter 2005; Ellsworth 1988; Fitzgerald and Ellsworth 1984; Oberer 1961; Young 2004) and that this placed capital defendants at an unfair risk of conviction. The U.S. Supreme Court held that courts would only be allowed to excuse potential jurors who expressed absolute opposition to the death penalty (Haines 1996, p. 33).<sup>48</sup>

The second era of capital punishment in America, lasting an entire century and responsible for nearly 10,000 executions, began with several shortly lived repeals of capital statutes in several states. Opponents and reformers of capital punishment were able to successfully lobby states to change their method of execution from hanging to electrocution, although use of the electric chair was not without strong criticism from the very beginning. Two other significant administrative changes occurred during this period as well, namely local authorities relinquishing control of executions to state authorities and the expansion of the role of federal appellate courts in the capital punishment process. State control of executions offered the promise of significantly reducing the likelihood of botched executions at the hands of inexperienced local authorities and

<sup>&</sup>lt;sup>48</sup> Abolitionists originally perceived *Witherspoon* to be a tremendous victory because the ruling was retroactive; that is, it applied to past cases in which jury selection did not meet the standard established by the court. Subsequent lower court rulings, however, significantly diluted the impact of *Witherspoon* (Haines 1996, p. 211 n.8)

further legitimating the practice capital punishment. The expansion of the role of the federal appellate courts allowed defendants convicted of capital crimes in state courts to challenge the constitutionality of their particular conviction and sentence and provided another avenue of relief for those facing the death penalty (see also Liebman et al. 2000c). Finally, the end of this era witnessed a significant shift in strategy by anti-death penalty attorneys in which conventional approaches of legislative lobbying, public education, and pressure for executive clemency were largely abandoned in favor of mounting a direct assault on the constitutionality of the death penalty, which included attacks based on racial and procedural grounds, as well capital punishment's apparent incompatibility with contemporary standards of decency (Haines 1996). While the Supreme Court would repeatedly refuse to address the constitutionality of the death penalty, itself, and the two key procedural flaws raised by anti-death penalty attorneys (i.e., single-verdict jury and the lack of sentencing standards), instead opting to decide cases on narrow procedural grounds, it did provide some hope to anti-death penalty attorneys at the end of the 1960s when it agreed to hear two additional cases in which similar constitutional arguments were central and that both involved the crime of murder.

## 2.3 The 1970s to the Present

As noted above, at the close of the 1960s, the Supreme Court failed to resolve the two central issues raised in *Maxwell* and *Boykin*: (1) single-verdict juries and (2) the absence of sentencing standards. The court did, however, offer some hope to death penalty opponents and reformers when it agreed to hear two new death penalty cases in which these issues were central. In 1970, when the Supreme Court issued its final ruling

in *Maxwell v. Bishop* (398 U.S. 262 [1970]) which sidestepped the central issues in *Maxwell* and *Boykin*, it also announced on the same day that to hear two cases, *Crampton v. Ohio* and *McGautha v. California*; the cases were consolidated as *McGautha v. California* (402 U.S. 183 [1971]).

In *Crampton*, the defendant was tried and convicted under the single-verdict jury trial in Ohio, which allowed juries to make two decisions—the first with respect to guilt/innocence and the second with respect to sentencing-after one deliberation. In an argument similar to United States v. Jackson, Crampton's attorneys objected to the single-verdict jury because they believed that such a system forced defendants to make a choice between two constitutional rights: the Fifth and Fourteenth Amendments. On the one hand, if a defendant decided to exercise her or his right against self-incrimination at the conviction phase, she or he would also have to forfeit the right to speak to the jury about the appropriateness of the death sentence in her or his case and would be prevented from illuminating aspects of her or his character, background, and the crime that might warrant a sentence less than death. Surrendering the right to remain silent at the conviction phase, on the other hand, would allow the defendant to introduce mitigating evidence that might lessen the severity of her or his sentence, but such testimony would likely dramatically increase the likelihood of conviction by introducing (or allowing the prosecution to introduce) evidence that was not directly relevant to the question of guilt/innocence. Crampton's attorneys argued that the only solution that was constitutionally acceptable was to mandate bifurcated capital trials in which there would be two separate hearings. Under this system, the first hearing would address the issue of guilt/innocence and only allow evidence relevant to this issue. The second hearing

(assuming that the first phase resulted in a capital conviction) would determine the appropriate punishment of the convicted defendant and allow a broader range of evidence that was usually not permissible at the guilt phase.

Unlike the *Crampton* case in Ohio, *McGautha* was tried in California, a state which had bifurcated capital trials; however in both *Crampton* and *McGautha*, capital juries were given a great deal of discretion in choosing the appropriate sentence and very little guidance as to how they should arrive at such a decision. That is, outside vague warnings indicating that the law forbids jurors from acting on conjecture or prejudice, juries were not told what factors *should* and *should not* be considered. Crampton's and McGautha's attorneys strongly objected to the absence of sentencing guidelines and the wide discretion enjoyed by capital jurors, suggesting that such broad discretion invited decisions based on caprice or bias. They further argued that the absence of sentencing guidelines violated their clients' Fourteenth Amendment rights to equal protection under the law because under such a system, defendants who committed equally serious crimes could be treated differently—particularly members of minority groups and the economically disadvantaged.

Both Crampton and McGautha's cases were handled by private attorneys, but LDF attorneys filed amicus curiae briefs in both cases and were extremely hopeful that their six years of constitutional attacks on capital punishment will finally score a major victory (Haines 1996). The U.S. Supreme Court, however, in a six-to-three vote, decided against both issues, ruling that neither the absence of sentencing guidelines nor singleverdict trial procedures were in violation of the Fourteenth Amendment. Justice John Harlan, the author of the majority opinion, wrote, with respect to sentencing guidelines, that it was both unwise and futile to attempt to determine, *a priori*, the factors that would warrant a death sentence. Furthermore, he stated that the "truly awesome responsibility" of taking a human life could not be reduced to legal formulae.<sup>49</sup> With respect to bifurcated trials, Justice Harlan believed that a defendant's decision to testify during trial is a strategic one, similar to all other decisions that must be made during trial, and is not without risk. He also commented that while he believed that bifurcated trials were superior means of dealing with capital cases, neither the Supreme Court nor the U.S. Constitution guaranteed trial procedures that were the best of both worlds.

With the rulings in *McGautha*, death penalty abolitionists feared that the *de facto* moratorium on the death penalty, which had began in 1967, would soon come to an end.<sup>50</sup> This would be particularly true if the Supreme Court refused to hear a direct Eighth Amendment challenge, a fact that seemed very likely considering the court's decision with respect to this issue two years earlier in *Boykin* and Justice Hugo Black's concurring opinion in *McGautha* which stated executions were common when the U.S. Constitution was written, and therefore, the cruel and unusual punishment clause was irrelevant to the question of the death penalty (Haines 1996).

However, while things were looking bleak for abolitionists with respect to an Eighth Amendment assault on capital punishment, several events occurred that increased the likelihood that the U.S. Supreme Court would have to hear the issue at least once more. First, on the very day that the *McGautha* decision was handed down, governors in

<sup>&</sup>lt;sup>49</sup> The majority's position in *McGautha* was at odds with the American Law Institute's (hereinafter, ALI) Model Penal Code that provided a list of factors for judges and juries to consider when sentencing convicted capital defendants. According to the ALI, the determination of factors that should influence the appropriate sentence in a death penalty case, *a priori*, was within human capacity and the best available option.

<sup>&</sup>lt;sup>50</sup> The hanging of Louis Jose Monge in Colorado on June 2, 1967 was the last execution before the *de facto* moratorium.

Ohio and Maryland announced that they would maintain a moratorium on executions until the Supreme Court resolved the issue as to whether the death penalty was cruel and unusual punishment. Second, the Fourth Circuit Court of Appeals ruled in late 1970 that death sentences for rape in which the crimes neither took nor endangered human lives were cruel and unusual punishment (Ralph v. Warden, 438 F.2d 786 [4th Cir. 1970]). In response to the Fourth Circuit's ruling, attorneys general of Maryland, the Carolinas, and Virginia filed appeals with U.S. Supreme Court. Third, Arkansas governor Winthrop Rockefeller commuted sentences of all the state's death row before leaving office and Pennsylvania's attorney general, Fred Speaker, unilaterally ordered the removal of the electric chair from the state's death chamber. Fourth, anti-death penalty bills reached the floor of state legislatures in California, Massachusetts, and the U.S. House of Representatives. And finally, by 1971, nine states had abolished the death penalty and four more had no one actively on death row. Collectively, these events added considerable pressure on the Supreme Court to issue a definitive ruling on the constitutionality of the death penalty (Haines 1996).

Less than two months after the *McGautha* ruling, partly in response to this pressure, the Supreme Court agreed to hear a package of four appeals and to issue a conclusive decision as to the constitutionality of capital punishment. All of these cases, in slightly different ways, challenged the constitutionality of their respective state's death penalty based on the belief that it violated the defendants' Eighth Amendment rights. Two of the cases, *Aikens v. California* and *Furman v. Georgia*, were murder cases. The other two, *Branch v. Texas* and *Jackson v. Georgia* were rape cases in which no life was taken. All of the defendants were black and all of the victims were white. The cases were consolidated as *Furman v. Georgia* (408 U.S. 238 [1972]).<sup>51</sup>

The legal challenges made by this group of cases were very similar to those in *McGautha*, and the issue at hand was whether capital punishment, *as administered under existing state law*, constituted cruel and unusual punishment. Attorneys for the appellants were well aware of the court's belief that the capital punishment, in of itself, was not cruel and unusual, so they elected to attack the respective states' implementation of capital punishment instead. The central question in *Furman* was whether the broad discretion afforded to capital juries was unconstitutional. LDF attorneys would specifically argue that the death penalty, as administered in the second half of the twentieth century, was inconsistent with evolving standards of decency resulting from its extremely infrequent application, extended delays between trial and execution that resulted in psychological torture of death-sentenced inmates, the arbitrariness and capriciousness of death noticed and death sentence cases, and the strong possibility of racial and economic bias in sentencing (see also Gottlieb 1961).

On June 29, 1972, approximately one year after the *McGautha* decision, the U.S. Supreme Court voted to strike down capital statutes that lacked sentencing guidelines for juries by the thinnest possible majority of five-to-four. This was somewhat surprising considering that the very same court—comprised of the same judges—had decided that the absence of sentencing standards was not a violation of the Fourteenth Amendment one year earlier in *McGautha*. Each Supreme Court Justice wrote a separate opinion,

<sup>&</sup>lt;sup>51</sup> The California Supreme Court struck down the death penalty as a violation of the California state constitution before the U.S. Supreme court ruled on the case (see *People v. Anderson*, 493 P.2d 880 [Cal. 1972], *cert. denied*, 406 U.S. 958). Aiken's sentence was commuted to life imprisonment and the appeal was dropped (*Aiken v. California*, 406 U.S. 813 [1971]). Nine months later, however, a public referendum was passed reinstating the death penalty in California.

citing different reasons for their decision, in what would be the longest Supreme Court opinion ever written. Although not entirely clear, it appears that the court believed that the Eighth and Fourteenth Amendments applied to capital punishment in very different ways. Whereas the Fourteenth Amendment argument presented in *McGautha* applied to the process of sentencing defendants without guidelines, the Eighth Amendment argument in *Furman* applied to the actual outcome. While the court believed that standardless juries were permissible under the due process clause of the Fourteenth Amendment, the product of this process was in violation of the Eight Amendment. In other words, the court appeared to believe that a constitutional process could still produce an unconstitutional result.

Only Justices William Brennan and Thurgood Marshall believed that capital punishment could not be considered constitutional in any form. Justice Brennan thought the capital punishment was an affront to human dignity because of its severity, arbitrariness and infrequency, incompatibility with contemporary standards of decency, and excessiveness. Justice Marshall, too, believed that capital punishment was excessive and that the same goals of capital punishment could be achieved with long-term imprisonment. He also thought that society had morally and socially progressed to the point that it viewed the death penalty as cruel and barbaric. Recognizing public support of capital punishment, Justice Marshall argued that this support was based upon citizens' lack of information about problems with the actual administration of the death penalty. In particular, he believed that if the public knew that the death penalty was no more of an effective deterrent than life imprisonment, was frequently imposed in a discriminatory manner, and that there was a strong likelihood that innocent persons are executed, then they would find the death penalty morally objectionable as well (Bedau 1997b, pp. 189– 95; Haines 1996).

The other Justices joining the majority, however, did not agree that capital punishment was inherently unconstitutional. Justice William Douglas rejected the "involving standards of decency" claim and cautioned that such an argument was dangerous in that it made the Eighth Amendment historically relative, potentially making the other nine amendments susceptible to reinterpretation as well. According to Justice Douglas, the flaw of the death penalty lies in the fact that it was so discretionary that discrimination against the poor and racial/ethnic minorities was inevitable. He believed that the absence of sentencing standards provided juries with the opportunity to act in a discriminatory manner and produce a pattern of death sentences that reflect the current social hierarchy that disadvantages individuals at the bottom of this hierarchy.<sup>52</sup> Justice Potter Stewart also argued that it was the arbitrary and capricious nature of the death penalty that violated the Eighth and Fourteenth Amendments, not capital punishment, per se. But while recognizing the randomness of the death penalty, he also believed that this problem was correctable and that racial discrimination in the administration of the death penalty had *not* been successfully demonstrated. In fact, Justice Stewart believed that current death sentencing patterns were completely unpredictable and could not be explained by any rational process, including intentional discrimination. Justice Byron White argued that the infrequent and arbitrary application of the death penalty undermined the deterrent effect of the punishment, therefore negating any legitimate

<sup>&</sup>lt;sup>52</sup> "We know that the discretion of judges and juries in imposing the death penalty enables the penalty to be selectively applied, feeding prejudices against the accused if he [*sic*] is poor and despised, and lacking political clout, or if he [*sic*] is a member of a suspect or unpopular minority, and saving those who by social position may be in a more protected position." *Furman v. Georgia*, 408 U.S. 238, 255 (Douglas, J., concurring).

function that the penalty might otherwise serve. According to Justice White, if the death penalty serves no other legitimate state function than the extinction of human life, it is cruel and unusual. Similar to Justices Douglas and Stewart, Justice White did not believe that capital punishment was inherently cruel and unusual, only that the justification for the punishment lay in its effectiveness in controlling crime and serving justice (Haines 1996). This effectiveness, then, was undermined by the lack uniformity in death sentencing decisions—a problem, in his opinion, that was correctable.

The immediate effect of *Furman* was that approximately 558 death row inmates had their sentences commuted to life imprisonment. Although there was widespread speculation by death penalty proponents that many of these inmates would kill again once released from prison, subsequent research revealed that only one of the 239 Furmancommuted inmates released from prison committed a second murder in the fifteen years following the *Furman* decision (see Marquart and Sorensen 1989). Following the ruling, many supporters and critics of capital punishment initially believed that *Furman* signaled the eventual abolition of the death penalty, although others were less certain (Banner 2002; Haines 1996). The fragile majority in *Furman*—only two of the Justices believed that the death penalty was inherently unconstitutional—left open the possibility for states to develop constitutionally acceptable capital statutes. In fact, in his dissent, Chief Justice Warren Burger suggested capital statutes that guided sentencing discretion, or eliminated discretion altogether for specific crimes, would meet the court's approval. In response to *Furman*, several state legislatures and the federal government crafted new capital statutes, adopting systems that either structured or eliminated discretion in capital sentencing.

On December 8, 1972, less than six months after *Furman*, Florida became the first state to officially restore capital punishment when Governor Reubin Askew signed new death penalty legislation (Haines 1996). Within six months, twelve more states had restored capital punishment and a total of thirty-five states and the federal government had reinstated capital punishment by 1976 (Banner 2002; Gillers 1980; Jacobs and Carmichael 2002).<sup>53</sup> Since the *Furman* decision left it unclear as to what type of capital punishment system would pass constitution muster, two forms of capital statues emerged: *mandatory* and *guided discretion*.

In an attempt to eliminate all arbitrariness from death penalty sentencing, several states enacted mandatory capital statutes requiring a death sentence for defendants convicted of certain types of offenses. As noted earlier, mandatory death sentences were a common feature of death penalty statutes in the first era of capital punishment, but they began to disappear in the mid-nineteenth century and had all but vanished by the time the Supreme Court began examining the constitutionality of the death penalty.<sup>54</sup> Recall that the original mandatory capital statutes were criticized because legislatures and judges feared that juries would rather find a defendant not guilty of the crime, although they actually believed that the defendant committed the crime, than convict the defendant if the jury believed that the defendant did not deserve a death sentence. Nonetheless, sixteen states responded to *Furman* by enacting mandatory death penalty laws: the Carolinas, Delaware, Idaho, Indiana, Kentucky, Louisiana, Mississippi, Nevada, New

<sup>&</sup>lt;sup>53</sup> The federal government passed legislation authorizing the death penalty for individuals convicted of aircraft hijacking resulting in death (Haines 1996).

<sup>&</sup>lt;sup>54</sup> A few instances of mandatory death penalties were present in Post-World War II America. Massachusetts authorized mandatory death sentences for persons convicted of rape-murders in 1951. Both Rhode Island and New York had laws that required death for individuals convicted of murder while currently serving prison sentences (Haines 1996). Although these laws were "on the books" in their respective states, it appears that they were rarely, if ever, applied to actual cases.

Hampshire, New Mexico, New York, Oklahoma, Rhode Island, Tennessee, and Wyoming (Haines 1996, p. 46). These statutes typically mandated the death penalty for the killing of law enforcement officials, murders-for-hire, murders committed by individuals already serving a life sentence, and multiple-victim homicides.

Other state legislatures elected to develop capital statutes with formalized sentencing guidelines—an approach that had been previously rejected by the Supreme Court in McGautha. These guided discretion statutes were of three basic types. The first type-adopted by Georgia, Illinois, Montana, Texas, and Utah-specified several aggravating circumstances and only permitted death sentences in cases where at least one of these factors was present. Most of the aggravating factors were borrowed from the ALI's Model Penal Code (see *supra*, note 49) and were very similar to the factors identified in the mandatory statutes. The second type of capital statute—developed by Arkansas, Colorado, Florida, and Nebraska—specified both aggravating and mitigating circumstances that judges and juries were required to consider. In these sentencing schemes, decision makers were required to weigh aggravating factors against mitigating factors, such as mental impairment, youthfulness at the time of the crime, emotional disturbance, et cetera, when deciding upon the appropriate sentence. Legislatures in California, Connecticut, Ohio, and Pennsylvania adopted the third type of guided discretion statute, which was actually a hybrid of mandatory and guided discretion systems. These sentencing schemes—sometimes referred to as "quasi-mandatory" statutes—also identified both aggravating and mitigating factors; however no "weighing" of these factors had to be done by the sentencer. Instead, these statutes required a death sentence in cases when aggravating, but no mitigating, factors were present.

Soon after the new death penalty laws were passed in their respective states, death rows began being populated by inmates condemned to death. Before any executions could occur, however, the Supreme Court had to decide whether these new sentencing schemes met the requirements of the *Furman* decision. As the differences in the newly minted capital sentencing schemes indicated, no one was entirely clear as to what type of capital statute, if any, would be considered constitutional.

Nearly four years after the *Furman* decision, the Supreme Court agreed to hear its first death penalty case: Fowler v. North Carolina, 428 U.S. 904 (1976). The case involved a defendant sentenced to death under North Carolina's new mandatory death penalty law. The North Carolina statute required a death sentence for defendants convicted of first-degree murder in which arson, rape, robbery, kidnapping, burglary, or any other felony was also committed or attempted.<sup>55</sup> Because mandatory death penalty laws were the most common type of post-Furman death penalty scheme adopted by states, *Fowler* would have widespread implications for inmates sentenced to death since the *Furman* decision. In a surprising twist, Solicitor General Robert Bork filed an amicus curiae brief in support of North Carolina's death penalty statute and cited a recent study by economist Isaac Ehrlich suggesting that the death penalty had a strong deterrent effect on homicide (see Ehrlich 1975). Opponents of capital punishment viewed Bork's involvement in the case as indication of the federal government's interest in-and support of—the death penalty (Haines 1996). Fowler, however, did not resolve the matter because the illness (and subsequent retirement) of Justice William Douglas left the court split four to four on the death penalty. Rather than rehear *Fowler* before a fully constituted court, the court agreed to rule on a package of five other cases after Douglas's

<sup>&</sup>lt;sup>55</sup> N.C. Gen. Statute § 14-17 (Cum. Supp. 1975).

successor had been confirmed. Three of the cases—*Gregg v. Georgia* (428 U.S. 123 [1976]), *Proffit v. Florida* (428 U.S. 242 [1976]), and *Jurek v. Texas* (428 U.S. 262 [1976])—involved statutes that authorized guided discretion, while the remaining two cases (*Woodson v. North Carolina*, 428 U.S. 280 [1976] and *Roberts v. Louisiana*, 428 U.S. 325 [1976]) involved mandatory death penalty laws. These five cases were considered "unusual" in two respects. First, unlike the vast majority of homicides at the time, each of the defendants in these cases had been sentenced to death for "felony murder," i.e., a homicide during the commission of another serious crime. Felony murders constituted less than one-third of all homicides during this period (Haines 1996). Second, all of the victims in these cases were white, although the majority of homicide victims in the United States were black. It was believed that these specific cases were chosen because no "side issues" (e.g., racial discrimination, mental retardation, *et cetera*) and the difference among them reflected the differences in the actual death penalty laws from their respective jurisdictions (Haines 1996).

Attorneys representing Gregg, Proffit, and Jurek filed briefs arguing that the new guided discretion death penalty laws were still unconstitutional under *Furman* because the aggravating factors specified in various statutes were defined too vaguely to rationalize the sentencing process *and* they had only minimal impact on one of the many decision points in the capital punishment process (Haines 1996). These attorneys also argued that, regardless of the procedural modifications made to pre-*Furman* death penalty statutes, capital punishment violated contemporary standards of decency and, therefore, was in violation of the Eighth Amendment.

On the other side of the aisle, state attorneys from Georgia, Florida, and Texas claimed that the remaining discretion in their states' new death penalty laws was necessary and justifiable, that racial discrimination was *not* a significant factor in the capital charging-and-sentencing process, and that capital punishment was not inherently cruel and unusual punishment. In addition to the briefs filed by the state attorneys, Solicitor General Robert Bork, again, submitted an amicus brief claiming that the death penalty was not cruel and unusual punishment and that capital punishment was an effective deterrent to would be murderers (Haines 1996).

In the three guided discretion cases—collectively know as the *Gregg* decision the court voted seven-to-two that the death penalty for murder did not by itself violate the Eighth Amendment. The majority opinion (authored by Justice Stewart) stated that the rapid response by state legislatures to reenact death penalty legislation after *Furman*, along with recent public opinion poll evidence citing widespread support for the death penalty, clearly indicated that the American people did not view the death penalty as cruel, barbaric, or morally reprehensible (Justices Brennan and Marshall dissented). This, of course, was of little surprise because all but two of the justices held this belief in the *Furman* decision. More importantly, the court held that all three of the capital statutes contained sufficient procedural reforms to warrant them constitutional under *Furman*. Interestingly, the court had no evidence suggesting that the new statutes eliminated arbitrariness and bias in capital sentencing; rather the court based its decision on whether the procedural reforms enacted in each statute were capable of producing outcomes different from those produced under the pre-*Furman* statutes. The death penalty statutes offered under *Gregg*, *Proffit*, and *Jurek*, although all guided discretion statutes, varied considerably. Georgia's statute in *Gregg* offered three procedural reforms that the court believed would be effective in producing fair death sentences: (1) a bifurcated hearing which split guilt/innocence and sentencing decisions into two separate phases; (2) a list statutory aggravating factors that judges and juries were required to consider; and (3) automatic review of death sentences by the Georgia Supreme Court to determine the appropriateness of the sentence (see also Chapter Five). Although Georgia's death penalty law did not explicitly list mitigating factors for the sentencer to consider, the statute allowed such factors to be presented to, and considered by, the sentencer.

Similar to Georgia's statute, Florida's statute under review in *Proffit* specified a list of aggravating circumstances that the sentencer must consider when deciding the appropriate punishment, required that capital trials to be bifurcated, and mandated the Florida Supreme Court to review all death sentences.<sup>56</sup> Florida's statute, however, differed from the Georgia's in two important ways. First, Florida law specified a list of mitigating factors that the trial court was to consider and weigh against the aggravating circumstances of the case when deciding upon an appropriate sentence. Second, Florida's statute allowed the trial judge to override the sentencing recommendation of the jury. Although the sentencing decision of the jury was presumed to be correct, the trial

<sup>&</sup>lt;sup>56</sup> Whereas Georgia's death penalty statute required the Georgia Supreme Court to consider specific factors when determining the appropriateness of a death sentence (see Chapter Five), Florida's death penalty law did not. The U.S. Supreme Court deemed this omission to be acceptable, however, because the Florida Supreme Court had previously defined several criteria necessary for comparative case review (see *State v. Dixon*, 283 So. 2d 1, 10 [Fla. 1973]). Shortly after the Florida's new death penalty statute was validated in *Proffit*, the U.S. Court of Appeals ruled that comparative proportionality review of death penalty cases was reserved for the Florida state courts, and should not be intruded upon by the federal judiciary (*Spinkellink v. Wainwright*, 578 F.2d 582, 613 [5th Cir. 1978]).

judge could increase or decrease the severity of the punishment suggested by the jury if she or he believed that jury's recommendation was inappropriate.

The death penalty statute developed by Texas and reviewed in *Jurek* represented the hybrid death sentencing system mentioned earlier. Similar to both Georgia and Florida, Texas required separate hearings for the determination of guilt/innocence and sentencing and all death sentences were subject to review by a court with statewide jurisdiction (i.e., the Texas Court of Criminal Appeals). Texas's statute differed from the other two sentencing systems in that following a determination of guilt during the first phase, the decision-maker was required to answer three specific questions in the penalty phase: (1) whether the conduct of the defendant that caused the death of the deceased was committed deliberately and with the reasonable expectation that the death of the deceased or another would result; (2) whether there is a probability that the defendant would commit criminal acts of violence that would constitute a continuing threat to society; and (3) if raised by the evidence, whether the conduct of the defendant in killing the deceased was unreasonable in response to the provocation, if any, by the deceased.<sup>57</sup> If the sentencer answered yes beyond a reasonable doubt to all three of these questions, a death sentence was mandatory. The majority in *Gregg* believed that Texas's death penalty law did not constitute a mandatory sentencing system because the "future dangerousness" of the defendant (question #2) required the sentencer to consider both the aggravating and mitigating factors of the case. According the court, it was the consideration of the unique *legitimate* aspects of the case that was a critical feature of a constitutionally permissible capital statute.

<sup>&</sup>lt;sup>57</sup> Tex. Code Crim. Proc. Art. 37.071 (Supp. 1975 – 1976).

The two remaining cases reviewed by the Supreme Court-Woodson v. North Carolina and Roberts v. Louisiana—involved mandatory capital statutes. As noted earlier, the constitutionality of mandatory sentencing systems was the central issue in *Fowler*, however Justice Douglas's illness and subsequent retirement from the court left the court dead-locked. States adopting mandatory death penalty laws drafted legislation that narrowly defined capital murder and required the death penalty for all individuals convicted of such crimes. Attorneys for Woodson and Roberts argued that these mandatory statues failed to adequately address the problems of arbitrariness and capriciousness in the administration of the death penalty that the Supreme Court identified in *Furman*. Similar to the arguments made by anti-death penalty attorneys in the guided discretion cases, attorneys for Woodson and Roberts argued that mandatory sentencing schemes did not eliminate discretion, but merely shifted discretion from juries to other actors (and other stages) in the capital charging-and-sentencing process. Under mandatory systems, prosecutors still retained broad discretion in charging defendants and making plea offers, judges wielded broad discretion in accepting or rejecting plea bargains in potentially capital cases, and juries—although denied discretion in the penalty phase—were still able to arbitrarily or discriminately decide who received the death sentences by deciding whether or not to convict the defendant of capital murder.

In a five-to-four decision, the Supreme Court held that mandatory death penalty laws were unconstitutional, as they violated the Eighth Amendment.<sup>58</sup> In particular, the court claimed that these sentencing schemes exhibited three major flaws. First, these statues violated contemporary standards of decency because, historically, most states had

<sup>&</sup>lt;sup>58</sup> Justices Stewart, Powell, Stevens, Brennan (concurring in judgment), and Marshall (concurring in judgment) formed the majority, with Chief Justice Burger and Justices Whites, Rehnquist, and Blackmun dissenting.

abolished mandatory penalties and preferred the "individualization" of punishment. These new mandatory laws, according to the court, were in response to crafting constitutionally acceptable statutes after *Furman*, rather than society's preference for them. Second, similar to the criticism of mandatory schemes in the mid-nineteenth century, the court believed that these laws still allowed considerable unguided discretion by juries by simply allowing them to refuse to convict the defendant of capital murder. And third, mandatory sentences failed to allow juries to consider the unique culpability of the defendant. The court believed that not all individuals convicted of capital murder deserved to die, therefore sentencers needed to be authorized to take into account the particular aspects of the defendant and the crime when deciding upon the appropriate punishment.

With the death penalty laws in Georgia, Florida, and Texas deemed constitutional by *Gregg*, lifting the official moratorium on executions established in *Furman*, both proponents and opponents of the death penalty believed that executions would be back in full swing. To the surprise of both sides of the debate, however, executions remained relatively infrequent until the mid-1980s. The first execution to take place in America since the *de facto* moratorium began in the middle of 1967 would be that of Gary Gilmore in Utah. Gilmore, who waived his judicial appeals and volunteered to be executed, was shot to death by a Utah firing squad on January 17, 1977, nearly ten years after the last execution in the United States. Executions per year would remain in the single digits for the next seven years: there were no executions in 1978, two in 1979, none in 1980, one in 1981, two in 1982, and five in 1983.

A major factor contributing to the infrequency of executions was the ambiguity of the *Gregg* decision. As in the *Furman* ruling, the Supreme Court left many important questions unanswered in *Gregg*. In attempt to reconcile the *Gregg* ruling with the *McGautha* decision that held single-verdict juries and the absence of sentencing standard were not in violation of the Fourteenth Amendment, the court stressed in *Gregg* that these two procedural modifications were not an absolute requirement of death penalty systems, merely that they were constitutional (Haines 1996). What, then, did the constitution require of death penalty statutes? Over the next several years following *Gregg*, the Supreme Court would continually wrestle with this and related questions, ultimately limiting the scope of these death statutes and expanding due process rights (Foley 2003, Chapter 4).

Shortly after the *Gregg* ruling, the Supreme Court considered the constitutionality of the death penalty for crimes in which no life was taken. Although this issue had been previously brought before the court, it had repeatedly sidestepped the issue. In *Coker v. Georgia* (433 U.S. 584 [1977]) and *Eberheart v. Georgia* (433 U.S. 917 [1977]), the court ruled that the death penalty was unconstitutional for the crimes of rape and kidnapping, respectively, when no killing occurred (see also Chapter Five). The following year, the court ruled that statutory restrictions on the types of mitigation evidence that a defendant could present on her or his behalf during sentencing were unconstitutional (*Lockett v. Ohio*, 438 U.S. 586 [1978])<sup>59</sup> and that the sentencer must find

<sup>&</sup>lt;sup>59</sup> Sandra Lockett was a getaway driver in an armed robbery of a pawn shop and while she was in the car, one of her accomplices killed the robbery victim. At her sentencing hearing, Lockett attempted to introduce mitigation evidence that she was neither the cause of the victim's death nor had she intended the victim's death. At the time of her trial, the Ohio death penalty statute enumerated three mitigating circumstances that the jury was allowed to consider: (1) the victim helped cause her or his own death; (2) the defendant acted under duress, coercion, or strong provocation; and (3) the offense primarily resulted from the defendant's mental impairment. The *Lockett* opinion included an limiting footnote stating that the

that there are legally valid aggravating circumstances to warrant a death sentence (*Presnell v. Georgia*, 439 U.S. 637 [1978]). Over the next few years the court ruled that juries were prohibited from basing death sentences on vaguely defined aggravating circumstances (*Godfrey v. Georgia*, 446 U.S. 420 [1980]), that a defendant could not be sentenced to death on retrial when the original trial resulted in a life sentence (*Bullington v. Missouri*, 451 U.S. 430 [1981]), and that a death sentence was unconstitutionally excessive for defendants who were not killers and who did not intend to kill (*Enmund v. Florida*, 458 U.S. 782 [1982]). In fact, in the first seven years after *Gregg*, the Supreme Court ruled in favor of 14 out of 15 death-sentenced inmates whose appeals were fully heard (Haines 1996). This trend, however, would soon end.

Beginning in 1983, the court's decisions in a series of death penalty cases signaled its growing uneasiness with telling states how to constitutionally administer the death penalty (Foley 2003). One of the first of these cases was *Barefoot v. Estelle* (463 U.S. 880 [1983]), in which the court ruled that it was not unconstitutional for juries to base their life or death decision on unreliable or questionable evidence regarding the defendants' "future dangerousness" *and* that federal circuit courts were not required to

holding did not limit "the traditional authority of the court to exclude, as irrelevant, evidence not bearing on the defendant's character, prior record, or the circumstances of the offense" (438 U.S. at 604 n.12). In *Franklin v. Lynaugh*, (487 U.S. 164 [1988]), the court held that a defendant did not have an Eighth Amendment right to a jury instruction allowing the jury to consider as a mitigating factor at the sentencing phase any residual doubt concerning the defendant's guilt that carried over from the guilt phase. Most states, however, tend allow the introduction of a broad range of mitigation evidence at the penalty phase and the Supreme Court has held that juries must be allowed "to exercise wide discretion," but states must also ensure that the process is "neutral and principled as to guard against bias or caprice in the sentencing decision" (*Tuilaepa v. California*, 512 U.S. 967, 973–74 [1994]).

Zimring (2003, pp. 53–55) suggests, somewhat paradoxically, that the *Lockett* decision has actually harmed capital defendants by setting the stage for subsequent rulings allowing the inclusion the victim impact evidence in the penalty phase of death penalty trials. According to Zimring, the penalty phase of capital trials have become "status competitions" between the offender and the victim's family, and therefore a "zero-sum" game between two private parties rather than a dispute between the defendant and the state. Under these circumstances, Zimring believes that jurors may feel that voting for mercy is, in effect, voting against the victim (cf. Kleck 1991).

grant stays of execution to death sentenced inmates until all of their appeals were completed. That same year the court would reverse its previous position in *Godfrey* and rule that juries could rely on an aggravating circumstance not specified in the actual death statute to sentence a defendant to death (*Wainwright v. Goode*, 464 U.S. 78 [1983]) and that trial judges may constitutionally overrule a jury's recommendation of a life sentence and impose a sentence of death (*Barclay v. Florida*, 463 U.S. 939 [1983]). The following year, the court ruled that proportionality review of death sentences to insure that the penalty was being applied in a consistent manner was not mandatory (*Pulley v. Harris*, 465 U.S. 37 [1984])<sup>60</sup> and it significantly narrowed the scope of ineffective assistance of trial counsel claims that could be made by defendants condemned to death (*Strickland v. Washington*, 466 U.S. 668 [1984]).

While all of these decisions were damaging to the death penalty defense bar, perhaps the biggest blow occurred in 1987 when the Supreme Court issued its ruling in *McCleskey v. Kemp*, (481 U.S. 279 [1987]). Prior to *McCleskey*, death penalty opponents primarily focused their efforts on two constitutional challenges to the capital-sentencing systems *as they were being applied* in the post-*Furman* era: (1) the comparative excessiveness of death sentences and (2) the lack of comparative-proportionality review in the supreme courts of some death penalty states (Baldus *et al.* 1990, p. 306). As noted above, the Supreme Court's early rulings were favorable to the death penalty defense bar, but became increasingly unsympathetic to these challenges starting in 1983. Not only did the court begin to refuse to hear many of these appeals, but it also began narrowly redefining its prior rulings that extended constitutional protections to defendants

<sup>&</sup>lt;sup>60</sup> The same year as the *Pulley* ruling, in *Cape v. Francis* (741 F.2d 1287 [11th Cir. 1984]), the U.S. Court of Appeals ruled that adequate proportionality review by the Georgia Supreme Court's did *not* require the Court to consider decisions in which a life sentence had been imposed.

sentenced to death. The court did, however, suggest a third avenue by which anti-death penalty attorneys could challenge the constitutionality of post-Furman statutes: racial discrimination (Baldus *et al.* 1990; Haines 1996). Although the majority in *Furman* believed that claims of racial discrimination had been unproven, their opinions suggested that such proof of purposeful discrimination would constitute cruel and unusual punishment.<sup>61</sup> Indeed, in the 1960s and 1970s, the Supreme Court had frequently applied the equal protection clause of the Fourteenth Amendment to condemn state discrimination on racial grounds, so there was a strong possibility that these decisions would also extend to the death penalty if the burden of proof could be meet (Baldus et al. 1990, p. 308; Blume, Eisenberg, and Johnson 1998). Almost immediately following the Furman decision, death penalty opponents began stressing the importance of "developing an arsenal of solid social science evidence on pivotal issues" (Haines 1996, p. 47) because very few empirical studies on capital punishment had been conducted at the time (see, e.g., Bedau 1964, 1965; Carter and Smith 1969; Garfinkel 1949; Johnson 1941; Koeninger 1969; Wolfgang, Kelly, and Nolde 1962). To be sure, death penalty opponents did not believe that this evidence would influence judges' decisions and settle the myriad of legal issues; however, they did believe that social science evidence would provide a solid foundation on which judges could justify their decisions (Haines 1996).<sup>62</sup> The *McCleskey* case provided the opportunity for such evidence to be heard.

Warren McCleskey, a black male, was sentenced to death in Atlanta, Georgia in 1978 for the murder of a white police officer, Frank Schlatt, while attempting to

<sup>&</sup>lt;sup>61</sup> In fact, in *Zant v. Stephens* (462 U.S. 862 [1983]), the court held that race-of-defendant or race-of-victim discrimination, whether overt or covert, is constitutionally impermissible.

<sup>&</sup>lt;sup>62</sup> In *Furman*, both Justice Douglas and Justice Marshall cited social scientific studies conducted in California (Carter and Smith 1969), New Jersey (Bedau 1964), Oregon (Bedau 1965), and Texas (Koeninger 1969) documenting racial disparities in the administration of capital punishment.

burglarize a furniture store. Following his conviction, McCleskey claimed that his death sentence was unconstitutional because it had been imposed in discriminatory manner, based on the race of the defendant and of the victim. McCleskey's appeal was handled by attorneys working for the LDF, the very organization that had been challenging the racially discriminatory application of the death penalty for over thirty-five years. McCleskey's attorneys presented evidence from a large-scale, methodological sophisticated statistical study indicating that Georgia's death penalty purposefully discriminated against black defendants accused of murdering white victims, therefore violating their client's Fourteenth Amendment right to equal protection. McCleskey's attorneys also argued that the discriminatory application of the death penalty constituted an arbitrary and capricious application of the death sentence and violated their client's Eighth Amendment right to protection from cruel and unusual punishment (Baldus *et al.* 1990).

In particular, the evidence presented in *McCleskey* was based on data collected for two separate studies on Georgia's capital charging-and-sentencing system (see also Chapter Four). The first study—The Procedural Reform Study (PRS)—was conducted solely for academic purposes, receiving financial support from the National Institute of Justice, the University of Iowa Law School, and Syracuse Law School's Center for Interdisciplinary Legal Studies. The PRS contained data on 700 murder convictions occurring in Georgia from 1969 to 1978 (156 pre-Furman cases and 594 post-Furman cases) and examined the last two decision phases in the capital charging-and-sentencing process: the prosecutor's decision to seek the death penalty following a conviction for a capital crime *and* the jury's decision to impose a life or death sentence after the penalty trial. The second study—The Charging and Sentencing Study (CSS)—was sponsored by the LDF through a grant from the Edna McConnell Clark Foundation, with the expectation that the study would be used to challenge the constitutionality of Georgia's capital-sentencing system (Baldus *et al.* 1990, p. 44; Haines 1996, p. 76). The CSS gathered data on a random sample of 1,066 defendants (out of a total of 2,484 defendants) convicted of murder or voluntary manslaughter in Georgia from 1973 to 1979 and examined five decision points in the capital punishment process: grand jury indictment; prosecutorial plea bargaining; guilt/innocence trial; prosecutorial discretion in seeking the death sentence after a guilty verdict; and jury penalty trial (Baldus *et al.* 1990, p. 45). Both studies were conducted under the direction of University of Iowa Law Professor David Baldus, considered an expert in the quantitative analysis of legal phenomena—particularly discrimination cases. The two studies, although differing in scope and time period under consideration, had over 300 cases in common (Baldus *et al.* 1990, p. 45).

Baldus and colleagues' data demonstrated that prosecutors sought the death penalty in 70 percent of cases involving black defendants and white victims, 32 percent of cases involving both white defendants and victims, 19 percent of cases involving white defendants and black victims, and 15 percent of cases involving both black defendants and victims. With respect to jury discretion, juries imposed the death sentence in 22 percent of cases involving black defendants and white victims, 8 percent of cases involving white defendants and victims, 3 percent of cases involving white defendants and black victims, and one percent of cases involving both black defendants (p. 315). The CSS, which examined over 230 legitimate and illegitimate case characteristics, constituted the bulk of the LDF's evidence on Georgia's capital punishment system. The core analysis of the CSS data simultaneously examined 39 case factors and discovered that defendants convicted of murdering white victims were 4.3 times more likely to be sentenced to death than defendants convicted of murdering black victims with similar levels of culpability (p. 316). Identical to the results from the CSS, analysis of the PRS data also revealed that—after simultaneously controlling for 23 case factors—defendants convicted of murdering white victims were 4.3 times more likely to receive a death sentence than defendants convicted of murdering black victims (p. 143). According Baldus *et al.*, "the race-of-victim effects in death sentencing observed among defendants indicted for murder were attributable principally to prosecutorial decisions made both before and after [guilt/innocence] trial" (p. 328). That is, prosecutors' decisions to seek the death penalty and make plea offers to defendants benefited killers of blacks.

The statistical evidence presented in *McCleskey* was twice rejected by the lower federal courts in Georgia. First, Federal District Court Judge Owen Forrester rejected McCleskey's claims, doubting the validity of the data, the statistical procedures employed to analyze the data, and the validity of the inferences drawn from such procedures (*McCleskey v. Zant*, 580 F. Supp. 338 [N.D. Ga. 1984]) (for a detailed discussion of Judge Forrester's critique, see Baldus *et al.* 1990, pp. 340–42, 450–78). The LDF's appeal to the Eleventh Circuit Court of Appeals was also rejected by a nineto-three vote. Although the court did not question the validity of the data, it believed that the statistical analyses were: (1) insufficient to demonstrate that racially bias decisionmaking was present in McCleskey's particular case; (2) incapable of taking into consideration the countless quantitative and qualitative differences between capital crimes and capital defendants; and (3) "insufficient to demonstrate discriminatory intent or unconstitutional discrimination in the Fourteenth Amendment context, [and] insufficient to show irrationality, arbitrariness and capriciousness under any kind of Eighth Amendment analysis" (*McCleskey v. Kemp*, 753 F.2d 891 [11th Cir. 1985]).<sup>63</sup>

The U.S. Supreme Court granted certiorari on April 22, 1987, and by a vote of five-to-four, the court affirmed the Eleventh Circuit's ruling rejecting the Eighth and Fourteenth Amendment challenges raised in McCleskey. Similar to the Court of Appeals, the majority in *McCleskey* believed the study was statistically valid, although Justice Powell-authoring the majority opinion-believed the data were incomplete and "failed to establish by a preponderance of the evidence that the data was [sic] trustworthy" (McCleskey v. Kemp, 481 U.S. at 288). Perhaps more damaging to McCleskey's case was the majority's belief that the standard methods for proving purposeful discrimination under the Fourteenth Amendment used in employment and jury discrimination cases did not apply in the capital-sentencing context (Baldus et al. 1990, p. 346; Gross and Mauro 1989). The majority held that capital-sentencing decisions and "the relationship of the statistics to that decision" were "fundamentally different" from others types of discrimination claims (McCleskey v. Kemp, 481 U.S. at 280). According to Justice Powell, the capital charging-and-sentencing process differed because it involved several relatively autonomous decision-makers and capital juries are unique in their composition

<sup>&</sup>lt;sup>63</sup> The court did decide, seven-to-five, to reverse the district court's decision and grant McCleskey a new trial because the prosecutor failed to inform the jury that a jailhouse informant who testified against McCleskey was promised favorable treatment by the state in exchanged for his testimony.

and are constitutionally required to consider innumerable factors about the defendant and the offense.

The court also rejected McCleskey's Eighth Amendment challenge on two grounds. First, the majority ruled that McCleskey failed to prove his claim that his death sentence was "comparatively excessive." The court argued that McCleskey could not prove that his sentence was excessive by merely identifying other similarly situated defendants who did not receive the death sentence. According to Justice Powell, "[t]he Constitution is not offended by inconsistency in results based on the objective circumstances of the crime" because there is substantial variation in many importance factors that influence the manner in which these circumstances are viewed (McCleskev v. Kemp, 481 U.S. at 307 n.28). As Baldus and colleagues (1990, p. 348) note, however, McCleskey never claimed that his sentence was excessive, merely that his sentence was likely the product of racial discrimination. Second, the court held that the evidence presented in *McCleskey* did not prove that racial considerations actually entered into the decision-making process; rather it only showed that race was a risk factor in some capital-sentencing decisions. According to the majority, although a discrepancy in sentencing appeared to be correlated with race, McCleskey did not demonstrate that race was the causal factor.

In addition to the majority's rejection of McCleskey's Eighth and Fourteenth Amendment challenges, it also cautioned that a ruling in favor of McCleskey's Eighth Amendment claim would result in a wave of Eighth Amendment challenges in both capital and non-capital cases on both racial and non-racial grounds (Baldus *et al.* 1990, p. 349) and concluded that McCleskey's arguments would be best presented to legislative bodies rather than the court.<sup>64</sup>

As with *Furman*, the court reached its decision in *McCleskev* with the thinnest possible majority. Justices Brennan, Blackmun, and Stevens all filed dissenting opinions, with Justice Marshall joining to varying degrees. Three of the dissenting justices— Brennan, Blackmun, and Marshall—believed that McCleskey established by a preponderance of evidence that race had been an influential factor in his *own* case. Justice Stevens believed that the evidence established a "strong probability" that McCleskey was sentenced to death because of race. Although Justices Brennan and Marshall held that the entire Georgia capital charging-and-sentencing process violated the Eighth Amendment, Justices Blackmun and Stevens did not believe that the evidence invalidated Georgia's capital punishment system. Justice Blackmun believed that the appropriate solution was to significantly narrow death eligibility to the most highly aggravated cases in which there was no evidence of a race effect. Justice Stevens suggested that the case be remanded to ascertain whether McCleskey's case was within the range of cases that presented an unacceptable risk that race played an important role in his particular case.

The court's decision in *McCleskey* marked the defeat of the last broad-based constitutional attack on the death penalty. After the ruling, many death penalty opponents believed that an ideological realignment of the Supreme Court would need to

<sup>&</sup>lt;sup>64</sup> According to Bright (1995a, p. 482), the U.S. Supreme Court's assertion in *McCleskey* that racial discrimination in the capital punishment system is an issue for state legislatures is ill advised. "[L]egislators respond to powerful interests. The poor person accused of a crime has no political action committee, no lobby, and often no effective advocate even in the court where his life is at stake...The constitutional buck of equal protection under law stops with the Supreme Court and with judges on lower courts throughout the land who have taken oaths to uphold the Constitution and the Bill of Rights even against the passions of the moment and the prejudices that have endured for centuries" (see also Cole 1994; but cf. Sunstein 1993).

occur if capital punishment was to be abolished as the result of litigation (Haines 1996). Over the next several years, the court displayed its continuing reluctance to interfere with the decisions of the lower courts in capital cases. The court would uphold the constitutionality of death sentences for accomplices who neither killed nor intended to kill (*Tison v. Arizona*, 481 U.S. 137 [1987]),<sup>65</sup> defendants with mental retardation (*Penry v. Lynaugh*, 492 U.S. 302 [1989]), and juveniles as young as sixteen (*Stanford v. Kentucky*, 492 U.S. 361 [1989]; *Wilkins v. Missouri*, 492 U.S. 361 [1989]).<sup>66</sup> It would

also rule that-despite their mandatory nature-laws requiring juries to impose the death

sentence if at least one aggravating circumstance and no mitigating factors were present

were constitutional (Blystone v. Pennsylvania, 494 U.S. 299 [1990])<sup>67</sup> and inmates

sentenced to death were not guaranteed the right to present new evidence proving their

innocence that had not been available at the time of the original trial (Herrera v. Collins,

506 U.S. 390 [1993]).<sup>68</sup>

<sup>&</sup>lt;sup>65</sup> In *Tison v. Arizona* (481 U.S. 137 [1987]), two co-defendants were sentenced to death although they were not at the scene when the murders took place. The court upheld their sentences, ruling that the death penalty was constitutional for individuals who neither murdered, attempted to murder, nor intend to murder when these individuals demonstrated a "reckless disregard for human life." The court's reckless disregard for human life standard, however, was extremely vague and *Tison* had the effect of significantly broadening the scope of individuals subject to the death sentence.

<sup>&</sup>lt;sup>66</sup> In 1988, the Supreme Court ruled that the death penalty was unconstitutional for juveniles under the age of 16 (*Thompson v. Oklahoma* 487 U.S. 815 [1988]). The common law tradition at the time of the adoption of the Eighth Amendment in 1791 permitted children over seven years old to be tried as adults for capital felonies, although a rebuttable presumption of incapacity existed for children between the ages of seven and fourteen depending on the child's maturity and experiences (Hale [1736] 1800).

<sup>&</sup>lt;sup>67</sup> The *Blystone* ruling was somewhat surprising considering that just three years earlier, in *Sumner v. Shuman* (483 U.S. 66 [1987]), the Supreme Court ruled that Nevada's capital punishment statute mandating the death penalty for individuals convicted of murder while already serving a life without the possibility of parole sentence was unconstitutional.

<sup>&</sup>lt;sup>68</sup> In *Herrera v. Collins*, the Court held that affidavits supporting the defendant's claim of innocence were insufficient to entitle the accused to federal habeas corpus relief from the death sentence. Particularly troubling to death penalty opponents was Chief Justice Rehnquist's remark, "due process does not require that every conceivable step be taken, at whatever cost, to eliminate the possibility of convicting an innocent person. To conclude otherwise would all but paralyze our system for enforcement of the criminal law" (p. 399).

Perhaps even more damaging to the capital defense bar than the Supreme Court's string of unfavorable rulings was the Anti-Terrorism and Effective Death Penalty Act of 1996, which included "the
Nearly twenty years would elapse between rulings by the Supreme Court that were favorable to the capital defense bar and that had broad application. On June 20, 2002, by a vote of six-to-three, the court reversed its earlier decision in *Penry v. Lynaugh* and ruled that it was unconstitutional to execute defendants suffering from mental retardation because it violated the Eighth Amendment ban on cruel and unusual punishment (*Atkins v. Virginia*, 536 U.S. 304 [2002]).<sup>69</sup> At the time of the *Penry* decision, only two states—Georgia and Maryland—prohibited the execution of mentally retarded individuals.<sup>70</sup> By the time of *Atkins*, eighteen states and the federal government had enacted laws prohibiting such executions.<sup>71</sup> The Supreme Court, however, allowed states to individually determine the evidentiary standard necessary for defendants to present an *Atkins* claim. While over half of death penalty states required defendants to meet the "preponderance of evidence" or "clear and convincing evidence" standard, other states required a "beyond a reasonable doubt" evidentiary standard. It has been argued that the "beyond a reasonable doubt" standard greatly increases the probability that

most far-reaching assault on habeas corpus since its expansion to the states in 1867" (Haines 1996, p. 198). The new law imposed a one-year time limit for filing petitions for habeas corpus following the completion of the inmates' state appeals *and* prohibits federal judges from overruling state court decisions in death penalty cases unless the those decisions clearly contradict established rulings of the Supreme Court.

<sup>&</sup>lt;sup>69</sup> Research also suggests that race may play a role in the execution of individuals with mental retardation. From 1976 to 2000, 34 inmates with mental retardation were executed: 22 blacks, 9 whites, and 3 Hispanics. Over 75 percent of the executions were for white victim cases (Death Penalty Information Center 2008; Reed 1993).

<sup>&</sup>lt;sup>70</sup> In *Penry*, the court held that execution of the mentally retarded did not violate the constitution, but mental retardation is a mitigating factor that jurors should consider when reaching a sentence in a capital case.

<sup>&</sup>lt;sup>71</sup> The eighteen states prohibiting executions at the time of *Atkins* were: Arizona, Arkansas, Colorado, Connecticut, Florida, Georgia, Indiana, Kansas, Kentucky, Maryland, Montana, Nebraska, New Mexico, New York, North Carolina, South Dakota, Tennessee, and Washington.

mentally retarded defendants will still be executed (see *Head v. Hill*, 587 S.E.2d 613 [Ga. 2003]).<sup>72</sup>

Four days following the Atkins decision the Supreme Court again reversed an earlier ruling (Walton v. Arizona, 497 U.S. 639 [1990]) and, by a vote of seven-to-two, held that juries, rather than judges, must decide the critical issues in a death penalty case (*Ring v. Arizona*, 536 U.S. 584 [2002]), including the existence of aggravating circumstances.<sup>73</sup> Following the *Furman* decision, nine states utilized judge-sentencing in one form or another: judges had sole sentencing discretion in Arizona, Idaho, and Montana; a three-judge panel was used in Colorado and Nebraska; and Alabama, Delaware, Florida, and Indiana authorized the trial judge to make a sentencing decision following a jury recommendation. The *Ring* decision invalidated the death sentencing laws in Arizona, Colorado, Idaho, Nebraska and Montana, and these states subsequently revised their statutes to comply with the ruling. Delaware and Indiana also revised their death penalty statutes following the *Ring* decision, although their laws were not deemed unconstitutional under the ruling. But while *Ring* clearly applied to all future death cases, the court was unclear as to whether the ruling would retroactively apply to everyone on death row sentenced under judge-sentencing statutes and, as a result, lower courts have

<sup>&</sup>lt;sup>72</sup> Following *Atkins*, the U.S. Court of Appeals for the Seventh Circuit held that life imprisonment for individuals suffering from mental retardation did not violate the Eighth Amendment (*Harris v. McAdory*, 334 F.3d 665 [7th Cir. 2003]).

<sup>&</sup>lt;sup>73</sup> A decade after *Walton*, in *Apprendi v. New Jersey* (530 U.S. 466 [2000]), the court ruled that a judge could not make findings that would increase a defendant's sentence beyond the maximum because such a finding would, in effect, be an additional conviction; therefore decisions that had the potential of increasing the sentence must be submitted to a jury and require proof beyond a reasonable doubt. The court, however, stated that its ruling in *Apprendi* did not apply to death penalty cases (see also *Blakely v. Washington*, 542 U.S. 296 [2004]). Three years after the *Ring* ruling, in a five-to-four decision, the court ruled that mandatory federal sentencing guidelines that allowed judges in enhance sentences based on facts not decided by juries violated the Sixth Amendment (see *United States v. Booker* 543 U.S. 220 [2005]). The court held that judges may only base sentences on facts decided by the jury. Any additional evidence not initially presented to juries that would increase the severity of a sentence must be presented to a jury at a separate hearing.

issued different interpretations on the retroactivity of *Ring*. The Supreme Court subsequently agreed to clarify the impact of *Ring*, and on June 24, 2004 the court ruled five-to-four that *Ring* did not apply retroactively to cases that had received final direct review (*Schriro v. Summerlin*, 542 U.S. 348 [2004]).<sup>74</sup>

With its rulings in *Atkins* and *Ring*, the Supreme Court went against its twentyyear trend of almost exclusively deciding in favor of states over capital defendants with respect to broader constitutional challenges to the administration of the death penalty. Perhaps most surprisingly, no "ideological realignment" had occurred—the court was considered more conservative at the time of *Atkins* and *Ring* than it was during the midto late-1980s and early-1990s. What had occurred, however, was increased public debate over specific issues concerning the administration of the death penalty and legislative responses to these debates (Dieter 1996; Haines 1996). It is almost certain that the growing number of death penalty states that prohibited the execution of mentally retarded individuals strongly influenced the court's decision in *Atkins* (Death Penalty Information Center 2008). Death penalty opponents also used the legal precedent established in *Atkins* to challenge the death penalty for individuals who were juveniles at the time of the crime: the death penalty for juveniles violates current standards of decency, constituting cruel and unusual punishment (Greenhouse 2004).<sup>75</sup> Citing strong public opposition to

<sup>&</sup>lt;sup>74</sup> Following *Ring*, many believed that ruling could possibly affect 800 death sentences in nine states (see Death Penalty Information Center 2008). It is unlikely, however, that the ruling would have influenced that many death sentences even if the Supreme Court had allowed *Ring* to be applied retroactively because the court only invalidated the death penalty laws of five states. The remaining four states that utilized judge sentencing but authorized juries to make sentencing recommendations (Alabama, Delaware, Florida, and Indiana) constituted nearly 80 percent of those 800 death sentences. Death sentences handed down in these states would have most likely been subject to *Ring* only if the jury recommended a life sentence, but the trial judge elected to impose a death sentence—a small percentage of the overall death sentences in those four states.

<sup>&</sup>lt;sup>75</sup> As of June 30, 2002, 83 juveniles were on death row across the United States. Seventy-five percent (75%) of these juveniles were 17 at the time of their crimes, the remaining 25 percent were 16 at the time

executing juveniles, as well as the fact that 19 of the 38 death penalty states and the federal death penalty did not allow individuals under the age of 18 at the time of their crime to be sentenced to death, and five other states set the minimum age for the death penalty at seventeen (Fagan and West 2005), opponents of the juvenile death penalty waged a massive public and legal campaign against the execution of juvenile offenders (Glass 2006).<sup>76</sup> On January 26, 2004, primarily in response to this tremendous public and legal pressure, the Supreme Court granted certiorari in a juvenile capital case involving a 17 year-old male sentenced to death in Missouri and announced that it would reconsider the constitutionality of the death penalty for defendants under the age of eighteen (Roper v. Simmons, 540 U.S. 1160 [2004]). Slightly over a year later on March 1, 2005, by a vote of five-to-four, the Court reversed the legal precedent established 15 years earlier in Stanford v. Kentucky and forbid the imposition of the death penalty on offenders who were under the age of 18 when their crimes were committed, holding that such a penalty violated the Eighth and Fourteenth Amendments (see Roper v. Simmons, 543 U.S. 551 [2005]).<sup>77</sup>

The modern era of the death penalty, spanning over thirty years and responsible for more than 1,000 executions,<sup>78</sup> has clearly been the most bazaar and unpredictable period since the colonies began executing individuals at the beginning of the seventeenth

of their crimes. Although two-thirds of juvenile death row inmates are members of minority groups (47 percent black and 18 percent Hispanic), two-thirds of the victims of these juvenile offenders are white (65 percent) (see Streib 2002b).

<sup>&</sup>lt;sup>76</sup> At the time of the *Roper* decision, nineteen states did not allow juveniles to be sentenced to death (California, Colorado, Connecticut, Illinois, Indiana, Kansas, Maryland, Missouri, Montana, Nebraska, New Jersey, New Mexico, New York, Ohio, Oregon, South Dakota, Tennessee, Washington, Wyoming), five states set the minimum age to 17 (Florida, Georgia, New Hampshire, North Carolina, Texas), and 14 states set the minimum age at 16 (Alabama, Arizona, Arkansas, Delaware, Idaho, Kentucky, Louisiana, Mississippi, Nevada, Oklahoma, Pennsylvania, South Carolina, Utah, Virginia).

<sup>&</sup>lt;sup>77</sup> Justices Kennedy, Souter, Ginsburg, Breyer, and Stevens (concurring) formed the majority, with Chief Justice Rehnquist and Justices O'Connor, Scalia, and Thomas dissenting.

<sup>&</sup>lt;sup>78</sup> On December 5, 2008, Joseph Gardner became the 1,136th inmate executed since the death penalty was reinstated in 1976 (Death Penalty Information Center 2008).

century. The period began with a crushing blow to death penalty abolitionists when, after sidestepping these issues for years, the Supreme Court decided in McGautha that neither single-verdict juries nor the absence of sentencing standards were unconstitutional under the Fourteenth Amendment. The following year, in *Furman*, the court would rule that the death penalty was unconstitutional as applied under the Eighth Amendment, mainly because of single-verdict juries and the absence of sentencing standards. The Furman decision resulted in the commutation of all death sentences in the country to life imprisonment and an official moratorium on the death penalty. Four years later in Gregg, the court approved three different "guided discretion" death penalty statutes enacted in Florida, Georgia, and Texas and ruled the mandatory death sentencing statutes enacted in Louisiana and North Carolina unconstitutional. Over the next several years, the court would extend many constitutional protections to capital defendants, only to repeal most of them a few years later. After receiving favorable rulings in the Supreme Court in 14 of 15 cases following Gregg, the capital defense bar would not receive a significant victory in the Supreme Court for nearly two decades.

Although death penalty abolitionists in the United States appeared to be losing ground in the modern era, the abolitionist movement was gaining significant momentum in the international community (Haines 1996; Whitman 2003).<sup>79</sup> Since the death penalty was reinstated in the United States after the *Gregg* decision, capital punishment has been either completely abolished or abolished for ordinary crimes in such notable countries as: Portugal ([1867] 1976), Denmark ([1933] 1978), Brazil (1979), Peru (1979), Norway ([1905] 1979), France (1981), The Netherlands ([1870] 1982), Argentina (1984),

<sup>&</sup>lt;sup>79</sup> Some of the first nations to abolish the death penalty include Venezuela (1863), San Marino (1865), and Costa Rica (1877).

Australia ([1984] 1985), Haiti (1987), Germany (1987),<sup>80</sup> New Zealand ([1961] 1989), Ireland (1990), Switzerland ([1942] 1992), Greece ([1993] 2004), Italy ([1947] 1994), Spain ([1978] 1995), Belgium (1996), Poland (1997), South Africa ([1995] 1997), Canada ([1976] 1998), United Kingdom ([1973] 1998), Chile (2001), Serbia (2002), Turkey ([2002] 2004), Mexico (2005), and Philippines (2006) (Amnesty International 2006; Simon and Blaskovich 2002).<sup>81</sup> Currently 89 countries have abolished the death penalty for all crimes, 10 nations abolished the death penalty for ordinary crimes, and 30 nations retain death penalty laws but have not carried out an execution in the past 10 years (Amnesty International 2006). In 2006, the United States was one of only twentysix nations to carry out executions (Amnesty International 2006). Moreover, just a few of these nations were responsible for the vast majority (90 percent) of these executions: China (1,010), Iran (177), Pakistan (82), Iraq (65), Sudan (65), and the United States (53) (Amnesty International 2006).<sup>82</sup>

Foreign nations have also been very critical of the U.S. Supreme Court's continuing dismissal of challenges to the execution of foreign nationals in the United States based on the Vienna Convention (Dieter 2003). At the close of 2003, there were approximately 118 foreign nationals from 30 different countries on death row in the United States, many of whom were not informed of their right obtain legal counsel from their home nations (Dieter 2003). The United States' continued use of the death penalty and its refusal to honor the rights foreign nationals established by international treaty has

<sup>&</sup>lt;sup>80</sup> West Germany abolished capital punishment in 1949.

<sup>&</sup>lt;sup>81</sup> Several nations retain the death penalty for exceptional crimes, such as crimes under military law. If a nation initially abolished its death penalty for ordinary crimes and then abolished its death penalty for all crimes at a later date, the year is listed in brackets.

<sup>&</sup>lt;sup>82</sup> According to Amnesty International, there were 1,591 reported executions worldwide in 2006. The 1,010 reported executions in China is an extremely low estimate, and it is suspected that the actual number may approach 8,000.

made many nations reluctant to extradite individuals arrested in their countries who are wanted for capital crimes in the United States (Bedau 1997b, pp. 246–48; Dieter 2003). Recall from Chapter One that, in 1999, the United Nations Commission on Human Rights passed a resolution calling for a moratorium death sentencing and asking all nondeath penalty nations to refuse to extradite suspects to countries that use capital punishment (Radelet and Borg 2000). The tension between the United States and the global community on the death penalty came to a head in 2002 when the president of Mexico, Vincenté Fox, cancelled a visit with President George W. Bush in protest of the execution of a Mexican national, Javier Suárez Medina, who was denied his rights under international treaty (Dieter 2003).<sup>83</sup>

The victories for death penalty opponents in *Atkins*, *Ring*, and *Roper*, as well as the broader success they have enjoyed in influencing public opinion and state legislatures, has been attributed to the reemergence of focused political activism (see Haines 1996) and the impact of social scientific scholarship on nearly every issue in the death penalty debate (Baldus 1995; Radelet and Borg 2000). As noted earlier, death penalty opponents recognized the need to collect solid social scientific evidence with respect to the actual administration of the death penalty in the early 1970s, and believed that such evidence would provide judges with a solid foundation on which to base their decisions. Death penalty opponents, however, never prevailed in a racial discrimination claim in a capital case, although the court has found racial discrimination in other types of cases with far less evidence (Blume *et al.* 1998). While several scholars have argued

<sup>&</sup>lt;sup>83</sup> Although Mexico officially abolished its death penalty in 2005, the last execution carried in the country was in 1937. Mexican Foreign Minister Jorge Castaneda noted that the forty-five Mexican nationals on death rows in the United States are "an important strain on bilateral relations" between Mexico and the United States (Shapiro 2001, p. 14).

that the court is unreceptive of social science evidence of racial discrimination in capital cases (see, e.g., Acker 1993; Ellsworth 1988; Gross 1985; Radelet and Vandiver 1986), neither *McCleskey* nor any other subsequent decision of the court precludes capital defendants from raising similar race-based challenges (Blume et al. 1998; Sorensen and Wallace 1999). The major obstacle to these challenges, however, is the level of proof necessary to "trigger the imposition of a burden on county-level decision makers" (Blume et al. 1998, p. 1808). In Furman, Justice Douglas and Justice Marshall cited numerous empirical studies documenting racial and economic disparities in the imposition death penalty (see Bedau 1964, 1965; Carter and Smith 1969; Koeninger 1969; Wolfgang et al. 1962). When the Supreme Court rejected McCleskey's racial discrimination claim by the thinnest of margins (five-to-four), three of the dissenting Justices-Brennan, Blackmun, and Marshall—believed by a preponderance of evidence that race had been an influential factor in McCleskey's particular case. The fourth dissenting judge—Justice Stevens believed that the evidence presented by McCleskey established a "strong probability" that his death sentence was attributable to his race. The failure of racial discrimination claims in the Supreme Court did not result from the majority of court's belief that social scientific evidence was invalid, inappropriate, or unreliable, but rather how "stark" such evidence needed to be.<sup>84</sup> The capital defense bar may have also erred in forgoing a well-

<sup>&</sup>lt;sup>84</sup> Justice Powell, writing for the majority in *McCleskey*, believed that stronger evidence of racial discrimination would warrant a reversal of a death sentence under similar circumstances. Years after the *McCleskey* decision, Justice Powell admitted that he had little understanding of statistics and he had come to believe that the death penalty continues to operate in an arbitrary and capricious manner and regretted his vote (Jeffries 1994, pp. 439, 451). Justice Antonin Scalia, on the other hand, did not agree with Justice Powell's position concerning the strength of statistical evidence demonstrating racial discrimination (Baldus *et al.* 1994b). Scalia, who also joined the majority in *McCleskey*, wrote a memorandum to then-Justice Marshall stating:

I do not share the view, implicit in the opinion [of Justice Lewis Powell], that an effect of racial factors upon sentencing, if it could be shown by sufficiently strong statistical evidence, would require reversal. Since it is my view that the unconscious operation of

organized public education campaign, opting, instead, to present the empirical evidence directly to the courts. The racial discrimination claims presented in *McCleskey* may have been successful had the evidence been used to attack the death penalty on the legislative and public education fronts (but see Bobo and Johnson 2004; Bright 1995a).

The constitutional challenge raised in *Atkins* was largely successful because of the public education and legislative lobbying efforts of anti-death penalty activists opposing the execution defendants with mental retardation. A tremendous amount of scientific evidence—indicating that capital defendants suffering from mental retardation have "diminished capacities" due to their impaired brain functioning—was used to pressure state legislatures to prohibit the execution of defendants with mental retardation. The court's recent decision in *Roper*, invalidating the death penalty for juveniles, primarily resulted from public pressure over the issue and the successful repeal of capital punishment for juveniles in 19 of 38 states with death penalty laws. Opponents of the juvenile death penalty also heavily relied on scientific evidence suggesting that, similar to individuals with mental retardation, juveniles have impaired brain functioning that contributes impulsive and irrational behavior and the inability to appreciate the gravity of the offense committed (Streib 2003, pp. 106–108).<sup>85</sup>

irrational sympathies and antipathies, including racial, upon jury decisions and (hence) prosecutorial decisions is real, acknowledged in the decisions of this court, and ineradicable, I cannot honestly say that all I need is more proof (Davis 1998, p. 50).
<sup>85</sup> At the time of the *Roper* decision, the following organizations had called for a repeal of the juvenile death penalty: American Academy of Child and Adolescent Psychiatry, American Academy of Psychiatry and the Law, American Bar Association, American Medical Association, American Psychiatric Association, American Society for Adolescent Psychiatry, and National Mental Health Association.

Fifteen years earlier, writing for the majority in *Stanford*, Justice Scalia rejected scientific evidence that 16-year-olds were not "adequately responsible or significantly deterred" by the death penalty, claiming that "socioscientific, ethicoscientific, or even purely scientific evidence [was] not an available weapon" against the juvenile death penalty and the Court had "no power under the Eighth Amendment to substitute [its] belief in [] scientific evidence for the society's apparent skepticism" (492 U.S. at 378).

To be sure, "traditional" scientific evidence such as the brain functioning of individuals with mental retardation played a pivotal role in the *Atkins* and *Roper*. Nonetheless, social science research also played an important role in the debate over the execution of individuals with mental retardation and juveniles. Social science evidence documenting the public's growing uneasiness with the execution of individuals with mental retardation figured prominently in *Atkins* and evidence of public disapproval with executing juveniles had a strong influence in *Roper*.<sup>86</sup> In fact, social science research has significantly influenced the entire death penalty debate in the current era (Baldus 1995; Radelet and Borg 2000). The next chapter discusses the major areas of the death penalty debate where social scientific scholarship has had a significant impact.

<sup>&</sup>lt;sup>86</sup> In the *Roper* decision, the Supreme Court held that although *Stanford v. Kentucky* rejected the proposition that the Constitution bars capital punishment for juvenile offenders younger than 18, a national consensus had developed against the execution of juveniles since *Stanford*.

### **Chapter 3: Social Science and the Debate Over Capital Punishment**

Legal scholars have noted that social science research is changing the way Americans debate the death penalty (Baldus 1995; Bedau 1997b). Social scientists have been conducting empirical research on various aspects of the death penalty for more than 75 years, and over the last 30 years, socio-legal research has figured prominently into the debate over the appropriateness and effectiveness of capital punishment in the United States (see Radelet and Borg 2000). In fact, following the U.S. Supreme Court's 1972 *Furman* ruling, death penalty scholars have devoted substantial time and resources to collecting social science evidence on key death penalty issues (see Chapter Two). As a result, considerable social research has been conducted on the six major issues that now dominate the death penalty debate in America: (1) deterrence, (2) incapacitation, (3) cost, (4) wrongful convictions/error rates, (5) retribution, and (6) caprice and bias (Baldus 1995; Dieter 1996; Lanier and Acker 2004; Radelet and Borg 2000; Zimring 2005). A discussion of these issues, as well as a review of the relevant empirical research, is presented below.

# 3.1 MAJOR ISSUES SURROUNDING THE DEATH PENALTY

# 3.1.1 Deterrence

Perhaps the major justification for the death penalty is its perceived ability to deter "would-be" murderers from killing. According to deterrence theory, criminal sanctions are most effective at preventing crime when the sanctions outweigh the benefits/gains from engaging in particular criminal activity (*severity*), there is a high probability that the offender will be caught if she or he engages in criminal activity

(*certainty*), and the sanction is administered promptly so there is a close connection between the criminal activity and the punishment (*celerity*) (Beccaria [1764] 1819; Bentham [1780] 1948; Gibbs 1975).<sup>87</sup> While death penalty proponents argue that capital punishment is a more effective deterrent than alternative sanctions (see, e.g., Carrington 1978; van den Haag 1975), death penalty abolitionists maintain that executions are no more effective than long-term imprisonment (Dieter 1996). As a result of this debate, no other issue related to the death penalty has received more systematic attention in legal and academic communities (Bailey and Peterson 1997, p. 135).

Early research was largely unsupportive of the deterrent effect of capital punishment (see, e.g., Sellin 1959; Sutherland 1925). Researchers comparing homicide rates of similar states with and without the death penalty and the homicide rates in jurisdictions before and after the abolition/reinstatement of the death penalty discovered that either the death penalty had no effect on murder rates or that homicide rates were often higher in death penalty states than non-death penalty states (Bailey and Peterson 1997). The debate on the deterrent effect of capital punishment appeared to be resolved

<sup>&</sup>lt;sup>87</sup> Robinson and Darley (2003, pp. 954–55) are highly skeptical about the ability to deter crime through the manipulation of criminal law rules and penalties. While they concede that having a punishment system does have a general effect on the conduct of potential offenders, the conditions under which criminal law manipulation can influence behavior (severity, certainty, and celerity) are unusual, rather than typical, in criminal justice systems in modern society: "We suggest that the infrequency of being able to achieve a meaningful deterrent effect through doctrinal manipulation reveals the deterrent-analysis tradition of modern criminal law scholars, judges, and law-makers to be seriously out of touch with the reality of its limitations [...] the tendency of system participants to undercut the deviation rules—be it through the exercise of prosecutorial or enforcement discretion, sentencing discretion, jury nullification, or other means—means, obviously, that the planned deterrence program will be frustrated" (pp. 1001–1002). There is also disagreement among criminologists as to whether criminal propensity is irrelevant for deterrence (Wright *et al.* 2004). Classical deterrence theorists posit that the motivation to commit crime is to be taken as a constant across persons, and individual differences in offending can be attributed differences between the perceived costs and benefits of crime rather than differences in personality, peer group association, and so forth (e.g., Becker 1968; Cornish and Clarke 1986). Other scholars, however, have suggested that the deterrent effect of criminal sanctions will vary depending on individuals' level of motivation or propensity to commit crime, although there is considerable disagreement as to whether criminally-prone persons are more or less likely to be deterred (Akers 1990; Baier and Wright 2001; Gottfredson and Hirschi 1990; Piquero and Tibbetts 1996; Sherman 1993; Stafford and Warr 1993; Wilson and Herrnstein 1985).

in the *academic* community until the mid-1970s when economist Isaac Ehrlich challenged earlier comparative studies that found no deterrent effect for capital punishment. He criticized these studies on numerous grounds, most notably their failure to take into account the certainty of execution and to control for numerous criminal justice and sociodemographic variables that were correlated with homicide rates. Ehrlich (1975) conducted his own study which corrected for the shortcomings of previous deterrence research and concluded that each execution prevented, on average, seven to eight homicides. His work received national attention when, as noted in Chapter Two, Solicitor General Bork cited his research in his amicus brief in *Fowler*, but was harshly criticized on a host of substantive and methodological grounds (see Baldus and Cole 1975). Critics faulted Ehrlich's work because he: (1) aggregated his data to the national level, ignoring the significant variations between states; (2) failed to differentiate between death penalty and non-death penalty states; (3) did not control for many relevant variables in his models; (4) chose to emphasize models finding a deterrent effect over models that did not report a deterrent effect without providing proper justification;<sup>88</sup> and (5) only found a deterrent effect from 1962 to 1969, but not from 1933 to 1961 (Bowers and Pierce 1975; Fox and Radelet 1989, pp. 39, 42). Although Ehrlich (1977) vigorously defended his work and attempted to address the aggregation bias problem by presenting additional analyses showing a deterrent effect at the state-level (see Ehrlich 1977), replications of his study failed to find a deterrent effect (see, e.g., Albert 1999; Bowers

<sup>&</sup>lt;sup>88</sup> According to McManus (1985, pp. 417–18), there is considerable uncertainty as to the "correct" empirical model that should be used to draw inferences in deterrence studies, and as a result, researchers typically try numerous alternative specifications before selecting and reporting a small subset that usually makes the strongest case for the analyst's prior hypothesis. Building on the work of Leamer (1978), McManus advocates using a Bayesian approach that incorporates the researcher's prior beliefs in order to examine the robustness of *ad hoc* model specifications (for an application of this approach in the death penalty charging context, see Weiss *et al.* 1999).

and Pierce 1975; Forst 1983; Leamer 1983; Passell and Taylor 1977).<sup>89</sup> Moreover, similar to the pre-Ehrlich studies, numerous studies discovered that executions often had a brutalization effect (Bailey 1998; Cochran and Chamblin 2000; Cochran, Chamblin, and Seth 1994; Stack 1994) on certain types of murders—that is, the number of homicides increased following an execution. Subsequent research discovered that neither the celerity of executions (Bailey 1980) nor execution publicity (Bailey 1990; Bailey and Peterson 1989; Stack 1994) had an impact on homicide rates. In fact, a survey conducted in 1996 revealed that over 80 percent of current or former presidents of the three major professional crime and law organizations—the Academy of Criminal Justice Sciences, the American Society of Criminology, and the Law and Society Association—believed that existing research failed to support a deterrence justification for capital punishment (Radelet and Akers 1996). Empirical research on the deterrent effect of the death penalty also appears to have had a strong impact on public opinion: 62 percent of Americans do not believe that the death penalty is a deterrent, up 21 percent from 1991 (Moore 2004).

It is important to note that, over the past several years, a handful of econometric studies have emerged, again challenging conventional wisdom and reporting a deterrent effect for executions (e.g., Cloninger and Marchesini 2001; Dezhbakhsh, Rubin, and Shepherd 2003; Ehrlich and Liu 1999; Mocan and Gittings 2003; Shepherd 2004; Yunker 2001; Zimmerman 2004). These studies suggest that each execution deters anywhere between five (Mocan and Gittings 2003; Shepherd 2004) and eighteen (Dezhbakhsh *et al.* 2003) homicides. Numerous analysts have noted, however, that these recent studies suffer from *at least* three serious problems that undermine the validity of their results

<sup>&</sup>lt;sup>89</sup> Economists John Donohue and Justin Wolfers (2006, p. 2) note that "a National Academy panel savaged Ehrlich's analysis" and "[i]t's modern-day impact beyond the University of Chicago campus is extremely limited" (see Klein, Forst, and Filatov 1978).

(Berk 2005b; Fagan 2006).<sup>90</sup> First, the deterrent effect identified in these studies appears to be the artifact of a few extreme cases. Data solely from Texas accounts for the deterrent effect of capital punishment, although Texas constitutes only two percent of the data points on executions from 1977 through 1997.<sup>91</sup> When Texas is removed from the analysis, there is no deterrent effect for 98 percent of the data (Berk 2005b, pp. 320–21). Not only are the deterrent effects based on the inclusion of Texas data not generalizable to the other states (or the nation as whole), but evidence for a deterrent effect in the state of Texas is also questionable because of the limited amount of data available to make reliable statistical inferences (see also Katz, Levitt, and Shustorovich 2003; but cf. Cloninger and Marchesini 2001). Second, misspecification of the functional form of the relationship between executions and subsequent homicides erroneously suggests a deterrent effect. An exploratory analysis of the bivariate relationship between executions and homicides reveals a nonlinear effect: executions have both slightly positive (suggesting brutalization) and slightly negative (suggesting deterrence) effects on homicides (Berk 2005b).<sup>92</sup> Specifically, the relationship is slightly positive for five executions or less and negative for more than five executions.<sup>93</sup> By incorrectly imposing

<sup>&</sup>lt;sup>90</sup> Fagan (2006, p. 316) notes that the new deterrence studies by these economists "us[e] core elements of identical data sets on executions, death sentences, and murders, and submit[] their papers to peer reviewed journals in economics and non-peer reviewed law reviews," but typically dismiss the contributions of sociologists to the empirical literature on the death penalty as ideological driven (see, e.g., Shepherd 2005; but cf. McManus 1985).

<sup>&</sup>lt;sup>91</sup> There were a total of 1,000 data points analyzed in most of these studies: 20 years multiplied by 50 states  $(20 \times 50 = 1000)$ . The 20 years of Texas data constitute two percent of the total data points  $(20 \div 1000 = 0.02)$ . (Technically, data for 21 years of data for each of the 50 states were available, resulting in 1050 observations, but the one-year lag for execution required all observations for 1977 to be discarded.

<sup>&</sup>lt;sup>92</sup> In particular, Berk (2005b) employs a generalized additive modeling (GAM) approach that allows each predictor variable to have its own functional relationship with the endogenous variable (see Hastie and Tibshirani 1990). GAMs allow the analyst to also "control" for the effect of potential confounding variables. Other research suggests that the deterrent effect of punishment on non-capital crimes may be *both* nonlinear and race-specific (see Yu and Liska 1993).

<sup>&</sup>lt;sup>93</sup> Berk (2005b, p. 313) notes that the relationship between executions and homicides becomes extremely unstable beyond seven executions because of the tiny sub-sample and could essentially be flat beyond

a linear relationship between executions and homicides, these studies allow a small number of extreme outliers to generate the appearance of a deterrent effect when, if fact, no such relationship exists (Berk 2005b; Donohue and Wolfers 2005). In particular, five or more executions occurred in only 11 of the 21 years of Texas data (ranging from 7 to 18 in any given year); therefore slightly over one percent of the data  $(11 \div 1050 =$ 0.0104) account for the deterrent effect discovered by these economists (Berk 2005b, p. 305). Third, these analysts incorrectly assume that using county-level and monthly execution and homicide data sufficiently correct for the distributional problems associated with analyzing the effect of executions (pp. 324–27). While these modifications do permit the analyst to investigate significantly more data points and potentially capture the ephemeral effects of the executions on subsequent homicide activity (see Shepherd 2004), they still reflect the same reality: executions are extremely infrequent and geographically concentrated events. The problems mentioned above seriously undermine the ability of these recent econometric studies to challenge the opinions of most criminal justice experts and the increasing number of Americans who doubt the deterrent value of the death penalty (Weisberg 2005).94

seven executions. Analyzing state-level data from the twenty-seven states that carried out at least one execution between 1977 and 1996, economist Joanna Shepherd (2005, p. 205) posits that executions deter future homicides when states carried out at least nine executions, but have an opposite or neutral effect when states execute less than nine individuals. According to Shepherd, capital punishment decreased murders in six states (Delaware, Florida, Georgia, Nevada, South Carolina, and Texas), increased murders in thirteen states (Arizona, Arkansas, California, Idaho, Illinois, Indiana, Louisiana, Maryland, Oklahoma, Oregon, Utah, Virginia, and Washington), and had no effect in eight states (Alabama, Mississippi, Missouri, Montana, Nebraska, North Carolina, Pennsylvania, and Wyoming). It should be noted, however, that economists John Donohue and Justin Wolfers (2005, p. 829) challenge the appropriateness of Shepherd's models, suggesting that her instrumental-variable estimates are invalid. <sup>94</sup> Like Ehrlich's earlier studies, Dezhbakhsh et al.'s analysis also misuses a sophisticated econometric technique—instrumental variables estimation—and the resulting misspecifications yield extremely misleading results (Donohue and Wolfers 2005, p. 827). With minor alternative model specifications, the impact of each execution ranges from 429 lives saved to 86 lives lost. Even when employing the same model specifications as Dezhbakhsh et al., but properly adjusting for the correlations across counties within a state or the correlation of relevant variables through time (i.e., clustering), their models suggests that the

### 3.1.2 Incapacitation

Even if some supporters of capital punishment are willing to concede that research on the deterrent effect of executions is largely unsupportive, they maintain that the specific deterrent effect of capital punishment is undeniable: executed murderers will not be able to kill again (van den Haag 1975). In fact, most public opinion polls indicate that incapacitation is the one of the most popular reasons for favoring the death penalty (Ellsworth and Gross 1994; Ellsworth and Ross 1983; Warr and Stafford 1984). As noted earlier, there was widespread concern that the Furman commuted inmates would kill again once released from prison. Subsequent research indicated, however, that *Furman*-commuted inmates were no more likely to re-offend for murder than the general inmate population: only 1.3 percent of Furman-commuted inmates committed a subsequent murder (Marquart and Sorensen 1989). Research also suggests that individuals convicted of homicide make significantly better adjustments in prison and, if released, exhibit lower rates of recidivism compared to other convicted felons (Radelet and Borg 2000). For example, in 1993, slightly less than 10 percent of death row inmates had a prior murder conviction, while over one-third of death row inmates had no prior felony conviction (Bedau 1997d). Death penalty proponents have also argued that prison inmates and prison staff are safer in death penalty states because inmates who murder while incarcerated risk execution for their crime, whereas inmates in abolitionist states face no such penalty; however, studies comparing the likelihood of prison homicides in death penalty versus abolitionist states suggest otherwise: prison murders are more common in jurisdictions with capital punishment than those without capital punishment

effect of each execution ranges from 119 lives saved to 82 lives lost (p. 835). Such questionable evidence has lead two well-known economists to recently remark, "The view that the death penalty deters is still the product of belief, not evidence" (Donohue and Wolfers 2006, p. 5).

(Bedau 1997d). Finally, supporters of capital punishment have commented that individuals sentenced to life imprisonment for murder only spend a few years behind bars. Interviews with jurors serving on capital trials in fourteen death penalty states across the nation also suggests many jurors grossly underestimate the amount of time that murder defendants will spend behind bars, leading them to vote for a sentence of death when they would have voted for a life sentence if properly informed (see, generally, Blume, Garvey, and Johnson 2001; Bowers and Steiner 1999; Eisenberg, Garvey, and Wells 2001a; Eisenberg and Wells 1993; Luginbuhl and Howe 1995; Steiner, Bowers, and Sarat 1999). This belief is echoed in the general population: public opinion polls indicate that most Americans believe that convicted murderers spend, on average, seven years in prison unless they are sentenced to death (Bedau 1997d, p. 181). This figure, however, is actually *half* of the actual amount of years, on average, that individuals sentenced to life imprisonment with the possibility of parole spend behind bars. Most retentionist and abolitionist jurisdictions also allow juries (and judges) to sentence inmates to life without the possibility of parole (LWOP). Specifically, all jurisdictions with the death penalty and 11 of the 12 states without the death penalty offer life without parole (Weisberg 2005).<sup>95</sup> Research suggests that 46 percent of Americans favor LWOP over the death penalty, up 14 percent from a decade ago (Dieter 1993; Moore 2004).

Proponents of capital punishment have also argued that society is much safer because of the "selective incapacitating" effect of executions (i.e., executions permanently remove individuals who have the highest risk of killing again from the

<sup>&</sup>lt;sup>95</sup> Texas became the most recent state to enact LWOP legislation when Governor Rick Perry signed the bill into law in June 2005. Prior to repealing its death penalty in March 2009, New Mexico was the only death penalty state that did not offer LWOP as a sentencing option. The District of Columbia, a non-death penalty jurisdiction, also offers LWOP as a sentencing option.

community) (van den Haag 1975, 1986). Abolitionists have criticized the selective incapacitation argument on the grounds that only a small number of the most highly aggravated homicides result in a death sentence. These death penalty opponents highlight a large body of evidence that suggests—although the level of aggravation in a homicide case strongly impacts the probability that the case is noticed for death and receives the death sentence—there is considerable variation in the outcomes of the most highly aggravated homicide cases. Furthermore, they note that other illegitimate case characteristics (e.g., race, gender, region) exert a strong influence on capital chargingand-sentencing decisions even after aggravation levels are held constant (Baldus and Woodworth 2003). Because prosecutors often fail to seek the death penalty and juries fail to impose the death sentence on a non-trivial number of these highly aggravated cases, and extra-legal factors improperly influence who is charged and sentenced to death among similarly culpable murder defendants, many convicted murderers who are at high risk of killing again are released back into the community. According to abolitionists, any benefit society may receive from executing an individual at high risk of killing again is offset by the considerable number of similarly culpable of convicted murderers who are released back into the community (see Radelet and Borg 2000).

# 3.1.3 Cost/Expense

The number of inmates in state and federal prisons and local jails has increased by 800 percent over the past 30 years. In 1971, there were fewer than 250,000 individuals behind bars on any given day (Currie 1998). On June 30, 2003, there were 2,078,570 prisoners incarcerated in prisons and local jails (U.S. Department of Justice 2004a). The skyrocketing criminal justice costs associated with the enormous expansion in the prison

population have placed a major strain of budgets of local, state, and federal governments (Currie 1998). For many years, death penalty proponents argued that executing convicted killers was much more cost effective than keeping them behind bars for life. In contrast, abolitionists claimed that life imprisonment was much more cost effective than the death penalty. Unfortunately this debate was rarely supported by solid research until detailed studies concerning the actual costs associated with the death penalty started being conducted in the late 1980s (Dieter 1994). Examinations of the costs associated with the death penalty in California, Colorado, Florida, Indiana, Kansas, New York, North Carolina, Texas, and the federal government all revealed that sentencing individuals to death was much more expensive than sentencing them to life imprisonment: California spends \$90 million on the death penalty each year; Colorado spends an average of \$2.5 million on each capital case; Florida spends an estimated \$51 million a year above and beyond what it would cost to sentence all first-degree murders to LWOP, Indiana spends 38 percent more on capital cases than on LWOP sentences; Kansas spends 70 percent more on capital cases than non-capital murder cases (\$1.26 million per capital case versus \$740,00 for non-capital cases through the end of incarceration); Texas spends an average of \$2.3 million per capital case, New York spends approximately \$1.5 million on death penalty cases; and the federal capital cases cost nearly four times more than comparable non-capital cases (Dieter 1994, 1996). In one of the most thorough examinations to date, Cook and Slawson (1993, pp. 77–78) discovered that North Carolina spends an additional \$216,000 per death sentence and an additional \$2.16 million per *execution* compared to non-capital cases resulting in life imprisonment. They estimate that, nationally, over \$1 billion more has been spent on the death penalty since 1976.<sup>96</sup> A more recent study

<sup>&</sup>lt;sup>96</sup> According to Cook and Slawson, this is a conservative estimate because it only considers costs to the

suggests that each additional capital trial causes an increase in county spending of 1.8 percent and an increase in county taxes of 1.6 percent, resulting in an increase of more than \$1.6 billion in between 1982 and 1997 (Baicker 2004). While the perceived high cost of life imprisonment, in of itself, is rarely cited by death penalty supporters as a reason for their position, these financial considerations, coupled with capital punishment's dubious deterrent effect and the fiscal crises that many states are facing, have many state and local officials rethinking their death penalty statutes (Dieter 1994; Rupp 2003).

#### 3.1.4 Wrongful Convictions/Error Rates

As noted earlier, one of the strongest criticisms of capital punishment made by death penalty opponents is that innocent people may be wrongfully executed. Although no legal or public official has admitted to a wrongful execution since 1887 (Bedau and Radelet 1987; Radelet, Bedau, and Putnam 1992), abolitionists point to the 118 individuals released from death row between 1973 and 2004 as strong evidence of the danger of mistaken executions (Dieter 2004). Abolitionists have also noted that the danger of mistaken executions has increased dramatically over the past ten years (Dieter 2004). Until a decade ago, death penalty retentionists strongly argued that mistaken executions were historical anomalies—resulting from adequate safeguards in the capital punishment process—and incapable of occurring in modern times (Cassell 1999; Radelet and Bedau 1998). More recently, however, retentionists have acknowledged the existence of such miscarriages of justice, but justify their continued support of the death penalty by contending that these miscarriages are extremely infrequent and that

state and local government, and does not include federal and private costs (p. 79).

abolitionists have never proven that an innocent person has been executed in the modern era of the death penalty.<sup>97</sup> Furthermore, retentionists argue that the procedural safeguards are functioning properly by identifying mistakes and preventing the innocent from being put to death (but see Liebman 2000). Abolitionists, however, have strongly criticized all of these assertions.

In contrast to retentionists' contentions that erroneous convictions in capital cases are extremely infrequent, death penalty opponents argue the number of exonerations to date grossly underestimates the actual number of individuals wrongfully sentenced to death. Highlighting evidence of wrongful convictions in non-capital murder cases, they suggest that many more people on death row are likely to be innocent because the factors influencing wrongful convictions (e.g., eyewitness error, prosecutorial misconduct, false or coerced confessions, ineffective assistance of defense counsel, and perjury) are very similar in both capital and non-capital cases (see Garrett 2008; Gross 1998b; Harmon 2001b; Harmon and Lofquist 2005; Huff 2002; Natapoff 2006; Scheck, Neufeld, and Dwyer 2000).<sup>98</sup> Between 1972 and 1996, over 400 people convicted of murder in both capital and non-capital cases in the United States were later found to be innocent of the crime (Bedau and Radelet 1987; Radelet *et al.* 1992). A recent study conducted by Gross

<sup>&</sup>lt;sup>97</sup> Renowned retentionist Ernest van den Hagg (1986, p. 1665) remarked, "Despite precautions, nearly all human activities, such as trucking, lighting, or construction, cost the lives of some innocent bystanders. We do not give up these activities, because the advantages, moral or material, outweigh the unintended losses. Analogously, for those who think the death penalty just, miscarriages of justice are offset by the moral benefits and the usefulness of doing justice."

<sup>&</sup>lt;sup>98</sup> In 2006, the Criminal Justice Section of the American Bar Association released a report on preventing wrongful convictions. The report was the result of a three-year research effort lead by an *ad hoc* group of defense attorneys, prosecutors, judges, academics, and representatives from the forensic and law enforcement communities, and its recommendations were adopted by the organization's policy-making body, the House of Delegates. The committee identified nine problem areas leading to wrongful convictions and made recommendations in each of these areas: (1) false (and coerced) confessions; (2) eyewitness identification procedures; (3) forensic evidence; (4) jailhouse informants (e.g., perjury); (5) defense counsel practices (i.e., ineffective assistance of counsel); (6) investigative policies and personnel; (7) prosecution practices (i.e., prosecutorial misconduct); (8) systemic remedies; and (9) compensation for the wrongfully convicted (American Bar Association 2006a).

and colleagues (2005) revealed that 61 percent of all exonerations for serious violent and non-violent crimes between 1989 and 2003 were for the crime of murder (199 exonerations). The proportion of exonerations for murder is particularly noteworthy considering that convicted murderers constitute only 13 percent of the total prison population in the United States. Moreover, death row inmates only comprise approximately 0.25 percent of the total prison population, but 22 percent of inmates exonerated.

Poveda's (2001) detailed analysis of murder convictions in New York revealed that 1.4 percent individuals convicted of murder between 1980 and 1987 were later found be innocent. If this percentage were applied to the 3,487 individuals currently on death row in the United States, 49 individuals ( $3,487 \times .014 = 48.82$ ) would have been erroneously sentenced to death. Admittedly, the wrongful conviction rate for capital cases is likely to be significantly lower than for non-capital cases because of the procedural safeguards (e.g., mandatory appeals) present in death penalty cases (but see Gross 1996).<sup>99</sup> But even if the erroneous murder conviction rate is halved (0.7 percent), nearly 25 individuals on death row still have been erroneously convicted.<sup>100</sup>

Contrary to retentionists, abolitionists have also argued that innocent people have, in fact, been executed. According to abolitionists, the refusal of public and legal officials

<sup>&</sup>lt;sup>99</sup> According to Gross (1996, pp. 494–96), capital cases may lead to *more* wrongful convictions than other non-capital murder cases because most capital defendants are *not* sentenced to death. He argues that capital cases are more likely to result in a conviction because of trial publicity, death qualified jurors who are more conviction prone than non-death qualified jurors, the increased likelihood of false confessions by innocent defendants to avoid the death sentence, the decreased likelihood that the actual murderer will confess to the crime, and the heinousness of capital murders. Many capitally charged defendants who are convicted but do not receive a death sentence are believed to have "already received the benefit of whatever doubts their cases may raise" (p. 498), and as a result, their cases fail to receive the "post-conviction" attention that capital cases resulting in a death sentence receive.

<sup>&</sup>lt;sup>100</sup> New York is often regarded as having one of the finest public defender systems in the nation, so the national erroneous conviction rate may be higher.

to admit executing an innocent person is not surprising, considering that such admissions would not only be professionally damaging, but it would also have serious legal and financial repercussions for the states and localities in which the wrongfully convicted are executed (Kirchmeier 2006; Radelet and Bedau 1998). Moreover, there is little incentive for prosecutors to admit wrongdoing in murder cases because misconduct is treated with great leniency—not one prosecutor has been disbarred in the 381 murder convictions that have been reversed because of misconduct (Scheck, Neufeld, and Dwyer 2000; Weinburg 2003). Nonetheless, examining 350 capital cases resulting in wrongful convictions in the United States between 1900 and 1987, Bedau and colleagues (Bedau and Radelet 1987; Radelet et al. 1992) discovered that 139 people were sentenced to death, 22 individuals came within 72 hours of being executed, and 23 believed to be innocent were executed (see also Radelet and Bedau 1998).<sup>101</sup> Since 1992, at least five individuals have been executed despite strong doubts about their guilt (Death Penalty Information Center 2008).<sup>102</sup> Some abolitionists contend that perhaps the best measure of the risk associated with executing the innocent is the ratio of exonerations to executions. From 1977 to 2008, 1,136 individuals were executed and 130 exonerated (Death Penalty Information Center 2008), or stated differently, one exoneration for every 8.7 executions. As noted earlier, Illinois offers the most extreme example of the risk of executing the innocent.

<sup>&</sup>lt;sup>101</sup> After a draft of the study was originally released in February 1987, then-Attorney General Edwin Meese—a strong death penalty advocate—ordered the Department of Justice to draft a response. The DOJ's response failed to offer any additional evidence to refute the initial study, instead focusing its sole attention on 10 of the 23 executions, arguing that the study was flawed because it was ideological driven, that the executed were convicted "beyond a reasonable doubt," and that no legal officials presiding over the cases admitted that the executed was innocent (Radelet and Bedau 1998). In contrast to Attorney General Meese's response, Bedau and colleagues' report was received favorably by U.S. Supreme Court Justice Harry Blackmun who cited the study as one of the reasons that he no longer supported the death penalty (see *Callins v. Collins*, 510 U.S. 1141 [1994]).

<sup>&</sup>lt;sup>102</sup> The executed individuals believed to be innocent were: Roger Keith Coleman in 1992 (Virginia), Joseph O'Dell (Virginia) in 1997, David Spence in 1997 (Texas), Leo Jones 1998 (Florida), and Gary Graham in 2000 (Texas).

Since Illinois reinstated the death penalty in 1977, it has exonerated more individuals than it has executed (13 exonerations and 12 executions) (see Johnson 2000a,b; Ryan 2003).

In response to the growing number of death row exonerations, retentionists have argued that these exonerations actually prove (a) that the procedural safeguards in capital trials and appeals are functioning correctly by preventing those wrongfully convicted from being wrongly executed, (b) that such exonerations are too infrequent to justify abolition of the death penalty, or (c) that abolitionists have failed to provide evidence that an innocent person has been executed (see, e.g., Cassell 2003, p. 208).<sup>103</sup> In particular, they contend that capital cases are held to the highest legal standards and undergo serious judicial scrutiny. By contrast, abolitionists note that the incredibly high rate of mistakes in the capital punishment system underscores the fact that the system is not operating properly to begin with and creates serious concern as to whether the courts will catch them all (Gross 1996; Liebman et al. 2000c). Abolitionists also posit that the wrongful conviction of the innocent constitutes only one type of miscarriage of justice. The second (and much more frequent) type of mistake arises when individuals are sentenced to death for capital murder when, in fact, they should have been convicted of second-degree murder or manslaughter, and as result, been ineligible for the death penalty in the first place (Liebman, Fagan, and West 2000b; Radelet and Borg 2000).<sup>104</sup> In 1991, the U.S.

<sup>&</sup>lt;sup>103</sup> Some abolitionists have found it ironic that many of the retentionists who currently applaud the procedural safeguards in the capital charging-and-sentencing process challenged the implementation of these very safeguards decades earlier on the grounds that they believed that the safeguards were unnecessary and would undermine the capital punishment process (for a discussion, see Liebman *et al.* 2000c).

<sup>&</sup>lt;sup>104</sup> This issue of "actual innocence" versus "innocent from the death penalty" was brought before the court in *Sawyer v. Whitley* (505 U.S. 333 [1992]). The petitioner in the case, Robert Wayne Sawyer, was sentenced to death in Louisiana for his participation in a murder in 1979. Sawyer was convicted of firstdegree murder (i.e., "intentional murder") and the jury found two aggravating circumstances: (1) murder

Senate Committee on the Judiciary authorized Columbia University Law Professor James Liebman to conduct a study on habeas corpus relief in capital cases. Liebman and colleagues (2000c) discovered that 68 percent of the 4,578 death sentences imposed in the United States between 1973 and 1995 receiving appellate review were overturned because of serious error.<sup>105</sup> They also discovered that three rounds of judicial inspection (state direct appeal, state post-conviction, and federal habeas corpus) were necessary to catch a significant portion of these serious mistakes. For example, while state courts overturned 47 percent of death sentences,<sup>106</sup> an additional 40 percent of the death sentences upheld by state courts were later overturned by federal courts (Liebman et al. 2000c). The high error rates placed many individuals at risk for wrongful executions: 82 percent of the death sentences overturned resulted in a sentence less than death upon retrial and seven percent were found to be innocent of the capital crime. Similar to studies of wrongful murder conviction, the Liebman study revealed that the most common reasons for reversal were egregiously incompetent defense lawyers and serious police and prosecutorial misconduct (Liebman et al. 2000c). The Liebman study also

occurred while the defendant was engaged in the commission of another felony; and (2) the murder was especially cruel, atrocious, and heinous. Sawyer claimed that although he was guilty of murder and the facts of the case made him "technically" eligible for the death penalty, he should have not been eligible for the death penalty because he should have not been convicted of first-degree murder, and he was not guilty of committing the contemporaneous felony of aggravated arson (the offense on which the statutory aggravating circumstances were based). Among other things, Sawyer pointed to evidence of his minimal involvement in the crime and that a witness claimed to have seen Sawyer trying to prevent his co-offender from setting the victim on fire. The Court affirmed Sawyer's death sentence, ruling unanimously ruled that Sawyer failed to satisfy the "actual innocence" exception.

<sup>&</sup>lt;sup>105</sup> Liebman *et al.* (2000c) limited their analyses to "serious errors" by exclusively focusing on errors that were (1) *prejudicial* because either the defendant has shown it probably affected the outcome of her or his case *or* because it is the type of error that almost always has that effect and (2) *properly preserved* by way of a timely objection at trial, reiterate in a timely new trial motion at the end of the trial, and timely and properly asserted on appeal.

<sup>&</sup>lt;sup>106</sup> Ironically, as Liebman and colleagues note, it is often the very same judge who presided over the death penalty trial in question who overturns the death conviction or sentence on appeal.

discovered that the high error rates were persistent over the entire time of the study<sup>107</sup> and were present across the entire nation.<sup>108</sup> In a follow-up study to determine why some states had higher error rates than others. Liebman and colleagues (2002) discovered that error rates were highest in states that: (1) impose the death penalty at higher rates; (2) have a homicide risk rate for whites that is near (or greater than) that of blacks; (3) have a higher proportion of blacks and welfare recipients (see also Jacobs and Carmichael 2002); (4) have a lower arrest rates for serious felony offenses (both violent and nonviolent); (5) have trial judges that are subject to highly partisan popular elections (but cf. Blume 1999); and (6) impose death sentences in cases that are *not* highly aggravated. Abolitionists argue that the prevalence and frequency of serious error in capital cases along with the incredible amount of time necessary to uncover such error (nine year national average)—clearly reveal a system in disrepair and runs the serious risk of executing the innocent (see, e.g., Liebman 2000). Death penalty opponents also note that a number of the death row exonerations in recent years have resulted from the emergence of new scientific technology (e.g., DNA testing) and the efforts of individuals and organizations working outside of the criminal justice system (e.g., journalists), rather than properly functioning legal mechanisms (Bedau 1997c; Smith 2005). The fact that several of these death sentences were affirmed during appellate review brings into question the adequacy of the current procedural safeguards in place in the capital punishment process and their ability to prevent the execution of the innocent (Dieter 2004).

<sup>&</sup>lt;sup>107</sup> Slightly over 50 percent of all death sentences reviewed in a single year were overturned in 20 of the 23 years studies and the error rate was over 60 percent in over half of the years studied.

<sup>&</sup>lt;sup>108</sup> Ninety-two percent of the death-sentencing states (24 states) had error rates over 52 percent, 85 percent of the states (22 states) had error rates of 60 percent or higher, and 61 percent of the states (15 states) had error rates of 70 percent or higher.

Research suggests that the concern over capital defendant's guilt has a strong impact on jurors' decision to impose a life or death sentence (Garvey 1998). An examination of actual jurors who sat on capital cases in 15 states across the United States revealed that lingering doubt about a defendant's guilt in the first phase of a capital trial (i.e., guilt/innocence phase) was the most important factor in deciding to vote for a life sentence in the penalty phase (Bowers, Sandys, and Steiner 1998; Garvey 1998; Sandys 1995).<sup>109</sup>

### 3.1.5 Retribution

In light of the increasing evidence demonstrating the growing risk of convicting and executing innocent individuals (as well as the other problems mentioned above), many death penalty advocates argue the societal benefit of the death penalty outweighs the societal harm on retributive grounds (for discussions, see Ellsworth and Gross 1994; Gross 1993; Warr and Stafford 1984).<sup>110</sup> Retentionists posit that the death penalty is necessary for justice and only through executing individuals who commit murder can justice be served (see, e.g., Carrington 1978; Cassell 2003; Pojman 2003; van den Haag 1975, 1986; cf. Sunstein and Vermeule 2005).<sup>111</sup> Unlike other debates surrounding the death penalty, retributive arguments primarily rely on a non-empirical justification (i.e., morality) and, therefore, social science research has not had much of an impact on this

<sup>&</sup>lt;sup>109</sup> A recent study by Unever and Cullen's (2005) revealed that over three-quarters of Americans believed that an innocent person had been executed in the past five years. They also discovered that this belief was associated with lower levels of support for capital punishment.

<sup>&</sup>lt;sup>110</sup> Citing recent Gallup Poll evidence, Radelet (2000) notes that 46 percent of respondents focused on the "eye for an eye" rationale to justify their support for capital punishment.

<sup>&</sup>lt;sup>111</sup> According to van den Haag (1986, p. 1663), the "[m]aldistribution between the guilty and innocent is, by definition, unjust...[but the] maldistribution of any punishment among those who deserve it is irrelevant to is justice or morality" (but see Laufer and Hsieh 2003).

aspect of the debate (Radelet and Borg 2000; but cf. Sunstein and Vermeule 2005).<sup>112</sup> One notable exception is Borg's (1998, p. 537) work on vicarious homicide victimization and support for capital punishment. She notes that individuals who support capital punishment on retributive grounds argue that the death penalty is necessary because it offers the family members and close friends of homicide victims a legitimized means for exacting revenge. Borg questions this assertion on four grounds. First, she notes that less than three percent of individuals convicted of homicide in the United States are sentenced to death, so the death penalty is either an unavailable or very remote punishment strategy for the family and close friends of homicide victims. Second, the vast majority of homicide victims are either related to or acquainted with their attackers (over 75 percent of homicide victims know their attackers, see Fox 2005) and family members and close friends of homicide victims who know the offender—which is rather common if the victim and the offender know one another-may be less likely to support capital punishment. Third, the vast majority of homicides are intra-racial (approximately 95 percent black-victim and 84 percent of white-victim homicides are intra-racial, see Fox 2005) and family and close friends of victims may be less like to support capital punishment when they share a strong cultural similarity to the offender. Finally, although blacks constitute nearly half of all homicide victims (Fox 2005), they are more likely to be distrustful of the criminal justice system (Anderson 1990; Cooney 1997a; Miller 1996) and, as a result, less likely to advocate for capital punishment (Young 1991, 2004).

<sup>&</sup>lt;sup>112</sup> Sunstein and Vermeule (2005, p. 706) suggest that if, indeed, capital punishment has a deterrent effect, then the refusal to impose the death penalty "ensur[es] the deaths of a large number of innocence people" and this kind of government inaction would constitute "a serious moral wrong." (but cf. Kaufman-Osborn 2006b). The authors note that they do not take a stand on the validity of the econometric studies supporting the deterrence hypothesis, rather they simply set out to explore the moral implications of government inaction if capital punishment is an effective deterrent. For a critique of Sunstein and Vermeule's consequentialist argument, see Steiker (2005).

Analyzing survey data from a nationally representative sample of 1,312 respondents, Borg discovered that blacks who were family and close friends of homicide victims were *less* likely to support capital punishment than blacks who were not family and friends of homicide victims. This result held after controlling for education, age, gender, region of the country (i.e., South/Non-South), and religious orthodoxy. Whites who experienced vicarious victimization, on the other hand, were more likely to support the death penalty, although white males were the most likely to support capital punishment, regardless of vicarious victimization status.

In the 1970s, over 47 Christian denominations, representing more than 10 million conservative Christians, supported the death penalty on biblical grounds, primarily citing the Old Testament (see Bedau 1997b, pp. 415–28). While most Fundamentalist and Pentecostal churches continue to support capital punishment, abolitionists have noted that much of the "moral leadership" in the United States now opposes the death penalty (Radelet 2000, p. 207). In recent years, for example, the Roman Catholic Church and many Protestant denominations (e.g., Episcopalians, Lutherans, Methodists, Presbyterians, and the United Church of Christ) have called for the abolition of the death penalty (Berg 2000; Religious Organizing Against the Death Penalty Project 1998). Similarly, abolitionists have highlighted much of the international community's rejection of the death penalty as evidence that the "morality" of the death penalty is questionable and that justice can be served without capital punishment. The concern over the unnecessary suffering of the condemned and trend towards more "humane" methods of execution dating back to the eighteenth century also appears to be inconsistent with the idea that the death penalty is needed for retributive justice (Radelet and Borg 2000).

Some abolitionists have made the rather unconventional argument that the death penalty is too lenient on convicted murderers and that these offenders should spend the rest of their lives in prisons (Brownlee, McGraw, and Vest 1997; Ibrahim 2004). But perhaps the strongest argument that abolitionists have made against the death penalty on retributive grounds is that it is unjust in that it too frequently punishes those who are not deserving of the death penalty (e.g., those who are innocence of a capital offense), and therefore fails to serve any function.

While miscarriages of justice (i.e., wrongful convictions/executions and error rates) have received the most public and media attention over the past several years (Huff 2002; Unnever and Cullen 2005) and the issue of deterrence has received the most scholarly attention (Bailey and Peterson 1997), studies of potential caprice and bias in the administration of the death penalty were among the earliest conducted by social scientists (see, e.g., Garfinkel 1949; Johnson 1941) and, as noted earlier, evidence from such studies has been directly used to challenged the constitutionality of the death penalty on the grounds that it is administered in a racially discriminatory manner (see Chapter Two). In the following chapter, a detailed review of the scholarly literature concerning the influence of extra-legal factors (particularly race/ethnicity) on the capital chargingsentencing process is presented. A proper examination of the role that race has played in the capital charging-and-sentencing process, however, first requires some historical information on the relationship between race and the law (particularly criminal law) in the United States. Therefore a brief discussion of this history, followed by a close examination of the research literature, appears in the next chapter.

# **Chapter 4: Extra-Legal Factors and the Death Penalty**

### 4.1 BACKGROUND ON RACE AND THE LAW IN THE UNITED STATES

#### 4.1.1 Early Colonial Period through the Civil War

The discriminatory treatment of blacks in the legal system has had a long history in the United States (Foner and Mahoney 1995; Williams 1991).<sup>113</sup> Enslaved Africans brought to the colonies enjoyed very few legal rights and almost no legal protections (David et al. 1976; Spindel 1989). Although white bondage also existed in the early days of the colonies in the form of indentured servitude, the legal status of white indentured servants was very different from blacks (Smith [1947] 1971). For example, unlike white endured servants, blacks were held in servitude for life, black children could be born slaves because they inherited their free/slave status from their mother, and conversion into Christianity did not alter the slave status of blacks (David et al. 1976; Skidmore 1993). Southern states developed statutes known as "slave codes" to legitimize the inferior legal and social status of blacks and allowed slaveholders broad discretion in controlling the slave population. First, by mandating that slaves were the legal private property of the slaveholders, slave owners had virtually unlimited power over their slaves without the threat of outside intervention (Schwartz 1988). Although some limitations were placed on these powers in later years (see Fogel and Engerman 1974; Spindel 1989), they were enforced at the discretion of the slaveholder (David et al. 1976). Second, slave codes prohibited blacks from enjoying nearly any constitutional protection

<sup>&</sup>lt;sup>113</sup> Wacquant (1997, p. 230) suggests that societies draw and enforce racial/ethnic boundaries through five "elementary" forms of domination: (1) categorization (prejudice and stigma), (2) discrimination (differential treatment based on imputed group membership), (3) segregation (group separation in physical and social space), (4) ghettoization (the forced development of parallel social and organizational structures), and (5) exclusionary violence (ranging from interpersonal intimidation and aggression, to lynching, riots and pogroms, and climaxing with racial warfare and extermination).

afforded whites. Slaves were not allowed to vote, own property, marry a partner of their own choosing, travel off of the plantation without approval from the plantation master, or work for themselves (David et al. 1976; Hindus 1980). Third, slave codes established severe punishments for a wide range of offenses committed by blacks that were not considered criminal for whites. For example, slaves were often whipped, beaten, tortured, burned, mutilated, and sometimes killed for such minor infractions as making direct eye-contact with whites, arguing with whites, not moving out of the way when approaching whites, learning to read or write, attempting to vote, and (for males) speaking with a white woman (Mullin 1972). These slave codes also differentially punished the criminal behavior of blacks much more severely than the criminal behavior of whites for the same offenses (Schwartz 1988). Many offenses committed by whites that were punishable by brief imprisonment or a small fine were punishable by whipping, branding, or mutilation if committed by blacks (see e.g., State of Georgia 1848). Slave codes also protected non-slaveholding whites from criminal punishments when victimizing blacks. Many slave codes prohibited blacks from testifying against whites and bringing criminal or civil cases against whites. In North Carolina, the killing of a slave by a white could not be prosecuted as murder until 1774 (Spindel 1989, p. 48). Since slaves were property of their masters, the slave master was the offended party if her or his slave was victimized or murdered by another white person (Fogel and Engerman 1974). In these situations, it was largely a civil matter and the owner of the slave was only entitled to financial restitution for the loss of her or his property (David et al. 1976).

Finally, slave codes enumerated many more capital offenses for blacks than for whites and blacks were much more likely to have their offenses tried as capital cases and be sentenced to death (Higginbotham and Jacobs 1992; Spindel 1989). Tennessee's 1858 Slave Code listed 124 capital crimes for slaves and free blacks, but only two capital offenses for whites (Vandiver, Giacopassi, and Curley 2001). In Virginia, slaves could receive the death penalty for 68 offenses, whereas whites could only be put to death for first-degree murder (Higginbotham and Jacobs 1992, p. 1022). Furthermore, in Virginia, slaves could be executed for any crime that, if committed by a white person, called for a sentenced sentence of not less than three years. From 1641 (the year of the first recorded execution of a black male) until the end of the Civil War, at least 1,890 blacks were legally executed, comprising 47.2 percent of all executions during this period (Espy and Smykla 2004). The proportion of blacks executed was 270 percent greater than their proportion of the population: from 1790 (when the federal government began collecting census data) to 1860, blacks, on average, comprised 17.5 percent of the nation's population (U.S. Bureau of the Census 2002b).

Moreover, the inferior legal status of blacks was neither limited to the institution of slavery nor the southern region of the United States (cf. Agamben [1995] 1998). In its landmark "Dredd Scott" decision (see *Scott v. Sanford*, 60 U.S. 393 [1857]), the U.S. Supreme Court ruled that free and enslaved blacks could never become U.S. citizens; therefore they had no rights that whites were required to respect. The court also held that the portion of the Missouri Comprise of 1820 that prohibited the expansion of slavery in the U.S. territories north and west of the state of Missouri was unconstitutional because enslaved blacks were the property of their owners and Congress could not prohibit slaves owners from taking their property into any territory owned by the United States.

### 4.1.2 Post Civil War Period through the 1960s

Immediately following the Civil War and the emancipation of millions of enslaved blacks, the former Confederate states began replacing slave codes with "Black Codes" (Ayers 1984). These Codes attempted to resurrect the legal and social order established during slavery (Foner and Mahoney 1995). Similar to slave codes, Black Codes defined both the legal and social status of freed blacks. Although somewhat less restrictive than the previous slave codes (e.g., blacks were allowed to own property, enter into marriage contracts, and file legal suits), Black Codes tried to severely limit the freedoms of the newly emancipated blacks and restore the racial hierarchy that existed under the previous slavocracy. Blacks were generally prohibited from voting, holding public office, serving on juries, testifying against whites, marrying whites, and owning weapons (Ayers 1984; Foner and Mahoney 1995). Blacks were also prohibited from entering certain non-agricultural occupations and were forced to sign labor contracts that allowed southern whites to exploit them. Not only were blacks not allowed to terminate these labor contracts before they expired, but these contracts allowed blacks to be beaten for poor performance on the job and for minor infractions such as insubordination and theft. Perhaps the most repressive aspects of the Black Codes were the vagrancy laws that they established. These vagrancy laws allowed both black adults and children who were unemployed or who did not have a labor contract with a white employer to be arrested, imprisoned, and forced into labor (Oshinksky 1996). Black Codes also ensured that blacks would be punished differently from whites (Ayers 1984). Rather than explicitly mentioning race in criminal statutes, southern whites ensured that blacks would receive harsher treatment by giving legal officials and juries greater discretion in

charging and sentencing (Ayers 1984; Williamson 1984). Southern whites also differentially disciplined blacks by prescribing more severe punishments for crimes that were generally perceived to be disproportionately committed by blacks than by whites (e.g., vagrancy, rape, arson, and burglary) and limited corporal punishment to black offenders (Ayers 1984). Similar the aforementioned labor and vagrancy laws, the penal systems and "convict-lease" policies of the former slave states supplied southern planters, businessmen, and financiers with the free (and forced) black labor that existed under slavery (Oshinksky 1996; Wacquant 2000).

As a result of the blatant mistreatment of blacks in the former Confederate states, the U.S. Congress passed the Fourteenth (1868) and Fifteenth Amendments (1870), allowing the federal government to protect the constitutional rights of all citizens and extending the right to vote to black men, respectively. These new constitutional protections, however, were frequently ignored or circumvented by the southern states (Behrens, Uggen, and Manza 2003; Kennedy 1998).<sup>114</sup> The vast majority of former Confederate states (and former slave states that did not secede) began enacting laws that prohibited individuals convicted of crimes from voting in an attempt to deny suffrage to blacks. For example, during the Reconstruction Era, felony disenfranchisement laws

<sup>&</sup>lt;sup>114</sup> Behrens and colleagues (2003) note that many Democrat-controlled Northern and Western states initially refused to ratify the Fourteenth and Fifteenth Amendments (Southern states were required to do so as a condition of readmission into the Union), and many Northern states that did support the two amendments did it primarily to punish the South rather than to achieve racial equality. Indeed, Abraham Lincoln's primary reason for going to war was not the dismantling of slavery, but rather the preservation of the Union. In his famous letter to *New York Tribune* editor, Horace Greeley, at the beginning of the Civil War, Lincoln wrote:

If there be those who would not save the Union, unless they could at the same time save slavery, I do not agree with them. If there be those who would not save the Union unless they could at the same time destroy slavery, I do not agree with them. My paramount object in this struggle is to save the Union, and is not either to save or destroy slavery. If I could save the Union without freeing any slave, I would do it; and if I could save it by freeing all the slaves, I would do it; and if I could save it by freeing all the slaves, I would do it; and if I could save it by freeing all the slaves, I what I do about slavery, and the colored race, I do because I believe it helps to save the Union; and what I forbear, I forbear because I do not believe it would help to save the Union (Lincoln [1862] 1953, p. 388).
were passed in Alabama (1867), Arkansas (1868), Missouri (1875), Florida (1868), Georgia (1868), Mississippi (1868), North Carolina (1876), South Carolina (1868), Tennessee (1871), and Texas (1869) (Behrens *et al.* 2003, pp. 565–66). The remaining five former slave states—Delaware, Kentucky, Louisiana, Maryland, and Virginia enacted felony disenfranchisement laws before the Civil War.<sup>115</sup> Not only were felony disenfranchisement statutes in these states among the most punitive in the country (Behrens *et al.* 2003), politicians in these states frequently made explicit appeals to the public to prevent blacks from voting (see Mendelberg 2001).<sup>116</sup> Southern states also began legalizing the practices of racial segregation in response to the intervention by the federal government to extend the full rights of citizenship to newly freed blacks. In response to these separatist practices and policies, blacks began challenging the constitutionality of racial segregation under the Fourteenth Amendment, but the U.S. Supreme Court consistently ruled in favor of the southern states (Williamson 1984).

The limited social, economic, and political gains made by blacks during the Reconstruction Era came to a staggering halt when Republican Rutherford B. Hayes was elected as the nation's nineteenth president (Foner and Mahoney 1995; Williamson 1984). The Reconstruction Era (1865 – 1877) effectively ended when Hayes withdrew northern military troops from the South in 1877. As a presidential hopeful, Hayes agreed

<sup>&</sup>lt;sup>115</sup> Delaware, Kentucky, Maryland, and Missouri did not leave the Union.

<sup>&</sup>lt;sup>116</sup> The leading professional association in law has long opposed felon disenfranchisement beyond imprisonment, although the practice remains widespread (American Bar Association 1980, 2003b). In 2004, 48 states and the District of Columbia prohibited inmates from voting (only Maine and Vermont allow incarcerated individuals to vote). Thirty-five states prohibit parolees and 31 states prohibit probationers from voting. In 14 states, a felony conviction can result in the permanent loss of voting rights. In 2000, 4.7 million Americans were disenfranchised because of a past or current felony conviction. Similar to the felony disenfranchisement laws of the Reconstruction Era, current felony disenfranchisement laws disproportionately affect blacks males. In Georgia, for example, one out of every eight black males is disenfranchised because of a felony conviction (12.5 percent). In Georgia's capital city, Atlanta, black males are legally ineligible to vote (or serve on juries) because of a prior or current felony conviction (see King and Mauer 2004).

to withdraw military force from the South in exchange for the support of Southern Democrats in winning the presidency. Without the presence of northern troops to ensure the recent constitutional protections granted by the U.S. Congress (i.e., the Fourteenth and Fifteenth Amendments), southern whites were free to reinstitute the racially oppressive policies and practices of the Antebellum South. Shortly after Hayes recalled the military from the South, southern states began enacting wide-sweeping "Jim Crow" laws that mandated the legal separation of blacks from whites (Delaney 1998). Twenty years after the widespread implementation of Jim Crow laws in the South, the U.S. Supreme Court upheld the legality of racial segregation in *Plessy v. Ferguson* (163 U.S. 537 [1896]), ruling that racial segregation was constitutional under the Fourteenth and Fifteenth Amendments as long as facilities for blacks were not inferior to those of whites (i.e., the "separate but equal" doctrine).<sup>117</sup> Shortly after *Plessy*, the Supreme Court began approving racially discriminatory systems of poll taxes and literacy tests in southern states that prevented the vast majority of blacks in these states from voting and serving on juries (see Williams v. Mississippi, 170 U.S. 213 [1898]). For nearly sixty years the Supreme Court would uphold the legality of racial segregation until it was finally struck down in Brown v. Board of Education of Topeka (347 U.S. 483 [1954]).<sup>118</sup> Ten years after *Brown*, Congress passed the Civil Rights Act of 1964, prohibiting discrimination

<sup>&</sup>lt;sup>117</sup> Thirteen years before the *Plessy* decision, in 1883, the U.S. Supreme Court struck down the constitutionality of the Civil Rights Act of 1875 that extended the reach of the Fourteenth Amendment and offered greater protection to blacks under the U.S. Constitution (Williamson 1984).

<sup>&</sup>lt;sup>118</sup> The desegregation of the U.S. Armed Forces would occur six years before the *Brown* decision when, on July 26, 1948, President Harry Truman issued Executive Order 9981, prohibiting racial discrimination in the military. For detailed discussions of *Brown v. Board of Education of Topeka* and its impact on racial integration and equality in the United States over the past 50 years, see Bell (2004), Ogletree (2004), and Rhode and Ogletree (2004).

based upon race, religion, or national origin.<sup>119</sup> Over the next several years, Congress passed the Voting Rights Act of 1965—making obstacles to restrict blacks from voting illegal (e.g., literacy tests)—and the Civil Rights Act of 1968, prohibiting discrimination in the sale, rental, and financing of housing (see, generally, Branch 1988).<sup>120</sup> It was not until 1967, however, that the Supreme Court finally outlawed anti-miscegenation statutes that had in been place since the colonies were initially established over three and a half centuries earlier (see *Loving v. Virginia*, 388 U.S. 1 [1967]).<sup>121</sup>

Many whites continued to violently victimize and murder blacks with very little legal risk after the Civil War—this victimization commonly came in the form of lynching (see Clarke 1998; Senechal de la Roche 1997b; Wells-Barnett 1895). Vigilante justice in the form of the lynch mob was disproportionately directed at freed blacks in the South, and used to uphold the racial social order established in slavery and preserved through Black codes and Jim Crow laws (Kaufman-Osborn 2006a). Moreover, whites participating in lynch mobs were rarely apprehended, charged, and convicted of any

<sup>&</sup>lt;sup>119</sup> Swinton (1990, p. 156) estimates the cost of labor market discrimination against blacks between 1929 and 1969 at \$1.6 trillion (in 1983 dollars).

<sup>&</sup>lt;sup>120</sup> Justice Stephen Breyer (2005, p. 32) remarked, "It took a civil war and eighty years of racial segregation before the slaves and their descendants could *begin* to think of the Constitution as theirs" (emphasis added) (cf. Sunstein 1993).

<sup>&</sup>lt;sup>121</sup> Congressional seniority rules and powerful committee chairmanships gave the one-party South tremendous leverage over legislation for the first two-thirds of the twentieth century (Manza 2000). From the 1930s through the 1960s, Southern Democracts vigorously fought (and often undermined) social policies that potentially threatened the racial hierarchy of the South (Goldfield 1997; Quadagno 1996). For example, Southern Democrats fought to limit social security insurance, unemployment insurance, and Aid to Dependent Children (forerunner to AFDC) (Lieberman 1998). Although these were race-neutral social programs available to both blacks and whites, many Southern Democrats were able to successfully convince their constituents that such programs unnecessarily extended the scope of government (cf. Sears, Sidanius, and Bobo 2000, Chapter 8) and that undeserving blacks would disproportionately benefit from these policies to the detriment of whites (see Gilens 2000; Hancock 2004; Mendelberg 2001; Wilson 1999; but cf. Huber and Lapinski 2006). Southern Democrats would also successfully oppose or significantly weaken key policy proposals related to community action, housing reform, welfare reform, and national childcare (Quadagno 1996).

serious offense (Curriden and Phillips 1999; Tolnay and Beck 1995).<sup>122</sup> Between 1880 and 1930, an estimated 2,462 black men, women, and children were lynched (constituting nearly 90 percent of all lynchings during that period), 94 percent of which were done by whites (Tolnay and Beck 1995). Although evidence suggests that most blacks were lynched for allegedly committing serious offenses (most commonly, raping a white woman), blacks were also lynched for such minor offenses as gambling, voting, arguing with whites, and dating white women (Tolnay and Beck 1995). Perhaps not surprisingly, lynch mobs were primarily used as a way to circumvent the formal judicial process (Messner, Baller, and Zevenbergen 2005). Lynch mobs would often remove their victims from the custody of legal officials and carry out their own "executions." For example, 80 percent of lynch mob victims in Georgia and 94 percent lynch mob victims in Virginia were taken from legal custody (Tolnay and Beck 1995). While lynch mobs often forcibly removed individuals from legal custody, it was not uncommon for legal officials to willfully turn over these individuals to the lynch mobs (see, e.g., Curriden and Phillips 1999).<sup>123</sup>

Not only were blacks disproportionately subjected to illegal executions at the hands of lynch mobs from Reconstruction to the Great Depression, they were also disproportionately subjected to legal executions at the hands of criminal justices officials. During this same period, in addition to the nearly 2,500 illegal hangings of blacks, over

<sup>&</sup>lt;sup>122</sup> Professor James H. Chadbourn's (1933) seminal study, sponsored by the Southern Commission on the Study of Lynching, revealed that less than one percent of whites who participated in lynchings between 1889 and 1932 were arrested and convicted.

<sup>&</sup>lt;sup>123</sup> Ogletree and Sarat (2006) note that there has been remarkable silence concerning America's history of lynching. Prior to the summer of 2005, the U.S. House of Representatives passed four anti-lynching bills and seven presidents, both Democrats and Republicans, endorsed legislation making lynching a federal crime; nonetheless these efforts were consistently opposed by southern senators who used filibuster to prevent the legislation from receiving a formal vote. It was not until June 2005 that the senate voted to formally apologize for failing to previously pass anti-lynching legislation.

three thousand blacks were legally executed (an average of over 50 per year) (Espy and Smykla 2004; Tolnay and Beck 1995). While blacks, on average, constituted 11.4 percent of the nation's population between 1870 and 1930 (U.S. Bureau of the Census 2002b), they represented 47.9 percent of all executions in the United States during same period (Espy and Smykla 2004).

The massive wave of hangings of blacks began to rapidly decline during the Great Depression, but blacks were still disproportionately subjected to death penalty from the time of the Great Depression until the official moratorium was placed on the death penalty after the *Furman* decision. In fact, Clarke (1998) and others (e.g., Kaufman-Osborn 2006a) have argued that lynching declined after the Great Depression primarily because state executioners replaced the lynch mobs in reinforcing the racial hierarchy established during slavery and preserved through various legal and non-legal mechanisms. From 1930 to 1972, 2,034 blacks were executed in the United States, constituting 52.3 percent of all executions (Espy and Smykla 2004), although they only comprised 10.4 percent of the total population during this period (U.S. Bureau of the Census 2002b).

## 4.1.3 The 1970s to the Present

While the civil rights legislation of the mid- and late-1960s did much to improve the legal status of blacks and other racial/ethnic minorities in the United States in the civil arena,<sup>124</sup> the same could not be said for the criminal justice arena (Reiman 1979). Since

<sup>&</sup>lt;sup>124</sup> Blacks would still face obstacles with respect to equal protection under the law well after the Civil Rights Movement. Twenty-four years after the Civil Rights Act of 1964, then-president Ronald Regan vetoed the Civil Rights Restoration Act of 1988 that expanded the reach of non-discrimination laws to private institutions receiving federal funding, although Congress would later override the veto. Three years later, then-president George H.W. Bush vetoed several versions of the Civil Rights Act of 1991—which

the early 1970s, the United States has steadily incarcerated larger proportions of its population (Currie 1998; Mauer 1999). Although the United States constitutes less than five percent of the world's population,<sup>125</sup> it houses nearly 25 percent of the world's incarcerated population (U.S. Department of Justice 2004a). In 1971, the incarceration rate (prison and jail) in the United States was 143 per 100,000. This rate increased to 241 per 100,000 in 1981, 482 per 100,000 in 1991, and 714 per 100,000 in 2003 (U.S. Department of Justice 2004a). In 2003, over 2 million individuals were incarcerated in prisons and local jails (U.S. Department of Justice 2004a)<sup>126</sup> and 4.5 million individuals were either on probation or parole (Feldman, Schiraldi, and Ziedenberg 2001).<sup>127</sup> The dramatic rise in incarceration has disproportionately impacted blacks and Hispanics, particularly black and Hispanic males. Blacks and Hispanics comprise nearly two-thirds (63 percent) of prison and jail inmates, although these two groups only comprise onequarter (25 percent) of the nation's population (U.S. Department of Justice 2004a). By contrast, in the 1950s, over two-thirds individuals incarcerated in the United States were white (Tonry 1995). While currently comprising approximately 13 percent of the United States population, blacks constitute 43 percent of state prison inmates, 40 percent of federal prison inmates, and 40 percent of local jail inmates (U.S. Department of Justice 2004a). In 2003, the incarceration rates for black and Hispanic males were well above the national average of 1,331 per 100,000 males. Blacks and Hispanics males were incarcerated at rates of 4,834 per 100,000 and 1,778 per 100,000, respectively, while

strengthened existing civil right laws and allowed individuals to receive damages in cases of intentional employment discrimination—before finally reversing himself and signing the bill into law.

<sup>&</sup>lt;sup>125</sup> The current population of the United States is approximately 290 million. The world's population is estimated around 6.3 billion (U.S. Bureau of the Census, Population Division/International Programs Center 2004).

<sup>&</sup>lt;sup>126</sup> In 2003, just four states—Texas, California, Florida, and New York—accounted nearly one-quarter of the total incarcerated population in the United States (U.S. Department of Justice 2004a).

<sup>&</sup>lt;sup>127</sup> Over 16 million people in the United States have a felony conviction (Manza and Uggen 2005, p. 9).

white males were imprisoned at nearly half of the national average (681 per 100,000) (U.S. Department of Justice 2004a). Twelve percent of black males and 3.7 percent of Hispanic males in their twenties are in prison or jail, compared with 1.6 percent of white males (U.S. Department of Justice 2004a) and one in three black males in their twenties is either incarcerated, on probation, or on parole (Mauer 1999; Miller 1996; U.S. Department of Justice 2004a). Black and Hispanic females are also incarcerated at a significantly higher rate than white females. In 2003, both black and Hispanic females were incarcerated at rates well above the national average of 119 per 100,000 (352 and 148, respectively), while white females were incarcerated at a significantly lower rate (75 per 100,000) (U.S. Department of Justice 2004a).<sup>128</sup> Although the causes of these drastic racial differences in incarceration rates continue to be debated by scholars (see Arvanites and Asher 1998; Bennett, DiIulio, and Walters 1996, pp. 43–46; Brown et al. 2003, Chapter 4; Dilulio 1996; Greenberg and West 2001; Jacobs and Carmichael 2001; Michalowski and Pearson 1990; Pettit and Western 2004; Sampson and Lauritsen 1997b; Smith 2004; Sorensen and Stemen 2002; Tonry 1995, Chapter 2; Western 2006, Chapter 2; Wilbanks 1987), a wide range of evidence suggests that these differences cannot be completely attributed to the differential criminal involvement of blacks and Hispanics, and are partly attributable to the differential treatment that blacks and Hispanics receive in the criminal justice system (Blumstein 1993; Chambliss 1999; Sampson and Lauritsen 1997a; Wacquant 2005a). For example, examining arrest and incarceration data in Pennsylvania between 1991 and 1995, Austin and Allen (2000) discover that only 42 percent of the racial imbalance in prison admissions was explained by arrest differentials.

<sup>&</sup>lt;sup>128</sup> In the 1990s, the probability of serving time in a state or federal prison was 4 percent for whites, 16 percent for Hispanics, and 29 percent for blacks (Tonry 1995).

With respect to non-violent drug offenses, they find that black arrest rates only accounted for 26 percent of the racially disproportionate drug sentences, suggesting that disparities are highest where discretion is greatest.

An enormous increase in the number of individuals on death row has also occurred since the early 1970s. In 1971 (the year before the Supreme Court commuted the all of the death sentences after the *Furman* ruling), 641 individuals were on death row (0.32 per 100,000) (Death Penalty Information Center 2008). At the end of 2003, there were 3,504 people awaiting execution in the United States (1.21 per 100,000)—an increase of 546 percent (Death Penalty Information Center 2008). Similar to the premoratorium years, blacks continue to be disproportionately represented among those sentenced to death and those executed since capital punishment was reinstated, although their proportion relative to whites has significantly decreased from earlier periods. Blacks constituted 12.3 percent (on average) of the total population from 1976 to 2002 (U.S. Bureau of the Census 2002b), but comprised 42 percent of death row inmates and 34 percent of total executions (281 executions) during the same time period (Death Penalty Information Center 2008).<sup>129</sup> Much stronger racial differences exist for the race of the victim for individuals executed since 1976. Since 1976, blacks have comprised approximately 50 percent of all homicide victims in the United States (Fox 2005), yet over 80 percent of executions have been carried out on individuals sentenced to death for killing white victims (Death Penalty Information Center 2008).

<sup>&</sup>lt;sup>129</sup> Although not as pronounced, racial disparities in the capital punishment system are evident for females as well. Although black females have a homicide victimization rate that is over four times greater than white females (Fox 2005), female offenders convicted of murdering white victims are over five times more likely to be on death row than female offenders convicted of murdering black victims (.63 versus .12) (Streib 2002a). Similarly, while black females comprise approximately twelve percent of all females in the United States, they constitute 31 percent of the females currently on death row (Streib 2002a).

Although scholars acknowledge that early data on executions in the United States are far from complete, they note that there have been at least 15,645 legal executions in the United States in the past four centuries (Death Penalty Information Center 2008; Espy and Smykla 2004). Over half of these executions have been of black offenders (50.5 percent), although blacks have never comprised over 20 percent of the nation's total population in any given year since the government began collecting population statistics and they have not constituted more than 15 percent of the nation's population since 1850 (U.S. Bureau of the Census 2002b). Racial differences in the administration of capital punishment become more pronounced when examining the death penalty for rape (White 1991). As mentioned in Chapter Two, anti-death penalty attorneys and scholars began attacking the constitutionality of the death penalty based upon racial disparities in the administration of capital punishment in southern states for the crime of rape in the early 1950s (see Wolfgang and Riedel 1975). During the 400 years of capital punishment in this country, at least 947 individuals have been executed for the crime of rape. Of these individuals, 89.2 percent (843) have been black and nearly all were convicted for raping white women (Espy and Smykla 2004).

Death penalty opponents point to the history of slavery, slave codes, Black Codes, Jim Crow laws, and the uneven administration of criminal justice in this country as evidence that the death penalty has always been imposed in a racially discriminatory manner and, like lynchings of old, have been and continue to be used to enforce the legal and social superiority of whites over blacks (see Bright 1995a; Ogletree 2002). Abolitionists also point to the disproportionate arrests and incarceration of blacks (see, e.g., Chiricos and Crawford 1995), as well as laws that differentially (and deleteriously) impact blacks offenders (see, e.g., Donziger 1996; Provine 2006), as evidence that the criminal justice system, as a whole, has yet to adopt "color-blind" practices and policies.<sup>130</sup> Furthermore, death penalty opponents argue continuing arbitrariness and capriciousness in the capital charging-and-sentencing process, in addition to racial bias, underscores the fact that the procedural changes implemented after *Furman* have done very little to guarantee the death penalty is being applied in a constitutionally acceptable manner (see, e.g., Bright 1995b; Nakell and Hardy 1987).

While acknowledging the existence of past problems of racial bias in the administration of criminal justice, in general, and the death penalty in the particular, proponents of capital punishment argue that capital punishment is both morally permissible and necessary (see Lakoff 2002, pp. 208–209; Pojman 2003, pp. 54–58), the procedural safeguards currently in place have eliminated arbitrariness and racial/ethnic discrimination in the criminal justice system (Wilbanks 1987), and the death penalty is

<sup>&</sup>lt;sup>130</sup> For example, in 1984 and 1988, two federal sentencing laws were enacted that made the punishment for selling crack cocaine 100 times more severe than the punishment for selling an equal amount of powder cocaine. A person convicted of selling five grams of crack cocaine received the same sentence as someone convicted of selling 500 grams of powder cocaine—a five-year mandatory minimum sentence. Although two-thirds of crack cocaine users are white or Hispanic, 84.5 percent of individuals convicted of selling crack are black (10.3 percent are white and 7.1 percent are Hispanic). In contrast, 58 percent of defendants convicted of selling powder cocaine were white, 26.7 percent were black, and 15 percent were Hispanic. In 1995, the USSC studied the racially disparate effects of these laws and recommended equalizing the punishment for powder and crack cocaine (see U.S. Sentencing Commission 1995). In response to the USSC's report, the DOJ, under then-Attorney General Janet Reno, urged Congress to reject the USSC's recommendation, citing the greater dangers associated with the trafficking of crack cocaine (U.S. Department of Justice 1995). The U.S. Congress ultimately followed the advice of the DOJ, marking the first time that Congress went against the recommendation of the commission. Following Congress's rejection of the recommendation, then-president William Jefferson Clinton signed the rejection into law (Feldman *et al.* 2001).

The deleterious impact of these differential punishments on the black community has been undeniable. Prior to 1986, the average drug sentence for blacks was 6 percent longer than that of whites, but four years later the average sentence for black drug offenders was 93 percent higher (Tonry 1995). In addition to lengthening sentences, the new "war on drugs" also had a significant impact on black imprisonment rates because police more aggressively enforced drug laws in areas where blacks and Hispanics resided, net of actual behavior (see Blumstein 1993, p. 753). For example, in 1983, the year before the aforementioned sentencing guidelines were enacted, 63 percent of the prison commitments in Virginia involved whites and 37 percent involved minorities (primarily blacks). Six years later the pattern reversed, with 65 percent of prison commitments involving minorities and 34 percent involving whites, although drug *use* by whites and minorities remained relatively constant during this period (Cross 2003).

now strictly reserved for the most heinous crimes and criminals (Cassell 2003, pp. 209– 12). In response to the disproportionate number of blacks and Hispanics currently on death row and that have been executed since *Furman*, retentionists posit that blacks and Hispanics are more likely to be perpetrators of criminal homicide, their homicides tend to be more aggravated (i.e., more heinous), and they tend to have more extensive (and violent) criminal backgrounds (Cassell 2003, pp. 201–205; Wilbanks 1987). Some analysts have also argued that the heinousness of the white-victim homicides accounts for the fact that most executions carried out since Furman (over 80 percent) have been of defendants convicted of killing white victims and the fact that most people currently on death row have murdered white victims (83 percent) (see, e.g., Kleck 1969, 1981; Klein and Rolph 1989). Unfortunately, early debates over the impact of race on the capital charging-and-sentencing process were rarely grounded in solid empirical research, as both abolitionists and retentionists relied primarily on anecdotal evidence and aggregate raw statistics when making their claims (see, generally, Baldus 1980). This began to change in the 1930s, however, when analysts began attempting to systematically assess the role of race and other extra-legal factors in the administration of the death penalty. Since that time a large body of research literature has emerged on the role of extra legal factors in the capital charging-and-sentencing process-particularly the role of race and region (see, generally, Baldus and Woodworth 1997, 2003; U.S. General Accounting Office 1990). These studies are discussed in greater depth below and their contribution to the question of the presence of arbitrariness and bias is assessed.

# 4.2 RESEARCH ON EXTRA-LEGAL FACTORS AND CAPITAL PUNISHMENT

#### 4.2.1 Pre-Furman Empirical Studies

Social scientists have been conducting research on the influence of extra-legal factors—particularly race/ethnicity—in criminal sentencing at least since the late 1920s (see, e.g., Sellin 1928).<sup>131</sup> These studies typically compared the sentence severity of white and non-white offenders (usually black offenders) for similar crimes. Although somewhat crude by contemporary scientific standards, several of these studies discovered that blacks were typically more likely to receive longer sentences than whites for similar offenses *and* that the impact of race on sentencing decisions was influenced by the region where the case was tried (see, e.g., Sellin 1935). In the 1930s, when the United States government started systematizing the collection, compilation, and publication of official crime statistics, the legal and academic communities (mainly the academic community, see Baldus et al. 1998) began increasingly turning their attention to apparent disparities in the administration of capital punishment. Scholars were also becoming increasingly aware that the racially discriminatory treatment of blacks in the criminal justice system appeared to be most pronounced when blacks were accused of challenging or violating the racial hierarchy—particularly victimizing whites (see Du Bois [1940] 1968; Wacquant 1997). Recognizing this fact, Johnson (1941) conducted one of the earliest studies of death penalty sentencing that simultaneously considered both the race of the offender and the race of the victim. Examining more than 300 murder cases from

<sup>&</sup>lt;sup>131</sup> In fact, much evidence suggests that social scientific research on racial disparities in criminal sentencing was being conducted by black sociologists at the turn of the twentieth century; however this body of work was largely ignored or negated by mainstream criminologists at the time (Gabbidon 2001; Wright 2002a, b). The Atlanta Sociological Laboratory's study, *Some Notes on Negro Crime, Particularly in Georgia*, published in 1904, discovered inequities in the length of sentences between blacks and whites and highlighted the strong racial bias in the convict-lease system (see Du Bois 1904).

Richmond, Virginia and five counties in North Carolina between 1930 and 1940, he discovered that 32 percent of black-offender/white-victim homicides received the death sentence compared to 13 percent of white-offender/white-victim homicides. Johnson also found that death sentences were imposed in 17.5 percent of all white-victim cases, but only in four-tenths of one percent of black-victim cases.<sup>132</sup> Garfinkel (1949) extended Johnson's analysis by examining 673 homicide cases (821 offenders) from ten countries in North Carolina during the same time period and looking at grand jury indictments and prosecutorial charging decisions, as well as sentencing decisions. Garfinkel found that blacks accused of murdering whites were significantly more likely to be indicted for first-degree murder, charged with first-degree murder by prosecutors (only first-degree murders were eligible for the death penalty in North Carolina during this period), and sentenced to death. In particular, black-offender/white-victim homicides were nine times more likely to result in a first-degree murder conviction than blackoffender/black-victim homicides (43 percent vs. 5 percent) and nearly three times more likely to result in a first-degree murder conviction than white-offender/white-victim homicides (43 percent vs. 15 percent). With respect to sentencing, black-offender/whitevictim homicides were over nine times more likely to result in a death sentence than black-offender/black-victim homicides (37 percent vs. 4 percent) and over three times more likely to result in a death sentence than white-offender/white-victim homicides (37 percent vs. 11 percent). Garfinkel also discovered that white-victim homicides, regardless of the race of the offender, were significantly more likely to result a firstdegree murder conviction than black-victim homicides (24 percent vs. five percent) and

<sup>&</sup>lt;sup>132</sup> Johnson (1941) discovered that there was only one homicide indictment for a white-offender/blackvictim homicide in Richmond, Virginia and only three homicide indictments for white-offender/blackvictim homicides in North Carolina during the entire 11 year period.

significantly more likely to result in a death sentence than black-victim homicides (18 percent vs. 4 percent). Early research also suggested that racial disparities in the capital punishment system were evident outside of the South. Examining 204 homicide cases in Philadelphia, Pennsylvania in 1970, Zimring and colleagues (1976) discovered that 65 percent of defendants convicted of murdering a white victim were sentenced to either death or life imprisonment, whereas only 25 percent of defendants convicted of murdering a black victim received a death or life sentence. Moreover, black defendants convicted or life imprisonment than black defendants convicted of murdering another black person.

Although these early studies consistently uncovered patterns of racial disparity in capital charging or sentencing, they were criticized for not being able to adequately differentiate between simple racial disparity and racial discrimination because they failed to consider the legitimate legal characteristics that could be correlated with the race of the defendant or the race of the victim. To address this shortcoming, Wolfgang and Riedel (1973, 1975) analyzed data on over 3,000 rape convictions from eleven southern states between 1945 and 1965, collecting detailed information about the offender, the victim, the facts of the crime, and the trial. They discovered that black offenders were over six times more likely to be sentenced to death for rape than white offenders and black offenders convicted of raping white victims were nearly eighteen times more likely to be sentenced to death than all other racial combinations (Wolfgang and Riedel 1973). After taking into account the presence of a contemporaneous felony (e.g., armed robbery, burglary, *et cetera*), Wolfgang and Riedel found that blacks were 20 times more likely to be sentenced to death than whites. Even after considering over two-dozen legally

relevant variables pertaining to the offender, the victim, the offense, and trial characteristics, they discovered that these racial differences remained and that race was the single most important predictor of whether a defendant received a death sentence (see also LaFree 1989).

While Wolfgang and Riedel focused on the death penalty for the crime of rape in which the victim was not killed, Baldus and colleagues (1990) investigated 300 homicide defendants in Georgia who were tried and convicted before Furman (see also Baldus and Woodworth 1983; Baldus, Woodworth, and Pulaski 1985). Similar to Wolfgang and Riedel's rape study, Baldus *et al.* collected data on the specific characteristics of the case in order to ascertain the heinousness of the crime and compare cases that were roughly similar with respect to their level of aggravation. In particular, they divided cases into six levels of aggravation (ranging from lowest to highest) and examined the proportion of all defendants sentenced to death at each level and then the proportion of defendants sentenced to death based on their race and the race of their victim(s) at each level. Baldus *et al.* argued that any racial differences *within* a level would be more likely attributable to the actual influence of race, rather than an important case characteristic (see, generally, Baldus 1980; Baldus and Cole 1977). They discovered that, although Georgia had one of the highest death sentencing rates in the nation, death sentences were extremely rare: only 15 percent of the 294 murder defendants under investigation received a death sentence. Without considering the level of aggravation of the case, black offenders were over twice as likely to be sentenced to death than white offenders (19 percent vs. 8 percent); killers of white victims were nearly twice as likely to be sentenced to death than killers of black victim (18 percent vs. 10 percent); and black-

offender/white-victim homicides were more than three times more likely to result in a death sentence than any other offender/victim combination. After taking into account the level of aggravation of the case, they discovered that black-offender/white-victim cases were, on average, nearly four times (3.9) more likely to result in a death sentence than white-offender/white-victim cases. The differences between black-offender/white-victim and white-offender/white-victim cases, however, were not uniform throughout the six levels of aggravation. At the lowest level of aggravation (level 1), almost no cases resulted in a death sentence and at the highest level of aggravation (level 6) almost all cases resulted in a death sentence.<sup>133</sup> Racial disparity, however, was greatest in the midrange cases: at level 3, black-offender/white-victim cases were over 14 times more likely to result in a death sentence than white-offender/white-victim cases; at level 5, blackoffender/white-victim cases were over twice as likely to result in a death sentence than white-offender/white-victim (Baldus et al. 1990, p. 144). Very few black-victim cases resulted in a death sentence—10 percent (12 out of 115 cases)—and only one whiteoffender/black-victim case resulted in a death sentence (at the fifth level of aggravation) (see also Baldus and Woodworth 2004).

In addition to analyzing the impact of race in charging, indictment, conviction, and sentencing decisions, scholars also examined racial differences in commutation decisions. No death sentence handed down by a judge or capital jury is absolutely final. In every death penalty state, death sentences may be commuted to a life sentence or a term of years by the governor or the state's pardons and parole board. These decisions

<sup>&</sup>lt;sup>133</sup> Five percent of black-offender/white-victim cases resulted in a death sentence at level 1, whereas no white-offender/white-victim cases resulted in a death sentence at level 1. At level 2 and level 5, black-offender/white-victim cases were slightly more likely to result in a death sentence than white-offender/white-victim cases (1.3 and 1.7 times more likely, respectively).

may be influenced by a host of political and social factors, including, potentially, the race of the defendant or victim (Pridemore 2000). In one of the earliest examinations of the role of race in the commutation decisions, Mangum (1940) discovered that—over a ten year period in Florida (1928 – 1938)—75 percent of blacks defendants sentenced to death were ultimately executed, whereas 56 percent of white defendants sentenced to death were executed. (Because commutation requests are made in nearly every single case, these percentages represent the proportion of requests rejected by the governor or state pardons and parole board.) Examining executions in North Carolina between 1933 and 1939, Johnson (1941) discovered that 74 percent of defendants sentenced to death for killing white victims were executed, whereas 65 percent of defendants sentenced to death for killing black victims were executed. Moreover, he found that 81 percent of black offenders sentenced to death were ultimately executed. Analyzing commutations in Texas between 1924 and 1968, Koeninger (1969) noted that 76 percent of all commutations were of white defendants and that 99 percent of black defendants sentenced to death were executed (see also Marquart, Ekland-Olson, and Sorensen 1994). Although evidence of racial disparity in commutation decisions appear to be primarily concentrated in the southern states (see Bedau 1964, 1965; Kleck 1981), similar disparities have also been discovered outside the South. For example, Wolfgang and colleagues (1962) found that whites sentenced to death for felony-murder were nearly three times more likely to have their death sentences commuted than blacks sentenced to death for felony murder (17.4 percent vs. 6.3 percent).

As noted in Chapter Two, statistical evidence of racial disparity and racial discrimination in the administration of the death penalty was first presented in *Hampton* 

v. Commonwealth (58 S.E.2d 288 [Va. 1950]). This case, known as the "Martinsville Seven," involved seven young black men sentenced to death for raping a white woman in Virginia in 1949.<sup>134</sup> Rather than attacking the death sentences on procedural grounds on appeal, the attorneys of the defendants presented statistical evidence showing 45 black men had been executed for rape between 1908 and 1949, but no white man had been executed (see Rise 1995). The attorneys also noted that twice as many black men had been sentenced to life imprisonment for rape than white men. The both the trial court and the Virginia Supreme Court of Appeals rejected the defendants' claim of racial discrimination and the U.S. Supreme Court denied certiorari (Hampton v. Virginia 339 U.S. 989 [1950]). One month after the Supreme Court's refusal to grant certiorari, all seven men were executed within 72 hours of one another. Although the defense's claims did not prevail, the case was largely credited for launching the NAACP's campaign against the death penalty (Rise 1995). Future attempts to use statistical evidence to challenge the constitutionality of the death penalty on racial grounds before the moratorium was placed on capital punishment would also be unsuccessful (see Chapter Two).

## 4.2.2 Post-Furman Empirical Studies

The U.S. Supreme Court decided to commute all existing death sentences and place a moratorium on the death penalty after the *Furman* decision because it believed that capital punishment, as administered, violated the Eighth Amendment (see Chapter Two). In particular, the Supreme Court held that the existing statutes failed to distinguish capital murder from non-capital murders *and* they failed to offer jurors any guidance with

<sup>&</sup>lt;sup>134</sup> The seven young men were: Joe Henry Hampton, Frances Desales Grayson, Frank Hairston, Jr., Howard Hairston, James Luther Hairston, Booker T. Millner, and John Clabon Taylor.

respect to sentencing in capital cases. Statistical evidence of the racially discriminatory administration of capital punishment was presented to the court in *Furman* (i.e., Bedau 1964, 1965; Carter and Smith 1969; Koeninger 1969; Wolfgang *et al.* 1962), but the court did not use this evidence as the basis for their decision. In fact, only Justice Douglas and Justice Marshall cited this statistical evidence in their opinions.<sup>135</sup> But although the majority of the Supreme Court Justices passed silently over the issue of racism in the administration of capital punishment, abolitionists were well aware that the collection of such data was still necessary because death states quickly revised their capital statutes to address the court's concerns in *Furman* and began sentencing defendants to death again (Haines 1996).

Shortly after states began reestablishing their capital punishment systems, scholars began conducting studies of the death penalty in order to determine whether the post-*Furman* capital statutes had eliminated the arbitrariness (and potential racial bias) that was rampant under the pre-*Furman* capital systems. Although some scholars found some evidence that the new statutes had eliminated or significantly reduced the amount of arbitrariness, caprice, and bias in the capital process in the years immediately following the *Furman* decision (for a discussion, see Baldus *et al.* 1990), the vast majority of studies discovered that race still had a strong influence at one or more of the decision stages in the capital charging-and-sentencing process (Baldus *et al.* 1998). In one of the first studies of post-*Furman* death sentences, Bowers and Pierce (1980) examined death sentences in Florida, Georgia, Ohio, and Texas in the five years following *Furman* (these four states were responsible for over 70 percent of the death

<sup>&</sup>lt;sup>135</sup> By comparison, when the Supreme Court ruled that racial discrimination in public schooling was unconstitutional in *Brown v. Board of Education of Topeka*, nearly 20 years before the *Furman* decision, its finding was heavily influenced by seven social science studies (see Rosen 1972).

sentences during this period) and discovered that black-offender/white-victim homicide cases were still more likely to result in a death sentence than any other offender/victim racial combination. Subsequent research also discovered that white-victim homicide cases (Barnett 1985; Berk and Lowery 1985; Bienen et al. 1988; Ekland-Olson 1988; Foley 1987; Gross and Mauro 1984, 1989; Nakell and Hardy 1987; Paternoster 1984; Paternoster and Kazyaka 1988) and black-offender/white-victim homicide cases (Baldus and Woodworth 1983; Bowers 1983; Paternoster 1984; Radelet 1981; Vito and Keil 1988; Zeisel 1981) were more likely to be noticed for the death penalty or sentenced to the death sentence in the post-*Furman* era. Still other evidence suggested that black defendants were more likely to have their sentences affirmed by the state's high court than white defendants (see, e.g., Radelet and Vandiver 1983). Baldus and colleagues (1986a) conducted one of the most wide-sweeping analyses by examining death sentences in 24 states from 1977 through 1984.<sup>136</sup> Comparing cases in which a contemporaneous felony was present, they discovered that white-victim cases were significantly more likely to result in a death sentence in ten states and white-victim cases were marginally more likely to result in a death sentence in eight other states. With respect to the race of the offender, they did not find a consistent pattern of racial discrimination—in some states, black offenders were more likely to be sentenced to death, in other states white offenders where likely to be sentenced to death, and still in other states, there were no statistically significant race-of-offender effects. Gross and Mauro (1989) also conducted a multiple-state study (Arkansas, Florida, Georgia, Illinois, Mississippi, North Carolina, Oklahoma, Virginia) of the death penalty from 1976 through

<sup>&</sup>lt;sup>136</sup> The 24 states were: Alabama, Arizona, Arkansas, California, Delaware, Florida, Georgia (1974 – 1984), Illinois, Indiana, Kentucky, Louisiana, Maryland, Mississippi, Missouri, Nebraska, Nevada, North Carolina, Ohio, Oklahoma, Pennsylvania, South Carolina, Texas, Virginia, and Washington.

1980 and considered the commission of a contemporaneous felony, the relationship between the offender and victim, and the number of victims when comparing cases (see also Gross and Mauro 1984). They discovered that white-victim homicides increased the likelihood of a death sentence by a factor of four in Illinois, five in Florida, and seven in Georgia. The methodological approaches of these early post-*Furman* studies significantly varied, however, and were criticized by many death penalty advocates for failing to consider a broad range of legal case characteristics that could be correlated with race and legitimately account for these racial patterns.

In an attempt to address these concerns, Baldus and colleagues (1990) conducted what was considered the most comprehensive study of the death at that time (see Kennedy 1988). As mentioned in Chapter Two, Baldus *et al.* collected data for two separate studies on Georgia's capital charging-and-sentencing process and these data were presented in to the U.S. Supreme Court in McCleskey (for a detailed description of the study, see Chapter Two). The first study, the Procedural Reform Study (PRS) contained data on 700 murder cases in Georgia from 1969 through 1978. The second study, the Charging-and-Sentencing Study (CSS) collected data on 1,066 murder and voluntary manslaughter convictions in Georgia from 1973 through 1979. Baldus et al. simultaneously considered 23 cases factors in the PRS and discovered that white-victim cases were 4.3 times more likely to result in a death sentence than black-victim cases. The larger of the two studies, the CSS, collected data on over 230 legitimate and illegitimate case characteristics. After controlling for 39 relevant case characteristics, Baldus *et al.* discovered that prosecutors were 3.2 times more likely to seek the death sentence in white-victim cases than black-victim cases and juries were 4.3 times more

likely to impose the death sentence in white-victim cases than black-victim cases (Baldus *et al.* 1990).<sup>137</sup> They also discovered that prosecutors treat black-victim cases with greater leniency with respect to plea-bargaining for a reduced charge. Prosecutors were 4.2 times more likely to accept a voluntary manslaughter plea in black-victim cases than white-victim cases (Baldus *et al.* 1990, p. 361 n.47).<sup>138</sup>

Although the Supreme Court rejected McCleskey's claims by ruling that Baldus *et al.*'s statistical data were incapable of showing purposeful discrimination on the part of prosecutors, judges, and capital juries, researchers continued to conduct studies examining the impact of race and other legal factors on the capital charging-and-sentencing process. Following the Supreme Court's decision in *McCleskey*, the U.S. General Accounting Office (GAO) commissioned a study to evaluate all of the available evidence on the role of race in the capital punishment process (see U.S. General Accounting Office 1990). The report revealed that 82 percent (23 of 28) of all methodologically rigorous (and non-duplicate) studies examining the capital charging-and-sentencing process revealed that the race of the victim had a significant effect on the

 $<sup>^{137}</sup>$  Although the odds-multiplier was smaller for the race-of-victim effect in the prosecutorial discretion model than in the jury discretion model (3.2 versus 4.3), race-of-victim had significantly more explanatory power in the prosecutorial discretion models (as measure by its contribution to overall model fit) (Baldus *et al.* 1990, p. 167).

<sup>&</sup>lt;sup>138</sup> Baldus *et al.* report an odds-multiplier (i.e., odds ratio) of .24 for white-victim cases with respect to prosecutorial discretion in accepting a plea for voluntary manslaughter. That is, white-victim cases were .24 times less likely to result in a plea for voluntary manslaughter compared to black-victim cases. The reciprocal of that odds-ratio  $(.24^{-1} = 4.16)$  represents the odds that black-victim cases result in a plea for voluntary manslaughter compared to white-victim cases. In other words, defendants charged with murdering black victims are over four times more likely to receive a plea for a reduced charge (i.e., voluntary manslaughter) than defendants charged with murdering white victims.

As mentioned in Chapter Two, Georgia prosecutors are also significantly more likely to seek the death penalty in cases that involve black offenders and white victims. Prosecutors sought the death penalty in 70 percent of black-offender/white-victim cases, 32 percent of white-offender/white-victim cases, 19 percent of white-offender/black-victim cases, and 15 percent of black-offender/black-victim cases. Georgia capital juries imposed the death sentence in 22 percent of black-offender/white-victim cases, 8 percent of white-offender/white-victim cases, 3 percent of black-offender/white-victim cases, 8 percent of white-offender/black-victim cases, 8 percent of black-offender/black-victim cases, 3 percent of white-offender/black-victim cases, and one percent of black-offender/black-victim cases.

probability of receiving a death sentence.<sup>139</sup> Offenders accused of murdering white victims, irrespective of their own race, were significantly more likely to be charged with capital murder and be sentenced to death. Studies conducted since the GAO report in California (Pierce and Radelet 2005; Rohrlich and Tulsky 1996; Weiss et al. 1999), Florida (Radelet and Pierce 1991), Illinois (Pierce and Radelet 2002), Maryland (Baldus and Woodworth 2001; Paternoster et al. 2004), Missouri (Sorensen and Wallace 1995), New Jersey (Baldus 1991; Weisburd and Naus 2001), North Carolina (Paternoster 1991; Unah and Boger 2001), Ohio (Williams and Holcomb 2001), Pennsylvania (Baldus et al. 1998), and South Carolina (Paternoster 1991; Songer and Unah 2006) all reveal that white-victim homicides are more likely to be noticed for the death penalty or result in a death sentence than black-victim homicides.<sup>140</sup> Furthermore, researchers examining the capital punishment process in Arizona (Thomson 1997), Illinois (Pierce and Radelet 2002), Maryland (Baldus and Woodworth 2001; Paternoster et al. 2004), Missouri (Sorensen and Wallace 1995), and New Jersey (Weisburd and Naus 2001) discovered that black-offender/white-victim homicides were more likely to be notice for the death penalty or result in a death sentence than any other offender/victim racial combination. In fact, the only studies conducted on the death penalty since the GAO report that did not

<sup>&</sup>lt;sup>139</sup> Studies were considered methodological rigorous if they received a satisfactory rating on five dimensions: (1) study design, (2) sampling, (3) measurement, (4) data collection, and (5) analytical technique. Of the 52 studies originally examined, 28 studies met these criteria: Arkin (1980), Baldus, Woodworth, and Pulaski (1990), Barnett (1985) Berk and Lowery (1985), Bienen, Weiner, Denno, Allison, and Mills (1988), Bowers and Pierce (1980), Bowers (1983), Ekland-Olson (1988), Foley (1987), Foley and Powell (1982), Gross and Mauro (1984), Keil and Vito (1989), Keil and Vito (1996), Kleck (1981), Klein *et al.* (1987), Klein and Rolph (1989), Klemm (1986), Lewis, Mannle, and Vetter (1979), Murphy (1984), Nakell and Hardy (1987), Paternoster and Kazyaka (1988), Radelet (1981), Radelet and Pierce (1985), Radelet and Vandiver (1983), Riedel (1976), Smith (1987b), Vito and Keil (1988), and Zeisel (1981).

<sup>&</sup>lt;sup>140</sup> Lee's (2007) recent study of death penalty charging decisions in California also revealed that defendants charged with murdering Hispanic-victims were significantly less likely to be noticed for the death penalty than defendants charged with murdering white-victims.

find race-of-victim or race-of-defendant effects were conducted in Colorado (Baldus, Woodworth, and Pulaski 1986b) and Nebraska (Baldus *et al.* 2002).<sup>141</sup>

Of particular note are the recent studies conducted in Maryland and North Carolina. These two studies represent the most comprehensive statistical examinations of the death penalty since the Baldus *et al.*'s (1990) landmark study in *McCleskey*.<sup>142</sup> Examining 1,311 death eligible cases in Maryland from 1978 through 1999 and considering a total of 123 legitimate and illegitimate case characteristics, Paternoster and colleagues (2004, pp. 35–39) discovered that white-victim cases were 1.8 times more likely to be noticed for the death penalty and three times more likely to have their death notices remain than black-victim cases. Black-offender/white-victim cases were over twice (2.1) as likely to be noticed for the death penalty than white-offender/white-victim cases, 2.6 times more likely to be noticed for the death penalty than white-offender/blackvictim cases, and 2.8 times more likely to be noticed for the death sentences, blackoffender/victim racial combinations. With respect to death sentences, blackoffender/white-victim cases were 2.6 times more likely to result in a death sentence than

<sup>&</sup>lt;sup>141</sup> Race-of-victim effects were not statistically significant in the Colorado study because of the extremely small sample size. Baldus *et al.* (1986b) were only able to identify 179 death-eligible cases over a five year period (1979 – 1984) and only four cases resulted in a death sentence. All four of these cases, however, were white-victim cases although 31 percent of all death-eligible cases were nonwhite-victim cases (see also Anderson 1991). A recent study conducted by Michael Radelet (2003) that examines Colorado's entire history of executions (1859 – 1972) reports that 25 percent of the state's 103 executions have been carried out against members of racial/ethnic minorities and 89.2 percent individuals executed in Colorado have were convicted of murdering white victims. In a follow-up to this study, Radelet and colleagues (2006) examine two decades of capital charging patterns in Colorado (1980 – 1999) and discover prosecutors were 4.2 times more likely to seek the death sentence in white-victim cases than in black-victim cases (p. 579).

<sup>&</sup>lt;sup>142</sup> A third comprehensive statistical study was conducted in New Jersey (see Weisburd and Naus 2001). The study examined over 400 death eligible cases since 1982 and considered 110 legitimate and illegitimate case characteristics. Due to the small number of death eligible cases and the small number of total death sentences (55) the statistical models were very unstable and the results varied considerably depending on the factors used in the analytical models. Nonetheless, tentative results indicate that white-victim cases were significantly more likely to advance to the penalty phase than nonwhite-victim cases (see also Baldus 1991).

white-offender/white-victim cases, 3.8 times more likely to result in a death sentence than black-offender/black-victim cases, and 11.3 times likely to result in a death sentence than other offender/victim racial combinations. Paternoster and Brame also discovered that the racial disparities that occur at the early stages of the capital charging-and-sentencing process were not corrected through the advancing stages (see also Paternoster *et al.* 2004). Unah and Boger's (2001) analysis of nearly 2,000 murder cases in North Carolina between 1993 and 1997 also revealed strong race-of-victim effects. After considering a total of 113 legal and extra-legal factors, they discovered that prosecutors were three times more likely to seek the death penalty and judges and juries were 3.5 times more likely to impose the death sentence in white-victim cases than non-white victim cases.

Expanding the GAO's original study to include research conducted through 1997, Baldus and Woodworth (1997) discovered that death penalty studies have been conducted in all 38 death penalty states. Ninety-three percent of these studies found evidence of a race-of-victim effect and nearly 50 percent of these studies find evidence of a race-of-defendant effect. Studies from eleven of these states have been "wellcontrolled" studies, meaning they have taken into account ten or more non-racial case characteristics (Baldus and Woodworth 1997, 2003; Baldus *et al.* 1998, p. 1661 n.71).<sup>143</sup> For each of the states in which a well-controlled study has been conducted, a less wellcontrolled study was also conducted. In six of these states, the racial disparities were

<sup>&</sup>lt;sup>143</sup> Well-controlled studies were conducted in California, Colorado, Georgia, Kentucky, Maryland, Mississippi, Nebraska, New Jersey, North Carolina, Pennsylvania, and South Carolina.

found to be stronger in the well-controlled studies.<sup>144</sup> In three states the race effects were weaker, but remained statistically significant (Baldus *et al.* 1998, pp. 1661–62).<sup>145</sup>

Not only does evidence suggest that the race of the victim and (to a lesser extent) the race of the defendant directly impacts capital charging-and-sentencing decisions (Baldus and Woodworth 2003; U.S. General Accounting Office 1990), death penalty appeals (Radelet and Vandiver 1983), commutation decisions (Marguart et al. 1994; Wolfgang et al. 1962), and executions (Jacobs et al. 2007; Keil and Vito 1992; Langbein 1999; Radelet 1989; Radelet and Mello 1986), it also suggests that race may have an *indirect* effect on the capital punishment process that increases the likelihood of a disproportionate death sentence. For example, a growing body of research literature suggests that: (1) prosecutors are more likely to proceed with capital cases with weak evidence when the defendant is a member of a minority group (Harmon 2001a, b; Parker, Dewees, and Radelet 2001);<sup>146</sup> (2) whites who hold racist attitudes towards blacks are more likely to support the death penalty (Aguirre and Baker 1993; Barkan and Cohn 1994; Mitchell and Sidanius 1995; Unnever and Cullen 2007), more likely to be allowed to serve on capital juries (Lynch 2006; Russell 1993; Young 2004), more likely to favor the conviction of the innocent over letting guilty defendants go free (Young 2004), and more likely to believe that the death penalty is the only appropriate punishment for convicted killers (Russell 1993); (3) prosecutors are more likely to use peremptory strikes

<sup>&</sup>lt;sup>144</sup> The six states were Colorado, North Carolina, New Jersey, Nebraska, Kentucky, Pennsylvania, and South Carolina.

<sup>&</sup>lt;sup>145</sup> These states were California, Georgia, and Mississippi.

<sup>&</sup>lt;sup>146</sup> Evidence of prosecutors' willingness to proceed with a capital prosecution with weak evidence when the defendant is a member of a minority groups has been discovered at both the state and federal levels. Bruck and colleagues' (2006) recent examination of the federal death penalty since it was reinstated in 1998 revealed that at least 15 defendants were who had their cases authorized for capital prosecution were subsequently acquitted or found innocent of the crime. Of these 15 defendants, 80 percent were racial/ethnic minorities (6 black, 6 Hispanic, and 3 white).

against potential black jurors during jury selection for capital trials—regardless of the race of the offender or victim—because black jurors are less likely to be conviction prone (Baldus et al. 1998, 2001; Bowers and Foglia 2003; Dieter 2005; Elliott-Engel 2008; Sommers and Norton 2007) and more likely to consider mitigating evidence (Bowers, Steiner, and Sandys 2001; Haney 2005); (4) black jurors are more likely to be intimidated by white jurors in jury deliberations-particularly black jurors who are least conviction prone (i.e., elderly black women) (Bowers et al. 2001; Eisenberg, Garvey, and Wells 2001b); (5) capital jurors frequently misunderstand instructions during trial (Bowers and Steiner 1999; Eisenberg and Wells 1993; Luginbuhl and Howe 1995) and racial discrimination is greatest when juror comprehension of courtroom instructions is poor (see Lynch and Haney 2000),<sup>147</sup> and jurors are more likely to vote for a death sentence when they are confused (Bowers and Foglia 2003; Dieter 2005; Frank and Applegate 1998; Garvey, Johnson, and Marcus 2000); (6) minority capital defendants, particularly blacks, tend to hold stigmatized status positions (net of actual individual behavior) (see Fleury-Steiner 2006; Kan and Phillips 2003),<sup>148</sup> and murderers of stigmatized individuals tend to be treated more leniently by prosecutors (Blume et al. 1998) and jurors (Baumer, Messner, and Felson 2000; Kleck 1991; Sundby 2003; but see Eisenberg, Garvey, and Wells 2003); and (7) minority capital defendants—particularly blacks and Hispanics—

<sup>&</sup>lt;sup>147</sup> A growing body of experimental evidence suggests that racial/ethnic stereotypes are especially relevant when individuals' ability to process information is systematically diminished or when judgments become increasingly complex. In particular, individuals use racial/ethnic stereotypes (when available and apparently relevant) as a central theme around which they organize evidence concerning defendants' culpability, emphasizing stereotype-consistent information and negating stereotype-inconsistent information as way of simplifying judgment tasks (Bodenhausen 1990; Bodenhausen and Lichtenstein 1987; Bodenhausen and Wyer 1985). Negative affective states may also significantly impact decision-making, with heightened stereotyping being observed among both angry (Bodenhausen, Sheppard, and Kramer 1994) and anxious individuals (Baron *et al.* 1992). These affective states may be particular common among jurors in death penalty trials (Garvey 2000).

<sup>&</sup>lt;sup>148</sup> For detailed discussions of the creation and diffusion of status value resulting from group membership, see Ridgeway (1991), Ridgeway and colleagues (1998) and Gould (2002).

are more likely to be represented by court appointed counsel, and capital defendants who are represented by court appointed counsel are more likely to receive a death sentence and more likely to have their appeals denied (see, generally, Beck and Shumsky 1997; Bright 1990, 1992, 1994; Dieter 1995; Gelman *et al.* 2004).<sup>149</sup>

In addition to focusing on the role of race and racial discrimination in the administration of capital punishment, scholars have examined the role that region plays in the capital charging-and-sentencing process. Since the moratorium on executions was lifted in 1976 after the *Gregg* decision, 82.2 percent of all executions have occurred in the South (American Civil Liberties Union 2004; Death Penalty Information Center 2008). Texas and Virginia, which currently incarcerate 13.7 percent of the nation's death row inmates, account for over 45 percent of executions since 1976. California and Pennsylvania, on the other hand, which house one-quarter the country's death row population, have only accounted for 1.4 percent of those executed during the same time period (Death Penalty Information Center 2008).

Although some scholars have hypothesized that there is a southern subculture of punitiveness and violence (see Ball-Rokeach 1973; Clarke 1998; Erlanger 1974, 1976; Zimring 2003), recent evidence suggests that there is little overall variation in support for capital punishment between Southerners and non-Southerners (Borg 1997). In fact, there appears to be tremendous variability in support for *and* use of capital punishment both across and within states (American Civil Liberties Union 2004; Baumer *et al.* 2003;

<sup>&</sup>lt;sup>149</sup> Over 90 percent of the individuals on death row could not afford a private attorney (see Tabak and Lane 1989). Iyengar's (2007) recent analysis of the federal indigent defense system discovered that minorities and immigrants reside in jurisdictions that disproportionately contract with private attorneys (known referred to as Criminal Justice Act attorneys) rather than use federal public defenders who are salaried employees of the court. The study also reveals that Criminal Justice Act attorneys systematically underperform federal public defenders with respect to conviction rates and sentence lengths, and their underperformance is primarily attributable to attorney experience, wages, law school quality, and average caseload.

Messner *et al.* 2005). Recent evidence suggests, however, that this variation can be partly accounted for by the structural characteristics of these jurisdictions (American Civil Liberties Union 2004; Baldus *et al.* 2002). Jacobs and Carmichael (2002), examining why states reinstated capital punishment after the *Furman* decision, discovered that a large minority presence, significant economic inequality, and political conservatism increased both the likelihood and rapidity of the return of capital statutes.<sup>150</sup> Similarly, Unah and Steenbergen (2005) discovered that state-level citizen ideology, institutional ideology, and political party competition influence the likelihood and rapidity of executions, net of defendant characteristics and location in the South.<sup>151</sup> Liebman and colleague's (2002) analysis of error rates in capital trials revealed that states and counties with high reversal rates had: (1) a homicide risk rate for whites equal to or greater than that of blacks; (2) higher proportion of blacks; (3) higher proportion of residents receiving public assistance; (4) lower arrest rates for Index offense (i.e., serious felonies); and (5) trial judges subjected to highly partisan elections.<sup>152</sup>

<sup>&</sup>lt;sup>150</sup> Jacobs and Carmichael controlled for the murder rate, violent crime rate, unemployment rate, level of urbanization, and region of the country. For a discussions of the impact of political conservatism on national criminal justice policy over the past several decades, see Abramsky (2002), Beale (1997), Beckett and Sasson (2000), and Simon (2007). Scholars have also discovered that state racial composition is associated with the passage of felon voting restrictions after the passage of the Fifteenth Amendment (Behrens *et al.* 2003, p. 596).
<sup>151</sup> Citizen ideology captures the public's dominant political values. Institutional ideology is distinct from

<sup>&</sup>lt;sup>151</sup> *Citizen ideology* captures the public's dominant political values. *Institutional ideology* is distinct from citizen ideology and measures the liberalism of a state's policies (i.e., restrictiveness of gun law policies, abortion laws, temporary assistance to needy families, tax progressivity, and permissiveness of unionization). *Political party competition* measured the level of inter-party competition for both state Houses. Case-level factors included defendant's race/ethnicity, marital status, education level, criminal history, and age.

<sup>&</sup>lt;sup>152</sup> For detailed discussions of the role of judicial politics in criminal proceedings, see Blume (1999), Bright (1997, 1998), Bright and Keenan (1995), Brooks and Raphael (2002), Helms and Jacobs (2002), Huber and Gordon (2004), and Schanzenbach (2007). See also Unah (2003) and Gordon and Huber (2002) for discussions of the impact of electoral politics on prosecutorial discretion in capital and non-capital cases, respectively. Boylan (2005) and Glaeser *et al.* (2000) provide thorough discussions of the potential impact of future "career aspirations" on prosecutorial discretion at the federal level.

Recent studies of capital punishment systems in California (Pierce and Radelet 2005), Colorado (Hindson et al. 2006), Georgia (Baldus et al. 1990; Kroll 1991), Illinois (Pierce and Radelet 2002), Maryland (Paternoster *et al.* 2004), Nebraska (Baldus *et al.* 2002), New Jersey (Baldus 1991; Weisburd and Naus 2001), Ohio (Williams and Holcomb 2001), and Texas (Brock, Cohen, and Sorensen 2000) all revealed that a small number of counties are responsible for the vast majority of death sentences in each state, net of legitimate case characteristics. In Maryland, for example, prosecutors in Baltimore County were 13 times more likely to seek the death penalty in death-eligible cases than prosecutors in Baltimore City (the state's largest city), five times more likely to seek the death penalty than prosecutors in Montgomery County, and three times more likely to seek the death penalty than prosecutors in Anne Arudel County, although Baltimore City, Montgomery County, and Anne Arudel County all had significantly higher homicide rates (Paternoster et al. 2004). Similarly, Baldus and colleagues' (2002) study of Nebraska revealed that prosecutors in urban counties in Nebraska were nearly 2.5 times more likely to seek the death penalty than prosecutors in rural counties. These findings are consistent with research on non-capital crimes which suggests that the larger social context is very important in understanding charging and sentencing decisions (see, e.g., Britt 2000; Helms and Jacobs 2002; Kautt 2002; Ulmer and Johnson 2004; Weidner, Frase, and Pardoe 2004).

### 4.2.3 Race and the Federal and Military Death Penalty

The first execution under a federal death sentence was that of Thomas Bird in 1790 (Little 1999). Since that time, at least 343 individuals have been executed under federal jurisdiction: 39 percent white, 35 percent black, 19 percent Native American, and

7 percent either Hispanic or of unknown race/ethnicity. From 1900 through 1999, 61 percent of individuals executed under federal law were members of minority groups (see Little 1999). As of July 2004, 68.6 percent of defendants on federal death row (22 of 32) were members of minority groups (21 black, 1 Native American) (Death Penalty Information Center 2008).

As with all other state capital statutes, the federal death penalty was also invalidated after the *Furman* decision in 1972. Unlike most states, however, the federal death penalty was not reinstated until 16 years after Furman when the Drug Kingpin Act was signed into law in 1988 (U.S. Congress 1994).<sup>153</sup> The federal death penalty primarily differs from capital statutes at the state level in that federal prosecutors must seek approval from the Attorney General to seek the death penalty. The period from 1998 to 1994 is referred to as the "pre-protocol era" because federal prosecutors were only required to notify the Attorney General when they affirmatively wished to seek the death penalty against a defendant (U.S. Congress 1994). From November of 1988 to March of 1994, federal prosecutors sought approval for 52 death penalty cases and received approval in 47 of these cases (U.S. Department of Justice 2000). The raw numbers suggest strong racial disparities in the federal capital punishment system. Of the initial 52 cases, 75 percent of the defendants were black, 13 percent where white, 10 percent were Hispanic, and 2 percent were members of another minority group (U.S. Congress 1994).

<sup>&</sup>lt;sup>153</sup> See *Anti-Drug Abuse Act of 1988*, 21 U.S.C. § 848(e) (1988). Shortly after the *Furman* ruling, the federal government reinstated the death penalty for individuals convicted of aircraft hijacking that resulted in death. Due to the infrequency of this type of offense, the federal government would not have an active death penalty statute until 1988.

Beginning in 1995, when the Federal Death Penalty Act was signed into law, the U.S. Department of Justice adopted a protocol that required federal prosecutors to submit information on all federal cases that were death eligible, regardless of whether the prosecutor intended to seek the death penalty (U.S. Department of Justice 2000).<sup>154</sup> From 1995 through 2000, federal prosecutors submitted information on 685 death eligible federal cases and sought the death sentence in 183 of these cases (U.S. Department of Justice 2000). Then-Attorney General Janet Reno authorized 86.8 percent of the cases submitted by prosecutors (159 of 183 cases) (U.S. Department of Justice 2000). In 75 percent of these cases the defendant was a member of a minority group and over 50 percent were black, although the death penalty authorization rate was higher for white offender cases (38 percent) than black-offender cases (25 percent) and Hispanic-offender cases (20 percent).<sup>155</sup> White-offender cases were much more likely to result in a plea agreement (48 percent) than black-offender (25 percent) and Hispanic-offender (28 percent) cases. It is impossible to determine, however, whether federal prosecutors differentially offered plea bargains based on the race of the offender. Plea agreements require the cooperation of the defendant and it is possible that black and Hispanic offenders are less willing to accept plea agreements than white offenders.

Racial differences appear to be more pronounced when examining the race of the victim. The death penalty authorization rate for white-victim was cases 15 percentage points higher than the authorization rate minority-victim cases (37 percent vs. 21 percent). Race-of-victim bias also appears evident in the actual imposition of the federal death penalty (Baldus 2001; but see Klein, Berk, and Hickman 2006). Among all death

<sup>&</sup>lt;sup>154</sup> See Federal Death Penalty Act of 1994, 18 U.S.C. § 3591 (1994).

<sup>&</sup>lt;sup>155</sup> Of the 312 defendants approved for capital prosecution since the federal death penalty was reenacted, 233 (75 percent) were members of minority groups (McNally 2004).

eligible cases, white-victim cases were over twice as likely to result in the death sentence compared to minority victim cases (5 percent vs. 2 percent). In the eleven states that actually imposed a federal death sentence, white-victim cases were also over twice as likely to result in the death sentence as minority-victim cases (17 percent vs. 8 percent). With respect to offender/victim racial combinations, black-offender/white-offender cases were almost four times more likely to result in a death sentence than cases with a black offender and a non-white victim (11 percent vs. 3 percent). In the eleven states that imposed a death sentence, black-offender/white-offender cases were over three times more likely to result in a death sentence than cases with a black-offender and non-white victim (24 percent vs. 7 percent) (U.S. Department of Justice 2000, 2001).

Similar to the administration of capital punishment at the state level, there is significant regional variation in the use of the federal death penalty. From 1995 through 2000, 42 percent (287 of 685) of submissions of death eligible cases were from five jurisdictions: Puerto Rico, Eastern District of Virginia, Maryland, Eastern District of New York, and the Southern District of New York. In fact, attorneys from 21 districts did not submit a single death-eligible case between 1995 and 2000 and attorneys from 40 districts never recommended seeking the death penalty against a defendant (U.S. Department of Justice 2000; see also Klein *et al.* 2006).<sup>156</sup>

There are also significant racial disparities in the administration of capital punishment in the U.S. Armed Forces. Currently, six of the seven individuals (85.7

<sup>&</sup>lt;sup>156</sup> Mannheimer (2006) notes that, since 2002, at least five individuals have been sentenced to death in federal court for conduct that occurred in states that do not authorize the death penalty, and argues that the federal death penalty may constitute "cruel and unusual" punishment when it is applied in states that have abolished capital punishment. In particular, he suggests that the Eighth Amendment ought to be read as a restraint on upon the federal government to punish in a way that conflicts with the norms of an individual state.

percent) on military death row are members of minority groups (5 blacks, 1 white, and 1 Asian/Pacific Islander) (Death Penalty Information Center 2008). Although no one has been executed by the U.S. Military since 1961, racial disparities in the administration of the death penalty have been present both before and after desegregation (Sullivan 1994). For example, during World War II, blacks accounted for less than 10 percent of the military population (Adams 1994), but constituted 78.5 percent of individuals executed during that period (55 of 70 executions) (Sullivan 1994).<sup>157</sup> After racial segregation in the military was outlawed in 1948, racial disparities in the administration of capital punishment actually increased. From 1954 to 1961 (the date of the last military execution), 11 of the 12 individuals executed by the military were black (91.6 percent) (Serrano 1994; Sullivan 1994). The offender/victim racial combination also appears to influence the military death penalty: every black inmate on military death row was convicted of killing a white victim (Serrano 1994).

In summary, although the codified racially discriminatory practices in the administration of criminal justice in the Antebellum South (1619 - 1865), the Reconstruction Era (1865 - 1876), and the Jim Crow Era (1877 - 1954) were struck down as unconstitutional during the middle of the twentieth-century, a voluminous research literature suggests that race continues to influence capital charging and sentencing decisions. The vast majority (93 percent) of studies examining the capital charging-and-sentencing process in states across the nation since *Furman* discovered that the race of the victim significantly influences the likelihood that a prosecutor seeks the death penalty against a defendant or the likelihood that a judge or jury imposes a death

<sup>&</sup>lt;sup>157</sup> Fourteen of the 18 soldiers executed (77.7 percent) in England during World War II were minorities (11 blacks, 3 Hispanics) (Lilly and Thomson 1997).

sentence, net of legitimate case characteristics (Baldus and Woodworth 1997). Moreover, 82 percent of methodologically rigorous studies conducted in eleven different states since Furman report statistically significant race-of-victim effects in capital charging or sentencing decisions (but cf. Morton and Rolph 2000). Numerous human rights, civil rights, and legal organizations—such as the American Bar Association (ABA), American Civil Liberties Union (ACLU), National Association for the Advancement of Colored People (NAACP), National Association of Criminal Defense Lawyers (NACDL), and the National Bar Association—have also publicly condemned the racially disparate treatment of minorities in the capital punishment system at both the state and federal levels. Ironically, to date, no racial discrimination claim in a capital case has prevailed before the U.S. Supreme Court (see Blume et al. 1998); though it seems unlikely the court will be able to continue to dismiss such claims in light of the mounting evidence documenting widespread and persistent racial bias in the administration of capital punishment since Furman. Perhaps as an indication of the Court's changing of direction, it recently ruled, eight-to-one, that lower courts are required to give capital defendants the opportunity to prove whether prosecutors violated their constitutional rights by engaging in racial discriminatory jury selection practices in their particular cases (*Miller-El v. Cockrell*, 537 U.S. 322 [2003]).<sup>158</sup> Authoring the majority opinion in *Miller-El*, Associate Justice Kennedy wrote:

<sup>&</sup>lt;sup>158</sup> In *Batson v. Kentucky* (476 U.S. 79 [1986]), the U.S. Supreme ruled that it was unconstitutional for prosecutors to use their peremptory challenges to strike potential jurors solely on the basis of their race (see also Bourke, Hingston, and Devine 2003). It was more than century earlier, in *Strauder v. West Virginia* (100 U.S. 303 [1880]), that the Supreme Court first held that the purposeful exclusion of blacks from jury service violated the Equal Protection Clause of the Fourteenth Amendment, but defendants challenging racial discrimination in *voir dire* were required to meet a prohibitively high burden of proof: a showing of *systematic* bias by the prosecutor (see *Swain v. Alabama*, 380 U.S. 202 [1965]). The *Batson* court overruled *Swain*, holding that a defendant was allowed to make a *prima facie* case of racial discrimination based on the prosecutor's conduct in the defendant's particular case, after which the prosecutor would be

Irrespective of the whether the evidence could prove sufficient to support a charge of systematic exclusion of African Americans, it reveals that the culture of the district attorney's office in the past was suffused with bias against African Americans in jury selections. This evidence, of course, is relevant to the extent it casts doubt on the legitimacy of the motives underlying the State's actions in the petitioner's case. Even if we presume at this stage that prosecutors in Miller-El's case were not part of this culture of discrimination, the evidence suggests they were likely not ignorant of it. Both prosecutors joined the District Attorney's Office when assistant district attorneys received formal training in excluding minorities from juries. The supposition that race was a factor could be reinforced by the fact that the prosecutors marked the race of each prospective juror on their juror cards...Our concerns here are heightened by the fact that, when presented with this evidence, the state trial court somehow reasoned that there was not even the inference of discrimination to support a prima facie case (537 U.S. at 354).<sup>159</sup>

Concern over racial discrimination in the capital punishment led several members

of the U.S. Congress to support the passage of racial justice legislation. Under such

allowed to rebut the challenge by offering a race-neutral reason for excluding the veniremember and the trial judge would decide whether the prosecutor's reason was legitimate or merely pretext. The Supreme Court would later rule that the exclusion of potential jurors based solely on ethnicity (*Hernandez v. New York*, 500 U.S. 352 [1991]) and gender (*J.E.B. v. Alabama ex rel. T.B.*, 511 U.S. 127 [1994]) was unconstitutional as well.

Several years later after the *Miller-El* ruling, the Supreme Court decided, eight-to-one (Justice Thomas dissented), that California's "more likely than not" burden of proof for defendants raising *Batson* challenges was unconstitutional and defendants were only required to produce evidence sufficient to "draw an inference that discrimination [had] occurred" to satisfy their prima facie case of discriminatory jury selection (see *Johnson v. California*, 545 U.S. 162 [2005]).

<sup>159</sup> In 2008, the Pennsylvania Supreme Court ruled that death row inmate, Robin Cook, failed to establish a *prima facie* case of purposeful racial discrimination with respect to the prosecutor's use of peremptory challenges to exclude black jurors during *voir dire* (Elliott-Engel 2008). Cook presented a training videotape created by then-Assistant District Attorney Jack McMahon during the year prior to Cook's trial that depicted McMahon instructing other prosecutors on how to successfully use racially-motivated peremptory challenges to strike black jurors. In the videotape, McMahon told prosecutors that blacks from lower income areas were not "good" jurors because were less likely to convict resulting from their "resentment for law enforcement" and "resentment for authority." He also advised prosecutors to develop pretextual race-neutral reasons for striking black jurors in the event defense counsel attempted to raise a *Batson* claim. The training videotape, recorded in 1987, was released in 1997 by the incumbent district attorney position.

Cook was convicted in 1988 of first-degree murder, criminal conspiracy to commit murder, and possessing an instrument of crime and robbery. During *voir dire* for Cook's trial, McMahon used 14 of his 19 peremptory strikes against potential black jurors. In a four-to-two ruling (one Justice did not participate in the case), the court concluded that McMahon was able to offer race-neutral reasons for 11 of the 14 potential black jurors he challenged, some thirteen years after Cook's trial. The court also noted that there were eight black jurors on Cook's jury and Cook was unable to prove that McMahon's race-neutral explanations were pretextual. The two dissenting Justices concluded that the videotape, showing McMahon explicitly instructing jurors to develop pretexual justifications for strikes of black jurors, seriously undermined McMahon's race-neutral reasons for striking black jurors.
legislation, individuals convicted of capital crimes would have been able to challenge their death sentences using conventional evidentiary standards established in civil racial discrimination cases (e.g., employment and housing discrimination) (Cole 1994). The proposed legislation also required states to maintain sufficient data on all potential capital cases in order to allow capital defendants and prosecuting attorneys to present and defend claims under such legislation (Baldus *et al.* 1994b, p. 378). The initial version of the legislation was introduced in 100th U.S. Congress by Representative John Convers, chairman of the Subcommittee on Criminal Justice of the House Judiciary Committee, but the bill died in committee.<sup>160</sup> During the following congressional term, a modified version of the legislation, known as the "Racial Justice Act" (RJA), was proposed in both houses of the 101st U.S. Congress (1989 – 1990), but also failed to make it out of committee.<sup>161</sup> Three subsequent versions of the RJA that were introduced in 1991 and 1993 also died in committee (Baldus, Pulaski, and Woodworth 1994a).<sup>162</sup> In 1994, due largely to the efforts of the Congressional Black Caucus, a weaker version of the RJA was passed in the House of Representatives, but failed to reach the Senate when Republicans in the Senate threatened to filibuster if the RJA had remained on the crime bill.<sup>163</sup> As Cole (1994) notes, passage of the RJA would have marked the first time since Reconstruction that the U.S. Congress has done anything to respond to racial discrimination in the criminal justice system. In 1998, Kentucky became the first state to

<sup>&</sup>lt;sup>160</sup> See H.R. 4442, 100th Congress, 1st Session (1988).

<sup>&</sup>lt;sup>161</sup> See S. 1696, 101st Congress, 1st Session (1989); H.R. 4618, 101st Congress, 2d Session (1990).

<sup>&</sup>lt;sup>162</sup> See S. 1249, 102d Congress, 1st Session (1991); H.R. 2851, 102d Congress, 1st Session (1991); H.R. 3329, 103<sup>rd</sup>, 1st Session (1993).

<sup>&</sup>lt;sup>163</sup> See H.R. 4017, 103<sup>rd</sup> Congress, 2d Session (1994). This version of the RJA was considered "watereddown" because although it provided condemned prisoners the opportunity to challenge their death sentence with statistical proof of racial discrimination, it established stricter requirements for proving a *prima facie* case (e.g., very strong statistically significant evidence of racial disparity) and did not require State officials to collect additional data on the administration of the death penalty (Baldus *et al.* 1994b, p. 379).

pass and enact such legislation.<sup>164</sup> Under Kentucky's RJA, defendants are allowed to present statistical or other evidence suggesting that their race, the race of their victim(s), or both, played a significant part in prosecutor's decision to seek the death sentence in their particular case. Such a claim must be made at a pre-trial conference, after which the court prescribes a time for the submission of such evidence by both the prosecution and defense. Under Kentucky's law, the defendant must provide clear and convincing evidence that race was the basis for the decision in her or his case. If the court finds in favor of the defendant, it is required to order that the death sentence cannot be sought in that particular case.

Following Kentucky's adoption of the RJA, similar legislation was introduced in Florida, Georgia, Illinois, Mississippi, Nebraska, North Carolina, Ohio, and Texas, but failed to pass in any of these states (American Bar Association 2003a, Appendix J; American Civil Liberties Union 2003; Mears 1998b).<sup>165</sup> The failure of the Georgia Legislature to pass racial justice legislation is particularly noteworthy considering no other state has been subject to more scrutiny than Georgia with respect to racial bias in the administration of the capital punishment in the modern era of the death penalty. As noted in Chapter Two, several Georgia death penalty cases challenging the constitutionally of capital punishment have set legal precedent. The history of Georgia's death penalty and its influence on the administration of the death penalty nationwide is presented in the next chapter.

<sup>&</sup>lt;sup>164</sup> See Kentucky Racial Justice Act, Ky. Rev. Stat. Ann. § 532.300 (1998).

<sup>&</sup>lt;sup>165</sup> See S. 1662, 102d Leg., Reg. Sess. (Fla. 2000); H.R. 324, 146th Gen. Assem., Reg. Sess. (Ga. 2001);
H.R. 1211, 146th Gen. Assem., Reg. Sess. (Ga. 2002); H.R. 129, 147th Gen. Assem., Reg. Sess. (Ga. 2003);
S. 1771, 92d Gen. Assem., Reg. Sess. (Ill. 2001); H.R. 4139, 92d Gen. Assem., Reg. Sess. (Ill. 2001); H.R. 2982, 93<sup>rd</sup> Gen. Assem., Reg. Sess. (Ill. 2003); H.R. 168, 2002 Reg. Sess. (Miss. 2002); H.R. 217, 2003 Reg. Sess. (Miss. 2003); LB 781, 98th Leg., 1st Reg. Sess. (Nebr. 2003); S. 171, 2001 Reg. Sess. (N.C. 2001); H.R. 140, 2001 Reg. Sess. (N.C. 2001); H.R. 102, 124th Gen. Assem., Reg. Sess. (Ohio 2001); H.R. 370, 78th Leg., Reg. Sess. (Tex. 2003); H.R. 866, 78th Leg., Reg. Sess. (Tex. 2003).

# **Chapter 5: Georgia and the Death Penalty**

#### 5.1 EXECUTIONS IN THE SOUTH, 1608 - 2008

The South has been a major focus of debate over the administration of the death penalty. While some scholars have suggested that Southerners are neither more likely to support the death penalty (Borg 1997) nor more likely to hold favorable attitudes towards violence (Erlanger 1974) than non-Southerners, empirical studies of the death penalty process consistently reveal that prosecutors are more likely to seek the death penalty, judges and juries are more likely to impose a death sentence, and executions are more likely to be carried out in the South (Blume, Eisenberg, and Wells 2004; Clarke 1998; Jacobs et al. 2007; Kroll 1991; Liebman et al. 2002; Ogletree 2002). The fifteen former slave states account for slightly over 80 percent of the 1,136 executions from 1976 through 2008: Alabama (38), Arkansas (27), Delaware (14), Florida (66), Georgia (43), Kentucky (3), Louisiana (27), Maryland (5), Mississippi (10), Missouri (66), North Carolina (43), South Carolina (40), Tennessee (4), Texas (423), and Virginia (102) (Death Penalty Information Center 2008). The territory of Oklahoma (88 executions since 1976) also allowed slavery but it did not receive its statehood until 1907, over forty vears after the abolition of slavery. The eleven states that secended the from the Union— Alabama, Arkansas, Florida, Georgia, Louisiana, Mississippi, North Carolina, South Carolina, Tennessee, Texas, and Virginia—account for 72 percent of the total number of executions since 1976 (Death Penalty Information Center 2008).

### 5.1.1 Executions in Georgia, 1735 – 2008

King George, II, of England granted Colonel James Edward Oglethorpe a charter for the founding of the colony of Georgia on June 9, 1732. Oglethorpe, the colony's first governor, was concerned about the conditions of debtors' prisons in London and established Georgia as a penal colony for such inmates. Slavery was originally prohibited in the colony, but the English Parliament repealed the prohibition on October 26, 1749, after repeated petitions from the colonists and slavery officially became legal in the colony on January 1, 1751. Four years later, on January 7, 1755, the first Royal Legislature of Georgia met in Savannah and adopted the colony's first slave code modeled after South Carolina's "Negro Act" of 1740-and formally declared that all Negroes, mulattoes, Mestizos (Indians), and their offspring who were slaves at the time of the act were assigned to slavery forever (Royal Legislature of Georgia [1755] 1910). The first recorded execution in the colony was that of female Irish indentured servant. Alice Riley, on January 19, 1735, for the murder of her master, William Wise.<sup>166</sup> Riley's boyfriend, Richard White, who assisted in the murder, was executed the following day (Espy and Smykla 2004). Nearly forty years later, on September 20, 1774, Jack (Lyford)<sup>167</sup> became the first slave (officially) executed in the colony of Georgia.<sup>168</sup> Mr. (Lyford) was convicted of arson and was burned to death (Espy and Smykla 2004).

From 1735 to 1924, the legal method of execution in Georgia was hanging. Executions were typically carried out by the sheriff in the county where the crime was

<sup>&</sup>lt;sup>166</sup> Riley was pregnant at the time of her trial and was allowed to deliver the baby before her execution. <sup>167</sup> The surname of "Lyford" is listed parenthetically because enslaved Africans were not officially allowed to have surnames; rather they were commonly identified by the surname of their slave masters. Typically name transmission among slaves was matronymic because slave masters obtained rights to the offspring of their females, but not their males (Gutman 1976).

<sup>&</sup>lt;sup>168</sup> Georgia was granted statehood in 1776, becoming the thirteenth state to join the Union, and its first constitution was signed the following year.

committed. On May 20, 1925, Gervis Bloodworth and Willie Jones (both white males) were the last two individuals officially executed by hanging in Georgia. In August 1924, death by hanging was abolished by the Georgia General Assembly and was replaced by the electric chair. The first electric chair was installed in the Georgia State Prison in Milledgeville, Georgia in 1924. Howard Hinton, a black male convicted of murder in Dekalb County, was the first person executed by electrocution in Georgia on September 13, 1924. Nearly fourteen years later, in 1938, the electric chair was moved to the new Georgia State Prison in Reidsville, Georgia and Archie Goodwin, a black male, was the first person to be executed at the new facility (Mears 1999). The electric chair was moved, once again, to the Georgia Diagnostic and Classification Center in Jackson, Georgia in 1980 and John Eldon Smith (a white male) became the first individual executed at the Jackson facility in 1983 (Georgia Department of Corrections 2002).<sup>169</sup>

As noted in Chapter Two, several states in the West and Southwest switched from electrocution to lethal gas (or lethal injection) in the 1930s, 1940s, and 1950s (Banner 2002). Most Southern states, however, resisted such reform and vigorously defended their use of the electric chair (see, e.g., Driggs 1993). Over the next half-century, opponents of the electric chair in Georgia continuously pressured the state legislature to change the state's method of execution to lethal injection. Death penalty opponents cited the mounting evidence showing that death by electrocution was, in fact, extremely painful and often resulted in significant disfigurement to the body of the convicted (see, e.g., Death Penalty Information Center 2008; Denno 1994).<sup>170</sup> On January 27, 2000, the

<sup>&</sup>lt;sup>169</sup> Smith was also the first person executed in Georgia under the state's post-*Furman* death penalty statute. <sup>170</sup> On December 12, 1984, Alpha Otis Stephens was executed at the Georgia Diagnostic and Classification Center in Jackson, Georgia. Stephens had to be given a second charge of electricity after the first twominute electric charge failed to kill him. Physicians had to wait six minutes after the first charge so

Georgia General Assembly officially changed the method of execution in the state from electrocution to lethal injection, but allowed individuals who committed a capital crime before the enactment of the new statute to be executed by electrocution.<sup>171</sup> However, on October 5, 2001, in *Dawson v. State* (554 S.E.2d 137 [Ga. 2001]), the Georgia Supreme Court held that death by electrocution constituted cruel and unusual punishment and all executions in the state had to be carried out by lethal injection (see Mears 2002, pp. 285–86). Three weeks later, on October 25, 2001, Terry Mincey (a white male) became the first person in Georgia put to death by lethal injection.<sup>172</sup>

5.1.2 Race and Executions in Georgia, 1774 – 2008

From 1774 (the year of the first recorded execution of a black person in Georgia)

through 2008, 992 official executions were carried out in Georgia. Of these nearly one

thousand executions, 737 of those executed were black (74.3 percent), 237 were white

Stephens's body could cool off enough to allow them to examine him and declare another jolt was needed. During that time, it is reported that Stephens took 23 breaths (Death Penalty Information Center 2008). <sup>171</sup> Ga. Code Ann. § 17-10-38.

<sup>&</sup>lt;sup>172</sup> The constitutionality of lethal injection was challenged soon after the Georgia Supreme Court outlawed executions by electrocution (Mears 2002, pp. 282–83). Of central concern was the absence of a licensed physician in the lethal injection process. Indeed, Georgia's new death penalty statute did not require the presence of a physician because the American Medical Association opposes physician involvement in executions. This is particularly problematic considering the difficulty in determining whether an individual who was administered the lethal drugs remains conscious after the injection. The drug used in Georgia, *Pavulon*, serves the primary function of paralyzing the body so that no twitching or seizures are observed by individuals viewing the execution (Mears 2004).

On April 16, 2008, the U.S. Supreme Court ruled, seven-to-two, that Kentucky's four-drug lethal injection protocol (*Valium*, sodium pentathol, *Pavulon*, and potassium chloride) did not violate the Eighth Amendment's prohibition against cruel and unusual punishment (*Baze v. Rees*, 128 S. Ct. 1520 [2008]). Justice John Paul Stevens, concurring in judgment, enumerated many problems that he believed continued to plague the administration of the death penalty and stated that the death penalty was inherently unconstitutional. Justice Stevens stated that he felt bound by court precedent to concur with the majority, but urged that it was time to once again reconsider the justifications for capital punishment.

The dissent, written by Justice Ruth Bader Ginsburg and joined by Justice David Souter, stressed that the consequences of mistake when administering the four-drug scheme are "horrendous and effectively undetectable," and therefore measures that can materially increase the likelihood that lethal injection will occur painlessly must be adopted by the state if readily available.

(23.9 percent), and 6 were Native American (0.6 percent).<sup>173</sup> Disaggregating these data into pre- and post-slavery time periods, however, reveals some stark differences in the administration of the death penalty in Georgia. Prior to the Emancipation Proclamation (circa 1863), 102 executions were carried out in Georgia. Of those executed during that time, 59 were white (57.8 percent), 37 were black (36.3 percent), and six were Native American (5.9 percent)—the proportion of executions that were of white offenders was approximately 11.5 percentage points greater than blacks (Espy and Smykla 2004). From 1863 through 2008, 890 individuals were executed in Georgia: 700 blacks (79.6 percent) and 177 whites (20.2 percent)—the proportion of executions carried out against blacks was nearly 60 percentage points greater than that of whites (Death Penalty Information Center 2008; Espy and Smykla 2004).<sup>174</sup> This dramatic swing in the proportion of executions carried out against blacks since the abolition of slavery has led some historians to suggest that, prior to emancipation, slave owners placed greater value on preserving black life because of the slaves' economic worth (Fogel and Engerman 1974). Other scholars argue that following emancipation, the discriminatory use of capital punishment in the former Confederate states, along with the proliferation of other racially motivated (and state endorsed) practices such as segregation laws, poll taxes, literacy tests, and felony disenfranchisement laws, served to forcibly and violently reestablish (and re-legitimize) the racial hierarchy that existed under slavery (Behrens *et al.* 2003; Clarke 1998; Foner and Mahoney 1995; Williamson 1984).

Recall that the U.S. Supreme Court placed a moratorium on executions in 1972 with its decision in *Furman* and reinstated the death penalty four years later in 1976

<sup>&</sup>lt;sup>173</sup> According to the Espy file, thirteen (1.3 percent) of these executions did not list the race of the person executed (Espy and Smykla 2004).

<sup>&</sup>lt;sup>174</sup> Thirteen cases (1.3 percent) are missing data on the race of the executed (Espy and Smykla 2004).

following its decisions in *Gregg*, *Proffit*, and *Jurek*. (Also recall that many states quickly began sentencing defendants to death only a few months after the Furman decision in 1972.) The Court believed that the new guided-discretion death penalty statutes were sufficient to eliminate the rampant arbitrariness and capriciousness that characterized the pre-Furman era.<sup>175</sup> However, in the aftermath of Gregg, executions in Georgia continue to be almost exclusively reserved for individuals convicted of murdering white victims. Georgia ranks sixth in the nation with respect to total executions since the death penalty was reinstated (Death Penalty Information Center 2008) and from 1977 through 2008, 43 people were executed in Georgia: 28 were white offenders convicted of murdering white victims (65.1 percent), 11 were black offenders convicted of murdering white victims (25.6 percent), and four were black offenders convicted of murdering black victims (9.3 percent). With respect to the race of the victim, nearly 91 percent of the executions during this period were of offenders convicted of murdering white victims (Death Penalty Information Center 2008; Espy and Smykla 2004). The proportion of executions for white victim cases is particularly startling considering that more than two-thirds (66.7 percent) of homicide victims in Georgia from 1976 to 2000 were black and over half of known homicide offenders were black (56.6 percent) (Fox 2005).

As noted in Chapters Two and Four, death penalty opponents argued that racial disparities in the administration of the death penalty were most pronounced for the crime of rape. The first recorded execution for the crime of rape in Georgia was that of an unidentified slave in 1847 (Espy and Smykla 2004). From 1847 until the last official

<sup>&</sup>lt;sup>175</sup> Although the majority in *Furman* believed that the existing death penalty statutes unnecessarily permitted arbitrariness and capriciousness, it did not hold that racial discrimination was demonstrated in the pre-*Furman* era. In fact, only Justice Marshall emphasized the legacy of the racially discriminatory administration of the death penalty in the United States (Mears 1999).

execution for rape was carried out in Georgia in 1961, the state executed 103 individuals for the crime of rape (Espy and Smykla 2004). Of these 103 executions, 101 (98.1 percent) have been of black men—the vast majority of which convicted of raping white women (Espy and Smykla 2004; Georgia Department of Corrections 2002).<sup>176</sup>

# 5.2 GEORGIA AND THE MODERN DEATH PENALTY

Georgia has been the most influential state in shaping national death penalty policy in the modern era of capital punishment (see Baldus *et al.* 1990). No less than seventeen cases originating in Georgia have set legal precedent with respect to the administration of capital punishment, including the *Furman* and *Gregg* cases which, respectively, were responsible for placing and lifting the moratorium on executions in the United States in the 1970s (Cook 1996).<sup>177</sup> A brief discussion of the most significant cases originating in Georgia and their impact on the capital punishment system in America is provided below.

# 5.2.1 Landmark Georgia Death Penalty Cases

*Furman v. Georgia (1972).* In August 1967, William Henry Furman, a black male, was arrested for the shooting death of William Micke, a white male Chatham County resident, during a botched burglary. Furman broke into the victim's house and

<sup>&</sup>lt;sup>176</sup> Recall from Chapter Four that from 1608 to 1977, 843 of the 947 official executions for individuals convicted of rape in America have been of black men (89.2 percent).

<sup>&</sup>lt;sup>177</sup> According to Cook (1996), the seventeen Georgia death penalty cases that have set legal precedent are (in chronological order): *Whitus v. Georgia* (385 U.S. 545 [1967]); *Furman v. Georgia* (408 U.S. 238 [1972]); *Gregg v. Georgia* (428 U.S. 153 [1976]); *Davis v. Georgia* (429 U.S. 122 [1976]); *Coker v. Georgia* (433 U.S. 584 [1977]); *Eberheart v. Georgia* (433 U.S. 917 [1977]); *Presnell v. Georgia* (439 U.S. 14 [1978]); *Green v. Georgia* (442 U.S. 95 [1979]); *Godfrey v. Georgia* (446 U.S. 420 [1980]); *Zant v. Stephens* (462 U.S. 862 [1983]); *Francis v. Franklin* (471 U.S. 307 [1985]); *McCleskey v. Kemp* (481 U.S. 279 [1987]); *Amadeo v. Zant* (486 U.S. 214 [1988]); *Ford v. Georgia* (498 U.S. 411 [1991]); *Dobbs v. Zant* (506 U.S. 357 [1993]); *Burden v. Zant* (510 U.S. 132 [1994]); and *Lonchar v. Thomas* (517 U.S. 314 [1996]).

fled when the victim came to investigate, tripping over an electrical cord and accidentally firing the gun he was carrying through a closed door in the house. The bullet passed through the door and killed Mr. Micke. Furman was found guilty by a jury and sentenced to death on September 20, 1968.<sup>178</sup> Furman subsequently appealed his sentenced to the Georgia Supreme Court, arguing that Georgia's death penalty statute violated his constitutional rights because it lacked sentencing guidelines and it was administered in a racially discriminatory manner. The Georgia Supreme Court unanimously disposed of Furman's constitutional challenge, holding that the state's death penalty statute had repeatedly been found *not* to constitute cruel and unusual punishment (see *Furman v*. State, 167 S.E.2d 628 [1969]). Furman appealed to the U.S. Supreme court and his case was consolidated with two additional cases, one from Georgia and the other from Texas, involving black men sentenced to death for raping white women.<sup>179</sup> Although the Court believed that Furman failed to prove racial bias in his case, it ruled that the lack of sentencing guidelines for juries in Georgia's death penalty statute was a violation of the Eighth Amendment and subsequently invalidated all capital punishment statutes that lacked such standards (see Chapter Two). At the time of the *Furman* decision, there were 43 individuals on death row in Georgia: 29 convicted of murder, 12 convicted of rape, and two convicted of armed robbery (Mears 1999, p. 16).

When the 1973 Georgia General Assembly convened the following January, drafting new death penalty legislation and reenacting the death penalty was its top

<sup>&</sup>lt;sup>178</sup> Furman, 26 years old at the time, was diagnosed as being "mentally deficient" and subject to psychotic episodes. The court, however, rejected Furman's insanity plea.

<sup>&</sup>lt;sup>179</sup> The rape cases involved twenty-one year-old Lucious Jackson (Georgia) and twenty-one year-old Elmer Branch (Texas). Jackson was sentenced to death on December 10, 1968 and Branch was sentenced to death on July 26, 1967, and their sentences were affirmed on appeal in state court (*Jackson v. State*, 171 S.E.2d 501 [Ga. 1969]; *Branch v. State*, 447 S.W.2d 932 [Tex. Crim. App. 1969]).

priority. In fact, several members of the General Assembly were prepared to defy the U.S. Supreme Court's Furman mandate and simply reenact the old death penalty statute (Mears 1999, p. 18). New death penalty bills were quickly filed in both the House and Senate. One of the most important provisions of the bill was the inclusion of a presentencing hearing in which prosecutors were required to prove certain aggravating circumstances relating to the crime or the defendant. Also, during this pre-sentencing hearing, the defendant would be allowed to present mitigation evidence suggesting why the death penalty should not be imposed.<sup>180</sup> The House and Senate versions of the new death penalty legislation were immediately challenged by members in both houses. Opponents of the proposed legislation argued that the changes to the statute were merely cosmetic and that the new legislation did very little to prevent the unconstitutional application of the death penalty, particularly with respect to poor and black defendants. Proponents of the new legislation believed that the three procedural reforms that were amended to the old statute made it constitutional, namely: (1) a bifurcated hearing for guilt/innocence and sentencing; (2) a list of statutory aggravating that juries were required to consider before imposing a sentence; and (3) automatic appellate review by the Georgia Supreme Court (see Chapter Two).<sup>181</sup> Despite the opposition to the bills, the legislation passed by a vote of 154 to 16 in the Georgia House of Representatives on February 13, 1973 and by a vote of 47 to 7 on February 22, 1973 in the Georgia Senate. Prior to the final vote on the new bill in the Senate, several amendments to make the

<sup>&</sup>lt;sup>180</sup> 1973 Ga. Laws 74, § 1; Ga. Code Ann. § 27-2534 (1973).

<sup>&</sup>lt;sup>181</sup> With respect to automatic appellate review, the Court was required to determine: (1) whether the sentence of death was imposed under the influence of passion, prejudice, or any other arbitrary factor; (2) whether, in cases other than treason or aircraft hijacking, the evidence supports the jury's or judge's finding of a statutory aggravating circumstance; and (3) whether the sentence of death was excessive or disproportionate to the penalty imposed in similar cases, considering both the crime and the defendant (1973 Ga. Laws 74, § 4; Ga. Code Ann. § 27-2537 [1973]).

death penalty mandatory were struck down (Mears 1999, p. 35). Slightly over a month

later, on March 28, 1973, Governor Jimmy Carter signed the bill into law and it

immediately became effective.<sup>182</sup>

Georgia's new death penalty was originally published in the Georgia Laws 1973

Session.<sup>183</sup> The legislation was codified and provided for eleven separate instances where

the death penalty could be imposed on someone convicted of a capital offense:

(a) The death penalty may be imposed for the offenses of aircraft hijacking or treason in any case.

(b) In all cases of other offenses for which the death penalty may be authorized, the judge shall consider, or he shall include in his instructions to the jury for it to consider, any mitigating circumstances or aggravating circumstances otherwise authorized by law and any of the following statutory aggravating circumstances which may be supported by the evidence:

(1) The offense of murder, rape, armed robbery, or kidnapping was committed by a person with a prior record of conviction for a capital felony, *or the offense of murder was committed by a person who has a substantial history of serious assaultive [sic] criminal convictions*;<sup>184</sup>

(2) The offense of murder, rape, armed robbery, or kidnapping was committed while the offender was engaged in the commission of another capital felony or aggravated battery, or the offense of murder was committed while the offender was engaged in the commission of burglary or arson in the first degree;

<sup>&</sup>lt;sup>182</sup> During his campaign for the U.S. Presidency in 1976, Carter issued a position paper that brings into question his comprehension of the Georgia death penalty statute that he signed into law. Carter wrote, "My position on the death penalty was spelled out as Governor. It should be retained for a few aggravated crimes like murder committed by an inmate with a life sentence. The penalty must be assessed by a jury and must be reviewed in each case by a three-judge panel of the State Supreme Court. Since there has not been an execution since 1967 in the United States, the death penalty actually means ineligibility for parole consideration" (Carter 1976). The Georgia statute, however, made no provisions for a three-judge review panel, nor was it limited to crimes like murder committed by an inmate currently serving a life sentence. <sup>183</sup> 1973 Ga. Laws 74, § 3; Ga. Code Ann. § 27-2534.1 (1973).

<sup>&</sup>lt;sup>184</sup> The italicized portion of the death penalty statute was declared unconstitutionally vague by the Georgia Supreme Court in *Arnold v. State*, 224 S.E.2d 386 (Ga. 1976).

(3) The offender, by his [*sic*] act of murder, armed robbery, or kidnapping, knowingly created a great risk of death to more than one person in a public place by means of a weapon or device which would normally be hazardous to the lives of more than one person;

(4) The offender committed the offense of murder for himself [*sic*] or another, for the purpose of receiving money or any other thing of monetary value;

(5) The murder of a judicial officer, former judicial officer, district attorney or solicitor-general, or former district attorney, solicitor, or solicitor-general was committed during or because of the exercise of his or her official duties;

(6) The offender caused or directed another to commit murder or committed murder as an agent or employee of another person;

(7) The offense of murder, rape, armed robbery, or kidnapping was outrageously or wantonly vile, horrible, or inhuman in that it involved torture, depravity of mind, or an aggravated battery to the victim;

(8) The offense of murder was committed against any peace officer, corrections employee, or firefighter while engaged in the performance of his [*sic*] official duties;

(9) The offense of murder was committed by a person in, or who has escaped from, the lawful custody of a peace officer or place of lawful confinement; or

(10) The murder was committed for the purpose of avoiding, interfering with, or preventing a lawful arrest or custody in a place of lawful confinement, of himself [*sic*] or another.

(c) The statutory instructions as determined by the trial judge to be warranted by the evidence shall be given in charge and in writing to the jury for its deliberation. The jury, if its verdict is a recommendation of death, shall designate in writing, signed by the foreman of the jury, the aggravating circumstance or circumstances which it found beyond a reasonable doubt. In non-jury cases the judge shall make such designation. Except in cases of treason or aircraft hijacking, unless at least one of the statutory aggravating circumstances enumerated in subsection (b) of this Code section is so found, the death penalty shall not be imposed.

At the time of *Furman*, the Criminal Code of Georgia defined capital offenses to include murder, rape, armed robbery, and kidnapping. As noted above, the legislation stated that once a defendant noticed for the death penalty was convicted of any capital offense, the trial would move to the pre-sentencing hearing where the prosecution must prove the existence of at least one of the ten statutory aggravating circumstances.<sup>185</sup> The defendant would also be allowed to present any mitigating circumstances or evidence in support of the contention that the death penalty should not be applied in her or his case. With very few changes, Georgia's death penalty legislation has remained in place since Governor Jimmy Carter first signed it into law.<sup>186</sup>

Jesse Lee Coley became the first person sentenced to death in Georgia after the new death penalty legislation was enacted. Coley, a black man, was convicted of raping a white woman in Wayne County, Georgia, and sentenced to death on April 27, 1973 less than a month after Governor Carter signed the new death penalty bill into law. Following his death sentence, Coley appealed to the Georgia Supreme Court. The Court held that the new death penalty statute "[did] not offend the principles of decisions of the U.S. Supreme Court in *Furman* and *Jackson*" (*Coley v. State*, 204 S.E.2d 612, 616–17 [Ga. 1974]). Although the Court ruled that the new legislation was not in violation of the U.S. Constitution or the Georgia Constitution, it held that Coley's death sentence was excessive when compared to the penalty imposed in similar cases. Coley's sentence was

<sup>&</sup>lt;sup>185</sup> The lone exception to this requirement was for the crime of aircraft hijacking or treason. For these crimes, no evidence of statutory aggravating circumstances was needed.

<sup>&</sup>lt;sup>186</sup> In 1996 and 1997, there were two unsuccessful proposals to lower the age of eligibility for the death penalty to sixteen. There were also two attempts to add an additional aggravating circumstance that would allow the death penalty in the event a person was convicted of the rape of a child under the age of twelve; however this legislation was also unsuccessful (Mears 1999, p. 46).

overturned and he was resentenced to life in prison.<sup>187</sup> After the *Coley* decision, it became clear to death penalty opponents that constitutionality of Georgia's new death penalty law would ultimately be tested in the U.S. Supreme Court.

*Gregg v. Georgia (1976).* Troy Leon Gregg, a white male, was convicted of murdering and robbing two men in Gwinnett County, Georgia. Gregg received four death sentences from a Gwinnett County jury: one for each murder and one for each armed robbery, and he appealed his sentences to the Georgia Supreme Court. The Court affirmed his two death sentences for the murders, but vacated his two death sentences for armed robbery, holding that the death penalty for armed robbery was excessive and disproportionate to the penalties imposed in similar cases (*Gregg v. State*, 210 S.E.2d 659 [Ga. 1974]).

The U.S. Supreme Court granted certiorari in Gregg's case in order to ultimately decide on the constitutionality of Georgia's new death penalty legislation. On July 2, 1976, approximately one and a half years after the Georgia Supreme Court approved the state's new death penalty law, the U.S. Supreme Court approved Georgia's death penalty legislation (by a vote of seven-to-two), holding that it satisfied the requirements of *Furman* (*Gregg v. Georgia*, 428 U.S. at 196–207, 220–24). As discussed in Chapter Two, the Court also approved the guided-discretion statutes adopted by Florida (*Proffit v. Florida*) and Texas (*Jurek v. Texas*), while invalidating the mandatory statutes from North Carolina (*Woodson v. North Carolina*) and Louisiana (*Roberts v. Louisiana*).

<sup>&</sup>lt;sup>187</sup> Shortly after the *Coley* decision, the Georgia Supreme Court reversed death sentences of two other defendants convicted of armed robbery, holding that such sentences were excessive and disproportionate to the sentences imposed in similar cases (see *Floyd v. State*, 210 S.E.2d 810 [Ga. 1974]; *Jarrell v. State*, 216 S.E.2d 258 [Ga. 1975]). In both of these cases, however, the defendants were also given death sentences for murder, and these sentences were affirmed by the Court. In *Jarrell*, the Court also affirmed the death sentence for the crime of kidnapping.

Coker v. Georgia (1977) and Eberheart v. Georgia (1977). The Georgia Supreme Court's decisions in *Coley* and *Gregg* seriously brought into question the appropriateness of the death penalty for rape and armed robbery, respectively. Although the Georgia Supreme Court's decisions in these cases did not officially invalidate the death penalty for rape and armed robbery, it clearly revealed the Court's belief that such a penalty for these crimes was excessive and inappropriate. Nonetheless, the Georgia General Assembly did not narrow the scope of capital crimes; therefore rape, armed robbery, and kidnapping, along with murder, remained death eligible offenses. The U.S. Supreme Court agreed to hear two death penalty cases originating out of Georgia that challenged the constitutionality of the death penalty for non-homicide offenses: Coker v. Georgia and Eberheart v. Georgia. The first case was that of Ehrlich Anthony Coker, a white male convicted of raping a white woman in Ware County, Georgia and sentenced to death. The second case was that of John Wallace Eberheart, a black man sentenced to death for the rape and kidnapping a white woman in Cook County, Georgia.<sup>188</sup> One year following the *Gregg* decision, the U.S. Supreme Court held that the death penalty for the crimes of rape and kidnapping *with* bodily injury in which no killing occurred was a violation of the Eighth and Fourteenth Amendments (see Chapter Two). Later that year, in Collins v. State (236 S.E.2d 759 [1977]), the Georgia Supreme Court applied the rationale of *Coker* and *Eberheart* to the offenses of armed robbery and kidnapping without bodily injury and held that the death penalty could not be imposed for these offenses as well.189

<sup>&</sup>lt;sup>188</sup> Eberheart's accomplice, John Wesley Hooks, was convicted of rape and also sentenced to death.
<sup>189</sup> While the scope of the death penalty has been significantly narrowed with respect to non-homicide offenses, economist Ilyana Kuziemko (2006, p. 137 n.15) suggests that any homicide is potentially a capital offense in Georgia (see also Rosen 1986).

Recall that the NAACP Legal Defense Fund began challenging the constitutionality of the death penalty for rape based upon racial disparities in the administration of capital punishment in Southern States in the early 1950s (see Chapter Two). In 1963, the U.S. Supreme Court denied certiorari in a case involving a black defendant and a white victim that challenged the constitutionality of the death penalty for rape (Rudolph v. Alabama, 375 U.S. 889).<sup>190</sup> The Coker case was the first time that the U.S. Supreme Court ruled on the constitutionality of the death penalty for rape, but the Court did not specifically address the issue of racial disparities in the imposition of the death penalty for rape, instead holding that rape "does not compare with murder" and "is an excessive penalty for the rapist, who, as such, does not take human life" (Coker v. Georgia, 433 U.S. at 598). It is probably not a coincidence that the case on which the Court decided to invalidate the death penalty for rape involved a white offender and a white victim, although these cases very rarely resulted in a death sentence (see LaFree 1989; Wolfgang and Riedel 1975). The Coker decision allowed the Court to sidestep the larger issue of rampant racial bias in the administration of the death penalty. As White (1991, p. 135) notes, "By blocking the death penalty for rape, the U.S. Supreme Court blocked the continued production of what had been the strongest evidence of racial bias in capital punishment."191

<sup>&</sup>lt;sup>190</sup> In 1969, the LDF also challenged the constitutionality of the death penalty for armed robbery in which no one was killed on grounds that it constituted cruel and unusual punishment because it was so infrequently and arbitrarily applied in such cases (see *Boykin v. Alabama*, 395 U.S. 238). The U.S. Supreme Court granted Boykin a new trial based on a procedure issue, but avoided addressing the question of whether the death penalty for armed robbery was a violation of the Eighth Amendment (see Chapter Two).

<sup>&</sup>lt;sup>191</sup> The United States has not executed anyone convicted of committing a non-homicide offense since the death penalty was reinstated in 1976. Fourteen states and the federal government *do* authorize the death penalty for non-homicide offenses, including the rape of a minor, treason, aggravated kidnapping and aircraft hijacking, but only Louisiana has actually sentenced an individual to death for an offense other than murder since *Coker v. Georgia* (D'Avella 2006). The Louisiana Supreme Court held that a statute allowing

*Presnell v. Georgia (1978).* In the same year as the *Gregg* decision, a jury in Cobb County, Georgia, sentenced Virgil Delano Presnell to death for the murder of an eight year-old girl. Presnell appealed his sentence, arguing that he was neither tried nor convicted of the aggravating circumstance on which the jury based his death sentence kidnapping with bodily injury and aggravated sodomy. In 1978, the U.S. Supreme Court reversed Presnell's death sentence, holding that the sentence violated Presnell's due process rights under the Fourteenth Amendment. According to the Court, juries must find legally valid aggravating circumstances to warrant a death sentence (*Presnell v. Georgia*, 439 U.S. 14). Presnell's case was remanded for resentencing and he was again sentenced to death in 1999.

*Francis v. Franklin (1985).* The year after the *Presnell* decision, Raymond Lee Franklin was sentenced to death for the kidnapping and murder of a Bibb County, Georgia resident. While serving a prison sentence for an unrelated crime, Franklin shot and killed his victim during a botched escape attempt after visiting a local dentist's office. Franklin claimed that the shooting was an accident and that the pistol fired when the victim unexpectedly slammed his door after Franklin demanded the key to the victim's car. The U.S. Supreme Court held that the trial judge improperly instructed the jury that Franklin had the burden of proof to show that he did not intend to kill the victim,

the death penalty for the non-fatal aggravated rape of a minor under the age of 12 (see La. Rev. Stat. Ann. § 14:42[c]) was constitutional and did not violate *Coker* (*State v. Wilson*, 685 So. 2d 1063 [La. 1996]); however the ruling was on a pre-trial order and not a conviction and sentence. The court reasoned that, given the nature of the crime and the severity of harm inflicted on the victim and society, the death penalty was not excessive. Moreover, the court believed that deference should be given to the legislature to determine the appropriateness of the death penalty for cases of rape of a child less than 12 years of age.

In June 2008, the U.S. Supreme Court held that the Louisiana statute authorizing the death penalty for the non-fatal rape of a minor violated the Eighth Amendment (*Kennedy v. Louisiana*, 554 U.S. \_\_\_\_\_, 128 S.Ct. 2641 [2008]). Writing for the majority, Justice Kennedy explained that only six jurisdictions permitted the death penalty for child rape, and therefore it was a disproportionate punishment for that crime. In his dissenting opinion, Justice Alito criticized the majority's focus on the absence of a national consensus because *Coker* had impaired legislative consideration of the issue for child victims.

(*Francis v. Franklin*, 471 U.S. 307 [1985]). According to the Court, the Fourteenth Amendment requires the State to prove every element of a criminal offense beyond a reasonable doubt. A capital defendant enjoys the presumption of innocence and does not have the burden of proof in a capital case; it is the prosecution's task to proof beyond a reasonable doubt that a defendant intended to commit the crime (see also *Enmund v. Florida*, 458 U.S. 782 [1982]; *Tison v. Arizona*, 481 U.S. 137 [1987]).

McCleskey v. Kemp (1987). As noted earlier, death penalty opponents had repeatedly attempted to challenge the racially discriminatory application of capital punishment since the early 1950s (see also Chapter Two). In 1968, the LDF presented data to the Eighth Circuit of the U.S. Court of Appeals demonstrating that black males convicted of raping white women were disproportionately sentenced in death in the South (see *Maxwell v. Bishop*, 398 F.2d 138). Although the Court acknowledged that the data demonstrated a historical pattern of racial bias in the administration of the death penalty, it affirmed Maxwell's sentence, holding that the data did not prove purposeful discrimination in his particular case. Four years after the Maxwell case, in the Furman decision, the U.S. Supreme Court held that racial bias had not been proven, but suggested that proof of purposeful discrimination would constitute cruel and unusual punishment. Eleven years later, the U.S. Supreme Court granted certiorari in another case from Georgia, Zant v. Stephens (462 U.S. 862 [1983]), and held that race-of-defendant or raceof-victim discrimination, whether covert or overt, was constitutionality impermissible (see Chapter Two).<sup>192</sup>

<sup>&</sup>lt;sup>192</sup> Alpha Otis Stephens was sentenced to death for murder and during his appeal he argued that one of the three statutory aggravating circumstances on which his sentence was based was invalid. The Court affirmed Stephens's death sentence, holding that his sentence was valid as long as the jury properly found at least one aggravating circumstance.

Following the *Furman* ruling, death penalty opponents began emphasizing the collection of social science evidence on pivotal issues surround the death penalty (Haines 1996). When Warren McCleskey—a black man sentenced to death for the 1978 murder of a white police officer during a burglary in Atlanta, Georgia—appealed his sentence to the U.S. Supreme Court on the grounds it had been imposed in a racially discriminatory manner, the LDF finally had its opportunity to present evidence of racial bias from a massive statistical study of Georgia's death penalty system (for a detailed discussion, see Chapter Two). McCleskey's attorneys' argued that the racially discriminatory administration of capital punishment in Georgia violated their client's right to due process and protection from cruel and unusual punishment under the Fourteenth and Eighth Amendments, respectively.<sup>193</sup>

The U.S. Supreme Court affirmed McCleskey's death sentence, by a vote of fiveto-four, ruling that the data presented on behalf of McCleskey failed to prove purposeful discrimination.<sup>194</sup> The Court ruled that the data presented in *McCleskey* combined the decisions of several relatively autonomous actors from multiple jurisdictions across the state, making it impossible to pinpoint any potential source of bias. More damaging to death penalty opponents, however, was the Court's belief that the standard methods of proving purposeful discrimination in employment and jury selection cases did not apply in the capital sentencing context (see also Acker 1993; Ellsworth 1988). Although the

<sup>&</sup>lt;sup>193</sup> Several empirical studies of Georgia's post-*Furman* capital punishment system had been published prior to *McCleskey*. Baldus and colleagues' (1983, 1985, 1986b) and Gross and Mauro's (1984) analyses revealed that Georgia's death penalty system continued to operate in an arbitrary and discriminatory manner. These pre-*McCleskey* studies were conducted for academic purposes, but as noted in Chapter Two, the LDF sponsored Baldus *et al.*'s (1990) "Charging and Sentencing Study" (CSS) to specifically challenge the constitutionality of Georgia's death penalty system. The statistical evidence presented in *McCleskey* consisted of both the "Procedural Reform Study" (PRS), which was conducted by Baldus *et al.* solely for academic purposes, and the CSS.

<sup>&</sup>lt;sup>194</sup> The State of Georgia executed Warren McCleskey on September 25, 1991.

Court did not foreclose the possibility of challenging racial bias in the administration of the death penalty, subsequent evidence based on the decision-making patterns of one individual or from one jurisdiction has also been rejected by the courts. According to Blume and colleagues (1998), the major obstacle to these challenges is the level of proof necessary to shift the burden of proof on death penalty decision-makers.<sup>195</sup>

In sum, the U.S. Supreme Court's rulings in Furman, Gregg, Coker, Eberheart,

*Presnell, Franklin*, and *McCleskey* have significantly influenced the administration of capital punishment across the country.<sup>196</sup> Though these seven cases have been the most influential cases originating in Georgia, legal analysts have noted that at least nine additional cases from Georgia have also had an important impact on the capital

<sup>&</sup>lt;sup>195</sup> Indeed, the evidentiary standard required by the Court to prove racial bias may be unreasonably high. For example, three years after *McCleskey*, the Georgia Supreme Court ruled that the trial court did *not* error in refusing to consider a post-sentence affidavit of a juror stating that she overheard two white jurors making racially derogatory comments about the defendant during deliberations (see *Spencer v. State*, 398 S.E.2d 179 [Ga. 1990]).

<sup>&</sup>lt;sup>196</sup> Prior to the U.S. Supreme Court's landmark *Batson* decision in 1986, in which it ruled that peremptory challenges may not be used to exclude jurors solely because of their race, the Court overturned the death sentence of Phil Whitus—a black man from Mitchell County, Georgia—on the grounds that blacks were systematically excluded from Georgia grand juries (see *Whitus v. Georgia*, 385 U.S. 545 [1967]). At the time of Whitus's trial, no black person had ever served on a grand jury or petit jury in Mitchell County and jury lists were derived from the County's tax returns that were color-coded based upon race (white sheets for whites and yellow sheets for blacks). The Court held that purposeful exclusion of blacks from the grand jury was unconstitutional (see also Snoddy 2002).

As part of the "Criminal Justice Act of 2005," the Georgia General Assembly recently passed bills reducing the number of peremptory jury strikes available to capital defense attorneys from twenty to ten (see H.R. 170 & S. 2, 148th Gen. Assem., Reg. Sess. [Ga. 2005]). The bills also give prosecuting attorneys the same number of peremptory strikes as capital defense attorneys. Previously, defense attorneys were allowed twice as many peremptory strikes as prosecutors, but the prosecutors challenged the statute allowing defendants a greater number of peremptory strikes (Ga. Code Ann. § 15-12-165 [2005]), claiming the statute was unconstitutional because it allowed defendants to use peremptory challenges in a discriminatory manner (Geerdes and Cox 2005, p. 151). The Georgia Supreme Court ruled that the statute was constitutional, noting that the Equal Protection Clause prohibited racial and gender discrimination in jury selection, and therefore the proper remedy was already available through *Batson* or *J.E.B.* motions (see Robinson v. State, 598 S.E.2d 466, 468 [2005]). The "Criminal Justice Act of 2005" allowed the Georgia General Assembly to circumvent the Georgia Supreme Court's ruling by revising the statute. As noted in Chapter Four, prosecutors in death penalty cases are significantly more likely to use peremptory strikes against black jurors because, irrespective of the race of the defendant, black jurors are less likely to vote for the sentence of death. Allowing prosecutors and capital defense attorneys an equal number of peremptory strikes is likely to significantly reduce the number of blacks serving on capital juries, particularly considering that blacks in Georgia are less likely to be able to serve on juries in the first place because of the State's of felony disenfranchisement laws (see King and Mauer 2004).

punishment process nationwide (for a discussion of these other cases, see Cook 1996). These cases have also placed Georgia's capital punishment system under a microscope in the death penalty community; nevertheless, it appears that race continues to figure prominently in Georgia's capital charging-and-sentencing process. Between 1976 and the U.S. Supreme Court's *McCleskey* ruling in April of 1987, 85.7 percent of executions in Georgia were of black offenders convicted of murdering white victims (6 of 7 executions). Since the *McCleskey* decision, only 16 percent of executions in Georgia have been of black offenders convicted of murdering white victims; however 32 of the 36 executions after the *McCleskey* ruling have been of individuals convicted of murdering white victims (89 percent) (Death Penalty Information Center 2008). Moreover, not a single white offender has been executed for the murder of a black victim in Georgia since the *Gregg* decision (see also Radelet 1989; Radelet and Mello 1986).<sup>197</sup>

## 5.2.2 Race/Ethnicity, Gender, Region, and Georgia's Death Penalty

*Race/Ethnicity.* As noted in Chapter Two, Baldus and colleagues (1990) discovered that prosecutors were significantly more likely to seek the death penalty and judges and juries were significantly more likely to impose the death sentence in murder cases involving white victims in Georgia, even after considering over 230 case factors. On January 31, 2009, there were 106 individuals on death row in Georgia: 54 whites (50.9 percent), 48 blacks (45.3 percent), three Hispanic (2.8 percent), and one Asian American (0.9 percent). Eighty-one were convicted of murdering white victims (77.4 percent), 23 were convicted of murdering black victims (21.9 percent), and one was

<sup>&</sup>lt;sup>197</sup> Nationally, only 1.3 percent of executions since 1976 have been of white offenders convicted of murdering black victims—the first was in 1991 in South Carolina and the most recent was in 2006 in Virginia (Death Penalty Information Center 2008).

convicted of murdering a Hispanic victim (0.95 percent).<sup>198</sup> Twenty-five of the 47 black death inmates row for which there was race-of-victim information were convicted of murdering one or more white victims (53.2 percent). This statistic is particularly noteworthy considering nearly 90 percent of black-offender homicides involved black victims in Georgia from 1976 through 2000. Fifty-three of the 54 white inmates on death row were convicted of murdering white victims (98.1 percent), which is similar to the 93 percent of white offender homicides involving white victims in Georgia during the same time period (Fox 2005).<sup>199</sup>

*Gender*. Female homicide offenders in Georgia are grossly underrepresented on death row. Only one of the 106 death row inmates in Georgia is female (0.9 percent). From 1976 through 2000, however, over 16 percent of homicide offenders in Georgia were female. One the other hand, female-victim homicide cases are overrepresented on death row. Currently, 59.4 percent of current death row inmates in Georgia were convicted of murdering *at least* one female victim, although between 1976 and 2000 less than one-quarter of all homicide cases involved a female victim (Fox 2005). There may also be an interaction between race/ethnicity and gender. Of the 57 death row inmates convicted of murdering at least one female victim, only 18 were convicted of murdering black females (31.5 percent).

*Region.* Death penalty scholars have consistently noted the dramatic jurisdictional variation with respect to death charging and sentencing in Georgia (see Baldus *et al.* 1990; Kroll 1991). Variation in the use of capital punishment only moderately corresponds to the actual homicide activity in these jurisdictions. Since 1973,

<sup>&</sup>lt;sup>198</sup> Race-of-victim information was missing for one case.

<sup>&</sup>lt;sup>199</sup> As is customary in the death penalty literature, homicides in which *at least* one of the victims is white are considered "white-victim" homicides (see Baldus *et al.* 1990).

there has been at least one homicide in 157 of Georgia's 159 counties (98.7 percent), yet only 55 counties (35 percent) are responsible for the state's entire death row population (see Fox 2005; Georgia Department of Corrections 2002). Moreover, 11 of those 55 counties (24 percent) are responsible for 53.7 percent of the death row population (57 of 106 death row inmates). Since Georgia's post-*Furman* death penalty statute was enacted in 1973, through 2008, 344 individuals have been sentenced to death. These 344 death sentences come from 104 of Georgia's 159 counties (65.4 percent), and nearly half of these of these death sentences (167 or 48.5 percent) come from just 14 counties.<sup>200</sup>

With respect to capital charging, prosecutors from 91 counties in Georgia (57 percent) have filed all of the 381 death penalty notices for homicides occurring between 1993 and 2000 (see Table 1). Three counties (1.9 percent)—Fulton (Atlanta), Dekalb (Decatur), and Chatham (Savannah)—accounted for nearly one-half of the reported homicides, but only 14 percent of death notices during this period. Particularly interesting is that nearly one-third of Georgia homicides occurred in Fulton County, although approximately five percent of death notices (and 1.8 percent of death sentences) came from Fulton County between 1993 and 2000. The nine counties with the highest death noticing rates (i.e., percentage of homicides noticed for the death penalty)— Effingham (100 percent), Morgan (90 percent), Appling (83 percent), Gilmer (67 percent), Putman (62 percent), Hart (60 percent), Oconee (58 percent), Fayette (57 percent), and Burke (56 percent)—account for only 3.3 percent of the state's population

<sup>&</sup>lt;sup>200</sup> The fourteen counties are: Baldwin, Bibb, Chatham, Clayton, Cobb, Dekalb, Douglas, Floyd, Fulton, Glynn, Gwinnett, Hall, Muscogee, and Richmond.

and 1.3 percent of homicides, but 12.6 percent of death notices between 1993 and 2000 (Georgia Bureau of Investigation 2001).<sup>201</sup>

# 5.2.3 Miscarriages of Justice: Innocence and Prejudicial Error

Innocence. As discussed in Chapter Three, between 1973 and 2008, 130 individuals have been released from death row after having their conviction overturned and being either acquitted at re-trial or having all charges dismissed or after being given an absolute pardon by based on new evidence of innocence (Death Penalty Information Center 2008). Since Georgia reinstated its death penalty in 1973, four death row inmates have been released because of their innocence—all of them have been black males (Mears 1999, pp. 169-80).<sup>202</sup> Earl Charles (convicted 1975; released 1978) was erroneously convicted for the double murder of two merchants in Chatham County. Charles was released after it was discovered that a police detective falsified evenitness identifications and persuaded a witness to lie in court. Charles, who was in Tampa, Florida at the time of the homicides, subsequently won a settlement from Savannah city officials for misconduct in the original investigation. Jerry Banks (convicted 1975; released 1987) was falsely convicted of a double homicide in Henry County, and later released when it was discovered that the prosecutor withheld exculpatory evidence and police planted physical evidence at the scene of the crime.<sup>203</sup> Robert Wallace (convicted 1980; released 1987) was sent to death row for the shooting death of a police officer. Wallace was acquitted at re-trial when it was discovered that he acted in self-defense.

<sup>&</sup>lt;sup>201</sup> These counties are also located in very rural areas, with only one—Fayette County—having more than 30 percent of its population living in an urban area.

<sup>&</sup>lt;sup>202</sup> Two of these cases involved white victims (Charles and Banks) and two involved black victims (Wallace and Nelson).

<sup>&</sup>lt;sup>203</sup> Banks committed suicide six months after being released from death row (Mears 1999).

Finally, Gary Nelson (convicted 1980; released 1991) was released after spending eleven years on death row for the alleged murder of a female acquaintance in Chatham County. Prosecutors in Nelson's case falsified physical evidence and concealed exculpatory evidence in the case and the police officer in the case lied about the murder weapon.

James Creamer and Henry Drake were also erroneously sentenced to death in Georgia, however they both were serving life sentences by the time they were released based on evidence of their innocence. Creamer, a white male, was sentenced to death by a Cobb County jury in 1973 and served two years on death row before his release. Creamer's death sentence was imposed under the same statute that was in effect at the time of the *Furman* decision, so his death sentence was commuted to life imprisonment following the U.S. Supreme Court's moratorium on the death penalty. The Georgia Supreme Court would later overturn Creamer's conviction after it was discovered that the prosecution withheld and destroyed exculpatory evidence. Soon after Creamer's conviction was reversed, another person confessed to the murders for which Creamer had been sentenced to death. Although Creamer was not on death row at the time his sentence was reversed, the sole reason that he was not on death row before his release was the U.S. Supreme Court's ruling in *Furman*, a completely unrelated case. Drake, also a white male, was sentenced to death in Madison County for an alleged murder and armed robbery. Drake's conviction was solely based the testimony of his co-defendant, William Campbell. Drake's initial conviction and sentence were reversed on federal appeal and, after a second trial, he was sentenced to life imprisonment. Six months after Drake's resentencing, after William Campbell recanted his original testimony and admitted that Drake was not involved in the homicide and a state medical examiner

presented evidence suggesting that Drake was not at the crime scene, the Georgia Board of Pardons and Parole released Drake based on the belief that he was innocent. Both Creamer and Drake were falsely convicted for murdering white victims (Mears 1999).

Prejudicial Error. Liebman and colleagues' (2000a) landmark study of serious prejudicial error in the capital punishment process revealed that 339 death sentences were imposed in Georgia between 1973 and 1995 (see also Liebman et al. 2002).<sup>204</sup> The authors define "serious prejudicial error" as error that the court believes "substantially undermines the reliability of the guilt finding or death sentence imposed at trial" (Liebman et al. 2000a, p. 6). Of those 339 cases, 309 (91 percent) had been reviewed on direct appeal, and 112 (36 percent) were reversed because of serious prejudicial error. Out of the 197 cases that were forwarded to state post-conviction, an additional 24 cases were subsequently reversed (12 percent). Collectively, 136 of the 309 death sentences (44 percent) reviewed during state direct appeal and state post-conviction were reversed. At the federal level, 63 of the 97 cases reviewed during federal habeas corpus were reversed (65 percent). Liebman *et al.* define the "overall error rate" as the proportion of capital judgments thrown out during the first (state direct appeal) inspection due to serious error, plus the proportion of the original judgments that survive the first inspection but are thrown out at the second (state post-conviction) inspection, plus the proportion of the original judgments that survive both state inspections but are thrown out at the final (federal habeas) stage. According to their calculations, Georgia's error rate of

<sup>&</sup>lt;sup>204</sup> It is important to note that the total number of capital defendants sentenced to death during this time was 270. Several capital defendants were sentenced to death multiple times. This would usually occur when a capital defendant was sentenced to death, her or his sentence and/or conviction was reversed on appeal because of reversible error, and she or he was subsequently resentenced to death (for a list of these cases, see Mears 1999, Appendix A).

80 percent was well above the national average of 68 percent.<sup>205</sup> Of this 80 percent, 39 percent of reversals were the result gross ineffectiveness on the part of defense counsel, 20 percent involved jury mis-instruction, 19 percent involved prosecutorial misconduct, and four percent involved biased judges or juries (Liebman *et al.* 2000a). It must be emphasized that serious prejudicial error is not commonly discovered in other types of criminal cases. For example, in Georgia between 1970 and 2003, criminal defendants alleged prosecutorial error or misconduct in 449 cases (22 of these cases were death penalty cases) (Weinburg 2003). Of these 449 cases, the court found harmful error and granted reversals in only 39 cases (8.7 percent).<sup>206</sup>

# 5.2.4 *Life Without the Possibility of Parole*

Currently all 35 states with capital punishment statutes offer life without the possibility of parole (LWOP) as a sentencing option in death penalty cases (see Chapter Three). Georgia's LWOP legislation became effective on May 1, 1993.<sup>207</sup> The year following the enactment of the legislation, Georgia's Board of Pardons and Parole challenged the statute, contending such legislation violated the state constitution's separation of powers provision.<sup>208</sup> The Georgia Supreme Court rejected the Board of

<sup>&</sup>lt;sup>205</sup> The range was from 18 percent (Virginia) to 91 percent (Mississippi). It is also noteworthy that Georgia had the fourth highest reversal rate at the federal habeas stage in the United States and the second highest among Southern states (among states with at least four federal habeas cases completed). Georgia also had the highest federal habeas reversal rate in Eleventh Circuit U.S. Court of Appeals (Alabama and Florida are the other two states in the Eleventh Circuit) (Gelman *et al.* 2004; Liebman *et al.* 2000a)

<sup>&</sup>lt;sup>206</sup> If one were to include the 17 additional cases in which at least one dissenting or concurring judge believed that harmful error was present, only 12.5 percent of the 449 cases would have been reversed do to serious prejudicial error on the part of the prosecution (Weinburg 2003).

<sup>&</sup>lt;sup>207</sup> 1993 Ga. Laws 569, § 4; Ga. Code Ann. § 17-10-30.1 (1993).

<sup>&</sup>lt;sup>208</sup> The Georgia Board of Pardons and Parole was created in 1943 and is granted the power to grant executive clemency, which includes reprieves, pardons, and commutations (Ga. Const. art. IV, § 2, pt. II(a)). The Board consists of five-members who are appointed by the Governor and subject to Senate confirmation. The Georgia Attorney General, who is also appointed by the Governor, serves as a legal advisor to the Board. Although the Governor is given the authority to appoint the Board, she or he has no direct influence to grant (or deny) clemency.

Pardons and Parole argument, holding that "the power to create crimes and to prescribe punishment therefore is legislative" (see *Freeman v. State*, 440 S.E.2d 181, 184 [Ga. 1994]). Citing a 64 year-old case (see *Johnson v. State*, 152 S.E. 76 [Ga. 1930]), the court went on to say that because LWOP legislation, like death penalty legislation, renders a defendant ineligible for parole in the first instance, it did not constitute a separation of powers violation.

Two years after the Georgia Supreme Court validated the state's life without parole legislation, it ruled six-to-one to restrict LWOP to death penalty cases. Writing for the majority, Justice Carole Hunstein wrote, "The unavoidable result of the legislative enactment is to bar the State from seeking life without parole unless the State has filed a notice of intent to seek the death penalty" (see *State v. Ingram*, 467 S.E.2d 523, 525 [Ga. 1996]). The Georgia Supreme Court, agreeing with an earlier ruling from Fulton County Superior Court Judge Elizabeth E. Long (see *State v. Ingram*, No. S96A0158), held that the sentencing laws represented a "coherent statutory plan" under which the same rules must govern the imposition of capital punishment and life without parole.<sup>209</sup>

The Board is required to collect as much information as possible about every inmate who may be eligible for relief under the powers of the Board. The information gathered by the board typically includes, but is not limited to, (1) circumstances of the crime and the sentence received, (2) the names of the judge, prosecutors, defense attorneys, and investigating officers, (3) the pre-sentencing report and any relevant previous court records, (4) any probation reports, and (5) any relevant social, physical, mental, or criminal documents/records about the inmate (Ga. Code Ann. § 42-9-41(a)(1)-(8)).

<sup>&</sup>lt;sup>209</sup> LWOP is an *automatic* sentencing option for other non-homicide offenses. Under Georgia's Sentence Reform Act of 1994 (i.e., the "Two Strikes" Law), any individual convicted of committing any of the "seven deadly sins"—murder, rape, armed robbery, kidnapping, aggravated sodomy, aggravated sexual battery, and aggravated child molestation—for the first time must serve a minimum of ten years in prison, or up to the maximum sentence allowed under law, all without possibility of parole. Individuals convicted of committing any one of the seven offenses for the second time, and who are not being tried for capital murder, are automatically sentenced to life without parole. In 2005, both houses of the Georgia legislature (H.R. 248 & S. 57, 148th Gen. Assem., Reg. Sess. [Ga. 2005]) introduced bills that would have mandated individuals convicted of murder who have previously been convicted of three or more felonies be sentenced to life without parole. The bill was adopted in the Senate, but was withdrawn and recommitted to the Rules Committee in the House of Representatives.

The Court's ruling in *Ingram* may have radically altered the administration of capital punishment in Georgia. Restricting LWOP to murder cases in which the prosecution has filed noticed of intent to seek the death penalty has led many prosecutors to seek the death sentence in cases they do not believe are deserving of the death penalty, but they do believe warrant LWOP. In these instances, LWOP becomes a "bargaining chip" for prosecutors (cf. Kuziemko 2006). Prosecutors are able to threaten defendants with the possibility of facing the death penalty if they do not choose to accept a plea bargain of LWOP. For example, following the jury's sentencing of Devonia Inman to LWOP in Alapaha County, Georgia in 2001, District Attorney Bob Ellis remarked, "Had we not sought the death penalty, we could have not gotten life without parole" (Failor 2001). Ellis told reporters that seeking only life in prison would have left the option open for eventual parole, but by seeking death the jury had the opportunity to deny parole to Inman.

While such a strategy may work well for prosecutors in some cases, allowing them to quickly dispose of murder cases to the satisfaction of the community and the family and friends of the victim(s), it also places many more murder defendants whose alleged crimes would not warrant the death penalty at a much greater risk for the death penalty. A capital defendant may opt to "take her or his chances" at a capital murder trial, hoping to be acquitted or sentenced to a life sentence *with* the possibility of parole (LS), rather than plea to LWOP and spend the rest of her or his natural life in prison.

On January 31, 2009, inmates serving LWOP sentences for the crime of murder comprised slightly more than one-half of one percent (0.54 percent) of the total incarcerated population in Georgia. Of the 300 inmates serving a LWOP sentence for the crime of murder, 205 were black (68.3 percent) and 94 were white (31.3 percent). On that same date, 4,674 inmates were serving LS sentences for murder: 3,244 blacks (69.4 percent), 1,394 whites (29.8 percent), 22 Asian Americans (0.5 percent) and 13 "Other" (0.27 percent). As noted earlier, slight over half of homicide offenders in Georgia from 1976 to 2000 were black (approximately 55 percent), so blacks appear to be somewhat over-represented with respect to LWOP and LS. The proportion of blacks serving LWOP and LS sentences, however, are nearly identical.<sup>210</sup>

One hundred and fifty-one capital defendants with crime dates between 1993 and 2000 received LWOP either by plea bargain or during the penalty trial phase (see Figure 1). Slightly over 65 percent of these capital defendants were black (99), 32.5 percent were white (49), and two percent were Hispanic (3). With respect to the race of the victim, 59.6 percent were cases with *at least* one white victim, 31.1 percent were cases with all black victims, and two percent were cases with Asian American victims.<sup>211</sup> Nearly an identical number of capital defendants were sentenced to LS with incident dates between 1993 and 2000 (see Figure 2). Of these 147 individuals, 58.5 percent were black (86), 38.8 percent were white (57), and 2.7 percent were Hispanic (4). In terms of the distribution of these with respect to the race of the victim, 59.5 percent were cases with *at least* one white victim, 32.1 percent were cases with all black victims, 2.7 percent

<sup>&</sup>lt;sup>210</sup> The overrepresentation of blacks for the crime of voluntary manslaughter is slightly greater. As of January 31, 2009, there were 1,190 inmates imprisoned for voluntary manslaughter in Georgia. Of these 1,190 inmates, 72.6 percent (864) were black, 26.1 percent (311) were white, 0.34 percent (4) were Asian American, and 0.84 percent (10) were "Other." With respect to the 127 inmates serving time for involuntary manslaughter, blacks comprised 59.8 percent (76) and whites comprised 40.2 percent (51). This overrepresentation of blacks for manslaughter may be the result of prosecutors' willingness to offer plea bargains of manslaughter to black offenders. As noted in Chapter Four, Georgia prosecutors are more likely to offer a reduced charged for black-offender/black-victim cases, net of legitimate case characteristics.

<sup>&</sup>lt;sup>211</sup> Race-of-victim information was missing for five cases (3.4 percent).

were cases with Hispanic victims, and two percent were cases with Asian American victims.

To date, no study has systematically examined the potential influence of extralegal factors on the likelihood that a capital defendant receives LWOP versus LS in Georgia. While it is understandable why death penalty scholars and legal activists have focused their attention on the "death/non-death" dichotomy, it is becoming increasingly important to examine more subtle distinctions in sentencing outcomes for capital cases because LWOP has become a popular sentencing alternative in the vast majority of death penalty states (see also Chapter Seven).<sup>212</sup>

#### 5.3 SUMMARY

Despite the U.S. Supreme Court's mandate in *Furman*, it appears that illegitimate factors may continue to play a prominent role in Georgia's capital charging-andsentencing process. The raw data suggest that prosecutors are more likely to seek the death penalty and judges and juries are more likely to impose the death sentence on individuals charged with and convicted of murdering white and female victims. Furthermore, these raw data strongly suggest that the jurisdiction in which a homicide occurs influences the probability that a case is noticed for the death penalty and results in a death sentence, net of the actual homicide rate of the jurisdiction. As noted in Chapter Four, racial justice legislation was introduced in Georgia, but failed to pass in either

<sup>&</sup>lt;sup>212</sup> Until recently, Texas and New Mexico were the only two death penalty states that did not offer LWOP as a sentencing option in capital cases. In June 2005, Texas Governor Rick Perry signed LWOP legislation into law. Interestingly, the very same bill failed by two votes in the state senate just two months earlier (S. 60, 79th Leg., Reg. Sess.). The initial defeat of the bill was particularly noteworthy because nearly 80 percent of Texans supported having LWOP as a sentencing alternative (Robison 2005). New Mexico completely repealed its capital statute in March 2009 (Associated Press 2009).

house of the General Assembly.<sup>213</sup> Georgia's proposed Racial Justice Act (RJA) would have mandated the collection and analysis of countywide data pertinent to the imposition of the death penalty by appropriate county agencies such as district attorney and public defender offices, as well as Superior Courts (Mears 1998b). In fact, as discussed in Chapter Four, the federal government adopted a similar protocol in 1995 requiring federal prosecutors to submit information on all death eligible cases, irrespective of the U.S. Attorney's intent to seek the death penalty (see U.S. Department of Justice 2000). Although the Department of Justice (DOJ) never officially attributed the change in protocol to pressure from RJA proponents, it may be more than mere coincidence that the DOJ adopted the new protocol the year after the RJA was approved in the U.S. House of Representatives.

In 2001 and 2003, resolutions for the creation of the House Study Committee on the Death Penalty<sup>214</sup> and the Georgia Capital Punishment Study Commission,<sup>215</sup> respectively, failed in the Georgia House of Representatives.<sup>216</sup> The proposed Capital Punishment Commission would have been established to specifically consider the issues outlined by the American Bar Association's (ABA) newly established death penalty protocols (see American Bar Association 2001). In February 2003, the ABA's "Death Penalty Moratorium Implementation Project" received a grant from the European Commission's European Initiative for Democracy and Human Rights to examine whether

<sup>&</sup>lt;sup>213</sup> In 2001, 2002, and 2003, different versions a RJA were introduced in the Georgia House of Representatives, but never made it out of committee (see H.R. 324, 146th Gen. Assem., Reg. Sess. [Ga. 2001]; H.R. 1211, 146th Gen. Assem., Reg. Sess. [Ga. 2002]; H.R. 129, 147th Gen. Assem., Reg. Sess. [Ga. 2003]).

<sup>&</sup>lt;sup>214</sup> H.R. Res. 1594, 146th Gen. Assem., Reg. Sess. (Ga. 2002).

<sup>&</sup>lt;sup>215</sup> H.R. Res. 546, 147th Gen. Assem., Reg. Sess. (Ga. 2003).

<sup>&</sup>lt;sup>216</sup> The House Study Committee on the Death Penalty was to be comprised of five members and the Capital Punishment Study Committee was to consist of eleven members from the Georgia House of Representatives.

jurisdictions in America with capital punishment statutes met the minimum standards of fairness and due process established by the ABA. Georgia was selected as one of its five initial "assessment projects" in September of the following year (see American Bar Association 2004)<sup>217</sup> and the ABA released its official assessment report in January 2006, urging the state to place a moratorium on seeking the death penalty after discovering significant flaws in Georgia's administration of capital punishment (American Bar Association 2006b).<sup>218</sup> Perhaps in response to the ABA's focus on Georgia, the 2005 – 2006 Legislative Session of the Georgia General Assembly adopted a resolution to create the Capital Punishment Study Commission. According the resolution, the purpose of the commission is to "study the death penalty" and "urge the suspension of executions until such time as a report from such study commission is submitted to the General Assembly and the General Assembly and governor act in response to recommendations for the study commission."<sup>219</sup> But in light of the General Assembly's pattern of establishing "unfunded mandates" in the area of criminal justice reform (see American Bar Association 2006b; Atlanta Bar Association 2001; Spangenberg Group 2002), it is

<sup>&</sup>lt;sup>217</sup> The four other assessment sites were Alabama, Arizona, Florida, and Tennessee. The ABA began a second set of assessments in January 2005 in Louisiana, Ohio, Oklahoma, South Carolina, Texas, and Virginia.

<sup>&</sup>lt;sup>218</sup> In particular, the ABA discovered seven major problems with the current Georgia death penalty system: (1) no guarantee of defense counsel for condemned inmates in habeas corpus appeals; (2) inexperienced and poorly prepared defense attorneys representing capital defendants; (3) inadequate proportionality review conducted by the State Supreme Court; (4) rampant miscomprehension of judicial instruction by capital jurors; (5) racial disparities in the imposition of the death penalty; (6) unreasonable evidentiary standard for capital defendants in mental retardation cases; and (7) death eligibility for unintentional murder (i.e., felony murder).

The Georgia Supreme issued 159 rulings that upheld the proportionality of death sentences imposed between 1992 and 2007. An investigation conducted by the Atlanta Journal-Constitution found that 129 of these rulings (approximately 80 percent) cited prior death cases that had been overturned. In total, the Georgia Supreme Court cited 76 cases that had been overturned. Particularly troubling was the fact that in 55 of these rulings, at least 25 percent of the cases cited them had been overturned. Only 14 rulings cited no overturned cases (Rankin, Vogel, and Wertheim 2007).

<sup>&</sup>lt;sup>219</sup> H.R. Res. 301, 148th Gen. Assem., Reg. Sess. (Ga. 2005); S. Res. 184, 148th Gen. Assem., Reg. Sess. (Ga. 2005).

questionable whether it will ultimately commit the necessary resources for such a commission.<sup>220</sup>

Considering the current racial/ethnic, gender, and regional distribution of capital cases in Georgia, and the vast empirical literature documenting the historical continuity of these disparities in Georgia's capital charging-and-sentencing process (see Chapter Four), it is very likely that the Georgia General Assembly's Capital Punishment Commission (if established) will discover that these disparities persist (as did the ABA Moratorium Project), even after taking into account a wide range of legitimate legal characteristics. Undoubtedly a plethora of evaluation studies and policy recommendations will emerge from these projects, primarily focusing on how to make the capital punishment system more "even-handed" and "effective." That is, it will be suggested that traditional legal tools should (and must) be used to solve traditional legal problems (Black 2002a, p. 117; Kan and Phillips 2003, p. 71). However, death penalty scholars have questioned whether these "incremental fixes" to the capital punishment system can ever correct or change the legacy of racial/ethnic bias in the administration of the death penalty in America (Howe 2004; Kan and Phillips 2003; Ogletree 2002). Moreover, some scholars have suggested that these minor procedural reforms may actually do more harm than good because they further entrench a failed capital

<sup>&</sup>lt;sup>220</sup> Two years after the General Assembly created Georgia's first statewide public defender system, the state Senate Judiciary Committee introduced several bills to give the executive and legislative branches significantly more control over spending for the newly created system. The proposed legislation would (1) transfer control over the governing body of the public defender system, the Georgia Public Defender Standards Council, from the judicial branch to the executive branch (S. 139, 149th Gen. Assem., Reg. Sess. [Ga. 2007]); (2) give state and local politicians more influence over the Council's operations, including its composition, qualifications and terms for new members, and hiring and firing decisions (S. 140, 149th Gen. Assem., Reg. Sess. [Ga. 2007]); (3) significantly restrict local public defenders' staff hiring (S. 142, 149th Gen. Assem., Reg. Sess. [Ga. 2007]); and (4) allow state judicial circuits to drop out of the system (S. 143, 149th Gen. Assem., Reg. Sess. [Ga. 2007]). Preston Smith, chair of the state Senate Judiciary Committee, specifically cited the cost of defending a high profile capital murder case in Atlanta as motivation for limiting indigent defense spending (Land 2007).
punishment system by providing the appearance of legality and impartiality (see, e.g., Kaufman-Osborn 2006b, p. 369; Ogletree 2002, p. 34).

Admittedly, some death penalty proponents posit that the post-*Furman* procedural safeguards have been successful at reducing pre-*Furman* problems in the administration of the death penalty (Cassell 2003; Marquis 2003), but there is little debate among most death penalty scholars that the vast majority of methodologically rigorous empirical studies of the capital charging-and-sentencing process since 1973 reveal patterns that are remarkably similar to the charging-and-sentencing patterns prior to *Furman* (see Chapter Four). Unfortunately, the bulk of research on the death penalty process has *not* been theoretically informed, and as a result, both advocates and critics of death penalty reform have been unable to offer an adequate explanation of why patterns of racial/ethnic, gender, and regional bias persist *and* why the past three decades of procedural reforms have failed to significantly reduce the amount of arbitrariness, capriciousness, and bias in the administration of capital punishment in America.<sup>221</sup>

Many scholars often attribute racially disparate outcomes to conscious (Sorensen and Wallace 1999) and/or unconscious (Gross and Mauro 1989; Howe 2004; Johnson 1988) racial prejudice among decision-makers in the capital charging-and-sentencing system (but cf. Kleck 1991). Advocates of the overt or conscious racism explanation

<sup>&</sup>lt;sup>221</sup> In 1994, in his dissenting opinion in *Callins v. Collins* (510 U.S. 1141), U.S. Supreme Court Justice Harry Blackmun wrote:

Twenty years have passed since this Court declared that the death penalty must be imposed fairly, and with reasonable consistency, or not at all, [*citation omitted*] and despite the effort of the States and courts to devise legal formulas and procedural rules to meet this daunting challenge, the death penalty remains fraught with arbitrariness, discrimination, caprice, and mistake. This is not to say that the problems with the death penalty today are identical to those that were present 20 years ago. Rather, the problems that were pursued down one hole with procedural rules and verbal formulas have come to the surface somewhere else, just as virulent and pernicious as they were in their original form (510 U.S. at 1144–45).

often highlight evidence of racially motivated misconduct by police, prosecutors, judges, and jurors, such as tampering with evidence, excluding blacks from capital juries, et *cetera* (see Chapter Four). While deliberate acts of racial discrimination certainly play an important role in some capital cases (see, e.g., Blume et al. 1998; Sorensen and Wallace 1999), the "disparities are too great and too wide-spread to be blamed entirely on racial bigots" (Gross and Mauro 1989, p. 110). Moreover, the most consistent finding in the literature is that white-victim cases are treated most punitively, irrespective of the race of the defendant (but cf. Klein and Rolph 1989). True, several studies have discovered that black offenders accused of murdering white victims are treated most severely, but blackoffender/white-victim homicides are relatively infrequent, and as a result, this finding has been difficult to replicate across jurisdictions in well-controlled studies (Baldus et al. 1998). Analysts have also suggested that white-victim cases are treated more severely because capital prosecutors, judges, and jurors: (1) display a blatant disregard for the lives of black victims (Hawkins 1997; Ogletree 2002, pp. 32-33; Sorensen and Wallace 1995), (2) feel more sympathy for white victims (Gross and Mauro 1989, pp. 113–14), and (3) experience more pressure from the community in white-victim cases (Baldus et al. 1998; Bright 1995a).<sup>222</sup> Recent data from interviews of former capital jurors (as opposed to jury simulation studies, see, e.g., Applegate et al. 1993; Lynch and Haney 2000) suggest, however, that the influence of race/ethnicity on decision-makers in the capital punishment process may be more complex. For example, white jurors particularly white male jurors—appear significantly more likely to vote for the death

<sup>&</sup>lt;sup>222</sup> Media coverage of homicides may also have an indirect impact on prosecutorial discretion to seek the death penalty via its impact on community awareness (and subsequent outrage) over the crime. Several researchers have discovered that homicides involving victims who where white, female, extremely youthful/elderly, and wealthy were most likely to receive coverage in the local media irrespective of the heinousness of the crime (see Hawkins, Johnson, and Michener 1994; Sorenson and Berk 1998).

sentence, irrespective of the race of the defendant or race of the victim (Bowers *et al.*) 2001). These interviews also revealed that black jurors are more likely to consider certain types of mitigation evidence (e.g., lingering doubt, remorsefulness, future dangerousness) than white jurors, and white jurors appear to be less receptive to mitigation evidence when the defendant is black (see Bowers and Foglia 2003; Bowers et al. 2001)—which is likely to be related to perceived culpability. Therefore, a combination of black-juror leniency and white-juror severity may be partly responsible for observed death sentencing patterns, irrespective of the race/ethnicity of the defendant(s) and victim(s) involved in these cases. Nonetheless, the legally legitimate characteristics of capital cases remain among the most important factors influencing jurors' decisions. Proponents of conscious and unconscious racism explanations, however, have been largely silent as to why (and under what conditions) blacks may be more receptive to mitigation evidence than whites, even with respect to intra-racial homicides. Racial prejudice explanations, while helpful, are only able to partially account for the observed capital charging-and-sentencing patterns. Relative to the total number of death eligible homicides in jurisdictions with capital punishment statutes, the death penalty is used rather infrequently (cf. Liebman 2000). Even if one were to assume that white-victim homicides are of primary concern to decision-makers in the death penalty system, only a small fraction of white-victim homicide cases are ever subject to the death penalty (Blume et al. 2004; Borg 1998).

The thirty years of empirical research on the modern capital punishment system in the United States has done a remarkable job identifying and describing patterns racial discrimination in the administration of the death penalty. Scholars have been less successful, however, at explaining why these patterns persist in light of all of the scrutiny that the capital punishment system has received during the last three decades. To be sure, death penalty scholars have done a much better job at explaining why the Courts have been unreceptive to statistical and other types of evidence documenting the arbitrary, capricious, and discriminatory application of the death penalty (see, e.g., Baldus et al. 1994b; Blume et al. 1998; Ellsworth 1988; Gross and Mauro 1989; Haney and Logan 1994), but they have largely failed (or neglected) to satisfactorily explain the complex ways in which race/ethnicity, gender, and region influence the administration of capital punishment in the United States. Comprehensive and valid explanations of the role of extra-legal factors are important for more than just scholarly reasons. An accurate understanding of the forces driving these patterns is absolutely necessary for effective policy recommendations and death penalty reform. Perhaps a major reason why procedural "fixes" to the capital punishment process over the past three decades have been unsuccessful is that they fail to address the actual underlying "sources" of these disparities (Howe 2004; Kan and Phillips 2003).<sup>223</sup> The following chapter describes why existing theories of the role of extra-legal factors in the criminal sentencing process have been inadequate in explaining the influence of these extra-legal factors—particularly race/ethnicity—on the capital charging-and-sentencing process and describes an alternative explanatory framework which is believed to overcome these shortcomings.

<sup>&</sup>lt;sup>223</sup> The term "sources" is used rather than "causes" because the term "cause" often implies temporal priority, spatio-temporal contiguity, and constant conjunction; whereas "source" merely implies a sufficient condition (see Chapter Six; see also Lieberson and Lynn 2002, pp. 11–12).

#### **Chapter 6: Presentation and Evaluation of Theoretical Perspectives**

The majority of research on the role of extra-legal factors—particularly race/ethnicity—in the criminal sentencing process has not been strongly guided by theory (Everett and Wojtkiewicz 2002; Mears 1998a; Unnever and Hembroff 1988). Traditionally, scholars have engaged in evaluation studies (i.e., studies of legal effectiveness) rather than research guided by theoretical propositions that attempt to order, predict, and explain variation in the sentencing outcomes (Black 1972). That is, these scholars attempt to examine whether the law is operating—or applied—in a manner consistent with larger abstract standards of how the law should operate (i.e., the concern over "ought" versus "is").<sup>224</sup> These standards (or ideals) arise from a myriad of sources, including constitutional doctrine, moral and other philosophical principles, and federal, state, and local statutes which tend to reify these more general principles (see, generally, Durkheim [1893] 1947). This focus on "law in books" (i.e., ideal law) versus "law in action" (i.e., real law)—referred to as *legal realism*—attracted a strong following when introduced in the late nineteenth- and early twentieth-centuries, and continues to influence contemporary legal scholarship (see Bechtler 1977; Hunt 1978, Chapter 3; Livingston 1982).<sup>225</sup> While the legal realist movement—popularized by Oliver Wendell

<sup>&</sup>lt;sup>224</sup> Black (1972) suggests that much of the sociology of law resembles "technocratic" thought, holding that problems of law reduce to problems of *technique*. A technique is considered desirable if it works and "what works" can be learned through science (pp. 1090–91). The primary focus, then, is to make law more efficient and more effective—to make law more closely realize these ideals (see, e.g., Schwartz 1978b; Tamanaha 1997, Chapter 2). For a similar critique of the sociology of punishment, see Garland (1990).
<sup>225</sup> U.S. Supreme Court Justice Benjamin Cardozo (1947, p. 10) remarked, "[t]he most distinctive product of the last decade [i.e., the 1920's] in the field of jurisprudence is the rise of a group of scholars styling themselves realists." It must be noted, however, that the term "legal realism" lacks a precise definition and, from the outset, many legal realists were sharply critical of one another (Rumble 1965, pp. 547). According to Cahill (1952, p. 97), legal realism "it is less a description than a slogan and carries emotional connotations rather than precise meanings." The early legal realists frequently disagreed about the role of pre-established legal norms in the judicial process and the feasibility of a "value-free" science of law, along the lines of natural science (Rumble 1965, pp. 553–54). Realists on either side of these issues claimed to follow the tradition established by Holmes, but as Rumble points out, Holmes' legacy is sufficiently

Holmes ([1881] 1963), Roscoe Pound (1910), and Karl Llewellyn ([1930] 1978)—is responsible for challenging conventional thinking about legal decision-making by placing a strong emphasis on social context, legal realists "engaged in little research and formulated little theory about the reality of law" (Black 1989, p. 5).

Similarly, most scholars continuing in the legal realist tradition have largely failed to advance or significantly contribute to a comprehensive theory of law (Black 1997; Luhmann 1989), and their research remains primarily concerned with critiquing current legal practices and advocating for legal reform (e.g., Baldus *et al.* 2001; Bowers *et al.* 2001; Gross 2001; Hawkins 1997; Johnson 1988; Langbein 1999; Liebman 2000; Radelet and Vandiver 1986; Whitehead 1998).<sup>226</sup> Perhaps is it not surprising, then, that while the empirical literature concerning the role of extra-legal factors in criminal sentencing in general—and the capital punishment process in particular—has proliferated, theoretical

ambiguous such that a clear resolution of this debate is unlikely (p. 566). Disagreements notwithstanding, the legal realist movement owes much of its intellectual tradition to Roscoe Pound's *sociological jurisprudence* (p. 548). According to Pound, the distinctive features of sociological jurisprudence were four-fold (two methodological and two substantive). With respect to his two methodological claims, Pound insisted that: (1) methods and concepts from the social sciences be used in the study of law and legal institutions and (2) the traditional subject-matter of legal study must be expanded beyond the case-method (i.e., focus on judicial opinions) to include scientific evidence of the relations of the law to society and of the needs and interests and opinions of the contemporary society. His two substantive claims were directed at the attitudes towards legal rules. Pound believed: (1) legal rules, in fact, have a very limited impact on determining legal decisions; and (2) antecedent rules should be as a means to the achievement of social ends (i.e., equitable application of the law) and that judges should be free to deal with the individual case, so as to meet the demands of justice between parties and accord with the general reason of ordinary men (Pound 1910, pp. 20, 35–36; Rumble 1965, pp. 548–49).

<sup>&</sup>lt;sup>226</sup> Undoubtedly, critical legal scholarship is necessary and research examining the effectiveness of our legal system is essential. Likewise, recommendations for policy reform emerging from this line of research are extremely important. Legal scholarship, however, involves more than the examination of the effectiveness of law. As Black (1989, p. 4) notes, legal scholars should also be concerned with constructing "a general theory capable of predicting and explaining legal behavior of every kind… unconcerned with policy and uncontaminated by practical considerations." That is, researchers should be concerned with developing a "pure science" of law (see also Cooney 1986).

understanding of legal behavior has not progressed much since the early formulations of Holmes and Pound (Black 1989; but cf. Evan 1990; Tamanaha 1997).<sup>227</sup>

Theory, however, is not completely absent from the voluminous research on extra-legal factors and sentencing. Theoretical propositions are found—either implicitly or explicitly—in a large number of studies; however, theory tends to plays a minor—if not trivial—role. In fact, most studies remain "legal effectiveness" studies that are driven more by ideology than theory (Schiff 1981), and do little more than apply theoretical explanations *post factum* to interpret empirical results.<sup>228</sup> As a result, there has been relatively little interplay between research and theory (for discussions, see Merton [1949] 1968, Chapters 4 and 5; Parsons 1938). According to Black (2000a, p. 346), "Research is often independent of theory, and theory is often independent of research." General theoretical propositions seldom dictate or guide the selection (or exclusion) of concepts used to explain variation in sentencing outcomes.<sup>229</sup> Similarly, theoretical concepts and propositions are rarely modified (i.e., restricted or expanded) or discarded in light of the growing body of supportive or disconfirming empirical research (Lenski 1988, pp. 166–67; Merton 1948, p. 506).

<sup>&</sup>lt;sup>227</sup> According to Black (2002a, p. 106), "Lawyers had studied law for at least a thousand years without ever formulating a major scientific idea about it."

<sup>&</sup>lt;sup>228</sup> For example, Kempf-Leonard and Sample (2001, p. 111) state, "the theoretical model of a rational case processing system is used *to guide the interpretation of our findings* and the recommendations we offer for improvement" (emphasis added).

Merton (1945, p. 468), however, strongly criticizes this approach to scientific explanation. "*Post factum* explanations remain at the level of *plausibility* (low evidential value) rather than leading to "compelling evidence" (a high degree of confirmation). The logical fallacy underlying the *post factum* explanation rests in the fact that there is available a variety of crude hypotheses, each with some measure of confirmation but designed to account for quite contradictory set of affairs. The method of *post factum* does not lend itself to nullifiability, if only because it is so completely flexible."

<sup>&</sup>lt;sup>229</sup> For example, rather than carefully and self-consciously selecting theoretically relevant variables and assessing their relative explanatory power, researchers merely include a long list of variables in an attempt to maximize the amount explained variation (for discussions of the shortcomings associated with this approach, see Horwitz 1983, p. 370; Mayhew 1981, p. 631).

Despite the aforementioned problems with current theoretical formulations concerning legal behavior, three different theoretical perspectives commonly employed to explain the relationship between race/ethnicity (and other extra-legal factors) and sentencing are presented below (see Section 6.1.2). Following their presentation, these theories are evaluated according to widely accepted scientific standards of theory construction, and their relative strengths and weaknesses are identified. After the presentation and evaluation of these perspectives, a fourth theoretical perspective is introduced and evaluated (see Section 6.2). Not only is it argued that this fourth perspective realizes these scientific criteria more successfully than the other approaches, but that it also introduces a research strategy (i.e., paradigm) that avoids several of the common shortcomings that plague socio-legal theory in general (cf. Scheppele 1994).<sup>230</sup>

### 6.1 DOMINANT THEORETICAL PERSPECTIVES

### 6.1.1 Scientific Criteria of Evaluation

Before discussing the major theoretical perspectives employed when analyzing the relationship between extra-legal factors and criminal sentencing, it is first necessary to outline the criteria on which these theories will be evaluated. According to Harris (1979, p. 5), it is important "to provide general criteria for distinguishing science from other ways of gaining knowledge and for distinguishing one research strategy from another." Although the requirements of "good scientific theory" are usually taken for granted by most social scientists, a close examination of the voluminous theoretical

<sup>&</sup>lt;sup>230</sup> A research strategy (or paradigm) is a set of general assumptions about entities and processes in a domain of study, and about the appropriate methods to be used for investigating the problems and constructing theories in that domain (see Kuhn 1970, 1977; Laudan 1977, p. 120).

literature in the social sciences reveals that many of the established standards of scientific theory are often neglected by theorists (for detailed discussions, see Black 1995; Jasso 1988; Lenski 1988; Mayhew 1981); therefore it is necessary to make explicit the standards on which theoretical formulations will be evaluated. Admittedly, scientists often differ on which criteria are most important in theory construction, however these disagreements tend to be trivial (see Braithwaite 1953; Nagel 1979). In fact, certain standards are nearly always emphasized and the numerous specific criteria articulated by some theorists have been shown to be derivative of more general standards (Nagel 1979, p. 29). Black (1995, pp. 831–45) and others (e.g., Akers 2000, pp. 6–13; Braithwaite 1953; McKinney 1966, pp. 18, 62–64; Tittle 1995, pp. 17–53) identify four major criteria on which scientific theory is commonly judged: (1) generality; (2) parsimony; (3) testability; and (4) empirical validity.<sup>231</sup>

<sup>&</sup>lt;sup>231</sup> Four additional criteria commonly articulated by scientists are *objectivity, logical coherence, clarity*, and *originality* (see Black 1995; Fuchs 1992; Mayhew 1981; Popper 1964). The objectivity of a theoretical formulation is its capacity to be "value-free" and unbiased (Nagel 1979, p. 485). The clarity of a concept or theory can be simply defined as its capacity to be understood. According to Mayhew (1981, p. 629), lack of clarity often boils down to a refusal (or failure) to discuss certain critical questions that must be answered in order for the theory to be intelligible. Logical coherence refers to both a logical consistency between parts of the theory (i.e., no internal contradictions) and the absence of spurious reasoning (Jasso 1988, p. 3). Objectivity, clarity, and logical coherence, however, can be subsumed under one or more of the four criteria mentioned above. For example, if objectivity is an observable characteristic of an idea, then it is an element of validity (Black 2000a, p. 351 n.23). Similarly, if a theory is unclear and logically incoherent, it cannot be parsimonious and testable. In fact, these problems (e.g., lack of clarity, lack of parsimony, logical incoherence) have a strong tendency to occur together (Mayhew 1981, pp. 629–30).

Originality simply refers to the novelty or innovativeness of an idea. "The importance of being original is so fundamental [to scientific theory] that it may be taken for granted and not even mentioned when scientific ideals and standards are listed. Yet science is obsessed with newness" (Black 1995, p. 846). It is widely acknowledged that the most celebrated scientists in history offered completely original insights to problems under investigation (e.g., Copernicus's sun-centered astronomy, Newton's theory of gravity, Einstein's general theory of relativity) (see Kuhn 1970). When original ideas first appear, however, they are often unpopular to the larger scientific community and may even be professionally damaging (see, generally, Barber 1961; Kuhn 1970). In fact, it has even been argued that the more original (i.e., unconventional) an idea, the more hostility and controversy it will attract (see Black 1995, 2000a). Opposition notwithstanding, novel approaches to the study of reality (whether physical, biological, psychological, or social) are what significantly advance science (Bronowski 1958). The originality of a theory or concept, however, becomes important only if it is first general, parsimonious, testable, and valid.

*Generality.* Many scientists and philosophers of science have argued that perhaps the most important aspect of scientific theory is its generality (see Braithwaite 1953; Harris 1979; Hempel 1965; Hempel and Oppenheimer 1948; Lenski 1988; Mayhew 1980, 1981; Nagel 1979). Generality (also referred to as "scope") is commonly understood as the capacity to explain the widest range of phenomena possible under a single explanatory framework or structure (Akers 2000, p. 6). Furthermore, propositions derived from a general theory are not restricted to limited regions of space and time (Braithwaite 1953, Chapter 9). It has even been argued that generality is necessary for science to progress (Nagel 1979, p. ix) and that generality is the "supreme virtue" of scientific theory (Lenski 1988, p. 166). Still others have maintained that generality is a "contemptuous attitude towards a particular case" and that science "craves" generality (Black 1995, p. 833).

Admittedly, social scientists in general—and sociologists in particular—are far from consensus on the importance and appropriateness of establishing "timeless" and "spaceless" theory (for discussions, see Black 2000c; Homans 1964; Lenski 1988; Somers 1998). In fact, many social scientists continue to argue that social processes are culturally and historically contingent, therefore prohibiting such general explanations (e.g., Frankford 1995; Tamanaha 1997). However, following Nagel (1979, pp. 506–507) and others (see Black 1979b; Mayhew 1981), it is argued here that the difficultly in establishing highly general social theory is not inherent in the study of social phenomena (i.e., social phenomena are not too complex to fully comprehend); rather the difficulty arises because social scientists have largely failed to establish more discriminating classifications of social phenomena that are of a transcultural nature. As a result, most theoretical formulations in the social sciences employ familiar "common sense"

distinctions and possess a very limited scope of valid application (Mayhew 1980, p. 352;

Nagel 1979, pp. 459, 503).

The fact that social processes vary with their institutional settings, and that the specific uniformities found to hold in one culture are not pervasive in all societies, does not preclude the possibility that these specific uniformities are specializations of relational structures invariant for all cultures...the "historically conditioned" character of social phenomena is no inherent obstacle to the formulation of comprehensive transcultural laws (Nagel 1979, pp. 462–64).

In the case of human associations, which are the most unlike imaginable in purposes and in total meaning, we find nevertheless similar formal relationships between individuals. Superiority and subordination, competition, imitation, division of labor, party structure, representation, inclusiveness towards members and at the same time exclusiveness towards non-members, and countless similar variations are found, whether in civic group or in a religious community, in a band of conspirators or an industrial organization, in an art school or in a family. However diverse, moreover, the interests may be from which the socializations arise, the forms in which they maintain their existence may nevertheless be similar (Simmel 1909, p. 299).<sup>232</sup>

Certainly, problems associated with narrow classifications of phenomena under

consideration are not particular to the social sciences. In fact, Braithwaite (1953) notes

that, in their early stages of development, all sciences were characterized by common

sense distinctions and low-level generalizations. As these sciences advanced, however,

more general (i.e., abstract) formulations were developed and narrow classifications

previously considered were either greatly expanded or discarded altogether:

As the hierarchy of hypotheses of increasing generality rises, the concepts with which the hypotheses are concerned cease to be properties of things

<sup>&</sup>lt;sup>232</sup> Simmel ([1908] 1950, pp. 21–22) initially conceived sociology as the science of social forms—"social geometry"—and suggested that sociologists should leave the examination of the content of social interaction to other social sciences (e.g., economics and psychology) much in the same way that geometry leaves the analysis of content to other physical sciences: "Geometric abstraction investigates only the spatial forms of bodies, although empirically these forms are given merely as the forms of some material content. Similarly, if society is conceived as interaction among individuals, the description of the forms of this interaction is the task of the science of society in its strictest and most essential sense."

[that] are directly observable, and instead become theoretical concepts (e.g., atoms, electrons), which are connected to the observable facts by complicated logical relationships (p.ix).

In fact, the very purpose of moving from the particular to the general is to improve our understanding of both. The specific entities of the social world—or, more precisely, specific facts about these entities—provide the basis on which generalizations must rest. In addition, we almost always learn more about a specific case by studying more general conclusions (King, Keohane, and Verba 1994, p. 35).

Highly general theory, then, achieves a greater level of *scienticity*<sup>233</sup> than theoretical

formulations that are culturally and historically contingent because it orders, predicts, and

explains a much wider range of phenomena under investigation. Highly general theory

also allows scientists to identify and understand relationships previously ignored or

believed to be completely incomprehensible (Bronowski 1958; Mills 1959).

Parsimony. Akers (2000, p. 7) defines parsimony as "the conciseness and

abstractness of a set of concepts and propositions.<sup>234</sup> The principle of parsimony, then,

is to use as few propositions as possible to explain the widest range of phenomena.<sup>235</sup> In

science, a simple (i.e., parsimonious) theoretical formulation that explains as much as a

complex one is always preferable.<sup>236</sup>

Since all models are wrong [i.e., all models are approximations], the scientist cannot obtain a "correct" one by excessive elaboration. On the contrary, following William of Occam he should seek an economical description of natural phenomena. Just as the ability to devise simple but evocative models is the signature of the great scientist, so overelaboration... is often the mark of mediocrity (Box 1976, p. 792).

The goal is to develop a theory that is at once simple and fruitful, a theory, that is, with a minimum of postulates and a maximum of predictions, the

<sup>&</sup>lt;sup>233</sup> According to Black (2000a, p. 351), science is a matter of degree. Scienticity is the capacity of a theory to realize scientific standards of generality, parsimony, testability, validity, and originality.

<sup>&</sup>lt;sup>234</sup> Others have remarked that "abstraction is the essence of scientific inquiry...." (Posner 2003, p. 17).

<sup>&</sup>lt;sup>235</sup> Nietzsche ([1888] 1998, p. 75) remarked, "It is my ambition to say to in ten sentences what other men say in whole books—what other men do *not* say in whole books." <sup>236</sup> Black (1995, p. 838) argues that science "loves simplicity and despises complexity."

latter including predictions for phenomena or relationships not yet observed (Jasso 1988, p. 1).

This is not to say, of course, that social life is not complex. Nor is it argued that the development of a parsimonious (and highly general) theory of social phenomena is an easy task. It is widely believed, however, that the ultimate purpose of science "is to simplify reality and find underlying patterns where reality first appears more complicated...[t]he more concisely such patterns are formulated, the more the goal of parsimony is realized" (Black 1995, p. 838).<sup>237</sup> The complexity of social life, then, does not require that theoretical propositions about social life be exceedingly complex or that parsimonious explanations of social phenomena are forever doomed (but cf. Sarat 1989; Tamanaha 1997). In fact, as the complexity of a theory increases, it becomes increasingly unclear.<sup>238</sup> Theories that rest on numerous assumptions and qualifications seldom order, predict, and explain social phenomena with reasonable accuracy—they are inefficient (Nagel 1979, p. 449).<sup>239</sup> Increasing complexity also invites divergent interpretations and increases the likelihood that the propositions will be misunderstood. If propositions are misunderstood, their testability—and ultimate validity—is also questionable. Stinchcombe (1968, p. 6) remarked, "Social theorists should prefer to be wrong rather than misunderstood. Being misunderstood shows sloppy theoretical work." Nagel (1979) also dismisses arguments that social phenomena are too complex to fully comprehend as misguided:

<sup>&</sup>lt;sup>237</sup> Weber ([1904] 1949, p. 58) remarked that the duty of sociology was to "analytically order[] empirical reality."

<sup>&</sup>lt;sup>238</sup> Edling (2002, p. 203) writes, "[S]ociological models should be precise, or else they cannot serve as hypothesis generators and consistency checkers in any substantial way; and those are two important functions for theoretical models."

<sup>&</sup>lt;sup>239</sup> According to Posner (2003, p. 17), "A theory that sought faithfully to reproduce the complexity of the empirical world in its assumptions would not be a theory—an explanation—but a description."

The complexity of a subject matter is at best not a precise notion, and problems that appear to be hopelessly complex before effective ways for dealing with them are invented often lose this appearance after the inventions have been made...though social phenomena may indeed be complex, it is by no means certain that they are in general more complex than physical and biological phenomena for which strictly universal laws have been established. Moreover, while it is true that responses to a given social situation are mediated by the variable interpretations [individuals] place upon it, this fact by itself does not explain why there are no universal laws relating each of the several interpretations placed upon a given type of social stimulus to a particular form of human response (p. 505).

Similarly, Mayhew (1981) notes:

Parsimony is critical to the development of any theory about social phenomena. A theory that is too complex to be understood (or too complex to permit the implementation of research on critical questions) is no more than an appeal to incomprehensibility, a claim that what is being studied cannot be understood (p. 629).

Testability. It is widely recognized in the scientific community that scientific

theory must be falsifiable (see Popper [1934] 1968). In order for a theory to be

falsifiable, it must satisfy two conditions. First, it must be predictive (Jasso 1988). As

Black (1995, p. 831) notes, "a prediction need not prophesy the future of anything, but is

simply an empirical pattern—something observable—logically implied by the theory."<sup>240</sup>

If prediction is not possible the theory cannot be proven false; it cannot be tested.

Second, it must be stated in quantitative language so that predictions can be evaluated by

measurement (see Braithwaite 1953; Zetterberg 1966). This measurement, however,

does not require a determination of precise differences (interval or ratio measurement);

rather it need only be a determination of whether more or less of something occurs

(ordinal measurement) or merely whether something occurs at all (nominal measurement)

(Black 1995, p. 831 n.14). Indeed, if science is the study of variation (i.e., change) in

<sup>&</sup>lt;sup>240</sup> McKinney (1966, p. 47) argues that scientific explanation consists of showing that a given event "had to be expected in view of the presence of certain factors prior and/or contemporaneous with it."

reality, as most scientists would suggest, a scientific theory that makes clear predictions about the quantity (or relative quantity) of a particular phenomenon occurring under specific conditions is superior to one that makes claims which are not subject to measurement (Akers 2000).<sup>241</sup> According to Nagel (1979, p. 447), the testability of social scientific theory is what distinguishes it from moral and social philosophy.<sup>242</sup>

*Empirical Validity.*<sup>243</sup> The validity of a theory is its truthfulness or accuracy; that

is, how well the theory is supported by empirical research evidence (Akers 2000).<sup>244</sup>

Although theories will inevitably encounter negative or disconfirming evidence, this does

not warrant the theory being judged wholly invalid (Harris 1979). It is only after

continued rigorous empirical testing that a theory's validity can truly be known (Popper

[1934] 1968). Through repeated testing, weaknesses in the theory are identified, often

resulting in the theory being modified or discarded. Most scientists agree, however, that

<sup>&</sup>lt;sup>241</sup> Lieberson and Lynn (2002) vigorously challenge the importance of prediction in sociological theory. They argue that predictability is a standard adopted from classical physics that is impractical in the social sciences and many of the natural sciences: "[Sociologists] are again trying to do something that [sociologists] have no business expecting to be able to do, at least in a world of complex influences that are not restrained as they would be in a true experiment. This again leads to unrealistic goals and forced efforts to have tests of predictability that are not appropriate...because they implicitly use a standard from classical physics rather than a more realistic standard of how it works in the wider set of sciences" (p. 10). Unfortunately, Lieberson and Lynn fail to offer a viable alternative for distinguishing between rival theories (or theoretical programs). By viewing predictability in relative rather than absolute terms, the predicative power of the theory remains a useful way to evaluate competing explanations of social phenomena. Moreover, the focus on predictability facilitates careful theory construction—attention to the development of clear predictions often improves the parsimoniousness and logical consistency (i.e., inherent coherence) of a theory.

<sup>&</sup>lt;sup>242</sup> Black (2000a, p. 360 n.43) suggests that much of the work of classical (e.g., Durkheim, Simmel, and Weber) and modern (e.g., Collins, Luhmann, and Parsons) sociologists is largely untestable (see also Lenski 1988, pp. 165–66). Cooney (1993, p. 2227) posits that "no flaw is more fatal to a scientific theory than lack of testability."

<sup>&</sup>lt;sup>243</sup> To be sure, empirical validity is but one of many different types of validity considered by scientists. Others include face validity, content validity, concurrent validity, convergent validity, and discriminant validity (see, e.g., Skrondal and Rabe-Hesketh 2004, p. 8), but these forms of validity are generally relevant to the measurement of theoretical concepts, not to the overall accuracy of theories. Thus when scientists speak of validity of a theory, they are usually referring to empirical validity.
<sup>244</sup> The validity of a scientific theory also increases with the precision of its predictions. "The validity of a

<sup>&</sup>lt;sup>244</sup> The validity of a scientific theory also increases with the precision of its predictions. "The validity of a theory is measurable with its precision: the degree to which the frequency and magnitude of its explanatory variable(s) matches the frequency and magnitude of the variable it seeks to explain. The highest validity is total precision" (Black 2000a, p. 351 n.26).

it is better for a theory to be false than it is for a theory to be untestable (see Black 2000a, p. 351 n.24). An incorrect theory, at minimum, advances science through alerting the researcher (and scientific community) what reality is not—it eliminates something (Black 1995, p. 841). An unfalsifiable theory, on the other hand, cannot be as judged right or wrong and, therefore, it cannot advance the understanding of the phenomenon under investigation.<sup>245</sup>

The four criteria described above—generality, parsimony, testability, and validity—constitute an "epistemological checklist" with which scientific theory should evaluated (see Black 1995, p. 847).<sup>246</sup> These criteria guide the critique of the theoretical perspectives presented below.

# 6.1.2 Description of Dominant Perspectives

Research examining the relationship between extra-legal factors and criminal sentencing that has been theoretically informed generally employs—or is believed to support—one of three theoretical perspectives: (1) formal legal; (2) conflict; and (3) interactionism (see Dixon 1995, pp. 1159–62; Pratt 1998, pp. 513–15; Urbina 2003, pp. 9–28).<sup>247</sup>

<sup>&</sup>lt;sup>245</sup> According to Francis Bacon (1875, p. 210), "Truth more readily emerges from error than from confusion."
<sup>246</sup> Epistemology is the philosophy of knowledge, including its nature and evaluation (see Black 1995, p.

<sup>&</sup>lt;sup>240</sup> Epistemology is the philosophy of knowledge, including its nature and evaluation (see Black 1995, p. 829 n.3; Nagel 1979, p. 30).

<sup>&</sup>lt;sup>247</sup> The *organizational maintenance* perspective, rooted heavily in the work of Michels ([1911] 1962), has also been used to explain sentencing outcomes. Organizational maintenance theorists posit that researchers must focus on the operational goals of legal officials, rather than the political (i.e., conflict), symbolic, or formal legal goals of sentencing. According to this perspective, the operational goal of courtroom elites (e.g., prosecutors and judges) is to maintain a stable and orderly sentencing system through the efficient processing and disposing of cases (Dixon 1995, p. 1162; Johnson 2006a, pp. 264–66; Kautt 2002, p. 642). Organizational maintenance theorists posit that an efficient sentencing system is maintained through the heavy use of plea-bargaining (see Baker and Mezitti 2001). (The U.S. Supreme Court specifically approved plea-bargaining as a means of managing overloaded criminal dockets, see *Santobello v. New York*, 404 U.S. 257, 260–61 [1971].)

Formal Legal Theory. The formal legal perspective conceives law,

fundamentally, as an affair of rules and logic.<sup>248</sup> According to this view, the explanation of legal behavior lies in the rules by which established facts are assessed (see Savelsberg 1992; Weber [1925] 1954, [1922] 1968). In the formal legal model, the social characteristics of actors in a particular case have no relevance unless these characteristics are mentioned in the rules themselves. The model also assumes that law is constant from one case to another, so it becomes possible to predict or anticipate the outcomes of cases based upon the assessment of facts (Trevino 1996, Chapter 5).<sup>249</sup>

Formal legal theorists, for example, acknowledge that members of certain

minority groups, on average, receive harsher sentences than whites, but suggest that this

Perhaps the strongest criticism of organization maintenance theory is that it is tautological (for a discussion of tautological explanations in the social sciences, see Akers 2000). Legal rules, case processing, and sentencing outcomes are all examples of legal behavior (see Black 1976). Therefore, not only do organizational maintenance theorists *explicitly* attempt to explain one form of legal behavior (case dispositions) with another form of legal behavior (case processing), they implicitly incorporate another form of legal behavior (i.e., formal legal rules) into their theory. Legal processing, then, becomes part of what is to be explained and is incapable of providing an adequate explanation of legal behavior, sentencing or otherwise (Black 1972; Horwitz 1983; Timasheff 1937).

Due to the aforementioned limitations of organizational maintenance theory, criminologists and sociolegal theorists tend to focus their attention on formal legal, conflict, and interactionists perspectives. <sup>248</sup> The formal legal model is also known as the *differential involvement* model (Pratt 1998, p. 513) and the *jurisprudential model* (Black 1989, pp. 19–22). <sup>249</sup> A discussion of the virtues and vices of rule-based legal reasoning is beyond the scope of this project;

see Scalia (1997) and Sunstein (1996) for contrasting arguments.

When attempting to explain racial disparities in sentencing, it is questionable, however, whether organizational maintenance theory offers a novel or sufficiently different theoretical perspective from formal legal and conflict perspectives. First, organizational maintenance theory cannot, by itself, explain racial disparities in criminal sentencing, so it is usually coupled with conflict perspectives (Chambliss and Seidman 1971). Organizational maintenance theorists argue that social elites are processed in ways that reduce their sentences, mainly through the use of plea bargains. Minority and powerless groups, on the other hand, are processed in ways that fail to reduce their sentences. From this perspective, processing and extra-legal variables (e.g., race/ethnicity) interact to affect sentencing (Dixon 1995). Second, legal rules still play a central role in organizational maintenance theory, although their role tends to be understated by organizational maintenance theorists. In fact, many legal statutes mandate certain procedures for specific crimes. For example, in Georgia, prosecutors must file a notice of intent to seek the death penalty prior to a grand jury indictment (Mears 1999). In North Carolina, any murder case in which one or more of several statutorily defined elements are present must become a death penalty case (Unah and Boger 2001). Organizational maintenance theorists also underestimate the role and importance of formal rules that are designed to streamline the legal process and make sentencing more efficient. In recent years, the U.S. Supreme Court and U.S. Congress, along with state supreme courts and state legislatures, have limited the scope of appeals and shortened the appellate process (see Haines 1996; Hertz and Liebman 2005; Williams 2000). The distinction between formal legal goals and operational goals, then, is dubious at best.

results from the fact that members of these minority groups are involved in more serious crimes and have more extensive criminal histories (see, generally, Wilbanks 1987; Wilson and Herrnstein 1985). According to this view, race/ethnicity and other extra-legal factors play little or no role in the sentencing process once legally relevant characteristics, such as offense severity and the offender's criminal history, are held constant (see Hagan 1974; Kleck 1981). Although supporters of this perspective tend to acknowledge that racial/ethnic bias in criminal sentencing was evident (and even widespread) in the past, even after racial/ethnic considerations were formally removed from legal codes, they hold that racial/ethnic neutrality now exists due to increased formalization and bureaucratization of the criminal judicial process (Pratt 1998, p. 515). Studies that continue to detect significant racial/ethnic differences are criticized by these theorists as being methodological flawed (Hagan 1974; Kleck 1969, 1981) and for incorrectly attributing such differences to racial/ethnic prejudice rather than other difficult to measure factors that are likely to be correlated with race/ethnicity (e.g., quality of legal representation) (Abrams and Yoon 2007; Schanzenbach 2005b).

*Conflict Theory.* The *conflict* perspective, largely rooted in the neo-Marxist tradition (e.g., Chambliss and Seidman 1971; Quinney 1970; Sellin 1938; Turk 1969), argues racial/ethnic, gender, and class discrimination in criminal sentencing is ubiquitous and results from power differentials between groups in society—differentials that are embedded in the institutional organization of society.<sup>250</sup> According to this view, racial/ethnic minorities, women, and the poor are members of politically and economically deprived groups who suffer from institutionalized discriminatory social

<sup>&</sup>lt;sup>250</sup> Pratt (1998) refers to the conflict model as the *direct-impact* model; Dixon (1995, p. 1160) refers to it as the *substantive political* model.

practices, particularly in the criminal justice system (see, e.g., Chambliss 1999; Du Bois [1940] 1968). Law, then, is either manipulated directly by powerful groups to maintain their advantage or it simply mirrors the basic tenets of the market economy (Hunt 1993; Savelsberg 1994; Spitzer 1983; Trevino 1996, Chapter 4; Turk 1969). Conflict theorists, for example, maintain that racial/ethnic minorities receive harsher sentences than whites even after important legal characteristics (e.g., prior record, offense severity) are held constant because law-making and law-finding differentially (and deleteriously) impacts members of politically and economically disadvantaged groups (see, e.g., LaFree 1989; Petersilia 1985; Zatz 1987). Revisions of conflict theory have identified certain mediating *and* moderating factors that impact the relationship between race/ethnicity and criminal punishment (see Hawkins 1987). These recent modifications of conflict theory, however, significantly blur the distinction between conflict and interactionist perspectives and, perhaps, are more accurately classified as integrated theories.<sup>251</sup>

*Interactionism.* The *interactionist model* emphasizes the social context in which criminal sentencing takes place and posits that the meaning and impact of legal and extra-legal case characteristics vary according to the setting in which such sentencing occurs (Tamanaha 1997, pp. 143–52). According to this perspective, which is rooted in symbolic interactionism (see Blumer 1969; Goffman 1963; Mead 1934; Simmel [1908] 1950; Thomas and Znaniecki [1921] 1958), the meaning and significance of legal and extra-legal factors are not abstract, fixed and unchanging; rather they are open to continual negotiation, reinterpretation, and recreation depending on the actors' goals and

<sup>&</sup>lt;sup>251</sup> In particular, Hawkins argues restatements of conflict theory were merely oversimplifications of its original formulation. Contrary to these previous oversimplifications, he emphasizes the importance that earlier conflict theorists placed on: (1) victim characteristics, (2) "race-appropriate" and "race-inappropriate" crime, (3) the differential mechanisms of "power-threat" and "subordination," and (4) region.

the setting in which interaction occurs (Brittan 1981; Emmelman 1994). For example, interactionists believe that the meaning and salience of race/ethnicity in a criminal sentencing hearing is likely to vary according to observable aspects of the case, such as the offender's prior criminal record (Thomson and Zingraff 1981; Zatz 1984), offense severity (Kempf and Austin 1986), prosecutorial decision-making (Keil and Vito 1989), the degree of urbanization and political climate of the jurisdiction (Hagan and Zatz 1985), and the structure of sentencing guidelines (Miethe and Moore 1986). The influence of race/ethnicity will also vary according to subjective, largely unobservable factors such as preexisting individual and group attitudes about members of a particular minority group (e.g., prejudices and stereotypes) and newly formed attitudes and definitions about racial/ethnic minorities that emerge in each decision-making process (Blumer 1969). Consequentially, an offender's racial/ethnic background will not uniformly disadvantage her or him when being sentenced; rather the specific characteristics of each particular case, the interpretation of these characteristics by decision-makers in the case, and the specific goals of the decision-makers determine how race/ethnicity operates in that situation.<sup>252</sup>

# 6.1.3 Evaluation of Dominant Perspectives

*Generality*. The first criterion on which these perspectives are evaluated is the generality of the proposed explanation. Conflict theory, considered one of the "grand" theories of society (Collins 1975, 2002), is much broader in scope than both formal legal

<sup>&</sup>lt;sup>252</sup> Some theorists have integrated elements from one or more of these three dominant perspectives. For example, both Albonetti (1991) and Steffensmeier and colleagues (Steffensmeier and Demuth 2000, 2001; Steffensmeier, Kramer, and Streifel 1993; Steffensmeier, Kramer, and Ulmer 1998) combine aspects of symbolic interactionism and formal legal theory in causal attribution and focal concerns theories, respectively.

and interactionist perspectives. Conflict theorists often conceptualize race/ethnicity, gender, and social class as proxies for economic and political power. They argue, for example, that power differentials between racial/ethnic groups explain racial/ethnic disparities in criminal processing at all stages of the criminal justice system (i.e., from law-making through sentencing) (see Turk 1969). Furthermore, they posit that these economic and power differentials explain racial/ethnic disparities in law-making, law-breaking, and the response to law-breaking throughout history and across different societies (Chambliss and Seidman 1971; Reiman 1979). Although the broad scope of the conflict perspective is largely regarded as one of its greatest strengths, its generality has been questioned because of earlier conflict theorists' nearly exclusive focus on class division and inequality resulting from capitalist economic systems (Akers 2000; Black 1993). Conflict theorists have been criticized for being unable to explain the existence of inequality in criminal justice systems in socialist and other societies that are not economically and politically stratified (or only have limited stratification in these areas).

Interactionists maintain that the essential meaning of race/ethnicity and its impact on criminal sentencing may change from situation to situation (cf. Bobo 1999). Furthermore, these theorists argue that extra-legal factors such as race/ethnicity are important primarily, if not exclusively, through their association with other legal and extra-legal factors. Interactionists, however, are largely silent as to how and when preexisting roles, norms, or ideologies constrain legal behavior. For example, although interactionists acknowledge that social actors are not always equal when they confront one another in the legal process (Brittan 1981, pp. 179–80), they offer little guidance as to how power and status differentials may systematically influence the legal process (but cf. Black and Baumgartner 1983; Cooney 1994). Because interactionists posit that concepts have to be continually redefined and reevaluated in every instance of interaction, their theory is criticized for not only "dispens[ing] with generalization" (Perdue 1986, p. 255), but also for offering virtually no simplification of reality (Black 1989; but cf. Tamanaha 1997).<sup>253</sup>

The cross-cultural applicability of interactionism is also limited because interactionists neglect explicitly addressing important differences in judicial traditions and legal cultures or haphazardly use Western interpretive frameworks to explain legal behavior in non-Western nations. Perhaps not surprisingly, interactionists have been primarily concerned with explaining legal processes and outcomes in legal systems based on English common law, most notably in the United States, the United Kingdom, Ireland, Australia, and Canada (see, e.g., Unnever and Hembroff 1988).<sup>254</sup> Systems based on English common law assign a preeminent position to case law (i.e., judge-made law) and emphasize an adversarial process in the courtroom. Legal systems based on common law, however, constitute a small proportion of the legal cultures of the world (David and Brierly 1985). For example, the legal systems of continental Europe, Central and South America, and Asia are based on the civil law tradition, which gives precedence to written law and emphasizes an inquisitorial process in the courts, assigning the court a pivotal role in the pre-trial preparation of evidence by police, the presentation of evidence at trial, and the examination of witnesses (see Royal Commission on Criminal Justice 1993). The majority of nations in Africa and the Middle East have legal systems that combine

<sup>&</sup>lt;sup>253</sup> Some legal sociologists have argued that the discipline will be unable to construct any useful substantive theory unless it "ceases to be preoccupied with the 'legal profession' and the behavior of jurors in particular social units" (Gibbs 1968, p. 446; see also Black 1997).

<sup>&</sup>lt;sup>254</sup> The state of Louisiana and the Canadian province of Quebec have legal systems that combine both common law and civil law.

elements of common or civil law with elements of customary or Islamic law, which are largely rooted in spiritual, philosophical, or religious traditions (David and Brierly 1985). These different legal cultures vary in the manner in which they conceptualize and structure judicial discretion and legal innovation (see Rosen 1989). As a result, the communicative process in legal proceedings, a central feature of the interactionist perspective, may be of limited utility in explaining legal behavior in certain legal cultures.

Interactionism may have limited cross-temporal applicability as well. The explanatory power of the interactionist model is likely to be dependent on the quantity and diversity of ideas found in any given society—the amount of culture.<sup>255</sup> Interactionists assume there exists a sufficient amount of different ideas in a social setting from which individuals can infer meaning about a social object. While this assumption is likely to hold in modern societies that have numerous subcultures and much individuality, it is problematic for the study of earlier societies. Many early societies, for example, had very little culture (in terms of quantity, not quality)—they had one language, one religion, one way of dressing, on way of preparing food, and so forth (Black 1976); therefore it is unlikely that interactionism could adequately account for variation in legal behavior in these early societies because of the sparseness and homogeneity of ideas in those contexts.

Formal legal theorists posit that the characteristics of a case outlined in preestablished rules primarily determine legal decision-making (but see Levi 1949). While acknowledging that certain extra-legal factors inappropriately influenced the legal

<sup>&</sup>lt;sup>255</sup> According to Black (1976, p. 61), "culture has an existence of its own, apart from the way people experience it;" therefore it is possible to measure the amount of culture in any given setting.

process in the past, formal legal theorists contend that legal decision-making has been formalized to the point where rules have become the only important predictors of legal outcomes (e.g., sentencing guidelines). The generality of this approach, however, has also been seriously questioned. First, by conceding that legal rules were ineffective at formalizing legal behavior in the past, formal legal theorists limit the generality of their approach because they must restrict the importance and function of legal rules to the modern period. Formal legal theory's generality is further restricted by its apparent logical inconsistency (i.e., internal contradiction): it suggests that rules were of limited importance in the past, but are of greater importance in the present because more restrictive rules have been applied to current legal behavior (see also Kan and Phillips 2003).

*Parsimony.* Black (2000a, p. 838), quoting Gell-Mann (1994, p. 28), notes, "It is not simple to define 'simple.'" Most scientists agree, however, that when comparing competing theoretical explanations, the theory that uses the fewest number of propositions and rests on the fewest assumptions to explain the widest range of phenomena is considered the most parsimonious (Friedman 1953; Mayhew 1981).

Formal legal theorists attempt to explain legal behavior by legal rules themselves. That is, they explain legal behavior by the application of legal rules to the facts in a particular case and attribute racial/ethnic, age, gender, and social class differences in sentencing to the differential involvement of members of these groups in serious crimes and the prevalence of extensive (or serious) criminal histories among certain members of these groups. While formal legal theory appears to offer a rather parsimonious explanation of sentencing differences, it rests on several assumptions concerning the creation and application of the law. Namely, formal legal theorists assume that (1) legal systems are becoming increasingly characterized by *formal rationality*,<sup>256</sup> emphasizing the universal application of established rules, rather than *substantive rationality* which is purposive and primarily concerned with the ends served by legal decisions (Engen *et al.* 2003; Savelsberg 1992; Tomlins 2007; Weber [1925] 1954); (2) laws, except in rare instances, are developed *independent* of considerations of extra-legal factors (see Chambliss and Seidman 1971); (3) laws can be directly applied to facts of a case independent of evaluations that, either explicitly or implicitly, rely on extra-legal factors (Black 1979a, 1989); and (4) laws are not—or cannot be—applied in a manner that is technically "legal" but still discriminatory (see Black 2002a, pp. 111–13; Bushway and Piehl 2001).<sup>257</sup> Some socio-legal theorists have found these assumptions extremely problematic (see Black 1989; Savelsberg 1992). These assumptions, then, seriously

<sup>&</sup>lt;sup>256</sup> Ewing (1987) identifies two dimensions of formal rationality: (1) *logically formal rationality* and (2) *sociologically formal rationality*. Logically formal rationality refers to a logical and gapless system of legal rules. Sociologically formal rationality, on the other hand, refers to the notion of uniformity and equal treatment under the law. Sentencing guidelines, for example, reflect both of these dimensions of rationality.

<sup>&</sup>lt;sup>257</sup> To be sure, all laws "discriminate"—that is, they focus on specific behaviors (and groups). In this context, however, researchers mean "socially undesirable" or "legally dubious" discrimination. For example, Bushway and Piehl (2001) discovered that judges make downward departures (from sentencing guidelines) more often in white-defendant cases than in black-defendant cases. Similarly, judges make upward departures more often in black-defendant cases than in white-defendant cases. Therefore, although these judges' actions are "legally permissible," they appear to be discriminatory because the sentencing guidelines already take into account the severity of the crime and the criminal history of the defendant (see also Johnson 2005; Mustard 2001). Also recall that all death penalty statutes require capital jurors to weigh statutory aggravating and mitigating evidence during the penalty phase (see Chapter Two). Interviews with former capital jurors reveal that white jurors are less receptive to mitigation evidence when the defendant is black (Bowers and Foglia 2003; Bowers et al. 2001), thereby allowing white jurors to "legally discriminate" against black defendants through the exercise of discretion at the sentencing stage (and perhaps the conviction stage as well). This neither implies that judges and jurors consciously act in a discriminatory manner (Johnson 1988) nor that discretion in the legal process will inevitably lead to discrimination (Baldus et al. 1994b); however it should be noted that discretion is desirable only to those who benefit from it and the robust empirical associations between race/ethnicity, upward/downward departures, and juror discretion at the penalty phase of a capital trial seriously bring into question the "appropriateness" of the observed sentence enhancements or reductions (for non-capital cases) and the death penalty.

undermine the ability of formal legal theorists to offer a parsimonious explanation of legal behavior.<sup>258</sup>

Both conflict and interactionist approaches also rest on several questionable assumptions concerning the nature of human beings and society that potentially undermine their degree of parsimony (see, generally, Black 2000a, pp. 345–48; Popper 1964). Conflict theorists attribute group differences in criminal justice processing to the domination and oppression of politically and economically disadvantaged groups by powerful elites. Underlying this perspective are the assumptions that (1) social groups have a history of shared identity and shared fate, (2) these groups believe themselves to be in zero-sum competition over valuable (and scarce) resources, and (3) social order is primarily maintained through coercion and oppression (see Collins 1975; Perdue 1986, pp. 303–305). Critics argue that these assumptions, while being sufficient to produce discrimination, are unnecessary (Sidanius and Pratto 1999, p. 17), unobservable, and unknowable (see Black 2000a, p. 346). In fact, conflict theorists' characterization of society as being rife with disagreement has been strongly challenged by social theorists for some time, particularly as it pertains to the creation and application of criminal laws (see Akers 2000; Durkheim [1893] 1947).

Interactionists posit that the impact of extra-legal factors on criminal sentencing is not fixed, but rather it is conditioned by the social context in which sentencing occurs and by the goals and motivations of the actors in that particular setting. Interactionists assume that (1) human beings act towards things on the basis of the meaning that things

<sup>&</sup>lt;sup>258</sup> Horwitz (1983, p. 372) extends the criticism of the formal legal theory even further, positing "whether legal rules predict [legal] behavior is an empirical rather than a conceptual question." According to Horwitz and others (e.g., Black 1972; Timasheff 1937), rules are to be taken as an object to be explained and not, themselves, what explains law.

have for them; (2) these meanings are a product of social interaction in human society; (3) these meanings are modified and handled through an interpretive process that is used by each individual in dealing with the signs he encounters; and (4) communication is the basis for the formation and maintenance of social order (see Blumer 1969; Cooley [1902] 1964; Mead 1934). Interactionists' assumptions and assertions about the human mind and the conscious and unconscious meanings and feelings individuals experience seriously undermine their ability to offer a parsimonious account of influence of race/ethnicity on criminal sentencing. Interactionists, largely, remain unclear as to how—and under what conditions—the various factors present in a situation influence the meanings actors assign to social objects and how these meanings, in turn, influence sentencing outcomes. According to Brittan (1981, p. 167), "it is impossible to define symbolic interactionism with any degree of precision. One thing we must be clear about is that we are not talking about a formal theoretical scheme." This may partly result from the fact that earlier interactionists (e.g., Mead 1934) were less concerned with specifying behavioral outcomes resulting from specific forms of social interaction, as they were with explaining the processes by which mind, self, and society are constructed (Perdue 1986, p. 239; but cf. Simmel [1908] 1950). While the illumination of these processes by interactionists is extremely important, ambiguities in the theory prevent the articulation of clear behavioral implications and, ultimately, seriously limit the theory's ability to offer a clear and parsimonious explanation of legal behavior.

*Testability.* As mentioned earlier, a testable theory must be both predictive and measurable. A theory's level of testability is also related to its parsimony (see Mayhew 1981). If a theory is exceedingly complex, resting on numerous caveats and qualifiers,

testing it becomes extremely difficult. Furthermore, if it makes contradictory predictions (i.e., logically inconsistent), all data can be interpreted as supportive. Both formal legal and conflict perspectives enjoy a higher degree of testability than interactionist theory, although neither perspective is without significant problems. Formal legal theorists postulate that, holding relevant legal factors constant, extra-legal factors such as race/ethnicity, age, and gender are not predictive of sentencing outcomes. In testing this perspective, researchers compare offenders with different social characteristics (e.g., members belonging to different racial/ethnic groups) who are similar with respect to their prior criminal records and offense severity and examine whether these offenders are treated differently. While this approach seems fairly straightforward, two major difficulties arise. First, as previously mentioned, the measurement of many legally relevant variables requires a subjective interpretation; that is they result more from an evaluation than mere description (see also Bernard 2002). These interpretations may, themselves, be influenced by the social characteristic under consideration. For example, the determination of culpability (i.e., an offender's level of blameworthiness), remorsefulness, and future dangerousness (i.e., a defendant's propensity to recidivate) is not simply a matter of description, but a value judgment (see, generally, Blume et al. 2001; Eisenberg, Garvey, and Wells 1998; Sundby 1998). Second, the impact of an extra-legal characteristic at earlier stages in the criminal justice process (i.e., arrest, charging, plea bargaining) may have a sizable impact on decisions at a later stage (Mears 1998a, p. 681; Piehl and Bushway 2007, p. 122). By the time a case reaches court, the particular extra-legal characteristic may have very little, if any, direct effect on sentencing. The extra-legal characteristic, however, may have a strong indirect effect on

sentencing resulting from the filtering process leading up sentencing. Cases that reach the sentencing stage are a very select group that only represents a small portion of similar cases that originally entered the system (Pratt 1998, p. 519). These cases tend to be the most serious cases, so similarly situated offenders may, indeed, be treated the same. Arrest, charging, and plea-bargaining decisions, however, may be influenced by the extra-legal characteristic such that individuals with certain devalued social characteristics who commit, objectively, identical offenses as individuals with valued social characteristics are charged with more severe offenses and offered less favorable plea bargains (Dixon 1995, pp. 1162–63). Since the discretion exercised by police and prosecutors is greatest in these earlier stages, the cumulative bias resulting from these decisions is masked when researchers only examine the sentencing stage (see Donziger 1996; Sorensen and Wallace 1995, 1999).

Conflict theorists are frequently criticized for making such open-ended predictions that any contradictory evidence can be interpreted or re-interpreted to support the theory (see Liska 1993).<sup>259</sup> For example, when faced with unsupportive evidence suggesting certain laws and institutional practices benefit disadvantaged groups, many conflict theorists argue that such laws and practices only *appear* to serve disadvantaged groups in the short-term to maintain societal stability, but ultimately serve the long-term interests of elites (see Collins 1975). The open-ended nature of predictions derived from the conflict perspectives, however, does not automatically preclude the theory from being falsifiable. Conflict theorists, for example, could specify, *a priori*, the conditions under which powerful elites would establish laws and practices that give the appearance of

<sup>&</sup>lt;sup>259</sup> In order for a theory to be testable (and falsifiable), there must be some evidence, if discovered, that would invalidate the theory. If a theory can treat all occurrences as confirmations, then it is not scientifically useful (Cooney 1993, p. 2222).

equality of neutrality. Nearly all theorists employing the conflict perspective, however, fail to establish such scope conditions when testing conflict theory (see, e.g., Hunt 1993).

Interactionist theory, because of its complexity, has been criticized for being virtually unfalsifiable (but cf. Tamanaha 1997). Interactionists argue that individual interpretation is essential understanding the role of legal and extra-legal factors in criminal sentencing. From this perspective, a particular case legal or extra-legal characteristic could be related to increased sentence severity in one case and leniency in another. While this is possible, interactionists fail to specify the conditions necessary for case factors to have a particular hypothesized effect. Straightforward testing of interactionist theory is also undermined by the fact that cases with identical legal and extra-legal characteristics can result in different outcomes because these factors may be differentially interpreted by decision-makers in the criminal sentencing process.

Empirical tests of racial disparities in criminal sentencing employing the interactionist model, for example, tend to emphasize the conditional nature of the relationship between race/ethnicity and sentencing outcomes. Interactionists interpret evidence of racial/ethnic characteristics interacting with other case factors, such as degree of urbanization or political climate (e.g., Hagan 1977), as supporting the theory. This interpretation, however, is questionable on several grounds. First, race/ethnicity can have both a direct (i.e., additive) and a conditional (i.e., interactive) effect on sentencing. Although race/ethnicity may have a greater impact some circumstances than others (e.g., urban versus rural context), race/ethnicity may still be similarly influential in both situations. While this may suggest that importance of race/ethnicity is conditional on the social context of the case, it does not necessarily suggest that the impact of race/ethnicity

operates solely through its interaction with the situational variable. Second, the influence of race/ethnicity may vary by social context for reasons other than changes in the meaning of race across these situations (see Black 1989; Liska 1993). Interactionists' reliance on the untestable assumption of negotiated meaning, however, precludes an examination of this critical question.<sup>260</sup> As Mayhew (1981) notes, it is no different than saying that "secret forces" are responsible for their decisions—both assertions are equally untestable. Finally, interactionists suggest that the legal and extra-legal factors present in a case may interact with an infinite number of other legal and extra-legal factors. This assertion, alone, makes the theory largely—if not completely—unfalsifiable because it would be impossible to hold constant the effects of every conceivable variable. Additionally, interactionists' focus on legal rules make the theory susceptible to many of the same criticisms of the formal legal perspective. The imprecision of predictions, along with its numerous unfalsifiable assumptions, undermine the testability of the interactionist model.

*Empirical Validity.* Problems with testability make statements about the validity of these three perspectives tentative at best. Nonetheless, a large body of empirical literature has accumulated on the influence of race/ethnicity and other extra-legal factors in the criminal sentencing process and some provisional conclusions can be drawn. A thorough survey of the empirical literature reveals support for all three perspectives (see Kempf-Leonard and Sample 2001; Pratt 1998; Urbina 2003; Weitzer 1996). Some

<sup>&</sup>lt;sup>260</sup> Although it may be possible to partly examine the causal mechanisms suggested by some theorists by explicitly measuring the intervening variable(s) identified in the theory (an approach increasingly being used when testing competing criminological theories that identify many of the same explanatory variables, but relate them to crime and delinquency in very different ways, see Agnew 1995), this approach seems less well suited for testing an interactionist account of the relationship between race/ethnicity (and other extra-legal variables) and sentencing. For example, it is unlikely that analysts could accurately measure "negotiated meaning" in a courtroom or experimental setting.

studies support the formal legal model, suggesting that race/ethnicity and other extralegal factors exert no effect (or very little effect) on sentencing decisions when legitimate legal factors are taken into account (e.g., Hagan 1974; Kleck 1981, 1985; Klein and Rolph 1989) and that the importance of race/ethnicity and other extra-legal factors, relative to legitimate legal characteristics, has greatly diminished over time (Pratt 1998).<sup>261</sup> Other studies provide evidence for the conflict perspective, revealing that race/ethnicity and other extra-legal factors significantly impact sentencing decisions net of relevant legal factors (e.g., Crawford, Chiricos, and Kleck 1998; LaFree 1989; Mustard 2001; Petersilia 1985; Smith and Damphouse 1996; Steffensmeier *et al.* 1998; Ulmer and Kramer 1996). Still others support the interactionist approach, indicating that race/ethnicity influences sentencing outcomes primarily through their interaction with other legal and extra-legal variables (e.g., Farnworth and Horan 1980; Hagan and Zatz 1985; Miethe and Moore 1986; Zatz 1984).

While these divergent findings may partly result from the conceptual ambiguities highlighted above, they also result from differences in methodological rigor (Bushway and Piehl 2001; Wooldredge 1998). In a meta-analysis of 47 race/ethnicity and sentencing studies, Pratt (1998, p. 518) discovered that differences in study design and analytical approach strongly influenced research findings sources of racial/ethnic disparities in criminal sentencing (cf. Schanzenbach and Yaeger 2006, p. 792).<sup>262</sup> In particular, the manner in which race/ethnicity was operationalized tremendously

<sup>&</sup>lt;sup>261</sup> Perhaps more problematic for formal legal theorists is that empirical studies reveal that legitimate legal characteristics explain only a modest proportion of the variation in criminal sentencing (see, e.g., Berk, Li, and Hickman 2005; Blumstein *et al.* 1983). Irrespective of the role of illegitimate case characteristics, such as race/ethnicity, age, and gender in criminal sentencing, the formal legal model posits that legally relevant case factors are primarily responsible for sentencing outcomes.

<sup>&</sup>lt;sup>262</sup> Meta-analysis provides for the statistical discovery of common patterns in the research literature, where inferences can be drawn on the basis of the "effect size" (or predictive capacity) of variables (Pratt 1998, p. 515; Raudenbush and Bryk 2002, p. 205).

influenced whether researchers found a statistically significant effect. Certain classifications of racial/ethnic tended to mask the true effect of race/ethnicity on sentencing (see also Steffensmeier and Demuth 2001, pp. 169–70).<sup>263</sup> Also, studies taking into account racial bias at earlier stages of criminal process (e.g., selection bias resulting from charging and plea-bargaining decisions) were more likely to find a significant race/ethnicity effect (see Albonetti 1997; Chiricos and Crawford 1995; Sampson and Lauritsen 1997a; Wooldredge 1998). Mitchell's (2005, p. 462) recent meta-analysis of race/ethnicity and sentencing research—expanding Pratt's study to include both published and unpublished sentencing studies (71 total)<sup>264</sup> and examining both continuous and discrete sentencing outcomes<sup>265</sup>—also discovered that blacks were sentenced more harshly than whites, independent of other measured factors, although the observed differences were generally small.

Theoretical and methodological limitations, notwithstanding, the bulk of research suggests that extra-legal factors continue to play a significant role in criminal sentencing, either directly or indirectly.<sup>266</sup> Mustard (2001), analyzing 77,236 federal offenders

<sup>&</sup>lt;sup>263</sup> Studies that operationalized race as "white/non-white" had the greatest likelihood of finding a significant effect. This largely resulted from the fact that "black/white" classifications grouped Hispanics and Native Americans with Anglos, and thereby masked the true race effect. This was particularly true in regions of the country that had large Hispanic populations (e.g., the southwestern region of the United States) (see also Zatz 1984).

<sup>&</sup>lt;sup>264</sup> Both published and unpublished studies were included to reduce the possibility of "publication bias," which may arise when published studies are not representative of all valid studies that have been undertaken. This bias can distort meta-analysis as published studies tend to report positive (i.e., statistically significant) findings rather than negative findings (Begg 1994). Mitchell discovered that published studies tended to report substantively and statistically larger effect sizes than unpublished studies (p. 457). Forty-five percent of the studies were published as journal articles, 16 percent were published as books or book chapters, and 39 percent were unpublished (e.g., dissertations, conference papers, *et cetera*) (p. 452).
<sup>265</sup> Mitchell's meta-analysis focused on five types of sentencing outcomes: (1) imprisonment decisions, (2) length of incarcerative sentence, (3) ordinal scales of sentence severity, (4) discretionary lenience (e.g., downward departures and stays of sentence), and (5) discretionary punitiveness (e.g., upward departures and enhanced sentencing provisions for eligible repeat offenders) (p. 444).

<sup>&</sup>lt;sup>266</sup> While the overwhelming majority of studies examining the role of extra-legal factors in the criminal charging-and-sentencing process have focused on defendant and victim characteristics, a limited number of studies have also explored the impact of judge characteristics on sentencing outcomes. Evidence suggests

sentenced under the United States Sentencing Commission (USSC) guidelines between October 1, 1991 and September 30, 1994, discovered that extra-legal factors continue to influence sentencing outcomes.<sup>267</sup> Controlling for numerous criminological, demographic, and socioeconomic variables, Mustard discovered that blacks were more likely to be imprisoned, receive longer sentences, and be subject to "upward" sentencing departures than whites (for similar findings, see Bushway and Piehl 2001).<sup>268</sup> Further analyses revealed that blacks were less likely to receive "downward" sentencing departures and, when receiving downward departures, blacks received smaller reductions than whites (see also Johnson 2005, pp. 785–86).<sup>269</sup>

that judges who are female (Spohn 1990a; Steffensmeier and Hebert 1999), older (Spohn 1990b), and have served more time on the bench (Hogarth 1971) tend to impose harsher sentences, net of legitimate case characteristics. Research on the impact of judge's race on sentencing severity, however, has produced mixed results (see, e.g., Spohn 1990a; Spohn 1990b; Steffensmeier and Britt 2001; Welch, Combs, and Gruhl 1988).

<sup>&</sup>lt;sup>267</sup> The centerpiece of the federal guidelines is a grid containing 258 boxes (termed the "Sentencing Table"). The grid's horizontal axis ("Criminal History Category") adjusts severity on the basis of the offender's past conviction record. The vertical axis ("Offense Level") reflects a base severity score for the crime committed, as further adjusted for those aspects of the crime that the guidelines deem relevant to sentencing. The guidelines instruct judges on how to calculate both "criminal history" and "offense level." The box at which the two factors intersect then determines the range within the judge may sentence the defendant. As required by law, the sentencing range in each box is small, the highest point being 25 percent more than the bottom, thus representing one source of discretion retained by judges. Judges may "depart" from the guidelines, formally, based on two circumstances. First is when the defendant offers substantial assistance in the prosecution of others (a downward departure). In this situation, the prosecutor must "authorize" the departure. The judge is then free to depart below the "box" range or any applicable statutory minimum sentence. Second is when the judge is able to demonstrate that there are factors in the case that have not been "adequately" factored into the guidelines and make the case "atypical." Departures under this circumstance may be either upward or downward (U.S. Sentencing Commission 1987, 1991). For a thorough discussion of judges' general use of guideline departures, see Schanzenbach (2005a). <sup>268</sup> In particular, Mustard controlled for the defendant's age, gender, level of education, number of dependents, citizenship status, income (adjusted for 1993 dollars), offense level, offense type, criminal history, and the federal district where the case was tried. Mustard also analyzed interactive effects of many of these variables with one another.

<sup>&</sup>lt;sup>269</sup> In November 2005, the Brennan Center for Justice and the National Institute for Law and Equity organized a focus group of twelve former U.S. Attorneys. The former U.S. Attorneys agreed that "conscious attention to the role of race in prosecutorial decision-making, as well as concerted efforts to monitor and improve the decision-making process, [was] essential for mitigating unwarranted racial disparities in the outcomes of federal criminal prosecutions" (Lu 2007, p. 195). Thirteen former U.S. Attorneys signed on to a set of prosecutorial guidelines for addressing racial disparities, which focused on five areas: (1) prosecutorial decision-making; (2) law enforcement/task forces; (3) training; (4) management/accountability; and (5) community (pp. 199–201).

Conducting a comprehensive review of the research literature on extra-legal factors and sentencing dating from the 1950s to the 1990s, both before and after the implementation of federal and state sentencing guidelines, Kempf-Leonard and Sample (2001) discovered significant racial/ethnic differences in 30 of 34 studies that analyzed racial/ethnic disparities, but not always in a direction that disadvantaged minority offenders (see also Steffensmeier and Demuth 2000, 2001). Furthermore, racial/ethnic disparities were found in every study conducted the since implementation of sentencing guidelines (see Crawford *et al.* 1998; Heaney 1991; Johnson 2003, 2005; Kramer and Steffensmeier 1993; Kramer and Ulmer 1996, 2002; Smith and Damphouse 1996; Steffensmeier *et al.* 1993; Steffensmeier *et al.* 1998; Ulmer and Kramer 1996). Kempf-Leonard and Sample (2001) also found that other extra-legal factors—gender, socioeconomic status, and familial status—significantly influenced sentencing outcomes in 46 of 49 studies, but not always in the expected direction (see also Schanzenbach 2005b).<sup>270</sup>

The "guided discretion" death penalty statutes have also failed to eliminate or significantly reduce racial disparities in the capital charging-and-sentencing process. As discussed in Chapter Four, the vast majority of studies of the capital charging-and-

<sup>&</sup>lt;sup>270</sup> Specific studies included in the review were: Albonetti (1991), Barry and Greer (1981), Bickle and Peterson (1991), Boritch (1992), Crawford, Chiricos, and Kleck (1998), Crew (1991), D'Alessio and Stolzenberg (1993), Daly (1987, 1989), Emmelman (1994), Farnworth and Horan (1980), Farrell and Swigert (1978), Farrington and Morris (1983), Figueira-McDonough (1985), Ghali and Chesney-Lind (1986), Griswold (1987), Gruhl and Welch (1984), Hagan, Nagel, and Albonetti (1980), Heaney (1991), Holmes *et al.* (1996), Johnston, Kennedy, and Shuman (1987), Kempf and Austin (1986), Kramer and Steffensmeier (1993), Kramer and Ulmer (1996), Kruttschnitt (1981, 1982), Kruttschnitt and McCarthy (1985), Miethe and Moore (1985, 1986), Moore and Miethe (1986), Myers and Talarico (1986), Nagel and Johnson (1994), Nobiling, Spohn, and Delone (1998), Peterson and Hagan (1984), Sloan and Miller (1990), Smith and Damphouse (1996), Spohn (1990b), Spohn and Cederblom (1991), Spohn, Gruhl, and Welch (1987), Spohn, Welch, and Gruhl (1985), Steffensmeier, Kramer, and Streifel (1993), Steffensmeier Kramer, and Ulmer (1998), Stolzenberg and D'Alessio (1994), Ulmer and Kramer (1996), Unnever (1982), Weisburd, Waring, and Wheeler (1990), Wheeler, Weisburd, and Bode (1983), Wolf and Weissman (1996), and Zatz (1984).

sentencing process in the United States over the past 70 years reveal that race/ethnicity plays a prominent role in the administration of the death penalty. Over 90 percent of studies examining the capital punishment process since the *Furman* decision revealed that race/ethnicity continues to influence the likelihood that the death penalty is sought and imposed, net of the legitimate legal characteristics of the case. Furthermore, over 80 percent of the methodological rigorous studies conducted in eleven death penalty states since *Furman* report significant race-of-victim effects in charging or sentencing decisions.<sup>271</sup>

In sum, the three theoretical perspectives evaluated above significantly differ in the diversity of facts they address (generality), the economy of their expression (parsimony), their capacity to predict facts (testability), and their conformity to those facts (validity)—their degree of scienticity (Black 2000a, p. 351). Unfortunately, the inability of these theories to fully satisfy the fundamental standards of scientific theory seriously undermines their capacity to offer a thorough explanation of legal behavior. Furthermore, formal legal and conflict perspectives appear to be particularly limited in their ability to anticipate or readily accommodate novel facts—an important feature of any progressive theory (see Harris 1979, pp. 17, 24; Jasso 1988, p. 1; Lakatos 1970, p. 131).<sup>272</sup>

<sup>&</sup>lt;sup>271</sup> Radelet and Pierce (1985) also discovered that prosecutors were more likely to "upgrade" the criminal charge in cases involving white victims by charging the defendant with a felony offense or some other aggravating factor, in addition to murder, that was not initially included in the police report.

<sup>&</sup>lt;sup>272</sup> Homans (1967, pp. 72–73) argues that important facts about human behavior are "obvious," "boring," and known to everyone. He claims that sociologists can no longer discover new facts; they can only explain known facts (p. 105). However Black (2002b, p. 669) claims that Homans is wrong and that his paradigm—pure sociology—is capable of making numerous predictions about unknown facts. Black's theory of law, for example, makes the novel discovery that law is situational rather than universal—only cases with the same location and direction have the same law. According to Black, the "rule of law" and "equality before the law" nowhere described legal reality. Discrimination in the law is not exceptional, it is normal, and law obeys the same sociological principles (i.e., the "laws of law") everywhere (p. 770). Black
These shortcomings, however, do not render these theories wholly invalid or useless. In fact, studies employing one or more of these general perspectives have illuminated important features of the criminal sentencing process. As noted above, research in the formal legal tradition (see, e.g., Kleck 1981; Kleck 1985; Klein and Rolph 1989) has revealed that legitimate legal case characteristics explain a modest portion of variation in sentencing outcomes and the increased formalization of the criminal justice process has had a perceivable impact on reducing bias, arbitrariness, and caprice in criminal sentencing over the past several decades *for certain crimes*. Formal legal theorists have also highlighted that race/ethnicity (and other frequently analyzed extralegal factors) tend to be highly correlated with important legitimate legal factors, and studies that do not properly control for these legitimate case characteristics are at risk of seriously overestimating the impact of extra-legal factors on criminal court decisions.

Research growing out of the conflict and interactionism traditions has also illuminated important relationships between extra-legal factors and the criminal justice process. Minority and economic threat perspectives, derived from the conflict tradition, explain how dominant racial and economic groups may employ "enhanced" criminal justice efforts in an effort to control rapidly growing subordinate groups (Blalock 1967; Mitchell 2005; Parker, Stults, and Rice 2005).<sup>273</sup> For example, research suggests that cities with a high minority presence (particularly blacks and Hispanics) have larger police forces, higher arrest rates, and spend more on prisons and jails, net of actual crime rates

has also used his pure sociology to make novel predictions about the behavior of ideas, art, violence, and God (see, e.g., Black 1995, 2004a).

<sup>&</sup>lt;sup>273</sup> It has also been argued that economic and political competition *breed* prejudice and that discrimination, prejudice, and negative stereotyping sharply increase as competition for scarce jobs increases (Aronson 1972, pp. 180–81).

(Fagan, West, and Holland 2003; Jackson 1986; Liska 1993).<sup>274</sup> Higher arrests rates, in turn, are associated with more severe criminal sentences, net of legitimate case characteristics (e.g., offense severity and prior criminal record). Moreover, conflict theorists have shown that growing minority presence in an area is associated with whites developing more conservative political ideologies that support harsher criminal punishments, net of actual crime (Baumer *et al.* 2003; Jacobs and Carmichael 2002; Jacobs, Carmichael, and Kent 2005; Jacobs and Tope 2007; King and Wheelock 2007).<sup>275</sup>

Interactionists have contributed to our understanding of the criminal sentencing process by revealing how actors' characteristics—both ascribed and achieved—appear to interact to influence sentencing outcomes. For example, interactionists have noted that the influence of race/ethnicity is likely to be conditioned by other characteristics, such as age and gender (see Spohn *et al.* 1985; Steffensmeier and Demuth 2006; Steffensmeier *et al.* 1998), employment status (Hagan *et al.* 1980; Nobiling *et al.* 1998; Spohn and Holleran 2000; Weisburd *et al.* 1990), level of formal education (Albonetti 1997; Mustard 2001), and marital/familial status (Bickle and Peterson 1991; Daly 1989; Ghali and Chesney-Lind 1986; Kruttschnitt and McCarthy 1985; Peterson and Hagan 1984). Steffensmeier and colleagues (1998), for example, note that although age, gender, and race/ethnicity all have a direct impact on criminal sentencing, net of legitimate case characteristics, certain age-gender-race combinations are more likely to receive harsher sentences than other combinations. In fact, young black males are the group most likely

<sup>&</sup>lt;sup>274</sup> Recent evidence suggests that the relationship between the minority presence and police strength may be limited to the United States (Kent and Jacobs 2004).

<sup>&</sup>lt;sup>275</sup> King's (2007) recent analysis of law enforcement responses to hate crime in the United States reveals that compliance with federal hate crime law is *less* likely in places with larger black populations, but the impact of black population size on compliance is contingent on region—a positive correlation in the Northeast, but an inverse relationship in the South.

to receive the most severe sentences, all else equal. Spohn and Holleran (2000) extend Steffensmeier *et al.*'s work by examining age, gender, race/ethnicity, *and* employment status combinations, and discover that young unemployed black and Hispanic males are most likely to receive harsher sentences, all else being equal.<sup>276</sup> Similar to conflict theorists, interactionists have also highlighted the importance of larger contextual factors such as arrest rate, unemployment rate, and urban/rural location (Myers and Talarico 1986; Thomson and Zingraff 1981; Weidner *et al.* 2004).

Although the various relationships uncovered by these three different perspectives are informative, each perspective is based upon different assumptions and makes different predictions. Perhaps the most significant shortcoming of these perspectives, then, is that none is fully capable of explaining the *robust* empirical relationships illuminated by the other perspectives. That is, none offers a sufficiently abstract explanatory framework from which the various empirically valid hypotheses could have been derived. Due to this major shortcoming, as well as the aforementioned problems with respect to each approach's level of scienticity, a fourth theoretical perspective is presented and evaluated below. This fourth perspective is believed to offer a superior explanation of legal behavior not only because of its ability to readily accommodate the facts predicted by these alternative perspectives, but also by achieving a greater degree of scienticity and avoiding several of the problems that readily plague social scientific theory.

<sup>&</sup>lt;sup>276</sup> Much research suggests that juveniles who are members of racial/ethnic minority groups (particularly black males) are more likely to have their cases transferred to juvenile court, net of offense seriousness and criminal/deviant background (Hsia, Bridges, and McHale 2004; Males and Macallair 2000; Pope, Lovell, and Hsia 2002).

## 6.2 THE BLACKIAN PARADIGM

In an effort to address the fundamental limitations of existing theories of law in particular—and sociological theory in general—theoretical sociologist Donald Black developed and advanced a research strategy for the study of human behavior operating solely at the social level. Black attempts to explain social life "through the lens of social relationships rather than through the myriad of biological, psychological, political, or economic factors that also influence social systems" (Horwitz 1990, p. ix; see also Simmel 1909, pp. 21–22). Black's paradigm represents a significant departure from conventional social scientific theory, and as a result, is unfamiliar—even shocking—to many social scientists (see Black 1995, pp. 864–69; 2002a, p. 119; Tucker 1999, p. 5).<sup>277</sup> The novelty of Black's approach, then, makes it necessary to carefully identify the assumptions on which it rests.

## 6.2.1 Assumptions of the Blackian Paradigm

Black's (1979b) paradigm rests on three key assumptions that need to be made explicit before entering into a specific discussion of this theoretical strategy. The first assumption concerns the structure of scientific explanation. Black, adopting the "covering-law model" of scientific explanation, posits that it is possible to formulate highly general propositions—both timeless and spaceless—that illuminate the basic properties of social life. Second, Black assumes that the various relationships between

<sup>&</sup>lt;sup>277</sup> According to Black (1995, p. 867), his working is shocking because it is epistemologically incorrect. "It violates conventional conceptions of social reality in general and legal and moral reality in particular. Therefore it shocks—epistemologically shocks—many on whom it is inflicted." Some scholars have even gone so far as to accused Black of contributing to a "nihilistic destruction of the legal tradition" (see Constable 1994, p. 19). Cooney (2003a, p. 1420) suggests that critics of the Blackian paradigm miss the "spirit" of his explanatory effort—a "theoretical nature of argument with all that implies in the way of trying to build a general, parsimonious, and new set of ideas that bring a measure of clarity and order to a field."

people underlie the variation in social life. His final assumption is that variation in social life is, in principle, quantifiable. These assumptions are discussed in greater detail below.

*Covering-Law Approach.* The covering-law model, popularized by Carl Hempel (1965, 1966) and others (e.g., Machlup 1955), has become the standard model of scientific explanation in the physical sciences (Nagel 1979) and is enjoying increased popularity in the social sciences as well (see Harris 1979; Lenski 1988; Zetterberg 1966).<sup>278</sup> Covering laws are either universal or probabilistic generalizations describing regularities (Hempel and Oppenheimer 1948). The model does not discuss singular cases; rather it understands causality to be couched in terms of what regularly happens between certain types of entities. These generalizations tell us that something is to be expected; however they are silent as to why something occurred (Nagel 1979, pp. 26–27). According to this approach, a theory is a restricted set of propositions—whose generality is not restricted to limited regions of space and time—yielding deductive implications that can be confirmed or refuted through empirical tests (Merton 1967, pp. 40, 66). Explanation occurs when observable variation in phenomena is deducible from the general propositions that comprise the theory (Homans 1964, pp. 811–12; Merton 1945, pp. 469–70).

Some scientists—physical, social, and biological—remain opposed to this approach and argue that the establishment of general laws is impossible or that the identification of causal mechanisms is essential to the explanation of phenomena (see, e.g., Cartwright 1983, 1989; Hëdstrom and Swedberg 1998; Machamer, Darden, and Craver 2000; Ragin 1987; Tittle 1995). For over two centuries, however, philosophers of

<sup>&</sup>lt;sup>278</sup> The covering-law model is also commonly referred to as the "hypothetical-deductive model," the "deductive-nominological method," or simply the "deductive model" (see Braithwaite 1953, p. 1; Horwitz 1983, p. 375; Nagel 1979, p. 21; Somers 1998, p. 736).

science have argued that science is incapable of answering questions as to why an event occurs or why things are related in certain ways—it only describes how or in accordance with what rules phenomena occur (Hume [1748] 1955). In fact, both Hume ([1739] 1975) and Nietzsche ([1888] 1998) believed that the concept of causality is a requirement of the human mind, rather than a necessity of science. They argued that the human mind was incapable of understanding or explaining *change* except as the result of some antecedent cause (see also Russell [1913] 1992). Moreover, Nietzsche suggested that people tend to search for a preferred explanation rather than an actual cause, and "acceptable" explanations reflect predilections of the audience rather than the accuracy of the claim (see also Mayhew 1980; Popper [1934] 1968).<sup>279</sup> As Cooney observes (1986, p. 269), "No matter how all-inclusive an explanatory framework purports to be, there are limits to the number of non-circular answers it can provide. Sooner or later, all theories become mute in the face of the persistent asking of the question, 'why'?" Similarly, Sobel (1995, p. 3) comments, "social scien[tists] often incorrectly equate explanation with causation...[but] many of the processes and phenomena that are of interest to social and behavioral scientists are not causal or at least not entirely causal." Even those scientists who believe that causal statements are possible recognize the necessity of first establishing highly general laws from which empirical hypotheses can be derived. Braithwaite (1953, p. 2), for example, notes:

To emphasize the establishment of general laws as the essential functions of science is not to overlook the fact that in many sciences the questions to which the scientist attaches most importance are historical questions about the causes of particular events rather than questions directly about general

<sup>&</sup>lt;sup>279</sup> Black (2002a, p. 105) posits that individuals' social distance to a subject matter explains their desire for the scientific analysis of that subject. Individuals who are very close to the subject tend to prefer other forms of knowledge—e.g., common sense, religion, metaphysics, and folklore—and openly question the appropriateness of a "scientific understanding" of that particular subject.

laws...[b]ut the statement that some particular event is the effect of a set of circumstances involves the assertion of a general law; to ask for the cause of an event is always to ask for a general law which applies to the particular event. Though we may be more interested in the application than in the law itself, yet we need to establish the law in order to know what law it is which we have to apply.<sup>280</sup>

In fact, Nagel (1979) suggests that all explanatory forms in science can be shown to

exemplify the covering-law model when implicit assumptions are made explicit.<sup>281</sup>

Reconceptualization of Human Behavior. The second assumption of Black's

paradigm is that social life is a reality in its own right, defined by human interaction, and

can be studied as a natural phenomenon without regard to psychology (Black 1979b,

1995). While some scholars have found this assertion problematic (e.g., Frankford 1995;

Greenberg 1983a; Tamanaha 1997), it has a rather long history in social thought. For

example, well over a century ago, Marx ([1857] 1973, p. 176) noted that society does not

consist of individuals, "rather it expresses the sum interrelations in which individuals

stand with respect to one another." Similarly, Radcliffe-Brown ([1935] 1965)

commented:

Individual human beings, the essential units of society, are connected by a definite set of social relations into an integrated whole. The continuity of the social structure is not destroyed by changes in the units. The continuity of [the social] structure is maintained through the process of social life, which consists of the activities and interactions of the individual human beings and of the organized groups in which they are united (pp. 179–80).

Durkheim ([1897] 1994) also remarked:

<sup>&</sup>lt;sup>280</sup> Similarly, Popper ([1934] 1968) argues that causality can *only* be a feature of universal laws (see also Jasso 1988, pp. 2, 6; Nagel 1979, p. 31; Turner 2002, p. 667). King and colleagues (1994, p. 34) note, "Even if explanation—connecting causes and effects—is the ultimate goal, description has a central role in all explanation, and is fundamentally important in and of itself."

<sup>&</sup>lt;sup>281</sup> Nagel (1979, pp. 21–26) identifies and describes four major patterns of explanation found in the sciences: (1) the deductive/covering-law model; (2) probabilistic explanations; (3) functional or teleological explanations; and (4) genetic explanations.

[S]ocial life should be explained not by the conceptions which the participants have of it, but by the fundamental causes which escape their consciousness...and these causes ought to be sought principally in the way in which associated individuals are grouped. It is only on this condition that history can become a science and sociology, consequently, exist (p. 126).

Black's ability to explain human behavior without reference to goals, motives, purposes,

values, needs, functions, interests, intentions, or anything else that is not directly observable results from this reconceptualization of individual action as social action (but cf. Durkheim [1912] 1995; Homans 1964; Weber [1922] 1968). For Black, what is normally considered individual behavior or the behavior of groups becomes the behavior of a particular form of social life.<sup>282</sup>

The interaction of individual actors, that is, takes place under such conditions that it is possible to treat such a process of interaction as a system in the scientific sense and subject it to the same order of theoretical analysis which has been successfully applied to other types of systems in other sciences (Parsons 1951, p. 3).

Therefore, a person injuring their spouse or a lynch mob setting fire to a jailhouse is reconceptualized as an increase of violence in a social situation, or the behavior of violence (see Phillips 2003; Senechal de la Roche 1997a, 2001).<sup>283</sup> A family going to church or a minister delivering a sermon is reconceptualized as an increase of religion in a social situation, or the behavior of religion. An individual calling the police, an arrest, a jury verdict, and an appeal are reconceptualized as increases of law in a conflict, or the behavior of law (see Black 1995, p. 859; 2000a, p. 347). When conceptualized in this manner, the explanation of behavior need not resort to untestable assumptions about the

<sup>&</sup>lt;sup>282</sup> By "behavior," Black simply means variation. "Every thing behaves, living or not, whether molecules, organisms, planets, or personalities. This applies to social life as well, to families, organizations, and cities, to friendship, conversation, government, and revolution" (Black 1976, p. 1).

<sup>&</sup>lt;sup>283</sup> "Violence is a social phenomenon with its own dynamic and laws. Individuals who kill and assault are merely agents of violence" (Cooney 2003a, p. 1421).

inner motivations of individuals or groups (cf. Parsons 1937). Social life, as defined by Black, has neither a mind nor emotions—it cannot be explained by its psychological characteristics.

Black, however, does not argue that psychological variables are unimportant in the explanation of human behavior (but see Mayhew 1980, p. 335). Nor does he suggest that his paradigm is at odds with psychological explanations of human behavior (Black 1976, pp. 7–8).<sup>284</sup> Rather, Black treats psychological and other non-sociological variables as constants in his theory (see also Lenski 1988, p. 163) and focuses specifically and exclusively on a *purely* sociological theory of human behavior and uses its own imagery, concepts, and framework of analysis (Black 1976, p. x).<sup>285</sup>

*Quantification of Social Variation*. Like older sciences, such as physics and chemistry, Black (1980, p. 217) argues that sociological theory should be maximally testable (see also Harris 1979, p. 17). As mentioned earlier, if a theory is to be testable, it must be stated in quantitative language so that predictions can be evaluated by measurement (i.e., counting). Black's paradigm assumes that variation in social reality is subject to quantification (see also Griffiths 1984, p. 39; Zetterberg 1966). Since measurement can occur at varying levels of precision (see previous discussion of types of

<sup>&</sup>lt;sup>284</sup> Cooney (1986, p. 294) argues that "[Individuals' and groups'] intentions and motivations are important in understanding their behavior/action, but not conclusive because people's understanding of their behavior cannot be assumed to be theoretically sound. Much work from psychoanalysis to post-structuralism has shown that human behavior can be explained by processes of which the actors may not be aware. Therefore, there is always a larger social context outside the individual that must be considered" (see also Cerulo 2002, p. 654; Miethe and Drass 1999). Benedictus de Spinoza ([1677] 1883, p. 108) famously remarked, "Men are mistaken in thinking themselves free; their opinion is made up of consciousness of their own actions, and ignorance of the causes by which they are determined. Their idea of freedom, therefore, is simply their ignorance of any cause for their actions." As Wacquant (2002, p. 1470) has explained, one of the "the proximate causes of the common limitations and liabilities of [social scientific explanation]...[is] naive acceptance of ordinary categories of perception as categories of analysis[.]" <sup>285</sup> According to Black (2000a, p. 348), "Anything pure is the most of itself, autonomous and free of

everything else" (see also Bourdieu [1992] 1996). As a result, Black (1979b) has labeled his paradigm, which operates solely at the social level, "pure sociology."

measurement), quantification need only identify if more or less of something is present in a particular setting or under certain conditions. Or, alternatively, if something is more likely to occur at all in a particular setting or under certain conditions. When social variation is conceptualized in this manner, quantification almost becomes intuitive (see Black 1980).<sup>286</sup> For example, a call to the police is more law than no call at all. Similarly, an arrest is more law than a simple call. Likewise, a prosecutor seeking an indictment is more law than an arrest. Law, then, can be seen as increasing as it progresses through various stages of the legal system (p. 211; see discussion below). Although the precise change in law from an arrest to indictment may be indeterminable, it is clear that an indictment is more law than an arrest. Variation in the years a defendant is sentenced in a criminal case or the amount of monetary compensation awarded in a civil case, on the other hand, does allow for the determination of precise differences.

The quantification of variation in all other forms of social life is possible as well. For example, it is possible to determine when more or less violence occurs (e.g., a shooting is more violence than a simple fist fight) by measuring the amount of injury—or potential injury—to the victim. The quantification of variation in religion, music, literature, medicine and science is also possible (see Black 1993, 2000a).<sup>287</sup>

To be sure, these three assumptions are debatable. According to Harris (1979, p. 20), however, non-empirical assumptions are necessary for *all* research strategies (see

<sup>&</sup>lt;sup>286</sup> Jasso (2006, p. 38) argues that, *inter alia*, a theory should "provide[] a foundation for measurement."
<sup>287</sup> Black (1980, p. 214) also suggests that the quantity of social phenomenon in a social setting can be measured indirectly, if necessary. He recognizes that it may not be possible to directly observe how a phenomenon varies, so an alternative method is to measure the phenomenon by its relationship to another that can more easily be observed and quantified (e.g., the behavior of mercury in a sealed glass tube as an indirect measure of the amount of heat).

also Dumont and Wilson 1967, pp. 987–88; Somers 1998, pp. 740–45).<sup>288</sup> Similarly, Popper ([1934] 1968, p. 27) argues that a theory must rest on assumptions because it can never be logically deduced from the facts it explains—logical induction of a theory is impossible (see also Wacquant 2002, pp. 1523–24; but cf. Glaser and Strauss 1967).<sup>289</sup> One of the strengths of Black's paradigm is that the assumptions on which subsequent theories are based still allow the theories to be highly general and maximally testable (Myers 1980). Black's explanatory structure, the covering-law model, has a long history in the natural sciences and is widely believed to be the best explanatory approach for yielding highly general and empirically verifiable propositions (Braithwaite 1953, p. 9; Harris 1979, p. 17; Lenski 1988, pp. 168–69).<sup>290</sup> His conceptualization of social life has it roots in earlier sociological thought as well (e.g., Marx [1859] 1970; Parsons 1951; Radcliffe-Brown [1935] 1965), and avoids problematic assumptions about human nature and unobservable internal states (e.g., goals, motivations, desires). His final assumption-the quantifiability of social variation-allows for the development of testable propositions pertaining to the widest range of social phenomena. Conversely, the assumptions on which formal legal, conflict, and interactionist perspectives are based deleteriously impact their generality, parsimony, testability, and validity.

<sup>&</sup>lt;sup>288</sup> Kuhn (1970, p. 148) remarked, "Neither side [of competing research strategies] will grant all the nonempirical assumptions that the other needs in order to make its case...the competition between paradigms is not the sort of battle can be resolved by proofs."

<sup>&</sup>lt;sup>289</sup> Science, according to Popper ([1934] 1968, p. 279), does not proceed through observations confirmed by verification; rather it proceeds through overarching conjectures that generalize beyond the data, but are always controlled and sharpened by refutation (see also Friedman 1953, p. 14).

<sup>&</sup>lt;sup>290</sup> Sherman (1978, p. 15) remarks, "For [Donald] Black, to predict is to explain. This position would seem more defensible than the more common situation: explanation without prediction" (see also Nagel 1979, p. 447).

## 6.2.2 Description of Black's Theory of Law

Now that the assumptions of Black's paradigm have been identified and his new conceptualization of social life has been articulated, attention turns to Black's theoretical strategy that explains variation in a particular aspect of social life: the behavior of *law*. Before presenting Black's theory of law, however, it is first important to discuss his conceptualization of law.<sup>291</sup>

*Black's Concept of Law.* Hempel (1965) identifies two major approaches to the meaning of concepts: nominalist and essentialist.<sup>292</sup> In the nominalist approach, concepts have no inherent meaning apart from their definition and are only true to the extent that they are useful (see also Nietzsche [1888] 1998). The purpose of a nominalist definition is to describe a class of phenomena in the world of experience in order to establish general principles (i.e., laws) by which the phenomena can be predicted and explained (see Cooney 1986, p. 266; Horwitz 1983, p. 370). In the essentialist approach, concepts must capture the "true nature" or "fundamental attributes" of phenomena. Essentialist concepts, however, are so vague that they are virtually useless to scientists. In fact, it is often argued that science is unable to explain the essential nature of anything, only how it varies (Nagel 1979).

Black adopts a nominalist approach to the concept of law, defining it simply as "governmental social control" (1972, p. 1096). He broadly defines social control as "the

<sup>&</sup>lt;sup>291</sup> "A definition [of a concept] is the first step toward identifying the empirical family to which it belongs, the theoretical jurisdiction responsible for its explanation, [and] the social process it may engender" (Black 2004b, p. 7).

<sup>&</sup>lt;sup>292</sup> Essentialist definitions are commonly referred to as "real" or "substantive" definitions, while nominal definitions have been called "functional" definitions.

definition of deviant behavior and the response to it" (Black 1976, p. 2).<sup>293</sup> Law, then, is but one kind of social control: "the normative life of a state and its citizens, such as legislation, litigation, and adjudication" (p. 2) (cf. Gibbs 1966; Schwartz 1978b).<sup>294</sup> According to Black (1972, p. 1091), law is not rules, rather it is the *observable* dispositions of legal agents (e.g., police, attorneys, judges, juries, *et cetera*) and, therefore, amenable to scientific inquiry (see also Black 1973, p. 128; Scheppele 1994, pp. 400–401; but cf. Frankford 1995).<sup>295</sup> Consistent with Black's paradigm, law is also a quantitative variable (cf. Espeland and Vannebo 2007, p. 25; Evan 1990, p. 157). The quantity of law is "the amount of governmental authority brought to bear on a person or group" (Black 1989, p. 8).<sup>296</sup> Law, therefore, increases and decreases, is found more in some settings than others, and varies across time and space.<sup>297</sup>

<sup>&</sup>lt;sup>293</sup> "Social control is the normative aspect of social life. It defines and responds to deviant behavior, specifying what ought to be: What is right or wrong, what is a violation, obligation, abnormality, or disruption" (Black 1976, p. 105). According to Ross ([1901] 1920, p. 106), law is the "most specialized and highly finished engine of social control employed by society."

<sup>&</sup>lt;sup>294</sup> Black's definition of law as governmental social control is very similar to Oliver Wendell Holmes's early formulation: "[I]n societies like ours the command of the public force is entrusted to the judges in certain cases, and the whole power of the state will be put forth, if necessary, to carry out their judgments and decrees...[t]he object of our study, then, is prediction of the incidence of the public force through the instrumentality of the courts" (Holmes 1897, p. 457).

Weber also locates law within a larger system of social control. "An order will be called 'law' when conformity with it is upheld by the probability that deviant action will be met by physical or psychic sanctions aimed to compel conformity or to punish disobedience, and applied by a group of men [*sic*] especially empowered to carry out this function" (Weber [1925] 1954, p. 127; see also Austin [1832] 1995; but cf. Hart [1961] 1994). Swedberg (2006, p. 66) notes that Weber believed law was no different from other social phenomena that also increase the probability that some action will take place.

<sup>&</sup>lt;sup>295</sup> Karl Llewellyn ([1930] 1978, p. 3) famously wrote, "What officials do about disputes is, in my mind, the law itself," and he strongly believed that legal decisions were very rarely decided by legal rules.

<sup>&</sup>lt;sup>296</sup> "More generally, the quantity of law is known by the number and scope of prohibitions, obligations, and other standards to which people are subject, and by the rate of legislation, litigation, and adjudication" (Black 1976, p. 3).

<sup>&</sup>lt;sup>297</sup> While law varies across time and space, general propositions predicting and explaining the behavior of law need not be restricted to limited regions of time and space (see, generally, Black 1976).

Black's definition of law has been criticized for not encompassing all that is considered to be law (e.g., rights, enablements, *et cetera*) (see Greenberg 1983a).<sup>298</sup> Black (1972, pp. 1096–97) argues, however, that his definition of law is only used as a means of delineating the subject matter of legal sociology, and is not the *only* proper definition of law (see also Cooney 1986, p. 266; Tamanaha 1997, p. 128).<sup>299</sup> Similarly, Horwitz (1983, ) suggests that while Black's definition does not embrace all existing conceptions of law, it encompasses a rather large and homogeneous body of phenomena considered legal and firmly "locates the subject matter of the sociology of law in the realm of social science...[rather than] within [the realm of] jurisprudence" (pp. 371–72).<sup>300</sup>

In addition to variation in the quantity of law, there is also variation in the *style* of law.<sup>301</sup> The style of law is the language and logic by which law defines and responds to

<sup>&</sup>lt;sup>298</sup> According to Black (1980, p. 221), the rights of individuals (e.g., the Bill of Rights) are not laws, but rather guarantees against laws because the more these rights are respected, the less law there is. Others scholars have expanded the conception of rights to include not only immunities from government intervention, but also powers to call on the government for help (see Holmes and Sunstein 1999). Weber ([1922] 1968, pp. 666–67) defines rights as a matter of degree, stating a right is "an increase of the probability that a certain expectation of the one whom the law grants that right will not be disappointed."

<sup>&</sup>lt;sup>299</sup> In series of articles, Wong (1995, 1998a,b) has criticized Black's concept of law on the basis that Black's definition of government is *unsociological*. In particular, Wong asserts that Black provides a descriptive—rather than an analytical—definition of government that erroneously views government as holistic or integrated, which either exists or does not exist. In contrast, Wong defines government as "formally institutionalized political authority" and suggests that government—as all social phenomena—is not absolute but relative (i.e., government is a matter of degree) (see also Weber [1925] 1954). Law, then, is "more or less governmental social control," resulting from the affirmative delegation of social control rights and responsibilities (Wong 1998b). Wong believes that his modifications of Black's concepts of government and law significantly extend the scope of Black's original formulation by "allowing it to reach out to private social control authority." But while purporting to expand the comprehensiveness of Black's concept, Wong does not challenge any of the relationships between law and social structure articulated in Black's original statement of the theory of law.

<sup>&</sup>lt;sup>300</sup> Gibbs (1968, p. 446) remarked, "Since jurisprudents show little inclination to engage in the construction of substantive theory (as opposed to purely conceptual analysis) and to undertake nomothetic research, the empirical questions about law could pass to the sociology of law by default."

<sup>&</sup>lt;sup>301</sup> Black (1993, p. 5) also introduces another variable aspect of social control: its form. The form of social control is "a mechanism by which a person or group expresses a grievance." The forms of social control generally divide into two categories: those involving only principles (with or without the help of supporters) and those involving third parties who relate to the conflict as an agent of settlement. In the first

deviant behavior. Black (1976, pp. 4–6; 1984, pp. 8–12) identifies four styles of law: (1) penal, (2) therapeutic, (3) compensatory, and (4) conciliatory.<sup>302</sup> The *penal* style of law (seen in criminal law), the *therapeutic* style (seen in juvenile law and psychiatric care), the *compensatory* style (seen in tort and contract law), and the *conciliatory* style (seen in marital and labor law) all differ from one another along several dimensions. Each style has its own standards, questions, and solutions. Furthermore, each style attributes a different identity to the deviant and has its own focus (see Black 1993). For example, in the penal style, the deviant is regarded as an offender who has violated a prohibition and who should suffer punishment, such as deprivation, humiliation, or even loss of life. The focus of penal law is typically not the offender as such, but the conduct itself—a specific act. In contrast, the therapeutic style views the deviant as a victim who is ill and needs to receive help or assistance. The focus of this style is the victim, rather than a specific act. Similarly, compensatory and conciliatory styles differ from penal and therapeutic styles,

category involving principles (and their supporters), law may be unilateral, flowing in a single direction from the aggrieved to the offender party, or bilateral, flowing in both directions where both parties pursue a grievance against one another (e.g., a duel or fight). In the second category, law is trilateral, requiring the intervention of a settlement agent who relates authoritatively to both sides (e.g., judges, mediators) (see also Black and Baumgartner 1983; Cooney 1998; Phillips and Cooney 2005).

The form of social control, however, did not initially appear in Black's theory of law. In fact, formas a variable aspect of social control systems—did not figure prominently in Black's writings until he began expanding his theory to non-legal forms of social control (see, generally, Black 1984, 1990). Nonetheless, Black (2002a, p. 111) views the law as *inherently* partisan because whenever a legal official or jury decides who "wins and loses" in a particular dispute, then the law has taken sides. Law does not take sides from the beginning; a legal trial must determine the state's partisanship. Because the process by which law takes sides typically takes time, Black refers to it as slow partisanship. Although Black believes that law is form of partisanship, he is not referring to the subjective experience or motivation of legal decision-makers. In fact, judges and juries may typically endorse the jurisprudential idea that law is autonomous and neutral. "[T]hose accustomed to conventional legal thought may have difficulty understanding the idea of law as a form of partisanship. Some might even find it outrageously wrong and ignorant...[but] the partisanship of legal officials should not be surprising. It is the primary role of third parties who intervene in conflicts of all kinds" (p. 112). For Black all that matters, from a sociological standpoint, is what legal officials actually do: they take sides. His theory, then, predicts and explains who take sides in a conflict-including conflicts involving law: partisan is a direct function of the social closeness and superiority of one side and the social remoteness and inferiority of the other (1993, p. 127). <sup>302</sup> According to Black (1993, p. 7) and others (e.g., Horwitz 1990, pp. 19–22), these four styles apply to both legal and non-legal types social control.

and one other, focusing on the consequences of the act and the relationship, respectively (Black 1984, p. 11).<sup>303</sup>

According to Black (1976) variation in social life, including law, can be explained by the shape of social space where it occurs—its *social geometry* (i.e., its social structure) (but cf. Frankford 1995; Luhmann 1989).<sup>304</sup> Black's theory of law predicts and explains the variable aspects of law—its style and quantity—based on the locations, directions, and distances of conflict in a multidimensional social space that is defined by "the characteristics of the people involved, their relationships with each other, and the larger social context in which they interact" (1979a, p. 19).<sup>305</sup> Black's multidimensional conception of social space is, in fact, a synthesis diverse sociological theories and concepts that have a long tradition in sociological thought (see also Horwitz 1983, p. 376; Tucker 1999, p. 6; Turner 2002, p. 664). He identifies five dimensions that define human interaction: (1) vertical, (2) horizontal, (3) symbolic, (4) corporate, and (5) normative. According to Klüver and Schmidt (1999, p. 311), "The number of dimensions of a 'space of experiences' is the number of independent options one has to take into consideration for a complete description of any experience [and that cannot be defined by combinations of other dimensions]...[g]eneral descriptions of social experiences are not dependent on the particularities of specific social groups or cultures." These five dimensions are discussed, in turn, below.

<sup>&</sup>lt;sup>303</sup> The different styles of social control (including law) also vary in severity, with penal control being the most severe, followed by therapeutic, compensatory, and conciliatory control (Black 1976, p. 106). <sup>304</sup> Klüver and Schmidt (1999, p. 313) suggest that it is possible to define the main concepts of theoretical sociology in geometric terms and make geometric models of social action determined by a network of rule governed interactions (cf. Caplow 1955).

<sup>&</sup>lt;sup>305</sup> Similarly, Wilson (1996) defines social structure as "the ordering of social positions (or statuses) and networks of social relationships that are based on the arrangement of mutually dependent institutions (economy, polity, family, education) of society" (pp. xiii–xiv; see also Whitmeyer 1994, p. 154).

*Vertical Space.* A vertical dimension of social space, emphasized by Marx ([1859] 1970, [1867] 1975), Tocqueville ([1840] 1969) and Weber ([1922] 1964), is present when there is an uneven distribution of wealth (i.e., social stratification).<sup>306</sup> A social phenomenon may be described by its vertical location, whether higher or lower in such a distribution. Social life may also have a vertical direction, moving upward (from a lower to higher elevation), downward (from a higher to lower elevation), or laterally (across the same elevation) (see also Tucker 2002). Finally, the difference in wealth itself, or vertical distance, varies as well. "Whether higher or lower in a distribution of wealth, or downward or upward in its direction, social phenomena may span a greater or lesser distance in vertical space" (Black 1993, p. 160).<sup>307</sup>

Law, according to Black, varies directly with rank (i.e., its location in vertical space). "This means that, all else constant, the lower ranks have less law than the higher ranks, and the higher or lower they are, the more or less they have" (1976, p. 17). Law varies with its direction in vertical space, such that downward law is greater than upward law. Again, all else constant, a complaint against someone of a lesser rank is likely to attract more law than a complaint against someone of a higher rank (see also Marx [1841]

<sup>&</sup>lt;sup>306</sup> Black's (1976, p. 11) conceptualization of "wealth" includes the material conditions of existence (e.g., food and shelter), the means by which these conditions are produced (e.g., land and raw materials), other property and luxuries that may be exchanged for the conditions of existence (e.g., money and livestock), and the ability to borrow wealth (i.e., credit). The various kinds of wealth in a society may be differentially distributed, so an individual or group may have a number of different ranks. It is possible, however, to combine these various ranks so that each individual or group has a general rank relative to all others (p. 16). Bonilla-Silva (1997, p. 469 n.5) offers a broader definition of "material": the economic, social, political, or ideological rewards or penalties received by social actors for their participation in social structural arrangements. Although Black's definition encompasses a narrower range of phenomena than Bonilla-Silva's, his conceptualization is more consistent with his emphasis on the observable aspects of social reality.

<sup>&</sup>lt;sup>307</sup> There are other variable aspects of stratification, as well. For example, the degree to which wealth is distributed in layers, the number of layers, and size of these layers are all variable. Other variable aspects may include the mechanisms of wealth distribution (e.g., inheritance, occupation, and gambling), the movement of people from one layer to another (vertical mobility), and the age of wealth itself (e.g., "old money" versus "new money") (Black 1976, pp. 11–12).

1975). This implies, then, that upward deviance is more serious than downward deviance. Finally, among cases with a direction between one rank and another, the difference between the ranks (i.e., vertical distance) predicts the quantity of law; however this depends on the direction of law in each case. Law with a downward direction varies directly with vertical distance. Upward law, however, varies inversely with vertical distance. For example, all else constant, the seriousness of an offense by a lower against a higher rank increases with the difference in wealth between the parties. Conversely, the seriousness of an offense by a higher against a lower rank decreases as this difference increases (Black 1976, pp. 24–25).

As noted above, the social geometry of a conflict also predicts the style of law (see also Seron and Munger 1996, p. 204). For example, downward law is more penal than upward law. "[When] the offender's rank is below the victim's, his [or her] conduct is likely to be punished as a crime than in a case where the direction is the opposite" (Black 1976, p. 29). Upward law is more compensatory than downward law, so an offender who ranks above her or his victim is more likely to be asked to pay for her or his damage than in a case with the opposite direction (Black 1987). Similarly, upward law is more therapeutic than downward law. In situations where an offender is of higher rank than the victim, the offender is more likely to be defined as sick and in need of treatment, whereas an offender is more likely to be punished in a conflict with the opposite direction. The style of law also varies with vertical distance. In a downward direction, penal law varies directly with vertical distance, whereas it varies inversely with vertical distance in an upward direction. The likelihood of compensatory and therapeutic law also increases as the wealth of a higher rank offender increases. Conciliatory law, on the other hand, decreases as stratification increases and is more likely to occur when parities are of equal rank (Black 1993, p. 163).

*Horizontal Space*. The horizontal dimension of social space (also referred to as morphological space) arises from the distribution of people to one another. This dimension, highlighted by Durkheim ([1897] 1951), Spencer (1876), and Simmel ([1908] 1950), includes integration, interdependence, and differentiation.<sup>308</sup> Black notes that every activity has a circle of participation, with people closer or further from the center, and therefore has a radial location. It may also have a radial direction, moving inward (centripetally) or outward (centrifugally) from the center (Black 1976, p. 49), as when a marginal individual makes a legal complaint against an individual who is more integrated. As with vertical space, there is also a radial distance—the difference in integration itself.<sup>309</sup>

Horizontal space has another variable aspect as well—the structure of intimacy (or relational distance). This is measured by the degree to which individuals participate in one another's lives, including the scope, frequency, and duration of their interaction and their linkages in a wider network (see Bonilla-Silva 1997, pp. 469–470; Caplow 1955, pp. 29–30; Sutherland 1947, pp. 6–7).<sup>310</sup>

Law varies directly with integration (i.e., its location in radial space). People in or near the center of social life have more law than those more marginal. For example, an

<sup>309</sup> Differentiation also varies across settings of every kind, including friendships, families, organizations, and societies. It is important to note that differentiation is not simply a function of the size of the population, as smaller groups may display more division of labor than larger groups (Black 1976, p. 38). <sup>310</sup> Structural analysis, or social network analysis, entails the representation of actors and/or objects linked together by social connections (e.g., two persons are friends or enemies) or shared experience (e.g., they go to the same school) (Edling 2002, p. 206). Much of the social network literature has been criticized, however, for being theoretically underdeveloped. Black avoids this problem by harnessing the explanatory power of structural analysis under a single social dimension—morphological space.

<sup>&</sup>lt;sup>308</sup> Differentiation (also know as division of labor) is a specialization of function across the parts of a whole (Durkheim [1893] 1947).

offense between two employed individuals is more likely to result in legal action than an offense between two individuals who are unemployed. Conversely, a conflict between two vagrants is less likely to be reported and even less likely to result in arrest. So, all else constant, an offense between marginal people attracts less law than an offense between people more integrated into social life. Radial direction is important as well. Centrifugal law (inward law) is greater than centripetal law (outward law). An offense of a marginal person or group against an integrated person or group is more serious (i.e., attracts more law) than an offense in the opposite direction.<sup>311</sup> Integration, of course, is a matter of degree, so the difference between people in horizontal space also predicts and explains legal variation. Inward law varies directly with radial distance, whereas outward law varies inversely with radial distance.

Law varies directly with the division of labor, to a point, then reverses itself: the relationship between law and differentiation is curvilinear (Black 1976, p. 39). Law increases with differentiation to a point of interdependence, but declines with symbiosis.<sup>312</sup> There is very little law at both extremes of differentiation: when people are undifferentiated by function and when people are completely dependent upon each other. The relationship between law and relational distance (i.e., intimacy) is also curvilinear. Law is relatively inactive among intimates and increases as the distance between people

<sup>&</sup>lt;sup>311</sup> Similar to wealth, another variable aspect of integration is its stability (or, conversely, its volatility) i.e., its age (see Bausman and Goe 2004). So, for example, legal action is more likely to occur among employed individuals with stable work histories than employed individuals with unstable work histories, all else constant. And by extension, an offense of an individual with a volatile employment history against an individual with a stable work history attracts more law than the reverse, all else being equal.

<sup>&</sup>lt;sup>312</sup> Unlike Black (1976), most theorists fail to specify the functional form of the interrelations between properties (for a discussion, see Thaxton and Agnew 2004). "[F]ew of our propositions ever state the exact function—which is one of the reasons why [sociology] is not an exact science" (Homans 1967, p. 20) and "rarer still is guidance about the specific functional form of any *a priori* nonlinearities" (Beck and Jackman 1998, p. 597). This "specification error" both undermines the theorist's ability to make precise predictions and the analyst's ability to accurately test hypotheses derived from the propositions (see Berk 2003, p. 94; Gujarati 1988, pp. 455–56).

increases until this distance reaches the point which people begin living in entirely separate worlds, at which time it decreases. Relational distance rarely reaches the point where people are entirely separate in modern society, so this relationship is basically linear; although in the past that was not always the case (Black 1976, p. 41). The shape of horizontal space, similarly, predicts and explains the style of law. For example, inward law (applied against someone who is more integrated) is more compensatory, whereas outward law (applied against someone who is more marginal) more penal.

Symbolic Space. Drawing heavily on the work of Hegel ([1821] 1952), Parsons (1951), and Sorokin (1937), Black identifies a symbolic dimension of social space, representing the expressive aspect of social life, or culture, whether moral, intellectual, or aesthetic (Black 1976, p. 61). It includes religion, ceremony, values, ideas, language, et *cetera*. Culture, too, is quantifiable. Culture has two variable aspects: content and conventionality. Its content is measured by the number of different languages, concepts, ideas, and in the volume of religions, folklore, science, values, customs, clothing, et *cetera* (p. 63). The quantity of culture is unevenly distributed both across and within societies. Black also argues that culture can be more or less conventional, measured by its frequency.<sup>313</sup> Conventionality is unevenly distributed across social settings, and social life varies with its location in symbolic space. Social life may also vary with its direction and distance symbolic space, from more to less (or less to more) conventionality. The value of Black's approach to culture, then, is that differences in what are traditionally seen as strictly qualitative phenomena can, in fact, be measured (see also Cerulo 2002; Mark 1998).

<sup>&</sup>lt;sup>313</sup> Ginzburg famously wrote "[i]t is significant that we say that something is 'high' or 'superior'—or conversely 'base' or 'inferior'—without considering why what we most praise (goodness, strength, and so on) must be located high" (1976, p. 31).

Law varies with both aspects of culture: its content and conventionality. Law varies directly with the content of culture. Where culture is sparse, so is law; where it is rich, law abounds. Law is also greater in a direction towards less culture than toward more culture. An offense by someone with less culture (e.g., education) than his or her victim attracts more law than an offense in the opposite direction. The magnitude of the difference in culture is also important. In a direction towards less culture, law varies directly with cultural distance. Conversely, in a direction toward more culture, law varies inversely with cultural distance.

Law also increases as it nears mainstream culture and decreases as it moves away: law varies directly with conventionality. Law is also greater in a direction towards less conventionality than towards more conventionality. The magnitude of this difference, again, is important. Downward law varies directly with cultural distance whereas upward law varies inversely with cultural distance. Cultural distance also predicts and explains the style of law. All else constant, penal law varies directly with cultural distance whereas conciliatory law varies inversely with cultural distance. The conditions for compensatory and therapeutic law, however, lie between those conditions most conducive to penal and conciliatory law (Black 1976, pp. 78–79).

*Corporate Space*. The corporate dimension of social space, emphasized by Weber ([1922] 1964, [1922] 1968) and Michels ([1911] 1962), refers to the capacity for collective action, or organization (see also Dowding 1996; Sampson 2006). This is found in all groups, whether a married couple or conglomeration of nations (e.g., NATO). The quantity (or level) of organization varies across social settings and legal behavior may be described by its location in such a distribution. Social life may also have an upward or downward direction in corporate space. Additionally, a social phenomenon may span greater distances of organization.

The capacity for collective action predicts and explains the quantity of law: law varies directly with organization. A conflict between organizations attracts more law than a conflict between individuals. "Between organizations, the loser loses more, an appeal by the plaintiff is more likely, and so is a reversal on his [or her] behalf" (Black 1976, p. 92). The direction of a conflict in corporate space also predicts and explains legal variation. Downward law (i.e., in a direction toward less organization) is greater than upward law (i.e., toward more organization). As with other dimensions, the magnitude of the difference predicts and explains legal variation. Downward law varies directly with organizational distance, whereas upward law varies inversely with organizational distance, is greater in a direction toward less organization than toward more. Compensatory and conciliatory law, on the other hand, is greater in a direction toward more organization.

*Normative Space.* The normative dimension, stressed by Ross ([1901] 1920), Cooley ([1902] 1964), and Sumner ([1906] 1940), is the final dimension of social space identified by Black. Normative space results from the operation of social control (Black

<sup>&</sup>lt;sup>314</sup> Hagan (1999, p. 364) suggests that the modern criminal justice system better serves corporate than individual interests, and agencies originally thought to have emerged for the protection of individuals against other individuals are currently devoting a significant share of their resources to the protection of large affluent corporate actors. He also posits that corporate domination over our everyday lives extends far beyond criminal law. Songer and Sheehan (1999, p. 339) also discover that parties with greater organizational status are significantly more likely to prevail in federal appellate courts. Examining both published and unpublished opinions from the Fourth, Seventh, and Eleventh circuits (N = 4,281), and controlling for area of law (criminal, civil liberties, economic regulation and labor relations, and diversity of citizenship), political composition of the panel of judges, and region, they discover that big businesses enjoy greater success over smaller business and the federal government experiences greater success over state and local governments.

1979b, p. 161). Just as social control defines who is deviant, it defines who is respectable as well (Black 1976, p. 111).<sup>315</sup> As such, normative space includes the respectability of individuals and groups: the degree to which they have been subject to social control in the past (see, e.g., Braithwaite 1989), *and* their authority—that is, the capacity for social control (see, e.g., Shaw and McKay 1942).<sup>316</sup> The quantity of social control varies across social settings. While law, itself, is a form of social control, other varieties of non-legal social control exist as well, such as violence, covert retaliation (e.g., vandalism and sabotage), avoidance, negotiation, toleration, gossip, and criticism (see Baumgartner 1984; Black 1990, 2000b; Ellickson 1991; Ewick and Silbey 2003; Levi 1980; Morrill, Zald, and Rao 2003).<sup>317</sup> As with other dimensions of social space, law varies according to its location, direction, and distance in normative space. Social life varies with different social locations of authority and respectability, may have either an upward or downward direction (e.g., from more to less respectability or authority), and may span greater or lesser distances in normative space.

<sup>&</sup>lt;sup>315</sup> Ziegert (1980, p. 60) argues that the history of humankind has been the history over the struggle for social control.

<sup>&</sup>lt;sup>316</sup> According to Louis Dumont (1981, p. 65), "[H]ierarchy is a ladder of *command* in which lower ranks are encompassed in the higher ones in regular succession...[h]ence it is a question of systematically graduated authority." Although not explicitly mentioned by Black, the capacity for respectability may be forward-looking as well: the degree to which a person or group can be viewed as moral or reputable. Irrespective of previous conduct, some individuals and groups hold the power of presumption while others do not (see Kan and Phillips 2003; Pager 2007). It is reasonable to believe, then, that members of seriously stigmatized groups may never be able to achieve similar levels of normative status to non-stigmatized groups, irrespective of their past, current, and future behavior (Cose 1993). According to Patterson (1982), it is impossible to understand the legacy of slavery and the social meaning of race in American without grasping the importance of honor. He notes that slavery is "the permanent, violent domination of natally alienated and generally dishonored persons," (p. 13) and suggests that the termination of the legal status of "slave," in itself, was insufficient to make slaves and their descendants into full members of society. <sup>317</sup> These other types of social control, however, may *not* necessarily behave according to the same principles as law, so they should be investigated as well (Black 1984, p. 16). Although not initially characterized as such, Merton's (1938) five adaptations to blocked legitimate opportunities for monetary success—conformity, innovation, ritualism, retreatism, and rebellion—are primary examples of the major forms of non-legal social control employed by social inferiors against social superiors. According to Baumgartner (1984), upward social control is commonly handled through more "passive" methods, primarily because there is usually a higher cost associated with exercising social control from below (see also Black 1992).

Law is stronger where other social control is weaker: it varies inversely with social control (see, e.g., Clear 2007; Fagan and Meares 2000; Pound 1942; but cf. Liska 1997). Where other types of non-legal social control are plentiful, law relaxes (Black 1976, p. 110). Law also behaves according to its location in normative space: law varies directly with respectability. All else constant, unrespectable people among themselves have less law than more respectable people. For example, ex-convicts are less likely to complain to the police about each other than are respectable people. Even when unrespectable people invoke law against one another, they are less likely to succeed (see Anderson 1999; Black 1983; Venkatesh 2006). Conflicts also have a normative direction if the offender is more or less respectable than her or his victim. Law is greater in a direction toward less respectability than toward more respectability. The less respectable person or group is more likely to be subject to law but less likely to have its benefits. The difference in respectability is a normative distance. Downward law (toward less respectability) varies directly with normative distance. Upward law, on the other hand, varies inversely with normative distance. "The less respectable an offender in relation to a complainant, a legal official, the members of a jury, or a witness, for instance, the more law to which he [or she] is likely to be subject" (Black 1976, p. 117). This principle applies to the complainant as well. That is, law also varies directly with the respectability of the victim (see Baumer et al. 2000; Chesney-Lind 1989; Sundby 2003). Any record of social control is a disadvantage, with a criminal record clearly being the worst.<sup>318</sup>

<sup>&</sup>lt;sup>318</sup> Schwartz and Skolnick's (1962) seminal study of the impact a criminal record on employment opportunities revealed that *any* contact with the criminal justice system reduced the likelihood of employment. Specifically, they sent four sets of resumes to prospective employers and varied the criminal record of applicants: (1) conviction and sentence for assault; (2) trial and acquittal for assault charge; (3) trial and acquittal for assault accompanied by a letter from the trial judge certifying the applicant's acquittal and innocence; and (4) no criminal charge or record. In all three "criminal" conditions, applicants were

Unlike physical distance, social distance (e.g., vertical distance, relational distance, cultural distance) is two-directional—measurable from both *A* to *B* and *B* to *A*; therefore it may be unequal in each direction (i.e., asymmetrical) (Black 2000a). For example, *A* may be relationally closer to *B* than *B* is to *A* if *A* participates more in *B*'s life than *B* participates in *A*'s life (cf. Gould 2002). Similarly, an asymmetrical distance in cultural space may arise if *A* speaks *B*'s native language, but *B* cannot speak *A*'s native language (Black 2000a, p. 349 n.18).<sup>319</sup>

In sum, Black's multidimensional conception of social space draws from five major aspects of social life that have long been emphasized in sociological theory (see also Collins 1994). By incorporating what is valuable from these various perspectives and discarding their peculiarities, Black demonstrates how these seemingly divergent approaches can contribute to a single body of theory. He recognizes, however, that other dimensions of social space may also be added to the theory if necessary and argues that his theory can readily accommodate these additions.<sup>320</sup> Black argues that law, like all other forms of social life, varies with its social geometry: its location, direction, and

less likely to be considered by employees relative to the group with no record of a criminal charge or conviction.

<sup>&</sup>lt;sup>319</sup> It is possible to define mathematically concepts like social nearness, degree of social cohesion, and so forth (Klüver and Schmidt 1999).

<sup>&</sup>lt;sup>320</sup> Villarreal (2002, p. 479 n.5), for example, argues that Black's framework neglects the "web of obligations" in which individuals are immersed and how this influences social control. According to Villarreal, differences in status dimensions identified by Black (e.g., stratification and structure of intimacy) do not explain the power that patrons have over their clients; rather this power derives from the unequal relations of exchange in which clients are forced to engaged (see, e.g., Emerson 1962; Lindblom 1977).

It is doubtful, however, that Villarreal's "web of obligations" constitutes a new or distinct dimension of social life; rather it can be subsumed under one or more of Black's existing dimensions. In fact, Black's (1976, p. 40) *horizontal space* (i.e., morphology) specifically addresses non-mutual dependency resulting from the uneven distribution of differentiation (i.e., division of labor) and interdependence in any given setting. Similarly, his conceptualization of two-directional, and potentially asymmetrical, social distances (see above) incorporates unequal exchange relations (see Black 2000a, p. 349).

distance in multidimensional social space (cf. Tomlins 2007).<sup>321</sup> Furthermore, he develops numerous propositions specifying how law varies along these various dimensions. That is, how variations in law are linked to variations in vertical, horizontal, symbolic, corporate, and normative space (see also Timasheff 1937, p. 227).<sup>322</sup> It must be emphasized, however, that Black's theory of law extracts bivariate relationships from multivariate situations (Horwitz 1990, p. x). In discussing specific relationships, Black treats all non-sociological variables, and other dimensions of social space identified in the theory, as constants. This is done in order to illustrate the basic properties of these relationships under ideal conditions when "all else is equal." Although some have criticized this approach as unrealistic (see Greenberg 1983b; Hunt 1983), it is nearly ubiquitous in science. Universal laws in the physical and biological sciences are nearly always established under ideal conditions that do not exist in the natural world and become probabilistic or statistical generalizations when applied outside of these ideal conditions (Nagel 1979, pp. 508–509). Similarly, Weber ([1922] 1968) noted that his "ideal types" were unlikely to be found in pure form in reality, but nonetheless allowed the analyst to study the real world. Social theory using the covering law approach casts predictions in probabilistic terms because it involves populations that are subject to many more influences than included in the theory and available in the data (Lenski 1988, p. 168). Once the fundamental logic of the theory has been articulated through these

<sup>&</sup>lt;sup>321</sup> Similarly, Hollingshead (1941, pp. 220, 222) notes, " the essence of social control is to be sought in the organization of people" and "[the interest of social control theorists] is to search for, define, and analyze the organizational systems' functioning in a culture to determine how these regulate the behavior of the person."

<sup>&</sup>lt;sup>322</sup> Luhmann (1989, p. 139) also recognized the relationship between the law and the other dimensions of social life: "Law is not politics and not the economy, not religion and not education; it produces no works of art, cures no illnesses, and disseminates no news, although it could not exist if all of this did not go on too."

bivariate associations (and subject to empirical tests), more complex (multivariate) relationships can be explored (see, e.g., Lessan and Sheley 1992; Sherman 1978).<sup>323</sup>

## 6.2.3 Evaluation of the Blackian Paradigm

Harris (1979) has argued that research strategies can be evaluated and compared even prior to the examination of their subsequent theories by their focus on discovering orderly relationships.<sup>324</sup> Research strategies that aim to discover the maximum amount of order in the universe are superior to competitors that fail to do so (Maxwell 1974a,b). As Maxwell (1974a, p. 152) notes, "we can assess in an *a priori* fashion the relative simplicity or intelligibility of rival paradigms [by the] promises which they hold out of realizing the basic metaphysical blueprint of the science" (cf. Somers 1998, pp. 758–61). Black (1995, p. 847) argues that his research strategy is scientifically superior to earlier research programs in sociology because it avoids several of their shortcomings, namely: (1) psychology, (2) one-dimensionality, (3) units of analysis, (4) individualism, and (5) teleology (see also 2000c, p. 705; 2002b, p. 668).

*Psychology.* As discussed above, Black's reconceptualization of individual action as social action allows for the explanation of social phenomena without regard to psychology. Conversely, nearly all sociological theory, both classical and modern, either explicitly or implicitly addresses human subjectivity (Black 1995, p. 848; Mayhew 1980, p. 335), although Durkheim ([1895] 1962) initially emphasized developing explanations operating solely at the sociological level without recourse to psychological processes and

<sup>&</sup>lt;sup>323</sup> The location of an individual or group in social space, relative to others, is a status (Black 1979b, p. 161). Fisek (1998) provides a thorough discussion of important issues concerning the combination of positive and negative status characteristics. Although his work is concerned with the social psychological theory of status characteristics and expectation states, many of the points raised in his work are relevant to Black's multidimensional conceptualization of social space.

<sup>&</sup>lt;sup>324</sup> Lakatos (1970, p. 155) argues that the history of science has been (and should continue to be) the history of competing research programs (see also Cohen 1985).

Simmel (1909) originally insisted on abstracting from *concrete* phenomena (Turner 2002, p. 664).<sup>325</sup> For example, much of sociology explains human behavior with the psychological impact of the social environment, making motives and meanings central (Black 2000a, p. 345). Black's paradigm, however, makes no assumptions or assertions about the goals, preferences, needs, values, attitudes, cognitions, or interests of individuals (but cf. Durkheim [1897] 1951; Weber [1922] 1968).<sup>326</sup> In fact, his paradigm entirely eliminates the individual from its formulations (see Black 1979b, 2000c). According to Black (2000a, p. 347), social phenomena, such as law, religion, and violence, cannot be explained by their psychological characteristics because they have no mind, no thoughts, and no subjectivity—they simply behave (i.e., vary). It is because the "unpsychological" nature of his approach, that Black believes his work is able to achieve a higher degree of scienticity than any previous sociological research strategy. His formulations require no psychological knowledge about individuals, therefore allowing his theories (e.g., his theory of law) to be tested by outward observation and direct measurement (see also Cooney 2002). Similarly, the removal of psychology from his paradigm permits his formulations to be highly general and parsimonious. Black (1995, p. 850) argues that highly general and parsimonious theory would be extremely difficult—if not entirely impossible—if it must address the nature of human subjectivity

<sup>&</sup>lt;sup>325</sup> "The classic sociologist most famous for insisting that sociology is different from psychology—Emile Durkheim—also addresses the subjectivity of goal-seeking individual. He psychologizes virtually every subject, even society[....] If Durkheimian sociology is not psychological, then Durkheim is not Durkheimian" (Black 2000a, p. 344).

Black (2002a, p. 102) rhetorically asks, "If it is possible to have psychology without biology, can we not have sociology without psychology as well?"

<sup>&</sup>lt;sup>326</sup> "Social science explanations usually have a motivational dimension but that offers no sound basis for holding that they must have such a dimension. What is a 'good reason' relates more to the psychological than to the logical aspects of theoretical explanation: it is a fact about the expectations of the consumers of social science rather than about the necessary ingredients of social science theories" (Cooney 1986, p. 269). Similarly, Klüver and Schmidt (1999, p. 322) remark that it is unnecessary for sociologists to introduce particular 'interests' of social actors in their explanations of human behavior.

(i.e., how everyone thinks and feels) in all situations, societies, and historical periods to which they apply (see also Nagel 1979, p. 542). "Moreover, since motivations and meanings are commonly said to be rooted in particular settings, passing over them in silence allows the theorist to focus on what is common about behavior, regardless of whether it occurs in ancient India, medieval Venice, pre-colonial Zimbabwe, or contemporary America" (Cooney 2002, p. 660).<sup>327</sup>

*One-Dimensionality*. Unlike earlier sociological traditions that attempt to explain social life solely with the distribution of wealth (Marx and Engels [1846] 1947) or the distribution of intimacy, interdependence, and integration (Durkheim [1893] 1947; Simmel [1908] 1964) or the distribution of culture (Bourdieu [1979] 1984) (i.e., one-dimensionality), Black's paradigm synthesizes these research strategies into his multidimensional conception of social space, "incorporating and harnessing the explanatory power of diverse theories and variables" (Black 1995, p. 851).<sup>328</sup> Although previous attempts at "paradigm mixing" in sociology have been unsuccessful because of the "substantial differences in the underlying assumptions about the nature of human beings, society, and sociology" (Perdue 1986, p. 271; see also Tittle 1995, Chapter 4), Black's synthesis is successful because his paradigm discards the many of problematic

<sup>&</sup>lt;sup>327</sup> For example, Michalski (2004, p. 652) notes, "The preoccupation with the psychology of violence and the focus on cultural orientations obscure the more salient features of social life that promote violence: the structure of interpersonal relationships."

<sup>&</sup>lt;sup>328</sup> Weber developed a multidimensional theory of stratification that incorporated class, status, and party (see Collins 1994, pp. 81–91). Classes represent the economic order and are defined in terms of market situation (e.g., class position, economic interests, life chances); status groups represent the social order and are determined by the distribution of social honor (e.g., titles, awards, and styles of life); and parties represent the legal/political order and usually pursue class or status interests. Weber argued that classes, status groups, and parties are phenomena of the distribution of *power* within a community (Gerth and Mills 1958, pp. 180–95). Weber ([1922] 1968, p. 28) defines power as the "probability that an actor realizes his will in a social relationship even against the will of someone else." His three components of power—class, status, and party—are very similar to Black's vertical, symbolic, and corporate spheres, respectively. Moreover, social power, itself, is very similar to Black's normative sphere: the distribution of authority. Unlike Black, however, Weber emphasizes the importance of power and domination over all other spheres, relegating these other dimensions to sources of social power.

assumptions that plagued these earlier approaches that made them (seemingly) incompatible with one another (e.g., Berger and Luckmann 1966; Parsons 1951). Black's synthesis avoids the problem of *eclecticism*, defined as the practice of "picking and choosing epistemological and theoretical principles to suit the convenience of each puzzle" (Harris 1979, pp. 287–88). Harris believes that "eclectism guarantees that...solutions will remain unrelated to each other by any coherent set of principles...[and] cannot lead to the production of a corpus of theories satisfying the criteria of parsimony and coherence" (p. 288). Eclectism typically results in theories that do not interpenetrate each other or that are incompatible, primarily because eclectic theories are capable of generating numerous contradictory hypotheses (see, e.g., Parsons and Shils 1951). While eclectic theorists may genuinely advocate the importance of the interplay between theory and research, they fail to develop and refine core principles "capable of directing research efforts consistently along lines that could conceivably produce a coherent corpus of interpenetrating theories" (Harris 1979, p. 291). By developing a highly general, parsimonious, and coherent theoretical framework—pure sociology—Black's synthesis results in single body of theory and avoids the pitfalls of eclecticism: "middle-range theories, contradictory theories, and unparsimonious theories without end" (p. 288).<sup>329</sup>

<sup>&</sup>lt;sup>329</sup> Harris remarked, "[E]clecticism is a prescription of perpetual scientific disaster" (p. 288). He notes, however, that eclecticism should not be confused with substantive or methodological versatility (p. 289). In fact, eclectic theories are seldom able to adequately explain the wide-range of phenomena they propose because of their contradictory assumptions and predictions (but cf. Stinchcombe 1968, pp. 4–6). Furthermore, "[t]he choice of methodology is an issue that is entirely separate from the choice of epistemological or theoretical principles. Methodologies are means one employs to test hypotheses and theories" (Harris 1979, p. 290).

Black explicitly rejects the theoretical supremacy of any dimension of social space over the other.<sup>330</sup> According to Black (2000a, pp. 354–55), it is impossible to rank the explanatory power of the various dimensions of social space because these dimensions lack a common unit of measurement (but see Bernard 1995, 2002). While he recognizes that it is possible, for the practical purposes of a single study, to rank the various dimensions because the comparisons reflect their measurement in one context, the dimensions cannot be ranked in a theory that applies across time and space (see also Kim and Ferree 1981).

Black's multidimensional conceptualization of social space greatly increases the comprehensiveness and generality of his work, while still remaining parsimonious and testable. Social life occurs in all societies throughout time, some of which might lack variability in vertical, horizontal, symbolic, organizational, or normative space. Theoretical approaches that attempt to explain social variation solely as a result of one dimension of social space are not only incomplete, but will have limited application in societies that are invariant along that dimension. Black's paradigm, however, identifies variability along all these dimensions, therefore tremendously broadening the scope and application of his formulations (see Black 2000a, p. 355).<sup>331</sup>

*Units of Analysis.* Sociological theories attempting to explain human behavior tend to focus on the microcosm (e.g., a person in a particular situation) or the macrocosm (e.g., a larger formation such as a society, region, or community), and as a result,

<sup>&</sup>lt;sup>330</sup> "Economics, the division of labor, networks, culture, organization, and social control have long vied for the pole position in the race to explain social life. In pure sociology none of these contenders wins the prize for being the ultimate explanation; they all share it equally. Every dimension of social space is as fundamental as every other" (Cooney 2006a, p. 56).

<sup>&</sup>lt;sup>331</sup> Cooney (2006a, p. 56) emphasizes that "pure sociology synthesizes...meta-theoretical systems paradigms of thought that can and have been applied to a wide variety of subjects."

explanations derive from the characteristics of these units (Collins 1981).<sup>332</sup> Black (1995, p. 858) argues, however, that neither persons nor societies are sources of human behavior. "Microcosms overpersonalize everything, and macrocosms oversocietalize everything." Black's shape of social space, which explains human behavior, is neither large nor small. Its size, boundaries, and duration are all variable (p. 853). Because the shape of social space is defined and measured by the social characteristics of everyone involved in every instance of human behavior (see also Goffman 1971), each has its own multidimensional location, direction, and distance in social space.

According to Black (1995), the shape of social space provides a better explanation than microcosms and macrocosms "because the precise location of social life is not persons or societies" and "to understand such phenomena as a consequence of persons or societies is inherently and incurably limited as an explanatory strategy" (pp. 857–58). He argues that people who use violence or law or religion are not violent, litigious, or religious in all social settings in which they participate. Similarly, violent or litigious or religious societies are not violent or litigious or religious in all their settings. Only particular conflicts, with particular locations, directions, and distances in social space, attract violence or law or religion (p. 857). Black suggests that the shape of social space is superior to every unit of analysis in sociology (e.g., a person, organization, or society) because it is both observable and measurable everywhere. The unit of analysis, then, disappears in social space (see also Black 2000a, p. 347 n.9).<sup>333</sup>

<sup>&</sup>lt;sup>332</sup> According to Somers (1998, p. 750), "all theories of knowledge make a more or less explicit ontological choice between either the individual or the social structure as the basic unit of analysis."

<sup>&</sup>lt;sup>333</sup> As Jasso (2006, p. 38) notes, a theory's predictions should span *all* levels of analysis.

Individualism. The subject matter of Black's paradigm is not the individual, rather it is the behavior of social life (i.e., social variation).<sup>334</sup> Social action replaces individual action and people disappear (cf. Parsons 1937). Human behavior, then, becomes a characteristic of social beings, rather than a characteristic of human beings with their own propensities.<sup>335</sup> The elimination of people from Black's paradigm greatly simplifies his explanation of human reality: the requirement of an understanding of human beings involved is replaced with a requirement of an understanding of the social beings involved (Black 1995, p. 860).<sup>336</sup> For example, his theory of law takes legal variation as its subject matter, not the behavior of people. While most socio-legal theorists attempt to explain diverse legal behavior such as calls to the police, arrests, lawsuits, verdicts, and appeals with the behavior of citizens, police, attorneys, and judges (see, e.g., Frankford 1995), Black's theory only requires an understanding of one single phenomenon-the behavior of law-which obeys the same principles throughout the social universe.<sup>337</sup> Under Black's framework, law is a natural phenomenon with its own patterns of behavior. Where the social structure of conflict varies, law varies. Where the social structure of conflict remains the same, law behaves the same. In Black's paradigm, it is the shape of social space, not people, that matters (Black 1995, p. 861).<sup>338</sup>

<sup>&</sup>lt;sup>334</sup> "Pure sociology is non-anthropocentric. Explanations make no reference to human beings as such. Social life behaves and people are merely its carriers" (Cooney 2006a, p. 53).

<sup>&</sup>lt;sup>335</sup> Popper (1964) argues that methodological individualism is "the quite unassailable doctrine that we must try to understand all collective phenomena as due to the actions, interactions, aims, hopes, and thoughts of individual men [*sic*], and as due to traditions created and preserved by individual men" (pp. 155–56) (see also Mayhew 1980, p. 356; Nagel 1979, pp. 542–43).

<sup>&</sup>lt;sup>336</sup> Mayhew (1981, p. 630) notes that much sociological theory is unnecessarily complex because sociologists take the individual as the unit of analysis and believe that only an extremely complex theory is able to connect individuals, as individuals, to the phenomena observed in a variety of groups and societies. <sup>337</sup> According to Mears (1998a, p. 673), researchers should "develop theories that view sentencing decisions as examples of more general phenomena rather than as empirically distinct outcomes, thereby encouraging an accumulation of knowledge rather than of facts."

<sup>&</sup>lt;sup>338</sup> Hunt's (1993) constitutive theory of law also urges a relational view in which the component elements of social life are not individuals or institutions, but combinations of economic, political, class, gender, and

*Teleology.* Teleological theories explain phenomena—social, physical, and biological—as a means to an end. Although the use of teleological explanations in the natural sciences has greatly diminished over time (see Burtt 1954), virtually all sociological theories explain human behavior as a means to an end. That is, sociologists assume or impute ends (e.g., goals, needs, values, purposes, interests, *et cetera*) and then explain human behavior as a means to those ends. For example, human behavior has been explained as a rational means to assumed goals or preferences (e.g., Akers 1990; Cornish and Clarke 1986; Crouch 1979; Kiser and Hechter 1991; McCarthy 2002; Parsons and Shils 1951; cf. Sunstein 2000),<sup>339</sup> a consequence of the opportunities available while in pursuit of assumed goals or preferences (e.g., Bennett 1991; Cohen and Felson 1979; Felson 1994), and an adaptation to a lack of opportunities to achieve assumed goals and preferences (e.g., Cloward and Ohlin 1960; Merton 1938; Messner and Rosenfeld 1994). In fact, teleological explanations are so fundamental to

legal relations; however his theory differs from Black's theory of law in three important ways. First, Hunt argues that the major goal of socio-legal scholarship is the investigation of how law constitutes social relations. Thus in his theoretical framework, law is the independent variable. Second, Hunt writes from a Marxists perspective, making class and economic factors central to his theory. Although he acknowledges the shaky empirical status of a Marxist theory of law, he largely ignores the vast empirical literature documenting how other factors (e.g., organizational affiliations, cultural conventionality, *et cetera*) are often more influential than class and economic variables. Finally, Hunt does not develop testable propositions about how other aspects of social life vary with law. While he does underscore the importance of law in forming, maintaining, altering, and dissolving social relations, he does not attempt to systematically explain how any particular aspect of social life varies. True, law may be an important explanatory variable, but a *theory* of anything must attempt to explain the phenomenon of interest. In contrast, Black's theory makes law the dependent variable and attempts to systematically predict and explain legal behavior with testable propositions about the relationships between law and the five dimensions of social space. Also, his work does not elevate any particular social dimension over the other, recognizing that the explanatory power of the difference dimensions will vary according to the distribution of these dimensions in a particular setting. Black's theory readily acknowledges that law can be an independent variable in social theory, as are other forms of social control, but Black offers a much more comprehensive theory of social life by paying equal attention to the simultaneous influence of the four other important dimensions of social space.

<sup>&</sup>lt;sup>339</sup> Cooney (1993, p. 2230) remarks, "It is striking that empirical scholars, particularly those whose work brings them into close and sustained contact with their subjects (e.g., many sociologists and most anthropologists) do not adopt a rational actor model of human behavior."

sociological research traditions, teleology has been called the "superparadigm" of sociology (Black 1995, p. 861).<sup>340</sup>

Black's paradigm, however, makes no assumptions about the ends of any kind—it avoids teleology. Rather than impute goals, preferences, purposes, functions, needs, *et cetera*, Black's approach simply predicts and explains social variation.<sup>341</sup> He abandons teleology because the ends of individuals, groups, and societies are not directly observable.<sup>342</sup> When the ends of people must be assumed or imputed, uncertainty is inevitably introduced into to the paradigm, severely limiting the degree to which it can be applied or falsified (i.e., its generality and testability) (Black 1995, p. 862). Teleological theory is also value-laden, with means to certain ends often becoming morally significant, even though such motivations are unobservable and unknowable (p. 863). Teleology, then, deleteriously impacts an explanatory strategy's degree of scienticity.

Black's theory of law avoids teleology by simply predicting and explaining legal variation with the shape of social space in which it occurs—its social geometry. Black makes no claims or assumptions concerning the ultimate ends, purposes, or functions of law (but cf. Evan 1990; Luhmann 1989). Nor is his theory concerned with any interests

<sup>&</sup>lt;sup>340</sup> Baumgartner (2002, p. 647) notes, for example, that although the classical theories about religion feature different consequences, they are all teleological. Durkheim explained religion as a means to promote social cohesion and overcome the dissociative tendencies of groups (Durkheim [1912] 1995). Marx claimed that religion was used by the ruling class a means to thwart the revolutionary tendencies of the poor by presenting them with hopes that their suffering would end and they would be rewarded in the afterlife (Marx and Engels 1964). Weber explained the early capitalistic behavior of Calvinist Protestants as a means by which people counteracted religious anxiety, specifically the fear of being predestined to damnation (Weber [1904] 1958).

While teleological explanations that attribute a mission or destiny to a society as a whole (e.g., Marx and Engels [1848] 1959) have been largely discredited and abandoned, most sociological explanations of human behavior remain teleological (Black 2000a, p. 346).

<sup>&</sup>lt;sup>341</sup> "[P]ure sociology lacks various conceptions found in other sociological paradigms, including social life as a system of interrelated parts, a struggle for domination, an outcome of opportunities, a survival of environmental selection, a rational choice, a product of motivations with social origins, or an exercise of free will. It is has a theoretical logic of its own" (Black 2002b, p. 668).

<sup>&</sup>lt;sup>342</sup> According to Black (1995, p. 863), "teleological explanation is not only bad science, but hardly science at all" (see also Nagel 1979, pp. 520–35).
promoted or undermined by the law (e.g., Chambliss and Seidman 1971) or the effectiveness of law (e.g., Savelsberg 1992). By avoiding problematic assumptions about the ultimate ends of individuals, societies, and law itself, Black's paradigm achieves higher levels of generality and testability than approaches that rest on teleological assumptions.<sup>343</sup>

Pure sociology is not at all teleological...and this may be why many sociologists cannot understand or appreciate it. Many cannot conceive of an explanation of human behavior that does not regard every human action as a means to an end. They seek to divine the purpose of everything human: Why, for example, do people handle legal cases as they do? What are they trying to accomplish? Why does a police officer make a particular arrest? Why does a judge make a particular decision? Virtually always the "why" refers to how each human action is a means to an end—its purpose. But such purposes lie beyond science (Black 2002a, pp. 107–108).

In sum, the scientific superiority of Black's paradigm over previous sociological

research strategies results from the removal of psychology, one-dimensionality, units of analysis, individualism, and teleology. Specifically, these five problematic assumptions undermine the ability of subsequent theoretical formulations derived from a research strategy to achieve a high level of scienticity.<sup>344</sup> Moreover, these assumptions make it difficult to reduce the number of substantive errors in theories derived from paradigms that rely on them (Lenski 1988). According to Lenski (1988, p. 169), the structure of the theories derived from a paradigm is of primary importance, not their content (see also

<sup>&</sup>lt;sup>343</sup> Kuhn (1970) criticized the covering law approach for being teleological, arguing that the model *is* a means to a set goal: a permanent scientific truth (pp. 171–73). This, however, would appear to be a criticism of positivism, in general, and not particular to the covering law approach (see Tamanaha 1997). As Braithwaite (1953, p. 9) suggests, "the hypothetico-deductive method [e.g., covering law model] applied to empirical material...is the essential feature of a science." Furthermore, philosophers of science have noted that causality is only a feature of universal laws (Popper [1934] 1968) and that science progresses when it can explain lower level generalizations by deducing them from more general hypotheses at a higher level (Braithwaite 1953; Lenski 1988).

<sup>&</sup>lt;sup>344</sup> "Pure sociology is more scientific than most sociology because it has a more scientific location. Its subjects are more distant: They are located all over the world and across history, and they are social entities rather than people" (Black 2002a, p. 106).

Jasso 1988, pp. 2–3).<sup>345</sup> If the structure and basic idea of a theory are sound, the content of the theory can be improved through the interplay of theory and research (Merton 1948).<sup>346</sup>

## 6.2.4 The Conceptualization of Race in the Blackian Paradigm

The legitimacy of "race" as a concept for scientific inquiry depends on the criteria for defining race and will, in turn, be related to the analytical purposes for which the concept is employed (Duster 2003, p. 258). As noted earlier in this chapter, formal legal, conflict, and interactionist explanations of the influence of extra-legal factors particularly race/ethnicity—on the criminal charging-and-sentencing process fail to fully realize the scientific ideals of generality, parsimony, testability, and empirical validity. The failure of these theoretical approaches partially stems from their ambiguous and incomplete conceptualizations of extra-legal factors (e.g., age, gender, race/ethnicity, and region) and their inability to coherently incorporate these conceptualizations (and the assumptions on which they rest) into their broader explanatory framework (Zatz and Rodriguez 2006).<sup>347</sup> Contemporary formal legal theorists, for example, pay scant

<sup>&</sup>lt;sup>345</sup> "[T]he word 'theory'...refer[s] not just to an explanation of a single phenomenon, but a cluster of explanations of related phenomena, when the explanations, the deductive systems, share some of the same general propositions" (Homans 1967, p. 26).
<sup>346</sup> In discussing why sociology should be modeled after evolutionary biology rather than classical physics,

<sup>&</sup>lt;sup>346</sup> In discussing why sociology should be modeled after evolutionary biology rather than classical physics, Lieberson and Lynn (2002, p. 4) remark, "In reviewing developments from Darwin to more recent periods, we are struck by how the theory has a tolerance for problems and incompleteness that gives it a certain durability and that enables one to better cope with errors. The evolutionists do not confuse fatal errors, on the one hand, with problems stemming from incompleteness, information that is still insufficient or not yet determined, or even unresolved. The latter cases are worrisome and certainly not to be glossed over. Yet, it does not necessarily mean that the theory is to be abandoned or that Darwin was *wrong*. In evolution, *incomplete* is not the same as *erroneous*" (emphases in original).

<sup>&</sup>lt;sup>347</sup> Race and ethnicity are often seen as overlapping, but conceptually distinct social constructs. According to Bonilla-Silva (1999, p. 902), race (and racism) is linked to the history of colonialism, whereas ethnicity (and ethnocentrism) is linked to the history of the nation-state. While racial identification and assertion gives primacy to phenotype (e.g., "What are you?"), ethnic identification and assertion gives primacy to place (e.g., "Where are you from?") (cf. Loveman 1999, pp. 895–96). Moreover, according to Bonilla-Silva, "race" is often an ascribed characteristic, whereas ethnicity is a matter of self-selection (cf. Gross

theoretical attention to race/ethnicity, except when it is explicitly identified in the legal code (e.g., Black Codes, Jim Crow and anti-miscegenation laws). That is, formal legal theorists simply treat race/ethnicity as a fixed, legally defined category and posit that its impact is primarily—if not exclusively—indirect, operating through the law itself. This assumption is particularly problematic considering that legal institutions have differentially conceptualized race/ethnicity over the years and there has never been societal or legal consensus as to what constitutes membership to (or exclusion from) a racial/ethnic group (see Duster 2003; Gross 1998a, 2003; Winant 2000a). In fact, judges in the nineteenth-century South repeatedly held that racial classification was not a matter of law and that juries, who represented community consensus, were most qualified to decide racial group membership (Gross 1998a, p. 109). Gotanda (1995, p. 257) notes that the U.S. Supreme Court has used at least four distinctive ideas of race: (1) race as an indicator of social status (status-race) (see also Bonilla-Silva 1997, 2003); (2) race as a biological category derived from skin-color and region of ancestry (formal-race) (see

<sup>2003).</sup> Ethnic groups tend to share non-genetic traits or characteristics, such as surnames, language, accent, religion, culture, and national origin. The non-genetic traits are often, but not necessarily, passed down with biological families.

However, with respect to governmental classification systems (including the legal system), race and ethnicity have often been used interchangeably. For example, Irish, Polish, Italian, Greek, and Jewish immigrants were categorized as both racial and ethnic groups. As these groups became more integrated into American society, however, they began to be viewed simply as "white" (Baldwin 1985, p. xix; Bonilla-Silva 2003; Winant 2000a). Similarly, Hispanics in this nation have been categorized both as a racial and ethnic minority group, as well as members of the white racial majority (see, e.g., Gross 2003). Moreover, since the 1970s the United States government has asked all Americans to identify their "race," and separately whether they are of Hispanic ancestry; therefore Hispanics can be of any race (e.g., white Hispanic, black Hispanic, mixed-race Hispanic, *et cetera*) (Logan 2004; U.S. Bureau of the Census 2003b). In 2002, there were approximately 38.8 Hispanics in the United States, with 1.7 million self-identifying as black Hispanic (U.S. Bureau of the Census 2003b). Compared to non-black Hispanics, black Hispanics tend to have socioeconomic profiles much more similar to blacks, are more likely to marry blacks, and are more likely to reside in neighborhoods that have nearly as many black as Hispanic residents (Logan 2004).

Because the bulk of previous research on race, ethnicity and the criminal justice system have failed to make these detailed distinctions with respect to race and Hispanic origin (i.e., ethnicity), I have elected to use the term "race/ethnicity" for comparability with previous studies and to fully capture both of these groups.

Duster 1990; see also Smith 2003);<sup>348</sup> (3) race as past and present subordination (historical-race) (see also Baldwin 1955; Guinier and Torres 2002; Pitts 1974; West 1992); and (4) race as beliefs, culture, community, and consciousness (cultural-race) (see also Appiah 1992; Dawson 1995, 2001; Gilroy 1993; Shelby 2005; Winant 2000b). The legal status of racial/ethnic minorities and their treatment by legal institutions, however, have rarely varied according to these classifications. Furthermore, despite the tremendous variability in the definition of race/ethnicity since the early seventeenth century, there has been remarkable stability in the influence of race/ethnicity on the criminal justice process (see, generally, Hawkins 1987, 1995).

Conflict theorists conceptualize race/ethnicity as a proxy for economic and political power (see Bonacich 1991). In fact many conflict theorists believe that racial/ethnic differences in levels and rates of criminal punishment primarily result from differences in social class between racial/ethnic majorities and minorities, rather than racial bias *per se* (see, e.g., Chambliss and Seidman 1971; Quinney 1970; Reiman 1979). This conceptualization of race/ethnicity is particularly problematic considering the voluminous literature highlighting the unique effect of race/ethnicity on social life, apart from its economic and political character (Cose 1993; Dawson 1995, 2001; Feagin and

<sup>&</sup>lt;sup>348</sup> Over three decades ago, Lewontin's (1972) seminal studied revealed that only six percent of genetic variation between human beings was attributable to "race" and that the genes responsible for "racial characteristics" (e.g., skin color, hair texture, *et cetera*) constitute a mere 1.5 percent of the human genetic make-up. More recently, studies employing a variety of methods and examining a broader range of populations suggest that race accounts for no more than 10 percent of the genetic variation between individuals and that genetic differences between individuals in the same racial group can be greater than differences across racial groups (see, e.g., Bamshad and Olson 2003; Brown and Armelagos 2001). But despite the dubious scientific validity of "genetic" or "biological" race, human beings continue to use race to sort their social groupings and to define their social and economic interactions, and as a result, socially constructed race groupings can (and often do) have significant biological consequences (Duster 2003). Although the effect of race on biological outcomes, such as physical and psychological health, is largely (if not exclusively) indirect, operating through more proximal factors such as socioeconomic status, racial discrimination, and so forth, this does not lessen the importance of the study the bio-social and psycho-social consequences of racial categorization, particularly as it pertains to the social organization of society.

McKinney 2003; Work 1984). For example, some analysts have discovered that race/ethnicity is a stronger predictor of arrest rates than social class in many American cities (Liska, Chamblin, and Reed 1985) and that the racial composition of communities influences both residents' perception of neighborhood crime (Quillian and Pager 2001) and police strength (i.e., per capital police force) (Kent and Jacobs 2005) net of the actual crime rate and economic condition of the neighborhood. Other studies have shown that poor racial/ethnic minorities are often treated with greater leniency than whites for certain offenses (Hawkins 1987; McAdams 1998). Conflict theorists' conceptualization of race/ethnicity also fails to consider potential differences in criminal processing between racial/ethnic minority groups and lower-class whites in the criminal justice system. This is problematic in light of the recent studies that move beyond simple "black versus white" comparisons and document significant differences between blacks, Hispanics, and Asians with respect to criminal sentencing after social class and legitimate case characteristics are taken into account (see Demuth 2003; Kan and Phillips 2003; Kramer and Ulmer 2002; Steffensmeier and Demuth 2000, 2001).

Interactionists emphasize that race/ethnicity is not defined by its content, but by the social relations that construct it (Omi and Winant 1986; Rawls 2000). According to these theorists, race/ethnicity may have as many meanings as there are social contexts (see also Lee 1995).<sup>349</sup> Although the greatest strength of this conceptualization is its ability to account for the malleability of the meaning of race/ethnicity across time and place, most interactionist models also ignore well-documented (and heavily theorized)

<sup>&</sup>lt;sup>349</sup> Duster (2003, p. 259) argues, "[T]hat a concept is variable in its meaning does not mean it has not important analytical use. Scientific inquiry abounds in such concepts, including a range that extends from 'genetic disorders' to 'economic markets.' On close examination, these apply to widely divergent empirical sites."

uniformities concerning the social realities of race/ethnicity (Machery and Faucher 2005). Admittedly, some interactionists have acknowledged the persistent effects of social stratification on racial prejudice resulting from competition over scarce socially valued goods and a commitment to a relative status positioning of groups in a racialized social order (e.g., Blumer 1955, 1958; Bobo 1999), yet they maintain that attitudes, feelings, and beliefs concerning the racial order that arise from this stratification, and not racial stratification itself, are central to their explanations. Interactionists appear unable to explain why the categorization of groups based on phenotypic features (e.g., skin color) has remained a major source of social division across cultures at least since the early sixteenth century (Machery and Faucher 2005), even though radically different conceptualizations of race and diverse racialized practices have developed across time and place (Bonilla-Silva 1997; Skidmore 1993). Interactionists also have difficulty explaining why the relative importance of biological sex categories and early "racial" categories based on origin (e.g., Irish) and religion (e.g., Jews) (Hirschfeld [1938] 1973; Winant 2000a) has diminished over time relative to phenotypic racial categories (Duster 1990; Root 2000). Interactionists have also largely ignored the work of many scholars of race/ethnicity who continually discover that race/ethnicity, like other social constructs, has an independent effect on social life apart from the meanings individuals assign to it and the way people personally experience it (see Bonilla-Silva 2003).

The Blackian paradigm overcomes these shortcomings by providing a completely sociological conceptualization of race/ethnicity that is fully incorporated into its broader explanatory strategy. Black's conceptualization of race/ethnicity, similar to his conceptualization of law, allows his theory of law to achieve a higher level of scienticity

than rival theories. According to Black, biological characteristics such as age, sex, and race are *not* social statuses because they are not locations in social space (but cf. Steffensmeier *et al.* 1998). As a result, racial/ethnic categories have no inherent sociological significance.<sup>350</sup> Nonetheless, race/ethnicity achieves social salience via its systematic association with a number of objective social conditions, particularly in the Western world (Horwitz 1990, p. 14). That is, race becomes significant because of the ways in which it has been used to structure social interaction and organize society (see also Park 1950). Race/ethnicity, according to Black (1989, p. 61), may serve as a crude indicator of social status in many situations. For example, in the aggregate, blacks and Hispanics in the United States have significantly less vertical, radial, relational, corporate, cultural, and normative status than whites (Black 1999, p. 68).<sup>351</sup>

With respect to vertical status (e.g., economic stratification), blacks lag far behind whites in virtually every category of economic wellbeing (Hacker 1992; Oliver and Shapiro 1995).<sup>352</sup> In the United States, for example, blacks comprise approximately thirteen percent of the population, but own less than one percent of privately owned rural land (Thomas, Pennick, and Gray 2004) and less than one percent of the nation's total wealth (Conley 1999). Relative to whites, blacks are three times more likely to live below the poverty level and are twice as likely to be unemployed (Bonilla-Silva 2003;

<sup>&</sup>lt;sup>350</sup> According to Machery and Faucher (2005), a satisfactory conceptualization of race must: (1) not assume that races exist; (2) not assume essentialism; (3) accommodate cultural and temporal variations; (4) account for similarity of racial classification based on phenotypic features; and (5) account for individual differences in racial categorizations (see also Winant 2000b).

<sup>&</sup>lt;sup>351</sup> According to Berk (2003, p. 211), "[r]ace is little more than a marker for other factors that are really behind the decision to seek the death penalty" and the "persistent associations between the race of the victim and the chances of a capital charge suggest that something structural is going on linked to cause-and-effect relationships."

<sup>&</sup>lt;sup>352</sup> Poverty within predominately black communities has also been remarkably persistent over time, although inner-city poverty, in general, has undergone dramatic changes over the past 30 years (Sampson and Morenoff 2006; see also DiPrete and Eirich 2006; Wilson 1987, Chapter 2).

Gleason and Cain 2004; Terkel 1992). Blacks are less likely than whites to be granted home loans (Hillier 2003; Massey and Denton 1993; Stuart 2003)<sup>353</sup> and receive college scholarships (U.S. General Accounting Office 1994), blacks have one-tenth of the net worth and two-hundredths the net financial assets of whites,<sup>354</sup> black households have a median income that is 59 percent of white households, black households are over twice as likely as white households to have zero or negative net worth,<sup>355</sup> and black-owned homes are worth 35 percent of what white-owned homes are worth (Dodson 2003; Oliver and Shapiro 1995).

Even among the college-educated and middle- and upper-class, blacks significantly lag behind whites (Grodsky and Pager 2001). For example, blacks males with a college degree earn 76 percent of what white males with similar education earn and are nearly twice as likely as their white counterparts to be unemployed (Conley 1999, p. 86; Gleason and Cain 2004). College-educated blacks also posses less than one-fourth (22 percent) the wealth as college-educated whites (\$15,000 versus \$67,000). Middleclass whites (i.e., annual household income between \$25,000 and \$49,999) have more than eight times the net financial assets as middle-class blacks, and among the two highest income quintiles (\$50,000 and above), blacks have one-fifth the net financial assets of whites (Oliver and Shapiro 2001).<sup>356</sup>

<sup>&</sup>lt;sup>353</sup> As noted above, the capacity to borrow wealth is a variable aspect of the vertical dimension of social space (see Black 1976, p. 11). <sup>354</sup> *Net worth* is calculated by subtracting total assets from total liabilities. These assets include financial

<sup>&</sup>lt;sup>354</sup> *Net worth* is calculated by subtracting total assets from total liabilities. These assets include financial resources such as savings accounts, mutual funds, and equity in homes and automobiles. *Net financial assets* is calculated in a similar fashion to net worth, except home and automobile equity is not included. Net financial assets is a better indicator of short-term financial viability because it concentrates on free cash resources (Conley 1999; Oliver and Shapiro 1995).

<sup>&</sup>lt;sup>355</sup> Nearly two-thirds of black households (63.2 percent) posses no net financial assets compared to 28 percent of white households (Oliver and Shapiro 1995).

<sup>&</sup>lt;sup>336</sup> According to Gittleman and Wolff (2004), much of the racial gap in wealth can be attributed to differences in inherited wealth, not income, savings, and investment dynamics (see also Conley 2001). As

Blacks, in the aggregate, also have significantly less radial, relational, and functional status (i.e., integration, intimacy, and interdependence) because they are much less integrated into mainstream society, are largely excluded from white social networks, and have infrequent meaningful contact with the vast majority of whites (Allport 1954; Barndt 1972; Drake and Cayton 1945; Loury 1977; Massey and Denton 1989; Pattillo-McCoy 1999; Wacquant 1997).<sup>357</sup> For example, a recent study revealed that 87 percent of whites admitted that none of their closest friends were black, 89 percent of whites have never had a romantic relationship with a black person, and 95 percent of whites have a white spouse (Bonilla-Silva 2003, pp. 105, 129). Even when whites attend highly integrated or predominately minority public schools, academic tracking almost guarantees that their experience in the classroom is mostly white (Anyon 1997; Kozol 1991). In fact, no other group in the history of the United States has ever been as segregated as blacks, even for a brief historical moment (Delaney 1998; Lieberson 1980), *and* levels of black-white segregation in the United States have actually increased at the

much as 80 percent of lifetime *net* wealth can be attributed to gifts from one generation to another and nearly 25 percent of the nation's current wealth distribution can be explained by the wealth distribution at the end of the Civil War (cf. Barsky *et al.* 2002). Due to slavery, with few exceptions, approximately twelve generations of black families were unable to accumulate and inherit any wealth. Using the price of enslaved blacks from 1790 (when the government began collecting census data) to 1860 as a proxy for slave capital, the total value of slave labor *for that seventy-year period* was estimated between \$1.4 trillion and \$4.7 trillion (in 1983 dollars), depending on the estimated rate of inflation and compound interest (Marketti 1990, p. 118). Considering that the first documented slaves in America arrived in 1619, some 171 years earlier, the total value of slave labor from 1619 to 1865 has been estimated to be between \$5 trillion and \$10 trillion (in 1993 dollars) (America 1993; Vedder, Gallaway, and Klingaman 1990). At the end of the first quarter of 2007, the *net* worth of all households in the United States was estimated at \$56 trillion (Federal Reserve Board 2007). According to Oliver and Shapiro (1995), the accumulation of disadvantages against blacks is so great that if all economic forms of racial discrimination against blacks ended today, it would take several hundred years for blacks to catch up to whites (see also Bonilla-Silva 2003, p. 79).

<sup>&</sup>lt;sup>357</sup> High levels of racial segregation and income inequality, interacting to produce concentrated poverty, result in blacks being exposed to far higher rates of social disorder and violence than whites. It has been argued that this long-term exposure to social disorder and violence produces a high allostatic load among blacks, leading to a variety of deleterious health outcomes (e.g., coronary heart disease and inflammatory disorders) and cognitive impairments (e.g., atrophy of memory, inhibition of synaptic learning, and suppression of neurogensis) (Massey 2004; Sampson 2003; see also Willie, Kramer, and Brown 1974).

neighborhood level over the past 50 years (Alba, Logan, and Stultz 2000; Massey and Hajnal 1995).<sup>358</sup> Blacks are also are more likely to be marginalized than whites (and other minority groups) because they are less likely to be employed (Wilson 1996),<sup>359</sup> more likely to be employed in low-skilled and unskilled occupations (Dodson 2003; Mincy 2006; Wilson 1996), more likely to have unmarried births (Martin *et al.* 2003; South and Crowder 1999; Western, Lopoo, and McLanahan 2004; Wilson and Neckerman 1986), and more likely to receive public assistance (Gilens 2000; Hancock 2004; Manza 2000).

As scholars have noted for some time, blacks' and Hispanics' capacity for collective action (corporate status) via the democratic process, relative to whites, has been severely diminished by felony disenfranchisement laws,<sup>360</sup> illegal vote suppression,

<sup>&</sup>lt;sup>358</sup> Since the 1970s, there has been increased integration at the state and county levels, but increased segregation at the neighborhood and (since 1950) municipal levels (Massey and Hajnal 1995). After examining trends in average black-white segregation across all metropolitan areas in the United States and noting a downward drift in the mean segregation values since the 1970s, Thernstrom and Thernstrom (1997) suggest that segregation is declining in importance. Their work has been criticized, however, for confounding two distinct trends: (1) increasing integration in newer and smaller metropolitan areas of the south and west with relatively small black populations (particularly in areas that contain colleges, military bases, large amounts of post-1970s housing); and (2) stable or slightly increasing segregation in older metropolitan areas that contain a disproportionate share of the nation's black population. By solely considering the average segregation level across all metropolitan areas (without adequately taking into account the size of the black population in these areas), the Thernstroms mischaracterize the situation of blacks because the declines in segregation are concentrated in places where few blacks live (Iceland, Weinberg, and Steinmetz 2002; Massey 2004, pp. 8–9).

<sup>&</sup>lt;sup>359</sup> In 2004, 72 percent of black males in their twenties who did not complete high school were jobless (i.e., unable to find work, not seeking work, or incarcerated). By comparison, 34 percent of white and 19 percent of Hispanic males in their twenties who were high school dropouts were jobless. Even when black high school graduates are considered, nearly half of black males in their twenties were jobless (Edelman, Holzer, and Offner 2006; Mincy 2006).

<sup>&</sup>lt;sup>360</sup> Uggen and Manza (2002, pp. 797–98), citing U.S. Department of Justice data, note that voting-age blacks are nearly 3.3 times more likely to be disenfranchised because of a felony conviction than the national average (7.5 percent versus 2.3 percent). In sixteen states (Alabama, Arizona, Delaware, Florida, Georgia, Iowa, Kentucky, Mississippi, Nebraska, Nevada, New Mexico, Rhode Island, Virginia, Washington, Wisconsin, and Wyoming), the felony disenfranchisement rate for blacks is in the double digits, whereas no single state has a felony disenfranchisement rate in the double digits for its total population. Moreover, in Iowa and New Mexico, nearly one-in-four voting-age blacks is disenfranchised (24.87 percent and 24.67 percent, respectively). The total felony disenfranchisement percentages for Iowa and New Mexico are 4.65 and 6.21, respectively. Research also suggests that the racial composition of the state and state prisons are strongly associated with the adoption of felon disenfranchisement laws (Behrens

low voter turnout, racial gerrymandering, multi-member legislative districts, election runoffs, annexation of predominately white areas, and at-large district elections (Behrens *et al.* 2003; Bonilla-Silva 2003; Brown *et al.* 2003; King and Mauer 2004; Uggen and Manza 2002). According to Dowding (1996), blacks lack social power primarily because they do not have the organizational capacity and resources to carry out a nationalist program (see also Myrdal 1944; Pitts 1974; cf. Jenkins, Jacobs, and Agnone 2003; Sampson *et al.* 2005a). Moreover, blacks and Hispanics are severely underrepresented at the highest levels of federal and state elected government (Dawson 2001) and in the federal judiciary (Gryski, Zuk, and Barrow 1994). Blacks and Hispanics, in the aggregate, also have significantly higher levels of residential instability (i.e., high residential turnover), thus increasing their level of social disorganization and undermining their ability to develop and maintain strong internal ties and act as a cohesive unit (Pattillo 2007; Sampson and Groves 1989; South and Crowder 1997; but cf. Venkatesh 2006; Wilson and Taub 2006).

Blacks, in the aggregate, also have less cultural status (conventionality) and normative status (respectability and authority). In relation to whites, they are more likely to have "non-standard" names (see Fryer and Levitt 2004), less likely to speak standard English because of isolation in the ghetto (Barndt 1972; Massey and Denton 1993; cf. McWhorter 1998),<sup>361</sup> have less formal education (Bonilla-Silva 2003), are more likely to receive inadequate formal education (Anyon 1997; Kozol 1991),<sup>362</sup> are less likely to

*et al.* 2003, pp. 596–97). Whether felon disenfranchisement laws actually reduce the turnout of African American men, however, remains a contested issue (compare, e.g., Manza and Uggen 2005; Miles 2004). <sup>361</sup> Recent research suggests that blacks are more likely to be subject to discrimination in the housing market *in the absence* of personal contact because rental agents use racial "cues" derived from Black Accented English (Massey and Lundy 2001).

<sup>&</sup>lt;sup>362</sup> Predominately minority schools districts are more likely to be underfunded and less likely to prevail in education finance litigation compared to predominately white districts (Edwards and Ahern 2004; Kozol

attend private schools (U.S. Department of Education 1995),<sup>363</sup> are less likely to graduate from college (Mincy 2006; Thernstrom and Thernstrom 2003) and law school (Ayres and Brooks 2005), are more likely to have a documented history of criminal involvement (Clear 2007; Miller 1996; Short 1997; Tonry 1995),<sup>364</sup> have lower levels of collective efficacy (Sampson, Morenoff, and Earls 1999; but see Cross 2003), and are less likely to depend on and benefit from police intervention (Anderson 1999; Black 1980; Pattillo-McCoy 1999; Venkatesh 2006). Also, with respect to cultural (and functional) status, the heroic, artistic, and scientific contributions of blacks to mainstream society have been largely ignored or negated by whites (for discussions, see Gates and West 2000; Graham 1999; Ross and Edwards 1998, pp. 31–32; Salzman, Smith, and West 1995).<sup>365</sup>

In accordance with the structure of his paradigm, Black does not relegate the social significance of race/ethnicity to one particular sphere of social life (i.e., onedimensionality). That is, unlike other theorists, Black does not attribute the importance of race/ethnicity primarily to economic inequality (Sampson and Morenoff 2006; Wilson 1980), culture (Akers 1998; Appiah 1992; Cernkovich, Giordano, and Rudolph 2000),

<sup>1991).</sup> Racially segregated schools have also been linked to academic underachievement and adult incarceration among blacks, and this impact has actually *increased* in recent decades (see, e.g., LaFree and Arum 2006).

<sup>&</sup>lt;sup>363</sup> In general, private schools provide better educational instruction than public schools and students perform better at private institutions. The relationship between private schools and educational performance remains after taking into account the income and educational attainment of the parents of students (see, generally, Bryk, Lee, and Holland 1993). In 1995, 10 percent of white children attended private schools, whereas only four percent of blacks attended private schools (U.S. Department of Education 1995).

<sup>&</sup>lt;sup>364</sup> In 2000, 32 percent of black male high school dropouts age twenty to forty were in jail or prison, compared to 7 percent of similarly situated white males (Western 2006, p. 17). For males born between 1965 and 1969, nearly 60 percent of black high school dropouts had been incarcerated by 1999, compared to 17 percent of white high school dropouts (p. 26).

<sup>&</sup>lt;sup>365</sup> Bonilla-Silva (2003) remarks that perceived differences between racial/ethnic groups that were once believed to have biological origins are now largely attributed to cultural differences, "Whites may no longer think that Africans, Arabs, Asian Indians or blacks from the West Indies are biological inferior, but they assail them for their presumed lack of hygiene, family disorganization, and lack of morality" (p. 39; see also Pieterse 1992).

integration and interdependence (Massey 1995), the capacity for collective action (Sampson, Raudenbush, and Earls 1997), the capacity for social control (Sampson 1987; Shihadeh and Steffensmeier 1994), and so forth. Rather, Black recognizes that the influence of race/ethnicity on social life—particularly the operation of social control must be understood in relation to all of the various dimensions of social space: vertical, horizontal, corporate, symbolic, and normative. Furthermore, Black's conceptualization of race/ethnicity neither rests on untestable assumptions of how individuals and groups perceive or experience their own racial/ethnic identity nor how these individuals and groups are perceived by others (but cf. Pitts 1974; West 1992).<sup>366</sup> Black's conceptualization is also sufficiently flexible to accommodate changes in salience of race/ethnicity across time and place. Also, recall that his unit of analysis-the shape of social space—allows for the simultaneous consideration of the micro- and macro-level dimensions of social phenomena (Horwitz 2002, pp. 642–43). Therefore, with respect to race/ethnicity, theories derived from Black's paradigm consider the influence of race/ethnicity at the micro- and macro-sociological levels (see, e.g., Kan and Phillips 2003).

It must be emphasized that Black does not suggest racial/ethnic groups actually exist. However, as a social construction, race can and does have a substantial effect on how people behave (see Duster 2003, p. 263). Similar to the interactionist model, Black believes "races" are best understood as relations and not as things (see also Weber [1922] 1968, p. 42). According to Omi and Winant (1986, p. 61), "racial formation is the

<sup>&</sup>lt;sup>366</sup> Bonilla-Silva (2003, p. 54) also challenges the usefulness of studying attitudes about race/ethnicity to explain racial disparities in society, arguing "the problem of racism is the problem of power, therefore the intentions of individual actors are largely irrelevant to the explanations of social outcomes." Quoting Fanon (1967, p. 77), Bonilla-Silva (1997, p. 465) notes, "The habit of considering racism as a mental quirk, as a psychological flaw, must be abandoned."

process by which social, economic, and political forces determine the content and importance of racial categories." Similarly, Bonilla-Silva (1999) suggests that "races are the effect of racial practices of opposition at the economic, political, social, and ideological levels" (p. 901) and the task of studying racial structures is to "uncover the particular social, economic, political, social control, and ideological mechanisms responsible for the reproduction of racial privilege in a society" (2003, p. 9). Race/ethnicity, then, is important inasmuch as it accurately locates individuals and groups within the social structure and influences their social relations.<sup>367</sup> Therefore it is possible that racial/ethnic categorization may be more influential for some groups than others. Some racial/ethnic groups, for example, may be more homogenous with respect to their locations and relations along Black's five dimensions of social space than others. The more homogenous the racial/ethnic group, the more likely classification in that particular racial/ethnic group will impact the lives of its members, all else equal. For example, Kan and Phillips (2003, p. 85) argue that individual blacks may be more disadvantaged than members of other groups, net of their particular social status, because "the aggregate characteristics of the group haunt or help its members...members of some groups are fighting an uphill battle, while members of other groups hold the power of presumption" (see also Kang 2005; Stanko 1981/1982; Sunstein and Jolls 2006). Pager's (2007) recent audit study of the impact of race and prior criminal convictions (i.e., respectability) on employment opportunities provides some indirect support for their assertion. Her study revealed that blacks were less than half as likely (41 percent) as whites to receive a callback from employers, although the applicants had identical

<sup>&</sup>lt;sup>367</sup> Anderson and Massey (2001, p. 12) suggest that racial stratification is "a multilevel process in which individual behavior is shaped by social structures that are firmly rooted in space" (see also Delaney 1998).

educational and work histories and no criminal record. Pager also discovered that whites with a criminal record were more likely to receive callbacks from employers than blacks without a criminal record, although these applicants also had identical educational and work histories. She suggests that employers may fear that blacks without criminal records may nevertheless have criminal tendencies.<sup>368</sup> Others have argued that the "race effect" tends to vary by degree of closeness of the racial/ethnic minority group to the dominant racial/ethnic group with respect to color (i.e., skin tone), culture, and so forth (see Bonilla-Silva 2003, p. 182; Hagan, Shedd, and Payne 2005, pp. 383–84; Portes and Rumbaut 2001, p. 47). It is questionable, however, whether skin color (i.e., phenotype) exerts an independent effect on social life, net of the status locations and relations articulated by Black's paradigm that define human interaction (but see Blair, Judd, and Chapleau 2004; Eberhardt et al. 2006).<sup>369</sup> While it is true that not all racial/ethnic minority groups share the same experiences, this is likely attributable to the fact that these different groups, in the aggregate, have different locations in the social structure. Furthermore, status mobility along these five dimensions of social space is likely to explain the changing character of racial/ethnic classifications and relations. As noted earlier, after Irish, Italians, Greeks, and Jews, in the aggregate, gained greater social

<sup>&</sup>lt;sup>368</sup> According to Western (2006, pp. 6–7), "No other group [than African Americans], as a group, routinely contends with long terms of forced confinement and bears the stigma of official criminality in all subsequent spheres of social life, as citizens, workers, and spouses. This is a profound social exclusion that significantly rolls back the gains to citizenship hard won by the civil rights movement...[i]ncarceration rates [of black males] are now so high that the stigma of criminality brands not only to individuals, but a whole generation of young black men with little schooling" (see also Harcourt 2007, pp. 162–65; Roberts 1999, p. 805; Wacquant 2005b).

<sup>&</sup>lt;sup>369</sup> Duster (2005, p. 1050) notes that "[t]here is a complex feedback loop and interaction effect between phenotype and social practices related to that phenotype." Citing a classic epidemiological study on hypertension discovering that, within the black population in America, darker-skinned blacks have, on average, higher blood pressure than lighter-skinned blacks (see Klag *et al.* 1991), Duster highlights that the authors of the study concluded that it was not the color of the skin that produced the relationship, but that darker skin color in the United States was associated with less access to scarce and valued resources.

status in the United States, they stopped being categorized as "racial" groups entirely (see, e.g., Winant 2000a).

An additional advantage of Black's conceptual framework is its ability to offer a more general and parsimonious explanation of the interactive effects of age, gender, and race/ethnicity—at both the micro- and macro-sociological levels—on legal behavior. As noted above, some research suggests that age, gender, and race/ethnicity often have both direct and conditional effects on legal decision-making (e.g., Frohmann 1997; Spohn and Holleran 2000; Spohn et al. 1985; Steffensmeier et al. 1998; Zatz 2000). These researchers posit that the motivations and goals of decision-makers are central to understanding these effects (e.g., decision-makers' concerns with offender blameworthiness, the protection of the community, and the practical implications of the actual sentencing decision) (see also Albonetti 1987, 1991). Death penalty researchers have also considered the interactive effects of victim characteristics on capital case outcomes. Holcomb and colleagues (2004), for example, discover that white femalevictim cases are most likely to result in a death sentence, even after holding the conduct of the victim and other legitimate cases characteristics constant. Similar to previous research on non-capital cases, Holcomb et al. assert decision-makers' interpretations of victim characteristics with respect to culpability and future dangerousness are of primary importance. It has also been suggested that these interpretations may be influenced by the larger social context in which these cases occur (e.g., electoral politics, correctional costs, and county racial composition) (see Baumer et al. 2000; Steffensmeier and Demuth 2000).

Black's theory of law also predicts that age, gender, and race/ethnicity have both direct and interactive effects on criminal processing; however the explanatory power of these variables results from the social positions that they indicate, rather than the variable interpretations that legal decision-makers place on them (Uggen and Kruttschnitt 1998). As noted above, age and gender, like race/ethnicity, are not social statuses themselves, but may serve as crude proxies for social location to the extent that they are correlated with Black's five social dimensions.<sup>370</sup> In modern America, for example, in the aggregate, youths have less vertical, radial, relational, cultural, and corporate status (see Black 1989). That is, holding gender and race/ethnicity constant, the young have less wealth, weaker social ties, less participation in the center of social life, less culture (e.g., education), and less capacity for collective action (e.g., youth are not allowed to vote). The normative status of youths, in the aggregate, is less straightforward—they have less capacity for social control (i.e., less authority), but they also tend to have less extensive criminal records (i.e., more respectability).<sup>371</sup> The situation for women is somewhat more complex—hold age and race/ethnicity constant, and women, in the aggregate, have less wealth and less authority than men, but their status locations are typically not as low as the young and most racial/ethnic minorities (Gilmore 1996; Gornick 2004; Smith 2002).

<sup>&</sup>lt;sup>370</sup> Similarly, Sampson and colleagues (2005b, p. 224) note, "Our theoretical framework does not view 'race' or 'ethnicity' as holding distinct scientific credibility...[r]ather, we argue they are markers for a constellation of external and malleable social contexts that are differentially allocated by racial/ethnic status in American society." Similarly, in commenting on the Federal Drug Administration's likely approval of the first "ethnic medicine" for heart disease targeted at African Americans, David Goldstein, director of Duke University's Center for Population Genomics and Pharmacogenetics, remarked, "Race for prescription is only an interim solution to carry us through a period of ignorance until we find the underlying causes" (Davies 2005).

<sup>&</sup>lt;sup>371</sup> Kurlycheck and Johnson (2004), for example, discover that juveniles transferred to adult court receiver *harsher* penalties than adults, controlling for offense severity, prior criminal history, race, gender, and conviction mode (i.e., plea/trial) for violent crimes, but not property or drug crimes. They suggest that juveniles convicted of violent crimes in adult court may be seen as *less* amenable to rehabilitation. Kurlycheck and Johnson also discovered that the explanatory power of offense severity and prior criminal record on sentencing severity is stronger for adults than juveniles (p. 504).

On the other hand, in the aggregate, women have closer interpersonal relationships (relational status), are less likely to have criminal histories (i.e., respectability), and are more likely to play a central role in family life (radial and functional statuses) (Crowell 2004; Daly 1987; Uggen and Kruttschnitt 1998). In recent years, women in America have also surpassed men in college enrollment (cultural status) and voter turnout (corporate status) (Center for American Women and Politics 2004; Dugger 2001).

According to Black's explanatory framework, the interactive effects of these extra-legal factors are important because, at least in modern times, they provide more social information about the parties in the conflict, and therefore more closely pinpoint an individual's location in the multidimensional social space. Rather than resting on largely untestable assumptions of how legal decision-makers differentially define and interpret these social characteristics in the context of a criminal case proceeding, Black simply posits that the social locations that these characteristics represent predict and explain how the cases will be handled. It should be noted that Black does not suggest that certain combinations of these extra-legal characteristics will uniformly disadvantage a defendant in a criminal case. In fact, throughout a criminal proceeding, more specific information may be presented that will differentially locate the individuals in the conflict along the various status dimensions. However, to the extent that this information is unknown or disputed, legal decision-makers may impute the aggregate characteristics of the group in a particular case, therefore disadvantaging individuals who are viewed as members of these groups (Bodenhausen and Wyer 1985; Kan and Phillips 2003; Stanko 1981/1982). Black's theory also predicts that as age, gender, and race/ethnicity become less reliable indicators of social status, they will become less important influences on case outcomes.

Although unlikely, it is possible that certain age-gender-racial/ethnic combinations that typically disadvantage individuals in criminal proceedings today may advantage members of these groups in the future as membership in these groups is associated with higher status locations. Black's theory readily anticipates and predicts these facts as well.

According to Black (1989, p. 61), the major disadvantages that blacks face in legal life appear to be shared with others of lower status. Blacks and Hispanics, in the aggregate, "suffer legal disadvantages of any group inhabiting the bottom of society" (see also Bell 1992). Conversely, individual blacks and Hispanics who are wealthy and well integrated into society are legally better off than poor and marginally integrated members of other racial/ethnic groups (Black 1989, p. 119 n.12; Reiman 1979, pp. 97–98). Black argues that uniformity in the application of law is reserved for two situations: (1) when cases are socially similar and (2) in cases in which the outcome is trivial (cf. Blumstein 1993; Spohn and Cederblom 1991). He notes, however, that the greater amount of potential law that may be applied to a case, the more social information about those in the conflict becomes involved (or gets collected). At one extreme (i.e., very little potential for law) is a parking ticket, where very little social information is known, collected, or exchanged. At the other extreme is a capital murder case where the defendant's (and often the victim's) entire life history may be presented (Black 1984, p. 20).<sup>372</sup> The increased knowledge of the social diversity of a case (which corresponds, to an extent, with the potential severity of the case), therefore, increases the degree of variation of how the case in handled. "Law...remains saturated with information about the social

<sup>&</sup>lt;sup>372</sup> It is also becoming increasingly common for capital defendants to present a "community health profile" during litigation—i.e., evidence of the effect of aggregate levels of drug-abuse, violence, sub-standard schooling, inadequate health-services, migration, *et cetera*, on individual and community well-being (Dudley and Leonard 2008).

characteristics of litigants and others involved in legal affairs...The abundance of this information, particularly in court, makes legal discrimination possible...The more social information, the more discrimination" (Black 1989, pp. 66–68).

## 6.2.5 Evaluation of Black's Theory of Law

Generality. Black (1976) argues that his theory of law is able to explain variation in law (i.e., its quantity and style) irrespective of time, place, or structural level (see also Cooney 1997b). He notes that the classical sociological scholars, such as Durkheim ([1893] 1947), Weber ([1922] 1968), and Ehrlich ([1913] 1936) attempted to develop general theories of law pertaining to observable facts, with propositions that would apply to all legal systems (Black and Mileski 1973, p. 2). Black's theory has *explicitly* been used to predict and explain diverse phenomena such as the role of gender and familial status in criminal sentencing in contemporary America (Kruttschnitt 1981, 1982; Kruttschnitt and McCarthy 1985); the production of police reports (Black 1970), police disrespect toward citizens (Mastrofski, Reisig, and McCluskey 2002), police discretion in arrests (Black 1980; Cooney 1992; Smith 1987a; Worden 1989), citizens' discretion in calling police in the United States (Avakame, Fyfe, and McCoy 1999; Doyle and Luckenbill 1991; Gottfredson and Hindelang 1979), Canada (Gartner and Macmillan 1995), and Australia (Braithwaite and Biles 1980), legal behavior in seventeenth century New England (Baumgartner 1978), support for capital punishment (Borg 1998), legal executions and incarceration in Post-Reconstruction Georgia (Massey and Myers 1989), homicide clearance rates in urban areas (Borg 2001), the ability of complainants to garner and use legal evidence in legal disputes (Cooney 1994), the reporting behavior of fraud victims (Copes et al. 2001), the legal systems of squatter communities in Brazil (Sousa

Santos 1977), historical changes in the use of law in the United States (Lessan and Sheley 1992), medical malpractice litigation (Mullis 1995), speech crimes in China's Qing dynasty (Wong 2000), and the relationship between centralized political systems and state executions (Cooney 1997b).<sup>373</sup>

As noted above, Black argues that the sociological relevance of race/ethnicity depends entirely on the degree to which it corresponds to a social location (see also Uggen and Kruttschnitt 1998, p. 346). His conceptualization of race/ethnicity allows his explanation of the relationship between race/ethnicity and the law to achieve a higher level of generality than formal legal, conflict, and interactionist perspectives because it is able to explain and predict the same facts as these other perspectives under a single explanatory structure. For example, similar to formal legal theorists, Black's theory would predict increasing uniformity, in the aggregate, in the application of law to minority groups. Black's theory, however, would attribute this uniformity to the fact that racial/ethnic minority groups are becoming structural similar to the racial/ethnic majority group. As noted above, some racial/ethnic minorities (e.g., blacks and Hispanics) continue to significantly lag behind whites in many aspects of social wellbeing, but the gaps between these groups (in most areas) have been steadily shrinking over time (cf. Western and Pettit 2005). Recall that Black argues that cases are handled similarly (i.e., legal rules matter) when the cases are structurally similar or when the amount of potential

<sup>&</sup>lt;sup>373</sup> As previously discussed, Black (1990) and others have expanded his theory to explain many forms of conflict management beyond law. For example, Black's theory has been used to explain individualistic violence among juveniles (Borg 1999), romantic partners (Michalski 2004), social elites (Cooney 1997a, 2003b), and prison inmates (Phillips 2003), as well as collective violence such as terrorism (Black 2004a, b), lynching (Senechal de la Roche 1997b), and race riots (Senechal de la Roche 1990). In addition to violence, his theory has also been used to successfully predict drug testing at the workplace (Borg and Arnold 1997), non-legal dispute resolution in suburbia (Baumgartner 1988), employee theft (Tucker 1989), and conflict management strategies among corporate executives (Morrill 1991) and nation-states in the world system (Borg 1992).

law in a case is trivial. Black's theory, then, specifies the conditions under which legal rules have their greatest influence (i.e., explanatory power). Black's theory also explains why racial/ethnic disparities in sentencing are most pronounced for the most serious crimes, such as murder and rape—these cases collect and present the most detailed information about parties in the dispute. It must be emphasized that Black's theory of law offers an explanation of legal behavior *in addition to* rather than *instead of* the formal legal perspective (Horwitz 1983; Sherman 1978).<sup>374</sup> That is, Black's theory of law explains legal behavior when legal rules and facts are held constant. But Black's theory also accomplishes more than just predicting legal behavior net of legal rules and facts his theory seeks to explain the development of the legal order (Horwitz 1983, p. 372). Recall that social control is both the *definition of* and *response to* deviance; therefore his theory of law explains both the creation of legal rules and the actual behavior of legal officials.<sup>375</sup> For example, Black's theory of law predicts that white-collar offenders and corporations are treated more leniently under the law, as a matter of law, net of the actual harm done by the offender, because they typically have greater vertical and corporate status, respectively (Black 1976; Shapiro 1990).

<sup>&</sup>lt;sup>374</sup> As Robinson and Darley (2003, p. 954) note, "[T]he legal rule is just one of hundreds of variables that influence a case disposition."

<sup>&</sup>lt;sup>375</sup> Legal sociology has been criticized for neglecting the empirical examination of formal legal rules (see, e.g., Ellickson 1991, pp. 147–149). Traditionally, legal sociologists have paid scant attention to the content of legal rules because much of the socio-legal literature suggests there is a substantial gap between what law says ought to occur and what actually occurs. As a result, their emphasis has been on *actual* behavior and the content of legal rules has been of secondary importance to them (Cooney 1993, p. 2221). Nonetheless, the process of enacting laws is a form of human behavior, thus the *content* of what is enacted is amenable to scientific explanation (see Parsons 1951, p. 3). As Cooney (1993, p. 2221) remarks, "[I]egal sociologists already have made considerable progress in developing a theory of legal behavior, but a complete theory of formal legal rules would constitute a major contribution to the sociology of law." As noted earlier, Black's (1976) theory of law predicts and explains the variable aspects of law: its style, form, and quantity (i.e., severity); therefore in a very meaningful way, his theory constitutes that "major contribution" to the sociology of law.

Similar to the conflict perspective, Black's theory recognizes the importance of economic and political inequality in explaining racial/ethnic disparities in criminal processing, even after holding legitimate case characteristics constant (Collins 2002). As noted above, blacks and Hispanics have considerably less wealth, capacity for collective action, and social control than whites. Economic and political power, however, are not the only important determinants of legal behavior. The horizontal (morphological) and symbolic (cultural) dimensions of social life exert an independent effect on legal behavior. Although in some contexts, the economic and political dimensions may be of primary importance, this need not necessarily be the case (Black 2000a). Black's full articulation of the social life (i.e., all five dimensions) allows this theory to explain what conflict theory explains, and more.

Both interactionist theory and Black's theory of law emphasize social relations. However, interactionists place primary importance to the ideas and meanings that are created during interaction (and influenced by the social structure), whereas Black's theory draws its explanatory power from the social positions of those in a conflict, not the motivations that presumably follow from these statuses (see Uggen and Kruttschnitt 1998, p. 345). The interactionist perspective's limited scope largely results from that fact that the meanings and motivations that are central to their paradigm are rooted in particular settings, and are unlikely to apply freely across time and place (see also Cooney 2002). While Black's theory does not discount the importance of motivations (and the middle-range theories that emphasize motivations), his focus is on the social positions and relations that give rise to these various motivations (Turner 2002, p. 667; Uggen and Kruttschnitt 1998, p. 345). Interactionists' overemphasis of the importance of the symbolic dimension of social life (i.e., meanings, ideas, values) leads to an understatement of the importance of the independent effects of the other objective features of social life (see Brittan 1981). By refusing to give theoretical supremacy to any particular dimension of social life, while simultaneously emphasizing the tremendous importance of each social sphere, Black's theory is able to elucidate all of the relationships derived from formal legal, conflict, and interactionist perspectives under a single, highly general, explanatory framework.

*Parsimony*. Under Black's theory of law, legal behavior—including law-making and the response to law-breaking—is explained the social geometry of the conflict. As noted earlier, this social geometry is comprised of the social characteristics of the people involved, the relationships these individuals have with one another, and the larger social context in which these individuals interact (Black 1979a). Black identifies five variables in his theory, corresponding to the five dimensions that define human interaction: vertical, horizontal, symbolic, corporate, and normative. He also argues, however, that a "simple" or "elegant" theory is only preferable when it explains as much or more than a complex formulation (Black 1995, p. 838). Black's theory of law is preferable to formal legal, conflict, and interactionist perspectives because it is capable of explaining all of the relationships identified by those perspectives much more succinctly.

*Testability*. As the predictability and measurability of a theory increases, so does its testability. The most testable of theoretical formulations are those that are both falsifiable and stated in quantitative language (see Black 1995; Popper [1934] 1968). Black states his theory of law is highly testable because he explicitly identifies both the direction and functional form of the relationship between his five dimensions of social

space and the law. Like most social theory, Black's theory does not precisely specify how a change in one aspect of social space produces a defined change in law (see Horwitz 1990, p. 15); however the nonlinear relationships that he specifies provide a general idea of when a relationship is either strengthening or weakening (see, generally, Blalock 1967; Tittle 1995). According to Braithwaite and Biles (1980, p. 334), "The strength of *The Behavior of Law* lay in the fact that because it was in explicit propositional [*sic*] form, it was eminently testable." Similarly, Myers (1980, p. 854) notes that Black's theory "admirably captures an essential facet of the scientific enterprise: the reduction of phenomena to fundamental testable principles."

Nonetheless, several analysts have criticized Black for failing to clearly establish the order, organization, and interrelationships among the independent variables in his theory (see Bernard 2002, pp. 650–51; Braithwaite and Biles 1980, pp. 334, 338). These failures, according to these analysts, prevent the theory from being maximally testable. Black (2000a) has defended his decision to not rank the relative explanatory power of his dimensions of social space, arguing that these dimensions lack a common unit of measurement *and* that imposing an order among the variables undermines the generality of the theory because the relative explanatory power of the various dimensions vary across time and space. Bernard (1995) argues, however, that the complexity of the theory requires Black, to a certain extent, to specify the order or organization among his independent variables. For example, Bernard (2002) notes that researchers have attempted to test several of Black's propositions by examining whether Black's dimensions of social space explain variation in citizen's complaints to the police (e.g., Avakame *et al.* 1999; Braithwaite and Biles 1980; Gottfredson and Hindelang 1979). These tests have been largely unsupportive of Black's theory, but Bernard (2002) argues that the unsupportive findings may primarily stem from confusion surrounding the order and organization of Black's independent variables (see also Mastrofski et al. 2002). As noted above, penal law is but one of the four styles of law, and as a result, there should be more penal law when there is less compensatory, therapeutic, or conciliatory law (Bernard 2002, p. 650). Therefore, if penal law (e.g., complaints to the police) is the dependent variable, the other three styles of law must become independent variables. An adequate (and accurate) test of Black's theory, then, would include eight independent variables: the five social dimensions and the three other styles of law. According to Bernard, access to the other styles of law would be most important in explaining the use of penal law. More generally, Bernard suggests that because law is "governmental social control," nongovernmental social control is the most important social dimension (i.e., the normative sphere). Because the vertical and normative dimensions of social space may be correlated (i.e., individuals with low vertical status may have less access to nongovernmental social control), a finding that individuals with less rank are more likely to mobilize penal law (e.g., call the police) would be consistent with Black's theory if these individuals had less access to nongovernmental social control. Indeed, Black (1976) expressly states that the mobilization of law remains relatively uncommon and individuals are much more likely to use non-legal forms of social control when they are available to them (see also Baumgartner 1978).

It is debatable, however, whether Black must necessarily order the theoretical importance of his social dimensions and whether the actual *styles* of law should constitute additional explanatory variables in the theory. Recall that Black specifies the bivariate

relationships of his theory under the condition that all non-sociological variables and the other dimensions of social space are held constant. Therefore a proper test of the relationship between the vertical dimension of social space (i.e., stratification) and law would require the analyst to hold normative status, which includes the capacity for nongovernmental social control (i.e., authority), constant. While the normative dimension of social space may be of primary importance in some contexts, this need not necessarily be the case across all contexts. For example, in contexts where there is limited variability in the distribution of authority, normative status may be of limited explanatory importance and one or more of the other dimensions of social space may be of greater significance. As noted earlier, an *a priori* ranking of the five dimensions of social space is unadvisable and would improperly limit the generality and potential accuracy of Black's theory. The previous failures of analysts to adequately consider, operationalize, and control for the variable aspects of normative status, which includes the capacity to mobilize various forms of non-governmental social control, may more accurately reflect deficiencies in methodological rigor rather than a shortcoming of Black's theory.

A similar criticism of Black's theory is its exclusive focus on the additive—rather than interactive—nature of the dimensions of social space (see Lessan and Sheley 1992, pp. 673–74). While Black's emphasis on the additive effects of his variables may fail to fully capture the complexity of interrelationships among his five social dimensions, it allows the basic properties of the theory to be explored in a straightforward manner. That is, Black does not require the analyst to explore both the additive and nonadditive nature of these relationships to test his fundamental propositions. If his theory did require the analyst to explore such relationships, as many as 31 relationships could potentially be included in the model (corresponding to all of the different relations among the five different social dimensions),<sup>376</sup> thus severely undermining both parsimony and testability. Nonetheless, Black's theory is fully capable of accommodating the various interactions among the five dimensions and their relationship to law. General statements about these interrelationships may not be advisable for the same reason that ranking the explanatory importance of the dimensions is unadvisable: such an enterprise would likely limit the generality of the theory. While the specification of such interrelationships may be useful for middle-range theories (see Turner 2002), they need not figure prominently into Black's broader explanatory framework.

It is also unadvisable to include the other styles of law as independent variables, as Bernard (2002) suggests. Black (1976, pp. 4–6) maintains that the style of law is a quantitative variable and his theory is able to explain variation in the style of law across time and space. Both the quantity of law, in general, and the quantity of each of these styles of law are aspects of the behavior of law—they are both dependent variables in Black's theoretical framework. Attempting to predict one style of law with another style of law would be circular—it reduces to "law predicts law." Bernard's analysis should be applauded, however, for underscoring the importance of not only the types, but also the various styles of nongovernmental social control. Because law varies inversely with other social control, all else constant (Black 1976, p. 107), Black's theory predicts that penal, compensatory, therapeutic, and conciliatory law will all vary inversely with penal, compensatory, therapeutic, and conciliatory styles of nongovernmental social control (Black 1983, 1990). By clearly holding the distinction between legal and non-legal (i.e.,

<sup>&</sup>lt;sup>376</sup> The formula for calculating the number of different possible combinations of the five social dimensions is as follows:  ${}^{n}C_{k} = n!/k!(n-k)!$ ; where *n* is the number of objects in the set and *k* is the number of objects that can be grouped (Finkelstein 1978, p. 34).

nongovernmental) forms of social control, Black's theory avoids the tautological theorizing that Bernard suggests.

It should also be emphasized that Black's form of theorizing, the covering law model, which is descriptive and focuses on identifying general patterns, is particularly consistent with the most popular analytical technique employed in the social and behavioral sciences: regression analysis. According to Berk (2003, p. 18), regression analysis is most powerful (and appropriately used) as a descriptive technique. Conventional regression analysis can provide only conditional distributions (e.g., conditional means and conditional variances). Causal inferences, on the other hand, "are based on information not in the data or the regression model [e.g., how the data were generated]...regression results are a way to describe important physical relationships of which causal links are justified elsewhere" (p. 9) (see also Heise 1975, pp. 12–17). In fact, the information that most social scientists are interested in, such as significance tests, confidence intervals, "explained variance," and causal relationships, rests on assumptions that actual social scientific data often fail to meet, but are required by the regression model.

As noted above, Black's theory of law avoids making causal statements (see Section 6.2.1). Also, when discussing supportive evidence for his theoretical propositions in *The Behavior of Law*, Black never mentions *t*-statistics, *p*-values, or the proportion of variance that should be explained. His theory simply specifies how variation in law corresponds with variation in each of the five dimensions of social space. Although his theory is explicitly stated in quantitative language and causal explanations are compatible with this theory, neither is central to the explanation of legal behavior.

269

*Empirical Validity.* As Cooney (1997b, p. 318) apply notes, a theory as general as Black's theory of law must incorporate findings from multiple disciplines, including anthropology, criminology, history, and political science. Indeed, in the original presentation of his theory (see Black 1976) and in subsequent explications (see Black 1995, 2000a), Black assembles an impressive body of research literature in support of the core propositions of this theory. The bulk of the evidence presented by Black is largely indirect, drawn primarily from anthropological and cross-cultural research; yet initial explicit tests of Black's theory appeared to produce mixed results. Gottfredson and Hindelang (1979) and Braithwaite and Biles (1980), for example, discovered that lowerstatus individuals were more likely to call the police than high-status individual, holding the level of perceived offense serious constant—a finding in direct contradiction of Black's theory of law. Myers (1980) analysis of several decision stages in the criminal charging-and-sentencing process not only failed to find support for Black's theory, but also challenged its applicability to multiple decision stages. Conversely, early tests by Kruttschnitt and colleagues' (1981, 1982, 1985) provided support for Black's theory, discovering that gender differences in pretrial sanctions and criminal sentencing were largely explained by stratification, integration, respectability, and level of familial social control. Dannefer's (1984) analysis of the juvenile justice system also provided strong support for Black's theory, noting that greater relational distance was associated with harsher sanctioning.

The debate over the ultimate validity of Black's theory continues after 25 years of empirical testing (for discussions, see Bernard 2002; Cooney 2002). Norris and colleagues' (2006) comparative study of the potential use of police force in Mexico,

Venezuela, and the United States, Avakame *et al.*'s (1999) examination of calls to the police using National Crime Victimization Survey data, Doyle and Luckenbill's (1991) analysis of the use of law for the social control of collective problems, and Massey and Myers (1989) investigation of the relationship between informal social control and executions in the Post-Reconstruction South all find very little support for Black's theory. These studies have been criticized, however, for being marred by fundamental errors. Cooney (2002, p. 660) suggests that "the twin flaws of accepting poor evidence and rejecting good evidence" undermines critics' claim that Black's theory has received only limited support. With respect to accepting poor evidence, he argues that many researchers fail to include many of the relevant variables, such as important defendant and victim characteristics and relationships, contextual factors, and information about third parties (e.g., witnesses, attorneys, judges, and jurors).<sup>377</sup> Horwitz (2002) maintains that most criminologists and legal sociologists are not accustomed to thinking sociologically, and as a result their preferred research methods, such as random sample surveys, ignore the relational and contextual aspects of social situations that Black's theory uses to explain and predict law and other types social control (see also King 1997, p. 5).<sup>378</sup> Horwitz states, "The rigorous study of social geometry barely exists in the

<sup>&</sup>lt;sup>377</sup> Similarly, Mears (1998a, p. 672) notes, "Only rarely are data about victims, court practitioners, or organization, culture, political, or social contexts considered at all, much less in a systematic manner." In recent years, however, researchers have paid increasing attention to the characteristics of third parties—particularly trial judges (see, e.g., Johnson 2006a; Spohn 1990a, b; Steffensmeier and Britt 2001: Steffensmeier, 1999 #727; Welch *et al.* 1988). Many of these studies reveal that judge characteristics do matter, but they often differ with respect to the direction and magnitude of these effects. Of particular note is Abram's (2006) recent study of the impact of defendant race on interjudge sentencing disparity in Cook County, Illinois between 1985 and 2005. Not only does Abram discover significant between-judge variation in ratio of sentence lengths between racial groups, but also that judge characteristics such as age and previous work experience as either a prosecutor or public defender predict their racial gap in sentencing.

<sup>&</sup>lt;sup>378</sup> There are numerous examples of seriously flawed studies reporting unsupportive results for Black's theory. Mooney (1986), for example, purports to test Black's theory and concludes that "[the] lack of empirical support...seriously calls into question the validity of Black's propositions and empirically

discipline [of sociology]" (p. 643). With respect to rejecting good evidence, both Cooney and Horwitz note that critics of Black's theory ignore or dismiss the substantial amount of supporting evidence presented by Black and others (see, e.g., Baumgartner 1992; Horwitz 1990) because much of this evidence is historical, cross-cultural, and qualitative. Sociology, they contend, remains highly ethnocentric—rarely investigating other times and places—and preoccupied with applying precise statistical techniques to large quantities of remote data, rather than the careful collection and analysis of valid data.<sup>379</sup>

In recent years, scholars have also produced a substantial amount of evidence in direct support of Black's theory. Hembroff's (1987) study of the perceived seriousness of crimes revealed that the social structure of a conflict predicted respondents' seriousness rankings, net of the actual harm caused by the crime. Cooney (1992) discovered that racial discrimination in police arrests were largely explained by differences in vertical and horizontal status. He also discovered that social status is

documents several conceptual and methodological weaknesses" (p. 742) and "Black's theory of law is an inadequate theoretical framework which ignores the complexity of the phenomena under investigation and the variables on which they are dependent" (p. 747). An examination of her methodological approach, however, seriously undermines any inferences that can be drawn from the study. Mooney surveys approximately 300 undergraduates, inquiring about their own "law use" (i.e., "any act or behavior the a student which, by its nature, initiated a legal procedure and/or invoked a legal norm for the purpose of an official decision being made) (p. 736), the "application of law" (i.e., "any act or behavior by a university official which, by its nature, imposed law on a system [member], that is, acted as a means of social control"), and measures their location on three of Black's five dimensions of social space (i.e., stratification, culture, and organization). Remarkably, Mooney fails to examine Black's key theoretical construct—the *shape* of social space—defined by the characteristics of the individuals involved, their relationships with one another, and the social context in which these individuals interact. By neglecting to include any information about the social characteristics of other principals (including supporters) and third parties, as well as failing to hold legal rules and legal facts constant, her analysis is wholly incapable of speaking to the empirical validity of Black's theory.

As noted earlier, Black's form of theorizing is consistent with the fundamental aim of regression analysis: the description of associations among variables. However, data subject to regression analysis must contain valid information about relational and contextual aspects of social situations. In fact, regression analysis has been used to explicitly model the relational and contextual associations highlighted by Black when the appropriate data are collected (e.g., Litwin 2004; Novak *et al.* 2002).

<sup>&</sup>lt;sup>379</sup> Cooney (2002, p. 660) also criticizes much of the qualitative work in sociology for being "driven more by a search for meaning than for explanation, holding that something useful will result from speculating about the contents of the sealed chamber that is the human mind, as it acts in particular settings."

directly related to the amount and strength of evidence and testimony that can be gathered and used in a case *and* that legal testimony about a social inferior was more likely to succeed than identical testimony about a social superior (Cooney 1994). Mullis's (1995) examination of medical malpractice litigation revealed that the vertical, relational, and organizational distances between the health care provider and the patient were important in predicting claims and case outcomes. Philips and Cooney's (2005) study of individuals incarcerated for homicide and serious assault discovered that the third-party structure of the conflict, based on the social locations of all of the third-parties present, predicted the type of involvement of the third-parties (i.e., partisans, settlement agents, or disinterested) and whether conflict escalated to violence. At the macro-sociological level, Borg's (2001) research supports Black's theory of law, discovering that aggregate measures of stratification, morphology, culture, organization, and nongovernmental social control explain variable homicide clearance rates across locations (see also Litwin 2004).

Much of the research on racial/ethnic bias and sentencing conducted independently of Black's theory confirms the theory's predictions. Similar to formal legal theorists, Black predicts that racial/ethnic differences in criminal charging-andsentencing will be attenuated when certain legitimate legal characteristics are taken into account; however Black's attributes this attenuation to the fact that many of the legal characteristics are themselves indicators of social status. For example, a defendant's prior criminal history is an indicator of her or his normative status (respectability) and an economic motive for a crime is often an indicator of the defendant's vertical status. Individuals' relational and radial statuses, indicated by their positions in, and strong ties to, their communities influence the likelihood of pre-trial detention and the amount of bail required for pre-trial release (Garber, Klepper, and Nagin 1983; Hagan and Bumiller 1983).<sup>380</sup> Even the heinousness of a crime, whether lethal or non-lethal, can be an indicator of cultural or normative status.<sup>381</sup> All crimes, including murder, may be conducted in a more or less conventional manner.<sup>382</sup> The more heinous or vile the murder, the less respectable and more unconventional the offender will appear to be, even after holding prior criminal history constant.

Much of the research on the impact of "quasi-legal" factors on the criminal charging-and-sentencing process also supports Black's theory (Kruttschnitt and McCarthy 1985; Nobiling *et al.* 1998). Defendant characteristics such as age, employment status/history, and familial status may be directly or indirectly built into some aspects of the criminal law (Hagan and Bumiller 1983). Formal legal theorists have difficulty clearly distinguishing between legitimate and illegitimate factors because what is considered "legitimate" often varies depending on social context and from one stage of criminal processing to the other (Hagan and Bumiller 1983, p. 5). Black's theory avoids this confusion by simply recognizing that these factors are indicators of the social geometry of the case. The influence of these factors, however, will depend on the precise location and direction of the conflict in multidimensional social space.

Nonetheless, legal and quasi-legal characteristics do not completely capture the social geometry of the case; therefore other aspects of the social organization of the case

<sup>&</sup>lt;sup>380</sup> Individuals who are granted pre-trial release are more likely to respect the legal system and comply with court orders, keep their jobs and their homes, and assist in the preparation of a meaningful defense (Colbert, Paternoster, and Bushway 2002, p. 1783).

<sup>&</sup>lt;sup>381</sup> Black (2004c) also suggests that the use and lethality of weapons can be predicted from the conflict structure—the more social distance adversaries are to each other, the greater the likelihood of deadly force. <sup>382</sup> Cooney (2006b, p. 6) notes that certain forms of homicide, such as dueling and lynching, attracted increasingly severe sanctions after they began declining in frequency.

may be captured by the racial/ethnic characteristics of the parties in the case. Like conflict theorists, Black predicts that groups relatively low in vertical, organizational, and normative status are disadvantaged in the criminal justice process; thus, research discovering that racial/ethnic minorities receive harsher treatment in the criminal sentencing process even after legitimate legal rules are taken into account also supports Black's theory.

As Hagan and Bumiller (1983, p. 6) note, "it is very important to simultaneously consider both individual process and structural context because it is impossible to determine the specific effects occurring at one particular level without considering the effects at the other level." Black's theory emphasizes the micro and macro dimensions of social life, making the social characteristics and relationship of the parties involved in disputes central to his explanation of legal behavior. However, by recognizing that motivations cannot be precisely derived from social contexts (see Miethe and Drass 1999) and that attitudes account for only a small part of the variation in human behavior (see Worden 1989), Black's emphasis on the observable aspects of human interaction allows him to present a systematic theory of how the social geometry of the conflict predicts and explains legal behavior.

Evidence believed to support the interactionist model also supports Black's theory of law; however Black makes no reference to the human mind or emotions. Similar to interactionists, Black predicts that extra-legal factors such as age, gender, and race/ethnicity neither uniformly benefit nor uniformly disadvantage certain groups (see, e.g., Kurlycheck and Johnson 2004; Spohn and Holleran 2000). But unlike interactionists who suggest that the "interactive" effects of these extra-legal factors results from the

unique ways in which legal decision-makers define and understand these social characteristics in particular contexts (see Steen, Engen, and Gainey 2005), Black argues that the sociological relevance of these factors depends on the degree to which these characteristics correspond to a social location. According to Black's theory, young black and Hispanic males receive the harshest sentences because, in the aggregate, they are the lowest in social status (Steffensmeier et al. 1998). Like interactionists, Black also predicts that the larger social context in which criminal case processing occurs influences legal behavior. Recall that the social geometry of a dispute is not only defined the characteristics of the parties involved and their relationships with one another, but also "the larger social context in which they interact" (Black 1979a, p. 19). Rather than focusing on the assumed psychological impact of the social environment in legal cases (e.g., Akers 1998), Black explains the impact of the social environment on legal behavior by the manner in which differentially locates a dispute in social space. Hold the direction and distance of a dispute constant, and the location of a conflict predicts and explains its fate; therefore downward law among the wealthy will be greater than downward law among the poor, even when the law spans the same distance. Without knowledge of the larger social context in which disputes occur, the location of the dispute in social space is unknown.

A consistent finding in the literature is that racial/ethnic disparities in criminal sentencing vary across urban and rural areas, net of legal considerations; however the direction of this relationship is unclear. Analyzing case dispositions in Alberta, Canada, Hagan (1977) discovered that minorities were sentenced more harshly in rural areas. Kramer's (2002) examination of sentencing decisions in Pennsylvania after the
implementation of sentencing guidelines also revealed that minority defendants were less likely to receive downward sentencing departures in rural areas. Conversely, Myers and Talarico's (1986) research in Georgia discovered that racial/ethnic disparities were greatest in urban areas. Despite these contradictory findings, these analysts attempt to explain urban and rural differences through the impact of bureaucratization on criminal sentencing. Some analysts hypothesize that racial differences in urban contexts will be less pronounced than in rural areas because increased bureaucratization in urban legal systems leads to more rational decision making, based primarily on legally relevant factors (see Austin 1981; Miethe and Moore 1986; Pope 1976). In contrast, other theorists have argued that increased bureaucratization in urban areas will exacerbate racial differences in criminal processing resulting from an increased emphasis on efficiency, which leads law enforcement agencies to disproportionately focus on members of weak and powerless groups (Chambliss and Seidman 1971; Myers and Talarico 1987). Moreover, the latter group posits that growing minority presence in urban areas causes members of the majority group (i.e., whites and powerful elites) to increasingly rely on the coercive power of the state to reduce the threat of competition (Blalock 1967; Hawkins 1987; Jacobs and Carmichael 2002). Thus, both camps attempt to explain these completely contradictory findings with the assumed goals decisionmakers in urban and rural areas.

Perhaps the primary reason for these contradictory predictions is that crude urban/rural distinctions ignore important structural differences between these social contexts (see Berk *et al.* 2005). According to Black's theoretical framework, it is not important whether a case is tried in an urban or rural location, but how these jurisdictions may systematically differ along the five social dimensions (see also Jackson 1986). He posits that differences in criminal processing can be predicted and explained by observable differences in the social organization of these jurisdictions, and without regard to the objectives of decision-makers. For example, with respect to economic stratification, rural communities have higher poverty rates than urban communities (14.3 percent versus 11.8 percent) (George, Pinder, and Singleton 2004). Also, the economic gap between blacks and whites is greater in rural communities than urban communities and the black poverty rate in rural communities is nearly 13 percentage points higher than the overall poverty rate of those same communities (George et al. 2004). Blacks in rural communities are also more likely to be chronically poor and rural minority communities comprise over half of all persistent poverty counties in the United States (George et al. 2004).<sup>383</sup> Residents of rural communities also tend to have fewer years of formal education than residents of urban communities (i.e., less cultural status). The South, which has one-third of the country's rural population, is home to half of all rural adults who have not completed high school. With respect to post-secondary education, 26 percent of adults in urban communities have completed college versus 15 percent of adults in rural communities. Rural blacks and Hispanics are half as likely to have completed college and twice as likely to lack a high school diploma as rural non-Hispanic whites. Moreover, blacks in urban communities are over twice as likely to hold a college degree as blacks in rural communities (the largest attainment difference among races), and Hispanics in rural areas have the lowest education attainment of any racial/ethnic group in the nation (U.S. Department of Agriculture 2003). In terms of morphological

<sup>&</sup>lt;sup>383</sup> Rural Minority Communities (RMCs) are defined as communities with at least a 33 percent minority population rate (George *et al.* 2004).

space, rural communities tend to be less racially integrated than urban communities. While whites comprise 69 percent of the total United States population, they comprise 82 percent of rural areas (George *et al.* 2004). These differences would suggest that blacks (and some Hispanics) are more likely to be "structurally disadvantaged" in these rural jurisdictions, and as a result, receive harsher penalties than whites, net of legally relevant factors.

It is also important to recognize that Blacks may not always be more disadvantaged in rural areas than urban areas. Although blacks are more racially isolated in rural areas, they may be more similar to whites in other respects than blacks in urban areas. Blacks and whites in rural areas are more likely to engage in similar activities outside of work, such as fishing and hunting, than black and white residents of urban areas (aspects morphological and symbolic space) (U.S. Department of the Interior 1999). Residential mobility among the poor in rural areas spans much shorter distances than in urban areas (Fitchen 1994), so blacks in rural areas may have greater capacities for collective action (corporate status) and collective efficacy (normative status). Differentiation (i.e., division of labor) also tends to increase with urbanization and population size and the severity of punishment increases with differentiation (Durkheim [1900] 1969). Conversely, there is less law where people are undifferentiated by function. Residents of rural areas, both black and white, are less likely to be interdependent than individuals in urban areas. This suggests that racial/ethnic disparities in criminal sentencing would be more pronounced in urban areas, holding the direction and distance of the conflict constant.

Although extremely general in scope, Black's theory of law identifies the most important aspects of social space and their relationships to legal behavior, thus providing a fairly nuanced understanding of the role of these various contexts in explaining racial differences in criminal sentencing. It must be emphasized that the preceding discussion was presented under the condition that the distance and direction of the conflict is held constant. The relative statuses of the parties in the conflict, third parities, and social distances that the conflict spans are also extremely important in understanding variation in legal behavior. Irrespective of whether a legal dispute occurs in a major metropolis or a rural township, a high status complainant will enjoy more legal advantages than her or his low status adversary. The social location of the conflict in a particular social context, however, will influence how large that legal advantage will be (Black 1976, pp. 27, 49, 68, 112).

Black's conceptualization of race/ethnicity as a proxy for social location avoids the significant problems associated with reifying race, and allows his theory of law to predict and explain the differential impact of race/ethnicity on criminal justice processing across time and place. The extensive research on racial/ethnic disparities in the capital charging-and-sentencing process reviewed in Chapter Four also supports Black's predictions. Studies conducted both before and after the *Furman* decision revealed that white-victim cases were significantly more likely to be noticed for the death penalty by the prosecutor, result in a guilty verdict, receive a sentence of death, be denied relief on appeal, and result in an execution, even after controlling for numerous legally relevant factors. These studies also discovered that black-offender/white-victim cases (both homicide and non-homicide capital cases) were the most likely to result in the death sentence. Again, rather than explain these differences with the motivations and goals of legal officials, Black explains these differences with the social organization of the case. For example, he hypothesizes that cases with a low-status offender and a high-status victim will attract the most law (e.g., black offender/white-victim), followed by cases between a high-status offender and a high-status victim (e.g., white-offender/white-victim), then between a low-status offender and a low-status victim (e.g., black-offender/black-victim), and the least amount law in cases with a high-status offender and a low-status offender and a low-status victim (e.g., black-offender/black-victim), and the least amount law in cases with a high-status offender and a low-status victim (e.g., black-offender/black-victim), then between a low-status offender/black-victim) (Black 1976, p. 116).

Much of the research on regional differences in the administration of capital punishment is also supportive of Black's theory. Scholars have long recognized that the former Confederate states execute a disproportionate percentage of their population, although there is very little difference in support for capital punishment between Southerners and non-Southerners (Borg 1997). Recent evidence suggests that certain structural characteristics, many of which are indicators of Black's five dimensions of social space, significantly influence the administration of capital punishment in the direction hypothesized by his theory of law. For instance, these structural characteristics influenced whether (and how quickly) states reinstated capital punishment after the *Furman* decision (Jacobs and Carmichael 2002) and how often death sentences were reversed on appeal (Gelman *et al.* 2004).

Finally, the actual *content* of capital statutes supports Black's predictions. As discussed in Chapter Two, the U.S. Supreme Court lifted the moratorium on capital punishment after approving the "guided-discretion" statutes in Georgia (*Gregg v. Georgia*), Florida (*Proffit v. Florida*), and Texas (*Jurek v. Texas*). These statutes

281

enumerated specific circumstances that prosecutors, judges, and jurors were required to consider when making charging and sentencing decisions. These circumstances were divided into two general groups: aggravating factors and mitigating factors. The aggravating factors were specific circumstances that were believed to make homicide defendants more culpable, whereas mitigating factors were circumstances believed to make the homicide defendants less culpable. Although some death penalty statutes do not include a specific list of mitigating factors, all death penalty statutes allow prosecutors, judges, and jurors to consider any mitigation evidence presented by the defendant (see Section 2.3).

Statutorily defined aggravating circumstances permit increases in the amount of law that can be applied to a case (*Ring v. Arizona*, 536 U.S. 584 [2002]); therefore according to Black's theory of law, they can be predicted and explained by the social geometry they reflect. Although these aggravating circumstances slightly vary from state to state, nearly all statutes include the following situations: (1) the offender had a prior conviction for a capital felony, (2) the offender was engaged in a contemporaneous felony, (3) the offender committed the murder while already incarcerated, (4) the murder was committed for the purpose of receiving money or something else of value, (5) the murder was committed against a peace officer, officer of the court, or firefighter during or because of exercise of her or his official duty, and (6) the murder was especially heinous, atrocious, and cruel or outrageously vile.<sup>384</sup>

<sup>&</sup>lt;sup>384</sup> "Most state laws now identify six to twelve aggravating circumstances, at least one of which must be present before a convicted murdered is eligible for a death sentence...[t]he two most important aggravating circumstances, however, are the 'contemporaneous offense' and the 'vile murder' circumstances. These two statutorily designated aggravating circumstances lead all others by far, in both numbers of defendants they make death eligible and in the number of cases in which appear that actually result in death sentences" (Baldus *et al.* 1990, p. 22; see also Baldus *et al.* 1986a, p. 138).

Note that only two of these factors specifically address the conduct of the offender or the actual harm inflicted by the offender on the victim (i.e., contemporaneous felony and the heinousness of the homicide).<sup>385</sup> The other four aggravating circumstances differentiate murder cases irrespective of the level of harm resulting from the crime. Recall that Black's propositions predict and explain legal variation net of offender conduct. The important question, then, is "Why were these particular factors developed and not others?" According to Black's theory, several of these circumstances reflect the particular social positions of the offender and/or victim and their social relations. For example, prior conviction for a capital felony and being incarcerated at the time of the murder are indicators of the normative and radial status of the offender (i.e., respectability and integration). Ex-offenders and individuals who are currently incarcerated are significantly more likely to be marginalized and viewed as disreputable than offenders without a prior criminal record and who are not incarcerated. Murder for financial gain is often indicative of the vertical status of the offender and victim. Murder for financial gain is frequently the result of a murder-for-hire arrangement, and in these situations the victim typically has higher vertical status than the offender. The offender, an "assassin," is also more likely to be marginalized and culturally unconventional by way of his profession. The aggravating circumstance for the murder of a peace officer,

Rosen (1986) argues that the "heinous, atrocious, and cruel" aggravating circumstance (commonly referred to as the HAC aggravator) violates the Eighth and Fourteenth Amendments due to its vagueness. Analyzing appellate decisions from various death penalty jurisdictions, he discovered that the HAC aggravator is applied inconsistently, thereby undermining the constitutional mandate established in *Furman*. According to Rosen and others (e.g., Blume *et al.* 2004), the HAC aggravator ostensibly makes every homicide death-eligible.

<sup>&</sup>lt;sup>385</sup> The U.S. Supreme Court has debated the legitimacy of considering certain background characteristics of offenders in capital cases. This issue was particularly salient in challenges over the constitutionality of the juvenile death penalty. Several Justices believed that it was necessary and proper to consider numerous factors beyond the actual injury caused by the crime in capital cases when assessing an offender's culpability. As Justice Brennan noted in his dissenting opinion in *Stanford v. Kentucky* (492 U.S. at 394 [1989]): "Proportionality analysis requires that we compare the gravity of the offense, understood to include not only the injury caused, but also the defendant's culpability, with the harshness of the penalty."

officer of the court, or firefighter is indicative of the horizontal status of the victim. Recall that an important aspect of morphological space is differentiation. Interdependence, defined as the degree to which individuals need one another to survive and prosper, is a direct result of differentiation (Black 1990). Certain social positions may be more functionally important to the group (or society) than others, and consequently, individuals who occupy these positions have greater functional status (Black 2000a). In modern society, police officers, district attorneys, and judges play a central role in the everyday functioning of society, and as a result, they have greater functional status (an aspect horizontal space). Individuals in these positions also tend to have greater normative status (i.e., authority).

Prosecutors are not required to seek the death penalty when there is evidence of one or more statutorily defined aggravating circumstances in a homicide case. Similarly, most death penalty statutes do not require judges and juries to sentence an individual to death when there is evidence of one or more aggravating circumstances. A few states require judges and juries to impose the death penalty if an aggravating factor is present and the judge or jury cannot find any mitigation factor that would justify leniency; however these states broadly define "mitigating factors," so judges and juries can easily vote for leniency if they so choose.

Mitigating factors typically do not come into play until the sentencing phase of capital trials. While it is true that prosecutors may consider these factors in the charging phase and judges and juries may consider these factors in the guilt/innocence phase, the primary purpose of mitigating evidence is to assist judges and juries in deciding the appropriate punishment for a defendant who has been found guilty. It is important to

note that most death penalty statutes stipulate that judges and juries only use this evidence as a general guideline, but they are not required to vote for leniency. In contrast, evidence of aggravating factors must be present in order for judges or juries to impose a death sentence. The most common types of mitigation evidence listed in death penalty statutes are: (1) the youthfulness of the defendant, (2) whether the victim's conduct contributed to her or his death, and (3) the defendant's level of involvement in the murder.

Recall that Black's theory of law predicts and explains variation in the style of law, which he defines as the language and logic by which law defines and responds to deviant behavior. He notes that one style of law is greater than another, with penal law generally the greatest, followed by therapeutic, compensatory, and conciliatory (Black 1976, p. 106). Black also notes that, in reality, law may deviate from these styles in their pure form, combining various elements from the different styles (p. 5). In fact, capital murder cases often combine elements of penal and therapeutic law.<sup>386</sup> Although a murder defendant in a capital case faces the most severe punishment that the state can impose, she or he may still be seen as someone who is ill and deserving of some form of help or assistance. Just as Black's theory of law predicts and explains the quantity of law with the social geometry of the case, it also explains the quantity of each of these styles of law in a case. For example, the youthfulness of an offender is seen as a mitigating factor because age is often a proxy for normative status. In the aggregate, young defendants are more respectable than older defendants in capital cases because they have less extensive criminal records (i.e., less likely to have been subject to governmental social control in

<sup>&</sup>lt;sup>386</sup> The compensatory style of law is present when families of murder victims choose to sue the offender for monetary damages. These lawsuits, however, occur in civil court, not in criminal proceedings.

the past). The more respectable the offender, the more likely she or he will be subject to therapeutic, rather than strictly penal, law. This largely explains the separate legal systems for juvenile and adult offenders—the former is primarily concerned with rehabilitation and conciliation, while the latter is primarily concerned with punishment/compensation.<sup>387</sup>

Hold constant the offender's behavior, and victim characteristics also predict and explain the quantity and style of *potential* law in a death penalty case (see Baumer *et al.* 2000; Sundby 2003). For example, with respect victim conduct, the victim's involvement in her or his death is often an indicator of the victim's normative and cultural status positions. A victim who assisted in her or his own death or was engaged in an activity that could reasonably result in her or his death is likely to be viewed as less respectable and less conventional (see also Luckenbill 1977; Wolfgang 1957). The less conventional and respectable the victim in a homicide case, the less amount of law the case attracts and the more likely that elements of therapeutic law will enter the case.

In sum, Black's theory of law provides a significantly better explanation of legal behavior than formal legal, conflict, and interactionist perspectives. His deliberate avoidance of psychology, one-dimensionality, microcosms and macrocosms, individualism, and teleology allows this theory to achieve the highest standards of scientific theory: generality, parsimony, testability, and empirical validity. Although the ultimate purpose of his theory is to identify permanent social processes (i.e., universal laws), the predictions derived from the theory "display a wide range of generality and observability, including predictions for spatially- and temporally-specific phenomena and

<sup>&</sup>lt;sup>387</sup> "Researchers should develop sociological theory that's applicable to juvenile justice, civil, and criminal court decision-making" (Mears 1998a, p. 717).

processes" (Jasso 1988, p. 2). As noted above, Black is able to predict and explain the same facts and relationships as the dominant theories of legal behavior under a single explanatory framework.

Black acknowledges that his theory of law is radically different from the psychological study of law. His theory predicts and explains legal behavior without regard to individuals and psychology (but cf. Frank 1930; Hunt 1993). He explicitly states, however, that his theory "is not at odds with psychological assumptions or theories…but is simply a different kind of explanation, a different way to predict the facts" (Black 1976, pp. 7–8). Following the early theoretical traditions of Durkheim ([1895] 1962) and Simmel (1909), Black advocates a purely sociological theory of law, distinct from psychology, with its own concepts, imagery, and framework of analysis (see also Mayhew 1980).

#### 6.3 RESEARCH HYPOTHESES

In *The Behavior of Law*, Black presents his theory as a set of testable propositions, specifying the direction and functional form of the relationship between the key variables of his theory and the style and quantity of law (see Section 6.2.2). The hypotheses outlined by Black, however, are extremely abstract and need to be tailored to address the analyst's specific research questions. With respect to the death penalty process, several core hypotheses can be derived from Black's theory.

The first *five* hypotheses presented below exclusively focus on the impact of extra-legal factors *not* specifically mentioned in death penalty statutes; that is, they predict relationships between extra-legal factors and legal decision-making irrespective

of the legally legitimate characteristics present in a case. This is done for two important reasons. First, Black argues that, with two exceptions (see Section 6.2.4), the social geometry of a case successfully predicts and explains legal behavior, independent of legal rules. It is important, then, to clearly distinguish between legally legitimate and illegitimate factors when testing the theory to assess the explanatory power of Black's five dimensions. Second, Black posits that legal rules, themselves, are dependant variables in his theory of law, and as a result, can be predicted by the theory (see Section 6.2.5). Technically, one cannot properly test the "causal" effect of explanatory factors (i.e., indicators of the dimensions of social space) by controlling for other factors that are, themselves, consequences of those explanatory factors (i.e., legal rules). By holding constant something that is itself affected by the causal variable(s) of interest, one removes precisely the effect one is attempting to study (see King and Zeng 2006, p. 148; Rosenbaum 2002, pp. 73–74).<sup>388</sup>

The remaining *three* hypotheses control for legally legitimate characteristics in addition to the legally illegitimate (and legally suspect) characteristics identified in the proceeding six hypotheses. According to Black, the social structure of a case predicts legal behavior when the legal characteristics of the case are held constant, therefore a complete test of the theory should control for these legal characteristics (i.e., the "technical core"). He suggests, however, the technical core only successfully predicts

<sup>&</sup>lt;sup>388</sup> Stated differently, if one or more control variables are "post-treatment" variables (i.e., variables that are, themselves, consequences of the treatment variable of interest), then the post-treatment variables are likely to change when the treatment variable changes, and therefore it becomes impossible to interpret the model as revealing the effect of the treatment variable on the outcome of interest when all other variables are held constant. Typically, analysts need to consider both post-treatment bias and omitted variable bias together, but unfortunately one usually cannot be addressed without making the other worse (King and Zeng 2006, pp. 148–49). Although there are no general solutions to the problem of post-treatment bias, analysts may examine "multiple-variable causal effects" (i.e., joint causal effects) because it does not require holding constant variables that not stay constant in nature (p. 149).

and explains legal behavior in cases with particular structural configurations. As noted above, controlling for an effect of the cause potentially presents an endogeneity problem and may likely lead to incorrect estimates of the impact of the social geometry of the case on legal behavior, nevertheless it is important to control for the legal factors in a case because the alternative theoretical approaches discussed earlier in this chapter, particularly formal legal and interactionist perspectives, posit that legal characteristics are primarily responsible for legal outcomes (see Section 6.1.2). The extent to which the endogeneity of legal variables impacts the results is examined in Chapter Eight.<sup>389</sup>

## 6.3.1 Hypothesis One

Black conceptualizes race/ethnicity as a proxy for social status and, in the aggregate, black offenders and victims have lower social status than their white counterparts (see Section 6.2.4). Black (1976, p. 17) suggests that the social organization of cases that attracts the most law is that with low-status offenders and high-status victims (e.g., black-offender/white-victim). Cases that attract the second greatest quantity of law consist of high-status offenders and high-status victims, followed by cases with low-status offenders and low-status victims, and finally cases high-status offenders and low-status victims (see *supra*, p. 281). According to Black's theory, *black-offender/white-victim cases have the greatest likelihood of being noticed for the death penalty, receiving unfavorable plea bargains, and receiving a more severe sentence at the penalty phase.*<sup>390</sup>

<sup>&</sup>lt;sup>389</sup> The endogeneity issue does not undermine the testability of Black's theory—i.e., its capacity to predict facts. Moreover, endogeneity is primarily a concern for "causal inference," and therefore only indirectly applicable to the covering-law approach employed by Black.

<sup>&</sup>lt;sup>390</sup> Due to the small number of cases involving Hispanic and Asian defendants/victims in Georgia, it is difficult to test research hypotheses concerning these groups. Recall from Chapter Five, that only three

Again, this does not suggest that race/ethnicity will uniformly disadvantage minority defendants or uniformly benefit white victims; however it does suggest that, in the aggregate, cases involving certain racial/ethnic combinations of defendants and victims will attract more or less law because of the social locations (i.e., status positions) these racial/ethnic groups tend to occupy and the social relations members of these groups tend to have with one another (see Section 6.2.4). Additionally, as some scholars have suggested, certain offenders and victims may be particularly advantaged (or disadvantaged), irrespective of their specific social status, because prosecutors, judges, and jurors may rely on stereotypes during decision-making based on the aggregate characteristics of racial groups (see Baumer et al. 2000; Kan and Phillips 2003; Stanko 1981/1982; cf. Bodenhausen 1990; Bodenhausen and Lichtenstein 1987; Bodenhausen et al. 1994). Moreover, because of the social structure of lethal violence (i.e., self-help in the form lethal violence is most often employed by those who lack access to the law), particular offender-victim racial combinations are more likely to reflect particular status configurations (see, especially, Black 1983; Cooney 1997a, 2003b), thereby further increasing the likelihood that they will be further disadvantaged in legal proceedings.

# 6.3.2 Hypothesis Two

Other extra-legal factors occurring at the case-level that are indicators of Black's five dimensions of social space will also influence the capital charging-and-sentencing decisions. Black (1989) notes that age, similar to race/ethnicity, serves as crude proxy for social status and, in the absence of more specific information about the structural

Hispanics and one Asian were on Georgia's death row. Only thirteen Hispanics and five Asian defendants were noticed for the death penalty in the time period under investigation for this study (see Chapter Seven). As noted in Chapter Four, Hispanic defendants may be at an even greater disadvantage than blacks in certain parts of the United States (i.e., the Southwest) with respect to criminal sentencing.

positions of disputants (and third-parties) in the case, can influence the administration of capital punishment. The relationship between chronological age and severity of law should be curvilinear: hold the age of the offender and the race/ethnicity of the parties constant, and cases involving very youthful victims or elderly victims will attract the most law (i.e., most likely to be noticed for the death penalty and result in unfavorable plea bargains and sentences at trial).<sup>391</sup> Due to the fact that all capital statutes place a nonnegotiable lower-limit on the age at which an alleged offender can be subject to the death penalty (see Section 2.3), much of the influence of the age of the offender has been eliminated because youthful offenders are most likely to receive leniency from judges and juries. Nevertheless, hold the age of the victim and the race/ethnicity of the parties constant, and cases with offenders with chronological ages that place them at (or very near) the minimum age for death-eligibility will attract the least amount of law. Also recall that Black's theory emphasized the relative status positions of disputants. As result, the interaction between offender and victim age should be most predictive: cases involving older offenders and youthful victims should attract the most law and cases involving younger offenders and older victims should attract the least amount of law, net of race/ethnicity and gender.

## 6.3.3 Hypothesis Three

Gender is also a crude proxy for social status (see Section 6.2.4). Unlike racial/ethnic status and chronological age, however, the association between gender and

<sup>&</sup>lt;sup>391</sup> Cases involving youthful victims and elderly victims (i.e., senior citizens) are likely to be handled in a similar fashion because in the aggregate, these victims—relative to their offenders—have more social status (cf. Sorenson and Berk 1998). Generally speaking, elderly victims tend to be more conventional (cultural status), respectable (normative status), integrated (radial status), and have had more time to accumulate wealth (vertical status).

the various social dimensions identified by Black are not nearly as robust—particularly in the last half-century. Nevertheless, because of the social structure of lethal violence, it is likely that female defendants and female victims in capital cases will, generally, occupy higher status locations, and as a result they enjoy greater advantages in legal proceedings. Black's theory emphasizes the relative status positions of parties involved in the conflict, so *cases involving male offenders and female victims will attract the most law and cases involving female offenders and male victims will attract the least law, net of age and race/ethnicity*.

## 6.3.4 Hypothesis Four

As noted above, the predictive power of race/ethnicity, age, and gender is limited in that they are mere proxies for social status. Black's theory predicts that other legally illegitimate (or legally suspect) factors, which are better indicators of social status, will have a stronger influence on legal decision-making.<sup>392</sup> Socioeconomic status and employment status are indicators of vertical status (i.e., wealth); marital status, number of children, and employment status are indicators of horizontal status (i.e., integration, intimacy, interdependence); level of education is an indicator of cultural status; and prior drug/alcohol use is an indicator of normative status (i.e., respectability) (see Section 6.2.2).<sup>393</sup>

<sup>&</sup>lt;sup>392</sup> Black's five dimensions of social space are extremely broad categories, and as such, several of the variables mentioned can be indicators of multiple dimensions.

<sup>&</sup>lt;sup>393</sup> Recall that organizational status, refers to the capacity for collective action and occurs when individuals or groups are drawn together to engage in a project (Black and Mileski 1973, p. 9). Measures of organization include the presence and number of administrative officers, the centralization and continuity of decision-making, and the frequency of collective action (Black 1976, p. 85). As a result, corporate status is more applicable when at least one of the disputants is a group. Nonetheless, in capital cases, the victim enjoys greater corporate status because the government is bringing the action against the defendant, and of course, the government is perhaps the most powerful collective actor (but cf. Zimring 2003). The

[T]he handling of cases always reflects the social characteristics of those involved in it. It applies whenever and wherever law is found. It is not merely a matter of differentials according to the race of the parties, their social class, gender, or other characteristics that nowadays attract attention. Many other kinds of discrimination exists as well, such as differentiation according to degree of intimacy between parties, the cultural distance between them, and their degree of organization, interdependence, integration, and respectability. No one has ever observed a legal system without social differentiation. Discrimination is ubiquitous. It is an aspect of the natural behavior of law (Black 1989, p. 21).

The aforementioned factors should exert a direct influence legal decision-making in the directions hypothesized by Black, net of the influence of race/ethnicity, age, and gender: *holding the social status of the victim constant, cases with defendants low in vertical, horizontal, cultural, and normative status will attract the most law.* 

## 6.3.5 Hypothesis Five

Black's theory of law suggests that all social phenomena may be described by their social location—their position, whether higher or lower, in a distribution. For example, hold the difference in wealth between disputants constant, and their location in the wealth distribution explains the quantity and style of law (Black 1976, p. 17). As a result, "similarly situated" cases may be handled differently depending on the jurisdiction where they are tried. Moreover, the variability in the handling of similarly situated defendants across jurisdictions can be predicted and explained by the structural characteristics of the jurisdictions in which they are tried. Black notes, for example, that

government's role, however, is that of a third-party supporter, not a principle (although the defendant has "technically" committed a crime against the State) (Black 1983).

Cases involving multiple offenders who work together, however, should attract the least amount of law. Conversely, cases in which multiple offenders turn against one another (i.e., incapacity for collective action) should result in the greatest quantity of law for at least one of the defendants because the government now has an additional cooperating actor. Multiple-offender cases, however, typically involve "heightened" homicides (e.g., homicides involving additional felonies), and as such, contain legally legitimate characteristics (i.e., statutory aggravating factors) that attract more law.

groups have wealth (i.e., rank) and *law varies directly with the rank of groups, not only among themselves but also in relation to individuals.* "It is even possible to rank entire societies among themselves, and also the areas of a society, its regions, communities, and neighborhoods. This may be done either according to the distribution of the wealth among the residents or according to the wealth of the society or area itself. In the first case, law varies with the proportion of the population that is more or less wealthy" (p. 20). Thus, the legal jurisdiction in which death penalty cases occur should influence the administration of capital punishment, net of extra-legal case-level characteristics.

Recall that Black's theory suggests the severity of law (i.e., its quantity) increases in conflicts among the wealthy, the economically stratified, the educated, the conventional, the organized, and the respectable, all else constant. Among the integrated and the intimate, the relationship is curvilinear: increasing to a certain threshold and then decreasing thereafter (p. 49). Also recall that law decreases among the empowered (i.e., those with the capacity for non-legal social control), all else constant (p. 112). Thus a jurisdiction's per capita income, poverty rate, and median home value (vertical status); labor force participation, residential mobility, and marriage rate (horizontal status); high school/college education rate and level of racial/ethnic integration (symbolic/cultural status); and crime rate (normative status) should influence the capital charging-andsentencing process—net of extra-legal individual case characteristics—in the direction specified by Black's theory (see Section 6.2.2).<sup>394</sup> *More specifically, holding case* 

<sup>&</sup>lt;sup>394</sup> Crime can be both a type of social control (i.e., an expression of a grievance) and an indication of a lack of social control, depending on the purpose of crime and the level of analysis (Black 1983). When individuals use crime as a conflict management strategy (i.e., self-help), such behavior can often be viewed as a type of social control (Black 1990). Moralistic crimes, however, are but one type of criminal activity and probably only constitute a small fraction of all criminal behavior (Cooney and Phillips 2002). A jurisdiction's crime rate, however, is most likely an indicator of the jurisdiction's lack of authority (i.e., capacity for social control). As Shaw and McKay (1942) and others (Fagan and Meares 2000; Kelling and

characteristics constant (i.e., case-level extra-legal variables), jurisdictions high in vertical and symbolic status will attract more law. Conversely, cases will attract less law in jurisdictions high in horizontal and normative status, net of case-level extra-legal variables. This hypothesis underscores Black's assertion that the *complete* social geometry of a case predicts and explains the quantity of law it attracts because "[m]icrocosms overpersonalize everything, and macrocosms oversocietalize everything" (Black 1995, p. 858).

It should be emphasized that a jurisdiction's status location is a *composite* of its aggregate characteristics with respect to a specific dimension of social space because no single proxy for social status will fully capture the theoretical construct. Additionally, prior research on the influence of structural factors on crime reveals it is necessary to "simplify the structural covariate space" to indices in order to reduce collinearity between structural covariates when examining their effects (Land, McCall, and Cohen 1990, p. 922). For example, the bivariate correlations for the circuit-level variables for a particular dimension of social space were extremely high (typically ranging from .8 to .9). Consequently, the aforementioned structural factors that correspond to particular dimensions of social space are combined into indices so the aggregate impact of these factors on legal decision-making are explored in the subsequent analyses (for a detailed description of these indices, see Section 8.1.2 and Table 4).

Coles 1996; Sampson 1987; Wilson and Kelling 1982) have persuasively argued, a persistently high crime rate in a community often results from the breakdown in familial and community social control (cf. Agnew 1999).

## 6.3.6 Hypothesis Six

Black (1976) posits that the social geometry of a case predicts and explains legal behavior when the legal characteristics of a case are held constant (see also Horwitz 1983), therefore an accurate test of the theory must hold legally legitimate characteristics constant. Black (1984) mentions that legal rules are important under two conditions: when cases are socially similar or when the amount of potential law in the case is trivial.<sup>395</sup> "Every case has a technical core—the rules in the face of the evidence—that can be meaningfully analyzed in the jurisprudential tradition. All else being equal, including the social characteristics of all concerned—the technical core is important to the handling of case" (p. 20). He notes, however, that legal doctrines and the facts are often ambiguous, with uncertain implications (cf. Breyer 2005; Scalia 1997); moreover, legal officials often disregard the written law (see, e.g., Black 1980, pp. 180–86; Weinburg 2003).<sup>396</sup> In fact, empirical studies conducted in numerous jurisdictions have consistently discovered that legally legitimate characteristics only account for a modest proportion of the variance in capital charging-and-sentencing decisions (Nakell and Hardy 1987; Paternoster et al. 2004; Weiss, Berk, and Lee 1996). Berk and colleagues (1993, 2005) even suggest that pure capriciousness in the death penalty process may be more important than the role of illegitimate factors such as race/ethnicity.<sup>397</sup> Black

<sup>&</sup>lt;sup>395</sup> Cases will also be handled similarly if the social locations and relationships of the disputants (and relevant third parties) are unknown (Black 1989, p. 64). The greater amount of potential law that may be applied to a case, however, the more social information about the case gets collected or becomes involved (Black 1984, p. 20).

<sup>&</sup>lt;sup>396</sup> In the opening paragraph of the seminal monograph, *An Introduction to Legal Reasoning*, Edward Levi (1949, p. 1) explains, "It is important that the mechanism of legal reasoning should not be concealed by its pretense. The pretense is that the law is a system of known rules applied by a judge; the pretense has long been under attack [citing (Frank 1930)]. In an important sense legal rules are never clear, and, if a rule had to be clear before it could be imposed, society would be impossible."

<sup>&</sup>lt;sup>397</sup> "It is difficult to imagine that a few covariates exist that if included as predictors would lead to clear and justified distinctions between defendants who are charged with a capital crime and defendants who are not;

(1989) suggests, however, that the handling of these cases is only capricious from a jurisprudential perspective, and can actually be systematically predicted and explained by the social geometry of the case.

Of course, many characteristics comprising the "technical core" of the case are also indicators of the social structure of the case (e.g., relative wealth, intimacy, integration, interdependence, respectability).<sup>398</sup> These factors provide more social information about the disputants involved in the case, and therefore attenuate the impact of crude proxies of social status, such as race/ethnicity and gender (Black 1989. pp. 61– 62, 108 n.48, 119 n.12). Nonetheless, the legal and quasi-legal characteristics of a case do not fully capture its social organization (see Section 6.2.5), *so the extra-legal characteristics of a case should still exert an impact on legal decision-making in the capital charging-and-sentencing process after controlling for these legally relevant factors*. In particular, *the impact of race/ethnicity, age, and gender, as predicted in Hypotheses One, Two, and Three, should remain after taking into account the technical core of the case.* 

### 6.3.7 Hypothesis Seven

Building on Hypothesis Five, Black's theory suggests that *the legal jurisdiction in which death penalty cases occur should influence the administration of capital punishment, net of legal and extra-legal characteristics*. That is, even after taking into

likewise for death sentences...if idiosyncrasies associated with the case, the defendant, or the adjudication process seem to determine a substantial part of the outcome, the adjudication process is suspect whether race is important or not" (Berk *et al.* 2005, p. 31).

<sup>&</sup>lt;sup>398</sup> As discussed earlier in this chapter, statutory aggravating circumstances permit increases in the amount of law (i.e., quantity) that can be applied to a case and, therefore, can be predicted and explained by the social geometry of the case (Horwitz 1983). Many of the statutory aggravating circumstances and mitigating circumstances are, themselves, indicators of the social status of the offender and victim, and do not specifically address the actual conduct of the offender or the amount of harm inflicted by the offender on the victim.

account the technical core of the case, the structural characteristics indicating a *jurisdiction's* vertical, horizontal, symbolic, and normative location should continue to influence legal decision-making in the direction specified by Black's theory. The legal characteristics of a case, *by design*, do not take into account the structural location of the conflict, and therefore they fail to fully reflect important elements of the social geometry of the case. So, as discussed previously (see Section 6.3.5), *cases occurring in jurisdictions high in vertical and symbolic status, and low in horizontal and normative status, will attract more law, net of case-level legal and extra-legal variables.* 

Recall that a jurisdiction's vertical status is indexed by per capita income, poverty rate, and median home value; its horizontal status is comprised of labor force participation, residential mobility, and marriage rate; its symbolic/cultural status is proxied by high school/college education rates and level of racial/ethnic integration; and its normative status is a function of its crime rate.

# 6.3.8 Hypothesis Eight

Black also predicts that the impact of case-level factors varies across social contexts.<sup>399</sup> According to the theory, the social location of a conflict influences how large legal advantages (or disadvantages) resulting from status differentials between parties in a case will be (Black 1976, pp. 49, 112). Structural characteristics (based on the five dimensions of social space), then, may either accentuate or attenuate the impact of status differences between disputants. In other words, *the impact of legal and extra-legal case-level factors on the administration of capital punishment should vary across* 

<sup>&</sup>lt;sup>399</sup> The variable impact of micro-level factors across macro units is often referred to as *causal heterogeneity* (see, e.g., Johnson 2006).

legal jurisdictions and this variability can be explained by the structural characteristics of these jurisdictions.

Specifically, a case's vertical, horizontal, symbolic, and normative location partially captured by jurisdictional variables mentioned above—will condition/moderate the impact of legal and extra-legal case-level factors on the administration of capital punishment. So, for example, *the impact of status differentials on the severity of law a cases attracts, as captured by differences in race/ethnicity, age, and gender should vary according to the jurisdiction's vertical, horizontal, normative, and cultural status, such that the effect of these status differences will be accentuated in jurisdictions that are high in vertical and cultural status, but attenuated in jurisdictions that high in horizontal and normative status.*<sup>400</sup>

Similarly, the *influence of legal variables should vary according to the jurisdiction's structural characteristics*. As noted earlier (see Section 6.2.5), most laws reflect the social geometry of case and, therefore, both their *content* and *impact* can be explained by Black's theory. Very few of the special circumstances outlined in capital statutes that allow a case to be eligible for the death penalty address the actual conduct of the offender or the harm inflicted by the offender on the victim; rather they reflect the particular social positions of the offender or the victim.<sup>401</sup> Those laws that specifically address the actual conduct of the offender (e.g., commission of a contemporaneous felony and heinous of a crime) are also explained by Black's theory—these factors are indicators of cultural or normative status. Since all crimes, including murder, can be conducted in a

<sup>&</sup>lt;sup>400</sup> The impact of the other indicators of the social positions of defendant and victims (e.g., socioeconomic status, marital status, employment status, *et cetera*) should also vary according to a jurisdiction's vertical, horizontal, normative, and cultural status.

<sup>&</sup>lt;sup>401</sup> Most mitigating factors also reflect the status positions of the offender or victim and not the actual level of harm resulting from the crime (see *supra*, 6.2.5).

more or less conventional manner (Cooney 1997), the more heinous or vile the murder, the less respectable and more unconventional the offender will appear to be, even after holding prior criminal history constant (see Section 6.2.5). It follows from this discussion that *the impact of the number of statutory aggravating circumstances, having a contemporaneous felony, a prior felony, and a prior violent felony on a case will vary according the jurisdiction's vertical, horizontal, normative, and cultural status, such that the effect of these case-level factors will be accentuated in jurisdictions that are high in vertical and cultural status, but attenuated in jurisdictions that high in horizontal and normative status.* 

It is important to reiterate that Black's theory does not specifically discuss the interrelations among the five social dimensions (see Lessan and Sheley 1992), and critics claim that the complexity of the theory requires Black to specify these interrelations (Bernard 1995). Black does discuss, in very general terms, micro-macro interrelationships *within* a particular social dimension (Black 1976, p. 49), but fails to articulate cross-dimension interrelationships. Some scholars have note, however, that the specification of these specific relationships is unnecessary at the level of abstraction at which Black theorizes, and that derivations of Black's general propositions is an appropriate task for middle-range-theorizing for more specific research programs (see, e.g., Turner 2002). The cross-level/cross-dimensional interactions articulated in *Hypothesis Eight*, then, are an extension of Black's theory.

## 6.3.9 Summary

These eight hypotheses comprise the core of Black's theory of law with respect to the administration of capital punishment and are examined using data collected on Georgia's capital charging-and-sentencing process from 1993 through 2000. As noted earlier, the capital punishment process provides an ideal test for Black's theory of law because the death penalty is the greatest quantity of law (i.e., most severe punishment) that can be imposed by the government and invites the most social information about the offender(s) and victim(s) into the legal proceedings (Black 1989). The specific data sources consulted and analytical models employed are discussed in detail in the next chapter.

### 6.4 A COMMENT ON VERIFICATION, FALSIFICATION, AND CORROBORATION

As noted above, formal legal, conflict, and interactionist theorists make similar predictions to those derived from Black's theory of law—e.g., the greatest quantity of law in black-offender/white victim cases and the least quantity of law in white offender/black-victim cases (Hypothesis One) or economic inequality influences the seriousness of punishment, net of individual case characteristics (Hypotheses Five and Seven), but for very different reasons. As previously discussed, all three of these perspectives have received some empirical support (see Section 6.1.3), but only Black's theory is able to predict these same facts and relationships under a single explanatory framework. It must be emphasized, however, that the scientific method cannot *verify* hypotheses or establish the one "true" theory. The scientific method can, however, assess what Popper ([1934] 1968, p. 419) has referred to as the *degree of corroboration*: the extent to which theories have been *least* falsified, but that are *most* falsifiable and most tested relative to competing theories (see also Harris 1979, p. 18). Moreover, only when theories are highly general, testable, conjectural, and concise, can they "maximally expose themselves to falsification" and substantially contribute to the scientific

community's understanding of the phenomena in question (p. 17). Popper's "degree of corroboration," therefore, is akin to Black's "degree of scienticity" (see Black 2000a, p. 351). Not only do the three dominant perspectives possess, comparatively, a substantially lower degree of scienticity than Black's theory of law, researchers have also produced a fairly large amount of disconfirming evidence for the core hypotheses of each perspective (Kempf-Leonard and Sample 2001; Weitzer 1996).<sup>402</sup>

Lakatos (1970, p. 119) argues that falsification, alone, cannot be used to decide whether a theory is scientific. A theoretical approach that is believed to be superior to rival theories must explain existing facts better, as well as explain new facts. According to Laudan (1977, p. 120), "All evaluations of research traditions and theories must be made within a comparative context. What matters is not, in some absolute sense, how effective or progressive a tradition or theory is, but, rather, how its effectiveness or progressiveness compares with its competitors." Black's conscious commitment to discovering orderly relationships at a very high level of abstraction—what Maxwell (1974a, p. 152) refers to as "aim-oriented empiricism"—allows this theory of law to be maximally effective and progressive, and therefore superior to rival research programs and theories lacking generality, concision, and coherence (see Harris 1979, p. 25).<sup>403</sup>

<sup>&</sup>lt;sup>402</sup> Attempts to integrate these theories aimed at reconciling many of these conflicting findings (i.e., eclecticism) have also failed to result in a single body of theory that is at once general, parsimonious, logically consistent, and falsifiable (e.g., Albonetti 1991; Dixon 1995; Hawkins 1987).

<sup>&</sup>lt;sup>403</sup> Maxwell (2005, p. 181) posits that aim-oriented empiricism is "a kind of synthesis of the views of Popper, Kuhn, and Lakatos, but is also an improvement over the views of all three." The basic idea behind aim-oriented empiricism is that science should be viewed as making not one, but a hierarchy of assumptions concerning the unity, comprehensibility, and knowability of the universe—with the assumptions becoming less and less substantial as one ascends the hierarchy, and thus more likely to be true. This hierarchy permits the disentanglement of what is most likely to be true, and not in need of revision, at the top of the hierarchy, from what is most likely to be false, and most in need of criticism and revision near the bottom on the hierarchy. The aim-oriented empiricism framework makes explicit metaphysical assumptions implicit in the manner in which physical theories are accepted and rejected, facilitating the critical assessment and improvement of these assumptions with the improvement of

knowledge by concentrating criticism where it is most needed: low down in the hierarchy (see also Braithwaite 1953).

## **Chapter 7: Data and Analytical Approach**

### 7.1 DATA COLLECTION

### 7.1.1 Years Considered

The current analysis of Georgia's capital charging-and-sentencing process focuses on homicide cases with incident dates between January 1, 1993 and December 31, 2000. This particular time frame was chosen for three substantive reasons. First, as discussed in Chapter Five, Georgia's life without the possibility of parole (LWOP) legislation was enacted in 1993. The legislation was specifically designed as a sentencing alternative in capital murder trials, therefore potentially having a tremendous impact on prosecutorial, judicial, and jury discretion.<sup>404</sup> Second, the Georgia Generally Assembly, along with the Georgia Supreme Court, established the Office of the Multi-County Public Defender (MPD) in October 1992 as a statewide agency to serve as a death penalty defense trial resource center and to actively monitor all death penalty cases in Georgia's 159 counties (Mears 1999).<sup>405</sup> MPD began collecting detailed information on cases that were noticed for the death penalty shortly thereafter, and therefore has a fairly complete list of cases in which the prosecutor initially filed a notice of intent to seek the death penalty since 1993. Third, cases with incidence dates through the end of 2000 are analyzed because this allows for the examination of eight years of cases with sufficient time to advance from the charging phase through the (initial) penalty phase.<sup>406</sup> Of the 381 cases noticed for the

<sup>404 1993</sup> Ga. Laws 569, § 4; Ga. Code Ann. § 17-10-30.1 (1993).

<sup>&</sup>lt;sup>405</sup> In 1995, Congress substantially cut federal funding for attorneys representing indigent inmates on death row, resulting in the closing of "Resource Centers" in twenty states that provided legal services to these inmates (Wiehl 1995).

<sup>&</sup>lt;sup>406</sup> Georgia law also requires all defendants convicted of murder receive, at minimum, a life sentence with the possibility of parole and serve *at least* fourteen years of their sentence before being eligible for parole.

death penalty between 1993 and 2000 in Georgia, 376 (98.6 percent) have progressed through the penalty phase.

### 7.1.2 Decision Stages Examined

The capital charging-and-sentencing process consists of several decision stages and several decision-makers, with each subsequent decision stage being dependent on the previous stage (for a description of the *entire* capital punishment process, see Appendix A). Focusing solely on later decision stages, such as the conviction and penalty decisions, ignores both the exercise of discretion at the earlier stages and the fact these cases represent a select group of cases that may not typically be representative of the universe of cases from which they are drawn (Gross and Mauro 1989, pp. 24–26). In many situations, this "differential selection" process can be explicitly incorporated into the analysis of later decision stages and valid inferences concerning these later decision stages can be made (Sorensen and Wallace 1999). It is important, however, to begin with the universe of all cases that *could* be prosecuted capitally (Baldus *et al.* 1990; Paternoster *et al.* 2004). The more comprehensive the universe of cases, the better measures of prosecutorial, judicial, and jury decision-making can be obtained. As noted in Chapter Four, previous research suggests that the influence of extra-legal factors is typically strongest in the earlier stages of the capital charging-and-sentencing process, therefore it is important to collect information about cases that were eligible for capital punishment, but in which the prosecutor chose not to seek the death penalty.

Recall from Chapter Two that Baldus and colleagues' (1990, p. 40) seminal study of Georgia's death penalty examined five decision points in capital punishment process: (1) indictment for murder, voluntary manslaughter, or a lesser crime; (2) plea bargaining and guilt plea; (3) guilt-trial decision for defendants not pleading guilty; (4) prosecutorial decision to advance to a penalty trial in cases that result in a capital murder conviction; and (5) jury death-sentencing decision for cases that advance to a penalty trial. Similarly, Paternoster and colleague's (2004) recent study of Maryland's capital punishment examined four key decision points (i.e., death notice, withdrawal of the death notice, advancement to the penalty phase, and disposition) and Sorensen and Wallace's (1995) analysis of Missouri's death penalty system examined three stages (i.e., death notice, conviction, and disposition). The current study examines three decision stages: (1) initial notice of the death penalty; (2) guilty plea and plea agreement; and (3) disposition at the penalty phase.

#### 7.1.3 Data Sources

It is important to begin with a comprehensive list of potentially capital cases from which discretion at the various stages of Georgia's capital charging-and-sentencing process can be properly identified and assessed. The research literature on capital punishment reveals two different approaches to collecting information on potentially capital cases from which prosecutors choose to seek the death penalty, each having its unique advantages and disadvantages. The first approach is to begin with all cases that resulted in a conviction for the crime of murder or voluntary manslaughter. Several scholars have advocated this approach because it provides a rudimentary control for the strength of evidence in the case (see, e.g., Baldus *et al.* 1990; Paternoster *et al.* 2004).<sup>407</sup>

<sup>&</sup>lt;sup>407</sup> For their Philadelphia study, Baldus and colleagues (1998) created quantifiable measures of strength of evidence for each statutory aggravating and mitigating circumstance. Baldus *et al.* classified the evidence supporting the aggravating factors on a four-point scale and the evidence supporting the mitigating circumstances on a three-point scale (cf. Nakell and Hardy 1987, Appendix A). While these measures are appropriate for conviction and sentencing decisions because such evidence has been presented before the

Indeed, it is possible that a case may appear to be death eligible, but the prosecutor believes that her or his case is very circumstantial or that key witnesses may be unreliable. The major disadvantage of this strategy, however, is that some cases are noticed for the death sentence but result in conviction for a crime other than murder or an acquittal. By exclusively focusing on cases that result in a murder conviction, researchers may fail to consider all of the cases that prosecutors originally consider for the death penalty. In fact, it is not uncommon for prosecutors to reduce the murder charge to a lesser offense in exchange for a capital defendant's testimony against a coconspirator(s). In Georgia, 25 death penalty cases with incident dates between 1993 and 2000 resulted in either an acquittal or a conviction for a crime other than murder.<sup>408</sup> Recent evidence also suggests that prosecutors are more likely to seek the death penalty in cases with weaker evidence when the defendant is a member of a minority group (Harmon 2001b; Parker et al. 2001). Eighty percent of the capital defendants in Georgia who were acquitted or convicted for a lesser offense were members of a racial/ethnic minority group: 16 black defendants (64 percent), five white defendants (20 percent), three Hispanic defendants (12 percent), and one Asian defendant (4 percent).<sup>409</sup>

*Supplementary Homicide Reports*. An alternative to focusing on cases resulting in a murder conviction is to simply begin with all homicide cases known to the police

court, it is not well suited for charging decisions. In Georgia, for example, the prosecuting attorney must announce her or his intention to seek the death penalty at a pre-trial conference shortly after a grand jury indictment, but prior to arraignment; therefore, all that is required for the prosecutor to seek the death penalty is an indictment for a capital felony (see Georgia Unified Appeal Procedure Rule II (C) (1)). The actual determination of whether the facts in the case warrant the case being "death-eligible" is made by the judge or jury if the case progresses to the penalty phase (see Chapter Five).

<sup>&</sup>lt;sup>408</sup> Of these 25 cases, 11 resulted in an acquittal (or dismissal) and the remaining 14 resulted in convictions other than murder.

<sup>&</sup>lt;sup>409</sup> Fifteen of the 25 cases were white-victim cases (60 percent), six cases were black-victim cases (24 percent), two were Hispanic-victim cases (8 percent), and two cases were missing information on the race of the victim (8 percent). Nine of the 16 black defendant cases (56 percent) involved white victims.

(see Brock et al. 2000; Gross and Mauro 1984; Holcomb et al. 2004; Sorensen and Wallace 1995; Thomson 1997). The Federal Bureau of Investigation (FBI) collects data on all homicides occurring in state, disaggregated by year, month, and county of occurrence, and compiles these data in its annual release of the Supplementary Homicide *Report* (SHR). The SHR contains data on the age, race/ethnicity, and gender of the victim(s) and offender(s), the circumstance under which the homicide took place (e.g., robbery, burglary, et cetera), the relationship of the victim to the offender (e.g., family, stranger, et cetera), and the weapon used in the homicide (Fox 2005). Clearly, the major strength of SHR data is that they document every homicide known to the police.<sup>410</sup> These data, however, have two major shortcomings: (1) a high proportion of missing data and (2) limited information on each homicide (Songer and Unah 2006, pp. 185–86). The SHR contain data on every homicide *known* to the police, not every homicide *arrest*, and as result, these data often lack information on homicide offenders and the relationship between the victim(s) and the offender(s). In Georgia, for example, as much as 30 percent of the homicides from 1993 to 2000 have missing information one at least one variable (Fox 2005). In response to the high rate of missing data in the SHR, analysts have recently developed ways to estimate missing information (see Fox 2004; Messner, Deane, and Beaulieu 2002; Regoeczi and Riedel 2003). The second problem-limited information on each case—is a much more significant obstacle to death penalty researchers because it is often difficult to determine which cases are, in fact, eligible for the death penalty. Recall from Chapter Five that Georgia's post-*Furman* death penalty statute lists ten circumstances that, if any one is present in a case, make the case eligible

<sup>&</sup>lt;sup>410</sup> Law enforcement agencies *voluntarily* report SHR data to the FBI on a monthly basis. Agencies may fail to report each month or at all, resulting in missing data on entire homicide incidents (Fox 2004).

for the death penalty. Similar to other death penalty states, Georgia's statute contains a "catch-all" circumstance that authorizes the death penalty for cases in which "the offense [of murder] was outrageously or wantonly vile, horrible or inhumane in that it involved torture, depravity of mind, or an aggravated battery to the victim."<sup>411</sup> Cleary homicides that involved multiple victims or that occurred along with a contemporaneous capital felony (e.g., robbery, kidnapping, *et cetera*) are death eligible under Georgia's statute and can be identified in the SHR data.<sup>412</sup> The other circumstances, however, are much more difficult to identify with the data available in the SHR. Although it is possible to argue that the catch-all circumstance ostensibly makes every homicide death eligible, it is very unlikely that prosecutors proceed as if every homicide were, in fact, eligible for capital punishment (see Paternoster *et al.* 2004).<sup>413</sup>

This study adopts the first approach, collecting information on every case resulting in a murder or manslaughter conviction. As Baldus *et al.* (1998, p. 1661) explain, a "well-controlled" study is one that has statistical controls for *at least* ten nonracial characteristics. It would be impossible to include the necessary non-racial control variables relying solely on the SHR data. Since prosecutors often rely on less information than is available to researchers who select cases that ultimately resulted in a

<sup>&</sup>lt;sup>411</sup> Ga. Code Ann. § 17-10-30(b)(7).

<sup>&</sup>lt;sup>412</sup> Pierce and Radelet (2005) employ this approach in their recent examination of California's capital punishment process (see also Gross and Mauro 1989). In their analyses, cases were categorized as having *one* aggravating circumstance if multiple victims OR a contemporaneous felony was present in the case; as having *two* aggravating circumstances if multiple victims AND a contemporaneous felony was present in the case; and as having *zero* aggravating circumstances if neither multiple victims NOR a contemporaneous felony was present in the case. According to the authors, these two types of aggravating circumstances are among the most common set of aggravating circumstances used by prosecutors, jurors, and judges to justify death sentences. Indeed, their examination the death sentencing in Illinois (see Pierce and Radelet 2002) revealed that the number of homicide victims was one of the strongest predictors of a death sentence, even after controlling for several other legally relevant and irrelevant case characteristics (see also Shatz and Rivkind 1997).

<sup>&</sup>lt;sup>413</sup> Baldus and colleagues (1986b, p. 152) suggest that approximately 10 percent to 25 percent of murders and non-negligent manslaughters annually reported qualify as death eligible homicides.

murder conviction (Baldus *et al.* 1990, Chapter 11), expanding the pool of cases to those that ultimately fall short of a murder conviction, but that were still likely to be deemed death-eligible prior to arraignment, allows the current study to achieve the requisite depth and breath.<sup>414</sup>

*Georgia Department of Corrections Records*. Previous research strongly suggests that such factors as the alleged offender's: prior criminal history, contemporaneous offenses (i.e., both felonies and misdemeanors), and prior incarcerations in the state have a strong impact on prosecutorial, judicial, and jury discretion in the capital charging-and-sentencing process (Paternoster *et al.* 2004; Unah and Boger 2001; Weisburd and Naus 2001). Many of these studies also discovered that several of these factors tend to be correlated with certain extra-legal factors (e.g., race/ethnicity, gender, age) (Baldus and Woodworth 2003). The importance of legal factors may be erroneously attributed to extra-legal variables if these legal factors are not explicitly taken into account in charging-and-sentencing research (Berk *et al.* 2005). Indeed, prior research suggests that the impact of race/ethnicity on the capital charging-and-sentencing process is somewhat attenuated when these legitimate legal characteristics are considered (see, generally, U.S. General Accounting Office 1990).

Not only does the Georgia Department of Corrections (GADOC) gather information on the legally relevant factors such as prior criminal history and offense characteristics, but it also gathers information on an inmate's educational history,

<sup>&</sup>lt;sup>414</sup> It is important to note that the unit of analysis for SHR data is the homicide *incident*, not the offender. For the purposes of our inquiry, the SHR data had to be disaggregated for multiple-offender homicides so the homicide case was the unit of analysis (at the micro level). Recall that this study analyzes the discretion of legal decision-makers as it pertains to *individual* homicide cases—prosecutors seek the death penalty, and judges and juries impose the death sentence, on individual defendants. In fact, it is likely that defendants in a multiple-offender homicide are treated differently from one another depending on their involvement in the crime, their helpfulness and usefulness to the prosecution, and their unique social characteristics (Paternoster and Brame 2003).

employment status at the time of the offense, alcohol/substance abuse, and family background. The GADOC contains information on every individual incarcerated in Georgia dating back to the turn of the twentieth century.<sup>415</sup> The major shortcoming of the GADOC data is that they do not contain victim information for homicide cases. Fortunately, the SHR data do contain such information, so SHR data and GADOC data were combined to benefit from the strengths of both data sources.

The SHR data do *not* list the names of alleged offenders, so matching is not as straightforward as one might wish, but the data contain information on the race, gender, and age of the alleged offender, as well as the date and location of the homicide.<sup>416</sup> The GADOC data contain information about the race/ethnicity, gender, and age of the (current or former) inmate and the date and location of the offense, making matching between SHR and GADOC data possible (cf. Songer and Unah 2006, p. 186).

*Records from the Clerk's Office of the State Supreme Court.* To accurately examine prosecutorial discretion in seeking the death penalty, a complete list of all cases noticed for the death penalty must also be obtained.<sup>417</sup> According to Georgia law, all notices of intent to seek the death penalty must be filed with the Office of the Clerk of the

<sup>&</sup>lt;sup>415</sup> Records dating back to 1909 were located in the GADOC records. The GADOC also has information on individuals sentenced to probation dating back to 1973. A GADOC official indicated, however, that the data are continuously being updated and that records for extremely old cases are less reliable. Because this study only focuses on cases with incident dates from 1993 through 2000, the GADOC records are reliable. <sup>416</sup> The SHR data contain information on the month and year of the offense, and within each month,

homicides are numbered chronologically in ascending order. Also, SHR data do not list the actual county where the offense occurred, rather they lists the name of the local agency that reported the homicide and a corresponding agency code. This agency code includes a Georgia county code that locates the agency in a particular Georgia county.

<sup>&</sup>lt;sup>41</sup> This list includes all cases noticed for the death penalty, irrespective of whether the notice was subsequently withdrawn. The rationale behind this approach is that primary interest is in the prosecutor's initial decision to seek the death penalty. There are a host of reasons why prosecutors may subsequently choose to withdraw the death penalty (e.g., local politics, strength of evidence, plea bargaining, *et cetera*). The decision to later withdraw the death notice, however, reflects a separate decision phase.

Georgia Supreme Court (henceforth, "Supreme Court's Clerk Office").<sup>418</sup> The Supreme Court's Clerk Office maintains a list of all the death notices filed, recording the name of the defendant, the date the notice was filed, the county in which the notice was filed, and the name of the prosecutor(s) filing the notice. For this study, the Supreme Court's Clerk Office was the primary source of information about death notices. This information was supplemented with other data sources. Routine searchers of news media sources (e.g., state and local newspapers and *LEXIS/NEXIS*) were made, as were periodic calls to the country clerks' offices and searches of GADOC records, in order to ensure that all cases noticed for the death penalty were included in the data.

The Supreme Court's Clerk Office also contained other important information on capital cases. Recall from Chapter Five that defendants sentenced to death are granted an automatic appeal before the Georgia Supreme Court. When these cases come before the Court for mandatory review, the entire record is sent to the Supreme Court's Clerk Office. Upon receipt, the record is transferred to microfilm and then either archived or returned to the county were it was initially tried. Included in the record is the report of the trial court judge. This report contains socio-demographic information about defendant and victim(s), defendant's criminal history, facts about the crime, information about the prosecutor and defense counsel, and facts about the trial. For example, the trial judge is required to indicate whether race appeared to be an issue at trial and which

<sup>&</sup>lt;sup>418</sup> Under Georgia Unified Appeal Procedure Rule II(C), a prosecutor must give notice of intent to seek death at the earliest possible opportunity after indictment but before arraignment. This is the first proceeding between the judge, attorneys, and the defendant. When a prosecutor gives notice of her or his intent to seek the death penalty, a written notice of such intention must be filed with the clerk of the Superior Court. Within ten days of receiving this notice, the clerk of the Superior Court must send a copy to the clerk of the Supreme Court. If the prosecutor may file another notice of intent to seek the death penalty against the same defendant (see also Unified Appeal Procedure Rule II(A). As of January 1, 2005, prosecutors are required to notify the Office of the Georgia Capital Defender (GCD) of their intent to seek the penalty for any person determined to be indigent (Ga. Code Ann. § 17-12-127).
aggravating circumstances were presented at trial and found by the jury. Data from the Supreme Court's Clerk Office was extremely helpful, although it was limited because the office only keeps information on cases resulting in a death sentence and only a small portion of death noticed cases result in a death sentence. Information that was missing from the record at the Supreme Court's Clerk Office was obtained from the appellate opinions of the Georgia Supreme Court.

*Georgia Supreme Court Appellate Opinions*. After the Georgia Supreme Court completes its mandatory review of a death penalty case, it produces an opinion that becomes a part of the record of the case (and published in the *South Eastern Reporter*). The court's opinions contain information about the trial judge, the county in which the case was tried, the name of the district attorney(s) who prosecuted the case, a brief summary of the facts surrounding the crime,<sup>419</sup> the date the death notice was filed, and a list of statutory aggravating circumstance sought by the prosecutor and found by the judge or jury.

Data from the SHR, GADOC, Supreme Court's Clerk Office, and Georgia Supreme Court opinions are extremely helpful, however, they have two shortcomings. First, they may be biased because they are all generated and compiled by official state agencies. It becomes extremely difficult to cross-validate these data because these different sources tend to build on one another. For example, the opinions of the Georgia Supreme Court are based primarily on the record obtain by the Supreme Court's Clerk Office which, in turn, is gathered for the county clerk offices. Second, there is some information that official agencies neglect to collect or are unable to collect. Therefore, it

<sup>&</sup>lt;sup>419</sup> These facts typically include the date of the crime, name(s) of the victims(s), and name(s) of codefendant(s), if applicable.

is important to cross-validate and supplement these data with information from those on the other side of the aisle: capital defense attorneys.

*Case Tracking via Capital Defenders.* Baldus and colleagues (1990) suggest that attorneys who defend clients facing the death penalty can be excellent sources of information (see also Zuckerman 1997). Statewide public defender agencies have a tremendous advantage in this regard because they personally handle the vast majority of capital and non-capital murder cases in their states.<sup>420</sup> Recall that MPD was established in 1992 as a trial resource center for capital defense attorneys in Georgia and has actively monitored all capital cases in Georgia since that time. The majority of capital defendants were represented by non-MPD attorneys, but MPD attorneys served as lead counsel/cocounsel or consultants in a significant number of death penalty cases with incident dates between 1993 and 2000 (approximately 30 percent of these cases). MPD collected data on these capital cases by mailing two forms to defense counsel whose client was noticed for the death penalty. The first form was mailed as soon as the office became aware of the case, requesting basic information about the case, such as: (1) the name of the prosecutor(s) trying the case, (2) the name of the judge presiding over the case, (3) the date the crime occurred, (4) the date the death penalty notice was filed, (5) the charges filed against the defendant, (6) the statutory aggravating circumstances sought by the prosecutor, (7) socio-demographic information about the defendant, (8) employment

<sup>&</sup>lt;sup>420</sup> Statewide agencies would not be able to provide counsel for "conflict of interest" cases and cases in which the alleged offender(s) opted for private counsel. Nevertheless, the overwhelming majority of capital defendants are either indigent or do not possess the resources to finance their defense, so state-appointed counsel usually handles capital murder cases. The same may not hold for conflict of interest cases. While the vast majority of homicide cases involve a single alleged offender (71 percent of homicides occurring in Georgia from 1993 to 2000 involved a lone defender), there still remain a significant number of cases that involve multiple defendants. Conflict of interest cases are readily available to them.

status of the defendant at the time of the incident, (9) names of and socio-demographic information about co-defendant(s) (if applicable) and victim(s), and (10) defendant's criminal history (if applicable). MPD mailed a follow-up form to capital defense attorneys after the office received news that the case had completed the penalty phase. This follow-up tracking form requested information from defense counsel concerning matters that would likely be known by the conclusion of the penalty stage. In particular, defense counsel was asked about: (1) the sentence her or his client received, (2) whether the case was disposed of by plea or trial, (3) the type of mitigation evidence presented at trial (if applicable), (4) educational history of the defendant, (5) the mental health status of the defendant (e.g., history of mental illness, mental retardation, *et cetera*), (6) whether the defendant had a history of alcohol or substance abuse, and (7) the status of the co-defendant's case (if applicable).<sup>421</sup>

By in large, defense attorneys were fairly cooperative in providing case information to MPD. When defense counsel did not fully complete the forms or return them in a timely manner, they were contacted via telephone and questionnaires were resent until the necessary information was obtained. In situations were defense attorneys continued to be unresponsive, it was often possible to get the necessary information from other sources (e.g., attorneys who were familiar with the case, attorneys working on the case during direct appeal, attorneys working on the case during state and federal habeas corpus, *et cetera*).

<sup>&</sup>lt;sup>421</sup> The Bureau of Justice Statistics (BJS) also collects annual data on all prisoners sentenced to death since 1973. These data include information on individuals whose sentences were later vacated or commuted. Basic socio-demographic information (e.g., age, gender, education) as well as criminal history is supplied (see U.S. Department of Justice 2004b).

U.S. Census and Official Crime Data. As noted in Chapter Six, scholars examining the criminal sentencing process have stressed the importance of simultaneously considering both case characteristics and structural contexts (see Hagan and Bumiller 1983; Ulmer 2000). Also recall that the "unit of analysis" in the Blackian paradigm is neither the microcosm nor the macrocosm, but rather the shape of social space—its social geometry. According to Black's theory of law (1979a) the social geometry of a case is defined and measured by the social characteristics of the actors involved, the relationships these actors have with one another, and the larger social context in which they interact. According to Lazarsfeld and Menzel (1969) and others (see Raudenbush and Sampson 1999), collective (i.e., contextual) properties may be analytical, structural, or global. *Analytical* properties are obtained by aggregating information from individual-level characteristics (e.g., racial/ethnic minority composition). Structural properties are based on the relational characteristics of members of the group (e.g., network density). Finally, *global* properties are characteristics of the collective itself that are not based on the properties of the individual members (e.g., type of indigent defense system). It is, therefore, very important to collect various types of detailed information on the characteristics of jurisdictions where these capital cases are tried.

Dozens of county-level variables, most of which are indicators of Black's five dimensions of social space—vertical, horizontal, symbolic, corporate, and normative were gathered from the U.S. Bureau of the Census and Georgia Bureau of Investigation (GBI). Economic and socio-demographic data for each of Georgia's 159 counties were collected from the U.S. Bureau of the Census. These publicly available county-level data were collected from both the 1990 and 2000 decentennial censuses and averaged across these two time periods for each county (see U.S. Bureau of the Census 1991, 1992, 2001a, b, 2002a, 2003a). These county-level variables include: (1) population size, (2) population density (i.e., population per square mile), (3) percent of population living in urban areas, (4) males per 100 females, (5) median age, (6) percent of population under 18 years of age, (7) percent population that is foreign born, (8) percent of population who only speak English, (9) racial/ethnic composition, (10) percent born in Georgia, (11) percent never married, (12) percent of high school and college graduates, (13) percent employed, (14) per capita income, (15) percent living below the poverty level, and (16) median home value.

The amount of crime, particularly violent crime, in a county is also likely to have a significant impact on the administration of criminal justice, especially capital punishment (Weidner *et al.* 2004). Also recall from Chapter Six that the amount of crime in a particular social setting is an indicator of normative space in the Blackian paradigm; therefore official crime statistics were collected for each county in Georgia from 1990 through 2000, and then averaged for each county. These data were gathered from the GBI and include information on reports to the police and arrests made by police for four types of serious violent offenses (also referred to "Index" or "Part I" offenses): murder and non-negligent manslaughter, forcible rape, robbery, and aggravated assault (see Georgia Bureau of Investigation 2001; U.S. Department of Justice 1991). The countylevel crime measures include: (1) violent crime report rate, (2) violent crime arrest rate, (3) homicide rate, and (4) proportion of total Georgia homicides occurring in the county.<sup>422</sup> Additional contextual variables were created from individual-level official homicide data (i.e., SHR data) by aggregating these characteristics to the county-level from (e.g., proportion of white victim cases, proportion of interracial homicides, *et cetera*). Collectively, these averaged data provide an accurate account of the economic, socio-demographic, and crime conditions in Georgia counties during the period of study (see Appendix B).<sup>423</sup>

Georgia's Administrative Office of the Courts (AOC) organized the state's 159 counties into 49 superior court judicial circuits (Judicial Council of Georgia 2003).<sup>424</sup> County-level data were aggregated to the judicial circuit level for two important reasons. First, in Georgia, there is one district attorney per judicial circuit. While large counties comprise a single judicial circuit, many smaller counties are grouped together to form a single judicial circuit. As a result, a single prosecutor may be responsible for charging and plea bargain decisions for several counties under her or his judicial circuit (see Table 2). Also, if a judicial circuit consists of multiple counties, trial judges rotate throughout these counties. Treating counties that share a single judicial circuit as if they were independent ignores the similarities they share in the administration of capital

<sup>&</sup>lt;sup>422</sup> Weidner and colleagues (2004) suggest using arrest data than report data because arrests are much better measure of the overall crime rate, given the high volume of drug and non-violence index offenses. Both report and arrest data for non-violent offenses, however, are questionable because drug offenses and other trivial crimes dominate these indexes. Violent crime and reports and arrests (including homicides) are used in these analyses because they are much better measures of the amount of serious crime in a jurisdiction and are most likely to have an impact on the administration of capital punishment.

<sup>&</sup>lt;sup>423</sup> Johnson (2006a, p. 273) notes that using interrelated levels of data provides the most comprehensive resource for examining the multilevel contexts of criminal sentencing.

<sup>&</sup>lt;sup>424</sup> Georgia's 49 judicial circuits are also organized into ten Judicial Administrative Districts: *East, Fulton, Macon, Middle, Northeast, Northwest, Southeast, Southwest, Stone Mountain,* and *West.* These districts were created by the Judicial Administration Act of 1976 to provide regional court administration to the superior courts of Georgia (Ga. Code Ann. § 15-5-2). The districts were created along Georgia Congressional District lines and each district is served by an administrative judge and district court administrator selected by the superior court judges and senior judge in each particular district. The primary function of the administrative judge is to assist chief judges in preparing, presenting, and managing local court budgets. These judges, however, do not exert any influence on the capital charging-and-sentencing process in the judicial circuits that comprise the judicial district.

punishment resulting from shared decision-makers (see also Unah and Boger 2002, p. 32). Second, death penalty cases are extremely rare events, so aggregating county-level data to the judicial circuit level allows one to observe more cases per contextual unit and better statistically estimate relationships occurring at both the case- and contextual-level without altering the dependence structure of the cases due to their clustering (see Raudenbush and Bryk 2002).

In sum, a thorough and accurate analysis of the capital charging-and-sentencing process requires data on prosecutorial discretion in seeking the death penalty and judicial and jury discretion in imposing the death penalty. Data from the (1) Supplementary Homicide Reports (SHR), (2) Georgia Department of Corrections (GADOC), (3) Office of the Clerk of the Georgia Supreme Court, (4) Georgia Supreme Court Appellate Opinions, (5) Office of the Multi-County Public Defender, (6) U.S. Bureau of the Census, and (7) Georgia Bureau of Investigation (GBI) were all important sources of information about the legal and extra-legal factors at both the case- and jurisdiction-level that potentially influence the capital charging-and-sentencing process in Georgia.<sup>425</sup>

<sup>&</sup>lt;sup>425</sup> Data from these seven different sources were stored in *Microsoft Access*®, a program in the *Microsoft Office*® software suite. The underlying structure of *Access*® is the spreadsheet package *Microsoft Excel*®, to which *Access*® essentially adds a graphical user interface (see Viescas 1999). Initially, three separate databases (i.e., spreadsheets) were created. The first database contained all information relevant to prosecutorial discretion in charging. The spreadsheet consisted of the modified SHR data and the GADOC data. The second database contained detailed information on all cases noticed for the death penalty with incident dates from 1993 through 2000. These data were collected from GADOC, records from the Clerk's Office, Georgia Supreme Court opinions, and death penalty case tracking questionnaires collected by MPD. The third database, the contextual database, was comprised of the economic, socio-demographic, and crime data collected from the U.S. Bureau of the Census and the GBI official crime statistics. Coding conventions are listed in Appendix B.

# 7.2 ANALYTICAL APPROACH

#### 7.2.1 Culpability Measures

In ideal situations, social researchers are able to isolate the impact of an explanatory variable on the outcome of interest by randomly assigning individuals into one or more "treatment" groups—a process that largely eliminates the influence of confounding variables because individuals, in principal, are similar across all other characteristics, save the explanatory variable of interest (Rosenbaum 2002).<sup>426</sup> Examining the relative influence of legal and extra-legal factors on the capital charging-and-sentencing process has been a difficult task for analysts because traditional experimental manipulations are impossible in field settings (Cook and Campbell 1979). Complicating matters, as the courts have repeatedly recognized, is the fact that it is extremely unlikely to find two capital cases that are exactly alike, save the particular variable of interest (Katz 1999, p. 231). As a result, death penalty analysts have focused their efforts on identifying the most important determinants of death penalty decision-making (both legal and extra-legal) to develop "culpability measures" that allow for the comparison of similarly situated defendants.

*Qualitative Measures.* Some analysts have relied on qualitative culpability measures to classify cases as similar or dissimilar based on legitimate characteristics that best explain the behavior of decision-makers in the death penalty process. For example, Barnett's (1985) "three-dimensional" classification system determines the presence or absence of thirty-five specific characteristics from the factual summaries of death penalty cases to categorize cases along the following dimensions: (1) certainty the defendant was

<sup>&</sup>lt;sup>426</sup> Berk (2005c) notes that proper implementation of randomized experiments is very demanding and, in some settings, alternative research designs may be preferable (see also Heckman and Smith 1995).

the deliberate killer (clearly no, clearly yes, or neither); (2) close prior relationship between defendant and victim (yes or no); and (3) vileness or heinousness of the killing (elements of self-defense, a vile killing, or neither) (pp. 1364–66). Each dimension is given a numeric score (zero, one, or two) and the classification scheme produces a total of eighteen potential categories of "legally" similar cases (see also Baldus *et al.* 1985, pp. 1381–82; Baldus *et al.* 1990, pp. 51–52). The scores are also summed to produce an overall "culpability score" to allow for comparison with more cases. While some have advocated this approach because of its intuitiveness (see, e.g., Keil and Vito 1989, 1996), the techniques used to identify these characteristics often vary considerably and the subjectivity of the measure undermines its replicability.

*Non-weighted Quantitative Measures.* Due to the shortcomings of qualitative classification systems, scholars have developed numerous quantitative measures to compare similarly situated defendants in the capital charging-and-sentencing process (see, e.g., Baldus 1980; Baldus and Cole 1977; Baldus and Woodworth 1983). Perhaps the most straightforward of the quantitative measures is the *legislative criteria* measure, which is simply the sum of the number of case characteristics that make the defendant death-eligible under a particular state's death penalty statute (i.e., statutory aggravating circumstances) (Baldus 1991; Baldus *et al.* 1985). For states with statutory mitigating circumstances, a measure indicating the number of these characteristics in each case can also be computed (Baldus *et al.* 1998). A related measure, the *salient factors* measure, relies on the appellate opinions of state courts' proportionality reviews to identify the

"most prominent statutory aggravating circumstance(s) and other relevant aggravators and mitigators" (Baldus *et al.* 1998, p. 1674).<sup>427</sup>

While informative, these measures suffer from several shortcomings. Perhaps the most obvious (and most disadvantageous) is the fact that cases with similar scores can be significantly different with respect to the actual characteristics present in the case. Although cases with increasing scores on the aggravation scale measure (and decreasing scores on the mitigation scale) are appropriately described as more culpable, prosecutors, judges, and jurors may view these cases very differently. Furthermore, decision-makers are likely to perceive some characteristics as more significant than others, so giving these characteristics equal weight in a summation scale fails to accurately reflect how these factors actually influence the decision-making process in death penalty cases.<sup>428</sup>

*Regression-Based Measures*. In an attempt to minimize the shortcomings associated with using qualitative culpability measures and non-weighted quantitative summary scales, death penalty researchers have increasingly turned their efforts to using regression-based measures (for a critical review, see Berk *et al.* 2005). In particular, these analysts have used multiple regression techniques to simultaneously estimate the impact of legitimate, illegitimate, and suspect case characteristics on the capital chargingand-sentencing process (Baldus *et al.* 1998, p. 1672). Regression models have been used to estimate the impact of various case characteristics at successive stages of the capital punishment process, as well as the effect of these factors on the entire series of decisions

<sup>&</sup>lt;sup>427</sup> Gross and Mauro's (1989, p. 59) aggravation scale included the three strongest non-racial predictors of capital sentencing—contemporaneous felony circumstance, relation of victim to offender, and number of victims—found in the SHR. The effects of the three variables varied across the eight states they examined (Arkansas, Georgia, Florida, Illinois, Mississippi, North Carolina, Oklahoma, and Virginia), but each circumstance was "more strongly associated with capital sentencing in each of the states than any other non-racial variable is in any of them" (see also Wolf 1964).

<sup>&</sup>lt;sup>428</sup> The salient factors method is particularly problematic because appellate review is an *ad hoc* process, and therefore it gives very little insight into the operation of the system as a whole (Baldus *et al.* 1998, p. 1674).

(see Paternoster *et al.* 2004; Unah and Boger 2002). Due to the fact that the variables of interest in death penalty research (e.g., whether or not a case is noticed for the death penalty) violate the classical linear regression assumptions, death penalty scholars have increasingly relied on regression models for limited and categorical (i.e., truncated, count, binary, ordered, and unordered) outcomes (see Long 1997; McCullagh and Nelder 1989). Similar to the classical linear regression model, these models (often referred to as *generalized linear regression models* or GLMs) allow the analyst to uncover the *independent* effects of case characteristics on the capital charging-and-sentencing process and summarize the relationship of these variables to the outcome(s) of interest in a mathematical equation.

The most common regression model employed in death penalty research is the logit or probit regression model, used for dichotomous outcomes (e.g., sentenced to death/not sentenced to death) (see, e.g., Baldus *et al.* 1998; Gross and Mauro 1989; Paternoster *et al.* 2004; Songer and Unah 2006; Unah and Boger 2001); however researchers have also used models for polytomous outcomes (e.g., ordered and multinomial logit/probit) (Sorensen and Wallace 1999). In fact, with the expansion of possible sentencing outcomes in death penalty cases—conviction of a non-capital offense, life with the possibility of parole, LWOP, and death—models moving beyond the simple death/not death distinction may more accurately capture the relationships between case characteristics and the outcomes of interest.

*Alternative Classification Methods.* Conventional regression approaches to studying the administration of capital punishment have been criticized by some scholars because death penalty data often fail to meet the requirements of the statistical models

(Berk 2003). Several analysts have advocated using propensity scores (Berk *et al.* 2005). classification and regression trees (CART) (Berk et al. 2005; Klein and Rolph 1989), and random forests (Berk et al. 2005) to more accurately asses the impact of extra-legal factors (e.g., race/ethnicity and region) on capital charging-and-sentencing decisions. *Propensity scores* have been used in observational studies to adjust for biases resulting from non-random selection into various treatment groups. Propensity scores are the estimated probability of membership into treatment groups-typically an experimental and control group—that are computed from a separate logit or probit regression equation with different predictors and functional forms from the equation for the outcome of interest (see Rosenbaum and Rubin 1983). The propensity score methods are more flexible than conventional regression adjustments and, therefore, may more effectively reduce the biases resulting from confounding (Smith 1997).<sup>429</sup> In death penalty research, racial predictors must be viewed as "treatments" and propensity scores are used to control for confounded variables (Berk et al. 2005). This method remains problematic for death penalty research, however, because there is no theoretical justification for specifying the selection equation for racial predictors due to the fact that individuals do not have their race/ethnicity "assigned" (Berk et al. 2005).<sup>430</sup> The propensity score methods also tend to compromise model fit—relative to conventional regression approaches—because the

<sup>&</sup>lt;sup>429</sup> Although propensity scoring methods may provide more flexibility than conventional regression approaches, these methods rely on rather strong assumptions that can be difficult to fulfill in practice (Imbens 2004).

<sup>&</sup>lt;sup>430</sup> Experimental studies randomly altering defendants' and victims' race/ethnicity have been used to investigate the role of race/ethnicity in the criminal sentencing process; however, these studies have been criticized on two important grounds. First, these experiments have limited generalizability because they almost exclusively rely on the responses of college students who tend to be dissimilar from the general population from which juries are drawn (Field and Barnett 1978). Second, experimental approaches do not alter the fact that race/ethnicity serves as a marker for other factors that are behind prosecutors' decisions to seek the death penalty and juries' decisions to imposed the death sentence (Berk 2003, p. 211). This implies that the most fruitful research must illuminate these important mediating/confounding factors perhaps at multiple levels of analysis.

emphasis of this approach is on isolating causal effects rather deriving a model that most closely fits the data.

The primary purpose of *classification and regression trees* and *random forests* models is to minimize classification errors rather than represent how the data were generated or how causal effects are produced (for detailed discussions of these methods, see Breiman 2001; Breiman *et al.* 1984). The goal of CART analysis is to use the relationships between the predictors and the response variable to subset the data, such that within each subset the values of the response variable are similar as possible. This is accomplished by partitioning the space defined by the data one partition at a time in recursive fashion; once a partition is constructed, it is not reconsidered when later partitions are defined (Berk 2005a).<sup>431</sup> CART and random forests models perform best when sample sizes are fairly large and variables are not extremely skewed because each subsequent stage splits the groups into smaller, more homogeneous, units. This requirement is particularly problematic for capital punishment research because of the infrequency in which homicide cases are noticed for the death penalty and advance to the penalty phase.

Berk and colleagues' (2005) recent reanalysis of the Maryland death penalty data (see Paternoster and Brame 2003) suggests that the race-effects reported in the study based on logistic regression—were unreliable and, using propensity score, CART, and random forests models, they discovered that racial/ethnic effects were substantially weaker. Specifically, they used CART models to measure the impact of race/ethnicity in terms of the extent to which the inclusion of the racial variables in the analysis increased

<sup>&</sup>lt;sup>431</sup> Classification trees apply to models with a categorical endogenous variable, whereas regression trees are appropriate when the endogenous variable is continuous. Both classification and regression trees allow for a mixture of categorical, interval, and continuous exogenous variables.

the accuracy of the model in predicting correct sentencing outcomes. They found that the inclusion of the racial variables had very little or no effect on the predictive power of the models, and strongly cautioned against using conventional regression approaches in death penalty charging-and-sentencing research. Conversely, Baldus and colleagues (1998) have cautioned against using CART analysis for assessing the impact of race/ethnicity on the capital charging-and-sentencing process because such analytical approaches potentially mask significant race effects that are detectable in a multiple regression framework. They argue that in order for a partition based on adding racial predictors to the model to improve the predictive power of the model in a CART analysis, the death sentencing rates for the two racial subgroups must straddle .50 (fifty percent). Baldus et *al.* suggest that there can be a substantial increase in the risk of death (e.g., from .05 to .15 or from .80 to .95) for one racial group compared to another in a particular category without any improvement in the prediction rate. "[I]n other words, as a metric, change in the correct prediction rate ignores increased risk of death associated with race, unless one race has a death-sentencing rate under .50 and the other has a death-sentencing rate over .50 in one or more (nonracial) categories" (p. 1666 n.80). The small sizes during the later stages of the capital charging-and-sentencing process in Berk et al.'s study also bring into question the reliability of their results based on CART and random forests models. Consequently, although Berk et al. strongly caution against using conventional regression approaches for causal modeling in death penalty research, their results do not seriously challenge the vast majority of death penalty studies reporting that racial/ethnic variables—particularly race-of-victim—significantly improve model fit and continue to be among the strongest predictors of capital charging-and-sentencing decisions, even

after controlling for dozens of legally legitimate case characteristics (Baldus *et al.* 1990; Paternoster *et al.* 2004; Unah and Boger 2001). Moreover, while propensity scores may be better able to reduce potential bias in the estimation of "causal" effects, and CART/random forests models may provide greater predictive accuracy (relative to conventional regression), these methods do not represent how the data were generated and are much more difficult to interpret (Berk 2003, p. 212). In other words, these models typically fail to offer answers to the questions that are of primary concern to death penalty analysts—e.g., *how* are charging-and-sentencing decisions made in the capital punishment process? In light of the serious shortcomings associated with these alternative approaches, along with the widespread use of conventional regression modeling in death penalty charging-and-sentencing research, regression-based models are used in this study.

### 7.2.2 Multilevel Regression Models

The growing emphasis on the role of context in the criminal sentencing process has encouraged scholars to develop more elaborate theories of how differences in individual sentencing outcomes across jurisdictions may result from the influence from environmental factors, net of case-level determinates (see Dixon 1995; Eisenstein, Flemming, and Nardulli 1988; Ulmer 2000). While death penalty analysts have readily acknowledged the importance of context on the capital punishment process for some time (see, e.g., Baldus *et al.* 1986a; Gross and Mauro 1984; Kleck 1981; Nakell and Hardy 1987), few have attempted to systematically explore *how* location impacts the administration of capital punishment.<sup>432</sup> As Berk and colleagues (2005, p. 31) note, "[I]f the concern is about illegitimate factors affecting capital cases, the impact of location needs to be studied in much greater depth. As now measured, a county or city is just a proxy for processes that are not analyzed." To be sure, researchers have employed multilevel models when exploring the impact of contextual-level factors on criminal sentencing outcomes in non-capital cases (see Britt 2000; Johnson 2005; Kautt 2002; Ulmer and Johnson 2004; Wooldredge and Thistlethwaite 2004). Only recently, however, have scholars developed the necessary estimation theory<sup>433</sup> and commercial

<sup>&</sup>lt;sup>432</sup> Three notable exceptions are Jacobs and Carmichael's (2002) examination of states' reimposition of the death penalty after the Furman decision, Liebman and colleagues' (2002) analysis of death sentence reversals, and Poveda's (2006) study of geographic disparities in the use of the death penalty in Virginia following the Furman case (see Chapter Four); however these three studies were not directly concerned with the initial charging-and-sentencing process and none explored cross-level effects. Jacobs and Carmichael used a pooled time-series analysis—a technique closely related to multilevel analysis (Berk 2003, p. 180)—to examine the influence of state-level characteristics on how quickly states' re-enacted death penalty legislation after the brief moratorium imposed by the U.S. Supreme Court. Liebman et al.'s data were amenable to multilevel analysis because of the hierarchical structure of the data (i.e., individual cases were nested into counties and states), but they aggregated their case-level data to the state and county levels and focused their analyses on the influence of contextual factors on state- and county-level deathsentencing *rates* (via poisson and negative binomial regression). In a subsequent analysis, Liebman and colleagues employed multilevel models on their aggregated data-nesting states within years-to predict the probability of a death penalty reversal (Gelman et al. 2004). They also disaggregated their county-level data to conduct case-level analyses to examine the impact of individual case factors on federal habeas review (their third stage of review). They discovered that the number of formal aggravating factors present in a case had a strong impact on the probability that a case was reversed—the more aggravated a case, the lower the probability of a reversal. Regrettably, this follow-up study also failed to simultaneously consider county- and case-level factors. Poveda (2006) used discriminant analysis to differentiate between "retentionist" (one or more death sentences/executions) and "abolitionist" (no death sentences/executions) jurisdictions in Virginia between 1978 and 2001. Poveda's unit of analysis was the county, so case-level factors (i.e., race of offender and victim, offender-victim relationship, and homicide circumstance) were aggregated to the county-level in the analysis. Poveda's analysis revealed that certain contextual level factors (i.e., population size and percentage black) significantly distinguished jurisdictions that employed the death penalty from jurisdiction that did not employ the death penalty.

<sup>&</sup>lt;sup>433</sup> For detailed reviews of restricted maximum likelihood (REML) approaches (e.g., marginal and penalized quasi-likelihood), see Wong and Mason (1985), Goldstein (1991), Schall (1991), Breslow and Clayton (1993), Longford (1993), and Rodríguez and Goldman (2001). Hedeker and Gibbons (1994), Pinheiro and Bates (1995), and Rabe-Hesketh *et al.* (2002b) discuss approximations to the marginal likelihood based on numerical integration using the Gauss-Hermite quadrature or adaptive Gauss-Hermite quadrature and Raudenbush *et al.* (2000) discuss the LaPlace approximation. For a generalization of the quasi-likelihood method to multivariate regression models, see Zeger *et al.*'s (1988) and Hardin and Hilbe's (2002) discussions of generalized estimating equations (GEE).

computer software<sup>434</sup> to analyze the types of data (i.e., limited and categorical variables) of primary interest to death penalty researchers.

There are important substantive reasons for employing multilevel models for nested data. The first is that such models allow the analyst to avoid committing *ecological* and *atomistic* fallacies (Luke 2004, pp. 5–6). Ecological fallacies occur when relationships observed in groups are assumed to hold for individuals. Conversely, atomistic fallacies occur when inferences about groups are incorrectly drawn from individuals. It is important to note that these fallacies are primarily a problem of *inference*, not measurement. In other words, it is permissible to *describe* higher-level data using lower-level information, but it is inappropriate to assume that relationships discovered at one particular level of analysis occur in the same fashion at some other level (King 1997). A second significant substantive reason for using multilevel models is the investigation of *causal heterogeneity*. Much social theory suggests that the causal effects of lower-level predictors are conditioned or moderated by higher-level predictors (i.e., cross-level effects)—multilevel models provide a framework under which analysts can systematically examine these effects (DiPrete and Forristal 1994).

Multilevel models also have several important statistical advantages over the classical regression model when analyzing nested data structures. First, they correctly adjust for the non-independence of observations resulting from the nesting of smaller units within larger units (i.e., clustering), therefore producing unbiased standard errors of

<sup>&</sup>lt;sup>434</sup> Numerous specialized and general purpose software packages to estimate multilevel nonlinear models have become readily available over the past decade. Specialized programs include: *aML* (Lillard and Panis 2003), *HLM* (Raudenbush *et al.* 2004), *LISREL* (Joreskog and Sorbom 1996), *MIXOR/MIXNO/MIXPREG* (Hedeker 1998, 1999; Hedeker and Gibbons 1996), *MLn/MLwiN* (Rasbash *et al.* 2004), *Mplus* (Muthén and Muthén 2004), and *VARCL* (Longford 1990). Popular routines in general software packages include: *nlme/glme* in *S-Plus* (Pinheiro and Bates 2000), *PROC MIXED* in *SAS* (Little *et al.* 1996), *GLLAMM* in *Stata* (Rabe-Hesketh and Skrondal 2005), and the Bayesian Markov Chain Monte Carlo (MCMC) package *WinBUGS* (Spiegelhalter *et al.* 2003).

parameter estimates. Units in the same cluster share cluster-specific influences; however it is unrealistic to include all cluster-specific influences as covariates in the model. This results in cluster-level unobserved heterogeneity leading to dependence between responses for units in the same cluster after conditioning on covariates (the association between units in the same cluster is often referred to as the *intra-class* or *intra-cluster* correlation). Ignoring the hierarchical structure of the data often results in standard errors that are biased towards zero, increasing the risk of Type I errors. Multilevel models account for this unobserved heterogeneity by including random effects (i.e., randomly varying intercepts and slopes) to account for parameter differences across the clusters. Random intercepts represent unobserved heterogeneity in the individual level outcome and random slopes represent unobserved heterogeneity in the effects of explanatory variables on the response variable (see Skrondal and Rabe-Hesketh 2004, pp. 49–50).

Second, they provide better estimates of individual-level effects within clusters by using all available data to produce estimates that are a weighted composite of the information from a particular cluster and the relations that exist in the overall sample. Estimating separate regression equations for each cluster is problematic because many clusters may have small sample sizes, so parameter estimates are likely to be extremely inefficient (i.e., they have very large standard errors) and it becomes difficult to develop reliable predictions. Using pooled data from across all clusters, ignoring the hierarchical structure of the data, is also likely to produce bias parameter estimates because of unmodeled cluster-specific influences (Raudenbush and Bryk 2002). The weighted estimates produced by multilevel models—often called *empirical Bayes* estimates (Lindley and Smith 1972) or "shrunken" estimates (James and Stein 1961)—lie between

the group mean (i.e., cluster-specific point estimate) and the grand mean. More weight is given to the group mean when the cluster is large because these clusters contribute more information to the model than clusters with only a small number of units (Luke 2004).

Third, they allow for the partitioning of variation in the individual-level outcome into within- and between-cluster effects (Skrondal and Rabe-Hesketh 2004, p. 51). Multilevel models have a complex error structure because the total variability in individual outcomes is comprised of two components: the within-cluster variance and the between-cluster variance. Decomposing the random part of the multilevel model into unit-specific and cluster-specific effects allows the analyst determine how much variability in the outcome can be attributed to each level. It is also possible to assess model fit (e.g., calculate a  $R^2$  statistic) for each level to assist in model building (Luke 2004, pp. 35–37).

Fourth, they offer a significant advance over traditional methods statistically linking macro-level data to individual outcomes. One traditional approach to account for cluster-specific influences is to add indictor variables for each cluster (except one) to the model to control for unobserved similarities among units in the same cluster.<sup>435</sup> The approach is limited in that it only accounts for subgroup differences, but does not allow the analyst to explain why these differences are present across clusters—an important concern of much social theory. Another traditional approach consists of adding interaction terms between individual- and cluster-level covariates in an attempt to explain variability in lower-level effects (see, e.g., Friedrich 1982). The major drawback of this approach is that it implicitly assumes that the interaction terms fully account for

<sup>&</sup>lt;sup>435</sup> Alternatively, indicator variables for all clusters can be included in the model, but no intercept can be estimated.

differences across the clusters because it only incorporates random error at the lowest level of analysis. (In other words, this model rests on the unlikely assumption that there is no cluster-level heterogeneity.) Ideally, analysts could combine these two approaches to exploit their unique advantages, but this is not possible because there are insufficient degrees of freedom to estimate both the interaction effects and the indicator variables in the same model. In multilevel models, random intercepts and slopes at lower-levels become outcomes at higher-levels (Raudenbush and Bryk 2002, p. 27)—random coefficient models allow the analyst to both estimate and model the variability in these regression coefficients across the clusters.

Finally, multilevel models recognize that contextual units in a sample are a subset of the population of interest, and allow for generalizations to the larger population of interest (see, generally, Aitkin and Longford 1986). Treating the cluster effect as a random factor is appropriate when making inferences regarding the population of clusters rather than the specific clusters in the dataset (Skrondal and Rabe-Hesketh 2004, p. 51). Inferences based on traditional approaches that estimate a separate fixed parameter for each cluster are limited to the clusters included in the sample, and not generalizable to the larger population from which the sample data were drawn.

Recent advances in multilevel modeling have allowed researchers to relax several of the assumptions on which classical linear regression is based to better analyze death penalty data. In addition to the improved modeling strategies for limited and categorical dependant variables and clustered data, extensions to the conventional regression model allow researchers to accurately study cross-classified (Raudenbush and Bryk 2002, pp. 373–98; Snijders and Bosker 1999, pp. 155–65), multiple-membership (Skrondal and

Rabe-Hesketh 2004, pp. 60–63), and spatially dependent (Zhang 2002) data. Nonetheless, the validity of inferences based on multilevel models depends on the tenability of assumptions about both the structural (fixed) and random (stochastic) aspects of the model. Serious problems may arise when the functional form of the relationship between the predictor variable(s) and the outcome variable is misspecified or when the error term is correlated with one or more of the explanatory variables in the model (Gujarati 1988, pp. 455–56). These specification assumptions apply at each level in a hierarchical model *and* misspecification at one level may bias estimates in another. Moreover, because higher-level equations may have correlated errors, the misspecification of one equation can bias the estimates in another (Raudenbush and Bryk 2002, p. 253). Fortunately, diagnostic tools for model selection and model adequacy continue to be developed and applied researchers have numerous empirical procedures to assist them in the model-building process (for detailed discussions, see Raudenbush and Bryk 2002, pp. 252–87; Skrondal and Rabe-Hesketh 2004, pp. 262–81; Snijders and Bosker 1999, pp. 86–98). Due to the aforementioned theoretical and statistical issues when analyzing nested data, multilevel regression models will be used when examining the relative impact of case- and county-level factors on Georgia's capital charging-andsentencing process. More specific details on these analytical models are provided in the next chapter.

# 7.2.3 Missing Data and Selection Bias Issues

Perhaps two of the most useful methodological advances in regression modeling for death penalty researchers have been with respect to the proper handling of missing data (Schafer 1997; van Buuren, Boshuizen, and Knook 1999) and corrections for sample selection bias (see Heckman 1976; Sartori 2003). *Missing Data* problems are extremely common in social science research, especially for analysts conducting survey research and using secondary data sources (Rubin 1996). The vast majority of statistical analyses must be performed on a full data matrix, therefore the common practice among social scientists is to perform *casewise* deletion by eliminating observations that have missing data on one or more variables (Little 1992). King and colleagues (2001, p. 49) report that 94 percent of political scientists use casewise deletion (also known as *listwise* deletion) to deal with missing data on one or more variables in their models and, on average, lose one-third of their observations.<sup>436</sup> Similarly, Royston (2004, p. 227) notes that complete-case analysis in medical research may omit as many as half of the available observations. Casewise deletion is problematic because it (1) forces researchers to potentially discard much useful information about the relationships between variables, (2) results in inefficient parameter estimates due to a reduction of sample size, and (3) may bias parameter estimates if the data are not missing complete at random (Allison 2002).

Rubin (1976) identifies three processes that generate missing data with respect to the information they provide about the unobserved data: missing completely at random (MCAR), missing at random (MAR), and missing not at random (MNAR). When data are MCAR, the probability of missing data on a particular variable is neither related to the value of the variable itself nor to the values of any other variable in the data set. In other words, the subset of cases that have complete data are simply a random sample of the universe of cases. Whenever it is possible to predict that a cell in a data matrix is

<sup>&</sup>lt;sup>436</sup> Many analysts also use *pairwise* deletion for statistical procedures that work with data in pairs, such as correlations and covariances. For these procedures, statistics can be computed with complete data for pairs of items that do not need to account for missing data with other variables. A major limitation of pairwise deletion is the difficulty in computing standard errors or other measures of uncertainty because parameters are estimated on different sets of units (Schafer and Graham 2002, p. 155).

missing, the MCAR assumption is violated. Data are MAR when the probability that a particular variable will be missing depends on the other observed variables, but not the variable itself. Unlike MCAR, the MAR assumption is untestable (Schafer and Graham 2002, p. 152). If MCAR or MAR assumptions are not met, the missing data mechanism is MNAR (also referred to as *nonignorable*). When data are nonignorable, the pattern of missingness is non-random and cannot be predicted by other variables in the data set, therefore it is necessary to model the missing data mechanism to obtain unbiased estimates.

As noted above, the SHR is missing data on a significant proportion of Georgia homicide cases (see also Messner *et al.* 2002, p. 458). If casewise deletion were performed on the SHR data, nearly three-fifths of the observations would be discarded from the analysis and statistical inference about the population of potentially death eligible homicides would be based on a subset of the data (i.e., the fully observed cases). If the missing data were MCAR, the subset of cases would be a random sample of the population and the resulting parameter estimates would not be biased, although their standard errors would be significantly larger (Schafer and Graham 2002, p. 156).<sup>437</sup> The advantage of examining the entire population of cases, rather than a random sample, is that statistical inference based upon sample statistics (i.e., *p*-values) need not apply in the conventional sense because there is no sampling error (see Berk *et al.* 2005; Steffensmeier and Demuth 2001).<sup>438</sup> However, as noted above, when it is possible to

<sup>&</sup>lt;sup>437</sup> Petrin (2005, p. 8) suggests that missing data may pose particular problems for parameter estimation and inference in multi-level models *even* when the data are MCAR because the sample sizes at each level (e.g., number and size of clusters) influence statistical power for all of the parameters in the model.

<sup>&</sup>lt;sup>438</sup> When examining the entire population of cases, one primarily focuses on the direction and magnitude of the observed relationships, rather than the generalizability of these relationships to the larger population, as indicated by tests of statistical significance (Berk 2003).

predict the probability that a variable is missing information for an observation (using information from other covariates in the data), the MCAR assumption is violated and casewise deletion may generate bias parameter estimates.<sup>439</sup> According to Regoeczi and Riedel (2003, p. 180), "analyses using listwise deletion for homicide data are on based on what are almost certainly not a random subset of the full range of cases." Complicating matters, death penalty cases in Georgia (and elsewhere) are *very* rare occurrences (see Chapter Five), so it is crucial to retain as many cases as possible in the analysis of the capital charging-and-sentencing process.<sup>440</sup>

Over the past two decades, methodologists have developed several approaches to "guess" the values of missing data by using information about the association of the variable of interest with other variables in the data. Hot deck imputation (Reilly 1993), predictive-mean matching (Landerman, Land, and Pieper 1997), and propensity scoring (Rosenbaum and Rubin 1983) are popular approaches to imputing missing data, but may perform poorly when the proportion of missing cases is large, missing data are multivariate, and the pattern of missingness is non-monotone (see Allison 2000; Pérez *et al.* 2002).<sup>441</sup> Messner and colleagues (2002) recently advocated using a log-

<sup>&</sup>lt;sup>439</sup> When the proportion of cases with missing data is small, departures from the complete case analysis assumptions often have to be quite large to have a substantive impact on the results (Brame and Paternoster 2003, p. 74).

<sup>&</sup>lt;sup>440</sup> Biostatisticians studying rare diseases commonly confront a similar problem because discarded cases may contain a large proportion of participants possessing the particular condition (Schafer and Graham 2002, p. 156).

<sup>&</sup>lt;sup>441</sup> Mander and Clayton (2005) developed an algorithm combining hotdeck imputation with propensity scoring. Classic hotdeck imputation is a semi-parametric method for imputing values in a single variable that is based on observed empirical distribution of the data and makes no other distributional assumptions. Mander and Clayton's procedure uses Rubin's (1987, p. 124) propensity scoring method to improve the accuracy of the imputations by stratifying the data based on the probability of missingness and imputing data within strata independently. They note, however, that this approach is appropriate when patterns of missingness are *monotone*. Data have a monotone missing pattern when a variable is missing for a particular individual implies that all subsequent variables are missing for the same individual (Schafer and Graham 2002, p. 150). When data have a monotone missingness pattern, the analyst has greater flexibility in the choice of imputation methods.

multiplicative association model to predict unknown victim-offender relationships for official homicide data. While Messner et al.'s approach has the advantage of being robust to the "ignorability" assumption necessary for most parametric methods, it seriously underestimates the uncertainty associated with imputation/classification algorithms. Messner *et al.* also concede that, as a practical matter, violations of the ignorability assumption will be inconsequential if the imputation model is well specified (see also Rubin 1996; Schafer 1997). In fact, Regoeczi and Riedel's (2003) analysis of homicide data in Chicago and Los Angeles suggests that the ignorability assumption is very reasonable, especially when a few important predictors (e.g., clearance status) are included in the imputation model. A comparison of Messner *et al.*'s log-multiplicative approach and regression-based approaches that assumed data were MAR (Pampel and Williams 2000) suggests that there is very little difference in the imputations derived from the different techniques.<sup>442</sup> The regression-based approaches, however, can be easily extended to more realistically reflect uncertainty surrounding the imputations and, therefore, offer a preferable alternative to hot deck, predictive-mean, and logmultiplicative association models.

Rubin's (1976, 1987) regression-based *multiple* imputation approach provides a significant improvement over simple imputation methods and traditional single imputation strategies. In brief, Rubin proposes a three-step approach to the estimation of incomplete data models. First, observed data are used to impute missing values and

<sup>&</sup>lt;sup>442</sup> Fox (2004, p. 240) notes that while the MAR assumption may be suitable for the age, race, and gender characteristics of perpetrators, it may be unsuitable for offender-victim relationship data because unsolved homicides are more likely to be committed by a stranger or an acquaintance, rather than a close friend or intimate even after controlling for the characteristics of the victim or the incident. He cautions, however, that techniques for estimating data that are nonignorable may perform worse than techniques which assume the data are MAR.

incorporate estimation uncertainty (resulting from analyzing a finite number of observations) and fundamental uncertainty (resulting from unmodeled variation in the dependent variable and represented by the stochastic component of the model) in their prediction of plausible values (see also King, Tomz, and Wittenberg 2000, pp. 348–49; Schafer and Graham 2002, p. 167).<sup>443</sup> This process is repeated M times (M > 1) to create M complete data sets, with each data set containing different plausible values for missing variables to account for the uncertainty surrounding the imputations.<sup>444</sup> Second, a complete-case analysis is repeated on each data set. Finally, the point estimate of each parameter is obtained by averaging across the M separate point estimates for that particular parameter (Rubin 1987). The variance of the point estimate is computed by averaging across the M estimated variances from within each completed data set, plus the sample variance in the point estimate across the data sets (multiplied by a factor that corrects for bias) (Little 1992, p. 1235).<sup>445</sup> King and colleagues (2001, p. 53) emphasize that the purpose of the imputation model is to create predictions for the distribution of each of the missing values, not causal explanation of parameter estimates.

Several different imputation algorithms have been developed to create plausible "guesses" of missing values (see Dempster *et al.* 1977; King *et al.* 2001; Meng and Rubin 1993; Schafer 1997; Tanner and Wong 1987; van Buuren *et al.* 1999; Vermunt *et al.* 

<sup>&</sup>lt;sup>443</sup> Rubin (1987) initially proposed using the Expectation Maximization (EM) algorithm, an iterative procedure, to compute imputations for each missing value in a data set. In simple applications, the EM algorithm proceeds by (1) estimating parameters based on available data, using these parameters to impute missing values, (2) updating the parameters based upon imputed values, and (3) iterating until the change in parameters is negligible (Dempster, Laird, and Rubin 1977, p. 11).

<sup>&</sup>lt;sup>444</sup> The fraction of missingness,  $\gamma$ , governs the number of imputations, M, needed to obtain efficient estimates. The efficiency of an estimate based on M imputations is approximately  $(1 + \gamma/M)^{-1}$  (Rubin 1987, p. 114). For example, when  $\gamma = .30$  and M = 10, the standard error is 2.9 percent larger than its minimum possible value.

<sup>&</sup>lt;sup>445</sup> Reiter and Raghunathan (2007) note Rubin's rules for combining point and variance estimates are inappropriate in those settings where data is missing to protect confidentiality or to correct for measurement error.

2008), each with its on unique advantages and disadvantages. The "fully conditional specification" (FCS) approach, developed by van Burren and colleagues (van Buuren et al. 1999), offers the greatest flexibility in creating multivariate imputation models because it allows for specialized methods that are impractical under the other approaches. The FCS algorithm imputes the data on a variable-by-variable basis by specifying an imputation model for each variable, which is then used in the imputation of the next variable (i.e., "chained equations"). This process repeats—using a Gibbs sampling procedure to impute missing values-until convergence (van Buuren et al. 2006). FCS allows the analyst to preserve unique features of the data such as bounds, skip patterns, interactions, bracketed responses, and appropriate constraints between different variables in order to avoid logical inconsistencies in the imputed data (Raghunathan et al. 2001).<sup>446</sup> FCS can also be modified to incorporate design features, such as stratification and clustering (Reiter, Raghunathan, and Kinney 2006; Yucel, Schenker, and Raghunathan 2006).<sup>447</sup> Although the statistical properties of FCS have been difficult to establish, simulation work has shown that FCS performs well in a variety of applications (see Cranmer 2007; Raghunathan et al. 2001; van Buuren et al. 2006).

It must be emphasized that imputation approaches are only appropriate when the missing values actually exist, but are unobserved in the data. Also, whenever possible, it

<sup>&</sup>lt;sup>446</sup> Since FCS permits separate models for continuous and categorical variables, it avoids the potential bias introduced by algorithms using a multivariate normal distribution that impute categorical variables as if they were continuous and require the analyst to "round" the value to the nearest discrete value in the dataset (see Horton, Lipsitz, and Parzen 2003; van Buuren 2007). As Schafer (1997, p. 148) explains, rounding is appropriate when the missing data take on many values and the marginal distribution is approximately unimodal and symmetric.

<sup>&</sup>lt;sup>447</sup> Ignoring the complex design features of the data during the imputation stage will result in invalid design-based inferences (Raghunathan *et al.* 2001, p. 93). Reiter and colleagues (2006) show that the inclusion of indicator variables in the FCS framework for cluster effects greatly reduces the bias relative to disregarding the hierarchical structure of the data. The indicator variable approach is even superior to hierarchical imputation models when data are missing from several continuous and discrete variables or when cluster effects follow a non-normal distribution.

is preferable to use external sources to discover the "true" values of the missing data rather than to estimate these values from the data. While it is reasonable (and advisable) to use external sources to find the true values for missing data for cases that enter the capital charging-and-sentencing process, it is much more difficult to find information about these true values for death-eligible cases that were not noticed for the death penalty or did not result in a homicide conviction because significantly less information is known about these cases. Recall that the GADOC contains information on every case resulting in a murder conviction, so these data can be used to supplement SHR data; however, GADOC data are usually limited to offender and crime characteristics and contain very little information about the homicide victim(s). Fortunately, much of the missing data in the SHR is with respect to victim characteristics, so GADOC data can be most useful in supplementing these data. Special care was taken to cross-validate the data using multiple sources, but it was impossible to uncover all the information about every single case. As noted earlier, MPD opened a file on every homicide case noticed for the death penalty with an incident date between 1993 and 2000 and attempted to collect all necessary information about the case, therefore the data are reliable for cases that actually *entered* the capital charging-and-sentencing process. Another important advantage of the data is that the outcome variables (e.g., death noticing decisions, sentencing decisions, et *cetera*) are fully observed, so there is no concern over whether they are imputed efficiently (Little 1992).

*Selection Bias.* As noted earlier in the chapter, sample selection bias in criminal sentencing research may arise when some component of the decision to convict or incarcerate is relevant to the sentence determining process (Mears 1998a; Wooldredge

1998). In other words, when some determinants of the incarceration decision are also influencing the length of the sentence received. When the relationship between the incarceration decision and sentence length is purely through observed variables, one may simply account for this relationship by including the appropriate variables in the sentencing model. However, if unobservable factors affecting the incarceration decision are correlated with the unobservable factors affecting the sentencing decision, the error term for the sample used in the sentencing-decision model does not have a conditional mean of zero and is correlated with sentencing length. When these unmeasured factors responsible for selection into a subsample are correlated with the observable factors, a crucial assumption of conventional regression is violated and failure to include an estimate of the unobservable factors will lead to incorrect inference regarding the impact of the observed variables on sentencing length. Stated more technically, the regression functions will confound the substantive parameters with the parameters of the function determining the probability of inclusion into the subsample and lead to bias parameter estimates (Berk 1983; Winship and Mare 1992). According to Sorensen and Wallace (1999):

Sample selection bias is most likely to result when the pool of cases is limited to those in which a decision was made overtly during the later stages of case processing, such as the sentencing decision in a pool of convicted first-degree murderers who have advanced to the penalty stage of a capital trial. When samples are limited in this manner, the effects of racial discrimination occurring at earlier decision points are not taken into consideration. Cases involving particular racial combinations of offenders and/or victims may be systematically included or excluded from the pool of convicted capital murder cases because of bias in the pretrial stages of case processing. At the same time, the sentencing decision may be found to lack racial bias, thus giving the appearance that the system of capital punishment in the jurisdiction studied is not influenced by race. If murders involving particular racial combinations (e.g., blacks who kill whites) are regularly selected for prosecution as capital murder, even in the least aggravated cases, then during the sentencing phase, when legally relevant case criteria are taken into consideration, one should expect a lower death-sentencing rate among those cases than among cases involving other offender/victim racial combinations. A finding of no difference in the rate of death sentencing among racial combinations actually could indicate racial bias, masked by researchers' failure to consider decisions made earlier in the process.

Studies that include a broader pool of cases and earlier decision-making stages are not immune from sample selection bias if the decisions are analyzed consecutively. Most studies that have analyzed both pretrial and trial decisions have not found evidence of racial bias in sentencing [*citations omitted*]. To determine whether sentencing decisions are influenced indirectly by earlier decisions, one must factor into subsequent models the effects of race on pretrial decision making [*citations omitted*]" (pp. 563–64).

Several sentencing studies of non-capital cases have revealed that failing to correct for

sample selection bias tends to underestimate the impact of race/ethnicity on the

sentencing outcomes (Albonetti 1997; Kempf-Leonard and Sample 2001; Peterson and

Hagan 1984). Unfortunately, the vast majority of death penalty studies (e.g., Baldus et

al. 1990; Gross and Mauro 1989) fail to correct for sample selection bias (for a notable

exception, see Unah and Boger 2002). Although many of these studies report that

race/ethnicity continues to exert an effect in the later stages of the process, their estimates

of this impact are likely to be grossly underestimated. As Sorensen and Wallace noted

above, selection bias models are often necessary to capture the true impact of

race/ethnicity (and other extra-legal factors) in the later stages of the death penalty

process.<sup>448</sup> Failing to properly account for selection bias may be particularly problematic

<sup>&</sup>lt;sup>448</sup> Conviction and "in/out" incarceration decisions occurring at earlier stages are not of particular concern in death penalty cases because the vast majority of homicide cases, once noticed for the death penalty, result in a conviction *and* most serious felony convictions in Georgia have mandatory minimum sentences attached to them. The few cases that were noticed for death, but did not result in a murder conviction may, in fact, differ from the other cases in unobserved ways, but these cases are so rare that they are unlikely to

because, as Paternoster and colleague's (2004) Maryland study suggests, disparities occurring at the early stages of the process are not likely to be corrected through the advancing stages.

In series of influential papers, Heckman (1974, 1976, 1979) identified the selection problem as omitted variable bias (i.e., specification error)<sup>449</sup> and advocated modeling the probability of selection with a probit equation (i.e., a selection equation) and using the predicted values as an additional regressor in the substantive equation.<sup>450</sup> Heckman's work was primarily concerned with continuous outcomes in the substantive equation (e.g., wages from work), but his work has been expanded to include dichotomous and polytomous logit models for the selection equation (see Dubin and Rivers 1990) and substantive models with discrete outcomes (Miranda and Rabe-Hesketh 2006; Sartori 2003; van de Ven and van Praag 1981). The primary restriction of Heckman's approach, however, is that there needs to be at least one predictor variable in the selection equation that is not correlated with the outcome variable in the substantive equation (after controlling for the other covariates) for the model to be identified (sometimes called the "exclusion restriction") (Bifulco 2002; Bushway, Johnson, and Slocum 2007; Puhani 2000). Many analysts have remarked that it is extremely difficult to find a variable that influences selection in a subsample but does not influence the

have a significant impact on parameter estimates. As a result, it appears most appropriate to adjust for potential selection bias at the death penalty charging and plea bargaining phases.

<sup>&</sup>lt;sup>449</sup> Omitted variable bias encompasses both selection bias and endogeneity bias because each of these forms of bias causes inferential problems by inducing a correlation between the explanatory variables and the disturbance term (Foster 1997; Greene 2000, Chapter 11; Heckman 1978; King and Zeng 2006; Miranda and Rabe-Hesketh 2006).

<sup>&</sup>lt;sup>450</sup> In subsequent work, Heckman (1980) suggested performing a statistical test for the presence of selection bias. Essentially this test allows the research to examine whether the error terms in the selection and substantive equations are correlated in the overall population. As Achen (1986) notes, however, even if the error terms are not correlated in the overall population (as indicated by significant tests), they are correlated in the sample and can bias parameter estimates (see also Stolzenberg and Relles 1997, pp. 503–504).

outcome of interest and, if fact, many theoretical models suggest that no such variable exists (Achen 1986; Vella 1998). When an identical set of explanatory variables is included in both the selection and substantive equations, the model is identified solely from the functional form of the hazard rate (i.e., selectivity index) and inferences are extremely sensitive to the choice of the nonlinear function used and the choice of predictor variables in the selection model (Clogg and Shihadeh 1994, pp. 173–76).<sup>451</sup> In light of this potential limitation, both the Heckman "two-step" model and an alternative sample selection bias correction approach based on a multivariate probit specification are employed in this study (see discussions in Sections 8.3.5 and 8.4).

# 7.3 SUMMARY

In conclusion, seven different data sources were consulted to gather case- and county-level information on Georgia's capital charging-and-sentencing process from 1993 through 2000. These data include a wide range of information with respect to the defendant, the victim(s), the crime, other important legal officials (e.g., judges and prosecutors), and the larger social environment in which these cases occurred. The greatest strength of these data is that multiple official sources, as well as information directly from capital defense attorneys, were used for cross-validation, improving both the richness and reliability of the data. As noted in Chapter Six, many of these factors are indicators of Donald Black's five dimensions of social space (i.e., vertical, horizontal, symbolic, corporate, and normative) and therefore are used to explicitly test several of his

<sup>&</sup>lt;sup>451</sup> The use of Heckman's selection bias correction in sentencing research is not with critics. Some scholars have strongly advocated Heckman's approach (see Peterson and Hagan 1984), while others have cautioned that the approach can possibly result in more harm than good when inclusion of the hazard rate leads to high multicollinearity among predictors in the model (Bushway *et al.* 2007; Stolzenberg and Relles 1997).

core hypotheses concerning the behavior of law. Drawing from previous research, primary attention is given to three decision points in the death penalty process: (1) initial death notice; (2) guilty plea and negotiated sentence; and (3) conviction at trial and sentence at penalty phase.

Following the work of Baldus and colleagues (1990) and others (see Paternoster *et al.* 2004; Unah and Boger 2001), multiple regression techniques are used to simultaneously explore the impact of dozens of legal and extra-legal factors on the death penalty process. As discussed earlier, regression analysis is preferred over alternative approaches (i.e., propensity scores, CART, and random forests) because these alternative approaches are likely to mask racial/ethnic effects that are detectable using multiple regression (Baldus *et al.* 1998). Moreover, these alternative approaches do not allow the analyst to explicitly test theoretical derived hypotheses and are very difficult to interpret when a large number of explanatory variables are considered (see Berk *et al.* 2005).

This study also offers three significant methodological improvements over previous death penalty studies. First, it explicitly accounts for the hierarchical nature of death penalty data and attempts to explain the contextual variation in death penalty decision-making, net of case-level factors, using multilevel analytical models (cf. Unah and Boger 2002). Second, the problem of missing data is addressed using multiple imputation approaches developed by Rubin (1987) and others (see van Buuren *et al.* 1999). Recall that traditional casewise deletion approaches to handling missing data may have serious consequences when the missing data are not a mere random sample of the population of cases. Multiple imputation provides a statistically appropriate way to fill in "guesses" about missing data and preserve valuable observed data from cases that would otherwise be excluded from the analysis. Finally, this study uses well-established methods to explicitly analyze non-randomly selected samples. Selection bias correction models are often necessary to capture the impact of racial/ethnic bias in the later stages of the death penalty process (Sorensen and Wallace 1999). To date, no single death penalty study has simultaneously considered these three important issues. The following chapter describes the specific models analyzed in this study and presents their results.

#### **Chapter 8: Results**

## 8.1 PRELIMINARY MATTERS

## 8.1.1 Multiple Imputation

Slightly less than three-fifths (58.3 percent) of the data contain a missing value on at least one variable. As discussed in the previous chapter, plausible guess of missing values were imputed via the FCS algorithm (see Section 7.2.3). All case-level variables were included in the imputation equation (for a description of the variables, see Appendix B and Table 3). To properly take into account the nested structure of the data, indicator variables were included for each cluster (minus one) (Reiter et al. 2006). An initial imputation model was specified with 100 cycles of sequential regression chains and the imputations made at each cycle for every variable was recorded (see Royston 2004). Imputed values were created based on a bootstrap estimation of the parameter estimates, rather than from the posterior distribution. The bootstrap method is more robust because it does not assume the distribution of the parameter estimate is multivariate normal. Following completion of the final cycle, a "time-series" of the imputations were plotted in order to determine when the imputations stabilized. Examination of these graphs revealed that imputations of variables generally stabilized between 30 and 50 cycles, depending on the distribution of the particular variable and the proportion of missingness. Ten complete datasets were created and quantities of interest were combined using the methods discussed in the previous chapter (see Section 7.2.3).<sup>452</sup>

<sup>&</sup>lt;sup>452</sup> The *HLM* software automates "Rubin's rules" for combining point estimates and standard errors, and making statistical inferences (Raudenbush *et al.* 2004, pp. 183–84). Unfortunately, *HLM* has very limited abilities to conduct different tests. Those tests that could not be performed in *HLM* were conducted in *Stata* 

## 8.1.2 Statistical Inference and Parameter Interpretation

Statistical Inference. Recall that the data consist of the entire population of homicide cases, and not a mere sample. Statistical inference based on sample statistics, in the conventional sense, does not apply when the analysis encompasses the entire population; rather, attention is given to the direction and magnitude of the effects (Berk 2003). Statistical inference based on the models estimated on these data fall somewhere in between sample-based and population-based methods because of the missing data issue discussed earlier. The multiple imputation approach employed permits use of the entire population of cases, but missing values were filled-in based on the conditional distribution of the variables. As a result, while all cases in the population are fully observed, not all cases have fully observed values and some of those values for certain variables had to be imputed. Treating the data as a subsample would likely present too conservative of a test, while treating the data as fully observed for the population may be problematic as well. In the discussion that follows, the parameter estimates are interpreted as being derived from the entire population because all cases are included and the imputed values are based upon the observed distribution of the variables from the entire population;<sup>453</sup> nonetheless, significant tests (i.e., *p*-values) are also included in the interest of comprehensiveness.<sup>454</sup>

<sup>8.2.</sup> *Stata* has several built-in routines for analyzing multiply imputed data, and perhaps more importantly, permits the to write simple programs that automate Rubin's rules to a wider range of applications. <sup>453</sup> Also recall that all outcome variables in the study (e.g., death notice, plea bargain, death sentence) are fully observed (see Section 7.2.3).

<sup>&</sup>lt;sup>454</sup> The "population-based" approach is preferable because no out-of-sample inferences are being made. The purpose of the study is to examine the capital charging-and-sentencing system in Georgia in the time period specified, and not to make inferences about non-Georgia death penalty systems or about different time periods. Indeed, capital punishment statutes and practices very considerable across states, so generalizing from one system to another—particularly a system in a different region of the country—is unadvisable.
Parameter Interpretation. It is common for analysts to include "standardized" parameter coefficients when presenting results from multiple regression in order to directly compare the effects of variables that have different metrics (Long 1997). These standardized coefficients are typically presented as either fully-standardized or semistandardized effects. In the fully-standardized context, both the endogenous and exogenous variables are standardized to have a mean of zero and a standard deviation of one. In the semi-standardized context, either the endogenous variable or the exogenous variable(s) are standardized, but not both. Usually the exogenous variables are standardized because the analyst wishes to directly compare those effects. Fullystandardized coefficients have the greatest applicability in the linear regression framework, while semi-standardized coefficients are used when the endogenous variable is discrete (e.g., binary, count, *et cetera*) because the latent variable is, essentially, already "standardized" (see Allison 1999). But standardized coefficients lack intuitive appeal when the exogenous variable(s) of interest is binary because the mean and standard deviation typically fail to provide the analyst with directly useful information.

In the models analyzed in this study, all but five case-level variables are dichotomous, so a "one unit" change in the binary exogenous variable is directly comparable to all other binary exogenous variables. The "continuous" variables are: number of defendants, age of the defendant, number of statutory aggravating circumstances (i.e., special circumstances) charged, number of victims, and age of the victim(s) (see Appendix B). All circuit-level variables are continuous as well (see Table 4). Due to the extremely high correlation between variables that are proxies for the different dimensions of social life, variables were combined in scales corresponding to each of Black's five social dimensions (see Section 6.3.5). Per capita income, percent of individuals and families living below the poverty line, and median home value variables were combined into a "Vertical Dimension" scale ( $\alpha = 0.87$ ); percent in labor force, percent living in a different home, and percent number been married variables were combined into a "Horizontal Dimension" scale ( $\alpha = 0.83$ ); violent crime rate (UCR index), murder rate, and proportion of state's total murder rate variables were combined into a "Normative Dimension" scale ( $\alpha = 0.97$ ); and percent of high school and college graduates, black-white exposure index, percent foreign born, percent who only speak English, and percent born in Georgia variables were combined into a "Cultural Dimension" scale ( $\alpha = 0.93$ ).<sup>455</sup> All contextual variables were standardized so their effects are directly comparable.

# 8.1.3 Death Eligibility

As discussed in the previous chapter, the initial challenge in examining the impact of legal and extra-legal factors on the capital charging-and-sentencing process is determining what cases qualify as "death eligible." Some analysts posit that nearly any homicide in Georgia could be deemed eligible for the death penalty because of Georgia's felony murder statute and the "catch-all" statutory aggravating circumstance: "In Georgia...any homicide is potentially a capital offense" (Kuziemko 2006, p. 137, n.15).<sup>456</sup> While this may be theoretically true, as a practical matter, this definition is overinclusive—it is extremely uncommon for a prosecutor to *solely* charge the "catch all"

<sup>&</sup>lt;sup>455</sup> For a discussion of the calculation and interpretation of the alpha ( $\alpha$ ) reliability measure, see Cronbach (1951).

 $<sup>^{456}</sup>$  Recall that felony-murder rule is a doctrine holding that any death resulting from the commission or attempted commission of a felony is murder (see Ga. Code Ann. § 16-5-1(c)).

special circumstance without charging other, more specific, statutory aggravators. Nonetheless, this overly broad classification is employed as one type of death-eligibility measure: every homicide case resulting in a murder conviction—proxy for strength of evidence—was deemed as eligible for capital punishment.

A different approach, which is becoming more common in the empirical literature on the death penalty, defines death eligible defendants as those who qualify for the death penalty under either the "B1" or "B2" statutory aggravating circumstances (see note 412). Recall that, under B1, the defendant is eligible for the death penalty when "[t]he offense of murder, rape, armed robbery, or kidnapping was committed by a person with a prior record of conviction for a capital felony."<sup>457</sup> According to the B2 statutory aggravating circumstance, a defendant is eligible for the death penalty when "[t]he offense of murder, rape, armed robbery, or kidnapping was committed while the offender was engaged in the commission of another capital felony or aggravated battery, or the offense of murder was committed while the offender was engaged in the commission of burglary or arson in the first degree" (see Section 5.2.1). Defendants who were convicted of contemporaneously committing these crimes were categorized as death eligible.<sup>458</sup> Defendants convicted of murdering multiple victims were also categorized as death eligible because multiple victim homicides satisfy the requirements of the B2 statutory aggravating

<sup>&</sup>lt;sup>457</sup> Also recall that, under the Georgia statute, capital felonies are defined as murder, rape, armed robbery, or kidnapping (see Chapter Five)

<sup>&</sup>lt;sup>458</sup> The major limitation of this measure is that the Georgia Department of Corrections does not technically distinguish between felonies committed during the actual commission of the murder and felonies that the defendant was convicted of during the same trial as the murder. To assess the plausibility of the "contemporaneous felony" assumption, I randomly selected several cases from the Department of Corrections website that lists the separate offenses (if a multiple offense case) by the incident date. In the vast majority of these cases, the murder and other felony were committed on the same day. This suggests that, for most cases, the B2 measure is valid.

circumstance.<sup>459</sup> Admittedly, these two factors are under-inclusive and ignore a host of other factors listed in the statute. Unfortunately, the SHR data only provide data on these two factors. It is worth noting, however, that prior research has identified these two factors as "the most commonly used factors in death sentence cases, and thus account[ing] for a high proportion of death eligible cases" (Pierce and Radelet 2002, p. 66).

Twenty-nine percent of the cases in the overall population of murder and manslaughter convictions in Georgia from 1993 through 2000 qualify as "death eligible" according to the B1 or B2 categorization. This statistic is consistent with the other studies examining the proportion of death eligible homicides. For example, Baldus *et al.* (1986b, p. 152) suggest that between 10 and 25 percent of murders and non-negligent manslaughters are death eligible. Similarly, Paternoster and Brame (2003) reported that 22 percent of homicides in Maryland between 1978 and 1999 were death eligible.

Since both approaches have their unique advantages and disadvantages, I analyze models predicting death-noticing decisions using both definitions of death eligibility (see Tables 5, 6 and 7 for summary statistics of the subsamples).

## 8.2 MODEL SPECIFICATIONS

As explained in Chapter Seven, the hypotheses presented in Chapter Six are examined through a series of multilevel models. In particular, a set of random-intercept and random-slope logistic regression models (when applicable) are estimated, each

<sup>&</sup>lt;sup>459</sup> To be sure, the multiple victim measure is imperfect because it is possible that a defendant murdered multiple victims on different days, or even possibly on the same day but in unrelated situations. The vast majority of cases, however, are single victim (87.3 percent), so this measure, in of itself, does not influence the categorization of most defendants. Perhaps more importantly, the B1 and B2 criteria are not mutually exclusive—in fact, they share considerable overlap. It is possible, even likely, then, that a defendant's categorization as death eligible will be valid on one or more of these measures.

testing specific predictions derived from Black's theory of law and applied to the capital charging-and-sentencing context. The random-intercept logistic regression model takes the following form:

$$P(Y_{ijt}=1|\mathbf{x}_{ijt}, \mathbf{w}_{j}, \gamma_{t}, \zeta_{j}) = F(\beta_{0j} + \beta_{1}X_{1ijt} + \beta_{2}W_{2j} + \dots + \beta_{q}X_{qijt} + \beta_{q}W_{qj} + \gamma_{t} + \zeta_{0j}), \quad [1]$$

where *F* is the cumulative logistic distribution,<sup>460</sup> *i* indexes the case, *j* indexes the judicial circuit, and *t* indexes the year. This model makes the following assumptions: (a) conditional on  $\mathbf{x}_{ijt}$  and  $\mathbf{w}_j$ ,  $Y_{ijt}$  is an independent Bernoulli random variable with probability given by [1]; (b)  $P(Y_{ijt} = 1)$  depends on  $\mathbf{x}_{ijt}$  and  $\mathbf{w}_j$  through the logistic function; (c)  $P(Y_{ijt} = 1)$  is governed by a vector  $\boldsymbol{\beta}$  of *q* common structural parameters, *T*-1 incident-year dummies, and  $\zeta_{0j}$  (*zeta*) (Raudenbush and Bryk 2002, pp. 457–58). *Zeta*, the deviation of the cluster-specific intercept from the mean, represents the combined effect of all omitted circuit-specific covariates that influence  $P(Y_{ijt} = 1)$ , where  $\zeta_j \sim N(0, \psi)$ . The total *conditional* variance of  $Y_{ijt}$  is  $Var(\zeta_{0j} + \varepsilon_{ijt}) = \psi + \theta$ , where  $\psi$  (*psi*) is the between-cluster variance and  $\theta$  (*theta*) is the within-cluster variance.<sup>461</sup> The intraclass correlation,  $\rho$  (*rho*), represents dependence among the dichotomous responses in the same cluster, given  $\mathbf{x}_{ij}$  and  $\mathbf{w}_{ij}$ , and is analogous to the proportion of total residual variance that is due to the between-cluster (i.e., between-circuit) residual variance:  $\rho = \psi/(\psi + \theta)$ .

In all of the analyses employed,  $\mathbf{x}$  and  $\mathbf{w}$  are, respectively, vectors of case and judicial circuit characteristics. This model specification, often referred to the *reduced*-or

<sup>&</sup>lt;sup>460</sup> The cumulative logistic function is defined as:  $F(z) = \exp(z) \times [1 + \exp(z)]^{-1}$  (Skrondal and Rabe-Hesketh 2004, p. 23).

<sup>&</sup>lt;sup>461</sup> The total *unconditional* variance of  $Y_{ijt}$  is equal to: Var( $\beta + \zeta_j + \varepsilon_{ij}$ ) (Rabe-Hesketh and Skrondal 2005, p. 7). Since logit model assumes  $\theta = \pi^2/3$  (Skrondal and Rabe-Hesketh 2004, p. 60),  $\varepsilon_{ijt}$  is typically excluded when writing out the equation for the model, as in [1].

*mixed-effects form* (Luke 2004), differs from the two-stage formulation popularized by Raudenbush and Bryk (2002), but yields identical results (Skrondal and Rabe-Hesketh 2004, p. 50).<sup>462</sup> The random-coefficient model relaxes the assumption that the clusterspecific intercept is uncorrelated with the regressors and permits slope estimates to vary across clustered units (e.g., judicial circuits). Equation [1] is extended to include a random intercept for the specified  $\mathbf{x}_{ii}$  regressors:

$$= F(\beta_{0j} + \beta_1 X_{1ijt} + \beta_2 W_{2j} + \dots + \beta_q X_{qijt} + \beta_q W_{qj} + \gamma_t + \zeta_{0j} + \zeta_{qj} X_{qijt}).$$
[2]

Equation [2] can be extended to include cluster-level predictors of the randomly varying slope(s):

$$= F(\beta_{0j} + \beta_1 X_{1ijt} + \beta_2 W_{2j} + \dots + \beta_q X_{qijt} + \beta_q W_{qj} + \gamma_t + \zeta_{0j} + \zeta_{qj} X_{qijt} + \beta_q X_{qijt} W_{qj}).$$
 [3]

The inclusion of a cluster-level predictor(s) of the varying slope estimate is commonly referred to as a "cross-level interaction" (Rabe-Hesketh and Skrondal 2005, p. 88; Raudenbush and Bryk 2002, p. 159).<sup>463</sup> The intra-class correlation,  $\rho$ , for randomly varying slopes can also be computed (Raudenbush and Bryk 2002, p. 78). In contrast to the random-intercept model, however, the random-slope model induces heteroscedastic responses since the conditional variance depends on the value of the  $X_{ijt}$ , as well as the intraclass correlation (Skrondal and Rabe-Hesketh 2004, p. 54):

$$= F[(\beta_{0j} + \zeta_{0j}) + (\beta_q + \zeta_{qj})X_{qijt} + \gamma_t).$$
 [4]

Now  $\zeta_{0j}$  is the residual deviation of the level-1 intercept,  $\beta_{0j}$ , from the population intercept after the level-1 intercept has been predicted from the level-2 predictor; therefore  $\zeta_{0j}$  represents the part of the deviation between the cluster-intercept and overall

<sup>&</sup>lt;sup>462</sup> The *HLM* software requires the two-stage formulation, whereas *Stata* permits both types of specifications (via the *GLLAMM* feature) (Rabe-Hesketh and Skrondal 2005, p. 88).

<sup>&</sup>lt;sup>463</sup> Vector notation for the random coefficient model is:  $y_{ijt} = \mathbf{x'}_{ijt} \mathbf{\beta} + \mathbf{z'}_{ijt} \mathbf{\zeta}_j + \varepsilon_{ijt}$ , where  $\mathbf{x}_{qijt}$  denotes covariates with fixed coefficients and  $\mathbf{z}_{qijt}$  covariates with random coefficients (Skrondal and Rabe-Hesketh 2004, p. 55).

intercept that cannot be predicted from the W (see Cohen *et al.* 2003, pp. 554–55). The random intercept variance and the correlation between the random intercept and slope are sensitive to the location of  $X_{ijt}$ , and the interpretation of these random components largely depends on the whether a "zero" value for the  $X_{qijt}$  (and  $W_{qj}$ ) is meaningful (Raudenbush and Bryk 2002, p. 32). As discussed early, all but five of the case-level covariates are binary, therefore based on the coding conventions adopted in this study, a "zero" value on the variable indicates an absence of that particular characteristic. The five non-binary case-level variables are grand-mean centered and, due to standardization of the contextual variable scales indicating Black's various social dimensions, those variables are centered at their means—consequently, a "zero" value is interpreted as the mean value of those particular variables.

It is worth nothing that level-one (case-level) slope coefficients may also be specified as non-randomly varying—i.e., the variation in the effect of the variable across clusters is solely a function of the  $W_{qj}$ —the cluster-level predictors. This approach is advisable when, after taking into account the effect of  $W_{qj}$ , the residual variation in  $\beta_{qj}$  is negligible (Raudenbush and Bryk 2002, p. 30).

These multilevel models are used estimate the impact of legal and extra-legal factors on the likelihood of a defendant: (1) being noticed for the death penalty (for both the murder conviction (MC) and death eligible (DE) subsamples; (2) pleading to LWOP after being noticed for the death penalty; and (3) receiving the death sentence at the penalty phase.<sup>464</sup> The result for each hypothesis articulated in Chapter Six is discussed, in turn, below.

<sup>&</sup>lt;sup>464</sup> The likelihood that a defendant accepts a plea bargain is also analyzed (for both the death eligible and death noticed subsamples. These models are not discussed in detail in the text, but are briefly mentioned

#### 8.3 Assessment of Hypotheses

In order to assess the appropriateness of employing the multilevel modeling framework, several null (i.e., intercept-only) models were estimated to determine whether there was significantly between-cluster variation in the outcome measures at the three decision stages (see top panel of Appendix C). These exploratory analyses strongly suggest that there is significant between-cluster variance to justify using mixed-effects models. With respect to prosecutorial discretion in seeking the death penalty, over 60 percent of the residual variation is between-cluster (67.9 percent and 62.9 percent for, respectively, the murder conviction and death eligible subsamples). For death-noticed cases that resulting in a plea bargain, 18.5 percent of the variability is between-cluster (Model 3). Finally, 19.2 percent of the variability on the likelihood of receiving a death sentence for case that advanced to the penalty phase is between-cluster (Model 4).

The models in the bottom panel of Appendix C account for the impact of three of the status dimensions mentioned above: vertical, normative, and horizontal. The correlation between the horizontal dimension and cultural dimension indices was extremely high (r = .92), so the cultural dimension was dropped from the analysis. As expected, the models reveal the inclusion of the judicial circuit-level predictors reduces total proportion of between-cluster residual variation (between five and eight percent reduction depending on the model), but significant the between-cluster residual variation remains.

during the discussion of model specification checks. The results from the plea bargaining models are presented in Appendix D.

The first three hypotheses rely on Black's theory to make predictions about the impact of race/ethnicity, age, and gender on capital charging-and-sentencing decisions. These hypotheses intentionally omit other important legal and extra-legal variables, only including controls for the number, race/ethnicity, age, and gender of defendants and victims the cases. More elaborate models that control for other relevant variables specified in the theory are examined later. Those fully specified models also employ special approaches that adjust for potential bias that may result from non-random selection due to death penalty noticing and plea-bargaining decisions (see also Section 7.2.3).

### 8.3.1 Hypothesis One

According to Hypothesis One ( $H_1$ ), black-offender/white-victim cases attract the most severe forms of law. Table 8 displays the effects of the racial organization of the case on the three decision points mentioned above. To make the interpretation of racial combination effects as intuitive as possible, cases are organized by their racial structure (i.e., race-of-defendant/race-of-victim) and compared to a reference category. Since the groupings use the same binary metric, both comparisons to the reference (omitted) category and the other groups are possible.<sup>465</sup> The three racial combination slopes estimated are black defendant/white victim (b/w), white defendant/white victim (w/w),

<sup>&</sup>lt;sup>465</sup> Parameterizing the racial groupings in this manner was done for other practical concerns as well. It is not uncommon for binary regression models to suffer from perfect prediction problems, resulting in numerical instability, when there are a limited number of unique covariate patterns (for a detailed discussion, see StataCorp 2003). Due to the small number of cases that were noticed for the death penalty during the period under investigation (381 cases), the covariate patterns are necessarily limited, particularly when controlling for numerous variables. When confronted with perfect prediction issues, the analyst typically must omit the variable(s) causing perfect prediction or reparameterize. The conventional approach to estimating defender/victim race interactions resulted in perfect prediction, so the current approach was adopted in order to properly estimate the models. These types of issues are not uncommon in death penalty research because of the infrequency in which the death penalty is used (Weiss *et al.* 1999).

and Hispanic defendant/white victim (h/w). The omitted category includes black defendant/black victim (b/b) and white defendant/black victim (w/b); however, because less than two percent of the cases involve a w/b combination, the reference group is essentially b/b.<sup>466</sup> It is also important to note the inferences about the effect of the h/w combination relative to other groups may be questionable because of the small number of Hispanic offenders in the data (less than three percent of offenders in the overall data and less than four percent in the death penalty subsample).

The coefficients presented in these models (logit coefficients or log odds) lack an intuitive interpretation. But these coefficients can be used to calculate the *percent change* in the odds of being noticed for death, receiving an LWOP plea, *et cetera*, with the formula:  $100 \times [(exp(\beta \times \delta) - 1];$  where *exp* is the antilog,  $\beta$  is the logit coefficient, and  $\delta$  is the value of the variable (Long 1997, p. 22). So, for example, based on this calculation, a b/w combination increased the odds of a case being noticed for death by 260 percent and 144 percent relative to a b/b combination, according to, respectively, Model 1 (MC) and Model 2 (DE).<sup>467</sup> Table 8 reveals that all three racial combinations had a greater likelihood of being charged as a death penalty case relative to the b/b combination. A comparison of slope coefficients for Model 1 reveals that b/w cases were most likely to be noticed for death, followed by h/w, and then w/w. These results are consistent with the prediction of  $H_1$ : low-status defendant/high-status victim cases. The

<sup>&</sup>lt;sup>466</sup> Again, I was necessary to parameterize this way in order to avoid numerical instability and nonsensical results. As a robustness check, the models were reanalyzed with the w/b category explicitly omitted and the results were nearly identical.

<sup>&</sup>lt;sup>467</sup> MC:  $100 \times (\exp(1.282) - 1) = 260.38$ ; DE:  $100 \times (\exp(.895) - 1) = 144.7$ .

coefficients from Model 2 imply a slightly different ordering (i.e., h/w, b/w, and w/w), but the overall implications are the same.

The same cannot be said, however, for other decision points in the capital charging-and-sentencing process. Model 3 suggests that b/w cases are more likely to result in a LWOP plea than b/b cases (84 percent increase in the odds), but h/w and w/w cases are less likely to result in a LWOP plea (32 and 16 percent decrease in the odds, respectively). With respect to Model 4, h/w and w/w cases are much more likely to result in a death sentence (360 and 50 percent increase in the odds, respectively), but there was, essentially, no difference between b/w and b/b cases (only a 5 percent decrease in the odds).

#### 8.3.2 Hypothesis Two

Hypothesis Two ( $H_2$ ) proposes that the relationship between age and the severity of law is curvilinear: cases involving very young and very old victims attract more law, independent of the race/ethnicity and gender. To assess functional form of the relationship, a quadratic term is included in the model. The "simple slope" (i.e., main effect) of victim age is interpreted as the marginal effect of the victim's age on the outcome variable (e.g., the likelihood of being noticed for the death penalty) when the victim's age equals "zero" (Aiken and West 1991, p. 73). As discussed earlier, the victim's age in centered at its mean, so the main effect is the impact of the victim's age when the victim's age is at its mean (which is 27.6 years for the uncentered victim age variable). The slope coefficient for the quadratic term is the change in the marginal effect of victim age for a one unit increase in the victim's age. For Models 1 and 2 in Table 8, the slope of main effect is negative and the slope of quadratic term is positive, meaning that the negative effect of age on the severity of law decreases (and ultimately changes direction) victim age increases. Model 3 (likelihood of a LWOP plea) suggests that the severity of law increases with the victim's age to a point, and then decreases, while Model 4 shows that the probability of receiving a death penalty decreases much more steeply as the victims get older.

Figure 3 visual depicts the nonlinear effect of victim-age on the severity of law. The graphs in the top panel present effect displays for the two death-notice models (Models 1 and 2) and the bottom panel presents the effect displays for the plea bargaining and death sentence models (respectively, Models 3 and 4). These figures suggest strong support for  $H_2$  for Models 1 and 2—cases with extremely young and extremely old victims have the greatest likelihood of being noticed for the death penalty. Models 3, however, does not support the predictions  $H_2$  and Model 4 only provides partial support (i.e., cases with extremely young victims have the highest probability of receiving the death sentence at the penalty phase, but cases with older victims have the lowest probability).

Recall that  $H_2$  also predicts that very youthful offenders will attract the least amount of law. Figure 4 presents the effect displays with respect to the relationship between defendant's age and severity of law. The main and quadratic effects are interpreted in the same fashion as previously discussed for the age-of-victim effects. The top panel, again, presents graphs for death noticing decisions, whereas the bottom panel presents effect displays for plea bargaining and death sentencing decisions. These figures reveal mixed support for  $H_2$ : the graphs in the top panel (i.e., likelihood of a death notice) directly contradict  $H_2$ , whereas the graphs in the bottom panel provide partial supportyouthful offenders attract less law than middle-aged offenders. The graph in the bottomright quadrant (i.e., likelihood of a death sentence at the penalty phase) reveals that cases involving extremely young *and* extremely old offenders attract the least law. In general, by the time offenders reach their late-30s (at the time of offense, not at trial), they are less likely to be noticed for the death penalty and receive the death sentence than youthful offenders. Only the model predicting pleading to LWOP provides the clearest support for  $H_2$ : youthful offenders have the lowest probability of pleading to LWOP and older offenders have the highest.

With respect to the interaction between offender and victim age,  $H_2$  predicts that cases with older defendants and youthful victims should attract the most law and cases with younger defendants and older victims should attract the least amount of law. The most straightforward to examine this relationship is to calculate the difference between age-of-defendant and age-of-victim for each case and include this difference as a regressor. If  $H_2$  is correct, then slope coefficient for this regressor should be positive because very small values (i.e., negative values) reflect a young defendant and a much older victim, whereas very large values indicated an older defendant and a much younger offender. The effect should also be nonlinear, so the slope actually increases as the disparity increases in the direction of older defendant/younger victim. The hypothesized positive and nonlinear relationship is captured with the inclusion of a quadratic term. Figure 5 presents the effect displays for Models 1, 2, 3, and 4. As with the previous nonlinear graphs, the top panel models death noticing decisions and the bottom panel models plea-bargaining and death-sentencing decisions. The two graphs on the top panel and the graph in bottom-right quadrant reveal mixed support for  $H_2$ : larger

defendant/victim age disparities increase the severity of law, irrespective of the direction of the disparity (i.e., regardless of whether the defendant is much older than the victim or the victim is much older than the defendant). The bottom-left panel directly contradicts  $H_2$ : the greater the age disparity, the less amount of law the case attracts. Only Model 4 (i.e., likelihood of a death sentence) provides unequivocal support for  $H_2$ .

## 8.3.3 Hypothesis Three

Hypothesis Three ( $H_3$ ) argues that male-defendant/female-victim cases will attract the most law and cases with female-defendants and male-victims will attract the least amount of law, holding the race/ethnicity and age of the parties constant. Similar to the models examining the impact of various defendant/victim racial combinations, cases are organized by their "gender structure" to assist in interpretation of gender combination effects. Effects are estimated for male-defendant/female-victim (m/f) and femaledefendant/male-victim (f/m), with male-defendant/male-victim (m/m) and femaledefendant/female-victim (f/f) serving as the reference category.<sup>468</sup> All four models reveal that m/f cases attract the most law (see Table 8). The percent *increase* in the odds of the outcome variable for Models 1, 2, 3, and 4 are, respectively, 173 percent, 161 percent, 69 percent, and 39 percent. Also consistent with  $H_3$ , f/m cases attracted the least amount of law, with the percent *decrease* in the odds of the outcome variable for Models 1, 2, 3, and 4 being 29 percent, 7 percent, 12 percent, and 58 percent, respectively.

As previously discussed, the variance components of the random-intercept model can be analyzed in order to determine how much of the total residual variance in the

<sup>&</sup>lt;sup>468</sup> Approximately three percent of the death eligible sample consisted of f/f cases, so the reference category is, essentially, m/m. As discussed early, it was necessary to parameterize the gender effects in this manner for numerical stability (see note 466).

outcome can be attributed to between-cluster variance—which, in our case represents judicial circuit (i.e., regional) effects. Table 8 reports the between-cluster residual variance ( $\rho$ ) for all of the models. This residual variance ranges from 63 percent (Model 1) to 12 percent (Model 3), indicating that there is significant between-circuit residual variation, after taking into the race/ethnicity, age, and gender of the defendants and victims.<sup>469</sup> It should be emphasized that  $\beta_{0j}$  represents the expected log odds of the outcome variable (e.g., being noticed for death) when the values of all predictor variables are zero (Luke 2004, p. 24). In the models previously estimated,  $\beta_{0j}$  is the expected log odds of the outcome for a case with the average number of defendants (1.5), average number of victims (1.2), a black male defendant of average age (approximately 27), and a black male victim of average age (approximately 36). The variance of  $\beta_{0j}$ ,  $\psi_{j}$ , is interpreted as the variability in the expected log odds for this type of case across judicial circuits.

## 8.3.4 Hypothesis Four

Recognizing that race/ethnicity, age, and gender are mere proxies for social status, Hypothesis Four ( $H_4$ ) predicts that other legally illegitimate/legally suspect factors, that are more proximal indicators of social status, will have an appreciable impact on legal decision-making *and* should attenuate the impact of race/ethnicity, age, and gender. As noted in Chapter Six, socioeconomic status and employment status are indicators of vertical status; marital status, number of children, and employment status are indicators of horizontal status; level of education is an indicator of cultural status; and prior drug

 $<sup>^{469}</sup>$  Raudenbush and Bryk (2002) note that a  $\rho$  higher than .1 indicates that multilevel modeling techniques may be appropriate.

use is an indicator of normative status. An ideal test of Black's theory would include variables for both the defendant and the victim, but current data limitations preclude such a test. Nonetheless, these measures, on balance, should influence legal decision-making at the various stages analyzed in this study.

Table 9 present the effects of these variables on death noticing, plea bargaining, and death sentencing. All of the models control for the effects previously analyzed for  $H_1$ ,  $H_2$ , and  $H_3$ . These results provide mixed support for Black's theory. In fact, only the effect of defendant's prior drug (normative status) and defendant being a stranger (horizontal status) is consistent with  $H_4$  across all four models. For example, consistent with the theory, being a high school graduate (cultural status) and employed (vertical status/horizontal status) decrease the likelihood that defendant receives an LWOP plea or is sentenced to death (Models 3 & 4), *but* increase the likelihood that a defendant is noticed for the death penalty (see Models 1 & 2). A defendant's socioeconomic status (vertical status) increases the likelihood of being noticed for the death penalty and is sentenced to death (inconsistent with  $H_4$ ), but decreases the likelihood of pleading to LWOP (versus LS) (consistent with  $H_4$ ). The effect of being married (horizontal status) is consistent  $H_4$  for all models except Model 3, while the effect of having children is contrary to the prediction of  $H_4$ , except for Model 3.

A possible explanation for these mixed results is that the models do not adequately take into account the social status of the victims. The previous models analyzing  $H_1$ ,  $H_2$ , and  $H_3$  were able to examine the *relative* characteristics of the parties involved—i.e., racial/ethnic, age, and gender *differences*. Black's theory places a premium on the *relative status* positions of the parties in a conflict, so the unsupportive

364

results mentioned above may result more from data limitations than with the actual theory itself; nevertheless, several of the predictions derived from Black's theory of law are supported by the data.

Recall that  $H_4$  also posits the models including the more proximal indicators of social status will better predict legal decision-making than models only including racial/ethnic, age, gender variables. Several measures are available for comparing the fit of competing models. Two statistics commonly used to assess model are the Akaike Information Criterion (AIC) and the Bayesian Information Criterion (BIC).<sup>470</sup> The AIC and BIC are defined as:

AIC = 
$$-2 \times L_1 + (2 \times k)$$
 [6],  
BIC =  $-2 \times L_1 + [k \times \log_e(N)]$  [7],

where  $L_1$  is the log-likelihood of the fitted model, k is the number of parameters estimated in the model (including the intercept) and N is the number of observations (Akaike 1973; Schwartz 1978a). The AIC favors model complexity, whereas the BIC favors model simplicity (Greene 2000). For the purposes of this study, the AIC statistic is preferred because it is important to take into many predictors in order to have a "well controlled" study (see Baldus *et al.* 1998); nonetheless, both AIC and BIC statistics are reported below the models in Tables 8 and 9. Comparison of the AIC statistics reveal that the three of the four models including the more proximal measures of social status—i.e.,

<sup>&</sup>lt;sup>470</sup> The likelihood ratio test (*LR*), commonly used to compare nested models, determines whether inclusion (or exclusion) of covariates improves the fit of the model:  $LR = -2 \times (L_1 - L_0)$ ; where  $L_0$  and  $L_1$  are the loglikelihood values association with the full and constrained models, respectively. The *LR* is approximately  $\chi^2$  distributed with  $d_0 - d_1$  degrees of freedom, where  $d_0$  and  $d_1$  are the model degrees of freedom associated with the full and constrained models, respectively (Gujarati 1988, p. 281). A *constrained* model is a model in which one or more estimated parameters is fixed to zero (i.e., excluded from the model). The constrained model, then, is the "simpler" model in that in includes less information than the unconstrained model.

socioeconomic status, marital status, *et cetera*—fit the data better than models without those variables. Model 3 is the exception, with the AIC favoring the model in Table 8.

An examination of *rho* ( $\rho$ ) also reveals that, even after taking into account several additional measures of social status, there remains significant between-cluster variation—ranging from 63 percent (Model 1) to 18 percent (Model 3).<sup>471</sup> As discussed earlier,  $\beta_{0j}$  must be interpreted as the expected log odds of the outcome for a case with the average number of defendants, average number of victims, a black male defendant of average age who did not graduate from high school, who was unemployed at the time of the offender, who is of lower socioeconomic status, has no children, does not have a history of drug abuse, and a black male victim of average age who knew the defendant.

# 8.3.5 Hypothesis Five

According to Hypothesis Five ( $H_5$ ), the legal jurisdiction where a case occurs influences legal decision-making in the capital charging-and-sentencing process, net of extra-legal case-level factors. The between-cluster residual variation, *rho* ( $\rho$ ), discussed earlier clearly indicates that a substantial portion of the variation in legal outcomes are attributable to judicial circuit-level influences (between 63 and 12 percent of the residual variance, depending on the model analyzed) (see Tables 8 and 9).  $H_5$  also predicts a jurisdiction's structural characteristics will influence the legal outcomes of cases involving similarly-situated defendants (see p. 293)—i.e., a judicial circuit's vertical, normative, horizontal, normative, and cultural status will exert influence on the administration of capital punishment, net of case-level characteristics. As discussed

<sup>&</sup>lt;sup>471</sup> The value of *rho* ( $\rho$ ) slightly increased for Model 3 (plea bargaining), suggesting that the betweencluster residual variance has increased with the addition of the more proximal social status variables in the model. This is not uncommon with multilevel models, and the increase suggests that one or more of the predictors included in the model has a differential effect across judicial circuits (i.e., causal heterogeneity).

earlier, the structural characteristics of the judicial circuits that were proxies for the various social dimensions were highly correlated, so the variables were combined into scales corresponding to the various dimensions of social space (see Section 8.1.2). The scales for the horizontal and cultural dimensions were highly (negatively) correlated (r = .92), so the cultural dimension variable was excluded from the estimated models.

Table 10 presents the results from the models estimating the impact of the structural factors on capital charging-and-sentencing decisions. A jurisdiction's vertical status has a negative impact on the likelihood that a case is noticed for the death penalty (see Models 1 and 2) (contrary to  $H_5$ ), but a positive impact on the likelihood of pleading to LWOP and receiving a death sentence (consistent with  $H_5$ ). According to Black, law is stronger where other social control is weaker, so the higher the jurisdiction's violent crime rate, the more law a case should attract net of case-level characteristics. The results from Table  $H_5$ , however, suggest the opposite: higher violent crime rates decrease the severity of law in three of the four models—only Model 3 (plea bargaining) is consistent with  $H_5$ . Also recall that  $H_5$  predicts the impact of a jurisdiction's horizontal status on the administration of capital punishment is curvilinear—increasing to a certain threshold and then decreasing thereafter. As stated before, to capture the nonlinear relationship, a higher-order polynomial (quadratic) is included in the models. The results in Table 10 provide support for this assertion: the slope coefficient of the quadratic term is negative across all four models, indicating that the negative effect horizontal status on severity law becomes "stronger" as horizontal status increases. Figure 6 presents the effect displays for the nonlinear relationship. These graphs reveal strong support for  $H_5$ : horizontal status increased the severity of law that a case attracts to a certain point, and

decreases thereafter. Jurisdictions that are more integrated, intimate, and interdependent attract the least amount of law, net of case-level extra-legal case-level characteristics.

Surprisingly, the inclusion of these judicial circuit-level variables fails to appreciably reduce the residual between-clustered variation. Moreover, a comparison of the contextual effects models to the previous models that did not include the contextual variables fail to substantial improve model fit according to AIC and BIC statistics. Granted, the AIC for the Model 4 (probability of receiving a death sentence) is slightly smaller than the similar model that does not include the circuit-level variables, but the improvement in model fit is minimal.

*Discussion*. Table 10, presenting results from the most complete specification and including all the case-level extra-legal and circuit-level predictors, provides mixed support for Black's theory of law. In fact, with a few exceptions, there is very little difference in model fit between specifications that include more control variables and those that do not. Important differences pertain to the effects of the racial structure of the case. After including additional covariates for the extra-legal variables and jurisdictional social structure (see  $H_4$  and  $H_5$ ), Black's theory is supported for all combinations except the b/w effect on death sentencing (Model 4). As noted earlier, these mixed findings may result from the fact that the victims' social status is not adequately taken into account in the analyses.

It must be emphasized that the above models likely suffer from misspecification because these models intentionally omit relevant legal variables. As noted earlier, the exclusion of these variables was done in order to determine the impact and predictive ability of all extra-legal variables. Since Black's theory also explains the *content* of legal rules, controlling for the legal variables would likely result in an attenuation of the effect of social geometry of the case because most legal rules, themselves, reflect the social organization of a case. These variables are included in the models presented below and diagnostic checks are performed to determine whether the inclusion of those variables distort the estimates of the extra-legal predictors (see Section 8.4).

Selection Bias. Another type of misspecification results from the fact that the plea-sentencing and trial-sentencing models analyze subsamples that are likely to result from two types of non-random selection. The first type of non-random selection occurs when a prosecutor decides to seek the death penalty against a defendant. The models presented earlier that analyze death penalty notice decisions strongly suggest that extralegal factors are predictive of these decisions. The potential problems that arise from this type of non-random selection were discussed in Section 7.2.3. Another form of nonrandom selection likely happens when a case advances to trial rather result in a plea bargain. Obviously a plea-bargain requires a "meeting of the minds" between the defendant and the prosecutor—one party must offer and the other party must accept (Baker and Mezitti 2001). Appendix D presents results from models predicting plea bargaining (see Models 2 and 3). Model 2 and Model 3 predict the likelihood of plea bargaining in, respectively, the "death eligible" and death noticed subsamples. These results suggest that extra-legal factors occurring at both the case- and circuit-level play an important role in reaching plea agreements. Moreover, the proportion of residual variation that is between-cluster is fairly large (.39 for Model 2 and .35 for Model 3).

As already explained (see Section 7.2.3), if unobservable factors influencing the decision to notice a case for death or a plea agreement are *also* relevant to the actual

sentencing decision, model misspecification will result and parameter estimates of the sentencing decision will be biased because they confound the substantive parameters with the parameters of the function determining the probability of inclusion into the subsample. In an attempt to correct for this possible bias in the plea-sentencing model (i.e., LWOP versus straight life plea), the joint probability of being noticed for the death penalty and entering a plea agreement was estimated via a bivariate probit and included as a regression in the multilevel analysis. A similar procedure was conducted for the trail-sentencing model (death sentence versus non-death sentence), in that the joint probability of being noticed for the death penalty and going to trial was estimated and included as a regressor in the multilevel analysis (see Johnson 2006). The death noticing and plea-bargaining models (i.e., the selection models) include all of the covariates in the sentencing models (i.e., substantive model), save the number of statutory aggravating circumstances sought by the prosecutor.<sup>472</sup> Inclusion of these probabilities estimated from the selection equation, in theory, should "absorb" the effect on any unobserved variables that are correlated with the outcome of interest and the other variables in the model.

It should be emphasized that the biased-corrected models are estimated only for the fully specified models including all case- and circuit-level extra-legal factors and legal variables. While it is possible to re-estimate the partial models  $(H_1 - H_5)$  with the sample-selection correction, the results from these models would be misleading because those models do not include the full set of covariates that appear in the death noticing and plea-bargaining models. The exclusion of these variables in the partial models alters

<sup>&</sup>lt;sup>472</sup> The number of statutory aggravators could not be included in the selection models because those variables are only observed for the cases actually noticed for death.

what the error term for each equation represents, and as a result, the multivariate probit is not properly adjusting for the shared unobservable characteristics relegated to the error terms (see, generally, Bollen 1989).

The estimation of the probability of selection in the subsample and inclusion of that probability in the sentencing models, generally referred to as the "two-step" approach, has been used to examine sentencing decisions in the multilevel context (Johnson 2006), but this procedure may be particularly problematic for the models analyzed in this study (see Bhattacharya, Goldman, and McCaffrey 2006). The procedure outlined above is employed when analyzing Hypotheses Six and Seven, and the robustness of these results is examined in Section 8.4.

### 8.3.6 Hypothesis Six

Legal variables are included in the models presented in Table 11. In particular, these models control for the presence of contemporaneous felonies, prior felony conviction, prior conviction for a violent felony, and for Models 3 and 4, the number of statutory aggravating circumstances charged by the prosecutor. It is also likely that *at least* two variables included in all of the partial models estimates—the number of defendants and victims—can also be considered legally relevant. The direction of the impact of number of victims is straightforward: cases with more victims should attract more law. The effect of the number of defendants, however, may be more dubious. One the one hand, a defendant and codefendant who *work with each other* should have more organizational status (capacity for collective action), and thereby attract less law, all else constant. On the other hand, defendants and codefendants who turn against one another (i.e., one or more of the defendants cooperate with the government) are likely to attract

more law. It is also likely that multiple defendant cases may be particularly aggravated (see note 393). Without more information about the interaction dynamics between codefendants in a particular case, it is impossible to accurately predict the direction of the effect.<sup>473</sup>

As the results in Table 11 reveal, the legal variables in the models sometimes have rather counterintuitive effects. Defendants with contemporaneous felony charges are *less* likely to be noticed for the death penalty (see Models 1 and 2), but more likely to plea to LWOP and be sentenced to death (Models 3 and 4). Defendants with prior felonies are *less* likely to be noticed for the death penalty, plea to LWOP, and be sentenced to death. Having a prior violent felony increases a defendant's likelihood of being noticed for the death penalty (Models 1 and 2) and pleading to LWOP (Model 3), but decreases the likelihood of a defendant being sentenced to death (Model 4).

The effect of the number of statutory aggravating circumstances charged against the defendant for plea-bargaining decisions and death sentencing decisions are presented in Models 3 and 4. In the plea bargaining model (Model 3), the number of statutory aggravators has the expected effect—it increases the likelihood of pleading to LWOP ( $\beta$ = 0.594). The same variable has the opposite effect in penalty phase sentencing model, though the effect size is very small ( $\beta$  = -0.098). There is a 81 percent *increase* in the odds of receiving an LWOP plea for an additional statutory aggravator. Is important to keep in mind, however, the effect a change in odds is multiplicative, not additive. As a result, there is a 494 percent *increase* in the odds of receiving an LWOP plea is a

<sup>&</sup>lt;sup>473</sup> The number of defendants in a case has a positive effect on the likelihood of being noticed for the death, but a negative effect on pleading to LWOP and being sentenced to death. This suggests a "collective action" effect because, typically, the later decision-stages (Models 3 and 4) permit the co-defendants to work together, whereas the death noticing decisions usually involve only the prosecutor. As expected, the number of victims has a positive effect on the severity of law across all decision stages.

defendant is charged with three statutory aggravators  $(100 \times (\exp(.594 \times 3) - 1) = 494.2)$ . With respect to death sentencing decisions, a one unit increase the in number of statutory aggravators results in a nine percent *decrease* in the odds of being sentenced to death, and a three unit increase results in a 25.5 percent decrease.<sup>474</sup>

As hypothesized, the inclusion of the legal variables has very little impact on the effect of race/ethnicity and gender across all four models (compare Tables 10 and 11). The change in the direction of the effect for the b/w racial combination variable results from the sample bias correction included in Model 4 in Table 11. Similarly, the legal variables do not mitigate the impact of several of the other extra-legal variables (e.g., marital, educational, socioeconomic status). Moreover, the inclusion of the legal variables does not improve model fit, as seen by comparing AIC and BIC statistics between the models Tables 10 and 11.

## 8.3.7 Hypothesis Seven

Table 11 shows the effects the various dimensions of social space on the capital charging-and-sentencing process, net of all case-level legal and extra-legal characteristics. As predicted, a jurisdiction's vertical status is positively related to the severity of law in three of the four models. The effect of a jurisdiction's normative status, again, runs counter to Black's theory (except for Model 3), while a jurisdiction's horizontal status is consistent with the theory (see Figure 6). These models clearly demonstrate that the social dimensions identified by Black's theory are important when studying the legal behavior. Also recall that the variables constituting the vertical, normative, and horizontal dimensions of a jurisdiction were standardized prior to

 $<sup>^{474}</sup>$  100 × (exp(-.098 × 1) - 1) = 9.33; 100 × (exp(-.098 × 3) - 1) = 25.47.

constructing the scale, so the scales have a mean of zero and approximately a standard deviation of one (see Section 8.1.2).<sup>475</sup> Due to this standardization, it is possible to *roughly* compare the coefficients of the structural effects with the coefficients of the legal variables. As Table 11 reveals, when you compare within models, several of the social dimensions have a stronger effect (in absolute value) on legal decision-making than the legal variables.

The proportion of the residual variance that is between-cluster remains rather high across the four models (ranging from .57 [Model 1] to .324 [Model 2]), suggesting that the random-intercept model is appropriate. Similar to the discussion of  $H_5$  (see Section 8.3.5), the inclusion of the circuit-level predictors slightly increases the between-cluster residual variance, suggesting that the effect of one or more the legal variables in the model varies across jurisdictions.

## 8.3.8 Hypothesis Eight

Black's theory proposes that the impact of case-level legal and extra-legal factors on the administration of capital punishment will vary across jurisdictions and the variation in these effects can be explained by the social location (i.e., social status) of the jurisdictions (see Section 6.3.8). Unfortunately, due to the very small sample sizes associated with plea bargaining and death sentencing decisions for death-noticed defendants, it is infeasible to estimate multiple random effects for Models 3 and 4. Instead, attention is focused on the decision to seek the death penalty for defendants who qualify as death-eligible according to the B1 or B2 statutory aggravators. While this is categorization is under-inclusive, as discussed earlier, it appears consistent with the

<sup>&</sup>lt;sup>475</sup> The standard deviations for the vertical, normative, and horizontal status scales are, respectively, .862, .974, and .91.

proportion of death-eligible case that studies in other jurisdictions have discovered (see 8.1.3).

Black's theory implies numerous cross-level interactions, but this study limits its focus to cross-level interactions with the racial/ethnic and sex/gender combination variables, as well as legally legitimate factors. This decision was made for both practical and substantive concerns. On the practical side, estimating more than one or two random effects when average cluster-size is small often results in numerical instability and the estimation algorithm typically fails to converge, so the random-slopes typically need to be estimated one at a time (Raudenbush and Bryk 2002). Examining a model for each potential cross-level interaction would require estimating 19 different models and potentially 57 cross-level interactions.

On the substantive side, the bulk of the discussion of the impact of extra-legal factors on legal decision-making—particularly as it pertains to proportionality (or lack thereof)—presented in prior chapters has focused on racial/ethnic, gender, and regional differences in the administration of capital punishment. While the exploration of cross-level interactions for other extra-legal (or legally suspect) variables included in the model is important, especially as it pertains to the development and refinement of Black's theory of law, such analyses are not central to the present aims of this study.

Table 12 reports the results from the cross-level interaction models. Again, the intercept in the model represents the expected log odds of being noticed for the death penalty when all covariates are at zero (see discussion in Section 8.3.4). The "main effect" for the b/w racial combination variable represents the marginal effect in a jurisdiction that is average in vertical, normative, and horizontal status, and reveals that

cases with a b/w racial combination increases the odds of being noticed for the death penalty by 102 percent  $(100 \times (\exp(.704) - 1) = 102.2)$ . A jurisdiction's normative and horizontal status conditions the "racial" effect in the hypothesized direction: when the level of social control decreases, the odds of a b/w case being noticed for death increases by 206 percent  $(100 \times (\exp(.704 + .416) - 1) = 206.5)$ . A jurisdiction's horizontal status attenuates the impact of the b/w combination: it only results in a 29 percent increase in the odds of being noticed for death when a circuit's horizontal status increases  $(100 \times (\exp(.704 - .444) - 1) = 29.7)$ . While the impact of a jurisdiction's vertical status on b/w cases is also in the predicted direction, the effect is extremely small, only increasing the odds that the b/w case is noticed for the death penalty by five additional percent  $(100 \times (\exp(.704 + .025) - 1) = 107.3)$ .

Model 2 presents the results from the cross-level interaction between a circuit's social status characteristics and the w/w racial combination. In jurisdictions with average vertical, normative, and horizontal status, w/w cases increase the odds of being noticed for the death penalty by 98 percent  $(100 \times (\exp(.686) - 1) = 98.6)$ . Consistent with Black's theory, normative status attenuates the effect;<sup>476</sup> however, contrary to Black's theory, horizontal status slightly accentuates and vertical status attenuates the effect.

Model 3, examines cross-level interactions between the m/f gender combination and a jurisdiction's social dimensions. In a jurisdiction with average levels of vertical, normative, and horizontal status, a m/f gender combination increases the odds that a defendant will be noticed for death by 195 percent  $(100 \times (\exp(1.082) - 1) = 245.2)$ . The gender disparity decreases as the vertical status of the jurisdiction increases, and

<sup>&</sup>lt;sup>476</sup> Recall that the normative status variable is a measure of the violent crime rate in a jurisdiction, so it represents a "lack" of social control.

increases as normative and horizontal status increases—all of these effects conditional effects conflict with Black's theory.

Models 4 and 5 show the conditional effects of prior felony and prior violent felony convictions on the likelihood of being noticed for the death penalty. The likelihood of a defendant with a prior felony conviction being noticed for the death penalty increases as the judicial circuit's normative and horizontal status increases, but decreases as vertical status increases (Model 4). These effects contradict Black's theory. A prior violent felony increases the odds of being noticed for the death penalty by 155 percent  $(100 \times (\exp(.939) - 1) = 155.7)$  in a jurisdiction with average vertical, normative, and horizontal status (Model 5). The conditional effects a prior violent felony conviction all support Black's theory of law: a unit increase in a jurisdiction's vertical status increases the effect of a prior violent conviction on the odds of being noticed for the death penalty by 322 percent  $(100 \times (\exp(.939 + .501) - 1) = 322.1)$ . Also, as aggregate normative status decreases, the effect of a prior violence conviction on the odds of a death notice increases by 281 percent  $(100 \times (\exp(.939 + .399) - 1) = 281.1)$ . Consistent with Black's theory, jurisdictional horizontal status decreases the effect of a prior violent felony on death noticing as well.

### 8.4 SPECIFICATION CHECKS

Multilevel models rest on several assumptions pertaining to both the structural (fixed) and random portions of the model (Raudenbush and Bryk 2002, pp. 255–56). Similar to traditional (i.e., pooled) regression models, the estimated models assume that case- and judicial-level predictors are uncorrelated with random effects within and between levels. Under those assumptions, the parameter estimates are unbiased. The accuracy of hypothesis testing (i.e., standard errors of parameter estimates and confidence intervals) depends on the tenability of assumptions with respect to the random portion of the model. As discussed earlier (see Sections 7.2.3 and 8.1.2), statistical inference based on the sampling distribution of the parameter estimates does not apply because these data analyzed consist of the entire population of cases. Moreover, no attempt is made to make inferences about the capital charging-and-sentencing process outside of Georgia and the timeframe explored in the study (see note 454).

Misspecification of the structural part of the model remains problematic because any omitted variable at the case-level that is related to both the outcome variable and one or more of the case-level covariates will lead to biased parameters. If that biased parameter estimate is modeled at the judicial circuit-level, that circuit-level model will also be biased. With respect to cross-level interactions, omitted-variable bias confounds inferences when three conditions hold: (1) the omitted variable is related to the outcome variable *after* controlling for other covariates in the model; (2) the omitted variable must be related to a case-level covariate in the model; and (3) the association between the omitted variable and the case-level covariate must vary from unit to unit and the strength of that association must be related to the circuit-level predictor (Raudenbush and Bryk 2002, p. 261).

As series of diagnostic checks were performed on the structural portion of the fully specified case-level models (see Appendix E). Several goodness-of-fit measures strongly suggest that the models adequately fit the data (Long 1997, Chapter 4). For example, the Hosmer-Lemeshow  $\chi^2$  goodness-of-fit statistic for the models rejects the hypothesis that the models are misspecified (for an explanation of the statistic, see Cameron and Trivedi 2009, pp. 457–58). The proportion of correctly classified responses ranged from .73 to .84—a figure consistent with previous death penalty research that included many more covariates (Unah and Boger 2001). Similarly, receiver operating characteristic (ROC) analyses also indicate that the models fit the data reasonable well (Cleves 2002). The area under a ROC curve is a measure of a model's predictive power, ranging from .5 (no predictive power) to 1 (perfect model fit). The area under the ROC curve for the fully specified models in this study ranged from .77 to .84. Finally, the pseudo- $R^2$  statistics reported in Appendix E provide another measure of model fit. The pseudo- $R^2$  statistic ranges from .27 to .41, indicating fairly good model fit considering that the pseudo- $R^2$  underestimates model fit relative to the  $R^2$  statistic in linear regression models (Long 1997).

Raudenbush and Bryk (2002) note that the analysis of model residuals plays an important role in assessing model specification at each level of a hierarchical model. Unfortunately, residual analysis from generalized linear models (e.g., logistic regression) is not as helpful as in the linear regression context because the residuals are non-normal and heteroskedastic. As a result, the numeric and graphical approaches to assessing model misspecification may not be particularly helpful when attempting exploring the possibility of omitted variable bias (Cohen *et al.* 2003, p. 514).

Model diagnostics of the level-2 equations also suggest that the equations do not suffer from any serious misspecification. The circuit-level equations predicting both the random intercept and the random slope include identical predictors, therefore any bias that would result from misspecification in one of equations is avoided (Raudenbush and Bryk 2002, p. 272–73). Since the data are unbalanced (i.e., cluster sizes vary), the

estimates of the random intercept and slope models are only asymptotically independent, and therefore only asymptotically unbiased. Graphical plots of the residuals examining the distance between the predicted and observed residuals for each group suggest that the random effects were approximately normally distributed (Raudenbush and Bryk 2002, p. 274).<sup>477</sup> As noted above, the departures from normality are not of primary concern because the data represent the entire population of cases.

*Selection Bias Specification.* As noted above, the previously estimated models take into account non-random selection by estimating the joint probability of being noticed for the death sentence and accepting/rejecting a plea-bargain and including those predicted probabilities as a regressor in the plea-sentencing and trial-sentencing models—the "two-step approach" (see Table 11: Models 3 and 4). Recent evidence suggests that a multivariate probit approach to correct for possible selection bias is superior to the two-step approach when (1) the outcome variable in the substantive equation is binary; (2) there are multiple processes influencing selection; and (3) the probability of the dependent variable is extremely low or extremely high (i.e., close to zero or one) (see Bhattacharya *et al.* 2006).<sup>478</sup> All three conditions are applicable to the current analysis because of the necessity to consider both the death-noticing and plea-bargaining decisions and the extremely small proportion of cases that are noticed for death and result in a death sentence.

<sup>&</sup>lt;sup>477</sup> Some of these analyses revealed that certain clusters were outliers (typically one to three at the most). Models were reestimated without these clusters and produced similar results, so the results including all clusters are presented.

<sup>&</sup>lt;sup>478</sup> It has been demonstrated that the two-step approach—estimating the probability of differential selection into the subsample and it including this probability as a regressor in the substantive equation (i.e., the sentencing equation)—is inconsistent because the second-stage probit equation is maximizing a misspecified likelihood (see Bhattacharya *et al.* 2006, p. 412).

In brief, these models take into account any shared unobserved factors that simultaneous influence all three decisions—that is, death charging, plea bargaining, and sentencing—by jointly estimating the correlation between the error terms across the models (see Greene 2000, p. 849). In principle, these models can be estimated within a multilevel modeling framework (see Grilli and Rampichini 2007; Rabe-Hesketh, Skrondal, and Pickles 2002a); in practice, however, this proved infeasible due to computational complexity and convergence problems when sample sizes are small because additional fixed-parameters and random effects (at both levels) must be estimated for the additional selection equations (Grilli and Rampichini 2007). As an alternative, the multivariate probit estimator is employed, via simulated maximum likelihood, without a multilevel specification (for a discussion of the algorithm, see Cappellari and Jenkins 2003).<sup>479</sup>

The results from these analyses are presented in Table 13, with the initial probit coefficients re-scaled to logit coefficients so coefficients across the models are directly comparable (Liao 1994, p. 25).<sup>480</sup> The *r* statistics at the very bottom of Table 13 for each model indicate the correlations of the error terms across the three equations for the pleabargaining and death-sentencing models.<sup>481</sup> Equations 1, 2, and 3 refer to, respectively, the sentencing decision, the death noticing decision, and plea bargaining decision. So  $r_{12}$ 

<sup>&</sup>lt;sup>479</sup> The simulated maximum likelihood estimator is sensitive to the number of random draws from the upper-truncated normal distribution, so setting the number of draws to the square root of the sample size is advisable. In the above analyses, the number of draws was set slightly higher than the square root of the sample size.

<sup>&</sup>lt;sup>480</sup> The parameter estimates of the sentencing models are of primary importance, so those results are presented in Table 13. Results for the equations predicting death penalty noticing and plea-bargaining decisions are omitted from Table 13, but can be found at Table 11 (Model 2) and Appendix D (Models 2 and 3).

<sup>&</sup>lt;sup>481</sup> Technically *r* should be represented as the Greek letter *rho* ( $\rho$ ), denoting a population statistic; however the sample statistic notation, *r*, is used to avoid confusion with previous definition of *rho* given in the multilevel context.

is the correlation between unobserved factors affecting both the sentencing and death noticing decisions,  $r_{13}$  is correlation between the unobserved factors at the sentencing and the plea-bargaining decisions, and  $r_{23}$  is the correlation between the unobserved factors influencing the death noticing and the plea bargaining decisions (Cappellari and Jenkins 2003). It is clear from Table 13 that unobserved factors the death noticing and pleabargaining decisions are also impacting the sentencing decisions.

The results of the multivariate probit estimator are similar to the results from the two-step with respect to the direction of the effects, although the magnitudes of the effects changed substantially for several variables. These models also suggest that, to a significantly degree, the results from the previous two-step specifications adequately model the direction of the effects, which are of primary importance in assessing whether the relationships predicted by Black's theory are observed in the data.

*Endogeneity*. There may also be a potential endogeneity problem with the models that include legal variables because, as previously stated, Black's theory also predicts that the *content* of legal rules are themselves explained by his theory (see Section 6.3). The estimated effect of a regressor on an outcome is inconsistent when that regressor is determined simultaneously with that outcome (Greene 2000). Recall that legal rules often reflect the social geometry of the case, and not the actual conduct of the defendant or the harm to the victim—this is particularly true for capital statutes. Specification tests were performed to assess any potential endogeneity bias that may result from inclusion of the legal variables. In particular, unconstrained models including the legal variables were compared with constrained models that excluded the legal variables in order to determine whether there were any systematic differences between the coefficients (Cameron and

Trivedi 2009, pp. 182–83).<sup>482</sup> This specification test, commonly referred to as a "Hausman specification test" (see Hausman 1978), follows a  $\chi^2$  distribution and is calculated as:

$$H = (b - B)'(V(b) - V(B))^{-1}(b - B), [8]$$

where *b* is the coefficient vector for the unconstrained model, *B* is the coefficient vector for the constrained model, V(b) is the covariance matrix for the unconstrained model, and V(B) is the covariance matrix for the constrained model. The estimate of (V(b) - V(B))may not be well-defined, particularly in small samples (Schreiber 2008), so a generalization of the Hausman test that is always admissible is preferable:

$$H = (b - B)'(V(b) - \operatorname{cov}(b, B) - \operatorname{cov}(B, b) + V(B))^{-1}(b - B), \quad [9]$$

where *b*, *B*, V(*b*), V(*B*) are defined as above, and cov(*b*, *B*) and cov(*B*,*b*) are the covariance matrices of the shared coefficients between the constrained and unconstrained models (Weesie 1999).<sup>483</sup> The null hypothesis is that the coefficient vectors systematically differ, so failure to reject the null hypothesis suggests the estimates are not biased by inclusion of the legal variables. These tests indicate that endogeneity bias is not present in the models predicting: (1) death noticing for the murder conviction sample ( $\chi^2 = 22.97$ , df = 29, p = .78); (2) death noticing for the death eligible sample ( $\chi^2 = 24.37$ , df = 29, p = .71); (3) pleading to LWOP for cases noticed for death ( $\chi^2 = 7.87$ , df = 30, p

<sup>&</sup>lt;sup>482</sup> "The Hausman test is best interpreted not as a test for the endogeneity or exogeneity of regressors per se but rather a test of the consequences of using different estimation methods on the same equation" (Baum 2006, p. 212).

<sup>&</sup>lt;sup>483</sup> The traditional Hausman test, unlike other tests, relies on asymptotic arguments not only for its distribution, but also for its ability to be computed. Weesie's (1999) generalization of the statistic relaxes the assumption that the covariance matrices of the two estimators are uncorrelated, thereby allowing the test to always be defined.

= 1.00); or (4) cases resulting in a death sentence that advanced to the penalty phase ( $\chi^2$  = 18.29, df = 30, p = .95).<sup>484</sup>

# 8.5 SUMMARY

The empirical analyses discussed in the preceding section provide mixed support for Black's theory of law. Many of the hypotheses derived from Black's theory of law and tested against the data receive moderate to strong support, while others were directly contradicted by the data (see Table 14 for a summary of results). The numerous models (and alternate specifications) clearly reveal that extra-legal factors remain important determinates of legal decision-making, but always in the direction predicted by Black's theory. The following chapter (Chapter Nine) describes some these results in greater detail and presents possible explanations for the mixed-results. Perhaps more importantly, the random-intercept and random-slope models strongly suggest that social contexts in which these cases occur influence the capital charging-and-sentencing process net of case-level characteristics. Although many of the direct effects on the structural characteristics of the jurisdiction on the various stages of legal decision making were weak to modest, they remained robust to the inclusion of many important case-level factors. The cross-level interactions were also very illuminating, particularly considering the small cluster sizes and restricted distributions (i.e., skewness) of most of the variables under study. At minimum, the analysis of the impact of the contextual factors explored in this study highlight the importance of the interplay between individual behavior and social structure.

<sup>&</sup>lt;sup>484</sup> The endogeneity tests compare models with and without the legal variables, collectively. Additional analyses were performed comparing models including and excluding individual legal variables and the results were similar.
In addition to underscoring the importance of micro- and macro-level dynamics in legal decision-making, the models also clearly demonstrate the importance of taking into consideration the impact of non-random selection into subsequent stages of the capital charging-and-sentencing process (see Unah and Boger 2002). The correlations between unobservable factors influencing the death noticing, plea-bargaining, and sentencing decisions (see Table 13) suggest that prior studies ignoring this differential selection have likely reported biased parameters estimates and drawn incorrect inferences from their regression analyses. As noted earlier (see Section 7.2.3), only one other study has attempted to account for non-random selection based on death-noticing and plea bargaining decisions. The study, examining the impact of both case- and county-level factors on the death penalty charging-and-sentencing process in North Carolina, used a bivariate probit estimator to correct for possible sample selection bias (Unah and Boger 2002); however, there are two problematic aspects of their model specifications that may undermine the validity of their results. First, the study did not explicitly model the hierarchical structure of the data—all intercepts and slopes were modeled as fixedeffects. A cross-level interaction between the prosecutor's political party affiliation and the percentage of the county's population that was non-white was also examined, but this interaction was modeled as non-randomly varying (see Raudenbush and Bryk 2002, p. 30).<sup>485</sup> Due to their estimation strategy, Unah and Boger's analyses fail to provide valuable information concerning possible between-cluster variation in their outcome measures or slope estimates.

<sup>&</sup>lt;sup>485</sup> The authors estimated robust standard errors to take into account clustering at the county-level (White 1980).

The second problem with Unah and Boger's analysis is that they incorrectly attempt to model a complex multivariate process as a simple bivariate process. The authors attempt to model *four* different selection processes—plea agreement, death penalty noticing, conviction, and death sentencing—within a bivariate probit framework. This is accomplished by using non-randomly selected samples in the *selection equations* to model non-random selection in outcome equations (Unah and Boger 2002, p. 16); however, the entire model is misspecified when the selection equation is misspecified (Manning, Duan, and Rogers 1987). The multivariate probit selection models employed in the current study properly estimate the interrelationships between the three selection processes and avoid the misspecifications of Unah and Boger's models.<sup>486</sup>

<sup>&</sup>lt;sup>486</sup> It is also puzzling that the authors report a regression coefficient for the selectivity index (i.e., the regressor obtained from the selection equation) from a probit sample selection model (see Unah and Boger 2002, Tables 2 and 4). The maximum-likelihood probit models with sample selection that the authors report estimating in *Stata 6* ("heckprob") does not provide that parameter. *Stata 6* estimates the correlation between the error terms of the selection and substantive equations (technically, *Stata* estimates the hyperbolic arctangent of the correlation for numerical stability, and then transforms this statistic to the correlation) (Baum 2009, p. 267). This statistic cannot be interpreted as a regression coefficient, as such, but as the relationship between the shared unobserved variables impacting the two decision processes (see Section 8.4). Unah and Boger "transform" the correlation of the effect of the index. This procedure cannot produce a coefficient that lends itself to any meaningful interpretation. The correlation is not a probit coefficient, and therefore cannot be re-scaled to logit coefficient. This interpretive *faux pas*, along with mispecifications of the selection equations, suggests the authors misunderstand the underlying statistical theory motivating the use of selection models.

## **Chapter 9: Summary and Discussion**

Science advances through the reasoned criticism of received knowledge (Sampson 2006, pp. 149–50). The aim of this project was to use improved theory, data, and methods to inform both the scholarly literature and the policy debate concerning the capital charging-and-sentencing process. Donald Black's theory of law was applied to the death penalty context to predict and explain legal decision-making *purely at the* sociological level—a form of theorizing in the tradition of early Marxian (Marx [1857] 1973), Durkheimian ([1897] 1994), and Simmelian (1909) social thought. Jonathan Turner (2002) has argued that the power of highly general and abstract theories is realized when analysts working in specialized fields derive hypotheses from the abstract theory and carefully link abstract propositions to concrete phenomena. Since the death penalty is the most severe form of punishment that can be applied by the government, the capital punishment process provides an ideal test of Black's theory.<sup>487</sup> The greater amount of potential law that may be applied to a case, the more social information about those in the conflict becomes involved (or gets collected) and the greater the potential for more discrimination (Black 1989, pp. 66–68).

Many socio-legal scholars refuse to accept that it is possible to predict and explain legal behavior without reference to psychology (e.g., Frankford 1995). Indeed, Kuhn (1970, p. 148) noted, "[n]either side [of competing research strategies] will grant all the non-empirical assumptions that the other needs in order to make its case." Nonetheless, Black's commitment to explaining human behavior without regard to human subjectivity

<sup>&</sup>lt;sup>487</sup> Analysts have explicitly employed Black's theory of law to examine variation in the commission of (and response to) lethal violence, including *attitudes* about capital punishment (Borg 1998; Kan and Phillips 2003), however, this study is the first to use Black's theory examine *legal decision-making* in the capital charging-and-sentencing process.

not only permits his theoretical propositions to be maximally general and parsimonious, but also firmly establishes sociology as a branch of social science unique from psychology-complete with its own concepts, imagery, and framework of analysis. A century ago Simmel ([1908] 1950, p. 21) stated "the [essential] task of the science of society" is to use "geometric abstraction" to investigate human interaction. And for quite some time, sociologists working outside of Black's theoretical tradition have also advocated a science of human behavior based on geometric principles of human interaction and without reference to human subjectivity (see, e.g., Mayhew 1980). Klüver and Schmidt (1999, pp. 312, 322) have explained that "[i]t is possible to define several main concepts of theoretical sociology in geometrical terms and to make geometrical models of social actors and interactions...[and it] is neither necessary to speculate about the influence of human nature upon history...nor to introduce particular 'interests' of social groups[.]"488 Geometric models of social interaction permit and facilitate the development and analysis of governing principles of human behavior: "[s]ocial systems are to be understood as sets of social actors whose interactions are determined by specific rules; these rules generate the dynamics of social systems" (p. 313).

Black's theory focuses on the observable dispositions of legal agents, rather than legal rules (cf. Llewellyn [1930] 1978); nevertheless, this project has demonstrated that the actual content of death penalty statutes can be predicted and explained by Black's theory of law. Capital statutes require the collection and consideration of social information about parties that reflects the social geometry of the case. Much of this

<sup>&</sup>lt;sup>488</sup> Klüver and Schmidt (1999, p. 311) identify three dimensions of social differentiation: segmentary ("us" vs. "them"), stratificatory ("above" vs. "below"), and functional ("action role" vs. "client role").

information has very little, if anything, to do with the conduct of the defendant or the actual harm to the victim, and much more to do with status positions of the parties.<sup>489</sup> To a non-trivial extent, then, this project highlighted the ability of Black's theory to contribute to the foundation of a theory of formal legal rules—a worthwhile endeavor that legal sociologists have largely abandoned (Cooney 1993). As noted earlier, Black's theory explains why white-collar criminals are treated more leniently *as a matter of law*, net of the actual harm done by the offender (see, e.g., Shapiro 1990). Black's theory also explains other aspects of legal decision-making with respect to the death penalty *apart* from the behavior of prosecutors, judges, and jurors. For example, Black's theory has successfully predicted how the larger social context influences: (1) the level of support for capital punishment; (2) the likelihood and rapidity of reinstatement of the death penalty following *Furman*; (3) the rate of death sentence reversals on appeal; and (4) the likelihood and frequency of executions and commutations.

Empirical studies of legal decision-making have consistently discovered that legally legitimate case characteristics account for only a modest proportion of the variation in criminal sentencing, but analysts have failed to adequately explain why and under what conditions do legal rules matter. Black's theory provides an answer to both of these questions. According to the theory, legitimate legal characteristics matter when they reflect the social structure of the case *and* will have the greatest explanatory power when cases are structurally similar or when the amount of law applied to the case is trivial.

<sup>&</sup>lt;sup>489</sup> In fact, capital defense attorneys who do not collect a wealth of information about their client(s) to present as mitigation evidence during the penalty phase of a death penalty trial may be deemed as providing ineffective assistance to their client(s) (see, generally, *Strickland v. Washington* (1984)).

Comprehensive sociological analysis and explanation requires a theory to be subject to both *epistemological* and *empirical* scrutiny (see Jasso 2006). Chapter Six argued that Black's overall explanatory framework, pure sociology, as well as his specific theory of law, were superior to the theoretical perspectives most frequently employed in criminal sentencing research—i.e., formal legal/jurisprudential, conflict, and symbolic interactionist theories. Specifically, it was demonstrated that Black's theory of law was more general, parsimonious, testable, and novel than those rival explanations (compare Sections 6.1.3 and 6.2.5). By extending the theoretical analysis well beyond Black's own epistemological defense of his work (see Black 1995, 2000), and reconceptualizing the sociological salience of race/ethnicity, gender, and geography in social life in general, and in conflict situations in particular, it was shown that Black's theoretical paradigm was able to offer a more complete understanding of these phenomena, and their interaction, with respect to legal behavior, while remaining completely consistent with the structure of his explanatory framework (see Section 6.2.4).

With respect to race/ethnicity, it was demonstrated that blacks, in the aggregate, lag far behind whites in almost every category of economic well-being (vertical status), even among the college-educated and upper-class (Conley 2001; Grodsky and Pager 2001; Oliver and Shapiro 1995), and poverty within predominately black communities has been remarkably persistent although inner-city poverty has undergone dramatic changes over the past three decades (DiPrete and Eirich 2006; Massey 2004; Sampson and Morenoff 2006). It was also shown that blacks have significantly less radial, relational, and functional status than whites (horizontal status), resulting from residential, professional, and interactional segregation (see, e.g., Bonilla-Silva 1997; Massey and Denton 1989). The legacy of residential instability, extreme high community disorganization, racially motivated vote suppression and felony disenfranchisement laws, and significant underrepresentation at the highest levels of government and the judiciary has also seriously hampered blacks' capacity for collective action (corporate status) (Gryski, Zuk, and Barrow 1994; Sampson and Groves 1989; South and Crowder 1997; Uggen and Manza 2002). Finally it was demonstrated that blacks, in the aggregate, have less cultural status (conventionality) and normative status (authority and respectability) due to their: use of "non-standard" names and "black-accented" English; lower levels of formal education; and higher levels of documented criminal histories (independent of aggregate levels of actual criminal involvement) (Austin and Allen 2000; Blumstein 1993; Fryer and Levitt 2004; Massey and Lundy 2001).

Black's theory was also applied to the interactive effects of race/ethnicity, gender, and region. Unlike symbolic interactionist perspectives, Black's theory does rest on the questionable assumptions about how legal decision-makers perceive these categories and how those perceptions vary across different contexts (see Section 6.2.3); rather Black's theory focuses on the status locations those characteristics indicate. According to Black, these characteristics will not uniformly advantage or disadvantage a criminal defendant in a case, but legal decision-makers are likely to impute the aggregate characteristics of the group when the specific information about the defendant is unknown (see, e.g., Kan and Phillips 2003; Stanko 1981/1982).

Similarly, Black's focus on both the micro- and macro-dimensions of human interaction underscores the fact that factors such as race/ethnicity and gender will have different effects depending on the larger social context. While other theorists have made contradictory predictions (and produced contradictory results) concerning the direct impact of structural features on legal decision-making, as well as the indirect effect of these structural factors on legal decision-making through their interaction with race/ethnicity, based on their differing assumptions about how these factors influence the motivations of actors, Black's theory simply focuses on how social contexts differ along the five social dimensions. For example, with respect to the "urban versus rural" distinction frequently examined in the criminal sentencing literature, it was argued that Black's theory is capable of explaining the contradictory results by simultaneously examining multiple dimensions of social organization. Blacks in rural areas typically have less vertical and cultural status, but more relational and functional status (i.e., horizontal status) (see Section 6.2.5). By highlighting the importance of considering all of the various social dimensions, Black's theory provides a more complete description of the social geometry of a case, as well as a more nuanced account of the relational dynamics potentially at play: "[t]he number of dimensions of a 'space of experiences' is the number of independent options one has to take into consideration for a complete description of any experience" (Klüver and Schmidt 1999, p. 311). As King et al. (1994, p. 34) note, "[D]escription has a central role in all explanation, and is fundamentally important in and of itself."

*Empirical Results*. Of course, the litmus test of any theory is how well it holds up to empirical scrutiny. The focus on empirical validity, however, does not minimize the importance of theoretical analysis; rather it assists in restricting or expanding theoretical concepts and propositions through the interplay of theory and research (Merton 1948)—the structure of the theories derived from a general explanatory framework is of primary

importance. All of the alternative theoretical perspectives previously described are informative in that they predict and explain some of the widely recognized empirical relationships, but only Black's theory is capable of predicting and explaining the wide *diversity* of facts under a single explanatory framework (see Chapter Six).

Similar to other direct tests of Black's theory of law in areas outside of capital punishment, the empirical results from the current study provide only weak to moderate support for the several of hypotheses derived from the theory and applied to the death penalty context (for a detailed description of the hypotheses and empirical results, see Chapters Six and Eight). While numerous hypotheses receive mixed support across models, several general observations can be made with respect to those findings that were either uniformly supportive or uniformly unsupportive. Results that were consistent with Black's theory across all models involved the racial and gender organization of the cases, defendants' marital status, defendants' prior drug use and violent criminal history, and the relationship between the defendant and the victim (see Tables 11 and 13: Models 1 and 2). The unsupportive results for the impact of prior criminal history and jurisdictional horizontal status were consistent across models with respect to their direct effects, but not their cross-level interactive effects (see Tables 12 and 14). The specific findings with respect to race/ethnicity, gender, and criminal history are discussed in greater detail below.

*Racial Structure*: The racial organization of cases was hypothesized to influence legal decision-making, net of case- and circuit-level legal and extra-legal variables. Although race/ethnicity is a crude proxy for social status, as explained earlier, it is strongly correlated with Black's five dimensions. If it were possible to take into account

393

all (or even most) of the direct/proximal measures of social status, then Black's theory would predict that these categories would be sociologically meaningless (at least with respect to explaining human behavior in conflict situations);<sup>490</sup> but this is highly unlikely and, as previously mentioned, the aggregate characteristic of a group are often assumed in situations where specific characteristics are unknown and the decision-making task is complex or likely to evoke negative emotions (e.g., Bodenhausen and Lichtenstein 1987; Bodenhausen, et al. 1994). The empirical analyses reveal that white-victim cases attract the most law across all models, although black-defendant/white-victim (b/w) cases did not always attract the most law relative to other white-victim cases (see Tables 11 and 13). The finding of white-victim bias is consistent with every study on capital punishment since 1990 except one (Nebraska) (Baldus and Woodworth 2003). Half of those studies (five out of ten) also discovered that b/w cases were more likely than any other offender/victim racial combination to be noticed for the death penalty or sentenced to death (see Section 4.2.2). It is worth noting that several of these studies examined well over one hundred legal and extra-legal variables (see, e.g., Paternoster et al. 2004; Unah and Boger 2001), while others only examined a few dozen, but the findings were very similar across the different model specifications. It is also important to note that, although the current study only examined three legal variables (or five if one also counts the number of offenders and victims), the predictive power of the models (i.e., the percent of case outcomes correctly classified) was very similar to those studies that analyzed over one hundred different covariates (see Unah and Boger 2001).

<sup>&</sup>lt;sup>490</sup> Black recognizes that it may impossible to observe how a phenomenon varies, so it must be measured by its relationship to other phenomena that are observable.

Although Black's theory does not explicitly discuss the interrelationships among the five social dimensions, either within or between levels of analysis (and has he been criticized for neglecting to do so), this project examined several cross-level/cross*dimensional interactions* with respect to the variable impact of the racial structure of a case across jurisdictions that should be consistent with Black's general framework (see Sections 6.3.8 and 8.3.8).<sup>491</sup> It was hypothesized that racial disparities will be greater in jurisdictions that are higher in vertical and normative status, and lower in horizontal status. The first set of racial cross-level models support these predictions: b/w cases were more likely than b/b cases to be noticed for the death penalty in judicial circuits high in vertical and normative status and low in horizontal status (see Tables 12 and 14). The second set of racial cross-level models examining w/w racial differences (relative to b/b) produced mixed results, however. Only the normative status of a jurisdiction interacted with w/w cases in the hypothesized direction: the higher the violent crime rate in a jurisdiction (i.e., low social control), the more powerful the racial disparity between w/w and b/b cases becomes.<sup>492</sup> The effect of a jurisdiction's vertical and horizontal status on the effect of w/w cases was essentially zero (-0.036 and 0.009, respectively).

These results concerning the racial organization death penalty have interesting implications for the capital charging-and-sentencing system in Georgia. First, the race/ethnicity of *both* defendant and victim continue to play a non-trivial role in legal decision-making at *every* stage of the process: death noticing, plea-bargaining, sentencing

<sup>&</sup>lt;sup>491</sup> Recall Black argues that, irrespective of social context, a high status complainant will enjoy more legal advantages than his or her low status adversary; nonetheless, the social location of a conflict will influence how large legal advantages will be (see Section 6.2.5).

<sup>&</sup>lt;sup>492</sup> As explained earlier, cross-level interactions between jurisdictional social status and h/w cases was infeasible because h/w cases only occurred in six of the forty-six judicial circuits.

for cases disposed by plea, and sentencing for cases advancing to the penalty phase. These relationships were robust to multiple model specifications.

Second, it appears that the LWOP legislation enacted in 1993 may have altered the impact of the racial organization of case on legal outcomes—at least with respect to b/w homicides. During the time-period under investigation in this study, LWOP could only be used as a sentencing option in death penalty cases (see Section 5.2.4), so LWOP became a "bargaining-chip" for prosecutors. Indeed, some Georgia prosecutors have openly admitted to seeking the death penalty in order for LWOP to be a sentencing option both during plea bargaining negotiations and during jury trials (see, e.g., Failor 2001). The empirical results reveal that b/w cases are much more likely to be noticed for the death penalty and receive a LWOP sentence resulting from a plea-bargain relative to w/w and b/b cases, but only slightly more likely to result in a death sentence relative to b/b cases and much less likely than w/w cases to result in a death sentence (compare Table 11: Models 1 and 2 with Table 13: Models 1 and 2). The b/w "racial-disparity effect" is significantly reduced by the time a case advances to the final phase of the capital charging-and-sentencing process, but at the price of b/w cases receiving the harshest plea-bargains. It is possible (even likely) that b/w cases advancing to the penalty phase are significantly "less-aggravated" than other cases that advance to the penalty phase of a capital trial (see Sorensen and Wallace 1999). The following evidence is suggestive of this fact: the average number of statutory aggravating factors charged in a b/w case that plead was 2.35, whereas the number for b/w that when to trial was 2.06—a difference of 0.29. By comparison, the average number aggravators charged in w/w cases that plead was 2.02, whereas the number for w/w that went to trial was 2.05. Since

prosecutors have tremendous (and largely unregulated) discretion in the plea-bargaining process (Scott and Stuntz 1992), it is plausible that the LWOP statute simply shifted *some of* the influence of racial bias to the plea bargaining phase. Also recall that recent evidence suggests that prosecutors, at both the state and federal levels, are more likely to seek the death penalty in cases with weaker evidence when the defendant is a member of a minority group (Bruck, Burr, and McNally 2006; Harmon 2001a, b; Parker, Dewees, and Radelet 2001).

Third, the cross-level interactions suggest that the variation in the effect of the racial combination on legal decision-making across jurisdictions is *not* completely random; rather, it can be partly explained by the aggregate social status of the legal jurisdiction along one or more of Black's five social dimensions. According to Black, the handling of cases is only "capricious" from a jurisprudential perspective, and can be predicted and explained by the social geometry of the case. These findings suggest that "racial disproportionality" must be explored at multiple levels of analysis. Baldus *et al.*'s (1990) landmark study of Georgia's capital punishment system in the 1970s documented geographic disparities between circuits with respect to death noticing and death sentencing rates, after adjusting for case-level factors, in the pre- and post-*Furman* eras; however they did not attempt to systematically link these disparities to structural features of the circuits.

Unfortunately, it is difficult to imagine how the Georgia Supreme Court could reasonably modify its proportionality review to take the structural features of judicial circuits into account. The court is already aware of the geographic disparities in the use of the death penalty, and it is unlikely that a more nuanced understanding of how the

397

effect of the racial organization of a case varies across judicial circuits will appreciably influence current practices by the court. Granted, the impact of the structural characteristics on the capital charging-and-sentencing process is likely to be viewed as problematic by the court, but it is doubtful that the court would believe that it would be competent to devise a workable method to mitigate those effects during proportionality screening. The "regional effects" of racial disparity in the capital-charging-andsentencing process is eerily similar to the regional problems associated with executions. Recall from Chapter Two that state authorities began centralizing executions at the turn of the twentieth century because local officials consistently "botched" executions. Racial disparities in sentencing are indicative of a "botched" charging-and-sentencing system in these jurisdictions.

*Gender Structure*. The gender organization of a case was also hypothesized to influence legal decision-making, net of case-level and circuit-level factors. Like race/ethnicity, gender is a proxy for social location according to Black's theory. Women, in the aggregate, have higher status along some dimensions (e.g., horizontal and cultural), but lower status along others (e.g., vertical and normative [authority]) (see Section 6.2.4), According to Black's theory, male-defendant/female-victim (m/f) cases should attract the most law and female-defendant/male-victim cases should attract the least law. Both of these predictions are strongly supported by the evidence (see Tables 11 and 13). Surprisingly, none of the hypotheses concerning the cross-level interactions with the gender organization of the case are supported by the data. In fact, all of the effects are opposite of the hypothesized, although those effect sizes are fairly small.

The robust direct of effects of the gender structure of a case is likely best explained in terms of cultural, relational, and normative status. All crimes, including murder, may be conducted in a more or less conventional manner (see Cooney 1997), and typically, the less conventional the crime, the more law the crime attracts. The conventionally of a crime can be measured by both the conduct *and* characteristics of the parties. Over 67 percent of homicides are intra-gender, so inter-gender crimes are unconventional. Relational status is also important, and many homicides committed by female defendants involve victims not only known to the defendant, but often under the care of the defendant. Female defendants are also likely to murder an intimate partner with whom there is a history of domestic violence against the defendant, thus the victim is likely to be lower in normative status (respectability), all else equal. These three factors-the unconventionality of the crime, the relational distance between the defendant and the victim, and the (lower) normative status of the victim-represent a social geometry that greatly increases the severity of law for m/f cases and greatly decreases the severity of law for f/m cases.

The gender disparities described have not attracted much attention from death penalty analysts or members of the capital prosecution and capital defense bars. Approximately ten percent of all murder and voluntary manslaughter convictions in Georgia during the time period under investigation in this study were of women. This figure mirrors the national data (Streib 2002). Only one woman has been executed in Georgia since 1900 and there is only woman currently on death row in Georgia. According to law professor Victor Strieb, women's lives are treated as being more valuable when they are either defendants or victims (Streib 2002). The thing that is most interesting about the persistence of (and blind-eye to) bias favoring women in the death penalty process is that it is an overt (and unapologetic) acceptance of the influence of an extra-legal factor. From a jurisprudential standpoint, any role that gender plays in the death penalty process is just as impermissible as race/ethnicity—it is a diffuse status characteristic that should have no bearing on a case apart from its association with doctrinally relevant factors in a case (e.g., level of aggravation and mitigation). This fact underscores the reality of both the force *and* acceptance of "extra-doctrinal" phenomena in legal decision-making. As Black (1989, p. 21) has noted, "The handling of cases always reflects the social characteristics of those involved in it. It applies whenever and wherever law is found.... No one has ever observed a legal system without social differentiation. Discrimination is ubiquitous. It is an aspect of the natural behavior of law.

*Normative Structure (Case-level)*. Most of the case factors deemed "legally legitimate" reflect the social organization of the case. Capital statutes enumerate many "special circumstances" that make a case eligible for the death penalty which have nothing to do with the actual conduct of the offender of harm done to the victim. Several of these "legitimate" factors address the defendant's past criminal/deviant conduct—the defendant's normative status. Black's theory predicts that defendants with criminal backgrounds, particularly serious documented criminal histories, will attract the most law. A defendant's currently charged criminal conduct is also an indicator of normative status under Black's theory because any social control is a disadvantage, including a complaint (Black 1976). From a formal legal/jurisprudential model standpoint, the effect of a defendant's prior criminal history and current criminal conduct should have a similar

impact across the various stages of the criminal justice process and across jurisdictions that is, the application of the law (rules) to the facts of the case should be uniform.

In contrast to the predictions of both formal legal theory and Black's theory of law, a defendant's prior felony conviction and the presence of a contemporary felony attracts *less* law across all models but one. The lone exception is the impact of a contemporaneous felony on the likelihood of receiving a death sentence for those cases advancing to the penalty phase. A prior violent felony conviction, on the other hand, attracts the most law across all models analyzed. The number statutory aggravating circumstances charged in a case, which may be both a "legitimate" legal characteristic and indicator of the social geometry of the case, also had the hypothesized effects on legal decision-making.<sup>493</sup>

Only the effect of two of the four "legal" variables significantly varied across jurisdictions—prior felony conviction and prior violent felony conviction. According to Black's theory, these heterogeneous effects can be partly explained by the vertical, normative, and horizontal status of the jurisdiction in which they occur. With respect to prior criminal conviction, only the interaction with a circuit's horizontal status was in the expected direction: the effect of a prior felony conviction on the likelihood of being noticed for the death penalty was weaker in jurisdictions with higher horizontal status. All of the cross-level interactions between a prior violent felony conviction and the circuit-level social dimensions were in the hypothesized direction: jurisdictional vertical status increased the effect, while jurisdictional normative and horizontal status decreased the effect.

<sup>&</sup>lt;sup>493</sup> Defendants with a history of prior drug use (normative status) also attracted the most law.

*Geographic Location*. Another finding that was consistent with Black's theory of law across all models was the fact that there was significant between-circuit variation, after controlling for legal and extra-legal factors at both the case- and circuit-level, in the likelihood that a case: (1) was noticed for the death penalty, (2) resulted in a plea bargain, (3) resulted in a LWOP sentence after a plea bargain, and (4) resulted in a death sentence after advancing to the penalty phase (see Table 11 and Appendix D). The proportion of between-circuit variation ranged from .30 to .58 in the fully specified models. The random-slope models analyzed in the study also revealed substantial between-circuit variation in the effects of those covariates on death noticing, ranging from .18 to .35 (see Table 12). Somewhat surprisingly, the circuit-level variables indexing vertical, normative, and horizontal status only account for a small proportion of that variation; nonetheless, these findings underscore the fact that the social context in which these cases occur is responsible for a significant amount of the variation in legal decision-making across Georgia and must be adequately explored when predicting and explaining legal outcomes.

Uniformly Unsupportive Results. Several findings were also uniformly unsupportive of specific hypotheses derived from Black's theory. Recall that Black's theory posits human behavior can be reconceptualized as social behavior, therefore a call to the police, and arrest, a jury verdict, and appeal, *et cetera* are all increases in law and can be predicted and explained by the social geometry of a case. This formulation of human behavior allows the theory to predict the behavior of diverse actors in the criminal justice process without reference to their subjectivity. This implies that the social geometry of a case should have a similar effect on legal decision-making across the various stages of the capital punishment process—i.e., noticing and sentencing decisions. Contrary to Black's theory, however, most of the variables have quite different effects depending on the stage of the process, and these differences remained after the models included adjustments for non-random selection into the subsequent stages of the death penalty process. This unsupportive finding mirrors the results of other tests of Black's theory that discover various factors of a case which reflect its social geometry have differential impact depending on the stage of the criminal justice process (see, e.g., Dannefer 1984; Myers 1980).

The aggregate normative status of a judicial circuit was almost consistently found to be positively related to the use of law (save Table 11: Model 3). This finding is unsupportive of Black's theory, but consistent with the findings of other studies directly examining Black's theory (see, e.g., Doyle and Luckenbill 1991). It must be noted, however, that his finding pertains to the *direct* effect and not the *conditioning* influence of normative status. As the prior discussion noted, a circuit's normative status had the hypothesized moderating effect in three of the five models examined (see Table 12: Models 1, 2, and 5).

*Mixed Findings and their Implications*. Many of the empirical results neither uniformly support nor uniformly contradict the hypotheses presented in Chapter Six. Table 14 presents a grid indicating how the hypotheses "stacked up" to the data for each of the fully specified models. Each variable (row) listed in the table corresponds with a theoretical prediction, and each column corresponds to a model. There are a total of 97 cells that correspond to a specific predictions derived from the theory—that is, a specific relationship between the variable and the severity of law.<sup>494</sup> Each cell in the table either contains an "S" (supported), "U" (unsupported), "M" (mixed support), or "N/A" (no specific prediction). Of the 97 cells, 63 cells indicate that the prediction was supported (65 percent), whereas 34 cells indicate the prediction was not supported. The two predictions that received mixed support were with respect to the hypotheses about the effect of the racial organization of the case. The results were mixed because, as hypothesized, b/w and w/w cases were more likely to receive the death sentence at the penalty phase than b/b cases; however, w/w cases were more likely than b/w cases to receive the death sentence—a finding in tension with the hypothesis. It was suggested earlier that the racial dynamics with respect to b/w cases and LWOP plea bargains may be responsible for this particular result.<sup>495</sup>

A description of the specific findings for all of these predictions is presented in Chapter Eight and will not be repeated here; the fact that less than two-thirds of hypotheses receive support in the data is problematic for Black's theory and requires an explanation. A plausible answer, and the most straightforward, is that the models do not adequately capture the social geometry of the cases. There is very little information about the victims in these cases, save their race/ethnicity, age, gender, and relationship with the defendant. By contrast, there is much more information about the defendants in these case. The problem of scant victim information is not idiosyncratic to this project, of course, and has plagued prior scholarship on capital punishment for decades. With the

<sup>&</sup>lt;sup>494</sup> As explained in Chapter Six, one of the greatest strengths of Black's theory it is ability to generate numerous predictions with few postulates (Jasso 2006, p. 38).

<sup>&</sup>lt;sup>495</sup> The aforementioned tabulation is actually misleading considering that the "death noticing" models essentially duplicate the same results and only the outcome of one prediction differs between them. Excluding the death noticing model for the murder conviction sample and recalculating the predictions yields the same proportions.

exception of a few well-funded projects (e.g., Baldus's initial study of Georgia and the recent studies in Maryland and North Carolina), most death penalty researchers have been limited in this respect. It is impossible to fully capture the direction, location, and distance (i.e., the social geometry) of a conflict without fairly complete information on the parties involved—particularly with respect to the specific variables under investigation. Since there was information on the race/ethnicity, gender, and age of both the defendant and victim, these general status differences could be explored in the study. It is worth highlighting that results concerning these differences were generally supportive of the hypotheses. This does not imply that Black's theory would have received more empirical support if more information was available about victims, but it does, at least, suggest that it is extremely important to focus on the relative status positions of the defendant and victim. Recall that predictions concerning the impact of defendants' socioeconomic, educational, and employment backgrounds were not corroborated by the evidence. It is highly likely that information about victims' backgrounds with respect to those very same variables was known to prosecutors prior to seeking the death penalty and offering a plea bargains, and known to jurors prior to imposing a death sentence.

The potential problems stemming from inadequate data may also account for the mixed support for the hypothesized direct and indirect effects of the indices of vertical and normative status. As explained in Chapter Eight, level-two equations (i.e., equations predicting random intercepts and slopes) will be misspecified if the level-one equation is misspecified. Many of the case-level extra-legal coefficients do not capture the status differentials between the parties, so the lack of support for the hypotheses predicting the

direct and indirect effects of Black's social dimensions may be an artifact of the data limitations.<sup>496</sup> With respect to vertical status, only four of the nine predictions received empirical support. The empirical results for the effect of normative status are even less favorable: only three of the nine predictions with respect to normative status support Black's theory. Even prior to including case-level covariates in the models, the effects of vertical and normative status in the various models were inconsistent (see Appendix C). This result, however, does not directly contradict Black's theoretical predictions because the theory treats all non-sociological variables and the other dimensions of social space as constants (see Section 6.2.2).<sup>497</sup> Nonetheless, these results are worrisome because many of the effects were in the opposite direction of Black's predictions. It is impractical for empirical analysts to take into account every variable identified in a social theory, and Black's posits that the testability of his theory is one of its strongest virtues. But the relational structure of the case is *the* central proposition of Black's theory, so unsupportive evidence from studies that fail to adequately capture these status differentials should not be viewed as identifying fatal errors in the theory (cf. Lieberson and Lynn 2002).

An equally plausible explanation for the inconsistent empirical results is that structural dynamics influencing legal behavior in the capital charging-and-sentencing process, at both the micro and macro-levels, have changed. If Klüver and Schmidt (1999) are correct, and social systems are able to change their rules of interaction

<sup>&</sup>lt;sup>496</sup> The inter-item correlations between the constituent variables in those scales, as well as the overall scale reliability (see Chapter Eight) strongly suggest that those indices were adequately capturing Black's social dimensions, so it also unlikely to be a measurement error problem.

<sup>&</sup>lt;sup>497</sup> In addition to the omission of several important case-level factors, the omission of a variable indexing corporate status may have negatively impacted the circuit-level models. But the results support seven of the nine predictions with respect to the effect of horizontal status, so it is doubtful that model misspecification would impact two, but not three, of the social dimensions.

according to the particular demands of their environment, it is possible that the relationships between Black's five dimensions and law have been altered. Klüver and Schmidt opine that the relationships between social systems have been fairly static because they "were just not pressed enough to change some of their basic values" (p. 322). Of particular relevance to Black's theory is Klüver and Schmidt's admonition concerning the development of an overly static geometric theory of social systems:

There are no master equations which may describe the dynamics of social systems...the reason for this is the fact that social actors and therefore social systems can change their rules, their dynamics and even of course their meta rules. In particular, social systems are self-referential systems which can model themselves and anticipate their possible future (p. 322).

To be sure, the dynamism of social systems is not fatal to Black's general explanatory framework. In fact, Black (1989) has acknowledged that certain crude proxies of social status (e.g., race/ethnicity, gender) may become less reliable overtime as membership into these groups becomes associated with higher status locations. The enduring strength of Black's approach is his identification and description of the five social dimensions that influence social behavior. Revision must be directed at the content of the substantive theories derived from the paradigm, and not the paradigm itself (Lenski 1988, p. 169). Black (1984, p. 16) has already commented that other types of social control may not behave according to the same principles of law and should be investigated. If social systems are self-referential systems that must be "pressed enough to change some of their basic values," then is plausible that certain social dynamics of the capital punishment process have changed according to the demands of their environment. Indeed, there was no ideological realignment of the U.S. Supreme Court during the *Atkins, Ring*, and *Roper* decisions that fundamentally altered the administration of the death penalty—there was

simply heightened public awareness and debate (nationally and globally) concerning the appropriateness of the death penalty for the mentally disabled (*Atkins*) and juveniles (*Roper*), as well as over capital defendants' Sixth Amendment rights (*Ring*) (see Section 2.3).

## 9.1 CONCLUSION

The present study revealed that race/ethnicity, gender, and region continue to influence the capital charging-and-sentencing process in Georgia. The enactment of Georgia's life without the possible of parole (LWOP) legislation in 1993 had very little impact on the charging-and-sentencing patterns that existed before the statute— defendants who plead guilty to murdering white victims and female victims received the harshest sentencing during plea bargaining and trial. There also remained significant regional variation in the death penalty process in Georgia and the level of racial/ethnic and gender disparity in a given jurisdiction could be partly explained by the structural characteristics of that jurisdiction. These results are also consistent with nearly every methodologically rigorous study conducted on the death penalty in the United States since 1990.

While surprising to some, and disappointing to many, the ineffectiveness of the post-*Furman* procedural reforms is completely consistent with Donald Black's theory of law. According to Black (1989, p. 21), "the handling of cases always reflects the social characteristics of those involved it. It applies whenever and wherever law is found. [...] No one has ever observed a legal system without social differentiation. Discrimination is ubiquitous. It is an aspect of the natural behavior of law." Black even suggests that the narrow focus on race/ethnicity, social class, and gender has resulted in analysts actually

ignoring the many other types of discrimination that impacts the handling of cases. The only way to eliminate, or substantially reduce, this discrimination is to limit the amount of social information to which legal decision-makers have access. This may be impractical from a policy standpoint; nonetheless, it is the strongest implication from Black's theory. Moreover, it underscores why the incremental fixes implemented since *Furman* have been unsuccessful—if legal rules are unable to predict the behavior of legal agents in a meaningful way, then reforming legal rules to modify the behavior of legal agents is largely pointless.

In *Furman*, Justice William Douglas believed the absence of sentencing standards permitted jurors to discriminate according to social hierarchies, but was optimistic that death penalty statutes could be revised to conform to the requirements of the constitution. When the U.S. Supreme Court lifted the moratorium on executions four years later in *Gregg*, its decision was based on the belief in the ability of the newly designed post*Furman* statutes to eliminate discrimination and caprice rather than any scientific evidence demonstrating that the statutes had the desired effect. Not only did the "guided discretion" provisions of the post*-Furman* have a minimal impact on juror discretion because the special circumstances were written so broadly, the provisions were inherently flawed because their influence was basically limited to only one of the many decision stages where discretion is exercised by legal agents.

Subsequent decisions by the Court also underscored its inability to appreciate the undesirable influence of social hierarchies on legal decision-making. Two years after *Gregg*, in *Lockett*, the Court held that statutory restrictions on the types of mitigation evidence that a defendant could present on her or his behalf during sentencing were

unconstitutional, thereby opening the floodgates for social information and discrimination (see Section 2.3). As Zimring (2003) has explained, the penalty phase of capital trials has become a status competition between offender and victim families. Recall from Chapter Four that, after the Civil War, rather than explicitly mentioning race in criminal statutes, Southern states began giving prosecutors and jurors greater discretion in charging and sentencing decisions in order to reinforce the pre-Emancipation racial order. Thirteen years after *Gregg*, in *Stanford*, several Justices believed it was necessary and proper for legal agents to consider numerous factors beyond the actual injury caused by the crime in capital cases when assessing an offender's culpability (see Section 4.1.2).

Perhaps it is axiomatic that discretion is only desirable to those who benefit from it. Both the capital prosecution and defense communities have repeatedly asked courts and legislatures to give their "side" more discretion, while simultaneously requesting that their opponents have less discretion. This tug-of-war has resulted in capital cases remaining just as saturated with social information as they were in the pre-*Furman* era. Twenty years after *Furman*, Justice Harry Blackmun, who dissented in *Furman*, concurred in *Gregg*, and dissented in *McCleskey*, famously wrote, "[D]espite the effort of the States and courts to devise legal formulas and procedural rules to [impose the death penalty fairly and consistently], the death penalty remains fraught with arbitrariness, discrimination, caprice, and mistake. This is not to say that the problems with the death penalty today are identical to those that were present 20 years ago. Rather, the problems that were pursued down one hole with procedural rules and verbal formulas have come to the surface somewhere else, just as virulent and pernicious as they were in their original form" (*Callins v. Collins*, 510 U.S. 1144, 1145 [1994]). It is likely the recognition of this inherent fallibility of death penalty systems—no matter how well crafted—has led many of the world's nations, and nearly all of our closest international allies, to dismantle their own capital punishment systems during the three decades that the United States has unsuccessfully attempted to repair what Justice Blackmun referred to as "the machinery of death."

## References

- Abrams, David S. 2006. "Essays on the Economics of Law, Crime and Discrimination." Ph.D. dissertation, Department of Economics, Massachusetts Institute of Technology, Cambridge, Mass.
- Abrams, David S. and Albert H. Yoon. 2007. "The luck of the draw: Using random case assignment to investigate attorney ability." *University of Chicago Law Review* 74:1145–1177.
- Abramsky, Sasha. 2002. *Hard Time Blues: How Politics Built a Prison Nation*. New York: St. Martin's Press.
- Achen, Christopher H. 1986. *The Statistical Analysis of Quasi-Experiments*. Berkeley and Los Angeles, Calif.: University of California Press.
- Acker, James R. 1993. "A different agenda: The Supreme Court, empirical research evidence, and capital punishment decisions, 1986-1989." *Law and Society Review* 27:65–88.
- Acker, James R. and Charles S. Lanier. 2003. "Beyond human ability? The rise and fall of death penalty legislation." Pp. 85–126 in *America's Experiment with Capital Punishment: Reflections on the Past, Present, and Future of the Ultimate Penal Sanction*, 2d ed., edited by J. R. Acker, R. M. Bohm, and C. S. Lanier. Durham, N.C.: Carolina Academic Press.
- Adams, Michael. 1994. *The Best War Ever: American in World War II*. Baltimore, Md.: Johns Hopkins University Press.
- Agamben, Giorgio. [1995] 1998. *Homo Sacer: Sovereign Power and Bare Life.* Translated by D. Heller-Roazen. Stanford, Calif.: Stanford University Press.
- Agnew, Robert. 1995. "Testing the leading crime theories: An alternative strategy focusing on motivational processes." *Journal of Research in Crime and Delinquency* 32:363–398.
- —. 1999. "A general strain theory of community differences in crime rates." Journal of Research in Crime and Delinquency 36:123–155.
- Aguirre, Adalberto and David V. Baker. 1993. "Racial prejudice and the death penalty: A research note." *Social Justice* 20:150–155.
- Aiken, Leona S. and Stephen G. West. 1991. *Multiple Regression: Testing and Interpreting Interactions*. Newbury Park, Calif.: Sage.

- Aitkin, Murray and Nicholas T. Longford. 1986. "Statistical modeling issues in school effectiveness studies." *Journal of the Royal Statistical Society* (Series A) 149:1– 43.
- Akaike, Hirotugu. 1973. "Information theory and an extension of the maximum likelihood principle." Pp. 267–281 in Second International Symposium on Information Theory, edited by B. N. Petrov and F. Csaki. Budapest, Hungary: Akadémiai Kiadó.
- Akers, Ronald L. 1990. "Rational choice, deterrence, and social learning theory in criminology: The path not taken." *Journal of Criminal Law and Criminology* 81:653–676.
- —. 1998. Social Learning and Social Structure: A General Theory of Crime and Deviance. Boston, Mass.: Northeastern University Press.
- —. 2000. *Criminological Theories: Introduction, Evaluation and Application*, 3d ed. Los Angeles, Calif.: Roxbury.
- Alba, Richard D., John R. Logan, and Brian J. Stultz. 2000. "How segregated are middleclass African Americans?" *Social Forces* 47:543–558.
- Albert, Craig J. 1999. "Challenging deterrence: New insights on capital punishment derived from panel data." *University of Pittsburgh Law Review* 60:321–371.
- Albonetti, Celesta A. 1987. "Prosecutorial discretion: The effects of uncertainty." *Law and Society Review* 21:291–313.
- —. 1991. "An integration of theories to explain judicial discretion." Social Problems 38:247–266.
- —. 1997. "Sentencing under the federal sentencing guidelines: Effects of defendant characteristics, guilty pleas, and departures on sentence outcomes for drug offenses, 1991-1992." *Law and Society Review* 31:789–822.
- Allison, Paul D. 1999. "Comparing logit and probit coefficients across groups." Sociological Methods and Research 28:186–208.
- —. 2000. "Multiple imputation for missing data: A cautionary tale." Sociological Methods and Research 28:301–309.
- 2002. Missing Data. Thousand Oaks, Calif.: Sage.
- Allport, Gordon W. 1954. The Nature of Prejudice. New York: Perseus Books.

- America, Richard F. 1993. *Paying the Social Debt: What White America Owes Black America*. Westport, Conn.: Greenwood Press.
- American Bar Association. 1980. "Civil Disabilities of Convicted Persons, Standard 23-8.4." ABA Standards for Criminal Justice: Legal Standards for Prisoners. American Bar Association, Washington, D.C.
- —. 2001. "Death Without Justice: A Guide for Examining the Administration of the Death Penalty in the United States (Protocols)." American Bar Association, Washington, D.C.
- —. 2003a. "Building Momentum: The American Bar Association Call for a Moratorium on Executions Takes Hold." American Bar Association, Washington, D.C.
- —. 2003b. "Prohibited Collateral Sanctions, Standard 19-2.6." ABA Standards for Criminal Justice: Discretionary Disqualification of Convicted Persons. American Bar Association, Washington, D.C.
- —, Death Penalty Moratorium Implementation Project. 2004. "Assessment Guide, Part I: The Collection of Information on Georgia's Death Penalty System." American Bar Association, Washington, D.C.
- -, Criminal Justice Section. 2006a. "Achieving Justice: Freeing the Innocent, Convicting the Guilty." American Bar Association, Washington, D.C.
- ---, Death Penalty Moratorium Implementation Project. 2006b. "Evaluating the Fairness and Accuracy in State Death Penalty Systems: The Georgia Death Penalty Assessment Report." American Bar Association, Washington, D.C.
- American Civil Liberties Union. 2003. "Race and the death penalty." New York: American Civil Liberties Union. Retrieved February 26, 2003 (http://www.aclu.org/DeathPenalty/DeathPenalty.cfm?ID=9312&c=62).
- —. 2004. "Scattered justice: Geographic disparities of the death penalty." New York: American Civil Liberties Union. Retrieved March 4, 2004 (http://www.aclu.org/DeathPenalty/DeathPenalty.cfm?ID=15182&c=62).
- Amnesty International. 2006. "Amnesty International Report 2006." Amnesty International Publishing, London.
- Anderson, Elijah. 1990. *Streetwise: Race, Class, and Change in an Urban Community*. Chicago, Ill.: University of Chicago Press.
- —. 1999. Code of the Street: Decency, Violence and the Moral Life in the Inner City. Boston, Mass.: Norton.

- Anderson, Elijah and Douglas S. Massey. 2001. "The sociology of race in the United States." Pp. 3–12 in *Problem of the Century: Racial Stratification in the United States*, edited by E. Anderson and D. S. Massey. Thousand Oaks, Calif.: Sage.
- Anderson, Scott. 1991. "As Flies to Wanton Boys': Death eligible defendants in Georgia and Colorado." *Trial Talk* 40:9–16.
- Anyon, Jean. 1997. *Ghetto Schooling: A Political Economy of Urban Educational Reform.* New York: Teachers College Press.
- Appiah, Kwame Anthony. 1992. In My Father's House: Africa in the Philosophy of *Culture*. New York: Oxford University Press.
- Applegate, Brandon K., John P. Wright, R. Gregory Dunaway, Francis T. Cullen, and John D. Wooldredge. 1993. "Victim-offender race and support for capital punishment: A factorial design approach." *American Journal of Criminal Justice* 18:95–115.
- Arkin, Stephen. 1980. "Discrimination and arbitrariness in capital punishment: An analysis of Post-*Furman* murder cases in Dade County, Florida, 1973-1976 (Note)." *Stanford Law Review* 33:75–101.
- Aronson, Elliot. 1972. The Social Animal. San Francisco, Calif.: W.H. Freeman.
- Arvanites, Thomas M. and Martin A. Asher. 1998. "State and county incarceration rates: The direct and indirect effects of race and inequality." *American Journal of Economics and Sociology* 57:207–222.
- Associated Press. 2009. "N.M. repeals death penalty." *Los Angeles Times*, March 19, p. A15.
- Atlanta Bar Association. 2001. "Moratorium Resolution." Atlanta Bar Association, Atlanta, Ga.
- Austin, John. [1832] 1995. *The Province of Jurisprudence Determined*. Cambridge: Cambridge University Press.
- Austin, Roy L. and Mark D. Allen. 2000. "Racial disparity in arrest rates as an explanation of racial disparity in commitments to Pennsylvania's prisons." *Journal of Research in Crime and Delinquency* 37:200–200.
- Austin, Thomas L. 1981. "The influence of court location on type of criminal offense: The rural-urban factor." *Journal of Criminal Justice* 9:305–316.

- Avakame, Edem F., James J. Fyfe, and Candace McCoy. 1999. "Did you call the police? What did they do?' An empirical assessment of Black's Theory of Mobilization of Law." Justice Quarterly 16:765–792.
- Ayers, Edward L. 1984. Vengeance and Justice: Crime and Punishment in the 19th-Century American South. New York: Oxford University Press.
- Ayres, Ian and Richard Brooks. 2005. "Does affirmative action reduce the number of black lawyers?" *Stanford Law Review* 57:1807–1854.
- Bacon, Francis. 1875. *The Works of Francis Bacon*, vol. 4, edited by J. Spedding, R. L. Ellis, and D. D. Heath. London: Longmans and Company.
- Baicker, Katherine. 2004. "The budgetary repercussions of capital convictions." Advances in Economic Policy and Analysis 4:1311.
- Baier, Colin J. and Bradley R. Entner Wright. 2001. "'If you love me, keep my commandments': A meta-analysis of the effect of religion on crime." *Journal of Research in Crime and Delinquency* 38:3–21.
- Bailey, William C. 1980. "Deterrence and the celerity of the death penalty: A neglected question in deterrence research." *Social Forces* 58:1308–1333.
- —. 1990. "Murder, capital punishment, and television: Execution publicity and homicide rates." *American Sociological Review* 55:628–633.
- —. 1998. "Deterrence, brutalization, and the death penalty: Another examination of Oklahoma's return to capital punishment." *Criminology* 36:711–733.
- Bailey, William C. and Ruth D. Peterson. 1989. "Murder and capital punishment: A monthly time-series analysis of execution publicity." *American Sociological Review* 54:722–743.
- —. 1997. "Murder, capital punishment, and deterrence: A review of the literature." Pp. 135–161 in *The Death Penalty in America: Current Controversies*, edited by H. A. Bedau. New York: Oxford University Press.
- Baker, Scott and Claudio Mezitti. 2001. "Prosecutorial resources, plea bargaining, and the decision to go to trial." *Journal of Law, Economics, and Organization* 17:149–167.
- Baldus, David C. 1980. "Quantitative methods for judging the comparative excessiveness of death sentences." Pp. 83–94 in *The Use/Nonuse/Misuse of Applied Social Research in the Courts*, edited by M. J. Saks and C. H. Baron. Cambridge, Mass.: Abt Books.

- —. 1991. "Death Penalty Proportionality Review Project: Final Report to the New Jersey Supreme Court." New Jersey Judiciary, Trenton, N.J.
- —. 1995. "The death penalty dialogue between law and social science." Indiana Law Journal 70:1033–1041.
- —. 2001. "Memorandum to the Honorable Russell D. Feingold: Re DOJ Report on the Federal Death Penalty System." University of Iowa College of Law, Iowa City, Iowa.
- Baldus, David C. and James W. L. Cole. 1975. "Statistical evidence on the deterrent effect of capital punishment: A comparison of the work of Thorsten Sellin and Isaac Ehrlich on the deterrent effect of capital punishment." *Yale Law Journal* 85:170–186.
- —. 1977. "Quantitative proof of intentional discrimination." *Evaluation Quarterly* 1:51– 86.
- Baldus, David C., Charles A. Pulaski, Jr., and George Woodworth. 1986a. "Arbitrariness and discrimination in the administration of the death penalty: A challenge to state supreme courts." *Stetson Law Review* 15:133–261.
- -. 1994a. "False attacks on the Racial Justice Acts." Des Moines Register, June 7, p. 9A.
- —. 1994b. "Reflections on the 'inevitability' of racial discrimination in capital sentencing and the 'impossibility' of its prevention." Washington and Lee Law Review 51:359–430.
- Baldus, David C. and George Woodworth. 1983. "Comparative review of death sentences: An empirical study of the Georgia experience." *Journal of Criminal Law and Criminology* 74:661–753.
- —. 1997. "Race Discrimination in America's Capital Punishment System Since Furman v. Georgia (1972): The Evidence of Race Disparities and the Record of Our Courts and Legislatures in Addressing this Issue." American Bar Association, Chicago, Ill.
- —. 2001. "Race of Victim and Race of Defendant Disparities in the Administration of Maryland's Capital Charging and Sentencing System (1979-1996): Preliminary Finding." Senate Judicial Proceeding Committee, Annapolis, Md.
- —. 2003. "Race discrimination in the administration of the death penalty: An overview of the empirical evidence with special emphasis on the post-1990 research." *Criminal Law Bulletin* 39:194–226.

- —. 2004. "Race discrimination and the legitimacy of capital punishment: Reflections on the interaction of fact and perception." *DePaul Law Review* 53:1411–1495.
- Baldus, David C., George Woodworth, and Charles A. Pulaski, Jr. 1985. "Monitoring and evaluating contemporary death sentencing systems: Lessons from Georgia." *University of California at Davis Law Review* 18:1375–1407.
- —. 1986b. "Arbitrariness and discrimination in Colorado's post-*Furman* capital charging and sentencing process: A preliminary report." University of Iowa, Iowa City, Iowa. Unpublished manuscript.
- —. 1990. Equal Justice and the Death Penalty: A Legal and Empirical Analysis. Boston, Mass.: Northeastern University Press.
- Baldus, David C., George Woodworth, Gary L. Young, and Aaron M. Christ. 2002.
  "Arbitrariness and discrimination in the administration of the death penalty: A legal and empirical analysis of the Nebraska experience (1973-1999)." *Nebraska Law Review* 81:486–753.
- Baldus, David C., George Woodworth, David Zuckerman, Neil A. Weiner, and Barbara Broffit. 1998. "Racial discrimination and the death penalty in the Post-Furman Era: An empirical and legal overview, with recent findings from Philadelphia." Cornell Law Review 83:1638–1770.
- —. 2001. "The use of peremptory challenges in capital murder trials: A legal and empirical analysis." University of Pennsylvania Journal of Constitutional Law 3:3–170.

Baldwin, James. 1955. Notes of a Native Son. Boston, Mass.: Beacon Press.

- —. 1985. Price of the Ticket. New York: St. Martin's.
- Ball-Rokeach, Sandra J. 1973. "Values and violence: A test of the subculture of violence thesis." *American Sociological Review* 38:736–749.
- Bamshad, Michael J. and Steve E. Olson. 2003. "Does race exist?" *Scientific American* 289:78–85.
- Banner, Stuart. 2002. *The Death Penalty: An American History*. Cambridge, Mass.: Harvard University Press.
- Barber, Bernard. 1961. "Resistance by scientists to scientific discovery." *Science* 134:596–602.
- Barkan, Steven E. and Steven F. Cohn. 1994. "Racial prejudice and support for the deathpenalty by whites." *Journal of Research in Crime and Delinquency* 31:202–209.

- Barndt, Joseph R. 1972. *Liberating Our White Ghetto*. Minneapolis, Minn.: Augsburg Fortress.
- Barnett, Arnold. 1985. "Some distribution patterns for the Georgia death sentence." University of California at Davis Law Review 18:1327–1374.
- Baron, Robert S., Mary L. Inman, Chuan Feng Kao, and Henrietta L. Logan. 1992.
  "Negative emotion and superficial social processing." *Motivation and Emotion* 16:323–346.
- Barry, Donald M. and Alexander Greer. 1981. "Sentencing versus prosecutorial discretion: The application of a new disparity measure." *Journal of Research in Crime and Delinquency* 18:254–271.
- Barsky, Robert, John Bound, Kerwin K. Charles, and Joseph Lupton. 2002. "Accounting for the black-white wealth gap: A nonparametric approach." *Journal of the American Statistical Association* 97:663–673.
- Baum, Christopher F. 2006. *An Introduction to Modern Econometrics using Stata*. College Station, Tex.: Stata Press.
- —. 2009. An Introduction to Stata Programming. College Station, Tex.: Stata Press.
- Baumer, Eric P., Steven F. Messner, and Richard B. Felson. 2000. "The role of victim characteristics in the disposition of murder cases." *Justice Quarterly* 17:281–307.
- Baumer, Eric P., Steven F. Messner, and Richard Rosenfeld. 2003. "Explaining spatial variation in support for capital punishment: A multilevel analysis." *American Journal of Sociology* 108:844–875.
- Baumgartner, M. P. 1978. "Law and social status in Colonial New Haven, 1639-1665."
  Pp. 153–178 in *Research in Law and Sociology: An Annual Compilation of Research*, vol. 1, edited by R. J. Simon. Greenwich, Conn.: JAI Press.
- —. 1984. "Social control from below." Pp. 303–339 in *Toward a General Theory of Social Control: Fundamentals*, vol. 1, *Studies on Law and Social Control*, edited by D. J. Black. New York: Academic Press.
- —. 1992. "The myth of discretion." Pp. 129–162 in *The Uses of Discretion*, edited by K. Hawkins. Oxford: Clarendon Press.
- —. 2002. "*The Behavior of Law*, or 'How to Sociologize with a Hammer'." *Contemporary Sociology* 31:644–649.

- Bausman, Kent and W. Richard Goe. 2004. "An examination of the link between employment volatility and the spatial distribution of property crime rates." *American Journal of Economics and Sociology* 63:665–695.
- Beale, Sara Sun. 1997. "What's law got to do with it? The political, social, psychological and other non-legal factors influencing the development of (federal) criminal law." *Buffalo Criminal Law Review* 1:23–66.
- Beccaria, Cesare. [1764] 1819. On Crimes and Punishment, 2d ed. Translated by E. D. Ingraham. Philadelphia, Pa.: Philip H. Nicklin.
- Bechtler, Thomas W. 1977. "American Legal Realism reevaluated." Pp. 1–49 in *Law in a Social Context*, edited by T. W. Bechtler. Norwell, Mass.: Kluwer.
- Beck, James C. and Robert Shumsky. 1997. "A comparison of retained and appointed counsel in cases of capital murder." *Law and Human Behavior* 21:525–538.
- Beck, Nathaniel and Simon Jackman. 1998. "Beyond linearity by default: generalized additive models." *American Journal of Political Science* 42:596–627.
- Becker, Gary S. 1968. "Crime and punishment: An economic approach." *Journal of Political Economy* 76:169–217.
- Beckett, Katherine and Theodore Sasson. 2000. *The Politics of Injustice: Crime and Punishment in America*. Thousand Oaks, Calif.: Pine Forge Press.
- Bedau, Hugo A. 1964. "Death sentences in New Jersey: 1907-1960." *Rutgers Law Review* 19:1–64.
- —. 1997a. "Background and developments." Pp. 3–35 in *The Death Penalty in America: Current Controversies*, edited by H. A. Bedau. New York: Oxford University Press.
- (Ed.). 1997b. *The Death Penalty in America: Current Controversies*. New York: Oxford University Press.
- —. 1997c. "Innocence and the death penalty: Assessing the danger of mistaken executions." Pp. 344–360 in *The Death Penalty in America: Current Controversies*, edited by H. A. Bedau. New York: Oxford University Press.
- —. 1997d. "Prison homicides, recidivist murder, and life imprisonment." Pp. 176–182 in *The Death Penalty in America: Current Controversies*, edited by H. A. Bedau. New York: Oxford University Press.
- Bedau, Hugo A. and Michael L. Radelet. 1987. "Miscarriages of justice in potentially capital cases." *Stanford Law Review* 40:21–179.
- Begg, Colin B. 1994. "Publication bias." Pp. 399–409 in *The Handbook of Research Synthesis*, edited by H. Cooper and L. V. Hedges. New York: Russell Sage Foundation.
- Behrens, Angela, Christopher Uggen, and Jeff Manza. 2003. "Ballot manipulation and the 'Menace of Negro Domination': Racial threat and felon disenfranchisement in the United States, 1850-2002." *American Journal of Sociology* 109:559–605.
- Bell, Derrick A. 1992. *Faces at the Bottom of the Well: The Permanence of Racism*. New York: Basic Books.
- —. 2004. Silent Covenants: Brown v. Board of Education and the Unfulfilled Hopes for Racial Reform. New York: Oxford University Press.
- Bennett, Richard R. 1991. "Routine activities: A cross-national assessment of a criminological perspective." *Social Forces* 70:147–163.
- Bennett, William J., John J. DiIulio, and John P. Walters. 1996. Body Count: Moral Poverty--and How to Win America's War Against Crime and Drugs. New York: Simon & Schuster.
- Bentham, Jeremy. [1780] 1948. *The Principles Morals and Legislation*, 6th ed. New York: Hefner.
- Berg, Thomas C. 2000. "Religious conservatives and the death penalty." *William and Mary Bill of Rights Journal* 9:31–60.
- Berger, Peter and Thomas Luckmann. 1966. *The Social Construction of Reality*. Garden City, N.Y.: Doubleday.
- Berger, Vivian. 1991. "Black box decisions' on life or death -- If they're arbitrary, don't blame the jury: A Reply to Judge Patrick Higginbotham." *Case Western Reserve Law Review* 41:1067–1092.
- Berk, Richard A. 1983. "An introduction to sample selection bias in sociological data." *American Sociological Review* 48:386–398.
- —. 2003. Regression Analysis: A Constructive Critique. Thousand Oaks, Calif.: Sage.
- —. 2005a. "An introduction to ensemble methods for data analysis." Sociological Methods and Research 34:263–295.

- —. 2005b. "New claims about executions and general deterrence: *Deja Vu* all over again?" *Journal of Empirical Legal Studies* 2:303–330.
- —. 2005c. "Randomized experiments as the bronze standard." Journal of Experimental Criminology 1:417–433.
- Berk, Richard A., Azusa Li, and Laura Hickman. 2005. "Statistical difficulties in determining the role of race in capital cases: A reanalysis of data from the state of Maryland." *Journal of Quantitative Criminology* 21:365–390.
- Berk, Richard A. and Joseph Lowery. 1985. "Factors Affecting Death Penalty Decisions in Mississippi." Unpublished manuscript.
- Berk, Richard A., Robert E. Weiss, and John C. Boger. 1993. "Chance and the death penalty." *Law and Society Review* 27:89–110.
- Bernard, Thomas J. 1995. "The black hole: Sources of confusion for criminologists in Black's theory." *Social Pathology* 1:219–227.
- —. 2002. "Donald Black's influence in criminology." Contemporary Sociology 31:650– 652.
- Bessler, John D. 1997. *Death in the Dark: Midnight Executions in America*. Boston, Mass.: Northeastern University Press.
- —. 2003. Legacy of Violence: Lynch Mobs and Executions in Minnesota. Minneapolis, Minn.: University of Minnesota Press.
- Bhattacharya, Jay, Dana Goldman, and Daniel McCaffrey. 2006. "Estimating probit models with self-selected treatments." *Statistics in Medicine* 25:389–413.
- Bickle, Gayle S. and Ruth D. Peterson. 1991. "The impact of gender-based family roles on criminal sentencing." *Social Problems* 38:372–394.
- Bienen, Leigh B., Neil A. Weiner, Deborah W. Denno, Paul D. Allison, and Douglas Mills. 1988. "The reimposition of capital punishment in New Jersey: The role of prosecutorial discretion." *Rutgers Law Review* 41:27–372.
- Bifulco, Robert. 2002. "Addressing self-selection bias in quasi-experimental evaluations of whole-school reform: A comparison of methods." *Evaluation Review* 26:545–572.
- Black, Donald. 1970. "Production of crime rates." *American Sociological Review* 35:733–748.
- —. 1972. "The boundaries of legal sociology." Yale Law Journal 81:1086–1100.

- —. 1973. "The mobilization of law." Journal of Legal Studies 2:125–149.
- —. 1979a. "Common sense in the sociology of law." American Sociological Review 44:18–27.
- —. 1979b. "A strategy of pure sociology." Pp. 149–168 in *Theoretical Perspectives in Sociology*, edited by S. G. McNall. New York: St. Martin's Press.
- —. 1983. "Crime as social control." American Sociological Review 48:34–45.
- —. 1984. "Social control as a dependent variable." Pp. 1–29 in Toward a General Theory of Social Control: Fundamentals, vol. 1, Studies on Law and Social Control, edited by D. J. Black. New York: Academic Press.
- —. 1987. "Compensation and the social structure of misfortune." Law and Society Review 21:563–584.
- —. 1989. Sociological Justice. New York: Oxford University Press.
- —. 1990. "The elementary forms of conflict management." Pp. 43–69 in New Directions in the Study of Justice, Law, and Social Control, edited by Arizona State University School of Justice Studies. New York: Plenum Press.
- —. 1992. "Social control of the self." Pp. 39–49 in Virginia Review of Sociology, Volume 1: Law and Conflict Management, edited by J. Tucker. Greenwich, Conn.: JAI Press.
- —. 1993. The Social Structure of Right and Wrong. New York: Academic Press.
- -. 1995. "The epistemology of pure sociology." Law and Social Inquiry 20:829-870.
- —. 1999. "Dispute settlement by the police." Pp. 61–95 in *The Social Organization of Law*, 2d ed., edited by M. P. Baumgartner. San Diego, Calif.: Academic Press.
- —. 2000a. "Dreams of pure sociology." Sociological Theory 18:343–367.
- 2000b. "On the origin of morality." Journal of Consciousness Studies 7:107-119.
- -. 2000c. "The purification of sociology." Contemporary Sociology 29:704–709.

- —. 2002a. "The geometry of law: An interview with Donald Black." International Journal of the Sociology of Law 30:101–129.
- —. 2002b. "Pure sociology and the geometry of discovery." *Contemporary Sociology* 31:668–674.
- -. 2004a. "The geometry of terrorism." Sociological Theory 22:14–25.
- —. 2004b. "Terrorism as social control." Pp. 9–18 in Terrorism and Counter-Terrorism: Criminological Perspectives, vol. 5, Sociology of Crime, Law, and Deviance, edited by M. Deflem. Oxford: Elsevier Science.
- —. 2004c. "Violent structures." Pp. 145–158 in Violence: From Theory to Research, edited by M. Zhan, H. H. Brownstein, and S. L. Jackson. Newark, N.J.: LexisNexis/Anderson Publishing.
- Black, Donald and M. P. Baumgartner. 1983. "Toward a theory of the third party." Pp. 84–114 in *Empirical Theories about Courts*, edited by K. O. Boyum and L. Mather. New York: Longman.
- Black, Donald and Maureen Mileski (Eds.). 1973. *The Social Organization of Law*. New York: Seminar Press.
- Blair, Irene V., Charles M. Judd, and Kristine M. Chapleau. 2004. "The influence of afrocentric facial features in criminal sentencing." *Psychological Science* 15:674– 679.
- Blalock, Hubert M. 1967. "Status inconsistency, social mobility, status integration and structural effects." *American Sociological Review* 32:790–801.
- Blume, John H. 1999. "Judicial politics, death penalty appeals, and case selection: An empirical study." *Southern California Law Review* 72:465–503.
- Blume, John H., Theodore Eisenberg, and Sheri L. Johnson. 1998. "Post-*McCleskey* racial discrimination claims in capital cases." *Cornell Law Review* 83:1771–1810.
- Blume, John H., Theodore Eisenberg, and Martin T. Wells. 2004. "Explaining death row's population and racial composition." *Journal of Empirical Legal Studies* 1:165–207.
- Blume, John H., Stephen P. Garvey, and Sheri L. Johnson. 2001. "Future dangerousness in capital cases: Always 'at issue'." *Cornell Law Review* 86:397–410.

- Blumer, Herbert. 1955. "Reflections on theory of race relations." Pp. 3–21 in Race Relations in World Perspective, edited by A. W. Lind. Honolulu, Hawaii: University of Hawaii Press.
- —. 1958. "Race prejudice as a sense of group position." *Pacific Sociological Review* 1:3–7.
- —. 1969. *Symbolic Interactionism: Perspective and Method*. Englewood Cliffs, N.J.: Prentice Hall.
- Blumstein, Alfred. 1993. "Racial disproportionality of U.S. prison populations revisited." University of Colorado Law Review 64:743–760.
- Blumstein, Alfred, Jacqueline Cohen, Susan E. Martin, and Michael Tonry (Eds.). 1983. *Research on Sentencing: The Search for Reform, Volume I.* Washington, D.C.: National Academy Press.
- Bobo, Lawrence. 1999. "Prejudice as group position: Microfoundations of a sociological approach to racism and race relations." *Journal of Social Issues* 55:445–472.
- Bobo, Lawrence and Devon Johnson. 2004. "A taste for punishment: Black and white Americans' view on the death penalty and the war on drugs." *DuBois Review* 1:151–180.
- Bodenhausen, Galen V. 1990. "Stereotypes as judgmental heuristics: Evidence of circadian variations in discrimination." *Psychological Science* 1:319–322.
- Bodenhausen, Galen V. and Meryl Lichtenstein. 1987. "Social stereotypes and information-processing strategies: The impact of task complexity." *Journal of Personality and Social Psychology* 52:871–880.
- Bodenhausen, Galen V., Lori A. Sheppard, and Geoffrey P. Kramer. 1994. "Negative affect and social judgment: The differential impact of anger and sadness." *European Journal of Social Psychology* 24:45–62.
- Bodenhausen, Galen V. and Robert S. Wyer, Jr. 1985. "Effects of stereotypes on decision-making and information-processing strategies." *Journal of Personality and Social Psychology* 48:267–282.
- Bollen, Kenneth A. 1989. *Structural Equations with Latent Variables*. New York: John Wiley and Sons.
- Bonacich, Edna. 1991. "Class approaches to ethnicity and race." Pp. 59–76 in *Majority* and *Minority: The Dynamics of Race and Ethnicity*, edited by N. R. Yetman. Boston, Mass.: Allyn and Bacon.

- Bonilla-Silva, Eduardo. 1997. "Rethinking racism: Toward a structural interpretation." *American Sociological Review* 62:465–480.
- —. 1999. "The essential social fact of race: Reply to Loveman." American Sociological Review 64:899–906.
- —. 2003. Racism Without Racists: Color-Blind Racism and the Persistence of Racial Inequality in the United States. Lanham, Md.: Roman and Littlefield.
- Borg, Marian J. 1992. "Conflict management in the modern world-system." *Sociological Forum* 7:261–282.
- —. 1997. "The southern subculture of punitiveness? Regional variation in support for capital punishment." *Journal of Research in Crime and Delinquency* 34:25–45.
- —. 1998. "Vicarious homicide victimization and support for capital punishment: A test of Black's theory of law." *Criminology* 36:537–567.
- —. 1999. "Using violence as social control: Applying a theory of conflict management to juvenile disputes." *Florida Journal of Law and Public Policy* 10:313–339.
- —. 2001. "Mobilizing law in urban areas: The social structure of homicide clearance rates." *Law and Society Review* 35:435–466.
- Borg, Marian J. and William P. Arnold, III. 1997. "Social monitoring as social control: The case of drug testing in a medical workplace." *Sociological Forum* 12:441–460.
- Boritch, Helen. 1992. "Gender and criminal court outcomes: An historical analysis." *Criminology* 30:293–325.
- Bourdieu, Pierre. [1979] 1984. *Distinction: A Social Critique of the Judgment of Taste*. Cambridge, Mass.: Harvard University Press.
- —. [1992] 1996. *The Rules of Art: Genesis and Structure of the Literary Field*. Stanford, Calif.: Stanford University Press.
- Bourke, Richard, Joe Hingston, and Joel Devine. 2003. "Black Strikes: A Study of the Racially Disparate Use of Peremptory Challenges by the Jefferson Parish District Attorney's Office." Louisiana Crisis Assistance Center, New Orleans, La.
- Bowers, William J. 1983. "The pervasiveness of arbitrariness and discrimination under Post-*Furman* capital statutes." *Journal of Criminal Law and Criminology* 74:1067–1100.

- —. 1984. Legal Homicide: Death as Punishment in America, 1864-1982. Boston, Mass.: Northeastern University Press.
- —. 1993. "Capital punishment and contemporary values: People's misgivings and the court's misperceptions." *Law and Society Review* 27:157–176.
- Bowers, William J. and Wanda D. Foglia. 2003. "Still singularly agonizing: Law's failure to purge arbitrariness from capital sentencing." *Criminal Law Bulletin* 39:51–86.
- Bowers, William J. and Glenn L. Pierce. 1975. "The illusion of deterrence in Isaac Ehrlich's research on capital punishment." *Yale Law Journal* 85:187–208.
- —. 1980. "Arbitrariness and discrimination under the Post-Furman capital statutes." Crime and Delinquency 26:563–635.
- Bowers, William J., Marla Sandys, and Benjamin D. Steiner. 1998. "Foreclosed impartiality in capital sentencing: Jurors' predispositions, guilt-trial experience, and premature decision making." *Cornell Law Review* 83:1476–1556.
- Bowers, William J. and Benjamin D. Steiner. 1999. "Death by default: An empirical demonstration of false and forced choices in capital sentencing." *Texas Law Review* 77:605–717.
- Bowers, William J., Benjamin D. Steiner, and Marla Sandys. 2001. "Death sentencing in Black and White: An empirical analysis of the role of jurors' race and jury racial composition." *University of Pennsylvania Journal of Constitutional Law* 3:171– 274.
- Box, George E. P. 1976. "Science and statistics." *Journal of the American Statistical Association* 71:791–799.
- Boylan, Richard T. 2005. "What do prosecutors maximize? Evidence from the careers of U.S. Attorneys." *American Law and Economics Review* 7:379–402.
- Braithwaite, John. 1989. *Crime, Shame and Reintegration*. Cambridge: Cambridge University Press.
- Braithwaite, John and David Biles. 1980. "Empirical verification and Black's *The Behavior of Law*." *American Sociological Review* 45:334–338.
- Braithwaite, Richard B. 1953. Scientific Explanation: A Study of the Function of Theory, Probability and Law in Science. Cambridge: Cambridge University Press.
- Brame, Robert and Raymond Paternoster. 2003. "Missing data problems in criminological research: Two case studies." *Journal of Quantitative Criminology* 19:55–78.

- Branch, Taylor. 1988. *Parting the Waters: America in the King Years, 1954-63*. New York: Simon and Schuster.
- Breiman, Leo. 2001. "Random Forests." Machine Learning 45:5-32.
- Breiman, Leo, Jerome H. Friedman, Richard A. Olshen, and Charles J. Stone. 1984. *Classification and Regression Trees*. Monterey, Calif.: Wadsworth.
- Breslow, Norman D. and David G. Clayton. 1993. "Approximate inference in generalized linear mixed models." *Journal of the American Statistical Association* 88:9–25.
- Breyer, Stephen G. 2005. *Active Liberty: Interpreting Our Democratic Constitution*. New York: Knopf.
- Bright, Stephen B. 1990. "Death by lottery--procedural bar of constitutional claims in capital cases due to inadequate representation of indigent defendants." *West Virginia Law Review* 92:679–725.
- —. 1992. "In defense of life: Enforcing the Bill of Rights on behalf of poor, minority and disadvantaged persons facing the death penalty." *Missouri Law Review* 57:849– 870.
- —. 1994. "Counsel for the poor: The death sentence not for the worst crime but the worst lawyer." Yale Law Journal 103:1835–1883.
- —. 1995a. "Discrimination, death and denial: The tolerance of racial discrimination in infliction of the death penalty." *Santa Clara Law Review* 35:433–483.
- —. 1995b. "The politics of crime and the death penalty: Not 'soft on crime,' but 'hard on the Bill of Rights'." *Saint Louis University Law Journal* 39:479–503.
- —. 1997. "Political attacks on the judiciary: Can justice be done amid efforts to intimidate and remove judges from office for unpopular decisions?" New York University Law Review 72:308–330.
- —. 1998. "Can judicial independence be attained in the South? Overcoming history, elections, and misperceptions about the role of the judiciary." *Georgia State University Law Review* 14:817–860.
- —. 2001. "Will the death penalty remain alive in the Twenty-First Century?: International norms, discrimination, arbitrariness and the risk of executing the innocent." *Wisconsin Law Review* 2001:1–27.

- Bright, Stephen B. and Patrick J. Keenan. 1995. "Judges and the politics of death: Deciding between the Bill of Rights and the next election in capital cases." *Boston University Law Review* 75:759–835.
- Britt, Chester L. 2000. "Social context and racial disparities in punishment decisions." *Justice Quarterly* 17:707–732.
- Brittan, Arthur. 1981. "The symbolic dimension of law and social control." Pp. 167–185 in *Sociological Approaches to Law*, edited by A. Podgorecki and C. J. Whelan. New York: St. Martin's Press.
- Brock, Dean, Nigel Cohen, and Jon R. Sorensen. 2000. "Arbitrariness in the imposition of the death sentences in Texas: An analysis of four counties by offense seriousness, race-of-victim, and race-of-offender." *American Journal of Criminal Law* 28:43–71.
- Bronowski, Jacob. 1958. Science and Human Values. New York: Harper and Row.
- Brooks, Richard R.W. and Steven Raphael. 2002. "Life terms or death sentences: The uneasy relationship between judicial elections and capital punishment." *Journal of Criminal Law and Criminology* 92:609–640.
- Brown, Michael K., Martin Carnoy, Elliot Currie, Troy Duster, David B. Oppenheimer, Marjorie M. Shultz, and David Wellman. 2003. *White-Washing Race: The Myth of a Color-Blind Society*. Berkeley and Los Angeles, Calif.: University of California Press.
- Brown, Ryan A. and George J. Armelagos. 2001. "Apportionment of racial diversity: A review." *Evolutionary Anthropology* 10:34–40.
- Brownlee, Shannon, Dan McGraw, and Jason Vest. 1997. "The place for vengeance." U.S. News and World Report, June 16, pp. 24–32.
- Bruck, David, Dick Burr, and Kevin McNally. 2006. "Summaries of Cases Authorized for the Death Penalty, 1988 - 2006: Federal Capital Defendants Who Were Found Not Guilty of the Capital Charge or Were Innocent." Lexington, Va.: Federal Death Penalty Resource Council. Retrieved July 17, 2006 (http://www.capdefnet.org/pdf\_library/5auth\_acquittal.pdf).
- Bryk, Anthony S., Valerie E. Lee, and Peter B. Holland. 1993. *Catholic Schools and the Common Good*. Cambridge, Mass.: Harvard University Press.
- Burtt, Edwin A. 1954. *The Metaphysical Foundations of Modern Science*, 2d ed. Garden City, N.Y.: Doubleday.

- Bushway, Shawn D. and Anne M. Piehl. 2001. "Judging judicial discretion: Legal factors and racial discrimination in sentencing." *Law and Society Review* 35:733–764.
- Bushway, Shawn, Brian D. Johnson, and Lee Ann Slocum. 2007. "Is the magic still there? The use of the Heckman Two-Step correction for selection bias in criminology." *Journal of Quantitative Criminology* 23:151–178.
- Cahill, Fred V. 1952. *Judicial Legislation: A Study in American Legal Theory*. New York: Ronald Press Co.
- Cameron, A. Colin and Pravin K. Trivedi. 2009. *Microeconometrics Using Stata*. College Station, Tex.: Stata Press.
- Caplow, Theodore. 1955. "The definition and measurement of ambiences." *Social Forces* 34:28–33.
- Cappellari, Lorenzo and Stephen P. Jenkins. 2003. "Multivariate probit regression using simulated maximum likelihood." *Stata Journal* 3:278-294.
- Cardozo, Benjamin N. 1947. *The Selected Writings of Benjamin Nathan Cardozo*, edited by M. E. Hall. New York: Fallon Law Book Co.
- Carrington, Frank G. 1978. Neither Cruel Nor Unusual. New Rochelle, N.Y.: Arlington House.
- Carter, James E. 1976. *Presidential Campaign, 1976*, vol. 1. Washington, D.C.: U.S. Government Printing Office.
- Carter, Robert M. and A. Lamont Smith. 1969. "The death penalty in California: A statistical and composite portrait." *Crime and Delinquency* 15:62–76.
- Cartwright, Nancy. 1983. How the Laws of Physics Lie. Oxford: Clarendon Press.
- Cassell, Paul G. 1999. "The guilty and the "innocent": an examination of alleged cases of wrongful conviction from false confessions." *Harvard Journal of Law and Public Policy* 22:523–603.
- —. 2003. "In defense of the death penalty." Pp. 183–217 in *Debating the Death Penalty:* Should America Have Capital Punishment?, edited by H. A. Bedau and P. G. Cassell. New York: Oxford University Press.
- CBS News Polls. 2005. "Looking Back at John Paul II and Ahead To a New Pope." New York: Columbia Broadcasting System, Inc. Retrieved April 17, 2005 (http://www.cbsnews.com/htdocs/pdf/pope poll 41705.pdf).

- Center for American Women and Politics. 2004. "Sex differences in voter Turnout." Eagleton Institute of Politics, The State University of New Jersey, New Brunswick, N.J.
- Cernkovich, Stephen A., Peggy C. Giordano, and Jennifer L. Rudolph. 2000. "Race, crime, and the American dream." *Journal of Research in Crime and Delinquency* 3:131–170.
- Cerulo, Karen A. 2002. "The behavior of culture...courtesy of Donald Black." Contemporary Sociology 31:652–655.
- Chadbourn, James H. 1933. *Lynching and the Law*. Chapel Hill, N.C.: University of North Carolina Press.
- Chambliss, William J. 1999. Power, Politics and Crime. Boulder, Colo.: Longview Press.
- Chambliss, William J. and Robert B. Seidman. 1971. Law, Order, and Power. Reading, Mass.: Addison-Wesley.
- Chesney-Lind, Meda. 1989. "Girls' crime and woman's place: Toward a feminist model of female delinquency." *Crime and Delinquency* 35:5–29.
- Chiricos, Theodore G. and Charles Crawford. 1995. "Race and imprisonment: A contextual assessment of the evidence." Pp. 281–309 in *Ethnicity, Race, and Crime: Perspectives Across Time and Place, New Directions in Crime and Justice Studies*, edited by D. F. Hawkins. Albany, N.Y.: SUNY Press.
- Clarke, James W. 1998. "Without fear or shame: Lynching, capital punishment and the subculture of violence in the American South." *British Journal of Political Science* 28:269–289.
- Clear, Todd R. 2007. Imprisoning Communities: How Mass Incarceration Makes Disadvantaged Neighborhoods Worse. New York: Oxford University Press.
- Cleves, Mario A. 2002. "Comparing areas under receiver operating characteristic curves from two or more probit or logit models." *Stata Journal* 2:301-313.
- Clogg, Clifford C. and Edward S. Shihadeh. 1994. *Statistical Models for Ordinal Variables*. Newbury Park, Calif.: Sage.
- Cloninger, Dale O. and Roberto Marchesini. 2001. "Execution and deterrence: A quasicontrolled experiment." *Applied Economics* 33:569–576.
- Cloward, Richard and Lloyd E. Ohlin. 1960. *Delinquency and Opportunity*. Glencoe, Ill.: Free Press.

- Cochran, John K. and Mitchell B. Chamblin. 2000. "Deterrence and brutalization: The dual effects of executions." *Justice Quarterly* 17:685–706.
- Cochran, John K., Mitchell B. Chamblin, and Mark Seth. 1994. "Deterrence or brutalization? An impact assessment of Oklahoma's return to capital punishment." *Criminology* 32:107–134.
- Cohen, I. Bernard. 1985. *Revolution in Science*. Cambridge, Mass.: Harvard University Press.
- Cohen, Jacob, Patricia Cohen, Stephen G. West, and Leona S. Aiken. 2003. *Applied Multiple Regression/Correlation Analysis for the Behavioral Sciences*, 3d ed. Mahwah, N.J.: Erlbaum.
- Cohen, Lawrence E. and Marcus Felson. 1979. "Social change and crime rate trends: A routine activities approach." *American Sociological Review* 44:588–608.
- Colbert, Douglas L., Raymond Paternoster, and Shawn D. Bushway. 2002. "Do attorneys really matter? The empirical and legal case for the right to counsel at bail." *Cardozo Law Review* 23:1719–1793.
- Cole, David. 1994. "Death, discrimination and delay." Recorder, May 11, p. 7.
- Collins, Randall. 1975. *Conflict Sociology: Toward an Explanatory Science*. New York: Academic Press.
- —. 1981. "On the microfoundations of macrosociology." American Journal of Sociology 86:984–1014.
- —. 2002. "Black's contributions to a general theory of conflict." Contemporary Sociology 31:655–658.
- Conley, Dalton. 1999. *Being Black, Living in the Red: Race, Wealth, and Social Policy in America*. Berkeley and Los Angeles, Calif.: University of California Press.
- —. 2001. "Decomposing the black-white wealth gap: The role for parental resources, inheritance, and investment dynamics." *Sociological Inquiry* 71:36–66.
- Constable, Marianne. 1994. "Genealogy and Jurisprudence: Nietzsche, nihilism, and the social scientification of law." *Law and Society Review* 19:551–590.

- Cook, Philip J. and Donna B. Slawson. 1993. "The Costs of Processing Murder Cases in North Carolina." Terry Sanford Institute of Public Policy, Duke University, Durham, N.C.
- Cook, Rhonda. 1996. "Georgia cases have set legal precedent." *Atlanta Journal-Constitution*, June 2, p. C4.
- Cook, Thomas D. and Donald T. Campbell. 1979. *Quasi-Experimentation: Design and Analysis Issues For Field Settings*. Boston, Mass.: Houghton Mifflin Company.
- Cooley, Charles H. [1902] 1964. *Human Nature and the Social Order*. New York: Schocken Books.
- Cooney, Mark. 1986. "Behavioural sociology of law: A defence." *Modern Law Review* 49:262–271.
- -. 1992. "Racial discrimination in arrest." Virginia Review of Sociology 1:99-119.
- —. 1993. "Why is economic analysis so appealing to law professors? (Review essay)." Stanford Law Review 45:2211–2230.
- -. 1997a. "The decline of elite homicide." Criminology 35:381-407.
- —. 1997b. "From warre to tyranny: Lethal conflict and the state." *American Sociological Review* 62:316–338.
- —. 1998. *Warriors and Peacemakers: How Third Parties Shape Violence*. New York: New York University Press.
- —. 2002. "Still paying the price of heterodoxy: *The Behavior of Law* a quarter-century on." *Contemporary Sociology* 31:658–661.
- 2003a. "Missing the mark: Reflections on Greenberg's comment." Criminology 41:1419–1426.
- —. 2006a. "The criminological potential of pure sociology." Crime, Law, and Social Change 46:51–63.
- —. 2006b. "How to Commit a Perfect Murder." Working Paper, University of Iowa Workshop on Theoretical Analysis, Iowa City, Iowa.

- Cooney, Mark and Scott Phillips. 2002. "Typologizing violence: A Blackian perspective." *International Journal of Sociology and Social Policy* 22:75–108.
- Copes, Heith, Kent R. Kerley, Karen A. Mason, and Judy van Wyk. 2001. "Reporting behavior of fraud victims and Black's theory of law: An empirical assessment." *Justice Quarterly* 18:343–363.
- Cornish, Derek and Ronald Clarke (Eds.). 1986. *The Reasoning Criminal: Rational Choice Perspectives on Offending*. New York: Springer-Verlag.
- Cose, Ellis. 1993. The Rage of a Privileged Class. New York: Harper-Collins.
- Cranmer, Skyler J. 2007. "Hot Deck imputation for discrete data." Working Paper, Research Workshop in Applied Statistics, Institute for Quantitative Social Science, Harvard University, Cambridge, Mass.
- Crawford, Charles, Theodore G. Chiricos, and Gary Kleck. 1998. "Race, racial threat, and sentencing of habitual offenders." *Criminology* 36:481–511.
- Crew, Keith B. 1991. "Sex differences in criminal sentencing: Chivalry or patriarchy." *Justice Quarterly* 8:59–84.
- Cronbach, Lee J. 1951. "Coefficient alpha and the internal structure of a test." *Psychometrika* 16:297–334.
- Cross, William E., Jr. 2003. "Tracing the historical origins of youth delinquency and violence: Myths and realities about black culture." *Journal of Social Issues* 59:67–82.
- Crouch, Ronald L. 1979. *Human Behavior: An Economic Approach*. North Scituate, Mass.: Duxbury Press.
- Crowell, Linda F. 2004. "Weak ties: A mechanism for helping women expand their social networks and increase their capital." *Social Science Journal* 41:15–28.
- Curriden, Mark and Leroy Phillips, Jr. 1999. Contempt of Court: The Turn-of-the-Century Lynching that Launched a Hundred Years of Federalism. New York: Anchor Books.
- Currie, Elliot. 1998. Crime and Punishment in America: Why Solutions to America's Most Stubborn Social Crisis Have Not Worked--and What Will. New York: Henry Holt.
- D'Alessio, Stewart J. and Lisa Stolzenberg. 1993. "Socioeconomic status and the sentencing of the traditional offender." *Journal of Criminal Justice* 21:61–77.

- D'Avella, Joanna H. 2006. "Death row for child rape? Cruel and unusual punishment under the Roper-Atkins 'Evolving Standards of Decency' framework." *Cornell Law Review* 92:129–156.
- Daly, Kathleen. 1987. "Structure and practice of familial-based justice in a criminal court." *Law and Society Review* 21:267–290.
- —. 1989. "Rethinking judicial paternalism: Gender, work-family relations, and sentencing." *Gender and Society* 3:9–36.
- Dannefer, Dale. 1984. "Who signs the complaint?' Relational distance and the juvenile justice process." *Law and Society Review* 18:249–272.
- David, Paul A., Herbert G. Gutman, Richard Sutch, Peter Temin, and Gavin Wright. 1976. *Reckoning With Slavery: A Critical Study in the Quantitative History of American Negro Slavery*. New York: Oxford University Press.
- David, Rene and John E. C. Brierly. 1985. *Major Legal Systems in the World Today*, 3d ed. London: Stevens.
- Davies, Kevin. 2005. "First Base: The Race Prescription Card." *Bio-IT World*. Retrieved June 5, 2007 (http://www.bio-itworld.com/archive/012105/firstbase.html).
- Davis, Angela J. 1998. "Prosecution and race: The Power of discretion." *Fordham Law Review* 67:13–66.
- Davis, David B. 1957. "The movement to abolish capital punishment in America, 1787-1861." *American Historical Review* 63:23–46.
- Dawson, Michael C. 1995. *Behind the Mule: Race and Class in African-American Politics*. Princeton, N.J.: Princeton University Press.
- —. 2001. Black Visions: The Roots of Contemporary African-American Political Ideologies. Chicago, Ill.: University of Chicago Press.
- Death Penalty Information Center. 2008. "Facts About the Death Penalty." Death Penalty Information Center, Washington, D.C.
- Delaney, David. 1998. Race, Place, and the Law. Austin, Tex.: University of Texas Press.
- Dempster, Arthur P., Nan M. Laird, and Donald B. Rubin. 1977. "Maximum likelihood from incomplete data via the EM algorithm." *Journal of the Royal Statistical Society* (Series B) 39:1–38.

- Demuth, Stephen. 2003. "Racial and ethnic differences in pretrial release decisions and outcomes: A comparison of Hispanic, black, and white felony arrestees." *Criminology* 41:873–908.
- Denno, Deborah W. 1994. "Is electrocution and unconstitutional method of execution? The engineering of death over the century." *William and Mary Law Review* 35:551–692.
- Dezhbakhsh, Hashem, Paul H. Rubin, and Joanna M. Shepherd. 2003. "Does Capital Punishment Have a Deterrent Effect? New Evidence from Postmoratorium Panel Data." *American Law and Economics Review* 5:344–376.
- Dieter, Richard C. 1993. "Sentencing for Life: Americans Embrace Alternatives to the Death Penalty." Death Penalty Information Center, Washington, D.C.
- —. 1994. "Millions Misspent: What Politicians Don't Say about the High Costs of the Death Penalty." Death Penalty Information Center, Washington, D.C.
- —. 1995. "With Justice for Few: The Growing Crisis in Death Penalty Representation." Death Penalty Information Center, Washington, D.C.
- —. 1996. "Twenty Years of Capital Punishment: A Re-Evaluation." Death Penalty Information Center, Washington, D.C.
- —. 2003. "International influence on the death penalty in the United States." *Foreign Service Journal* 80:31–38.
- —. 2004. "Innocence and the Crisis in the American Death Penalty." Death Penalty Information Center, Washington, D.C.
- —. 2005. "Blind Justice: Juries Deciding Life and Death With Only Half the Truth." Death Penalty Information Center, Washington, D.C.
- Dilulio, John J. 1996. "My black crime problem and ours." *City Journal*, Spring, pp. 14–28.
- DiPrete, Thomas A. and Gregory M. Eirich. 2006. "Cumulative advantage as a mechanism for inequality: A review of theoretical and empirical developments." *Annual Review of Sociology* 32:271–297.
- DiPrete, Thomas A. and Jerry D. Forristal. 1994. "Multilevel Models: Methods and Substance." *Annual Review of Sociology* 20:331–357.
- Dixon, Jo. 1995. "The organizational context of criminal sentencing." *American Journal* of Sociology 100:1157–1198.

- Dodson, Edward J. 2003. "The Wealth of Our Nation and Our Cities." Council of Georgist Organizations, July, Bridgeport, Conn.
- Donohue, John J. and Justin Wolfers. 2005. "Uses and abuses of empirical evidence in the death penalty debate." *Stanford Law Review* 58:791–846.
- —. 2006. "The death penalty: No evidence for deterrence." *Economists' Voice*, Berkeley, Calif.: Berkeley Electronic Press. Retrieved May 9, 2006 (www.bepress.com/ev/vol3/iss5/art3/).
- Donziger, Steven R. 1996. The Real War on Crime: The Report of the National Criminal Justice Commission. New York: Harper-Collins.
- Dowding, Keith. 1996. Power. Minneapolis, Minn.: University of Minnesota Press.
- Doyle, Daniel P. and David F. Luckenbill. 1991. "Mobilizing law in response to collective problems: A test of Black's Theory of Law." *Law and Society Review* 25:103–116.
- Drake, St. Clair and Horace R. Cayton. 1945. *The Black Metropolis: A Study of Negro Life in a Northern City*. New York: Harcourt, Brace and World.
- Driggs, Kenneth. 1993. "A current of electricity sufficient in intensity to cause immediate death: A Pre-Furman history of Florida's electric chair." Stetson Law Review 22:1169–1208.
- Du Bois, William Edward Burghardt (Ed.). 1904. Some Notes on Negro Crime, Particularly in Georgia. Atlanta, Ga.: Atlanta University Press.
- —. [1940] 1968. Dusk of Dawn: An Essay Toward an Autobiography of a Race Concept. New York: Schocken.
- Dubin, Jeffrey A. and R. Douglas Rivers. 1990. "Selection bias in linear regression, logit and probit models." Pp. 410–442 in *Modern Methods of Data Analysis*, edited by J. Fox and J. S. Long. Newbury Park, Calif.: Sage.
- Dudley, Richard G., Jr. and Pamela Blume Leonard. 2008. "Getting it right: Life history investigation as the foundation for a reliable mental health assessment." *Hofstra Law Review* 36:963-988.
- Dugger, Karen. 2001. "Women in higher education in the United States: Has there been progress?" *International Journal of Sociology and Social Policy* 21:118–130.
- Dumont, Louis. 1981. *Homo Hierarchius: The Caste System and Its Implications*, 2d ed. Chicago, Ill.: University of Chicago Press.

- Dumont, Richard G. and William J. Wilson. 1967. "Aspects of concept formation, explication, and theory construction in sociology." *American Sociological Review* 32:985–995.
- Durkheim, Emile. [1893] 1947. *The Division of Labor in Society*. Translated by G. Simpson. New York: Free Press.
- —. [1897] 1951. Suicide: A Study in Sociology. Translated by J. A. Spaulding and G. Simpson. New York: Free Press.
- -. [1895] 1962. Rules of Sociological Method. Glencoe, Ill.: Free Press.
- —. [1900] 1969. "Two laws of penal evolution." University of Cincinnati Law Review 38:32–60.
- [1897] 1994. "Review of Antonio Labriola, Essais sur la conception matérialiste de l'histoire." Pp. 123–130 in Emile Durkheim on Institutional Analysis, Reprint ed., edited by M. Traugott. Chicago, Ill.: University of Chicago Press.
- -. [1912] 1995. The Elementary Forms of Religious Life. New York: Free Press.
- Duster, Troy. 1990. Backdoor to Eugenics. New York: Routledge.
- —. 2003. "Buried alive: The concept of race in science." Pp. 258–277 in *Genetic Nature/Culture: Anthropology and Science Beyond the Two-Culture Divide*, edited by A. H. Goodman, D. Heath, and M. S. Lindee. Berkeley and Los Angeles, Calif.: University of California Press.
- Eberhardt, Jennifer L., Paul G. Davies, Valerie J. Purdie-Vaughns, and Sheri Lynn Johnson. 2006. "Looking deathworthy: Perceived stereotypicality of black defendants predicts capital-sentencing outcomes." *Psychological Science* 17:383– 386.
- Edelman, Peter, Harry J. Holzer, and Paul Offner. 2006. *Reconnecting Disadvantaged Young Men.* Washington, D.C.: Urban Institute Press.
- Edling, Christofer R. 2002. "Mathematics in sociology." *Annual Review of Sociology* 28:197–220.
- Edwards, Yohance C. and Jennifer Ahern. 2004. "Unequal treatment in state supreme courts: Minority and city schools in education finance reform litigation." *New York University Law Review* 79:326–366.

- Ehrlich, Eugene. [1913] 1936. Fundamental Principles of the Sociology of Law. Cambridge, Mass.: Harvard University Press.
- Ehrlich, Isaac. 1975. "The deterrent effect on capital punishment: A question of life and death." *American Economic Review* 65:397–417.
- —. 1977. "Capital punishment and deterrence: Some further thoughts and evidence." *Journal of Political Economy* 85:741–788.
- Ehrlich, Isaac and Joel C. Gibbons. 1977. "On the measurement of the deterrent effect of capital punishment and the theory of deterrence." *Journal of Legal Studies* 6:35–50.
- Ehrlich, Issac and Zhiqiang Liu. 1999. "Sensitivity analyses of the deterrence hypothesis: Let's keep the 'econ' in econometrics." *Journal of Law and Economics* 42:455–487.
- Eisenberg, Theodore, Stephen P. Garvey, and Martin T. Wells. 1998. "But was he sorry? The role of remorse in capital sentencing." *Cornell Law Review* 83:1599–1637.
- —. 2001a. "The deadly paradox of capital jurors." *Southern California Law Review* 74:371–397.
- —. 2001b. "Forecasting life and death: Juror race, religion, and attitude toward the death penalty." *Journal of Legal Studies* 30:277–313.
- —. 2003. "Victim characteristics and victim impact evidence in South Carolina capital cases." Cornell Law Review 88:306–342.
- Eisenberg, Theodore and Martin T. Wells. 1993. "Deadly confusion: Juror instructions in capital cases." *Cornell Law Review* 79:1–17.
- Eisenstein, James, Roy B. Flemming, and Peter F. Nardulli. 1988. *The Contours of Justice: Communities and Their Courts*. Boston, Mass.: Little, Brown and Company.
- Ekland-Olson, Sheldon. 1988. "Structured discretion, racial bias and the death penalty: The first decade after *Furman* in Texas." *Social Science Quarterly* 69:853–873.
- Ellickson, Robert C. 1991. Order Without Law: How Neighbors Settle Disputes. Cambridge, Mass.: Harvard University Press.
- Elliott-Engel, Amaris. 2008. "Justices OK juror challenges involving prosecutor from controversial training tape." *The Legal Intelligencer*, Aug. 5.

- Ellsworth, Phoebe C. 1988. "Unpleasant facts: The Supreme Court's response to empirical research on capital punishment." Pp. 177–211 in *Challenging Capital Punishment: Legal and Social Science Approaches*, vol. 24, edited by K. C. Haas and J. A. Inciardi. Newbury Park, Calif.: Sage.
- Ellsworth, Phoebe C. and Samuel R. Gross. 1994. "Hardening of the attitudes: Americans' views on the death penalty." *Journal of Social Issues* 50:19–52.
- Ellsworth, Phoebe C. and Lee Ross. 1983. "Public opinion and capital punishment: A close examination of the views of abolitionists and retentionists." *Crime and Delinquency* 29:116–169.
- Emerson, Richard. 1962. "Power-dependence relations." *American Sociological Review* 40:252–257.
- Emmelman, Debra S. 1994. "The effects of social class on the adjudication of criminal cases: Class-linked behavior tendencies, common sense, and the interpretive procedures of court-appointed defense attorneys." *Symbolic Interactionism* 17:1–20.
- Engen, Rodney L., Randy R. Gainey, Robert D. Crutchfield, and Joseph G. Weis. 2003. "Discretion and disparity under sentencing guidelines: The role of departures and structured sentencing alternatives." *Criminology* 41:99–130.
- Erlanger, Howard S. 1974. "The empirical status of the subculture of violence thesis." *Social Problems* 22:280–291.
- —. 1976. "Is there a 'subculture of violence' in the South?" *Journal of Criminal Law and Criminology* 66:483–490.
- Espeland, Wendy Nelson and Berit Irene Vannebo. 2007. "Accountability, quantification, and law." *Annual Review of Law and Social Science* 3:21–43.
- Espy, M. Watt and John Ortiz Smykla. 2004. *Executions in the United States, 1608-2002: The Espy File*. [MRDF]. Tuscaloosa, Ala.: University of Alabama [producer]. Ann Arbor, Mich.: Inter-university Consortium for Political and Social Research, University of Michigan [distributor].
- Evan, William M. 1990. Social Structure and Law: Theoretical and Empirical Perspectives. Newbury Park, Calif.: Sage.
- Everett, Ronald S. and Roger A. Wojtkiewicz. 2002. "Difference, disparity, and race/ethnic bias in federal sentencing." *Journal of Quantitative Criminology* 18:189–211.

- Ewick, Patricia and Susan Silbey. 2003. "Narrating social structure: Stories of resistance to legal authority." *American Journal of Sociology* 108:1328–1372.
- Ewing, Sally. 1987. "Formal justice and the spirit and capitalism: Max Weber's Sociology of Law." *Law and Society Review* 21:487–512.
- Fagan, Jeffrey. 2006. "Death and deterrence redux: Science, law and causal reasoning on capital punishment." *Ohio State Journal of Criminal Law* 4:255–320.
- Fagan, Jeffrey and Tracey L. Meares. 2000. "Punishment, Deterrence and Social Control: The Paradox of Punishment in Minority Communities." Working Paper No. 010, Columbia Law School Public Law and Legal Theory Group, New York.
- Fagan, Jeffrey and Valerie West. 2005. "The decline of the juvenile death penalty: Scientific evidence of evolving norms." *Journal of Criminal Law and Criminology* 95:427–500.
- Fagan, Jeffrey, Valerie West, and Jan Holland. 2003. "Reciprocal effects of crime and incarceration in New York City neighborhoods." *Fordham Urban Law Journal* 30:1551–1602.
- Failor, Peter. 2001. "Man gets life without parole for Adel murder." *Valdosta Daily Times*, June 27, p. A1.
- Fanon, Frantz. 1967. Black Skin, White Masks. New York: Grove.
- Farnworth, Margaret and Patrick M. Horan. 1980. "Separate justice: An analysis of race differences in court processes." *Social Science Research* 9:381–399.
- Farrell, Ronald A. and Victoria L. Swigert. 1978. "Prior offense record as a self-fulfilling prophecy." *Law and Society Review* 12:437–453.
- Farrington, David P. and A. M. Morris. 1983. "Sex, sentencing and reconviction." British Journal of Criminology 23:229–248.
- Feagin, Joe R. and Kathryn D. McKinney. 2003. *The Many Costs of Racism*. Landam, Md.: Rowan and Littlefield.
- Federal Reserve Board. 2007. Flow of Funds Accounts of the United States. Washington, D.C.: U.S. Government Printing Office.
- Feldman, Lisa, Vincent Schiraldi, and Jason Ziedenberg. 2001. Too Little, Too Late: President Clinton's Prison Legacy. Washington, D.C.: Justice Policy Institute.
- Felson, Marcus. 1994. Crime and Everyday Life: Insight and Implications for Society. Thousand Oaks, Calif.: Pine Forge Press.

- Field, Hubert S. and Nona J. Barnett. 1978. "Simulated jury trials: Students vs. 'real' people as jurors." *Journal of Social Psychology* 104:287–293.
- Figueira-McDonough, Josefina. 1985. "Gender differences in informal processing: A look at charge bargaining and sentence reduction in Washington, D.C." *Journal of Research in Crime and Delinquency* 22:101–133.
- Finkelstein, Michael O. 1978. *Quantitative Methods in Law: Studies in the Application of Mathematical Probability and Statistics to Legal Problems*. New York: Free Press.
- Fisek, M. Hamit. 1998. "The combining of status information." *Journal of Mathematical Sociology* 22:333–353.
- Fitchen, Janet. 1994. "Residential mobility among the rural poor." *Rural Sociology* 59:416–436.
- Fitzgerald, Robert and Phoebe C. Ellsworth. 1984. "Due process vs. crime control: Death qualification and jury attitudes." *Law and Human Behavior* 8:31–51.
- Fleury-Steiner, Benjamin. 2006. "Death in "Whiteface": Modern race minstrels, official lynching, and the culture of American Apartheid." Pp. 150–181 in *From Lynch Mobs to the Killing State: Race and the Death Penalty in America*, edited by C. J. Ogletree, Jr. and A. Sarat. New York: New York University Press.
- Fogel, Robert W. and Stanley L. Engerman. 1974. *Time on the Cross: The Economics of American Negro Slavery*. Boston, Mass.: Little, Brown and Company.
- Foley, Linda A. 1987. "Florida after the *Furman* decision: The effect of extra-legal factors on the processing of capital offense cases." *Behavioral Sciences and the Law* 5:457–465.
- Foley, Linda A. and Richard Powell. 1982. "The discretion of prosecutors, judges, and juries in capital cases." *Criminal Justice Review* 7:16–22.
- Foley, Michael A. 2003. Arbitrary and Capricious: The Supreme Court, the Constitution, and the Death Penalty. Wesport, Conn.: Praeger.
- Foner, Eric and Olivia Mahoney. 1995. *America's Reconstruction: People and Politics after the Civil War*. New York: Harper Collins.
- Forst, Brian E. 1983. "Capital punishment and deterrence: Conflicting evidence." *Journal* of Criminal Law and Criminology 74:927–942.

- Foster, E. Michael. 1997. "Instrumental variables for logistic regression: An illustration." *Social Science Research* 26:487–504.
- Fox, James A. 2004. "Missing data problems in the SHR: Imputing offender and relationship characteristics." *Homicide Studies* 8:214–254.
- —. 2005. Uniform Crime Reports [United States]: Supplemental Homicide Reports, 1976-2002. [MRDF]. Boston, Mass.: College of Criminal Justice, Northeastern University [producer]. Ann Arbor, Mich.: Inter-university Consortium for Political and Social Research, University of Michigan [distributor].
- Fox, James A. and Michael L. Radelet. 1989. "Persistent flaws in econometric studies of the deterrent effect of death penalty." *Loyola of Los Angeles Law Review* 23:29– 44.
- Frank, James and Brandon K. Applegate. 1998. "Assessing juror understanding of capital-sentencing instructions." *Crime and Delinquency* 44:412–433.
- Frank, Jerome. 1930. Law and the Modern Mind. New York: Brentano's Publishers.
- Frankford, David M. 1995. "Donald Black's *Social Structure of Right and Wrong:* Normativity without agents." *Law and Social Inquiry* 20:787–804.
- Frazier, Harriet C. 2006. *Death Sentences in Missouri, 1803-2005: A History and Comprehensive Registry of Legal Executions, Pardons, and Commutations.* Jefferson, N.C.: McFarland.
- Friedman, Milton. 1953. *Essays in Positive Economics*. Chicago, Ill.: University of Chicago Press.
- Friedrich, Robert J. 1982. "In defense of multiplicative terms in multiple regression equations." *American Journal of Political Science* 26:797–833.
- Frohmann, Lisa. 1997. "Convictability and discordant locales: Reproducing race, class, and gender ideologies in prosecutorial decision-making." *Law and Society Review* 31:531–556.
- Fryer, Roland G., Jr. and Steven D. Levitt. 2004. "The causes and consequences of distinctly black names." *The Quarterly Journal of Economics* 119:767–805.
- Fuchs, Stephen. 1992. "A sociological theory of objectivity." Science Studies 11:4-26.
- Gabbidon, Shaun L. 2001. "W.E.B. DuBois: Pioneering American criminologist." *Journal of Black Studies* 31:581–599.

- Garber, Steven, Steven Klepper, and Daniel Nagin. 1983. "The role of extralegal factors in determining criminal case disposition." Pp. 129–183 in *Research on Sentencing: The Search for Reform, Volume II*, edited by A. Blumstein, J. Cohen, S. Martin, and M. Tonry. Washington, D.C.: National Academy Press.
- Garfinkel, Harold. 1949. "Research note on inter- and intra-racial homicides." *Social Forces* 27:369–381.
- Garland, David. 1990. *Punishment and Modern Society: A Study in Social Theory*. Chicago, Ill.: University of Chicago Press.
- Garrett, Brandon L. 2008. "Judging Innocence." Columbia Law Review 108:55-142.
- Gartner, Rosemary and Ross Macmillan. 1995. "The effect of victim-offender relationship on reporting crimes of violence against women." *Canadian Journal* of Criminology 37:393–430.
- Garvey, Stephen P. 1998. "Aggravation and mitigation in capital cases: What do jurors think?" *Columbia Law Review* 98:1538–1578.
- —. 2000. "The emotional economy of capital sentencing." New York University Law Review 75:26–73.
- Garvey, Stephen P., Sheri L. Johnson, and Paul Marcus. 2000. "Correcting deadly confusion: Responding to jury inquiries in capital cases." *Cornell Law Review* 85:627–655.
- Gates, Henry Louis, Jr. and Cornel West. 2000. *The African American Century: How Black Americans Have Shaped Our Country*. New York: Free Press.
- Geerdes, Holly L. and Nikki Cox. 2005. "Death Penalty Law." *Mercer Law Review* 57:139–171.
- Gell-Mann, Murray. 1994. The Quark and the Jaguar: Adventures in the Simple and the Complex. New York: W. H. Freeman.
- Gelman, Andrew, James S. Liebman, Valerie West, and Alexander Kiss. 2004. "A Broken System: The persistent patterns of reversals of death sentences in the United States." *Journal of Empirical Legal Studies* 1:209–261.
- George, Lance, Jennifer Pinder, and Theresa Singleton. 2004. "Race, Place, and Housing: Housing Conditions in Rural Minority Communities." Housing Assistance Council, Washington, D.C.
- Georgia Bureau of Investigation. 2001. *Georgia Uniform Crime Reporting Statistics*. Decatur, Ga.: Georgia Bureau of Investigation.

- Georgia Department of Corrections, Office of Planning and Analysis. 2002. A History of the Death Penalty in Georgia, Executions by Year 1924-2001. Atlanta, Ga.: Department of Corrections.
- Gerth, Hans and C. Wright Mills. 1958. From Max Weber: Essays in Sociology. New York: Oxford University Press.
- Ghali, Moheb and Meda Chesney-Lind. 1986. "Gender bias and the criminal justice system: An empirical investigation." *Sociology and Social Research* 70:164–171.
- Gibbs, Jack P. 1966. "The sociology of law and normative phenomena." *American Sociological Review* 31:315–325.
- —. 1968. "Definitions of law and empirical questions." Law and Society Review 2:429–446.
- Gilens, Martin. 2000. Why Americans Hate Welfare: Race, Media, and the Politics of Antipoverty Policy. Chicago, Ill.: University of Chicago Press.
- Gillers, Stephen. 1980. "Deciding who dies." University of Pennsylvania Law Review 129:1–124.
- Gilmore, David D. 1996. "Above and below: Toward a social geometry of gender." *American Anthropologist* 98:54–66.
- Gilroy, Paul. 1993. *The Black Atlantic: Modernity and Double Consciousness*. Cambridge, Mass.: Harvard University Press.
- Ginzburg, Carlo. 1976. "High and low: The theme of forbidden knowledge in the Sixteenth and Seventeenth centuries." *Past and Present* 73:28–41.
- Gittleman, Maury and Edward N. Wolff. 2004. "Racial differences in patterns of wealth accumulation." *Journal of Human Resources* 39:193–227.
- Glaberson, William. 2004. "4-3 ruling effectively halts death penalty in New York." *New York Times*, June 25, p. A1.
- Glaeser, Edward L., Daniel P. Kessler, and Anne Morrison Piehl. 2000. "What do prosecutors maximize? An analysis of the federalization of drug crimes." *American Law and Economics Review* 2:259–290.
- Glaser, Barney G. and Anselm L. Strauss. 1967. *The Discovery of Grounded Theory: Strategies for Qualitative Research*. New York: Aldine de Gruyter.

- Glass, Robert F. 2006. "*Roper v. Simmons*: A dead-end for the juvenile death penalty." *Mercer Law Review* 57:1341–1355.
- Gleason, Philip M. and Glen C. Cain. 2004. "Earnings of black and white youth and their relation to poverty." Pp. 449–475 in *Race, Poverty, and Domestic Policy*, edited by C. M. Henry. New Haven, Conn.: Yale University Press.
- Goffman, Erving. 1963. *Stigma: Notes on the Management of Spoiled Identity*. Englewood Cliffs, N.J.: Prentice-Hall.
- —. 1971. Relations in Public: Microstudies of the public order. New York: Basic Books.
- Goldfield, Michael. 1997. *The Color of Politics: Race and the Mainsprings of American Politics*. New York: New Press.
- Goldstein, Harvey. 1991. "Nonlinear multilevel models with an application to discrete response data." *Biometrika* 78:45–51.
- Gordon, Sanford C. and Gregory A. Huber. 2002. "Citizen oversight and the electoral incentives of criminal prosecutors." *American Journal of Political Science* 46:334–351.
- Gornick, Janet C. 2004. "Women's economic outcomes, gender inequality and public policy: Findings from the Luxembourg Income Study." *Socio-Economic Review* 2:213–238.
- Gottfredson, Michael R. and Michael J. Hindelang. 1979. "A study of the behavior of law." *American Sociological Review* 44:3–18.
- Gottfredson, Michael R. and Travis Hirschi. 1990. *A General Theory of Crime*. Stanford, Calif.: Stanford University Press.
- Gottlieb, Gerald. 1961. "Testing the death penalty." *Southern California Law Review* 34:268–281.
- Gould, Roger V. 2002. "The origins of status hierarchies: A formal theory and empirical test." *American Journal of Sociology* 107:1143–1178.
- Graham, Lawrence O. 1999. *Our Kind of People: Inside America's Black Upper Class*. New York: Harper-Collins.
- Greenberg, David F. 1983a. "Donald Black's sociology of law: A critique." *Law and Society Review* 17:337–368.

- —. 1983b. "Resistance to conventionality in the sociology of law: A rejoinder to Horwitz." *Law and Society Review* 17:385–390.
- Greenberg, David F. and Valerie West. 2001. "State prison populations and their growth, 1971–1991." *Criminology* 39:615–654.
- Greene, William H. 2000. *Econometric Analysis*, 4th ed. Englewood Cliffs, N.J.: Prentice-Hall.
- Greenhouse, Linda. 2004. "Justices weigh executions of young killers." *New York Times*, October 14, p. A1.
- Griffiths, John. 1984. "The division of labor in social control." Pp. 37–70 in *Toward a General Theory of Social Control: Fundamentals*, vol. 1, edited by D. J. Black. New York: Academic Press.
- Grilli, Leonardo and Carla Rampichini. 2007. "A multilevel multinomial logit model for the analysis of graduates' skills." *Statistical Methods and Applications* 16:381–393.
- Griswold, David B. 1987. "Deviation from sentencing guidelines: The issue of unwarranted disparity." *Journal of Criminal Justice* 15:317–329.
- Grodsky, Eric and Devah Pager. 2001. "The structure of disadvantage: Individual and occupational determinants of the black-white wage gap." *American Sociological Review* 66:542–567.
- Gross, Ariela J. 1998a. "Litigating whiteness: Trials of racial determination in the nineteenth century south." *Yale Law Journal* 108:109–188.
- Gross, Samuel R. 1985. "Race and death: The judicial evaluation of evidence of discrimination in capital sentencing." *University of California at Davis Law Review* 18:1275–1325.
- —. 1993. "The romance of revenge: Capital punishment in America." *Studies in Law, Politics, and Society* 13:71–104.
- —. 1996. "The risks of death: Why erroneous convictions are common in capital cases." Buffalo Law Review 44:469–500.
- -. 1998b. "Lost lives: Miscarriages of justice in capital cases." *Law and Contemporary Problems* 61:125–152.

- -. 2001. "Race, peremptories, and capital jury deliberations." University of Pennsylvania Journal of Constitutional Law 3:283–296.
- Gross, Samuel R., Kristen Jacoby, Daniel J. Matheson, Nicholas Montgomery, and Sujata Patil. 2005. "Exonerations in the United States, 1989 through 2003." *Journal of Criminal Law and Criminology* 95:523–560.
- Gross, Samuel R. and Robert Mauro. 1984. "Patterns of death: An analysis of racial disparities in capital sentencing and homicide victimization." *Stanford Law Review* 37:27–153.
- —. 1989. Death and Discrimination: Racial Disparities in Capital Sentencing. Boston, Mass.: Northeastern University Press.
- Gruhl, John and Susan Welch. 1984. "Women as criminal defendants: A test for paternalism." *Western Political Quarterly* 37:456–467.
- Gryski, Gerard S., Gary Zuk, and Deborah J. Barrow. 1994. "A bench that looks like America? Representation of African Americans and Latinos on the Federal Courts." *The Journal of Politics* 56:1076–1086.
- Guinier, Lani and Gerald Torres. 2002. The Miner's Canary: Enlisting Race, Resisting Power, Transforming Democracy. Cambridge, Mass.: Harvard University Press.
- Gujarati, Damodar N. 1988. Basic Econometrics, 2d ed. New York: McGraw-Hill.
- Gutman, Herbert G. 1976. *The Black Family in Slavery and Freedom, 1970-1925*. New York: Pantheon.
- Hacker, Andrew. 1992. Two Nations: Black and White, Separate, Hostile, Unequal. New York: Scribner.
- Hagan, John. 1974. "Extra-legal attributes and criminal sentencing: An assessment of a sociological viewpoint." *Law and Society Review* 8:357–383.
- —. 1977. "Criminal justice in rural and urban communities: A study of the bureaucratization of justice." *Social Forces* 55:597–612.
- —. 1999. "The corporate advantage: A study of the involvement of corporate and individual victims in a criminal justice system." Pp. 343–370 in *The Social Organization of Law*, 2d ed., edited by M. P. Baumgartner. San Diego, Calif.: Academic Press.
- Hagan, John and Kristin Bumiller. 1983. "Making sense of sentencing: A review and critique of sentencing research." Pp. 1–54 in *Research on Sentencing: The Search*

*for Reform, Volume II*, edited by A. Blumstein, J. Cohen, S. Martin, and M. Tonry. Washington, D.C.: National Academy Press.

- Hagan, John, Ilene H. Nagel, and Celesta A. Albonetti. 1980. "The differential sentencing of white-collar offenders in ten federal district courts." *American Sociological Review* 45:802–820.
- Hagan, John, Carla Shedd, and Monique R. Payne. 2005. "Race, ethnicity, and youth perceptions of criminal injustice." *American Sociological Review* 70:381–407.
- Hagan, John and Marjorie S. Zatz. 1985. "The social organization of criminal justice processing: An event history analysis." *Social Science Research* 14:103–125.
- Haines, Herbert H. 1996. Against Capital Punishment: The Anti-Death Penalty Movement in America, 1972-1994. New York: Oxford University Press.
- Hale, Matthew. [1736] 1800. *Historia Placitorum Coronae: History of the Pleas of the Crown*, vol. 1, edited by G. Wilson and T. Dogherty. London: E. Rider.
- Hancock, Ange-Marie. 2004. *The Politics of Disgust: The Public Identity of the Welfare Queen*. New York: New York University Press.
- Haney, Craig. 2005. *Death by Design: Capital Punishment As a Social Psychological System*. New York: Oxford University.
- Haney, Craig and Deana D. Logan. 1994. "Broken promise: The Supreme Court's response to social science research on capital punishment." *Journal of Social Issues* 50:75–101.
- Harcourt, Bernard E. 2007. *Against Prediction: Profiling, Policing, and Punishing in an Actuarial Age*. Chicago, Ill.: University of Chicago Press.
- Hardin, James and Joseph Hilbe. 2002. *Generalized Estimating Equations*. Bocan Raton, Fla.: Chapman and Hall.
- Harmon, Talia R. 2001a. "Guilty until proven innocent: An analysis of post-*Furman* capital errors." *Criminal Justice Policy Review* 12:113–139.
- 2001b. "Predictors of miscarriages of justice in capital cases." Justice Quarterly 18:946–968.
- Harmon, Talia R. and William S. Lofquist. 2005. "Too late for luck: A comparison of post-*Furman* exonerations and executions of the innocent." *Crime and Delinquency* 51:498–520.

- Harris, Marvin. 1979. *Cultural Materialism: The Struggle for a Science of Culture*. New York: Random House.
- Hart, H. L. A. [1961] 1994. The Concept of Law, 2d ed. Oxford: Oxford University Press.
- Hastie, Trevor J. and Robert J. Tibshirani. 1990. *Generalized Additive Models*. Bocan Raton, Fla.: Chapman and Hall.
- Hausman, Jerry A. 1978. "Specification tests in econometrics." *Econometrica* 46:1251–1271.
- Hawkins, Darnell F. 1987. "Beyond anomalies: Rethinking the conflict perspective on race and criminal punishment." *Social Forces* 65:719–745.
- (Ed.). 1995. *Ethnicity, Race, and Crime: Perspectives Across Time and Place*. Albany, N.Y.: SUNY Press.
- —. 1997. "Which way toward equality? Dilemmas and paradoxes in public policies affecting crime and punishment." Pp. 238–252 in *African Americans and the Public Agenda: The Paradoxes of Public Policy*, edited by C. Herring. Thousand Oaks, Calif.: Sage.
- Hawkins, Darnell F., John W. C. Johnson, and Arthur Michener. 1994. "Race, social class, and newspaper coverage of homicide." *National Journal of Sociology* 9:113–140.
- Heaney, Gerald W. 1991. "The reality of guidelines sentencing: No end to disparity." *American Criminal Law Review* 28:161–228.
- Heckman, James J. 1974. "Shadow prices, market wages, and labor supply." *Econometrica* 42:679–694.
- —. 1976. "The common structure of statistical models of truncation, sample selection and limited dependent variables and a simple estimator for such models." *Annals of Economic and Social Measurement* 5:475–492.
- —. 1978. "Dummy endogenous variables in a simultaneous equation system." *Econometrica* 46:931–959.
- —. 1979. "Sample selection bias as a specification error." *Econometrica* 47:153–162.
- —. 1980. "Addendum to 'Sample selection bias as specification error'." Pp. 69–74 in *Evaluation Studies Review Annual*, vol. 5, edited by E. Stormsdorder and G. Farkas. Beverly Hills, Calif.: Sage.

- Heckman, James J. and Jeffrey A. Smith. 1995. "Assessing the case for social experiments." *Journal of Economic Perspectives* 9:85–110.
- Hedeker, Don. 1998. *MIXPREG: A computer program for mixed-effects Poisson regression*. Chicago, Ill.: School of Public Health, University of Illinois.
- —. 1999. "MIXNO: A computer program for mixed-effects nominal logistic regression." Journal of Statistical Software 4:1–92.
- Hedeker, Don and Robert D. Gibbons. 1994. "A random-effects ordinal regression model for multilevel analysis." *Biometrics* 50:933–944.
- —. 1996. "MIXOR: A computer program for mixed-effects ordinal regression analysis." Computer Methods and Programs in Biomedicine 49:157–176.
- Hëdstrom, Peter and Richard Swedberg. 1998. Social Mechanisms: An Analytical Approach to Social Theory. Cambridge: Cambridge University Press.
- Hegel, Georg Wilhelm Friedrich. [1821] 1952. *Hegel's Philosophy of Right*. New York: Oxford University Press.
- Heise, David R. 1975. Causal Analysis. New York: John Wiley and Sons.
- Helms, Ronald and David Jacobs. 2002. "The political context of sentencing: An analysis of community and individual determinants." *Social Forces* 81:577–604.
- Hembroff, Larry A. 1987. "The seriousness of acts and social contexts: A test of Black's theory of the behavior of law." *American Journal of Sociology* 93:322–347.
- Hempel, Carl G. 1965. "Aspects of scientific explanation." Pp. 331–496 in Aspects of Scientific Explanation and Other Essays in the Philosophy of Science, edited by C. G. Hempel. New York: Free Press.
- Hempel, Carl G. and Paul Oppenheimer. 1948. "Studies in the logic of explanation." *Philosophy of Science* 15:135–175.
- Hertz, Randy and James S. Liebman. 2005. *Federal Habeas Corpus Practice and Procedure*, 5th ed. Newark, N.J.: Matthew Bender and Company.
- Higginbotham, A. Leon, Jr. and Anne F. Jacobs. 1992. "The 'Law Only as an Enemy': The legitimization of racial powerlessness through colonial and antebellum criminal laws of Virginia." *North Carolina Law Review* 70:969–1070.

- Hillier, Amy E. 2003. "Who received loans? Home owners' loan corporation lending and discrimination in Philadelphia in the 1930s." *Journal of Planning History* 2:3–24.
- Hindson, Stephanie, Hillary Potter, and Michael L. Radelet. 2006. "Race, gender, region and death sentencing in Colorado, 1980-1999." *University of Colorado Law Review* 77:549–594.
- Hindus, Michael S. 1980. Prison and Plantation: Crime, Justice, and Authority in Massachusetts and South Carolina, 1767-1878. Chapel Hill, N.C.: University of North Carolina Press.
- Hirschfeld, Magnus. [1938] 1973. *Racism.* Translated by E. Paul and C. Paul. Port Washington, N.Y.: Kennikat Press.
- Hogarth, John. 1971. Sentencing as Human Process. Toronto: University of Toronto Press.
- Holcomb, Jefferson E., Marian R. Williams, and Stephen Demuth. 2004. "White female victims and death penalty disparity research." *Justice Quarterly* 21:877–902.
- Hollingshead, August B. 1941. "The concept of social control." *American Sociological Review* 6:217–224.
- Holmes, Malcolm D., Harmon M. Hosch, Howard C. Daudistel, Delores A. Perez, and Joseph B. Graves. 1996. "Ethnicity, legal resources, and felony dispositions in two southwestern jurisdictions." *Justice Quarterly* 13:11–30.
- Holmes, Oliver Wendell. 1897. "The path of law." Harvard Law Review 10:457-478.
- —. [1881] 1963. The Common Law, edited by M. DeWolfe. Cambridge, Mass.: Harvard University Press.
- Holmes, Stephen and Cass R. Sunstein. 1999. *The Cost of Rights: Why Liberty Depends* on Taxes. Boston, Mass.: Norton.
- Homans, George C. 1964. "Bringing men back in." *American Sociological Review* 29:809–818.
- Hood, Roger. 2003. *The Death Penalty: A Worldwide Perspective*, 3d ed. New York: Oxford University Press.
- Horton, Nicholas J., Stuart R. Lipsitz, and Michael Parzen. 2003. "A potential for bias when rounding in multiple imputation." *American Statistician* 57:229-232.

- Horwitz, Allan V. 1983. "Resistance to Innovation in the sociology of law: A response to Greenberg." *Law and Society Review* 17:369–384.
- —. 1990. The Logic of Social Control. New York: Plenum Press.
- —. 2002. "Toward a new science of social life: A retrospective examination of *The Behavior of Law*." Contemporary Sociology 31:641–644.
- Howe, Scott. 2004. "The futile quest for racial neutrality in capital selection and the Eighth Amendment argument for abolition based on unconscious racial discrimination." *William and Mary Law Review* 45:2083–2166.
- Hsia, Heidi M., George S. Bridges, and Rosalie McHale. 2004. *Disproportionate Minority Confinement: Year 2002 Update*. NCJ 201240. Washington, D.C.: U.S. Department of Justice, Office of Juvenile Justice and Delinquency Prevention.
- Huber, Gregory A. and Sanford C. Gordon. 2004. "Accountability and coercion: Is justice blind when it runs for office?" *American Journal of Political Science* 48:247–263.
- Huber, Gregory A. and John S. Lapinski. 2006. "The 'race card' revisted: Assessing racial priming in policy contests." *American Journal of Political Science* 50:421–440.
- Huff, C. Ronald. 2002. "Wrongful convictions and public policy: The American Society of Criminology presidential address." *Criminology* 40:1–18.
- Hume, David. [1748] 1955. An Inquiry Concerning Human Understanding. New York: Liberal Arts Press.
- Hunt, Alan. 1978. *The Sociological Movement in Law*. Philadelphia, Pa.: Temple University Press.
- —. 1983. "Behavioral sociology of law: A critique of Donald Black." *Journal of Law and* Society 10:19–46.
- —. 1993. *Explorations in Law and Society: Toward a Constitutive Theory of Law*. New York: Routledge.
- Ibrahim, Paul. 2004. "Capital Punishment: Good for America?" *Cornell Review*, March 14, p. 1. Retrieved September 10, 2004 (http://www.cornellreview.org/viewart.cgi?num=350).

- Iceland, John, Daniel H. Weinberg, and Erika Steinmetz. 2002. *Racial and Ethnic Residential Segregation in the United States, 1980–2000.* Washington, D.C.: U.S. Bureau of the Census.
- Imbens, Guido W. 2004. "Nonparametric estimation of average treatment effects under exogeneity: A review." *The Review of Economics and Statistics* 86:4–29.
- Iyengar, Radha. 2007. "An Analysis of the Performance of Federal Indigent Defense Counsel." Working Paper No. 13187, National Bureau of Economic Research, Cambridge, Mass.
- Jackson, Pamela I. 1986. "Black visibility, city size, and social control." *Sociological Quarterly* 27:185–203.
- Jacobs, David and Jason T. Carmichael. 2001. "The politics of punishment across time and space: A pooled time-series analysis of imprisonment rates." *Social Forces* 80:61–89.
- —. 2002. "The political sociology of the death penalty: A pooled time-series analysis." American Sociological Review 67:109–131.
- Jacobs, David, Jason T. Carmichael, and Stephanie L. Kent. 2005. "Vigilantism, current racial threat, and death sentences." *American Sociological Review* 70:656–677.
- Jacobs, David, Zhenchao Qian, Jason T. Carmichael, and Stephanie L. Kent. 2007. "Who survives on death row? An individual and contextual analysis." *American Sociological Review* 72:610–632.
- Jacobs, David and Daniel Tope. 2007. "The politics of resentment in the post-Civil Rights Era: Minority threat, homicide, and ideological voting in Congress." *American Journal of Sociology* 112:1458–1494.
- James, W. and C. Stein. 1961. "Estimation with quadratic loss." Pp. 361–379 in Proceedings of the Fourth Berkeley Symposium on Mathematical Statistics and Probability, edited by J. Neyman. Berkeley, Calif.
- Jasso, Guillermina. 1988. "Principles of theoretical analysis." *Sociological Theory* 6:1–20.
- —. 2006. "Formal theory." Pp. 37-68 in *Handbook of Sociology Theory*, edited by J. H. Turner. New York: Springer.
- Jeffries, John C., Jr. 1994. Justice Lewis F. Powell, Jr.: A Biography. New York: Charles Scribner's Sons.

- Jenkins, J. Craig, David Jacobs, and Jon Agnone. 2003. "Political opportunities and African-American protest, 1948–1997." *American Journal of Sociology* 109:277– 303.
- Johnson, Brian D. 2003. "Racial and ethnic disparities in sentencing departures across modes of conviction." *Criminology* 41:449–490.
- —. 2005. "Contextual disparities in guideline departures: Courtroom social contexts, guidelines compliance, and extralegal disparities in criminal sentencing." *Criminology* 43:761–796.
- 2006a. "The multilevel context of criminal sentencing: Integrating judge and county level influences." *Criminology* 44:259–298.
- Johnson, David T. 2006b. "Where the state kills in secret: Capital punishment in Japan." *Punishment and Society* 8:251–285.
- Johnson, Dirk. 2000a. "Illinois, citing verdict errors, bars executions." *New York Times*, February 1, pp. A1, A16.
- —. 2000b. "No executions in Illinois until system is repaired." New York Times, May 21, p. A20.
- Johnson, Guy B. 1941. "The Negro and crime." Annals of the American Academy of Political and Social Science 217:93–104.
- Johnson, Sheri L. 1988. "Unconscious racism and the criminal law." *Cornell Law Review* 73:1016–1037.
- Johnston, J. B., T. D. Kennedy, and I. G. Shuman. 1987. "Gender differences in the sentencing of felony offenders." *Federal Probation* 51:49–55.
- Jones, Jeffrey M. 2005. "Americans' view on the death penalty more positive this year." *Gallup Poll News Service*, May 19, p. 1.
- Joreskog, Karl G. and Dag Sorbom. 1996. *LISREL 8 User's Reference Guide*. 2d ed. Chicago, Ill.: Scientific Software International.
- Judicial Council of Georgia. 2003. *Your Guide to the Georgia Courts*. Atlanta, Ga.: Administrative Office of the Courts.
- Kan, Yee W. and Scott Phillips. 2003. "Race and the death penalty: Including Asian Americans and exploring the desocialization of law." *Journal of Ethnicity in Criminal Justice* 1:63–92.

Kang, Jerry. 2005. "Trojan horses of race." Harvard Law Review 118:1491–1593.

- Katz, Jack. 1999. "Hunting for bias: Notes on the evolution of strategies for documenting invidious discrimination." Pp. 210–257 in Social Science, Social Policy, and the Law, edited by P. Ewick, R. A. Kagan, and A. Sarat. New York: Russell Sage Foundation.
- Katz, Lawrence, Steven D. Levitt, and Ellen Shustorovich. 2003. "Prison conditions, capital punishment, and deterrence." *American Law and Economics Review* 5:318–343.
- Kaufman-Osborn, Timothy V. 2004. "Capital punishment, proportionality review, and claims of fairness (with lessons from Washington State)." *Washington Law Review* 79:775–880.
- —. 2006a. "Capital punishment as legal lynching?" Pp. 21–54 in *From Lynch Mobs to the Killing State: Race and the Death Penalty in America*, edited by C. J. Ogletree, Jr. and A. Sarat. New York: New York University Press.
- —. 2006b. "A critique of contemporary death penalty abolitionism." *Punishment and* Society 8:365-383.
- Kautt, Paula M. 2002. "Location, location: Inter-district and inter-circuit variation in sentencing outcomes for federal drug-trafficking offenses." *Justice Quarterly* 19:633–671.
- Keedy, E.R. 1949. "History of the Pennsylvania statute creating degrees of murder." University of Pennsylvania Law Review 97:749–777.
- Keil, Thomas J. and Gennaro F. Vito. 1989. "Race, homicide severity, and application of the death penalty: A consideration of the *Barnett* Scale." Criminology 27:511– 535.
- —. 1992. "The Effects of the *Furman* and *Gregg* Decisions on Black-White Execution Ratios in the South." *Journal of Criminal Justice* 20:217–226.
- —. 1996. "Race and the death penalty in Kentucky murder trials: 1976-1991." *American Journal of Criminal Justice* 20:17–36.
- Kelling, George L. and Catherine M. Coles. 1996. *Fixing Broken Windows: Restoring* Order and Reducing Crime in Our Communities. New York: Simon and Schuster.
- Kempf-Leonard, Kimberly and Lisa L. Sample. 2001. "Have federal sentencing guidelines reduced severity? An examination of one circuit." *Journal of Quantitative Criminology* 17:111–144.
- Kempf, Kimberly and Roy L. Austin. 1986. "Older and more recent evidence on racial discrimination in sentencing." *Journal of Quantitative Criminology* 2:29–47.
- Kennedy, Randall. 1988. "*McCleskey v. Kemp*: Race, capital punishment, and the Supreme Court." *Harvard Law Review* 101:1388–1411.
- —. 1998. Race, Crime, and the Law. New York: Vintage.
- Kent, Stephanie L. and David Jacobs. 2004. "Social divisions and coercive control in advanced societies: Law enforcement strength in eleven nations from 1975 to 1994." *Social Problems* 51:343–361.
- —. 2005. "Minority threat and police strength from 1980 to 2000: A fixed-effects analysis of nonlinear and interactive effects in large U.S. cities." *Criminology* 43:731–760.
- Kim, Jae-On and G. Donald Ferree. 1981. "Standardization in causal analysis." Sociological Methods and Research 10:187–210.
- King, Gary. 1997. A Solution to the Ecological Inference Problem: Reconstructing Individual Behavior from Aggregate Data. Princeton, N.J.: Princeton University Press.
- King, Gary, James Honaker, Anne Joseph, and Kenneth Scheve. 2001. "Analyzing incomplete political science data: An alternative algorithm for multiple imputation." *American Political Science Review* 95:49–69.
- King, Gary, Robert O. Keohane, and Sidney Verba. 1994. *Designing Social Inquiry: Scientific Inference in Qualitative Research*. Princeton, N.J.: Princeton University Press.
- King, Gary, Michael Tomz, and Jason Wittenberg. 2000. "Making the most of statistical analyses: Improving interpretation and presentation." *American Journal of Political Science* 44:341–355.
- King, Gary and Langche Zeng. 2006. "The dangers of extreme counterfactuals." *Political Analysis* 14:131–159.
- King, Ryan D. 2007. "The context of minority group threat: Race, institutions, and complying with hate crime law." *Law and Society Review* 41:189–224.
- King, Ryan D. and Darren Wheelock. 2007. "Group threat and social control: Race, perceptions of minorities and the desire to punish." *Social Forces* 85:1255–1280.

- King, Ryan S. and Marc Mauer. 2004. "The vanishing black electorate: Felony disenfranchisement in Atlanta, Georgia." The Sentencing Project, Washington, D.C.
- Kirchmeier, Jeffrey L. 2002. "'Another Place Beyond Here': The death penalty moratorium movement in the United States." *University of Colorado Law Review* 73:1–116.
- —. 2006. "Dead innocent: The death penalty abolitionist search for a wrongful execution." *Tulsa Law Review* 42:403–435.
- Kiser, Edgar and Michael Hechter. 1991. "The role of general theory in comparativehistorical sociology." *American Journal of Sociology* 97:1–30.
- Klag, M. J., P. K. Whelton, J. Coresh, C. E. Grim, and L. H. Kuller. 1991. "The association of skin color with blood pressure in US blacks with low socioeconomic status." *Journal of the American Medical Association* 265:599– 602.
- Kleck, Gary. 1969. "A study of the California penalty jury in first-degree murder cases." *Stanford Law Review* 21:1297–1366.
- —. 1981. "Racial discrimination in criminal sentencing: A critical evaluation of the evidence with additional evidence on the death penalty." *American Sociological Review* 46:783–805.
- —. 1985. "Life support for ailing hypotheses: Modes of summarizing the evidence for racial discrimination in sentencing." *Law and Human Behavior* 9:271–285.
- —. 1991. "Equal Justice and the Death Penalty: A Legal and Empirical Analysis (Book Review)." Contemporary Sociology 20:598–599.
- Klein, Lawrence, Brian E. Forst, and Victor Filatov. 1978. "The deterrence effect of capital punishment: An assessment of the estimates." Pp. 336–360 in *Deterrence* and Incapacitation: Estimating the Effects of Criminal Sanctions on Crime Rates, edited by A. Blumstein, J. Cohen, and D. Nagin. Washington, D.C.: National Academy Press.
- Klein, Stephen P., Allen Abrahamse, and John E. Rolph. 1987. "Racial equity in prosecutor requests for the death penalty." RAND Corporation, Santa Monica, Calif.
- Klein, Stephen P., Richard A. Berk, and Laura Hickman. 2006. "Race and the decision to seek the death penalty in federal cases." RAND Corporation, Santa Monica, Calif.

- Klein, Stephen P. and John E. Rolph. 1989. "Relationship of offender and victim race to death penalty sentences in California." RAND Corporation, Santa Monica, Calif.
- Klemm, Margaret Fae. 1986. "The Determinants of Capital Sentencing in Louisiana, 1975-1984." Ph.D. dissertation, Department of Political Science, University of New Orleans, New Orleans, La.
- Klüver, Jürgen and Jörn Schmidt. 1999. "Social differentiation as the unfolding dimensions of social systems." *Journal of Mathematical Sociology* 23:309–325.
- Koeninger, Rupert C. 1969. "Capital Punishment in Texas, 1924-1968." Crime and Delinquency 15:132–141.
- Kozol, Jonathan. 1991. Savage Inequalities: Children in America's Schools. New York: Crown.
- Kramer, John H. and Darrell Steffensmeier. 1993. "Race and imprisonment decisions." Sociological Quarterly 34:357–376.
- Kramer, John H. and Jeffrey T. Ulmer. 1996. "Sentencing disparity and departure from guidelines." *Justice Quarterly* 13:81–105.
- —. 2002. "Downward departures for serious violent offenders: Local court "corrections" to Pennsylvania's sentencing guidelines." *Criminology* 40:897–932.
- Kroll, Michael. 1991. "Chattahoochee Judicial District: The Buckle on the Death Belt-The Death Penalty in Microcosm." Death Penalty Information Center, Washington, D.C.
- Kruttschnitt, Candace. 1981. "Social status and sentences of female offenders." *Law and Society Review* 15:247–265.
- —. 1982. "Women, crime, and dependency: An application of the Theory of Law." Criminology 19:495–513.
- Kruttschnitt, Candace and Daniel McCarthy. 1985. "Familial social control and pretrial sanctions: Does sex really matter?" *Journal of Criminal Law and Criminology* 76:151–175.
- Kuhn, Thomas S. 1970. *The Structure of Scientific Revolutions*, 2d ed. Chicago, Ill.: University of Chicago Press.
- —. 1977. "Second thoughts on paradigms." Pp. 459–481 in *The Structure of Scientific Theories*, 2d ed., edited by F. Suppe. Urbana, Ill.: University of Illinois Press.

- Kurlycheck, Megan C. and Brian D. Johnson. 2004. "The juvenile penalty: A comparison of juvenile and young adult sentencing outcomes in criminal court." *Criminology* 42:485–515.
- Kuziemko, Ilyana. 2006. "Does the threat of the death penalty affect plea bargaining in murder cases? Evidence from New York's 1995 reinstatement of capital punishment." *American Law and Economics Review* 8:116–142.
- LaFree, Gary and Richard Arum. 2006. "The impact of racially inclusive schooling on adult incarceration rates among U.S. cohorts of African Americans and Whites since 1930." *Criminology* 44:73–104.
- LaFree, Gary D. 1989. *Rape and Criminal Justice: The Social Construction of Sexual Assault*. Belmont, Calif.: Wadsworth.
- Lakatos, Imre. 1970. "Falsification and methodology of scientific research programmes." Pp. 91–195 in *Criticism and the Growth of Knowledge*, edited by I. Lakatos and A. Musgrave. Cambridge: Cambridge University Press.
- Lakoff, George. 2002. *Moral Politics: How Liberals and Conservatives Think*, 2d ed. Chicago, Ill.: University of Chicago Press.
- Land, Greg. 2007. "Public defense system eyed for overhaul." *Daily Report*, February 16, p. 1.
- Land, Kenneth C., Patricia L. McCall, and Lawrence E. Cohen. 1990. "Structural covariates of homicide rates: Are there any invariances across time and social space?" *American Journal of Sociology* 95:922–963.
- Landerman, Lawrence R., Kenneth C. Land, and Carl F. Pieper. 1997. "An empirical evaluation of the predictive mean matching method for imputing missing values." *Sociological Methods and Research* 26:3–33.
- Langbein, Laura I. 1999. "Politics, rules, and death row: Why states eschew or execute executions." *Social Science Quarterly* 80:629–647.
- Lanier, Charles S. and James R. Acker. 2004. "Capital punishment, the moratorium movement, and empirical questions: Looking beyond innocence, race, and bad lawyering in death penalty cases." *Psychology, Public Policy, and Law* 10:577– 616.
- Laudan, Larry. 1977. Progress and Its Problems: Towards a Theory of Scientific Growth. Berkeley and Los Angeles, Calif.: University of California Press.
- Laufer, William S. and Nien-he Hsieh. 2003. "Choosing equal injustice." *American Journal of Criminal Law* 30:343–361.

- Lazarsfeld, Paul F. and Herbert Menzel. 1969. "On the relation between individual and collective properties." Pp. 499–516 in *A Sociological Reader on Complex Organizations*, edited by A. Etzioni. New York: Holt, Rinehart and Winston.
- Leamer, Edward E. 1978. Specification Searches: Ad Hoc Inference with Nonexperimental Data. New York: John Wiley and Sons.
- —. 1983. "Let's take the con out of econometrics." *American Economic Review* 73:31–43.
- Lee, Catherine Y. 2007. "Hispanics and the death penalty: Discriminatory charging practices in San Joaquin County, California." *Journal of Criminal Justice* 35:17–27.
- Lee, Jayne Chong-Soon. 1995. "Navigating the topology of race." Pp. 441–448 in *Critical Race Theory: The Key Writings that Formed the Movement*, edited by K. Crenshaw, N. Gotanda, G. Peller, and K. Thomas. New York: New Press.
- Lenski, Gerhard. 1988. "Rethinking macrosociological theory." *American Sociological Review* 53:163–171.
- Lessan, Gloria T. and Joseph F. Sheley. 1992. "Does law behave? A macrolevel test of Black's propositions on change in law." *Social Forces* 70:655–678.
- Levi, Edward H. 1949. *An Introduction to Legal Reasoning*. Chicago, Ill.: University of Chicago Press.
- Levi, Ken. 1980. "Homicide as conflict resolution." Deviant Behavior 1:281–307.
- Lewis, Peter, Henry Mannle, and Harold Vetter. 1979. "A Post-*Furman* profile of Florida's condemned: A question of discrimination in terms of race of the victim and a comment on *Spenkelink v. Wainwright*." *Stetson Law Review* 9:1–45.
- Lewontin, Richard C. 1972. "The apportionment of human diversity." *Evolutionary Biology* 6:381–398.
- Liao, Tim Futing. 1994. Interpreting Probability Models: Logit, Probit, and other Generalized Linear Models. Thousand Oaks, Calif.: Sage.
- Lieberman, Robert C. 1998. *Shifting the Color Line: Race and the American Welfare State*. Cambridge, Mass.: Harvard University Press.
- Lieberson, Stanley. 1980. A Piece of the Pie: Black and White Immigrants Since 1880. Berkeley and Los Angeles, Calif.: University of California Press.

- Lieberson, Stanley and Freda B. Lynn. 2002. "Barking up the wrong branch: Scientific alternatives to the current model of sociological science." *Annual Review of Sociology* 28:1–19.
- Liebman, James S. 2000. "The overproduction of death." *Columbia Law Review* 100:2030–2156.
- —. 2007. "Slow dancing with death: The Supreme Court and capital punishment, 1963-2006." Columbia Law Review 107:1–130.
- Liebman, James S., Jeffrey Fagan, Andrew Gelman, Valerie West, Garth Davies, and Alexander Kiss. 2002. "A Broken System, Part II: Why There Is So Much Error in Capital Cases, and What Can Be Done About It." The Justice Project, New York.
- Liebman, James S., Jeffrey Fagan, and Valerie West. 2000a. "A Broken System: Error Rates in Capital Cases, 1973-1995." The Justice Project, New York.
- Liebman, James S., Jeffrey Fagan, Valerie West, and Jonathan Floyd. 2000c. "Capital attrition: Error rates in capital cases, 1973-1995." *Texas Law Review* 78:1839–1865.
- Lillard, Lee A. and Constantijn W. A. Panis. 2003. *aML: Multilevel Multiprocess Statistical Software*. Los Angeles, Calif.: EconWare.
- Lilly, J. Robert and J. Michael Thomson. 1997. "Executing U.S. soldiers in England, World War II: Command influence and sexual racism." *British Journal of Criminology* 37:262–288.
- Lincoln, Abraham. [1862] 1953. "Letter to Horace Greeley." Pp. 388–389 in *The Collected Works of Abraham Lincoln*, vol. 5, edited by R. P. Basler. New Brunswick, N.J.: Rutgers University Press.
- Lindblom, Charles E. 1977. *Politics and Markets: The World's Political Economic Systems*. New York: Basic Books.
- Lindley, Dennis V. and Adrian F. M. Smith. 1972. "Bayes estimates for the linear model." *Journal of the Royal Statistical Society* (Series B) 34:1–41.
- Liptak, Adam. 2004. "Kansas' death penalty law is declared unconstitutional." *New York Times*, December 18, p. A16.
- Liska, Allen E. 1993. "Social structure and social control: Building theory." *Law and Society Review* 27:345–354.

- —. 1997. "Modeling the relationships between macro forms of social control." Annual Review of Sociology 23:39–61.
- Liska, Allen E., Mitchell B. Chamblin, and Mark D. Reed. 1985. "Testing the economic production and conflict models of crime control." *Social Forces* 64:119–138.
- Little, Ramon, George Milliken, Walter Stroup, and Russel Wolfinger. 1996. SAS System for Mixed Models. Cary, N.C.: SAS Institute, Inc.
- Little, Roderick J. A. 1992. "Regression with Missing X's: A Review." Journal of the American Statistical Association 87:1227–1237.
- Little, Rory. 1999. "The Federal Death Penalty: History and some thoughts about the Department of Justice's Role." *Fordham Urban Law Journal* 26:347–508.
- Litwin, Kenneth J. 2004. "A multilevel multivariate analysis of factors affecting homicide clearances." *Journal of Research in Crime and Delinquency* 41:327– 351.
- Livingston, Debra. 1982. "Round and 'round the bramble bush: From legal realism to critical legal scholarship." *Harvard Law Review* 95:1669–1690.
- Llewellyn, Karl N. [1930] 1978. *The Bramble Bush: Our Law and Its Study*. New York: Oceana Publications.
- Logan, John R. 2004. "How race counts for Hispanic Americans." Sage Race Relations Abstracts 29:7–19.
- Long, J. Scott. 1997. *Regression Models for Categorical and Limited Dependent Variables*. Thousand Oaks, Calif.: Sage.
- Longford, Nicholas T. 1990. VARCL: Software for Variance Component Analysis of Data with Hierarchically Nested Random Effects (Maximum Likelihood). Princeton, N.J.: Educational Testing Service.
- —. 1993. Random Coefficient Models. New York: Oxford University Press.
- Loury, Glen C. 1977. "A dynamic theory of racial income differences." Pp. 153–186 in Women, Minorities, and Employment Discrimination, edited by P. A. Wallace and A. LeMund. Lexington, Mass.: Lexington Books.
- Loveman, Mara. 1999. "Is 'race' essential?: Comment on Bonilla-Silva." *American Sociological Review* 64:891–898.

- Lu, Lynn D. 2007. "Prosecutorial discretion and racial disparities in federal sentencing: some views of former U.S. Attorneys." *Federal Sentencing Reporter* 19:192–201.
- Luckenbill, David. 1977. "Criminal homicide as a situated transaction." *Social Problems* 25:176–186.
- Luginbuhl, James and Julie Howe. 1995. "Discretion in capital sentencing instructions: Guided or misguided." *Indiana Law Journal* 70:1161–1181.
- Luhmann, Niklas. 1989. "Law as a social system." *Northwestern University Law Review* 83:136–150.
- Luke, Douglas A. 2004. Multilevel Modeling. Thousand Oaks, Calif.: Sage.
- Lynch, Mona. 2006. "Stereotypes, prejudice, and life-and-death decision-making: Lessons from laypersons in an experimental setting." Pp. 182–210 in From Lynch Mobs to the Killing State: Race and the Death Penalty in America, edited by C. J. Ogletree, Jr. and A. Sarat. New York: New York University Press.
- Lynch, Mona and Craig Haney. 2000. "Discrimination and instructional comprehension: Guided discretion, racial bias, and the death penalty." *Law and Human Behavior* 24:337–358.
- Machamer, Peter, Lindley Darden, and Carl Craver. 2000. "Thinking about mechanisms." *Philosophy of Science* 67:1–25.
- Machery, Edouard and Luc Faucher. 2005. "Why do we think racially? A critical journey in culture and evolution." Pp. 1009–1033 in *Handbook of Categorization in Cognitive Science*, edited by H. Cohen and C. Lefebvre. Amsterdam: Elsevier.
- Machlup, Fritz. 1955. "The problem of verification in economics." *Southern Economic Journal* 21:1–21.
- Males, Mike and Dan Macallair. 2000. *The Color of Justice: An Analysis of Juvenile Adult Court Transfers in California*. Washington, D.C.: Justice Policy Institute.
- Mander, Adrian and David G. Clayton. 2005. "Weighted hotdeck imputation." MRC Biostatistics Unit, Cambridge, Mass. Unpublished manuscript.
- Mangum, Charles S., Jr. 1940. *The Legal Status of the Negro*. Chapel Hill, N.C.: University of North Carolina Press.
- Mannheimer, Michael. 2006. "When the federal death penalty is 'cruel and unusual'." University of Cincinnati Law Review 74:819–885.

- Manning, Willard G., Naihua Duan, and William H. Rogers. 1987. "Monte Carlo evidence on the choice between sample selection bias and two-part models." *Journal of Econometrics* 35:59–82.
- Manza, Jeff. 2000. "Race and the underdevelopment of the American Welfare State." *Theory and Society* 29:819–832.
- Manza, Jeff and Christopher Uggen. 2005. Locked Out: Felon Disenfranchisement and American Democracy. New York: Oxford University Press.
- Mark, Noah. 1998. "Beyond individual differences: Social differentiation from first principles." *American Sociological Review* 63:309–330.
- Marketti, James. 1990. "Estimated present value of income diverted during slavery." Pp. 107–124 in *The Wealth of Races: The Present Value of Benefits from Past Injustices*, edited by R. F. America. Westport, Conn.: Greenwood Press.
- Marquart, James W., Sheldon Ekland-Olson, and Jon R. Sorensen. 1994. *The Rope, the Chair, and the Needle: Capital Punishment in Texas, 1923-1990.* Austin, Tex.: University of Texas Press.
- Marquart, James W. and Jon R. Sorensen. 1989. "A national study of the *Furman*commuted inmates: Assessing the threat to society from capital offenders." *Loyola of Los Angeles Law Review* 23:5–28.
- Marquis, Joshua K. 2003. "Truth and consequences: The penalty of death." Pp. 117–151 in *Debating the Death Penalty: Should America Have Capital Punishment?*, edited by H. A. Bedau and P. G. Cassell. New York: Oxford University Press.
- Marshall, Carolyn. 2006. "California Assembly sidelines a moratorium on executions." *New York Times*, January 20, p. A12.
- Martin, Joyce A., Brady E. Hamilton, Paul D. Sutton, Stephanie J. Ventura, Fay Menacker, and Martha L. Munson. 2003. *Births: Final data for 2002. National Vital Statistics Reports*, vol. 52. Hyattsville, Md.: National Center for Health Statistics.
- Marx, Karl. [1859] 1970. *A Contribution to the Critique of Political Economy*. New York: International Publishers.
- —. [1857] 1973. *Grundrisse: Foundations of the Critique of Political Economy*, vol. 1. Translated by M. Nicolaus. New York: Vintage.
- —. [1867] 1975. Capital: A Critique of Political Economy. New York: International Publishers.

- —. [1841] 1975. "Debates on the law of thefts of wood." Pp. 224–263 in Karl Marx-Friedrich Engels: Collected Works, vol. 1. New York: International Publishers.
- Marx, Karl and Friedrich Engels. [1846] 1947. *The German Ideology*. New York: International Publishers.
- —. [1848] 1959. "Manifesto of the Communist Party." Pp. 1–41 in Marx and Engels: Basic Writings on Politics and Philosophy, edited by L. S. Feuer. Garden City, N.Y.: Anchor Books.
- —. 1964. Marx and Engels on Religion, edited by R. Niebuhr. New York: Schocken Books.
- Massey, Douglas S. 1995. "Getting Away with Murder': Segregation and violent crime in urban America." *University of Pennsylvania Law Review* 143:1203–1232.
- —. 2004. "Segregation and stratification: A biosocial perspective." *Du Bois Review* 1:7–25.
- Massey, Douglas S. and Nancy A. Denton. 1989. "Hypersegregation in U.S. metropolitan areas: Black and Hispanic segregation along five dimensions." *Demography* 26:373–392.
- —. 1993. American Apartheid: Segregation and the Making of the Underclass. Cambridge, Mass.: Harvard University Press.
- Massey, Douglas S. and Zoltan L. Hajnal. 1995. "The changing geographic structure of black-white segregation in the United States." *Social Science Quarterly* 76:527–542.
- Massey, Douglas S. and Garvey Lundy. 2001. "Use of Black English and racial discrimination in urban housing markets: New methods and findings." *Urban Affairs Review* 36:452–469.
- Massey, James L. and Martha A. Myers. 1989. "Patterns of repressive social control in Post-Reconstruction Georgia, 1882-1935." *Social Forces* 68:458–488.
- Mastrofski, Stephen D., Michael D. Reisig, and John D. McCluskey. 2002. "Police disrespect toward the public: An encounter-based analysis." *Criminology* 40:519–553.
- Mauer, Marc. 1999. Race to Incarcerate. New York: New Press.
- Maxwell, Nicholas. 1974a. "The rationality of scientific discovery, Part I: The traditional rationality problem." *Philosophy of Science* 41:123–153.

- —. 1974b. "The rationality of scientific discovery, Part II: An aim-oriented theory of scientific discovery." *Philosophy of Science* 41:247–295.
- —. 2005. "Popper, Kuhn, Lakatos and aim-oriented empiricism." *Philosophia* 32:181-239.
- Mayhew, Bruce H. 1980. "Structuralism versus individualism, Part I: Shadowboxing in the dark." *Social Forces* 59:335–375.
- —. 1981. "Structuralism versus individualism, Part II: Ideological and other obfuscations." *Social Forces* 59:627–648.
- McAdams, John C. 1998. "Racial disparity and the death penalty." *Law and Contemporary Problems* 61:153–170.
- McCarthy, Bill. 2002. "New economics of sociological criminology." *Annual Review of* Sociology 28:417–442.
- McCullagh, Peter and John A. Nelder. 1989. *Generalized Linear Models*, 2d ed. Bocan Raton, Fla.: Chapman and Hall.
- McGarrell, Edmund F. and Marla Sandys. 1996. "The Misperception of public opinion toward capital punishment." *American Behavioral Scientist* 39:500–513.
- McKinney, John C. 1966. *Constructive Typology and Social Theory*. New York: Appleton-Century-Crofts.
- McManus, Walter S. 1985. "Estimates of the deterrent effect of capital punishment: The importance of the researcher's prior beliefs." *Journal of Political Economy* 98:417–425.
- McNally, Kevin. 2004. "Race and the federal death penalty: A nonexistent problem gets worse." *Depaul Law Review* 53:1615–1645.
- McWhorter, John H. 1998. Word on the Street: Debunking the Myth of "Pure" Standard English. New York: Perseus Publishing.
- Mead, George Herbert. 1934. *Mind, Self, and Society: From the Standpoint of a Social Behaviorist*. Chicago, Ill.: University of Chicago Press.
- Mears, Daniel P. 1998a. "The sociology of sentencing: Reconceptualizing decisionmaking processes and outcomes." *Law and Society Review* 32:667–724.
- Mears, Michael. 1998b. "Georgia needs a Racial Justice Act." *Daily Report*, September 25, p. 1.

- —. 1999. *The Death Penalty in Georgia: A Modern History, 1970-2000.* Atlanta, Ga.: Georgia Indigent Defense Division of Professional Education.
- —. 2004. "Lethal injection and the Georgia Supreme Court's new millennium." The Champion, January/February, p. 33.
- Mendelberg, Tali. 2001. *The Race Card: Campaign Strategy, Implicit Messages, and the Norm of Equality.* Princeton, N.J.: Princeton University Press.
- Meng, Xiao-Li and Donald B. Rubin. 1993. "Maximum likelihood estimation via the ECM algorithm: A general framework." *Biometrika* 80:267-278.
- Merton, Robert K. 1938. "Social structure and anomie." *American Sociological Review* 3:672–682.
- -. 1945. "Sociological theory." American Journal of Sociology 50:462-473.
- —. 1948. "The bearing of empirical research upon the development of social theory." *American Sociological Review* 13:505–515.
- -. 1967. On Theoretical Sociology: Five Essays, Old and New. New York: Free Press.
- Messner, Steven F., Robert D. Baller, and Matthew P. Zevenbergen. 2005. "The legacy of lynching and southern homicide." *American Sociological Review* 70:633–655.
- Messner, Steven F., Glenn Deane, and Mark Beaulieu. 2002. "A log-multiplicative association model for allocating homicides with unknown victim-offender relationships." *Criminology* 40:457–480.
- Messner, Steven F. and Richard Rosenfeld. 1994. *Crime and the American Dream*. Belmont, Calif.: Wadsworth.
- Michalowski, Raymond J. and Michael A. Pearson. 1990. "Punishment and social structure at the state level: A cross-sectional comparison of 1970 and 1980." *Journal of Research in Crime and Delinquency* 27:52–78.
- Michalski, Joseph H. 2004. "Making sociological sense out of trends in intimate partner violence: The social structure of violence against women." *Violence Against Women* 10:652–675.
- Michels, Robert. [1911] 1962. Political Parties: A Sociological Study of the Oligarchial Tendencies of Modern Democracy. New York: Collier Books.

- Miethe, Terance D. and Charles A. Moore. 1985. "Socioeconomic disparities under determinate sentencing systems: A comparison of preguideline and postguideline practices in Minnesota." *Criminology* 23:337–363.
- —. 1986. "Racial differences in criminal processing: The consequences of model selection on conclusions about differential treatment." *Sociological Quarterly* 27:217–237.
- Miethe, Terrence D. and Kriss A. Drass. 1999. "Exploring the social context of instrumental and expressive homicides: An application of Qualitative Comparative Analysis." *Journal of Quantitative Criminology* 15:1–21.
- Miles, Thomas J. 2004. "Felon disenfranchisement and voter turnout." *Journal of Legal Studies* 33:85–129.
- Miller, Jerome G. 1996. Search and Destroy: African American Males in the Criminal Justice System. Cambridge: Cambridge University Press.
- Mills, C. Wright. 1959. *The Sociological Imagination*. New York: Oxford University Press.
- Mincy, Ronald B. (Ed.). 2006. *Black Males Left Behind*. Washington, D.C.: Urban Institute Press.
- Miranda, Alfonso and Sophia Rabe-Hesketh. 2006. "Maximum likelihood estimation of endogenous switching and sample selection models for binary, ordinal, and count variables." *Stata Journal* 6:285–308.
- Mitchell, Michael and Jim Sidanius. 1995. "Social hierarchy and the death penalty: A social-dominance perspective." *Political Psychology* 16:591–619.
- Mitchell, Ojmarrh. 2005. "A meta-analysis of race and sentencing research: Explaining the inconsistencies." *Journal of Quantitative Criminology* 21:439–466.
- Mocan, H. Naci and R. Kaj Gittings. 2003. "Getting off death row: Commuted sentences and the deterrent effect of capital punishment." *Journal of Law and Economics* 46:453–478.
- Moon, Melissa, John P. Wright, Francis T. Cullen, and Jennifer Pealer. 2000. "Putting kids to death: Specifying public support for juvenile capital punishment." *Justice Quarterly* 17:663–684.
- Mooney, Linda A. 1986. "The Behavior of Law in a private legal system." *Social Forces* 64:733–750.

- Moore, Charles A. and Terance D. Miethe. 1986. "Regulated and unregulated sentencing decisions: An analysis of first-year practices under Minnesota's felony sentencing guidelines." *Law and Society Review* 20:253–277.
- Moore, David W. 2004. "Public divided between death penalty and life imprisonment without parole." *Gallup News Service*, June 2.
- Moran, Richard. 2002. Executioner's Current: Thomas Edison, George Westinghouse and the Invention of the Electric Chair. New York: Knopf.
- Morrill, Calvin. 1991. "Conflict management, honor, and organizational change." *American Journal of Sociology* 97:585–621.
- Morrill, Calvin, Mayer N. Zald, and Hayagreeva Rao. 2003. "Covert political conflict in organizations: Challenges from below." *Annual Review of Sociology* 29:391–415.
- Morton, Sally C. and John E. Rolph. 2000. "Racial bias in death sentencing: Assessing the statistical evidence." Pp. 94–116 in *Public Policy and Statistics: Case Studies From Rand*, edited by S. C. Morton and J. E. Rolph. New York: Springer-Verlag.
- Mullin, Gerald W. 1972. *Flight and Rebellion: Slave Resistance in Eighteenth-Century Virginia*. New York: Oxford University Press.
- Mullis, Jeffery. 1995. "Medical malpractice, social structure, and social control." *Sociological Forum* 10:135–163.
- Murphy, Elizabeth. 1984. "The application of the death penalty in Cook County." *Illinois Bar Journal* 93:90–95.
- Mustard, David B. 2001. "Racial, ethnic, and gender disparities in sentencing: Evidence from the U.S. Federal Courts." *Journal of Law and Economics* 44:285–314.
- Muthén, Bengt O. and Linda K. Muthén. 2004. *Mplus Version 3 User's Guide*. Los Angeles, Calif.: Muthén and Muthén.
- Myers, Martha A. 1980. "Predicting the behavior of law: A test of two models." *Law and Social Inquiry* 14:835–857.
- Myers, Martha A. and Susette M. Talarico. 1986. "Urban justice, rural injustice: Urbanization and its effect on sentencing." *Criminology* 24:367–390.
- -. 1987. The Social Contexts of Criminal Sentencing. New York: Springer-Verlag.
- Myrdal, Gunnar. 1944. An American Dilemma: The Negro Problem and Modern Democracy. New York: Harper and Brothers.

- Nagel, Ernest. 1979. *The Structure of Science: Problems in the Logic of Scientific Explanation*. Indianapolis, Ind.: Hackett.
- Nagel, Ilene H. and Barry L. Johnson. 1994. "The role of gender in a structured sentencing system: Equal treatment, policy choices, and the sentencing of female offenders of the United States sentencing guidelines." *Journal of Criminal Law* and Criminology 85:181–221.
- Nakell, Barry and Kenneth A. Hardy. 1987. *The Arbitrariness of the Death Penalty*. Philadelphia, Pa.: Temple University Press.
- Natapoff, Alexandra. 2006. "Beyond unreliable: How snitches contribute to wrongful convictions." *Golden Gate University Law Review* 37:107–129.
- Nietzsche, Friedrich W. [1888] 1998. Twilight of the Idols, or How to Philosophize with a Hammer, edited by D. Large. New York: Oxford University Press.
- Nobiling, Tracey, Cassia Spohn, and Miriam Delone. 1998. "A tale of two counties: Unemployment and sentence severity." *Justice Quarterly* 15:459–485.
- Norris, Michael, Christopher Birkbeck, and Luis Gerardo Gabaldón. 2006. "Social geometry and force: A partial test of Black's Theory of Law with Mexican, U.S., and Venezuelan police." *Journal of Contemporary Criminal Justice* 22:324–346.
- Novak, Kenneth J., James Frank, Brad W. Smith, and Robin Shepard Engel. 2002. "Revisiting the decision to arrest: Comparing beat and community officers." *Crime and Delinquency* 48:70–98.
- Oberer, Walter E. 1961. "Does disqualification of jurors for scruples against capital punishment constitute denial of fair trial on issue of guilt?" *Texas Law Review* 39:545–567.
- Ogletree, Charles J., Jr. 2002. "Black man's burden: Race and the death penalty in America." *Oregon Law Review* 81:15–38.
- —. 2004. All Deliberate Speed: Reflections on the First Half-Century of Brown v. Board of Education. Boston, Mass.: Norton.
- Ogletree, Charles J., Jr. and Austin Sarat (Eds.). 2006. From Lynch Mobs to the Killing State: Race and the Death Penalty in America. New York: New York University Press.
- Oliver, Melvin L. and Thomas M. Shapiro. 1995. *Black Wealth/White Wealth: A New Perspective on Racial Inequality*. New York: Routledge.

- —. 2001. "Wealth and racial stratification." Pp. 222–251 in America Becoming: Racial Trends and Their Consequences, vol. 2, edited by N. Smelser, W. J. Wilson, and F. Mitchell. Washington, D.C.: National Academies Press.
- Omi, Michael and Howard Winant. 1986. *Racial Formation in the United States: From the 1960s to the 1980s*. New York: Routledge.
- Oshinksky, David M. 1996. Worse than Slavery: Parchman and the Ordeal of Jim Crow Justice. New York: Free Press.
- Pager, Devah. 2007. Marked: Race, Crime, and Finding Work in an Era of Mass Incarceration. Chicago, Ill.: University of Chicago Press.
- Pampel, Fred C. and Kirk R. Williams. 2000. "Intimacy and homicide: Compensating for missing data in SHR." *Criminology* 38:661–680.
- Park, Robert E. 1950. Race and Culture. Glencoe, Ill.: Free Press.
- Parker, Karen F., Mari A. Dewees, and Michael L. Radelet. 2001. "Racial bias and the conviction of the innocent." Pp. 114–134 in *Wrongly Convicted: Perspectives on Failed Justice*, edited by S. D. Westervelt and J. A. Humphrey. Piscataway, N.J.: Rutgers University Press.
- Parker, Karen F., Brian J. Stults, and Stephen K. Rice. 2005. "Racial threat, concentrated disadvantage and social control: Considering the macro-level sources of variation in arrests." *Criminology* 43:1111–1134.
- Parsons, Talcott. 1937. *The Structure of Social Action: A Study in Social Theory*. New York: McGraw-Hill.
- —. 1938. "The role of theory in social research." American Sociological Review 3:13–20.
- —. 1951. The Social System. New York: Free Press.
- Parsons, Talcott and Edward A. Shils (Eds.). 1951. *Toward a General Theory of Action*. Cambridge, Mass.: Harvard University Press.
- Passell, Peter and John B. Taylor. 1977. "The deterrent effect of capital punishment: Another view." *American Economic Review* 67:445–451.
- Paternoster, Raymond. 1984. "Prosecutorial discretion in requesting the death penalty: A case of victim-based racial discrimination." *Law and Society Review* 18:347–478.
- —. 1991. "Prosecutorial discretion and capital sentencing in North and South Carolina." Pp. 39–52 in *The Death Penalty in America: Current Research*, edited by R. M. Bohm. Cincinnati, Ohio: Anderson.

- Paternoster, Raymond and Robert Brame. 2003. "An Empirical Analysis of Maryland's Death Sentencing System with Respect to the Influence of Race and Legal Jurisdiction." Department of Criminology, University of Maryland, College Park, Md. Unpublished manuscript.
- Paternoster, Raymond, Robert Brame, and Sarah Bacon. 2007. *The Death Penalty: America's Experience with Capital Punishment*. New York: Oxford University Press.
- Paternoster, Raymond, Robert Brame, Sarah Bacon, and Andrew Ditchfield. 2004. "Justice by geography and race: The administration of the death penalty in Maryland, 1978-1999." University of Maryland Law Journal of Race, Religion, Gender and Class 4:1–97.
- Paternoster, Raymond and Ann Marie Kazyaka. 1988. "The administration of the death penalty in South Carolina: Experiences over the first few years." *South Carolina Law Review* 39:245–414.
- Patterson, Orlando. 1982. *Slavery and Social Death: A Comparative Study*. Cambridge, Mass.: Harvard University Press.
- Pattillo-McCoy, Mary. 1999. *Black Picket Fences: Privilege and Peril Among the Black Middle Class*. Chicago, Ill.: University of Chicago Press.
- Pattillo, Mary. 2007. *Black on the Block: The Politics of Race and Class in the City.* Chicago, Ill.: University of Chicago Press.
- Perdue, William D. 1986. *Sociological Theory: Explanation, Paradigm, and Ideology*. Palo Alto, Calif.: Mayfield Publishing.
- Pérez, Adriana, Rodolfo J. Dennis, Jacky F. A. Gil, Martín A. Rondón, and Adriana López. 2002. "Use of the mean, hot deck, and multiple imputation techniques to predict outcome in intensive care unit patients in Colombia." *Statistics in Medicine* 21:3885–3896.
- Peters, Jeremy W. 2007. "Corzine Signs Bill Ending Executions, Then Commutes Sentences of 8." *New York Times*, December 18, p. B03.
- Petersilia, Joan. 1985. "Racial disparities in the Criminal Justice System: A summary." *Crime and Delinquency* 31:15–34.
- Peterson, Ruth D. and John Hagan. 1984. "Changing conceptions of race: Towards an account of anomalous findings of sentencing research." *American Sociological Review* 49:56–70.

- Petrin, Robert A. 2005. "Item non-response and multiple imputation for hierarchical linear models: An overview of issues and solutions." Working Paper, NORC/University of Chicago, Chicago, Ill.
- Pettit, Becky and Bruce Western. 2004. "Mass imprisonment and the life course: Race and class inequality in U.S. incarceration." *American Sociological Review* 69:151–169.
- Phillips, Scott. 2003. "The social structure of vengeance: A test of Black's model." *Criminology* 41:673–708.
- Phillips, Scott and Mark Cooney. 2005. "Aiding peace, abetting violence: Third parties and the management of conflict." *American Sociological Review* 70:334–354.
- Piehl, Anne M. and Shawn D. Bushway. 2007. "Measuring and explaining charge bargaining." *Journal of Quantitative Criminology* 23:105–125.
- Pierce, Glenn L. and Michael L. Radelet. 2002. "Race, region, and death sentencing in Illinois." *Oregon Law Review* 81:39–96.
- —. 2005. "The impact of legally inappropriate factors on death sentencing for California homicides, 1990-1999." Santa Clara Law Review 46:1–47.
- Pieterse, Jan Nederveen. 1992. White on Black: Images of Africa and Blacks in Western Popular Culture. New Haven, Conn.: Yale University Press.
- Pinheiro, José C. and Douglas M. Bates. 1995. "Approximations to the log-likelihood function in the nonlinear mixed-effects model." *Journal of Computational and Graphical Statistics* 4:12–35.
- Piquero, Alex R. and Stephen Tibbetts. 1996. "Specifying the direct and indirect effects of low self-control and situational factors in an offenders' decision making: Toward a more complete model of rational offending." *Justice Quarterly* 13:481–510.
- Pitts, James P. 1974. "The study of race consciousness: Comments on new directions." *American Journal of Sociology* 80:665–687.
- Pojman, Louis P. 2003. "Why the death penalty is morally permissible." Pp. 51–75 in Debating the Death Penalty: Should America Have Capital Punishment?, edited by H. A. Bedau and P. G. Cassell. New York: Oxford University Press.

- Pope, Carl E. 1976. "The influence of social and legal factors on sentencing dispositions: A preliminary analysis of offender based transaction statistics." *Journal of Criminal Justice* 4:203–221.
- Pope, Carl E., Rick Lovell, and Heidi M. Hsia. 2002. Disproportionate Minority Confinement: A Review of the Research Literature from 1989 through 2001. NCJ 198428. Washington, D.C.: U.S. Department of Justice, Office of Juvenile Justice and Delinquency Prevention.
- Popper, Karl. 1964. The Poverty of Historicism. New York: Harper and Row.
- -. [1934] 1968. The Logic of Scientific Discovery, 2d ed. New York: Harper and Row.
- Portes, Alejandro and Rubén G. Rumbaut. 2001. *Legacies: The Story of the Immigrant Second Generation*. Berkeley, Calif. and New York: University of California Press and Russell Sage Foundation.
- Posner, Richard A. 2003. Economic Analysis of Law, 6th ed. New York: Aspen.
- Possley, Maurice and Steve Mills. 2003. "Clemency For All." *Chicago Tribune*, January 12, p. A1.
- Post, Leonard. 2006. "More states review death penalty laws." *National Law Journal*, January 16, p. 4.
- Pound, Roscoe. 1910. "Law in books and law in action." *American Law Review* 44:12–34.
- -. 1942. Social Control Through Law. New Haven, Conn.: Yale University Press.
- Poveda, Tony G. 2001. "Estimating wrongful convictions." *Justice Quarterly* 18:689–708.
- —. 2006. "Geographic location, death sentences and executions in post-Furman Virginia." *Punishment and Society* 8:423–442.
- Pratt, Travis C. 1998. "Race and sentencing: A meta-analysis of conflicting empirical research results." *Journal of Criminal Justice* 26:513–523.
- Pridemore, William Alex. 2000. "An empirical examination of commutations and executions in post-*Furman* capital cases." *Justice Quarterly* 17:159–183.
- Provine, Doris Marie. 2006. "Creating racial disadvantage: The case of crack cocaine." Pp. 277–294 in *The Many Colors of Crime: Inequalities of Race, Ethnicity and Crime in America.*, edited by R. D. Peterson, L. Krivo, and J. Hagan. New York: New York University Press.

- Puhani, Patrick A. 2000. "The Heckman correction for sample selection bias and its critique: A short survey." *Journal of Economic Surveys* 14:53–68.
- Quadagno, Jill. 1996. The Color of Welfare: How Racism Undermined the War on Poverty, 2d ed. New York: Oxford University Press.
- Quillian, Lincoln and Devah Pager. 2001. "Black neighbors, higher crime? The role of racial stereotypes in evaluations of neighborhood crime." *American Journal of Sociology* 107:717–767.
- Quinney, Richard. 1970. *The Social Reality of Crime*. Boston, Mass.: Little, Brown and Company.
- Rabe-Hesketh, Sophia and Anders Skrondal. 2005. *Multilevel and Longitudinal Modeling Using Stata*. College Station, Tex.: Stata Press.
- Rabe-Hesketh, Sophia, Anders Skrondal, and Andrew Pickles. 2002a. "Multilevel Selection Models Using *GLLAMM*." Stata User Group, May, Maastricht, Netherlands.
- 2002b. "Reliable estimation of generalized linear mixed models using adaptive quadrature." *Stata Journal* 2:1–21.
- Radcliffe-Brown, Alfred R. [1935] 1965. *Structure and Function in Primitive Society: Essays and Addresses*. New York: Free Press.
- Radelet, Michael L. 1981. "Racial characteristics and the imposition of the death penalty." *American Sociological Review* 46:918–927.
- —. 1989. "Executions of Whites for crimes against Blacks: Exceptions to the rule?" Sociological Quarterly 30:529–544.
- —. 2003. "Capital punishment in Colorado, 1859-1972." University of Colorado Law Review 74:885–1010.
- Radelet, Michael L. and Ronald L. Akers. 1996. "Deterrence and the death penalty: The view of the experts." *Journal of Criminal Law and Criminology* 87:1–16.
- Radelet, Michael L. and Hugo A. Bedau. 1998. "The execution of the innocent." *Law and Contemporary Problems* 61:105–124.

- Radelet, Michael L., Hugo A. Bedau, and Constance E. Putnam. 1992. *In Spite of Innocence: Erroneous Convictions in Capital Cases*. Boston, Mass.: Northeastern University Press.
- Radelet, Michael L. and Marian J. Borg. 2000. "The changing nature of death penalty debates." *Annual Review of Sociology* 26:43–61.
- Radelet, Michael L. and Michael Mello. 1986. "Executing those who kill blacks: An *unusual* case study." *Mercer Law Review* 37:911–925.
- Radelet, Michael L. and Glenn L. Pierce. 1985. "Race and prosecutorial discretion in homicide cases." *Law and Society Review* 19:587–621.
- —. 1991. "Choosing those who will die: Race and the death penalty in Florida." University of Florida Law Review 43:1–34.
- Radelet, Michael L. and Margaret Vandiver. 1983. "The Florida Supreme Court and death penalty appeals." *Journal of Criminal Law and Criminology* 74:913–926.
- —. 1986. "Race and capital punishment: An overview of the issues." Crime and Social Justice 25:94–113.
- Raghunathan, Trivellore E., James M. Lepkowski, John van Hoewyk, and Peter Solenberger. 2001. "A multivariate technique for multiply imputing missing values using a sequence of regression models." *Survey Methodology* 27:85-95.
- Ragin, Charles. 1987. *The Comparative Method: Moving Beyond Qualitative and Quantitative Strategies*. Berkeley and Los Angeles, Calif.: University of California Press.
- Rankin, Bill, Heather Vogel, and Alice Wertheim. 2007. "High court botched death reviews." *Atlanta Journal-Constitution*, Sept. 26, p. A1.
- Rasbash, Jon, Fiona Steele, William Browne, and Bob Prosser. 2004. A User's Guide to MLwiN 2.01. London: Institute of Education, University of London.
- Raudenbush, Stephen W. and Anthony S. Bryk. 2002. *Hierarchical Linear Models: Applications and Data Analysis Methods*, 2d ed. Thousand Oaks, Calif.: Sage.
- Raudenbush, Stephen W., Anthony S. Bryk, Yuk Fai Cheong, and Richard Congdon. 2004. HLM 6: Hierarchical Linear and Nonlinear Modeling. Lincolnwood, Ill.: Scientific Software International, Inc.
- Raudenbush, Stephen W. and Robert J. Sampson. 1999. "Ecometrics: Toward a science of assessing ecological settings, with application to the systematic social

observation of neighborhoods." Pp. 1–41 in *Sociological Methodology*, vol. 29, edited by R. M. Stolzenberg. Oxford: Blackwell.

- Raudenbush, Stephen W., Meng-Li Yang, and Matheos Yosef. 2000. "Maximum likelihood for generalized linear models with nested random effects via high-order, multivariate Laplace approximation." *Journal of Computational and Graphical Statistics* 9:141–157.
- Rawls, Anne Warfield. 2000. ""Race" as an interaction order phenomenon: W.E.B. Du Bois's "Double Consciousness" thesis revisited." Sociological Theory 18:241– 274.
- Reed, Emily F. 1993. *The Penry Penalty: Capital Punishment and Offenders with Mental Retardation*. New York: University of American Press.
- Regoeczi, Wendy C. and Marc Riedel. 2003. "The application of missing data estimation models to the problem of unknown victim/offender relationships in homicide cases." *Journal of Quantitative Criminology* 19:155–183.
- Reilly, Marie. 1993. "Data analysis using hot deck multiple imputation." *The Statistician* 42:307–313.
- Reiman, Jeffrey H. 1979. The Rich Get Richer and the Poor Get Prison: Ideology, Class, and Criminal Justice. New York: John Wiley and Sons.
- Reiter, Jerome P. and Trivellore E. Raghunathan. 2007. "The multiple adaptations of multiple imputation." *Journal of the American Statistical Association* 102:1462-1471.
- Reiter, Jerome P., Trivellore E. Raghunathan, and Satkartar K. Kinney. 2006. "The importance of modeling the sampling design in multiple imputation for missing data." *Survey Methodology* 32:143-150.
- Religious Organizing Against the Death Penalty Project. 1998. "The Death Penalty: The Religious Community Calls for Abolition." American Friends Service Committee, Philadelphia, Pa.
- Rhode, Deborah L. and Charles J. Ogletree, Jr. (Eds.). 2004. *Brown at 50: The Unfinished Legacy*. Washington, D.C.: American Bar Association.
- Ridgeway, Cecilia L. 1991. "The social construction of status value: Gender and other nominal characteristics." *Social Forces* 70:367–386.
- Ridgeway, Cecilia L., Elizabeth H. Boyle, Katherine J. Kuipers, and Dawn T. Robinson. 1998. "How do status beliefs develop? The role of resources and interaction." *American Sociological Review* 63:331–350.

- Riedel, Marc. 1976. "Discrimination in the imposition of the death penalty: A comparison of the characteristics of offenders sentenced Pre-*Furman* and Post-*Furman*." *Temple Law Quarterly* 49:261–287.
- Rise, Eric W. 1995. *The Martinsville Seven: Race, Rape, and Capital Punishment*. Charlottesville, Va.: University of Virginia Press.
- Roberts, Dorothy E. 1999. "Foreword: Race, vagueness, and the social meaning of ordermaintenance policing." *Journal of Criminal Law and Criminology* 89:775–836.
- Robinson, Paul H. and John M. Darley. 2003. "The role of deterrence in the formulation of criminal law rules: At its worst when doing its best." *Georgetown Law Journal* 91:949–1002.
- Robison, Clay. 2005. "Life without parole option falls two votes short in the Senate." *Houston Chronicle*, April 6, p. B3.
- Rodríguez, Germán and Noreen Goldman. 2001. "Improved estimation procedures for multilevel models with binary response: A case study." *Journal of the Royal Statistical Society* (Series A) 164:339–355.
- Rohrlich, Ted and Fredric N. Tulsky. 1996. "Not all L.A. murder cases are equal." *Los Angeles Times*, December 3, p. A1.
- Root, Michael. 2000. "How we divide the world." Philosophy of Science 67:S628-S639.
- Rosen, Lawrence. 1989. *The Anthropology of Justice: Law as Culture in Islamic Society*. Cambridge: Cambridge University Press.
- Rosen, Paul A. 1972. *The Supreme Court and Social Science*. Urbana, Ill.: University of Illinois Press.
- Rosen, Richard A. 1986. "The 'especially heinous' aggravating circumstance in capital cases--the standardless standard." *North Carolina Law Review* 64:941–992.
- Rosenbaum, Paul R. 2002. Observational Studies, 2d ed. New York: Springer-Verlag.
- Rosenbaum, Paul R. and Donald B. Rubin. 1983. "The central role of the propensity score in observational studies for causal effects." *Biometrika* 70:41–55.
- Ross, Edward Alsworth. [1901] 1920. Social Control: A Survey of the Foundations of Order. New York: Macmillan.
- Ross, Lee E. and Willie J. Edwards. 1998. "Publishing among African American criminologists: A devaluing experience?" *Journal of Criminal Justice* 26:29–40.

- Royal Commission on Criminal Justice. 1993. *Report of the Royal Commission on Criminal Justice*. Report Cm 2263. London: HMSO.
- Royal Legislature of Georgia. [1755] 1910. "An act for the better ordering and governing negroes and other slaves in this province." Pp. 102–144 in *The Colonial Records* of the State of Georgia: Statutes Enacted by the Royal Legislature of Georgia from its First Session in 1754 to 1768, vol. 18, edited by A. D. Candler. Atlanta, Ga.: Chas. P. Byrd.
- Royston, Patrick. 2004. "Multiple imputation of missing values." *Stata Journal* 4:227–241.
- Rubin, Donald B. 1976. "Inference in missing data." Biometrika 63:581–592.
- —. 1987. *Multiple Imputation for Nonresponse in Surveys*. New York: John Wiley and Sons.
- —. 1996. "Multiple imputation after 18+ years." *Journal of the American Statistical Association* 91:473–489.
- Rumble, Wilfrid E., Jr. 1965. "Legal realism, sociological jurisprudence and Mr. Justice Holmes." *Journal of the History of Ideas* 26:547–566.
- Rupp, Ashley. 2003. "Death penalty prosecutorial charging decisions and county budgetary restrictions: Is the death penalty arbitrarily applied based on county funding?" *Fordham Law Review* 71:2735–2780.
- Russell, Bertrand. [1913] 1992. "On the notion of cause." Pp. 193–210 in *The Collected Papers of Bertrand Russell: Logical and Philosophical Papers*, vol. 6, edited by J. G. Slater. London: Routledge Press.
- Russell, Gregory D. 1993. *The Death Penalty and Racial Bias: Overturning Supreme Court Assumptions*. Westport, Conn.: Greenwood Press.
- Ryan, George. 2003. "'I must act'." Pp. 218–234 in *Debating the Death Penalty: Should America Have Capital Punishment*?, edited by H. A. Bedau and P. G. Cassell. New York: Oxford University Press.
- Sack, Kevin. 1997. "No death sentence for drunken driver in student killings." *New York Times*, May 7, p. A1.
- Salzman, Jack, David L. Smith, and Cornel West (Eds.). 1995. *Encyclopedia of African American Culture and History*. New York: MacMillan.

- Sampson, Robert J. 1987. "Urban black violence: The effect of male joblessness and family disruption." *American Journal of Sociology* 93:348–382.
- —. 2003. "The neighborhood context of well-being." Perspectives in Biology and Medicine 46:S53–S64.
- —. 2006. "Collective efficacy theory: Lessons learned and directions for future inquiry." Pp. 149–167 in *Taking Stock: The Status of Criminological Theory*, vol. 15, *Advances in Criminological Theory*, edited by F. T. Cullen, J. P. Wright, and K. R. Blevins. New Brunswick, N.J.: Transaction.
- Sampson, Robert J. and W. Byron Groves. 1989. "Community structure and crime: Testing Social Disorganization Theory." *American Journal of Sociology* 94:774–802.
- Sampson, Robert J. and Janet Lauritsen. 1997a. "Racial and ethnic disparities in crime and criminal justice in the United States." Pp. 311–374 in *Ethnicity, Crime, and Immigration: Comparative and Cross National Perspectives*, vol. 21, edited by M. Tonry. Chicago, Ill.: University of Chicago Press.
- Sampson, Robert J. and Janet L. Lauritsen. 1997b. "Racial and ethnic disparities in crime and criminal justice in the United States." Pp. 311–374 in *Ethnicity, Crime, and Immigration: Comparative and Cross-National Perspectives*, edited by M. Tonry. Chicago, Ill.: University of Chicago Press.
- Sampson, Robert J., Doug McAdam, Heather MacIndoe, and Simón Weffer-Elizondo. 2005a. "Civil society reconsidered: The durable nature and community structure of collective civic action." *American Journal of Sociology* 111:673–714.
- Sampson, Robert J. and Jeffrey D. Morenoff. 2006. "Durable inequality: Spatial dynamics, social processes, and the persistence of poverty in Chicago neighborhoods." Pp. 176–203 in *Poverty Traps*, edited by S. Bowles, S. Durlauf, and K. Hoff. Princeton, N.J. and New York: Princeton University Press and Russell Sage Foundation.
- Sampson, Robert J., Jeffrey D. Morenoff, and Felton Earls. 1999. "Beyond social capital: Spatial dynamics of collective efficacy for children." *American Sociological Review* 64:633–660.
- Sampson, Robert J., Jeffrey D. Morenoff, and Stephen W. Raudenbush. 2005b. "Social anatomy of racial and ethnic disparities in violence." *American Journal of Public Health* 95:224–232.
- Sampson, Robert J., Stephen W. Raudenbush, and Felton Earls. 1997. "Neighborhoods and violent crime: A multilevel study of collective efficacy." *Science* 277:918– 924.

- Sandys, Marla. 1995. "Cross-overs: Capital Jurors who change their minds about the punishment: A litmus test for sentencing guidelines." *Indiana Law Journal* 70:1183–1221.
- Sandys, Marla and Edmund F. McGarrell. 1995. "Attitudes toward capital punishment: Preference for the penalty or mere acceptance?" *Journal of Research in Crime and Delinquency* 32:191–213.
- Sanger, Robert M. 2003. "Comparison of the Illinois Commission Report on Capital Punishment with the capital punishment system in California." *Santa Clara Law Review* 44:101–234.
- Sarat, Austin. 1989. "Donald Black discovers legal realism: From pure science to policy science in the sociology of law." *Law and Society Review* 14:765–785.
- Sartori, Anne E. 2003. "An estimator for some binary-outcome selection models without exclusion restrictions." *Political Analysis* 11:111–138.
- Savelsberg, Joachim J. 1992. "Law that does not fit society: Sentencing guidelines as a neoclassical reaction to the dilemmas of substantive law." *American Journal of Sociology* 97:1346–1381.
- —. 1994. "Knowledge, domination, and criminal punishment." American Journal of Sociology 99:911–943.
- Scalia, Antonin. 1997. *A Matter of Interpretation: Federal Courts and the Law*, edited by A. Gutmann. Princeton, N.J.: Princeton University Press.
- Schafer, Joseph L. 1997. *Analysis of Incomplete Multivariate Data*. Bocan Raton, Fla.: Chapman and Hall.
- Schafer, Joseph L. and John W. Graham. 2002. "Missing data: Our view of the state of the art." *Psychological Methods* 7:147–177.
- Schall, Robert. 1991. "Estimation in generalized linear models with random effects." *Biometrika* 78:719–727.
- Schanzenbach, Max M. 2005a. "Have federal judges changed their sentencing practices?: The shaky empirical foundations of the Feeney Amendment." *Journal of Empirical Legal Studies* 2:1–48.
- —. 2005b. "Racial and sex disparities in prison sentences: The effect of district-level judicial demographics." *Journal of Legal Studies* 34:57–92.

- Schanzenbach, Max M. and Emerson H. Tiller. 2007. "Strategic judging under the United States Sentencing Guidelines: Positive political theory and evidence." *Journal of Law, Economics, and Organization* 23:24–56.
- Schanzenbach, Max M. and Michael L. Yaeger. 2006. "Prison time, fines, and federal white-collar criminals: The anatomy of a racial disparity." *Journal of Criminal Law and Criminology* 96:757–793.
- Scheck, Barry, Peter Neufeld, and Jim Dwyer. 2000. Actual Innocence: Five Days to Execution and Other Dispatches From the Wrongfully Convicted. New York: Doubleday.
- Scheppele, Kim L. 1994. "Legal theory and social theory." *Annual Review of Sociology* 20:383–406.
- Schiff, David N. 1981. "Law as a social phenomenon." Pp. 151–166 in Sociological Approaches to Law, edited by A. Podgorecki and C. J. Whelan. New York: St. Martin's Press.
- Schreiber, Sven. 2008. "The Hausman test statistic can be negative even asymptotically." *Journal of Economics and Statistics* 228:394-405.
- Schwartz, Gideon. 1978a. "Estimating the dimension of a model." *Annals of Statistics* 6:461–464.
- Schwartz, Phillip J. 1988. *Twice Condemned: Slaves and the Criminal Courts of Virginia,* 1705-1865. Baton Rouge, La.: Louisiana State University Press.
- Schwartz, Richard D. 1978b. "Moral order and sociology of law: Trends, problems, and prospects." *Annual Review of Sociology* 4:577–601.
- Schwartz, Richard D. and Jerome Skolnick. 1962. "Two studies of legal stigma." *Social Problems* 10:133–142.
- Sears, David O., James Sidanius, and Lawrence Bobo (Eds.). 2000. *Racialized Politics: The Debate About Racism in America*. Chicago, Ill.: University of Chicago Press.
- Sellin, Thorsten. 1928. "The negro criminal: A statistical note." *Annals of the American Academy of Political and Social Science* 140:52–64.
- —. 1935. "Race prejudice in the administration of justice." American Journal of Sociology 41:212–217.
- -. 1938. Culture, Conflict and Crime. New York: Social Science Research Council.
- -. 1959. The Death Penalty. Philadelphia, Pa.: American Law Institute.

- Senechal de la Roche, Roberta. 1990. *The Sociogenesis of a Race Riot: Springfield, Illinois, in 1908.* Chapel Hill, N.C.: University of North Carolina Press.
- —. 1997b. "The sociogenesis of lynching." Pp. 48–76 in Under Sentence of Death: Lynching in the South, edited by W. F. Brundage. Chapel Hill, N.C.: University of North Carolina Press.
- -. 2001. "Why is collective violence collective?" Sociological Theory 19:126–144.
- Seron, Carroll and Frank Munger. 1996. "Law and inequality: Race, gender...and, of course, class." *Annual Review of Sociology* 22:187–212.
- Serrano, Richard A. 1994. "A grim life on military death row." *Los Angeles Times*, July 12, p. A1.
- Shapiro, Bruce. 2001. "Dead Reckoning: A world effort to force an end to the U.S. death penalty is gaining strength." *The Nation*, August 6, p. 14.
- Shapiro, Susan P. 1990. "Collaring the crime, not the criminal: Reconsidering the concept of White-Collar crime." *American Sociological Review* 55:346–365.
- Shatz, Steven F. and Nina Rivkind. 1997. "The California death penalty: Requiem for *Furman?*" *New York University Law Review* 72:1283–1343.
- Shaw, Clifford and Henry D. McKay. 1942. *Juvenile Delinquency in Urban Areas*. Chicago, Ill.: University of Chicago Press.
- Shelby, Tommie. 2005. *We Who Are Dark: The Philosophical Foundations of Black Solidarity*. Cambridge, Mass.: Harvard University Press.
- Shepherd, Joanna M. 2004. "Murders of Passion, Execution Delays, and the Deterrence of Capital Punishment." *Journal of Legal Studies* 33:283–321.
- —. 2005. "Deterrence versus brutalization: Capital punishment's differing impacts among states." *Michigan Law Review* 104:203–255.
- Sherman, Lawrence W. 1978. "Review of *The Behavior of Law*." *Contemporary Sociology* 7:11–15.
- —. 1993. "Defiance, deterrence, and irrelevance: A theory of the criminal sanction." Journal of Research in Crime and Delinquency 30:445–473.

- Shihadeh, Edward S. and Darrell J. Steffensmeier. 1994. "Economic inequality, family disruption, and urban black violence: Cities as units of stratification and social control." *Social Forces* 73:729–751.
- Short, James F. 1997. *Poverty, Ethnicity, and Violent Crime*. Boulder, Colo.: Westview Press.
- Sidanius, James and Felicia Pratto. 1999. Social Dominance: An Intergroup Theory of Social Hierarchy and Oppression. Cambridge: Cambridge University Press.
- Simmel, Georg. 1909. "The problem of sociology." *American Journal of Sociology* 15:289–320.
- —. [1908] 1950. The Sociology of Georg Simmel, edited by K. H. Wolff. New York: Free Press.
- —. [1908] 1964. *Conflict and the Web of Group Affiliations*. Translated by K. H. Wolff and R. Bendix. New York: Free Press.
- Simon, Jonathan. 2007. *Governing Through Crime: How the War on Crime Transformed American Democracy and Created a Culture of Fear*. New York: Oxford University Press.
- Simon, Rita James and Dagny A. Blaskovich. 2002. A Comparative Analysis of Capital Punishment: Statutes, Policies, Frequencies, and Public Attitudes the World Over. Oxford: Lexington Books.
- Skidmore, Thomas E. 1993. "Bi-racial USA vs. Multi-racial Brazil: Is the contrast still valid?" *Journal of Latin American Studies* 25:373–386.
- Skrondal, Anders and Sophia Rabe-Hesketh. 2004. *Generalized Latent Variable Modeling: Multilevel, Longitudinal, and Structural Equation Modeling.* Bocan Raton, Fla.: Chapman and Hall.
- Sloan, John J. and JoAnn L. Miller. 1990. "Just deserts, the severity of punishment and judicial sentencing decisions." *Criminal Justice Policy Review* 4:19–38.
- Smith, Abbot Emerson. [1947] 1971. Colonists in Bondage: White Servitude and Convict Labor in America, 1607-1776. New York: Norton.
- Smith, Brent L. and Kelly R. Damphouse. 1996. "Punishing political offenders: The effect of political motive on federal sentencing decisions." *Criminology* 34:289– 319.
- Smith, Bruce P. 2005. "The history of wrongful execution." *Hastings Law Journal* 56:1185–1233.

- Smith, Douglas A. 1987a. "Police response to interpersonal violence: Defining the parameters of legal control." *Social Forces* 65:767–782.
- Smith, Herbert L. 1997. "Matching with multiple controls to estimate treatment effects in observational studies." Pp. 325–353 in *Sociological Methodology*, vol. 27, edited by A. E. Raftery. Oxford: Blackwell.
- Smith, Kevin B. 2004. "The politics of punishment: Evaluating political explanations of incarceration rates." *The Journal of Politics* 66:925–938.
- Smith, M. Dwayne. 1987b. "Patterns of discrimination in assessments of the death penalty: The case of Louisiana." *Journal of Criminal Justice* 15:279–286.
- Smith, Rogers M. 2003. "Black and white after *Brown*: Constructions of race in modern Supreme Court decisions." University of Pennsylvania Journal of Constitutional Law 5:709–733.
- Smith, Ryan A. 2002. "Race, gender, and authority in the workplace: Theory and research." *Annual Review of Sociology* 28:509–542.
- Snijders, Tom A. B. and Roel J. Bosker. 1999. *Multilevel Analysis: An Introduction to Basic and Advanced Multilevel Modeling*. Thousand Oaks, Calif.: Sage.
- Snoddy, Mariah. 2002. "The fight of southern attorney P. Walter Jones: Overturning "Yellow Paper-Black Man" discrimination in the Georgia jury system and desegregating the Georgia prison system." *Juris*, Spring, pp. 28–32.
- Sobel, Michael E. 1995. "Causal inference in the social and behavioral sciences." Pp. 1– 38 in *Handbook of Statistical Modeling for the Social and Behavioral Sciences*, edited by G. Arminger, C. C. Clogg, and M. E. Sobel. New York: Plenum Press.
- Somers, Margaret R. 1998. "We're no angels': Realism, rational choice, and relationality in social science." *American Journal of Sociology* 104:722–784.
- Sommers, Samuel R. and Michael I. Norton. 2007. "Race-based judgments, race-neutral justifications: Experimental examination of peremptory use and the *Batson* challenge procedure." *Law and Human Behavior* 31:261-273.
- Songer, Donald R. and Reginald S. Sheehan. 1999. "Who wins on appeal? Upperdogs and Underdogs in the United States Courts of Appeals." Pp. 323–342 in *The Social Organization of Law*, 2d ed., edited by M. P. Baumgartner. San Diego, Calif.: Academic Press.

- Songer, Michael J. and Issac Unah. 2006. "The effect of race, gender, and location on prosecutorial decisions to seek the death penalty in South Carolina." *South Carolina Law Review* 58:161–209.
- Sorensen, Jon R. and Don Stemen. 2002. "The effect of state sentencing policies on incarceration rates." *Crime and Delinquency* 48:456–475.
- Sorensen, Jon R. and Donald H. Wallace. 1995. "Capital punishment in Missouri: Examining the issue of racial disparity." *Behavioral Sciences and the Law* 13:61–80.
- —. 1999. "Prosecutorial discretion in seeking death: An analysis of racial disparity in the pretrial stages of case processing in a Midwestern county." *Justice Quarterly* 16:559–577.
- Sorenson, Susan B. and Richard A. Berk. 1998. "News media portrayals and the epidemiology of homicide." *American Journal of Public Health* 88:1510–1514.
- Sorokin, Pitirim A. 1937. *Social and Cultural Dynamics*. New York: American Book Company.
- Sousa Santos, Boaventura de. 1977. "The law of the oppressed: The construction and reproduction of legality in Pasargada." *Law and Society Review* 12:5–126.
- South, Scott J. and Kyle D. Crowder. 1997. "Escaping distressed neighborhoods: Individual, community, and metropolitan influences." *American Journal of Sociology* 102:1040–1085.
- —. 1999. "Neighborhood effects on family formation: Concentrated poverty and beyond." *American Sociological Review* 64:113–132.
- Spangenberg Group. 2002. "Status of Indigent Defense in Georgia: A Study for the Chief Justice's Commission on Indigent Defense, Part I." The Spangenberg Group, West Newton, Mass.
- Spencer, Herbert. 1876. *The Principles of Sociology*, vol. 1. London: Williams and Norgate.
- Spiegelhalter, David, Andrew Thomas, Nicky Best, and Dave Lunn. 2003. *WinBUGS User Manual: Version 1.4.* Cambridge: MRC Biostatistics Unit.
- Spindel, Donna J. 1989. *Crime and Society in North Carolina, 1663-1776.* Baton Rouge, La.: Louisiana State University Press.
- Spinoza, Benedictus de. [1677] 1883. *The Ethics*. Translated by R. H. M. Elwes. London: Dover.

- Spitzer, Steven. 1983. "Marxist perspectives in the sociology of law." *Annual Review of Sociology* 9:103–124.
- Spohn, Cassia C. 1990a. "Decision making in sexual assault cases: Do black and female judges make a difference." *Women and Criminal Justice* 2:83–105.
- —. 1990b. "The sentencing decisions of black and white judges: Expected and unexpected similarities." *Law and Society Review* 24:1197–1216.
- Spohn, Cassia C. and Jerry Cederblom. 1991. "Race and disparities in sentencing: A test of the liberation hypothesis." *Justice Quarterly* 8:305–328.
- Spohn, Cassia C., John Gruhl, and Susan Welch. 1987. "The impact of ethnicity and gender of defendants on the decision to reject or dismiss felony charges." *Criminology* 25:175–191.
- Spohn, Cassia C. and David Holleran. 2000. "The imprisonment penalty paid by young, unemployed black and Hispanic male offenders." *Criminology* 38:281–306.
- Spohn, Cassia C., Susan Welch, and John Gruhl. 1985. "Women defendants in court: The interaction between sex and race in convicting and sentencing." *Social Science Quarterly* 66:178–185.
- Stack, Steven. 1994. "Execution publicity and homicide in Georgia." *American Journal* of Criminal Justice 18:25–39.
- Stafford, Mark C. and Mark Warr. 1993. "A reconceptualization of general and specific deterrence." *Journal of Research in Crime and Delinquency* 30:123–135.
- Stanko, Elizabeth A. 1981/1982. "The impact of victim assessment on prosecutors' screening decisions: The case of the New York County District Attorney's Office." *Law and Society Review* 16:225–239.
- StataCorp. 2003. Stata Statistical Software: Release 8. College Station, Tex.: Stata Press.
- State of Georgia. 1848. "Slave Codes of the State of Georgia." in A Codification of the Statute Law of Georgia: Including the English Statutes of Force, 2d ed., edited by W. A. Hotchkiss. Atlanta, Ga.: Charles E. Grenville.
- Steen, Sara, Rodney L. Engen, and Randy R. Gainey. 2005. "Images of danger and culpability: Racial stereotyping, case processing, and criminal sentencing." *Criminology* 43:435–468.

- Steffensmeier, Darrell and Chester L. Britt. 2001. "Judge's race and judicial decisionmaking: Do black judges sentence differently?" Social Science Quarterly 82:749– 764.
- Steffensmeier, Darrell and Stephen Demuth. 2000. "Ethnicity and sentencing outcomes in the U.S. federal courts: Who is punished more harshly?" *American Sociological Review* 65:705–729.
- —. 2001. "Ethnicity and judges' sentencing decisions: Hispanic-Black-White comparisons." Criminology 39:145–178.
- —. 2006. "Does gender modify the effects of race-ethnicity on criminal sanctioning? Sentences for male and female white, black, and Hispanic Defendants." *Journal of Quantitative Criminology* 22:241–261.
- Steffensmeier, Darrell and Christopher G. Hebert. 1999. "Women and men policymakers: Does the judge's gender affect the sentencing of criminal defendants." *Social Forces* 77:1163–1196.
- Steffensmeier, Darrell J. and Emilie A. Allan. 1995. "Age-inequality and property crime: The effects of age-linked stratification on status-attainment processes on patterns of criminality across the life course." Pp. 95–115 in *Crime and Inequality*, edited by J. Hagan and R. Peterson. Stanford, Calif.: Stanford University Press.
- Steffensmeier, Darrell, John H. Kramer, and Cathy Streifel. 1993. "Gender and imprisonment decisions." *Criminology* 31:411–443.
- Steffensmeier, Darrell, John H. Kramer, and Jeffrey T. Ulmer. 1998. "The interaction of race, gender, and age in criminal sentencing: The punishment costs of being young, black, and male." *Criminology* 36:763–798.
- Steiker, Carol S. 2005. "No, capital Punishment is not morally required: Deterrence, deontology, and the death penalty." *Stanford Law Review* 58:751–790.
- Steiner, Benjamin D., William J. Bowers, and Austin Sarat. 1999. "Folk knowledge as legal action: Death penalty judgments and the tenet of early release in a culture of mistrust and punitiveness." *Law and Society Review* 33:461–505.
- Stinchcombe, Arthur L. 1968. Constructing Social Theories. Chicago, Ill.: University of Chicago Press.
- Stolzenberg, Lisa and Stewart J. D'Alessio. 1994. "Sentencing and unwarranted disparity: An empirical assessment of the long-term impact of sentencing guidelines in Minnesota." *Criminology* 32:301–310.

- Stolzenberg, Ross M. and Daniel A. Relles. 1997. "Tools for intuition about sample selection bias and its correction." *American Sociological Review* 62:494–507.
- Streib, Victor L. 2002a. "Death Penalty for Female Offenders--January 1, 1973 to June 30, 2002." Death Penalty Information Center, Washington, D.C.
- —. 2002b. "The Juvenile Death Penalty Today: Death Sentences and Executions for Juvenile Crimes--January 1, 1973 to June 30, 2002." Death Penalty Information Center, Washington, D.C.
- —. 2003. "Standing between the child and the executioner: The special role of defense counsel in juvenile death penalty cases." *American Journal of Criminal Law* 31:67–116.
- Stuart, Guy. 2003. Discriminating Risk: The U.S. Mortgage Lending Industry in the Twentieth Century. Ithaca, N.Y.: Cornell University Press.
- Sullivan, Dwight H. 1994. "The last line of defense: Federal habeas review of military death penalty cases." *Military Law Review* 144:1–76.
- Sumner, William Graham. [1906] 1940. Folkways: A Study of the Sociological Importance of Usages, Manners, Customs, Mores, and Morals. New York: New American Library.
- Sundby, Scott E. 1998. "The capital jury and absolution: The intersection of trial strategy, remorse, and the death penalty." *Cornell Law Review* 83:1557–1597.
- —. 2003. "The capital jury and empathy: The problem of worthy and unworthy victims." Cornell Law Review 88:343–377.
- Sunstein, Cass R. 1993. *The Partial Constitution*. Cambridge, Mass.: Harvard University Press.
- -. 1996. Legal Reasoning and Political Conflict. New York: Oxford University Press.
- (Ed.). 2000. Behavioral Law and Economics. New York: Cambridge University Press.
- Sunstein, Cass R. and Christina Jolls. 2006. "The law of implicit bias." *California Law Review* 94:969–998.
- Sunstein, Cass R. and Adrian Vermeule. 2005. "Is capital punishment morally required? Acts, omissions, and life-life trade-offs." *Stanford Law Review* 58:703–750.
- Sutherland, Edwin H. 1925. "Murder and the death penalty." *Journal of the American Institute of Criminal Law and Criminology* 51:522–529.

—. 1947. Principles of Criminology, 4th ed. Philadelphia, Pa.: Lippincott.

- Swedberg, Richard. 2006. "Max Weber's contribution to the economic sociology of law." Annual Review of Law and Social Science 2:61–81.
- Swinton, David H. 1990. "Racial inequality and reparations." Pp. 153–162 in *The Wealth* of *Races: The Present Value of Benefits from Past Injustices*, edited by R. F. America. Westport, Conn.: Greenwood Press.
- Tabak, Ronald J. and J. Mark Lane. 1989. "The execution of injustice: A cost and lackof-benefit analysis of the death penalty." *Loyola of Los Angeles Law Review* 23:59–146.
- Tamanaha, Brain Z. 1997. *Realistic Socio-Legal Theory: Pragmatism and a Social Theory of Law.* Oxford: Clarendon Press.
- Tanner, Martin A. and Wing Hung Wong. 1987. "The calculation of posterior distributions by data augmentation." *Journal of the American Statistical Association* 82:528–540.
- Terkel, Studs. 1992. *Race: How Blacks and Whites Think and Feel about the American Obsession*. New York: The New Press.
- Thaxton, Sherod and Robert Agnew. 2004. "The nonlinear effects of parental and teacher attachment on delinquency: Disentangling strain from social control explanations." *Justice Quarterly* 21:763–792.
- Thernstrom, Abigail and Stephan Thernstrom. 2003. *No Excuses: Closing the Racial Gap in Learning*. New York: Simon and Schuster.
- Thernstrom, Stephan and Abigail Thernstrom. 1997. *America in Black and White: One Nation, Indivisible.* New York: Simon and Schuster.
- Thomas, Miessha, Jerry Pennick, and Heather Gray. 2004. "What is African-American Land Ownership?", East Point, Ga.: The Federation of Southern Cooperatives/Land Assistance Fund. Retrieved March 5, 2005 (http://www.federationsoutherncoop.com/aalandown04.htm).
- Thomas, William I. and Florian Znaniecki. [1921] 1958. *The Polish Peasant in Europe* and America, 2d ed. New York: Dover.
- Thomson, Ernie. 1997. "Discrimination and the death penalty in Arizona." *Criminal Justice Review* 22:65–76.
- Thomson, Randall J. and Matthew T. Zingraff. 1981. "Detecting sentencing disparity: Some problems and evidence." *American Journal of Sociology* 86:869–880.

- Timasheff, Nicholas S. 1937. "What is 'Sociology of Law?'." American Journal of Sociology 43:225–235.
- Tittle, Charles R. 1995. *Control Balance: Toward a General Theory of Deviance*. Boulder, Colo.: Westview Press.
- Tocqueville, Alexis de. [1840] 1969. *Democracy in America*. Garden City, N.Y.: Anchor Books.
- Tolnay, Stewart E. and E. M. Beck. 1995. A Festival of Violence: An Analysis of Southern Lynchings, 1882-1930. Urbana, Ill.: University of Illinois Press.
- Tomlins, Christopher. 2007. "How autonomous is law?" *Annual Review of Law and Social Science* 3:45–68.
- Tonry, Michael. 1995. *Malign Neglect: Race, Crime, and Punishment in America*. New York: Oxford University Press.
- Trevino, A. Javier. 1996. *The Sociology of Law: Classical and Contemporary Perspectives.* New York: St. Martin's Press.
- Tucker, James. 1989. "Employee theft as social control." Deviant Behavior 10:319-334.
- —. 1999. The Therapeutic Corporation. New York: Oxford University Press.
- -. 2002. "Becoming a pure sociologist." Contemporary Sociology 31:661-664.
- Turk, Austin T. 1969. Criminality and Legal Order. Chicago, Ill.: Rand McNally.
- Turner, Jonathan H. 2002. "Why are elegant theories under-utilized by sociologists?" *Contemporary Sociology* 31:664–668.
- U.S. Bureau of the Census. 1991. *Census of Population and Housing, 1990: Summary Tape File 1*. Washington, D.C.: U.S. Government Printing Office.
- —. 1992. Census of Population and Housing, 1990: Summary Tape File 3. Washington, D.C.: U.S. Government Printing Office.
- —. 2001a. Census 2000 Summary File 1 [Georgia]. Washington, D.C.: U.S. Government Printing Office.
- -. 2001b. *Census 2000 Summary File 2 [Georgia]*. Washington, D.C.: U.S. Government Printing Office.
- 2002a. 2000 Census of Population and Housing: Summary File 3. Washington, D.C.: U.S. Government Printing Office.
- —. 2002b. Historical Census Statistics on Population Totals by Race, 1790 to 1990, and by Hispanic Origin, 1970 to 1990, for the United States, Regions, Divisions, and States. Washington, D.C.: U.S. Government Printing Office.
- —. 2003a. 2000 Census of Population and Housing: Summary File 4. Washington, D.C.: U.S. Government Printing Office.
- —. 2003b. U.S. Census Bureau Guidance on the Presentation and Comparison of Race and Hispanic Origin Data. Washington, D.C.: U.S. Government Printing Office.
- U.S. Congress. 1994. House Subcommittee on Civil and Constitutional Rights & Committee on the Judiciary. *Racial Disparities in Federal Death Penalty Prosecutions, 1988-1994:* Staff Report. 103rd Cong., 2d sess., pp. 1–8.
- U.S. Department of Agriculture. 2003. *Rural Education at a Glance*. Rural Development Research Report No. 98. Washington, D.C.: U.S. Government Printing Office.
- U.S. Department of Education, National Center for Educational Statistics. 1995. Use of School Choice. NCES 95-742R. Washington, D.C.: U.S. Government Printing Office.
- U.S. Department of Justice. 1991. Uniform Crime Reports. Washington, D.C.: U.S. Government Printing Office.
- ---, Office of Public Affairs. 1995. Justice Department Urges Congress to Retain Heavier Penalties for Crack Cocaine Offenses. 95-209. Washington, D.C.: U.S. Government Printing Office.
- —. 2000. The Federal Death Penalty System: A Statistical Survey, 1988-2000. Washington, D.C.: U.S. Government Printing Office.
- -... 2001. The Federal Death Penalty System: Supplementary Data, Analysis and Revised Protocols for Capital Case Review. Washington, D.C.: U.S. Government Printing Office.
- ---, Bureau of Justice Statistics. 2004a. *Prison and Jail Inmates at Midyear 2003*. Washington, D.C.: U.S. Government Printing Office.
- U.S. Department of Justice, Bureau of Justice Statistics. 2004b. *Capital Punishment in the United States, 1973-2002.* [MRDF]. Washington, D.C.: U.S. Department of Commerce, Bureau of the Census [producer]. Ann Arbor, Mich.: Inter-university Consortium for Political and Social Research, University of Michigan [distributor].

- U.S. Department of the Interior, Fish and Wildlife Service. 1999. 1980-1995 Participation in Fishing, Hunting, and Wildlife Watching: National and Regional Demographic Trends. Report 96-5. Washington, D.C.: U.S. Government Printing Office.
- U.S. General Accounting Office. 1990. *Death Penalty Sentencing: Research Indicates Patterns of Racial Disparities*. GGD-90-57. Washington, D.C.: U.S. Government Printing Office.
- —. 1994. Information on Minority-Targeted Scholarships. Washington, D.C.: U.S. Government Printing Office.
- U.S. Sentencing Commission. 1987. Supplementary Report on the Initial Sentencing Guidelines and Policy Statements. Washington, D.C.: U. S. Government Printing Office.
- -. 1991. Guidelines Manual. Washington, D.C.: U. S. Government Printing Office.
- —. 1995. Special Report to Congress: Cocaine and Federal Sentencing Policy. Washington, D.C.: Government Printing Office.
- Uggen, Christopher and Candace Kruttschnitt. 1998. "Crime in the breaking: Gender differences in desistance." *Law and Society Review* 32:339–366.
- Uggen, Christopher and Jeff Manza. 2002. "Democratic contraction? Political consequences of felon disenfranchisement in the United States." *American Sociological Review* 67:777–803.
- Ulmer, Jeffrey T. 2000. Social Worlds of Sentencing: Court Communities Under Sentencing Guidelines. Albany, N.Y.: State University of New York Press.
- Ulmer, Jeffrey T. and Brian D. Johnson. 2004. "Sentencing in context." *Criminology* 42:137–177.
- Ulmer, Jeffrey T. and John H. Kramer. 1996. "Court communities under sentencing guidelines: Dilemmas of formal rationality and sentencing disparity." *Criminology* 34:383–407.
- Unah, Issac. 2003. "Electoral incentives and prosecutorial decision to seek the death penalty." Presented at the annual meeting of the American Political Science Association, September, Philadelphia, Pa.
- Unah, Issac and John C. Boger. 2001. "Race and the Death Penalty in North Carolina: An Empirical Analysis, 1993-1997." The Common Sense Foundation, Chapel Hill, N.C.

- —. 2002. "Race, politics, and the process of capital punishment in the South." Presented at the annual meeting of the American Political Science Association, September, Boston, Mass.
- Unah, Issac and Marco Steenbergen. 2005. "Culture, race, and the implementation of the death penalty in the United States." Presented at the annual meeting of the Southern Political Science Association, January, New Orleans, La.
- Unnever, James D. 1982. "Direct and organizational discrimination in the sentencing of drug offenders." *Social Problems* 30:212–225.
- Unnever, James D. and Francis T. Cullen. 2005. "Executing the innocent and support for capital punishment: Implications for public policy." *Criminology and Public Policy* 4:3–38.
- —. 2007. "The racial divide in support for the death penalty: Does white racism matter?" *Social Forces* 85:1281–1301.
- Unnever, James D. and Larry A. Hembroff. 1988. "The prediction of racial/ethnic sentencing disparities: An expectation states approach." *Journal of Research in Crime and Delinquency* 25:53–82.
- Urbina, Martin G. 2003. Capital Punishment and Latino Offenders: Racial and Ethnic Differences in Death Sentences. New York: LFB Scholarly Publishing.
- van Buuren, Stefan. 2007. "Multiple imputation of discrete and continuous data by fully conditional specification." *Statistical Methods in Medical Research* 16:219–242.
- van Buuren, Stefan, Hendriek. C. Boshuizen, and Dick L. Knook. 1999. "Multiple imputation of missing blood pressure covariates in survival analysis." *Statistics in Medicine* 18:681–694.
- van Buuren, Stefan, Jaap P. L. Brand, C. G. M. Groothuis-Oudshoorn, and Donald B. Rubin. 2006. "Fully conditional specification in multivariate imputation." *Journal of Statistical Computation and Simulation* 76:1049–1064.
- van de Ven, Wynand P. M. M. and Bernard M. S. van Praag. 1981. "The demand for deductibles in private health insurance: A probit model with sample selection." *Journal of Econometrics* 17:229–252.
- van den Haag, Ernest. 1975. *Punishing Criminals: Concerning a Very Old and Painful Question*. New York: Basic Books.

- Vandiver, Margaret, David Giacopassi, and Mazie Curley. 2001. "Slave Codes of Tennessee." Presented at the annual meeting of the American Society of Criminology, November, Atlanta, Ga.
- Vedder, Richard K., Lowell Gallaway, and David C. Klingaman. 1990. "Black exploitation and white benefits: The Civil War income revolution." Pp. 125–137 in *The Wealth of Races: The Present Value of Benefits from Past Injustices*, edited by R. F. America. Westport, Conn.: Greenwood Press.
- Vella, Francis. 1998. "Estimating models with sample selection bias: A survey." *The Journal of Human Resources* 33:127–169.
- Venkatesh, Sudhir Alladi. 2006. Off the Books: The Underground Economy of the Urban Poor. Cambridge, Mass.: Harvard University Press.
- Vermunt, Jeroen K., Joost R. van Ginkel, L. Andries van der Ark, and Klaas Sijtsma. 2008. "Multiple imputation of incomplete categorical data using latent class analysis." Pp. 369-397 in *Sociological Methodology*, vol. 38, edited by Y. Xie. Oxford: Blackwell.
- Viescas, John. 1999. Running Microsoft Access 2000. Redmond, Wash.: Microsoft Press.
- Villarreal, Andres. 2002. "Political competition and violence in Mexico: Hierarchical social control in local patronage structures." *American Sociological Review* 67:477–498.
- Vito, Gennaro F. and Thomas J. Keil. 1988. "Capital sentencing in Kentucky: An analysis of the factors influencing decision making in the Post-*Gregg* period." *Journal of Criminal Law and Criminology* 79:483–508.
- Wacquant, Loïc. 1997. "Towards an analytic of racial domination." *Political Power and Social Theory* 11:221–234.
- —. 2000. "The new 'peculiar institution': On the prison as a surrogate ghetto." *Theoretical Criminology* 4:377–389.
- —. 2002. "Scrutinizing the street: Poverty, morality, and the pitfalls of urban ethnography." *American Journal of Sociology* 107:1468–1532.
- —. 2005a. "The great penal leap backward: Incarceration in America from Nixon to Clinton." Pp. 3–26 in *The New Punitiveness: Trends, Theories, Perspectives*, edited by J. Pratt, D. Brown, M. Brown, S. Hallsworth, and W. Morrison. London: Willan.

- Wagner, John. 2009. "O'Malley Begins Quest to Repeal the Death Penalty." *Washington Post*, January 16, p. A01.
- Warr, Mark and Mark C. Stafford. 1984. "Public goals of punishment and support for the death penalty." *Journal of Research in Crime and Delinquency* 21:95–111.
- Weber, Max. [1904] 1949. "Objectivity in social science and social policy." Pp. 50–112 in *The Methodology of the Social Sciences*, edited by E. A. Shils and H. A. Finch. New York: Free Press.
- —. [1925] 1954. Max Weber on Law in Economy and Society, 2d ed., edited by M. Rheinstein. Cambridge, Mass.: Harvard University Press.
- —. [1904] 1958. *The Protestant Ethic and the Spirit of Capitalism*. New York: Scribner and Sons.
- —. [1922] 1964. The Theory of Social and Economic Organization, edited by T. Parsons. New York: Free Press.
- —. [1922] 1968. Economy and Society: An Outline of Interpretive Sociology, edited by G. Roth and C. Wittich. Berkeley and Los Angeles, Calif.: University of California Press.
- Weesie, Jerome. 1999. "sg121: Seemingly unrelated estimation and the cluster adjusted sandwich estimator." *Stata Technical Bulletin* 52:34–47.
- Weidner, Robert R., Richard Frase, and Iain Pardoe. 2004. "Explaining sentencing severity in large urban counties: A multilevel analysis of contextual and caselevel factors." *The Prison Journal* 84:184–207.
- Weinburg, Steve. 2003. *Harmful Error: Investigating America's Local Prosecutors*. Washington, D.C.: Center for Public Integrity.
- Weisberg, Robert. 2005. "The death penalty meets social science: Deterrence and jury behavior under new scrutiny." *Annual Review of Law and Social Science* 1:151–170.
- Weisburd, David and Joseph Naus. 2001. "Report to Special Master David Baime: Re Systemic Proportionality Review." New Jersey Judiciary, Trenton, N.J.
- Weisburd, David, Elin Waring, and Stanton Wheeler. 1990. "Class, status, and the punishment of white-collar criminals." *Law and Social Inquiry* 15:223–243.
- Weiss, Robert E., Richard A. Berk, and Catherine Y. Lee. 1996. "Assessing the capriciousness of death penalty charging." *Law and Society Review* 30:607–626.

- Weiss, Robert E., Richard A. Berk, Wenzhi Li, and Margaret Farrell-Ross. 1999. "Death penalty charging in Los Angeles County: An illustrative data analysis using skeptical priors." *Sociological Methods and Research* 28:91–115.
- Weitzer, Ronald. 1996. "Racial discrimination in the criminal justice system: Findings and problems in the literature." *Journal of Criminal Justice* 24:309–322.
- Welch, Susan, Michael Combs, and John Gruhl. 1988. "Do black judges make a difference?" *American Journal of Political Science* 32:126–136.
- Wells-Barnett, Ida B. 1895. A Red Record: Tabulated Statistics and Alleged Causes of Lynchings in the United States, 1892, 1893, 1894. Chicago, Ill.: Donohue and Henneberry.
- West, Cornel. 1992. "Black leadership and the pitfalls of racial reasoning." Pp. 390–401 in *Race-ing Justice, En-gendering Power: Essays on Anita Hill, Clarence Thomas, and the Construction of Social Reality*, edited by T. Morrison. New York: Pantheon.
- Western, Bruce. 2006. *Punishment and Inequality in America*. New York: Russell Sage Foundation.
- Western, Bruce, Leonard M. Lopoo, and Sara S. McLanahan. 2004. "Incarceration and the bonds between parents in fragile families." Pp. 21–45 in *Imprisoning America: The Social Effects of Mass Incarceration*, edited by M. Patillo, D. Weiman, and B. Western. New York: Russell Sage.
- Western, Bruce and Becky Pettit. 2005. "Black-White wage inequality, employment rates, and incarceration." *American Journal of Sociology* 111:553–578.
- Wheeler, Stanton, David Weisburd, and Nancy Bode. 1983. "Sentencing the white-collar offender: Rhetoric and reality." *American Sociological Review* 47:641–659.
- White, Halbert. 1980. "A heteroskedasticity-consistent covariance matrix and a direct test for heteroskedasticity." *Econometrica* 48:817–838.
- White, Welsh S. 1991. *The Death Penalty in the Nineties: An Examination of the Modern System of Capital Punishment*. Ann Arbor, Mich.: University of Michigan Press.
- Whitehead, John T. 1998. "Good ol' boys' and the chair: Death penalty attitudes of policy makers in Tennessee." *Crime and Delinquency* 44:245–256.
- Whitman, James Q. 2003. *Harsh Justice: Criminal Punishment and the Widening Divide Between America and Europe*. Oxford: Oxford University Press.

- Whitmeyer, Joseph. 1994. "Why actor models are integral to structural analysis." *Sociological Theory* 12:153–165.
- Wiehl, Lis. 1995. "A program for death-row appeals is facing elimination." *New York Times*, Aug. 11, p. B16.
- Wilbanks, William. 1987. *The Myth of a Racist Criminal Justice System*. Monterey, Calif.: Brooks/Cole Publishing.
- Williams, Kenneth. 2000. "The deregulation of the death penalty." *Santa Clara Law Review* 40:677–728.
- Williams, Marian R. and Jefferson E. Holcomb. 2001. "Racial disparity and death sentences in Ohio." *Journal of Criminal Justice* 29:207–218.
- Williams, Patricia J. 1991. *The Alchemy of Race and Rights*. Boston, Mass.: Harvard University Press.
- Williamson, Joel. 1984. The Crucible of Race: Black-White Relations in the American South since Emancipation. New York: Oxford University Press.
- Willie, Charles V., Bernard M. Kramer, and Betram S. Brown (Eds.). 1974. *Racism and Mental Health*. Pittsburgh, Pa.: University of Pittsburgh Press.
- Wilson, James Q. and Richard J. Herrnstein. 1985. *Crime and Human Nature*. New York: Simon and Schuster.
- Wilson, James Q. and George L. Kelling. 1982. "The police and neighborhood safety." *Atlantic Monthly*, March, pp. 29–38.
- Wilson, William Julius. 1980. *The Declining Significance of Race: Blacks and Changing American Institutions*, 2d ed. Chicago, Ill.: University of Chicago Press.
- —. 1987. *The Truly Disadvantaged: The Inner City, the Underclass, and Urban Poverty.* Chicago, Ill.: University of Chicago Press.
- —. 1996. *When Work Disappears: The World of the New Urban Poor*. New York: Knopf.
- —. 1999. *The Bridge Over the Racial Divide: Rising Inequality and Coalition Politics*. Berkeley and Los Angeles, Calif.: University of California Press.
- Wilson, William Julius and Kathryn M. Neckerman. 1986. "Poverty and family structure: The widening gap between evidence and public policy issues." Pp. 232–259 in *Fighting Poverty: What Works and What Doesn't*, edited by S. H. Danziger and D. H. Weinberg. Cambridge, Mass.: Harvard University Press.

- Wilson, William Julius and Richard P. Taub. 2006. *There Goes the Neighborhood: Racial, Ethnic, and Class Tensions in Four Chicago Neighborhoods and Their Meaning for America.* New York: Knopf.
- Winant, Howard. 2000a. "Race and race theory." *Annual Review of Sociology* 26:169–185.
- —. 2000b. "The theoretical status of the concept of race." Pp. 181–190 in *Theories of Race and Racism*, edited by L. Back and J. Solomos. New York: Routledge.
- Winship, Christopher and Robert D. Mare. 1992. "Models for sample selection bias." Annual Review of Sociology 18:327–350.
- Wolf, Edwin D. 1964. "Analysis of jury sentencing in capital cases: New Jersey, 1937-1961 (Abstract of)." *Rutgers Law Review* 19:56–64.
- Wolf, Elaine and Marsha Weissman. 1996. "Revising federal sentencing policy: Some consequences of expanding eligibility for alternative sanctions." *Crime and Delinquency* 42:192–205.
- Wolfgang, Marvin E. 1957. "Victim-precipitated criminal homicide." *Journal of Criminal Law, Criminology, and Police Science* 48:1–11.
- Wolfgang, Marvin E., Arlene Kelly, and Hans C. Nolde. 1962. "Comparison of the executed and the commuted among admissions to Death Row." *Journal of Criminal Law, Criminology, and Police Science* 53:301–311.
- Wolfgang, Marvin E. and Marc Riedel. 1973. "Race, judicial discretion, and the death penalty." *Annals of the American Academy of Political and Social Science* 407:119–133.
- —. 1975. "Rape, race, and the death penalty in Georgia." American Journal of Orthopsychiatry 45:658–668.
- Wong, George Y. and William M. Mason. 1985. "The hierarchical logistic regression model for multilevel analysis." *Journal of the American Statistical Association* 80:513–524.
- Wong, Kam Chow. 1995. "Black's theory on the behavior of law revisited." *International Journal of the Sociology of Law* 23:189–232.
- —. 1998a. "Black's theory on the behavior of law revisited II: A restatement of Black's concept of law." *International Journal of the Sociology of Law* 26:75–119.

- —. 1998b. "Black's theory on the behavior of law revisited III: Law as more or less governmental social control." *International Journal of the Sociology of Law* 26:365–392.
- —. 2000. "Black's theory on the behavior of law revisited IV: The behavior of Qing Law." International Journal of the Sociology of Law 28:327–374.
- Wooldredge, John D. 1998. "Analytical rigor in studies of disparities in criminal case processing." *Journal of Quantitative Criminology* 14:155–179.
- Wooldredge, John D. and Amy Thistlethwaite. 2004. "Bilevel disparity in court dispositions for intimate assault." *Criminology* 42:417–456.
- Worden, Robert E. 1989. "Situational and attitudinal explanations of police behavior: A theoretical reappraisal and empirical assessment." *Law and Society Review* 23:667–711.
- Work, John W. 1984. *Race, Economics, and Corporate America*. Wilmington, Del.: Scholarly Resources, Inc.
- Wright, Bradley R. Entner, Avshalom Caspi, Terrie E. Moffit, and Raymond Paternoster. 2004. "Does the perceived risk of punishment deter criminally-proned individuals? Rational choice, self-control, and crime." *Journal of Research in Crime and Delinquency* 41:180–213.
- Wright, Earl, II. 2002a. "The Atlanta Sociological Laboratory 1896-1924: A historical account of the first American school of sociology." *The Western Journal of Black Studies* 26:165–174.
- —. 2002b. "Why black people tend to shout! An earnest attempt to explain the sociological negation of the Atlanta Sociological Laboratory despite its possible unpleasantness." Sociological Spectrum 22:335–361.
- Young, Robert L. 1991. "Race, conceptions of crime and justice, and support for the death penalty." *Social Psychology Quarterly* 54:67–75.
- —. 2004. "Guilty until proven innocent: Conviction orientation, racial attitudes, and support for capital punishment." *Deviant Behavior* 25:151–167.
- Yu, Jiang and Allen E. Liska. 1993. "The certainty of punishment: A reference group effect and its functional form." *Criminology* 31:447–464.
- Yucel, Recai M., Nathaniel Schenker, and Trivellore E. Raghunathan. 2006. "Multiple imputation for incomplete multilevel data with SHRIMP." Presented at the annual meeting of the New Methods for the Analysis of Family and Dyadic Processes, October, Amherst, Mass.

- Yunker, James A. 2001. "A new statistical analysis of capital punishment incorporating U.S. postmoratorium data." *Social Science Quarterly* 82:297–311.
- Zatz, Marjorie S. 1984. "Race, ethnicity, and determinate sentencing: A new dimension to an old controversy." *Criminology* 22:141–171.
- —. 1987. "The changing forms or racial/ethnic biases in sentencing." *Journal of Research in Crime and Delinquency* 24:69–92.
- —. 2000. "The convergence of race, ethnicity, gender, and class on court decisionmaking: Looking toward the 21st century." Pp. 503–552 in *Criminal Justice 2000: Policies, Processes, and Decisions of the Criminal Justice System*, vol. 3, edited by J. Horney. Washington, D.C.: National Institute of Justice.
- Zatz, Marjorie S. and Nancy Rodriguez. 2006. "Conceptualizing race and ethnicity in studies of crime and criminal justice." Pp. 39–53 in *The Many Colors of Crime: Inequalities of Race, Ethnicity and Crime in America.*, edited by R. D. Peterson, L. Krivo, and J. Hagan. New York: New York University Press.
- Zeger, Scott L., Kung-Yee Liang, and Paul S. Albert. 1988. "Models for longitudinal data: A generalized estimating equation approach." *Biometrics* 44:1049–1060.
- Zeisel, Hans. 1981. "Race bias in the administration of the death penalty: The Florida experience." *Harvard Law Review* 95:456–468.
- Zetterberg, Hans L. 1966. *On Theory and Verification in Sociology*. Totowa, N.J.: Bedminster Press.
- Zhang, Hao. 2002. "On estimation and prediction for spatial generalized linear mixed models." *Biometrics* 58:129–136.
- Ziegert, Klaus. 1980. "A sociologist's view." Pp. 60–85 in *Law and Social Control: Ideas and Ideologies*, edited by E. Kamenka and A. E.-S. Tay. London: Edward Arnold.
- Zimmerman, Paul R. 2004. "State executions, deterrence, and the incidence of murder." *Journal of Applied Economics* 7:163–193.
- Zimring, Franklin E. 2003. *The Contradictions of American Capital Punishment*. Oxford: Oxford University Press.
- —. 2005. "The unexamined death penalty: Capital punishment and reform of the model penal code." Columbia Law Review 105:1396–1416.

- Zimring, Franklin E., Joel Eigen, and Shelia O'Malley. 1976. "Punishing homicide in Philadelphia: Perspectives on the death penalty." *University of Chicago Law Review* 43:227–252.
- Zuckerman, David. 1997. "Building a capital arbitrariness claim from the ground up: A 'How To' primer based on the Pennsylvania experience." Presented at the annual meeting of the Capital Punishment Training Conference, August, Warrenton, Va.

#### Cases

- Aiken v. California, 406 U.S. 813 (1971)
- Amadeo v. Zant, 486 U.S. 214 (1988)
- Apprendi v. New Jersey, 530 U.S. 466 (2000)
- Arnold v. State, 224 S.E.2d 386 (Ga. 1976)
- Atkins v. Virginia, 536 U.S. 304 (2002)
- Barclay v. Florida, 463 U.S. 939 (1983)
- Barefoot v. Estelle, 463 U.S. 880 (1983)
- Batson v. Kentucky, 476 U.S. 79 (1986)
- Baze v. Rees, 128 S. Ct. 1520 (2008)
- Blakely v. Washington, 542 U.S. 296 (2004)
- Blystone v. Pennsylvania, 494 U.S. 299 (1990)
- Boykin v. Alabama, 395 U.S. 238 (1969)
- Branch v. State, 447 S.W.2d 932 (Tex. Crim. App. 1969), rev'd, 408 U.S. 238 (1972)
- Brown v. Board of Education of Topeka, 347 U.S. 483 (1954)
- Bullington v. Missouri, 451 U.S. 430 (1981)
- Burden v. Zant, 510 U.S. 132 (1994)
- Callins v. Collins, 510 U.S. 1141 (1994)
- Cape v. Francis, 741 F.2d 1287 (11th Cir. 1984), cert. denied, 474 U.S. 911 (1985)
- Coker v. Georgia, 433 U.S. 584 (1977)
- *Coley v. State*, 204 S.E.2d 612 (Ga. 1974)
- Collins v. State, 236 S.E.2d 759 (Ga. 1977)
- Davis v. Georgia, 429 U.S. 122 (1976)

- Dawson v. State, 554 S.E.2d 137 (Ga. 2001)
- Dobbs v. Zant, 506 U.S. 357 (1993)
- *Eberheart v. Georgia*, 433 U.S. 917 (1977)
- Enmund v. Florida, 458 U.S. 782 (1982)
- Floyd v. State, 210 S.E.2d 810 (Ga. 1974), cert. denied, 431 U.S. 949 (1977)
- *Ford v. Georgia*, 498 U.S. 411 (1991)
- Fowler v. North Carolina, 428 U.S. 904 (1976)
- Francis v. Franklin, 471 U.S. 307 (1985)
- Franklin v. Lynaugh, 487 U.S. 164 (1988)
- Freeman v. State, 440 S.E.2d 181 (Ga. 1994)
- Furman v. Georgia, 167 S.E.2d 628 (Ga. 1969), rev'd, 408 U.S. 238 (1972)
- Godfrey v. Georgia, 446 U.S. 420 (1980)
- *Green v. Georgia*, 442 U.S. 95 (1979)
- Gregg v. Georgia, 210 S.E.2d 659 (Ga. 1974), aff'd, 428 U.S. 123 (1976)
- Hamilton v. Alabama, 386 U.S. 52 (1961)
- Hampton v. Commonwealth, 58 S.E.2d 288 (Va. 1950), cert. denied, 399 U.S. 989 (1950)
- Harris v. McAdory, 334 F.3d 665 (7th Cir. 2003), cert. denied, 541 U.S. 992 (2004)
- Head v. Hill, 587 S.E.2d 613 (Ga. 2003)
- Hernandez v. New York, 500 U.S. 352 (1991)
- Herrera v. Collins, 506 U.S. 390 (1993)
- *In re Kemmler*, 136 U.S. 436 (1890)
- J.E.B. v. Alabama ex rel. T.B., 511 U.S. 127 (1994)
- Jackson v. State, 171 S.E.2d 501 (Ga. 1969), rev'd, 408 U.S. 238 (1972)

Jarrell v. State, 216 S.E.2d 258 (Ga. 1975), cert. denied, 428 U.S. 910 (1976)

- Johnson v. California, 545 U.S. 162 (2005)
- Johnson v. State, 152 S.E. 76 (Ga. 1930)
- Jurek v. Texas, 428 U.S. 262 (1976)
- Kennedy v. Louisiana, , 554 U.S. , 128 S.Ct. 2641 (2008)
- Lockett v. Ohio, 438 U.S. 586 (1978)
- Lonchar v. Thomas, 517 U.S. 314 (1996)
- Louisiana ex rel. Francis v. Resweber, 329 U.S. 459 (1947)
- Loving v. Virginia, 388 U.S. 1 (1967)
- Maxwell v. Bishop, 398 F.2d 138 (8th Cir. 1968), rev'd on other grounds, 398 U.S. 262 (1970)
- *McCleskey v. Kemp*, 580 F. Supp. 338 (N.D. Ga. 1984), *aff'd*, 753 F.2d 877 (11th Cir. 1985), *aff'd*, 481 U.S. 279 (1987)
- McGautha v. California, 402 U.S. 183 (1971)
- Miller-El v. Cockrell, 537 U.S. 322 (2003)
- Penry v. Lynaugh, 492 U.S. 302 (1989)
- People v. Anderson, 493 P.2d 880 (Cal. 1972), cert. denied, 406 U.S. 958
- People v. Lavalle, 817 N.E.2d 341 (N.Y. Ct. App. 2004)
- Plessy v. Ferguson, 163 U.S. 537 (1896)
- Presnell v. Georgia, 439 U.S. 637 (1978)
- *Proffit v. Florida*, 428 U.S. 242 (1976)
- Pulley v. Harris, 465 U.S. 37 (1984)
- Ralph v. Warden, 438 F.2d 786 (4th Cir. 1970), cert. denied, 408 U.S. 942
- Ring v. Arizona, 536 U.S. 584 (2002)

Roberts v. Louisiana, 428 U.S. 325 (1976)

Robinson v. State, 598 S.E.2d 466 (2005)

Rudolph v. Alabama, 375 U.S. 889 (1963)

Santobello v. New York, 404 U.S. 257 (1971)

Sawyer v. Whitley, 505 U.S. 333 (1992)

Schriro v. Summerlin, 542 U.S. 348 (2004)

Scott v. Sanford, 60 U.S. 393 (1857)

Spencer v. State, 398 S.E.2d 179 (Ga. 1990), cert. denied, 500 U.S. 960 (1991)

- Spinkellink v. Wainwright, 578 F.2d 582 (5th Cir. 1978), cert. denied, 440 U.S. 976 (1979)
- Stanford v. Kentucky, 492 U.S. 361 (1989)
- State ex rel. Simmons v. Roper, 112 S.W.3d 397 (Mo. 2003), cert. granted, 540 U.S. 1160 (2004), aff'd, 543 U.S. 551 (2005)

State v. Dixon, 283 So. 2d 1 (Fla. 1973)

State v. Ingram, 467 S.E.2d 523 (Ga. 1996)

State v. Marsh, 102 P.3d 445 (Kan. 2004), rev'd and remanded, 548 U.S. 163 (2006)

State v. Wilson, 685 So. 2d 1063 (La. 1996), cert. denied, Bethley v. Louisiana, 520 U.S. 1259 (1997)

*Strauder v. West Virginia*, 100 U.S. 303 (1880)

*Strickland v. Washington*, 466 U.S. 668 (1984)

Sumner v. Shuman, 483 U.S. 66 (1987)

Swain v. Alabama, 380 U.S. 202 (1965)

*Thompson v. Oklahoma*, 487 U.S. 815 (1988)

*Tison v. Arizona*, 481 U.S. 137 (1987)

*Trop v. Dulles*, 365 U.S. 86 (1958)

*Tuilaepa v. California*, 512 U.S. 967 (1994)

United States of America v. Burns, [2001] S.C.R. 283, 360

United States v. Booker, 543 U.S. 220 (2005)

United States v. Jackson, 390 U.S. 570 (1968)

Wainwright v. Goode, 464 U.S. 78 (1983)

Walton v. Arizona, 497 U.S. 639 (1990)

Whitus v. Georgia, 385 U.S. 545 (1967)

Wilkerson v. Utah, 99 U.S. 130 (1878)

Wilkins v. Missouri, 492 U.S. 361 (1989)

Williams v. Mississippi, 170 U.S. 213 (1898)

Witherspoon v. Illinois, 391 U.S. 510 (1968)

Woodson v. North Carolina, 428 U.S. 280 (1976)

Zant v. Stephens, 462 U.S. 862 (1983)

County	Death Notices	% of Total	County	Death Notices	% of Total
Appling	5	1.32	Houston	1	0.26
Baldwin	5	1.32	Irwin	1	0.26
Barrow	2	0.53	Jackson	2	0.52
Bartow	9	2.37	Jasper	1	0.26
Ben Hill	1	0.26	Jeff Davis	3	0.52
Bibb	7	1.85	Jefferson	1	0.26
Bryan	1	0.26	Jenkins	1	0.26
Bulloch	4	1.03	Jones	2	0.52
Burke	5	1.29	Laurens	1	0.26
Butts	2	0.52	Lee	1	0.26
Camden	1	0.26	Liberty	4	1.03
Candler	1	0.26	Long	5	1.29
Carroll	3	0.77	Lowndes	6	1.55
Catoosa	1	0.26	Macon	2	0.52
Charlton	1	0.26	Marion	1	0.26
Chatham	10	2.84	McIntosh	1	0.26
Cherokee	2	0.52	Meriwether	1	0.26
Clarke	10	2.58	Monroe	2	0.52
Clayton	19	4.9	Morgan	8	2.06
Cobb	11	2.84	Muscogee	14	3.61
Coffee	3	0.52	Newton	3	0.77
Colquitt	1	0.26	Oconee	7	1.8
Columbia	4	1.03	Oglethorpe	1	0.26
Cook	3	0.77	Paulding	2	0.52
Crawford	1	0.26	Pike	3	0.77
Crisp	2	0.52	Polk	3	1.03
DeKalb	25	6.7	Putnam	8	2.06
Dougherty	7	1.8	Rabun	1	0.26
Douglas	3	0.77	Richmond	19	5.15
Early	3	0.77	Rockdale	4	1.03
Effingham	2	0.52	Screven	2	0.52
Elbert	2	0.52	Spalding	8	2.06
Emanuel	1	0.26	Stephens	1	0.26
Fayette	4	1.03	Sumter	1	0.52
Floyd	5	1.29	Terrell	1	0.26
Forsyth	1	0.26	Thomas	2	0.52
Fulton	19	5.15	Tift	6	1.8
Gilmer	1	0.52	Toombs	3	0.77
Glynn	6	1.55	Troup	1	0.26
Greene	1	0.26	Upson	2	0.52
Gwinnett	13	3.35	Walker	2	0.52
Hall	11	3.09	Walton	2	0.52
Hancock	1	0.26	Ware	7	1.8
Harris	1	0.26	Washington	1	0.26
Hart	6	1.55	Wayne	1	0.26
Henry	5	1.29	Whitfield	1	0.26

Table 1: Death Notices in Georgia by County (1993–2000)

Total Death Notices: 381

Percent of all counties filing a death notice: 58%

Judicial Circuit	Death Notices	% of Total
Alapaha	3	0.77
Alcovy	5	1.29
Appalachian	1	0.52
Atlanta	19	5.15
Atlantic	11	2.84
Augusta	28	7.47
Blue Ridge	3	0.77
Brunswick	15	3.61
Chattahoochee	16	4.12
Cherokee	9	2.32
Clayton	19	4.9
Cobb	11	2.84
Conasauga	1	0.26
Cordele	3	0.77
Coweta	5	1.29
Dougherty	7	1.8
Douglas	3	0.77
Dublin	1	0.26
Eastern	10	2.84
Flint	9	2.32
Griffin	17	4.38
Gwinnett	13	3.35
Houston	1	0.26
Lookout Mountain	3	0.77
Macon	8	2.06
Middle	7	1.8
Mountain	2	0.52
Northeastern	11	3.09
Northern	9	2.32
Ocmulgee	26	6.7
Ogeechee	9	2.32
Pataula	2	0.52
Piedmont	4	1.03
Rockdale	4	1.03
Rome	5	1.29
South Georgia	2	0.52
Southern	9	2.32
Southwestern	4	1.29
Stone Mountain	25	6.7
Tallapoosa	5	1.55
Tifton	7	2.06
Waycross	12	2.84
Western	17	4.38

 Table 2: Death Notices in Georgia by Judicial Circuit (1993–2000)

Total Death Notices: 381

Percent of all judicial circuits filing a death notice: 93%

Variable Name	<u>N</u>	<u>Mean</u>	Std. Dev.	<u>Min</u>	Max
DP Notice	3318	0.114	0.318	0	1
Verdict/Plea (Verdict)	3315	0.418	0.493	0	1
Murder Conviction	3316	0.601	0.490	0	1
Death Sentence (Yes)	3318	0.017	0.128	0	1
Sentence (LS; LWOP;DS)	1991	1.144	0.423	1	3
LWOP Plea	3318	0.032	0.176	0	1
# of Defendants	3315	1.524	1.011	1	11
Defendant Black	3318	0.726	0.446	0	1
Defendant White	3318	0.238	0.426	0	1
Defendant Native Amer.	3318	0.001	0.025	0	1
Defendant Asian	3318	0.007	0.085	0	1
Defendant "Other"	3318	0.002	0.043	0	1
Defendant Hispanic	3318	0.027	0.161	0	1
Defendant Sex (Male)	3318	0.901	0.299	0	1
Defendant Age (Yrs)	3315	27.631	10.533	17	76
Defendant HS Grad	3316	0.157	0.363	0	1
Defendant Employed	3315	0.560	0.496	0	1
Defendant SES (Middle)	3315	0.427	0.495	0	1
Defendant Married	3318	0.288	0.453	0	1
Defendant has Children	3316	0.628	0.484	0	1
Defendant Drug Use	3315	0.440	0.496	0	1
Contemp. Felony	3316	0.547	0.498	0	1
Prior Felony	3316	0.176	0.381	0	1
Felony-related (Yes)	3315	0.389	0.488	0	1
Prior Violent Crime	3318	0.390	0.488	0	1
# of Stat Aggs	378	2.069	0.850	1	6
# of Victims	3315	1.218	0.762	1	14
Victim Black	3315	0.632	0.482	0	1
Victim White	3315	0.348	0.476	0	1
Victim Native Amer.	3315	0.003	0.055	0	1
Victim Asian	3315	0.018	0.132	0	1
Victim Female	3315	0.275	0.447	0	1
Victim Age	3315	32.324	15.854	0	93
Victim Stranger	3315	0.244	0.430	0	1

## Table 3: Summary Statistics for Case-Level Variables

Variable Name	<u>N</u>	<u>Mean</u>	Std. Dev.	Min	Max
Per Capita Income	46	14900.500	3200.693	11157.900	24227.500
% Below Poverty Level	46	16.050	6.052	4.850	28.279
% Fam. Below Pov. Level	46	12.623	5.113	3.350	22.964
Median Home Value	46	91961.260	29081.300	55371.430	180700.000
Violent Crime Rate (Reps)	46	443.780	353.064	108.488	2459.853
Murder Rate	46	8.127	3.491	2.700	21.640
% of GA murders	46	0.022	0.036	0.004	0.226
% Living in Diff. Home	46	45.777	6.439	33.866	59.200
% Never Been Married	46	23.952	4.576	16.241	36.738
% HS Grad	46	65.845	3.698	60.382	73.654
% College Grad	46	15.135	7.980	7.618	36.493
% Foreign Born	46	1.170	0.853	0.234	3.985
% English Only	46	94.924	2.859	85.463	98.045
% Born in Georgia	46	70.827	11.979	41.563	88.681
Vertical Status ( $\alpha$ = 0.87)	46	0.000	0.862	-1.840	1.646
Normative Status ( $\alpha$ = 0.97)	46	0.000	0.974	-1.046	1.861
Horizontal Status ( $\alpha$ = 0.83)	46	0.000	0.910	-1.383	1.343
Cultural Status ( $\alpha = 0.93$ )	46	0.000	0.968	-1.722	1.542

## Table 4: Summary Statistics for Judicial Circuit-Level Variables

Variable Name	<u>N</u>	<u>Mean</u>	<u>Std. Dev.</u>	<u>Min</u>	Max
DP Notice	2138	0.177	0.382	0	1
Verdict/Plea (Verdict)	2136	0.559	0.497	0	1
Murder Conviction*	2138	0.932	0.251	0	1
Death Sentence (Yes)	2138	0.026	0.158	0	1
Sentence (LS; LWOP;DS)	1984	1.145	0.423	1	3
LWOP Plea	2138	0.050	0.217	0	1
# of Defendants	2136	1.609	1.065	1	8
Defendant Black	2138	0.712	0.453	0	1
Defendant White	2138	0.258	0.437	0	1
Defendant Native Amer.	2138	0.000	0.022	0	1
Defendant Asian	2138	0.006	0.078	0	1
Defendant "Other"	2138	0.002	0.048	0	1
Defendant Hispanic	2138	0.021	0.144	0	1
Defendant Sex (Male)	2138	0.933	0.250	0	1
Defendant Age (Yrs)	2136	27.397	10.175	17	73
Defendant HS Grad	2136	0.163	0.369	0	1
Defendant Employed	2136	0.565	0.496	0	1
Defendant SES (Middle)	2136	0.438	0.496	0	1
Defendant Married	2138	0.283	0.451	0	1
Defendant has Children	2136	0.632	0.482	0	1
Defendant Drug Use	2136	0.465	0.499	0	1
Contemp. Felony	2138	0.668	0.471	0	1
Prior Felony	2138	0.207	0.405	0	1
Felony-related (Yes)	2136	0.481	0.500	0	1
Prior Violent Crime	2138	0.498	0.500	0	1
# of Stat Aggs	378	2.069	0.850	1	6
# of Victims	2136	1.273	0.861	1	14
Victim Black	2136	0.575	0.495	0	1
Victim White	2136	0.402	0.490	0	1
Victim Native Amer.	2136	0.002	0.044	0	1
Victim Asian	2136	0.015	0.122	0	1
Victim Female	2136	0.330	0.470	0	1
Victim Age	2136	33.430	16.340	0	93
Victim Stranger	2136	0.291	0.454	0	1

#### Table 5: Summary Statistics for Murder Conviction Subsample

\*Note: The murder conviction subsample also includes an additional 145 cases that were death eligible under B1 or B2 of Georgia's capital statute or where the prosecutor sought the death penalty but the defendant was not ultimately convicted of murder.

Variable Name	<u>N</u>	<u>Mean</u>	Std. Dev.	<u>Min</u>	Max
DP Notice	1096	0.346	0.476	0	1
Verdict/Plea (Verdict)	1094	0.476	0.500	0	1
Murder Conviction	1096	0.868	0.339	0	1
Death Sentence (Yes)	1096	0.050	0.218	0	1
Sentence (LS; LWOP;DS)	951	1.287	0.566	1	3
LWOP Plea	1096	0.090	0.287	0	1
# of Defendants	1094	1.678	1.072	1	8
Defendant Black	1096	0.662	0.473	0	1
Defendant White	1096	0.315	0.465	0	1
Defendant Native Amer.	1096	0.001	0.030	0	1
Defendant Asian	1096	0.005	0.067	0	1
Defendant "Other"	1096	0.000	0.000	0	1
Defendant Hispanic	1096	0.017	0.131	0	1
Defendant Sex (Male)	1096	0.939	0.240	0	1
Defendant Age (Yrs)	1094	27.983	9.946	17	65
Defendant HS Grad	1094	0.213	0.410	0	1
Defendant Employed	1094	0.583	0.493	0	1
Defendant SES (Middle)	1094	0.463	0.499	0	1
Defendant Married	1096	0.296	0.457	0	1
Defendant has Children	1094	0.644	0.479	0	1
Defendant Drug Use	1094	0.501	0.500	0	1
Contemp. Felony	1096	0.834	0.372	0	1
Prior Felony	1096	0.281	0.450	0	1
Felony-related (Yes)	1094	0.633	0.482	0	1
Prior Violent Crime	1096	0.695	0.461	0	1
# of Stat Aggs	378	2.069	0.850	1	6
# of Victims	1094	1.534	1.145	1	14
Victim Black	1094	0.453	0.498	0	1
Victim White	1094	0.503	0.500	0	1
Victim Native Amer.	1094	0.002	0.049	0	1
Victim Asian	1094	0.019	0.137	0	1
Victim Female	1094	0.366	0.482	0	1
Victim Age	1094	35.204	17.499	0	93
Victim Stranger	1094	0.334	0.472	0	1

## Table 6: Summary Statistics for Death Eligible Subsample

Variable Name	<u>N</u>	<u>Mean</u>	<u>Std. Dev.</u>	<u>Min</u>	Max
DP Notice	379	1.000	0.000	1	1
Verdict/Plea (Verdict)	379	0.434	0.496	0	1
Murder Conviction	379	0.926	0.262	0	1
Death Sentence (Yes)	379	0.145	0.353	0	1
Sentence (LS; LWOP;DS)	352	1.705	0.723	1	3
LWOP Plea	214	0.387	0.488	0	1
# of Defendants	379	1.892	1.141	1	8
Defendant Black	379	0.588	0.493	0	1
Defendant White	379	0.372	0.484	0	1
Defendant Native Amer.	379 379	0.000	0.000	0	1
Defendant Asian	379	0.005	0.073	0	1
Defendant "Other"	379	0.000	0.000	0	1
Defendant Hispanic	379	0.034	0.182	0	1
Defendant Sex (Male)	379	0.942	0.234	0	1
Defendant Age (Yrs)	379	26 886	9 149	17	65
Defendant / ge (113)	379	20.000	0.140	17	00
Defendant HS Grad	379	0.251	0.434	0	1
Defendant Employed	379	0.624	0.485	0	1
Defendant SES (Middle)	379	0.511	0.501	0	1
Defendant Married	379	0.285	0.452	0	1
Defendant has Children	379	0.682	0.466	0	1
	379				
Defendant Drug Use	379	0.534	0.499	0	1
Contemp. Felony	379	0.797	0.403	0	1
Prior Felony	379	0.201	0.401	0	1
Felony-related (Yes)	379	0.950	0.219	0	1
Prior Violent Crime	379	0.734	0.443	0	1
	379				
# of Stat Aggs	379	2.069	0.850	1	6
# of Victims	379	1.632	1.347	1	14
Victim Black	379	0.300	0.459	0	1
Victim White	379	0.667	0.472	0	1
Victim Native Amer.	379	0.005	0.073	0	1
	379				
Victim Asian	379	0.030	0.170	0	1
Victim Female	379	0.492	0.501	0	1
Victim Age	379	36.008	19.885	0	90
Victim Stranger	379	0.402	0.491	0	1

## Table 7: Summary Statistics for Death Noticed Subsample

	(1) <u>DPN-MC</u>	(2) <u>DPN-DE</u>	(3) <u>BP</u>	(4) <u>DS</u>
# of Defs	0.352***	0.320***	-0 357*	-0 136
	(0.060)	(0.074)	(0.165)	(0.234)
Def Age	0.017	0.004	0.008	0.079
Bonnigo	(0.010)	(0.011)	(0.025)	(0.042)
Def Age (Squared)	-0.002**	-0.001	0.001	-0.006*
	(0.001)	(0.001)	(0.001)	(0.003)
# of Vics.	0.479***	0.135*	-0.006	0.448*
	(0.072)	(0.068)	(0.114)	(0.190)
Vic. Age	-0.005	-0.011*	0.012	-0.014
	(0.004)	(0.005)	(0.010)	(0.015)
Vic. Age (Squared)	0.001***	0.001***	0003	-0.000
	(0.000)	(0.000)	(.0003)	(0.000)
Def. X Vic. Age	-0.001**	-0.001	0.001	-0.001
	(0.000)	(0.000)	(0.001)	(0.002)
Def. B/Vic. W.	1.282***	0.895***	0.612	-0.003
	(0.187)	(0.212)	(0.437)	(0.606)
Def. W/Vic. W	0.834***	0.676***	-0.178	1.087*
	(0.169)	(0.189)	(0.389)	(0.543)
Def. H./Vic. W.	0.929*	2.419**	-0.387	1.780
	(0.410)	(0.737)	(1.187)	(1.122)
Def. M./Vic. F.	1.006***	0.960***	0.526	0.151
	(0.141)	(0.158)	(0.331)	(0.442)
Def. F./Vic. M.	-0.352	-0.075	-0.137	-2.560
	(0.357)	(0.390)	(0.925)	(1.541)
Constant	-2.457***	-1.470***	-1.392**	-1.926*
	(0.233)	(0.262)	(0.527)	(0.813)
Vera Dummine	V	V	V	V
rear Dummies	Y 0.690***	Y 0.641***	Y 0 272	Y 1 004***
S.D. OF FANDOM ETTECT	0.002^^^	0.641***	0.373	1.234***
κιιο (ρ)	0.020	0.556	0.119	0.405
IN # of Judicial Circuits	2130	1094	214	104
	40	40	40	39
AIC	1643.107	1229.208	287.487	214.591
BIC	1/62.10/	1334.158	354.619	279.688

#### Table 8: Random Intercept Logistic Regression Models (H1-H3)

	(1) <u>DPN-MC</u>	(2) <u>DPN-DE</u>	(3) <u>BP</u>	(4) <b>DS</b>
# of Defs.	0.328***	0.312***	-0.435*	-0.175
	(0.062)	(0.076)	(0.178)	(0.262)
Def. Age	0.009	-0.007	0.035	0.083
	(0.012)	(0.013)	(0.029)	(0.047)
Def. Age (Squared)	-0.001*	-0.000	-0.000	-0.006
	(0.001)	(0.001)	(0.001)	(0.003)
# of Vics.	0.485***	0.142*	-0.005	0.475*
	(0.073)	(0.070)	(0.118)	(0.199)
Vic. Age	-0.008	-0.012*	0.004	-0.017
	(0.005)	(0.005)	(0.011)	(0.017)
Vic. Age (Squared)	0.001***	0.001***	-0.000	-0.000
	(0.000)	(0.000)	(0.000)	(0.001)
Def. X Vic. Age	-0.001**	-0.001	0.001	-0.001
	(0.000)	(0.000)	(0.001)	(0.002)
Def. B/Vic. W.	1.084***	0.725**	0.444	-0.257
	(0.198)	(0.224)	(0.482)	(0.672)
Def. W/Vic. W	0.865***	0.697***	0.164	1.727**
	(0.178)	(0.199)	(0.431)	(0.635)
Def. H./Vic. W.	1.061*	2.647***	0.200	2.328*
	(0.437)	(0.769)	(1.244)	(1.179)
Def. M./Vic. F.	1.107***	1.076***	0.687*	0.143
	(0.148)	(0.167)	(0.347)	(0.497)
Def. F./Vic. M.	-0.253	0.043	0.093	-2.711
	(0.377)	(0.408)	(0.948)	(1.824)
Def. HS Grad	0.740***	0.348	-0.573	-0.676
	(0.173)	(0.197)	(0.433)	(0.581)
Def. Employed	0.412**	0.385*	-0.229	-0.181
	(0.149)	(0.168)	(0.362)	(0.538)
Def. SES	0.051	0.160	-0.537	0.819
	(0.143)	(0.162)	(0.364)	(0.509)
Def. Married	-0.122	-0.097	0.309	-1.057
	(0.168)	(0.189)	(0.451)	(0.633)
Def. has Children	0.310*	0.413*	-0.157	2.312**
	(0.158)	(0.182)	(0.381)	(0.830)
Def. Drug Use	0.477***	0.412**	0.482	-0.057
-	(0.139)	(0.158)	(0.339)	(0.527)
Vic. Stranger	0.686***	0.606**	0.840	0.759
-	(0.172)	(0.188)	(0.458)	(0.599)
Constant	-3.465***	-2.498***	-1.583*	-4.044**
	(0.302)	(0.338)	(0.688)	(1.316)

# Table 9: Random Intercept Logistic Regression Models (H4)

# Table 9 (cont.)

Year Dummies	Y	Y	Y	Y
S.D. of random effect	0.699***	0.674***	0.498	1.09
Rho (ρ)	0.629	0.569	0.178	0.361
Ν	2138	1096	214	164
# of Judicial Circuits	46	46	40	39
AIC	1594.866	1208.059	289.623	298.769
BIC	1753.507	1347.940	380.251	408.798

	(1) <u>DPN-MC</u>	(2) <u>DPN-DE</u>	(3) <u>BP</u>	(4) <u>DS</u>
# of Defs.	0.325***	0.314***	-0.442*	-0.230
	(0.062)	(0.076)	(0.178)	(0.250)
Def. Age	0.009	-0.006	0.031	0.071
	(0.012)	(0.013)	(0.030)	(0.042)
Def. Age (Squared)	-0.001*	-0.000	0.000	-0.004
	(0.001)	(0.001)	(0.001)	(0.003)
# of Vics.	0.483***	0.143*	-0.002	0.652**
	(0.072)	(0.069)	(0.118)	(0.200)
Vic. Age	-0.008	-0.012*	0.005	-0.021
	(0.005)	(0.005)	(0.011)	(0.015)
Vic. Age (Squared)	0.001***	0.001***	-0.000	-0.000
	(0.000)	(0.000)	(0.000)	(0.000)
Def. X Vic. Age	-0.001**	-0.001	0.001	-0.001
	(0.000)	(0.000)	(0.001)	(0.002)
Def. B/Vic. W.	1.092***	0.724**	0.478	-0.163
	(0.198)	(0.223)	(0.514)	(0.641)
Def. W/Vic. W	0.844***	0.651**	0.244	1.724**
	(0.181)	(0.202)	(0.478)	(0.629)
Def. H./Vic. W.	1.038*	2.641***	0.371	2.202*
	(0.436)	(0.778)	(1.252)	(1.085)
Def. M./Vic. F.	1.106***	1.074***	0.720*	0.148
	(0.148)	(0.167)	(0.358)	(0.491)
Def. F./Vic. M.	-0.277	0.017	0.036	-3.013
	(0.377)	(0.407)	(0.989)	(1.756)
Def. HS Grad	0.726***	0.337	-0.667	-0.830
	(0.173)	(0.196)	(0.447)	(0.568)
Def. Employed	0.404**	0.373*	-0.156	0.058
	(0.149)	(0.168)	(0.366)	(0.527)
Def. SES	0.034	0.138	-0.574	0.895
	(0.143)	(0.163)	(0.372)	(0.498)
Def. Married	-0.133	-0.108	0.362	-0.966
	(0.167)	(0.188)	(0.457)	(0.591)
Def. has Children	0.321*	0.422*	-0.137	2.414**
	(0.158)	(0.182)	(0.386)	(0.846)
Def. Drug Use	0.473***	0.412**	0.497	0.169
	(0.139)	(0.158)	(0.342)	(0.497)
Vic. Stranger	0.673***	0.593**	0.867	0.908
	(0.172)	(0.187)	(0.469)	(0.560)
Vertical	-0.062	-0.045	0.350	0.133
<b>N</b> <i>H</i>	(0.207)	(0.207)	(0.364)	(0.530)
Normative	-0.538	-0.498	0.608	-1.375
	(0.295)	(0.283)	(0.391)	(0.799)
Horizontal	0.190	0.103	-0.329	-0.140
	(0.278)	(0.276)	(0.427)	(0.629)

## Table 10: Random Intercept Logistic Regression Models (H5)

# Table 10 (cont.)

Horizontal (Squared)	-0.600**	-0.540*	-0.278	-0.779
	(0.229)	(0.228)	(0.336)	(0.535)
Constant	-3.641***	-2.670***	-1.446*	-4.589**
	(0.317)	(0.346)	(0.710)	(1.500)
Year Dummies	Y	Y	Y	Y
S.D. of random effect	0.616***	0.624***	0.655	0.998
Rho (r)	0.574	0.538	0.225	0.314
Ν	2138	1096	214	164
# of Judicial Circuits	46	46	40	39
AIC	1595.320	1209.424	293.207	206.818
BIC	1770.958	1364.293	393.904	305.621

	(1)	(2)	(3)	(4)
	<u>DPN-MC</u>	<u>DPN-DE</u>	<u>BP</u>	<u>DS</u>
# of Defs.	0.360***	0.322***	-0.246 (0.329)	-0.239 (0.252)
Def. Age	0.015	0.010	0.018	0.071
Def. Age (Squared)	-0.001 (0.001)	-0.001 (0.001)	0.000 (0.002)	-0.004 (0.003)
# of Vics.	0.479***	0.144*	0.131	0.621**
	(0.076)	(0.071)	(0.188)	(0.206)
Vic. Age	-0.008	-0.011*	-0.003	-0.023
	(0.005)	(0.005)	(0.016)	(0.016)
Vic. Age (Squared)	0.001***	0.001***	0.000	-0.000
	(0.000)	(0.000)	(0.001)	(0.001)
Def. X Vic. Age	-0.001*	-0.001	0.000	-0.001
	(0.000)	(0.000)	(0.001)	(0.002)
Def. B/Vic. W.	1.068***	0.698**	1.374	0.115
	(0.206)	(0.228)	(0.728)	(0.976)
Def. W/Vic. W	0.921***	0.576**	0.916	2.031*
	(0.190)	(0.208)	(0.631)	(0.847)
Def. H./Vic. W.	1.168*	2.434**	3.257	3.003
	(0.456)	(0.770)	(1.715)	(2.153)
Def. M./Vic. F.	1.087***	1.128***	1.291	0.219
	(0.152)	(0.170)	(1.062)	(0.521)
Def. F./Vic. M.	-0.283	-0.293	-0.242	-2.817
	(0.395)	(0.413)	(0.985)	(1.795)
Def. HS Grad	0.745***	0.402*	-0.326	-0.754
	(0.179)	(0.203)	(0.611)	(0.612)
Def. Employed	0.446**	0.391*	0.132	-0.017
	(0.155)	(0.171)	(0.516)	(0.545)
Def. SES	0.050	0.158	-0.613	0.750
	(0.148)	(0.166)	(0.413)	(0.521)
Def. Married	-0.225 (0.174)	-0.108 (0.192)	0.334 (0.490)	-1.161 (0.613)
Def. has Children	0.276 (0.164)	0.336 (0.186)	0.138 (0.531)	(0.892)
Det. Drug Use	(0.144)	0.390 <sup>-</sup> (0.161)	(0.469)	(0.527)
Vic. Stranger	0.434 <sup>*</sup> (0.178)	(0.193)	(0.757)	(0.605)
Contemp. Fel.	-0.046 (0.209)	-0.955*** (0.246)	0.990	1.653 (0.924)
Prior Fel.	-0.222 (0.188)	-0.657** (0.200)	-0.944 (0.723)	-0.399 (0.617)
VIO. Prior	1.437***	0.943***	0.646	-0.677
	(0.194)	(0.210)	(0.942)	(0.772)

## Table 11: Random Intercept Logistic Regression Models (H<sub>6</sub>-H<sub>7</sub>)

#### Table 11 (cont.)

# Stat. Aggs.			0.594*	-0.098
# eta. 1 .990			(0.251)	(0.386)
Vertical	-0.102	0.019	0.230	0.072
	(0.215)	(0.213)	(0.502)	(0.546)
Normative	-0.594	-0.472	0.467	-1.498
	(0.307)	(0.291)	(0.463)	(0.841)
Horizontal	0.261	0.019	-0.337	0.055
	(0.290)	(0.284)	(0.440)	(0.676)
Horizontal (Squared)	-0.678**	-0.569*	-0.682	-0.681
	(0.239)	(0.234)	(0.572)	(0.564)
Constant	-4.360***	-2.309***	-3.531**	-5.299**
	(0.362)	(0.404)	(1.217)	(1.703)
Year Dummies	Y	Y	Y	Y
Sel. Bias Corrected	N/A	N/A	Y	Y
S.D. of random effect	0.645***	0.545***	0.824*	1.223
Rho $(\rho)$	0.578	0.473	0.324	0.383
N	2138	1096	214	164
# of Judicial Circuits	46	46	40	39
AIC	1509.646	1183.684	297.040	211.101
BIC	1702.282	1353.540	417.877	322.254

	(1) <b>DPN</b>	(2) <u>DPN</u>	(3) <u>DPN</u>	(4) <u>DPN</u>	(5) <u>DPN</u>
Def. B/Vic. W.	0.704**				
B/W X Vert.	(0.238) 0.025				
B/W X Norm.	(0.376) 0.416				
B/W X Horiz.	(0.357) -0.445				
Def. W/Vic. W	(0.444)	0.686**			
W/W X Vert.		(0.249) -0.036			
W/W X Norm.		(0.345) 0.240			
W/W X Horiz.		(0.389) 0.009			
Def. M./Vic. F.		(0.410)	1.082***		
M/F X Vert.			(0.183) -0.209		
M/F X Norm.			(0.299) -0.284		
M/F X Horiz.			(0.306) 0.285		
Prior Fel.			(0.374)	-0.684**	
Prior Fel. X Vert.				(0.212) -0.123	
Prior Fel. X Norm.				(0.348) -0.208	
Prior Fel. X Horiz.				(0.334) 0.241	
Vio. Prior				(0.424)	0.939***
Prior Vio. X Vert.					(0.237) 0.501
Prior Vio. X Norm.					(0.329) 0.399
Prior Vio. X Horiz.					(0.368) -0.893*
Constant	-2.343*** (0.408)	-2.324*** (0.404)	-2.303*** (0.406)	-2.288*** (0.406)	(0.405) -2.340*** (0.423)

#### Table 12: Cross-Level Interactions (H<sub>8</sub>)

# Table 12 (cont.)

Year Dummies	Y	Y	Y	Y	Y
S.D. of Random Intercept	0.628***	0.761***	0.817***	0.761***	0.551***
S.D. of Random Slope	0.818*	0.543	0.750*	0.932*	0.787*
Rho (Random Intercept)	0.509	0.539	0.545	0.549	0.343
Rho (Random Slope)	0.316	0.179	0.301	0.332	0.355
Rho $(\rho_{21})$	-0.222	-0.687	-0.666	-0.382	-0.441
Ν	1092	1092	1092	1092	1092
# of Judicial Circuits	46	46	46	46	46
AIC	1187.285	1189.073	1188.699	1189.251	1184.332
BIC	1372.128	1373.916	1373.542	1374.094	1369.176

\*p<.05; \*\*p<.01; \*\*\*p<001 Standard errors in parentheses

The above models include controls for all case-level covariates.

	(1) <u>BP</u>	(2) <u>DS</u>
# of Defs.	-0.010	-0.078
	(0.125)	(0.091)
Def. Age	0.027	0.021
	(0.016)	(0.029)
Def. Age (Squared)	0.000	-0.002
	(0.002)	(0.002)
# of Vics.	0.011	0.225***
	(0.082)	(0.066)
Vic. Age	0.001	-0.019**
	(0.006)	(0.008)
vic. Age (Squared)	0.000	0.000
Def X Via Age	(0.000)	(0.000)
Dei. A vic. Age	(0.000)	0.000
Def B//ic W	0.000)	(0.000)
	(0.250)	(0.336)
Def W/Vic W	0.360	0.000)
	(0.240)	(0.386)
Def. H./Vic. W.	0.280	1.803**
	(0.850)	(0.699)
Def. M./Vic. F.	0.942***	0.497***
	(0.198)	(0.286)
Def. F./Vic. M.	-0.398	-0.680
	(0.512)	(0.606)
Def. HS Grad	0.155	-0.309
	(0.206)	(0.352)
Def. Employed	-0.058	0.195
	(0.213)	(0.243)
Def. SES	-0.222	0.484*
	(0.219)	(0.237)
Def. Married	-0.083	-0.622*
Def has Children	(0.166)	(U.280) 1 701***
Der. has children	-0.259	(0.475)
Def Drug Lise	0.195)	(0.473)
Del. Diug Ose	(0.214)	(0.132
Vic Stranger	0.531**	0.629*
vier example	(0.178)	(0.322)
Contemp. Fel.	-0.086	0.614
•	(0.270)	(0.501)
Prior Fel.	-0.280	-0.301
	(0.211)	(0.293)
Vio. Prior	0.627*	0.171
	(0.302)	(0.349)

## Table 13 (cont.)

Vertical	-0.051	0.018	
	(0.131)	(0.227)	
Normative	-0.091	-0.758***	
	(0.100)	(0.224)	
Horizontal	-0.008	0.150	
	(0.159)	(0.302)	
Horizontal (Squared)	-0.174	-0.470	
	(0.132)	(0.256)	
Constant	-2.233***	-5.7328***	
	(0.296)	(0.648)	
Year Dummies	Y	Ŷ	
Ν	1037	1096	
$r_{21}$	0.664***	0.629***	
<i>r</i> <sub>31</sub>	0.734***	0.732***	
<i>r</i> <sub>23</sub>	0.193**	-0.007	

	DPN-MC	DPN-DE	BP	<u>DS</u>
# of Defs.	N/A	N/A	N/A	N/A
Def. Age	S	S	U	S
# of Vics.	N/A	N/A	N/A	N/A
Vic. Age	S	S	U	S
Def/Vic Age Difference	S	S	U	S
Def. B/Vic. W.	S	S	S	М
Def. W/Vic. W	S	S	S	М
Def. H./Vic. W.	S	S	S	S
Def. M./Vic. F.	S	S	S	S
Def. F./Vic. M.	S	S	S	S
Def. HS Grad	U	U	U	S
Def. Employed	U	U	S	U
Def. SES	U	U	S	U
Def. Married	S	S	S	S
Def. has Children	U	U	S	U
Def. Drug Use	S	S	S	S
Vic. Stranger	S	S	S	S
Contemp. Fel.	U	U	U	S
Prior Fel.	U	U	U	U
Vio. Prior	S	S	S	S
# Stat. Aggs.	N/A	N/A	S	S
Vertical	U	S	U	S
Normative	U	U	U	U
Horizontal	S	S	S	S
Cross-Level Hypotheses				
B/W X Vert.		S		
B/W X Norm.		S		
B/W X Horiz.		S		
W/W X Vert.		U		
W/W X Norm.		S		
W/W X Horiz.		U		
M/F X Vert.		U		
M/F X Norm.		U		
M/F X Horiz.		U		
Prior Fel. X Vert.		U		
Prior Fel. X Norm.		U		
Prior Fel. X Horiz.		U		
Prior Vio. X Vert.		S		
Prior Vio. X Norm.		S		
Prior Vio. X Horiz.		S		

## Table 14: Summary of Results for Hypotheses

KEY: S = Supportive; U = Unsupportive; M = Mixed Support





Figure 2: Capital Cases Resulting in Straight Life Sentences




Figure 3: Nonlinear Effect of Victim's Age

Figure 4: Nonlinear Effect of Defendant's Age





Figure 5: Nonlinear Effect of Defendant/Victim Age Difference

Figure 6: Nonlinear Effect of Circuit-Level Horizontal Status



Arrest, First Appearance, and Commitment Hearing	Accused presented before a magistrate judge within 48 (warrant) or 72 (w/o warrant) hours (Unif. Super. Ct. R. 26.1 [2007]).
Grand Jury Indictment	Grand jury returns an indictment charging capital offense. Once indicted, accused possibly eligible for the appointment of counsel.
Appointment of Counsel	Accused eligible for appointed counsel if indigent. Pursuant to the Georgia Indigent Defense Act of 2003 (GIDA). If the accused is eligible, he must be appointed two attorneys before he is called upon to plea to the charges, which generally occurs at the arraignment.
Pretrial Conference: Notice of Intent to Seek the Death Penalty and Qualifications of Defense Counsel	Pretrial conference must be held as soon as possible after indictment and before arraignment, and the conference must be recorded and transcribed. Prosecuting attorney must announce intention to seek the death penalty and then file a notice of intent with the clerk of the superior court. The superior court must then transmit the notice to the clerk of the Supreme Court of Georgia (Unif. App. R. IIC(1) [2007])
Arraignment, Pleas, Special Plea of Mental Incompetency to Stand Trial and Notice of the Defense's Intention to Raise the Issue of Insanity or Mental Illness	During the arraignment, the court must read the indictment and ask the defendant to plead to the capital felony and any lesser-included offenses charged. The defendant is allowed to plead guilty, not guilty, or mentally incompetent to stand trial; <i>nolo contendere</i> pleas are disallowed (Ga. Code Ann. § 17-7-95(a)).
Selection of a Capital Jury	Capital defendant(s) have a right to a trial by jury. The court must empanel forty-two prospective jurors from which the state and defense must select a total of twelve jurors and one or more alternative jurors, if deemed necessary by the judge (Ga. Code Ann. §§ 15-12-160, 168).
Capital Trial	Capital cases are heard before the superior court (Ga. Const. Art. 6, § 4, ¶ I) and conducted in two phases: the guilt/innocence phase and, if the defendant is found guilty of a capital felony, the penalty phase. Immediately prior to the conviction phase, the court must conduct a conference with the state, defense counsel, and the defendant to resolve several matters, including, <i>inter alia</i> , any last minute motions, stipulations, and objections to defense counsel. If the defendant is convicted of capital murder at the conclusion of the guilt/innocence phase, the case proceeds to the penalty phase where both the prosecutor and defense counsel may present witnesses and evidence regarding the statutory aggravating circumstances, as well as non-statutory aggravating and mitigating circumstances. The jury <i>may</i> sentence the defendant to death if, and only if, they find one or more statutory aggravating circumstance beyond a reasonable doubt (Ga. Code Ann. § 17-10-31.1(c)). Following a conviction for a capital felony and a sentence of death, the defendant may challenge her conviction or death sentence by: (1) filing a motion for a new trial with the superior court, or (2) filing a direct appeal with the Georgia Supreme Court (Ga. Code Ann. § 17- 10-35). If the defendant does not initiate any sort of review, the case will automatically be appealed to the Georgia Supreme Court within ten days of the filing of the trial transcript by the court reporter of the superior court. This automatic review will occur even if the defendant does not wish to appeal her conviction or sentence.

## Appendix A: Progression of Georgia Death Penalty Case (Abridged)

## **Appendix B: Variable Definitions**

VARIABLE NAME	VARIABLE DESCRIPTION	DATA SOURCE(S)
	Case-Level Variables	
IncidentYear	Year of Incident	GDC; CO; GCD
DP Notice	Death Penalty Notice Filed (Yes=1)	CO; GCD
Sentence	Disposition of Case (Life Sentence; Life Sentence w/o Parole; Death Sentence)	GDC; CO; GCD
Plea/Verdict	Case Disposed by Trial or Plea Bargain (Trial=1)	GDC; CO; GCD
# of Defendants	Total Number of Co-defendants	CO; GCD
Defendant's Race	Defendant's Race (Asian, Black, Hispanic, Other, White)	GDC; CO; GCD
Defendant's Sex	Defendant's sex/gender (Male=1)	GDC; CO; GCD
Defendant's Age	Defendant's Age at Time of Incident (in Years)	GDC; CO; GCD
Defendant HS Grad	Defendant Graduated from High School (Yes=1)	GDC; CO; GCD
Defendant Employed	Defendant was Employed at time of incident (Yes=1)	GDC; GCD
Defendant Married	Defendant was Married at the time of the incident (Yes=1)	GDC; GCD
Defendant has Children	Defendant has Children (Yes=1)	GDC; GCD
Defendant's Drug Use	Defendant has history of drug use (Yes=1)	GDC; GCD
Contemp. Felony	Defendant was convicted of committing a contemporaneous felony (Yes=1)	CO; GCD
Prior Felony	Defendant had prior felony conviction (Yes=1)	CO; GCD
Felony Circumstances	Murder was committed during the course of a another felony (Yes=1)	CO; GCD; SHR
Murder Conviction	Defendant Convicted of Murder (Yes=1)	GDC; CO
Prior Violent Crime	Defendant had prior conviction for a violent crime	GDC; GCD
# of Stat Aggs	Number of Statutory Aggravating Circumstances Alleged	GDC; GCD

Appendix B (cont.)	-	
VARIABLE NAME	VARIABLE DESCRIPTION	DATA SOURCE(S)
# of Victims	Number of deceased victims in the case	CO; GCD; SHR
Victim Race	Victim's race (Asian, Black, Hispanic, Other, White)	CO; GCD; SHR
Victim Sex	Victim's sex/gender (0=No; 1=Yes)	CO; GCD; SHR
Victim Age	Victim's age at time of incident (in Years)	CO; GCD; SHR
Victim Stranger	Victim's and defendant were strangers (Yes=1)	CO; GCD; SHR
	Circuit-Level Variables	
County	County in which the trial took place	GDC; CO; GCD
Circuit	Circuit in which the trial took place	GDC; CO; GCD
Трор	Total Population	CENSUS
PopSM	Population per square mile	CENSUS
Pwhite	Percent White	CENSUS
Pblack	Percent Black	CENSUS
Pother	Percent "Other"	CENSUS
Medage	Median Age	CENSUS
PU18	Percent of Population Under 18 years of age	CENSUS
PCapInc	Per Capital Income	CENSUS
PIBPL	Percent of Population with Income below Poverty Level	CENSUS
PFBPL	Percent Families with Income Below Poverty Level	CENSUS
ValHome	Median Value of Homes	CENSUS
MPFem	Males per 100 Females	CENSUS
PHSGrad	Percent HS graduate or higher (25 and older)	CENSUS
PCGrad	Percent College graduate or higher (25 and older)	CENSUS
PNBM	Percent Never been married (15 and older)	CENSUS
PLDH	Percent living in different home from 1995	CENSUS
PUrban	Percent living in urban area	CENSUS

Appendix B (cont.)				
VARIABLE NAME	RIABLE NAME VARIABLE DESCRIPTION			
PSEO	Percent who only speak English (18 and over)	CENSUS		
PBSOR	Percent Born in Georgia	CENSUS		
PFB	Percent Foreign Born	CENSUS		
PLabor	Percent in labor force (16 and older)	CENSUS		
VRUCR	County Violent Crime Rate (Reports)	GBI		
VAUCR	County Violent Crime Rate (Arrests)	GBI		
MRUCR	County Homicide Rate	GBI		
LEGEND: U.S. Bureau of the Census (CENSUS); Ga. Department of Corrections (GDC); Ga. Sup.				
Ct. Clerk's Office (CO); Ga. Bureau of Investigation (GBI); Office of the Georgia Capital Defender				
GCD); Supplementary Homicide Reports (SHR).				

Null Model (No Predictors)	(1) <u>DPN-MC</u>	(2) <u>DPN-DE</u>	(3) <u>BP</u>	(4) <u>DS</u>
Constant	-1.331*** (0.185)	-0.332 (0.208)	-1.128** (0.396)	898*** (0.456)
Year Dummies	Y	Y	Y	Y
S.D. of random effect	0.717***	0.722***	0.474	0.553
Rho (ρ)	0.679	0.629	0.185	0.192
N	2138	1096	214	164
# of Judicial Circuits	46	46	40	39
AIC	1882.424	1330.979	284.737	209.772
BIC	1893.759	1340.978	291.451	215.972
	(5) <b>DPN-MC</b>	(6) <u>DPN-DE</u>	(7) <b>BP</b>	(8) <b>DS</b>
Contextual Variables Only				
Vertical	0.061	-0.042	0.238	0.275
	(0.205)	(0.216)	(0.364)	(0.337)
Normative	-0.675*	-0.590	0.649	-0.501
	(0.3090)	(0.319)	(0.436)	(0.529)
Horizontal	0.076	0.034	-0.494	-0.199
	(0.284)	(0.298)	(0.442)	(0.501)
Horizontal (Squared)	-0.355	-0.401	-0.293	-0.442
	(0.228)	(0.241)	(0.367)	(0.465)
Constant	-1.286***	-0.303	-1.314**	-0.948
	(0.178)	(0.203)	(0.414)	(0.473)
Year Dummies	Y	Y	Y	Y
S.D. of random effect	0.617***	0.578***	0.549	0.495
Rho (p)	0.622	0.571	0.224	0.157
N	2138	1094	214	164
# of Judicial Circuits	46	46	40	39
AIC	1880.275	1330.987	288.695	212.152
BIC	1908.613	1355.984	305.478	230.752

## Appendix C: Random Intercept Logistic Regression Models (Null & Partial)

\*p<.05; \*\*p<.01; \*\*\*p<001 Standard errors in parentheses

	(1)	(2)	(3)
	DS-RI	PB-DE	PB-DN
# of Defs.	-0.217	0.153*	0.207
	(0.263)	(0.068)	(0.123)
Def. Age	0.077	0.016	0.024
	(0.051)	(0.012)	(0.022)
Def. Age (Squared)	-0.006	-0.000	0.001
	(0.003)	(0.001)	(0.001)
# of Vics.	0.429*	-0.091	-0.067
	(0.199)	(0.063)	(0.090)
Vic. Age	-0.018	0.012*	0.005
	(0.017)	(0.005)	(0.008)
Vic. Age (Squared)	-0.000	-0.000	0.000
	(0.001)	(0.000)	(0.000)
Def. X Vic. Age	-0.001	-0.001	-0.000
	(0.002)	(0.000)	(0.001)
Def. B/Vic. W.	-0.094	0.055	0.184
	(0.681)	(0.200)	(0.349)
Def. W/Vic. W	1.958**	0.044	0.301
	(0.690)	(0.180)	(0.327)
Def. H./Vic. W.	2.242	-0.150	-0.573
	(1.277)	(0.569)	(0.685)
Def. M./Vic. F.	0.145	0.328*	0.213
	(0.515)	(0.145)	(0.251)
Def. F./Vic. M.	-2.354	-0.485	0.418
	(1.825)	(0.364)	(0.726)
Def. HS Grad	-0.636	0.190	0.101
	(0.630)	(0.175)	(0.297)
Def. Employed	-0.222	-0.087	-0.323
	(0.549)	(0.144)	(0.263)
Def. SES	0.628	0.238	-0.034
	(0.522)	(0.143)	(0.258)
Def. Married	-1.235	-0.301	-0.269
	(0.658)	(0.164)	(0.292)
Def. has Children	2.668**	-0.700***	-1.283***
	(0.876)	(0.158)	(0.300)
Def. Drug Use	-0.112	0.187	0.059
	(0.539)	(0.136)	(0.246)
Vic. Stranger	0.858	-0.001	-0.123
	(0.697)	(0.163)	(0.312)
Contemp. Fel.	2.026*	-0.891***	-0.441
	(0.945)	(0.208)	(0.418)
Prior Fel.	-0.291	-0.090	-0.318
	(0.646)	(0.167)	(0.329)
Vio. Prior	-1.190	0.459**	0.439
	(0.832)	(0.166)	(0.386)

Appendix D: Supplemental Analyses (Partial & Plea Bargaining Models)

# Stat. Aggs.	-0.102		0.045
	(0.344)		(0.160)
IMR	-0.757**		
	(0.271)		
Vertical	(0.271)	-0 205	0 123
Vertical		(0.160)	(0.272)
Neverative		(0.100)	(0.272)
Normative		-0.164	0.193
		(0.185)	(0.334)
Horizontal		0.076	0.026
		(0.206)	(0.335)
Horizontal (Squared)		0.066	-0.355
		(0.165)	(0.295)
Constant	-4 089**	0.942*	1.301**
Constant	(1 519)	(0.3/17)	(0.631)
	(1.010)	(0.047)	(0.001)
Year Dummies	Y	Y	Y
S.D. of random effect	1.185	0.407**	0.608*
Rho (p)	0.384	0.391	0.351
N	164	1096	379
# of Judicial Circuits	39	46	46
AIC	212.136	1469.536	529.486
BIC	310.939	1644.387	670.951

\*p<.05; \*\*p<.01; \*\*\*p<001 Standard errors in parentheses

	(1) DPN-MC	(2) DPN-DE	(3) BP	(4) DS
		DINDE	<u> </u>	<u>50</u>
# of Defs.	0.347***	0.296**	-0.447	-0.183
	(0.086)	(0.103)	(0.248)	(0.172)
Def. Age	0.009	0.004	0.019	0.066
	(0.014)	(0.014)	(0.033)	(0.049)
Def. Age (Squared)	-0.001	-0.001	0.000	-0.005
	(0.001)	(0.001)	(0.001)	(0.003)
# of Vics.	0.457***	0.133*	0.024	0.413**
	(0.085)	(0.062)	(0.106)	(0.155)
Vic. Age	-0.008	-0.010	0.004	-0.013
	(0.006)	(0.006)	(0.011)	(0.013)
Vic. Age (Squared)	0.001***	0.001***	-0.000	-0.000
	(0.000)	(0.000)	(0.000)	(0.001)
Def. X Vic. Age	-0.001	-0.001	0.001	-0.001
	(0.001)	(0.001)	(0.001)	(0.003)
Def. B/Vic. W.	1.186***	0.871***	0.889	-0.068
	(0.232)	(0.256)	(0.572)	(0.705)
Def. W/Vic. W	1.112***	0.807**	0.547	1.461**
	(0.259)	(0.256)	(0.440)	(0.689)
Def. H./Vic. W.	1.292***	2.259***	1.580	1.859
	(0.372)	(0.544)	(1.217)	(1.074)
Def. M./Vic. F.	1.064***	1.116***	1.496	0.156
	(0.185)	(0.180)	(0.384)	(0.424)
Def. F./Vic. M.	-0.189	0.119	-0.353	-2.383
	(0.331)	(0.295)	(0.893)	(1.779)
Def. HS Grad	0.696***	0.279	-0.500	-0.597
	(0.167)	(0.179)	(0.349)	(0.578)
Def. Employed	0.458***	0.450***	-0.242	-0.171
	(0.114)	(0.125)	(0.402)	(0.479)
Def. SES	0.091	0.232	-0.620	0.614
	(0.202)	(0.238)	(0.328)	(0.404)
Def. Married	-0.236	-0.115	0.429	-1.122
	(0.141)	(0.190)	(0.406)	(0.635)
Def. has Children	0.283	0.348	-0.140	2.695*
	(0.193)	(0.232)	(0.343)	(1.141)
Def. Drug Use	0.475***	0.380**	0.506	-0.063
	(0.119)	(0.138)	(0.369)	(0.445)
Vic. Stranger	0.380**	0.508***	0.513	0.630
	(0.130)	(0.144)	(0.630)	(0.643)
Contemp. Fel.	-0.076	-0.933***	0.088	1.968*
	(0.174)	(0.197)	(0.795)	(0.994)
Prior Fel.	-0.239	-0.663**	-0.139	-0.388
	(0.205)	(0.236)	(0.546)	(0.534)
Vio. Prior	1.344***	0.801**	0.226	-1.058
	(0.232)	(0.283)	(0.694)	(0.937)

# Appendix E: Supplemental Analyses (Pooled Estimates)

### Appendix E (cont.)

# Stat. Aggs.			0.407	-0.077	
Constant	-4.156*** (0.370)	-2.197*** (0.381)	(0.233) -1.959* (0.980)	(0.201) -4.592** (1.540)	
Year Dummies	Y	Y	Y	Y	
Sel. Bias Corrected	N/A	N/A	N	N	
Ν	2134	1092	212	164	
Pseudo-R <sup>2</sup>	0.325	0.271	0.305	0.411	
AIC	1579.303	1227.310	291.019	210.430	
BIC	1749.276	1377.183	395.073	306.146	

\*p<.05; \*\*p<.01; \*\*\*p<001

Standard errors in parentheses